HILDEBRAND

Advanced Calculus for Applications

FRANCIS B. HILDEBRAND

Associate Professor of Mathematics Massachusetts Institute of Technology

Advanced Calculus for Applications

PRENTICE-HALL, INC.

Englewood Cliffs, New Jersey 1962

PRENTICE-HALL MATHEMATICS SERIES

Dr. Albert A. Bennett, Editor

PRENTICE-HALL INTERNATIONAL, INC. London • Tokyo • Sydney • Paris PRENTICE-HALL OF CANADA, LTD. PRENTICE-HALL DE MEXICO, S. A.

A revision of Advanced Calculus for Engineers by F. B. Hildebrand

© 1948, 1949, 1962 by PRENTICE-HALL, Inc., Englewood Cliffs, New Jersey

All rights reserved. No part of this book may be reproduced in any form, by mimeograph or any other means, without permission in writing from the publisher. Printed in the United States of America. Library of Congress Catalog Card Number 62–17425 01116-C

Preface

The purpose of this text is to present an integrated treatment of a number of those topics in mathematics which can be made to depend only upon a sound course in elementary calculus, and which are of common importance in many fields of application.

An attempt is made to deal with the various topics in such a way that a student who may not proceed into the more profound areas of mathematics may still obtain an intelligent working knowledge of a substantial number of useful mathematical methods, together with an appropriate awareness of the foundations, interrelations, and limitations of these methods. At the same time, it is hoped that a student who is to progress, say, into a rigorous course in mathematical analysis will be provided, in addition, with increased incentive and motivation in that direction, as, for example, when he is confronted by the phrase "It can be shown" within the derivation of a useful concrete result, or when he is led to sense that a certain new concept is a fertile one and is deserving of being expanded and made more precise.

The book is a revision of Advanced Calculus for Engineers, published in 1949, incorporating not only a number of minor changes for the purpose of increased clarity or precision, but also some added textual material, as well as a very substantial number of additional problems.

The first four chapters are concerned chiefly with ordinary differential equations, including analytical, operational, and numerical methods of solution, and with special functions generated as solutions of such equations. In particular, the material of the first chapter can be considered as either a systematic review or an initial introduction to the elementary concepts and techniques, associated with linear equations and with special solvable types of nonlinear equations, which are needed in subsequent chapters. The fifth chapter deals with boundary-value problems governed by ordinary differential equations, with the associated characteristic functions, and with series and integral representations of arbitrary functions in terms of these functions.

Chapter 6 develops the useful ideas and tools of vector analysis; Chapter 7 provides brief introductions to some special topics in higher-dimensional calculus which are rather frequently needed in applications. The treatment here occasionally consists essentially of indicating the plausibility and practical significance of a result and stating conditions under which its validity is rigorously established in listed references.

In Chapter 8, certain basic concepts associated with the simpler types of partial differential equations are introduced, after which, in Chapter 9, full use is made of most of the tools developed in earlier chapters for the purpose of formulating and solving a variety of typical problems governed by the partial differential equations of mathematical physics.

The concluding chapter deals with topics in the theory of analytic functions of a complex variable, including residue calculus, conformal mapping, and applications. Although certain developments in preceding chapters could be made more elegant and more complete if they were made to depend upon this treatment, introduced at an earlier stage, it is felt that, in some cases, the knowledge based on a brief initial study of analytic functions may not be sufficiently firm to support significantly dependent treatments of the other topics, but that such knowledge then may better serve to clarify the other topics when subsequently provided. However, since most of the treatments of Chapter 10, as well as most of those of Chapters 6 and 7, are independent of the content of preceding chapters, material from these chapters can indeed be introduced at an earlier stage in a given course, at the discretion of the instructor. It has been considered reasonable to assume knowledge of certain elementary properties of complex numbers in the earlier chapters, even though the solution of the equation $x^4 + 1 = 0$ then may occasion a personal review on the part of the reader.

Extensive sets of problems are included at the end of each chapter, grouped in correspondence with the respective sections with which they are associated. In addition to more-or-less routine exercises, there are numerous annotated problems which are intended to guide the reader in developing results or techniques which extend or complement treatments in the text, or in dealing with a particularly challenging application. Such problems may serve as focal points for extended discussions or for the introduction of additional (or alternative) material into a chapter, permitting the text to serve somewhat more flexibly in courses of varied types. Answers to all problems are either incorporated into the statement of the problem or listed at the end of the book.

The author is particularly indebted to Professor E. Reissner for valuable collaboration in the preliminary stages of the preparation of the original edition and for many ideas which contributed to whatever useful novelty some of the treatments may possess, to Professor G. B. Thomas for helpful suggestions and criticisms, and to Miss Ruth Goodwin for assistance in the preparation of the original manuscript.

F. B. HILDEBRAND

Contents

CHAPTER 1 Ordinary Differential Equations

1.1 Introduction, 1. 1.2 Linear dependence, 3. 1.3 Complete solutions of linear equations, 4. 1.4 The linear differential equation of first order, 7. 1.5 Linear differential equations with constant coefficients, 8. 1.6 The equidimensional linear differential equation, 13. 1.7 Properties of linear operators, 16. 1.8 Simultaneous linear differential equations, 19. 1.9 Particular solutions by variation of parameters, 25. 1.10 Reduction of order, 29. 1.11 Determination of constants, 32. 1.12 Special solvable types of nonlinear equations, 33.

CHAPTER 2 The Laplace Transform

2.1 An introductory example, 51. 2.2 Definition and existence of Laplace transforms, 53. 2.3 Properties of Laplace transforms, 56. 2.4 The inverse transform, 60. 2.5 The convolution, 62. 2.6 Singularity functions, 63. 2.7 Use of table of transforms, 65. 2.8 Applications to linear differential equations with constant coefficients, 68. 2.9 The Gamma function, 73.

CHAPTER 3 Numerical Methods for Solving Ordinary Differential Equations

3.1 Introduction, 93. 3.2 Use of Taylor series, 94. 3.3 The Adams method, 96. 3.4 The modified Adams method, 100. 3.5 The Runge-Kutta method, 102. 3.6 51

93

CHAPTER 3 Numerical Methods for Solving Ordinary Differential Equations—Continued

Picard's method, 106. 3.7 Extrapolation with differences, 108.

CHAPTER 4 Series Solutions of Differential Equations: Special Functions

4.1 Properties of power series, 119. 4.2 Illustrative examples, 123. 4.3 Singular points of linear second-order differential equations, 127. 4.4 The method of Frobenius, 129. 4.5 Treatment of exceptional cases, 135. 4.6 Example of an exceptional case, 138. 4.7 A particular class of equations, 140. 4.8 Bessel functions, 142. 4.9 Properties of Bessel functions, 150. 4.10 Differential equations satisfied by Bessel functions, 155. 4.11 Ber and bei functions, 156. 4.12 Legendre functions, 159. 4.13 The hypergeometric function, 165. 4.14 Series solutions valid for large values of x, 167.

CHAPTER 5 Boundary-Value Problems and Characteristic-Function Representations

5.1 Introduction, 187. 5.2 The rotating string, 189. 5.3 The rotating shaft, 193. 5.4 Buckling of long columns under axial loads, 198. 5.5 The method of Stodola and Vianello, 200. 5.6 Orthogonality of characteristic functions, 206. 5.7 Expansion of arbitrary functions in series of orthogonal functions, 210. 5.8 Boundary-value problems involving nonhomogeneous differential equations, 213. 5.9 Convergence of the method of Stodola and Vianello, 214. 5.10 Fourier sine series and cosine series, 216. 5.11 Complete Fourier series, 221. 5.12 Fourier-Bessel series, 226. 5.13 Legendre series, 231. 5.14 The Fourier integral, 236.

CHAPTER 6 Vector Analysis

6.1 Elementary properties of vectors, 262. 6.2 The scalar product of two vectors, 265. 6.3 The vector product of two vectors, 266. 6.4 Multiple products, 268. 6.5 Differentiation of vectors, 270. 6.6. Geometry of a space curve, 271. 6.7 The gradient vector, 275. 6.8 The vector operator ∇ , 277. 6.9 Differentiation formulas, 278. 6.10 Line integrals, 281. 6.11 The potential function, 284.

119

262

CHAPTER 6 Vector Analysis—Continued

6.12 Surface integrals, 287. 6.13 The divergence theorem, 290. 6.14 Green's theorem, 293. 6.15 Interpretation of curl. Laplace's equation, 294. 6.16 Stokes's theorem, 295. 6.17 Orthogonal curvilinear coordinates, 298. 6.18 Special coordinate systems, 303. 6.19 Application to twodimensional incompressible fluid flow, 305. 6.20 Compressible ideal fluid flow, 309.

CHAPTER 7 Topics in Higher-Dimensional Calculus

7.1 Partial differentiation. Chain rules, 335. 7.2 Implicit functions. Jacobian determinants, 340. 7.3 Functional dependence, 344. 7.4 Jacobians and curvilinear coordinates. Change of variables in integrals, 346. 7.5 Taylor series, 348. 7.6 Maxima and minima, 351. 7.7 Constraints and Lagrange multipliers, 352. 7.8 Calculus of variations, 355. 7.9 Differentiation of integrals involving a parameter, 359. 7.10 Newton's iterative method, 362.

CHAPTER 8 Partial Differential Equations

8.1 Definitions and examples, 376. 8.2 The quasi-linear equation of first order, 379. 8.3 Special devices. Initial conditions, 384. 8.4 Linear and quasi-linear equations of second order, 388. 8.5 The homogeneous linear equation of second order, with constant coefficients, 389. 8.6 Other linear equations, 392. 8.7 Characteristics of linear firstorder equations, 394. 8.8 Characteristics of linear firstorder equations, 400. 8.9 Singular curves on integral surfaces, 406. 8.10 Remarks on linear second-order initial-value problems, 409. 8.11 The characteristics of a particular quasi-linear problem, 410.

CHAPTER 9 Solutions of Partial Differential Equations of Mathematical Physics

9.1 Introduction, 426. 9.2 Heat flow, 428. 9.3 Steadystate temperature distribution in a rectangular plate, 430. 9.4 Steady-state temperature distribution in a circular annulus, 433. 9.5 Poisson's integral, 437. 9.6 Axisymmetrical temperature distribution in a solid sphere, 439. 9.7 Temperature distribution in a rectangular parallelepiped, 441. 9.8 Ideal fluid flow about a sphere, 443. 9.9 Vibration of 335

376

CHAPTER 9 Solutions of Partial Differential Equations of Mathematical Physics—Continued

a circular membrane, 446. 9.10 Heat flow in a rod, 449. 9.11 The superposition integral, 451. 9.12 Traveling waves, 454. 9.13 The pulsating cylinder, 458. 9.14 Examples of the use of Fourier integrals, 460. 9.15. Application of the Laplace transform to the telegraph equations for a long line, 464. 9.16 Formulation of problems, 469. 9.17 Supersonic flow of ideal compressible fluid past an obstacle. 475.

CHAPTER 10 Functions of a Complex Variable

10.1 Introduction. The complex variable, 509. 10.2 Elementary functions of a complex variable, 511. 10.3 Other elementary functions, 514. 10.4 Analytic functions of a complex variable, 521. 10.5 Line integrals of complex functions, 523. 10.6 Cauchy's integral formula, 528. 10.7. Taylor series, 528. 10.8 Laurent series, 531. 10.9 Singularities of analytic functions, 535. 10.10 Singularities at infinity, 542. 10.11 Significance of singularities, 545. 10.12 Residues, 547. 10.13 Evaluation of real definite integrals, 551. 10.14 Theorems on limiting contours, 555. 10.15 Indented contours, 558. 10.16 Conformal mapping, 562. 10.17 Application to two-dimensional fluid flow, 566. 10.18 Basic flows, 568. 10.19 Other applications of conformal mapping, 572. 10.20 The Schwarz-Christoffel transformation, 574.

Appendix A	617
Appendix B	619
Answers to Problems	621
Index	637

Ordinary Differential Equations

1.1. Introduction. A differential equation is an equation relating two or more variables in terms of derivatives or differentials. Thus, the simplest differential equation is of the form

$$\frac{dy}{dx} = f(x),\tag{1}$$

where f(x) is a given function of the independent variable x. The solution is obtained immediately by integration, in the form

$$y = \int^{x} f(x) \, dx + C, \qquad (2)$$

where a convenient lower limit is assumed in the integral, and C is an arbitrary constant. Whether or not it happens that the integral can be expressed in terms of simple functions is incidental, in the sense that we define a solution of a differential equation to be any functional relation, not involving derivatives or integrals of unknown functions, which implies the differential equation. Similarly, in an equation of the form

$$F(x) G(y) dx + f(x) g(y) dy = 0, (3)$$

we may separate the variables and obtain a solution by integration in the form

$$\int^{x} \frac{F(x)}{f(x)} dx + \int^{y} \frac{g(y)}{G(y)} dy = C.$$
 (4)

Usually we desire to obtain the *most general* solution of the differential equation; that is, we require *all* functional relations which imply the equation. In the general case it is often difficult to determine when *all* such relations have indeed been obtained. Fortunately, however, this difficulty does not exist in the case of so-called *linear* differential equations, which are of most frequent occurrence in applications and which are to be of principal interest in what follows.

A differential equation of the form

$$a_0(x)\frac{d^n y}{dx^n} + a_1(x)\frac{d^{n-1} y}{dx^{n-1}} + \dots + a_{n-1}(x)\frac{dy}{dx} + a_n(x) y = f(x)$$
 (5)

is said to be a *linear* differential equation of order n. The distinguishing characteristic of such an equation is the absence of products or nonlinear functions of the *dependent variable* (unknown function) y and its derivatives, the highest derivative present being of order n. The coefficients $a_0(x), \ldots, a_n(x)$ may be arbitrarily specified functions of the independent variable x.

For a linear equation of the *first* order,

$$a_0(x)\frac{dy}{dx} + a_1(x) y = f(x),$$

it is shown in Section 1.4 that if both sides of the equation are multiplied by a certain determinable function of x (an "integrating factor"), the equation can always be put in the simpler form

$$\frac{d}{dx}\left[p(x)\ y\right] = F(x),$$

where p(x) and F(x) are simply expressible in terms of a_0 , a_1 , and f, and hence can then be solved directly by integration.

Although no such simple general method exists for solving linear equations of higher order, there are two types of such equations which are of particular importance in applications and which can be completely solved by direct methods. These two cases are considered in Sections 1.5 and 1.6. In addition, this chapter presents certain techniques that are available for treatment of more general linear equations.

Many of the basically useful properties of linear differential equations do not hold for *nonlinear* equations, such as

$$\frac{dy}{dx} = x + y^2, \qquad \frac{d^2y}{dx^2} + \sin y = 0, \qquad \frac{d^2y}{dx^2} + x\left(\frac{dy}{dx}\right)^2 + y = e^x.$$

A few special types of solvable nonlinear equations are dealt with briefly in Section 1.12.

The equations to be considered in this chapter are known as ordinary differential equations, as distinguished from *partial* differential equations, which involve *partial derivatives* with respect to two or more independent variables. Equations of the latter type are treated in subsequent chapters.

Before proceeding to the study of linear ordinary differential equations, we next briefly introduce the notion of *linear dependence*, which is basic in this work.

1.2. Linear dependence. By a linear combination of n functions $u_1(x)$, $u_2(x), \ldots, u_n(x)$ is meant an expression of the form

$$c_1 u_1(x) + c_2 u_2(x) + \dots + c_n u_n(x) \equiv \sum_{k=1}^n c_k u_k(x),$$
 (6)

where the c's are constants. When at least one c is not zero, the linear combination is termed *nontrivial*. The functions u_1, u_2, \ldots, u_n are then said to be *linearly independent* over a given interval (say $a \le x \le b$) if over that interval no one of the functions can be expressed as a linear combination of the others, or, equivalently, if no *nontrivial* linear combination of the functions are said to be *linearly dependent*.

As an example, the functions $\cos 2x$, $\cos^2 x$, and 1 are linearly dependent over *any* interval because of the identity

$$\cos 2x - 2\cos^2 x + 1 \equiv 0.$$

It follows from the definition that *two* functions are linearly dependent over an interval if and only if one function is a constant multiple of the other over that interval. The necessity of the specification of the interval in the general case is illustrated by a consideration of the two functions x and |x|. In the interval x > 0 there follows $x - |x| \equiv 0$, whereas in the interval x < 0we have $x + |x| \equiv 0$. Thus the two functions are linearly dependent over any continuous interval not including the point x = 0; but they are linearly independent over any interval including x = 0, since no single linear combination of the two functions is identically zero over such an interval.

Although in practice the linear dependence or independence of a set of functions generally can be established by inspection, the following result is of some importance in theoretical discussions. We assume that each of a set of n functions u_1, u_2, \ldots, u_n possesses n finite derivatives at all points of an interval I. Then, if a set of constants exists such that

$$c_1u_1+c_2u_2+\cdots+c_nu_n=0$$

for all values of x in I, these same constants also satisfy the identities

$$c_{1}\frac{du_{1}}{dx} + c_{2}\frac{du_{2}}{dx} + \dots + c_{n}\frac{du_{n}}{dx} = 0,$$

$$c_{1}\frac{d^{2}u_{1}}{dx^{2}} + c_{2}\frac{d^{2}u_{2}}{dx^{2}} + \dots + c_{n}\frac{d^{2}u_{n}}{dx^{2}} = 0,$$

$$\dots$$

$$c_{1}\frac{d^{n-1}u_{1}}{dx^{n-1}} + c_{2}\frac{d^{n-1}u_{2}}{dx^{n-1}} + \dots + c_{n}\frac{d^{n-1}u_{n}}{dx^{n-1}} = 0.$$

Ł

Thus the *n* constants must satisfy *n* homogeneous linear equations. However, such a set of equations can possess *nontrivial* solutions only if its coefficient determinant vanishes. Thus it follows that if the functions u_1, u_2, \ldots, u_n are linearly dependent in an interval I, then the determinant

$$W(u_{1}, u_{2}, \dots, u_{n}) = \begin{vmatrix} u_{1} & u_{2} & \cdots & u_{n} \\ \frac{du_{1}}{dx} & \frac{du_{2}}{dx} & \cdots & \frac{du_{n}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{2}}{dx} & \cdots & \frac{du_{n}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} \\ \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} & \frac{du_{1}}{dx} \\ \frac{du_{1}}{dx} & \frac{$$

vanishes identically in I. This determinant appears frequently in theoretical work and is called the Wronskian (or Wronskian determinant) of the functions. Thus we see that if the Wronskian of u_1, u_2, \ldots, u_n is not identically zero in I, then the functions are linearly independent in I.

To illustrate, since the value of the determinant

$$W(1, x, x^{2}, ..., x^{n}) = \begin{vmatrix} 1 & x & x^{2} & x^{3} & \cdots & x^{n} \\ 0 & 1! & 2x & 3x^{2} & \cdots & nx^{n-1} \\ 0 & 0 & 2! & 6x & \cdots & n(n-1)x^{n-2} \\ 0 & 0 & 0 & 3! & \cdots & n(n-1)(n-2)x^{n-3} \\ \cdots & \cdots & \cdots & \cdots & n! \end{vmatrix}$$

is merely the product of the nonvanishing constants appearing in the principal diagonal and hence cannot vanish, it follows that the functions appearing in the first row are linearly independent (over *any* interval).

Unfortunately, the converse of the preceding theorem is *not* true. That is, the vanishing of the Wronskian is *necessary* but *not sufficient* for linear dependence of a set of functions. For an example establishing the insufficiency, see Problem 7.

1.3. Complete solutions of linear equations. The most general linear differential equation of the *n*th order can be written in the form

$$\frac{d^n y}{dx^n} + a_1(x) \frac{d^{n-1} y}{dx^{n-1}} + \cdots + a_{n-1}(x) \frac{dy}{dx} + a_n(x) y = h(x).$$
(8)

Here it is assumed that both sides of the equation have been divided by the coefficient of the highest derivative. We will speak of this form as the standard

form of the equation. This equation is frequently written in the abbreviated form

$$L y = h(x), \tag{9}$$

where L here represents the linear differential operator

$$L = \frac{d^n}{dx^n} + a_1(x)\frac{d^{n-1}}{dx^{n-1}} + \dots + a_{n-1}(x)\frac{d}{dx} + a_n(x).$$
(10)

The problem of solving Equation (8) consists of determining the most general expression for y which, if substituted into the left-hand side of (8), or if operated on by (10), gives the prescribed right-hand side h(x). When a relationship of the form y = u(x) implies Equation (8), it is conventional to say that either the relation y = u(x) or the function u(x) is a solution of that equation.

If all the coefficients $a_1(x), \ldots, a_n(x)$ were zero, the solution of Equation (8) would be accomplished directly by *n* successive integrations, each integration introducing an independent constant of integration. Thus it might be expected that the general solution of (8) also would contain *n* independent arbitrary constants. As a matter of fact, it is known that in any interval I in which the coefficients are continuous, there exists a continuous solution to Equation (8) involving exactly *n* independent arbitrary constants; furthermore, there are no solutions of Equation (8) valid in I which cannot be obtained by specializing the constants in any such solution.

It should be noticed that this is a property peculiar to *linear* differential equations. To illustrate, the nonlinear differential equation

$$\left(\frac{dy}{dx}\right)^2 - 2\frac{dy}{dx} + 4y = 4x - 1 \tag{11}$$

is of first order. A solution containing one arbitrary constant is of the form

$$y = x - (x - c)^2,$$
 (12)

as can be verified by direct substitution. However, this is not the most general solution, since the function y = x also satisfies the differential equation but cannot be obtained by specializing the arbitrary constant in the solution given. The additional solution y = x is called a *singular solution*. Such solutions can occur only in the solution of nonlinear differential equations.

We consider first the result of replacing the function h(x) by zero in Equation (8). The resulting differential equation, L y = 0, is said to be homogeneous, since each term in the equation then involves the first power of y or of one of its derivatives. In this case, from the linearity of the equation, it is easily seen that any linear combination of individual solutions is also a solution. Thus, if n linearly independent solutions $u_1(x), u_2(x), \ldots, u_n(x)$ of the associated homogeneous equation

$$L y_H = 0 \tag{13}$$

are known, the general solution of Equation (13) is of the form

$$y_H(x) = c_1 u_1(x) + c_2 u_2(x) + \dots + c_n u_n(x) = \sum_{k=1}^n c_k u_k(x),$$
 (14)

where the c's are the *n* required arbitrary constants. That is, all solutions of the homogeneous equation associated with (8) are obtained by suitably specializing the constants in Equation (14).

In this connection, it should be explained that we refer to a function as a *solution* of a differential equation in a given interval *I* if and only if that function satisfies the differential equation at *all points* of *I*. Thus, in the case of the homogeneous equation

$$\frac{d^2y}{dx^2}=0$$

we say that the general solution is of the form $y = c_1 + c_2 x$. It may be argued that since the function y = |x| is a linear function in any interval not including the point x = 0, its second derivative is zero and hence it is a "solution" which cannot be obtained by specializing c_1 and c_2 . However, it is clear that the first derivative of this function does not exist at x = 0, and hence the second derivative also does not exist at that point. Consequently, since the left-hand side of the equation does not exist at x = 0, the equation is not satisfied at this point, and y = |x| cannot be said to be a solution in any interval including x = 0. In any interval not including x = 0, the function y = |x| may be replaced by either +x or -x and hence is obtained from the general solution by setting $c_1 = 0$ and either $c_2 = 1$ or $c_2 = -1$.

Now suppose that one *particular solution* of Equation (8), say $y = y_P(x)$, can be obtained by inspection or otherwise, so that

$$L y_P = h(x). \tag{15}$$

Then the *complete solution* of Equation (8) is of the form

$$y = y_{H}(x) + y_{P}(x) = \sum_{k=1}^{n} c_{k} u_{k}(x) + y_{P}(x), \qquad (16)$$

since this expression contains n independent arbitrary constants and satisfies the differential equation

$$L y = L(y_H + y_P) = L y_H + L y_P = h(x).$$
(17)

Thus it is seen that the process of solving an ordinary linear differential equation can be conveniently divided into two parts. First, n linearly independent solutions of the associated homogeneous equation may be obtained;

then, if any one particular solution of the complete equation is found, the complete solution is given by Equation (16).

It is frequently convenient to say, " $y_H(x)$ is a homogeneous solution of L y = h," in abbreviation of the statement, " $y = y_H(x)$ is a solution of the associated homogeneous equation L y = 0." The term "complementary solution" also is used.

It will be shown in Section 1.9 that if the general homogeneous solution of an *n*th order linear equation is known, a particular solution can always be obtained by n integrations. In Sections 1.5 and 1.6 we consider important special cases in which the homogeneous solution is readily obtained.

1.4. The linear differential equation of first order. The linear equation of first order is readily solved in general terms, without determining separately homogeneous and particular solutions. For this purpose, we attempt to determine an integrating factor p(x) such that the standard form

$$\frac{dy}{dx} + a_1(x) y = h(x) \tag{18}$$

is equivalent to the equation

$$\frac{d}{dx}(py) = ph. \tag{19}$$

Since Equation (19) can be written in the form

$$\frac{dy}{dx} + \left(\frac{1}{p}\frac{dp}{dx}\right)y = h(x),$$

it follows that Equations (18) and (19) are equivalent if p satisfies the equation

$$\frac{1}{p}\frac{dp}{dx}=a_1(x),$$

and hence an integrating factor is

$$p = e^{\int a_1(x) \, dx}.\tag{20}$$

The solution of Equation (19) is obtained by integration:

$$py=\int ph\,dx+C,$$

so that the general solution of (18) is of the form

$$y = \frac{1}{p} \int ph \, dx + \frac{C}{p},\tag{21}$$

where p is the integrating factor defined by Equation (20), and C is an arbitrary constant.

Example 1. To solve the differential equation

$$x\frac{dy}{dx}+(1-x)y=xe^{x},$$

we first rewrite the equation in the standard form,

$$\frac{dy}{dx} + \left(\frac{1}{x} - 1\right)y = e^x.$$

An integrating factor is then

$$p = e^{\int \left(\frac{1}{x}-1\right) dx} = e^{\log x - x} = x e^{-x},$$

no constant being added in the integration, since only a particular integrating factor is needed. The solution is then given by Equation (21),

or

$$y = \frac{e^x}{x} \int x \, dx + C \frac{e^x}{x},$$

$$y = \frac{x}{2} e^x + C \frac{e^x}{x}.$$

It may be noticed that the general homogeneous solution of Equation (18) is $y_H = \frac{C}{p}$, whereas a particular solution is $y_P = \frac{1}{p} \int ph \, dx$.

1.5. Linear differential equations with constant coefficients. The simplest and perhaps the most important differential equation of higher order is the linear equation,

$$L y = \frac{d^{n}y}{dx^{n}} + a_{1}\frac{d^{n-1}y}{dx^{n-1}} + \dots + a_{n-1}\frac{dy}{dx} + a_{n}y = h(x), \qquad (22)$$

in which the coefficients a_k are constants.

We first attempt to determine n linearly independent solutions of the corresponding homogeneous equation. The appearance of the equation suggests homogeneous solutions of the form e^{rx} , where r is a constant, since all derivatives of e^{rx} are constant multiples of the function itself,

$$\frac{d^m}{dx^m}\,e^{rx}=\,r^m\,e^{rx}$$

We then have

 $L e^{rx} = (r^n + a_1 r^{n-1} + \dots + a_{n-1} r + a_n) e^{rx}.$ (23)

This result shows that e^{rx} is a solution of the homogeneous equation associated with Equation (22) if r is one of the *n* roots r_1, r_2, \ldots, r_n of the *characteristic* equation,

$$r^{n} + a_{1}r^{n-1} + \cdots + a_{n-1}r + a_{n} = 0.$$
 (24)

It should be noticed that this equation is obtained from the associated homogeneous differential equation by formally replacing $\frac{d^k y}{dx^k}$ by r^k , with the convention that $\frac{d^0 y}{dx^0} \equiv y$. If the *n* roots of Equation (24) are distinct, exactly *n* independent solutions $e^{r_1 x}, \ldots, e^{r_n x}$ of the homogeneous equation are so obtained and the general homogeneous solution is

$$y_H = \sum_{k=1}^n c_k e^{r_k \alpha}.$$
 (25)

However, if one or more of the roots is repeated, less than *n* independent solutions are obtained in this way. To find the missing solutions we may proceed as follows. Suppose that $r = r_1$ is a double root of Equation (24). Then Equation (23) is of the form

$$L e^{rx} = (r - r_1)^2 (r - r_3) \cdots (r - r_n) e^{rx}$$

and it follows that not only the right-hand member itself but also its (partial) derivative with respect to r must vanish when $r = r_1$. The same must then be true for the left-hand member. Thus we conclude that in this case we have both

$$L[e^{rx}]_{r=r_1} = Le^{r_1x} = 0$$
$$L\left[\frac{\partial}{\partial r}(e^{rx})\right]_{r=r_1} = Lxe^{r_1x} = 0,$$

and

so that the part of the homogeneous solution corresponding to a double root r_1 can be written in the form

$$c_1 e^{r_1 x} + c_2 x e^{r_1 x} = e^{r_1 x} (c_1 + c_2 x).$$

By a simple extension of this argument, it can be shown that the part of the homogeneous solution corresponding to an *m*-fold root r_1 is of the form

$$e^{r_1x}(c_1+c_2x+c_3x^2+\cdots+c_mx^{m-1})$$

Hence, to each of the n roots of Equation (24), repeated roots being counted separately, there is a corresponding known homogeneous solution, and the general homogeneous solution is determined as a linear combination of these n solutions.

Example 2. For the equation

$$\frac{d^3y}{dx^3}-\frac{dy}{dx}=0,$$

the characteristic equation is $r^3 - r = r(r + 1)(r - 1) = 0$, from which there follows $r = 0, \pm 1$. The general solution is then

$$y = c_1 + c_2 e^x + c_3 e^{-x}.$$

Example 3. For the differential equation

$$\frac{d^3y}{dx^3} - 5 \frac{d^2y}{dx^2} + 8 \frac{dy}{dx} - 4y = 0,$$

the characteristic equation is $(r - 1)(r - 2)^2 = 0$, from which there follows r = 1, 2, 2. The general solution is then

$$y = c_1 e^x + e^{2x} (c_2 + c_3 x)$$

If Equation (24) has *imaginary roots* and if the coefficients of Equation (24) are real, the roots must occur in conjugate pairs. Thus, if $r_1 = a + ib$ is one root, a second root must be $r_2 = a - ib$. The part of the solution corresponding to these two roots can be written in the form

$$A e^{(a+ib)x} + B e^{(a-ib)x} = e^{ax}(A e^{ibx} + B e^{-ibx}).$$

In order that this expression be real, the constants A and B must be imaginary By making use of *Euler's formula*,*

$$e^{i\theta} = \cos\theta + i\sin\theta, \qquad (26)$$

we find that the solution becomes

$$e^{ax}[A(\cos bx + i \sin bx) + B(\cos bx - i \sin bx)]$$

and hence can be written in the more convenient form,

$$e^{ax}(c_1\cos bx + c_2\sin bx),$$

where c_1 and c_2 are new arbitrary constants replacing (A + B) and i(A - B), respectively. Thus real values of c_1 and c_2 correspond to values of A and B which are conjugate complex. Similarly, if $a \pm ib$ are *m*-fold roots, the corresponding 2m terms in the homogeneous solution can be written in the real form,

$$e^{ax}[(c_1 + c_2x + \cdots + c_mx^{m-1})\cos bx + (c_{m+1} + c_{m+2}x + \cdots + c_{2m}x^{m-1})\sin bx].$$

Example 4. The equation

$$\frac{d^2y}{dx^2}+2\frac{dy}{dx}+5y=0$$

has the characteristic equation $r^2 + 2r + 5 = 0$, from which $r = -1 \pm 2i$; hence

$$y = e^{-x}(c_1 \cos 2x + c_2 \sin 2x).$$

* Familiarity with this important relation, and with the elementary algebra of complex numbers, is assumed. Such topics are reviewed in the preliminary sections of the last chapter.

Example 5. The equation

$$\frac{d^4y}{dx^4}+2\frac{d^2y}{dx^2}+y=0$$

has the characteristic equation $(r^2 + 1)^2 = 0$, from which $r = \pm i, \pm i$; hence

$$y = (c_1 + c_2 x) \cos x + (c_3 + c_4 x) \sin x.$$

General methods for obtaining a particular solution of the complete nonhomogeneous Equation (22) are given in Sections 1.7 and 1.9. A shorter method which can be applied in many practical cases is that of *undetermined coefficients*. This method may be used when the right-hand side of Equation (22) involves only terms of the form x^m , where *m* is an integer, terms of the form $\sin qx$, $\cos qx$, and e^{px} , and/or products of two or more such functions. The reason for the success of the method is the fact that each of these functions, or any product of a finite number of these functions, has only a finite number of linearly independent derivatives.

If we define the *family* of a function f(x) as the set of linearly independent functions of which the function f(x) and its derivatives with respect to x are linear combinations, the following families may be listed:

Term	Family	
x^m	$x^{m}, x^{m-1}, x^{m-2}, \ldots, x^{2}, x, 1$	
$\sin qx$	sin qx, cos qx	
$\cos qx$	sin qx, cos qx	
e^{px}	e^{px}	

The family of a function consisting of a product of *n* terms of this type is readily seen to consist of all possible products of *n* factors, in which one factor in each product is taken from the family of each factor in the parent function. Thus, it may be verified that the family of $x^2 \sin 3x$ is composed of two-factor products of terms in the families $\{x^2, x, 1\}$ and $\{\sin 3x, \cos 3x\}$, one term from each family appearing in each product:

$${x^2 \sin 3x, x \sin 3x, \sin 3x, x^2 \cos 3x, x \cos 3x, \cos 3x}$$
.

The method of undetermined coefficients may now be outlined as follows. It is assumed that the general homogeneous solution of the differential equation has already been obtained, and that any cosh or sinh functions occurring in it, or in the right-hand member h(x), are replaced by equivalent linear combinations of exponential functions.

(1) Construct the family of each term (or product) of which h(x) is a linear combination.

(2) If any family has a member which is a homogeneous solution of the differential equation, replace *that* family by a new family in which *each* member

of the original family is multiplied by x, or by the lowest integral power of x for which no member of the new family is a homogeneous solution. Only members of the offending family are so modified. It should also be noticed, for example, that the presence of e^x or sin x in the homogeneous solution does not require modification of a family containing the *product* $e^x \sin x$ unless that product itself is also a homogeneous solution.

(3) Assume as a particular solution a linear combination of all members of the resultant families, with undetermined literal coefficients of combination, and determine these coefficients by requiring that the differential equation be identically satisfied by this assumed solution.

It will be found that in all cases the number of coefficients to be determined will equal the number of linearly independent functions whose coefficients must be matched, and that the resultant equations always have a solution. The detailed proof of this general statement is rather lengthy and is omitted. A special case is treated, for the purpose of illustration, at the end of Section 1.7.

It should be emphasized that this procedure does not generally apply unless the differential equation has constant coefficients and has a right-hand member possessing a finite family.

Example 6. Consider the differential equation

$$\frac{d^3y}{dx^3}-\frac{dy}{dx}=2x+1-4\cos x+2e^x.$$

The general homogeneous solution is

$$y_H = c_1 + c_2 e^x + c_3 e^{-x}.$$

The families of the terms x, 1, $\cos x$, and e^x on the right-hand side of the equation are, respectively,

$$\{x, 1\}, \{1\}, \{\cos x, \sin x\}, \{e^x\}$$

The second family is contained in the first, and is discarded. Since the first family has the representative 1 in the homogeneous solution, it is replaced by the family $\{x^2, x\}$. Similarly, the last family is replaced by $\{x e^x\}$. A particular solution is then assumed in the form

$$y_P = Ax^2 + Bx + C\cos x + D\sin x + Ex e^x.$$

When y is replaced by y_P , the differential equation becomes

$$-2Ax - B - 2D\cos x + 2C\sin x + 2Ee^x = 2x + 1 - 4\cos x + 2e^x$$

By equating the coefficients of x, 1, $\cos x$, $\sin x$, and e^x , there follows

$$A = -1, B = -1, D = 2, C = 0, E = 1.$$

A particular solution thus is

$$y_P = -x^2 - x + 2\sin x + x e^x$$

and the general solution is

$$y = c_1 + c_2 e^x + c_3 e^{-x} - x^2 - x + 2 \sin x + x e^x.$$

1.6. The equidimensional linear differential equation. An equation of the form

$$Ly = x^{n} \frac{d^{n}y}{dx^{n}} + b_{1}x^{n-1} \frac{d^{n-1}y}{dx^{n-1}} + \dots + b_{n-1}x \frac{dy}{dx} + b_{n}y = h(x), \quad (27)$$

where the b's are constants, has the property that each term on the left is unchanged when x is replaced by cx, where c is a nonzero constant. Thus the physical dimension of x is irrevelant in each term on the left and, if the b's are dimensionless, each term on the left has the dimensions of y. For this reason we shall refer to this equation as the equidimensional linear equation. The equation is also variously called "Euler's equation," "Cauchy's equation," and the "homogeneous linear equation," although each of these terms also has other connotations.

One method of solving this equation consists of introducing a new independent variable z by the substitution

$$x = e^z, \qquad z = \log x. \tag{28}$$

There then follows

$$\frac{d}{dx}=\frac{dz}{dx}\frac{d}{dz}=\frac{1}{e^z}\frac{d}{dz},$$

and hence

$$x^m \frac{d^m}{dx^m} = e^{mz} \left(\frac{1}{e^z} \frac{d}{dz}\right)^m.$$

Thus, in particular, we obtain

$$x \frac{dy}{dx} = \frac{dy}{dz},$$

$$x^2 \frac{d^2y}{dx^2} = \frac{d^2y}{dz^2} - \frac{dy}{dz} = \frac{d}{dz} \left(\frac{d}{dz} - 1\right) y,$$

$$x^3 \frac{d^3y}{dx^3} = \frac{d^3y}{dz^3} - 3 \frac{d^2y}{dz^2} + 2 \frac{dy}{dz}$$

$$= \frac{d}{dz} \left(\frac{d}{dz} - 1\right) \left(\frac{d}{dz} - 2\right) y,$$

and, in general, it is found that

$$x^{m}\frac{d^{m}y}{dx^{m}} = \frac{d}{dz}\left(\frac{d}{dz}-1\right)\left(\frac{d}{dz}-2\right)\cdots\left(\frac{d}{dz}-m+1\right)y.$$
 (29)

The transformed equation thus becomes linear with constant coefficients, and y then can be determined in terms of z by the methods of the preceding section

if the new right-hand member is zero or if it has a finite family (with respect to z-differentiation). The final result is obtained by replacing z by log x.

Example 7. To solve the differential equation

$$x^{2}\frac{d^{2}y}{dx^{2}}-2x\frac{dy}{dx}+2y=x^{2}+2,$$

we make use of Equations (28) and (29) to obtain the transformed equation

$$\frac{d^2y}{dz^2} - 3\frac{dy}{dz} + 2y = e^{2z} + 2.$$

The solution is found, by the methods of Section 1.5, to be

$$y = c_1 e^z + c_2 e^{2z} + z e^{2z} + 1,$$

or, returning to the variable x,

$$y = c_1 x + c_2 x^2 + x^2 \log x + 1.$$

If the right-hand member is zero, there is a more convenient alternative procedure which consists of directly assuming a homogeneous solution of the form

$$y_H = x^r$$
,

corresponding to the assumption $y_H = e^{rz}$ in the transformed equation. By making use of the relationship

$$x^m \frac{d^m x^r}{dx^m} = r(r-1)\cdots(r-m+1) x^r,$$

there follows, with the notation of Equation (27),

$$L x^{r} = \{ [r(r-1)\cdots(r-n+1)] + b_{1}[r(r-1)\cdots(r-n+2)] + \cdots + b_{n-1}r + b_{n} \} x^{r}.$$

Hence x^r is a homogeneous solution if r satisfies the characteristic equation

$$[r(r-1)\cdots(r-n+1)] + b_1[r(r-1)\cdots(r-n+2)] + \cdots + b_{n-1}r + b_n = 0.$$
(30)

This equation can be obtained from the left-hand side of Equation (27) by formally replacing $x^m \frac{d^m y}{dx^m}$ by the *m*-factor product

$$r(r-1)\cdots(r-m+1).$$

Let the *n* roots of Equation (30) be denoted by r_1, r_2, \ldots, r_n . If these roots are distinct, the general homogeneous solution is of the form

$$y_{II} = \sum_{k=1}^{n} c_k x^{r_k}.$$
 (31)

In analogy with the results of the preceding section, we find that the second homogeneous solution corresponding to a double root r_1 is

$$\left[\frac{\partial}{\partial r}\left(x^{r}\right)\right]_{r=r_{1}}=x^{r_{1}}\log x$$

and the part of the homogeneous solution corresponding to an *m*-fold root r_1 is

$$x^{r_1}[c_1 + c_2 \log x + c_3 (\log x)^2 + \cdots + c_m (\log x)^{m-1}].$$

Further, to a conjugate pair of imaginary roots $r = a \pm ib$ there corresponds the solution

$$x^{a}[c_{1}\cos(b\log x) + c_{2}\sin(b\log x)].$$

The extension to the case of repeated imaginary roots is obvious.

Except in those cases in which the right-hand member is a linear combination of powers of x (and in certain other cases of little practical interest), particular solutions of nonhomogeneous equations of type (27) usually cannot be obtained by the method of undetermined coefficients. However, it is readily shown by using the substitution (28) that a particular solution corresponding to a right-hand member of the form x^s is given by $y_P = Ax^s$, where A is a constant to be determined by substitution, unless x^s is a homogeneous solution. If x^s is a homogeneous solution, the trial particular solution should be of the form $y_P = Ax^s (\log x)^k$, where k is the smallest positive integer for which this expression is not a homogeneous solution. In other cases, particular solutions can be obtained by the method of Section 1.9.

Example 8. For the equation

$$x^2\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + 2y = x^2 + 2,$$

of Example 7, the characteristic equation (30) becomes

$$r(r-1) - 2r + 2 = r^2 - 3r + 2 = 0,$$

from which r = 1, 2. The homogeneous solution is thus

$$y_{II}=c_1x+c_2x^2.$$

Since x^2 is a homogeneous solution, we assume a solution $y_P = Ax^2 \log x$ corresponding to the right-hand term x^2 and a solution $y_P = B$ corresponding to the constant term, and hence write

$$y_P = Ax^2 \log x + B.$$

Substitution into the given differential equation gives A = B = 1, and the complete solution is

$$y = c_1 x + c_2 x^2 + x^2 \log x + 1.$$

1.7. Properties of linear operators. We now consider more critically certain properties of linear differential operators of the general form

$$L = a_0(x) \frac{d^n}{dx^n} + a_1(x) \frac{d^{n-1}}{dx^{n-1}} + \dots + a_{n-1}(x) \frac{d}{dx} + a_n(x).$$
(32)

An expression of this sort has no intrinsic meaning by itself, but when it is followed by a function u(x), the result L u is defined to be a new function of x defined by the relationship

$$L u \equiv \left(a_0 \frac{d^n}{dx^n} + a_1 \frac{d^{n-1}}{dx^{n-1}} + \cdots + a_{n-1} \frac{d}{dx} + a_n\right) u$$
$$\equiv a_0 \frac{d^n u}{dx^n} + a_1 \frac{d^{n-1} u}{dx^{n-1}} + \cdots + a_{n-1} \frac{du}{dx} + a_n u.$$

We speak of L u as the result of operating on u by the operator L.

Further, if L_1 and L_2 are two linear operators, we write L_2L_1u to indicate the operation $L_2(L_1u)$, that is, the result of operating on L_1u by L_2 , and similarly for three or more successive operations. The abbreviations $L^2 u = L L u$, $L^3 u = L L L u$, and so on are frequently used. In particular, if the operator $\frac{d}{dx}$ is written as D,

$$D = \frac{d}{dx}, \qquad (33)$$

there follows

$$D^2 = \frac{d}{dx}\left(\frac{d}{dx}\right) = \frac{d^2}{dx^2}, \qquad D^3 = \frac{d}{dx}\left(\frac{d^2}{dx^2}\right) = \frac{d^3}{dx^3}, \qquad \dots,$$

and in general $D^m = \frac{d^m}{dx^m}$. Thus Equation (32) can be written in the equivalent

form

$$L = a_0(x) D^n + a_1(x) D^{n-1} + \dots + a_{n-1}(x) D + a_n(x)$$

= $\sum_{k=0}^n a_{n-k}(x) D^k.$ (34)

The operations L_2L_1u and L_1L_2u should be carefully distinguished from each other, since the two operations are not, in general, equivalent. To illustrate, let $L_1 = \frac{d}{dx}$ and $L_2 = x \frac{d}{dx}$. Then $L_2 L_1 u = x \frac{d}{dx} \left(\frac{du}{dx} \right) = x \frac{d^2 u}{dx^2},$ $L_1 L_2 u = \frac{d}{dx} \left(x \frac{du}{dx} \right) = x \frac{d^2 u}{dx^2} + \frac{du}{dx}.$ and

If the order in which the operators L_1 and L_2 are applied is immaterial, that is, if $L_1L_2u = L_2L_1u$, the two operators are said to be *commutative*. Similarly, we say that a *set* of operators is commutative if each pair of operators in the set is commutative.

It is clear that any two operators of the form D^m and D^n are commutative; so also are two operators of the form $a_m D^m$ and $a_n D^n$, where a_m and a_n are constant. From this fact it follows easily that the set of linear operators with constant coefficients is commutative.

The commutativity of two linear equidimensional operators, for which $a_k(x) = b_k x^k$, is seen to depend upon the commutativity of any two operators of the form $L_1 = b_m x^m \frac{d^m}{dx^m}$ and $L_2 = b_n x^n \frac{d^n}{dx^n}$. But with the substitution (28), Equation (29) shows that L_1 and L_2 become linear operators with constant coefficients, and hence are commutative. Thus it follows that the set of equidimensional linear operators is commutative.

It may be remarked, however, that commutativity is the exception rather than the rule. Thus, for example, the above illustration shows that linear operators with constant coefficients and homogeneous linear operators are not in general commutative with each other.

The distributive property of linear operators,

$$(c_1L_1 + c_2L_2 + \dots + c_nL_n)u = c_1L_1u + c_2L_2u + \dots + c_nL_nu, \quad (35)$$

as well as the distributive property of linear operations,

$$L(c_1u_1 + c_2u_2 + \dots + c_nu_n) = c_1L u_1 + c_2L u_2 + \dots + c_nL u_n, \quad (36)$$

of which use has already been made, is easily established.

In many cases it is possible to *factor* a linear operator into the product of n linear factors. If the factors are commutative, the result of factoring is unique, the *order* in which the factors are written being arbitrary. Otherwise, the component factors will differ in form according to the position they occupy in the product.

To illustrate, the operator $D^2 - 3D + 2$ can be factored uniquely in the forms (D-2)(D-1) = (D-1)(D-2). However, the operator $xD^2 + D$ factors in two ways, into the products (D)(xD) and (xD + 1)(D). By this statement we mean, of course, that

$$(D)(xD) u = (xD + 1)(D) u = (xD^2 + D) u$$

for any twice-differentiable u.

Use is frequently made of the factoring process in solving linear differential equations. Thus, in the case of the homogeneous linear equation with constant coefficients, the operator

$$L = D^n + a_1 D^{n-1} + \cdots + a_{n-1} D + a_n$$

can be factored uniquely into the linear factors,

$$L = (D - r_1)(D - r_2) \cdots (D - r_n),$$

where r_1, r_2, \ldots, r_n are roots of the formal equation L = 0. Thus the equation $L y_H = 0$ becomes

$$(D-r_1)(D-r_2)\cdots(D-r_n) y_H=0,$$

where each operator operates on the expression to its right. Hence the complete expression will be zero if the result of the first operation is zero. Since any one of the n operators can be written immediately before y, it follows that a solution of any one of the n equations

$$(D-r_k) y_H = 0$$
 $(k = 1, 2, ..., n)$

is a solution of $L y_H = 0$. But these equations are equivalent to

$$\frac{dy_H}{dx} - r_k y_H = 0 \qquad (k = 1, 2, \ldots, n)$$

and are readily solved to give the solutions

$$y_H = c_k e^{r_k x}$$
 $(k = 1, 2, ..., n).$

By superimposing these solutions, the general homogeneous solution is obtained in the case where the roots are distinct, in accordance with the results of Section 1.5. The part of the solution corresponding to *m*-fold roots can be obtained as the solution of $(D - r_1)^m y_H = 0$.

Analogous procedures can be applied in other cases. In particular, the general solution of any linear differential equation with constant coefficients and *arbitrary* right-hand side can be obtained by a method illustrated by the following example.

Example 9. To solve the differential equation

$$\frac{d^2y}{dx^2}-3\frac{dy}{dx}+2y=f(x),$$

we write the equation in the operational form

$$(D-2)(D-1)y = f(x).$$

Writing next

$$y_1 = (D - 1)y_2$$

the differential equation becomes

$$(D-2)y_1 = f(x)$$
 or $\frac{dy_1}{dx} - 2y_1 = f(x)$.

This linear equation is of first order and is solved, by using the results of Section 1.4, in the form

$$y_1 = e^{2x} \int e^{-2x} f(x) \, dx + c_1 e^{2x}.$$

Next, replacing y_1 by (D - 1)y, we obtain a second first-order equation,

$$\frac{dy}{dx} - y = e^{2x} \int e^{-2x} f(x) \, dx + c_1 e^{2x},$$

with solution

$$y = e^x \int e^x \left[\int e^{-2x} f(x) dx \right] dx + c_1 e^{2x} + c_2 e^x.$$

Similarly, it can be shown that the linear equidimensional operator of Equation (27) can be factored into the commutative factors

$$(x D - r_1)(x D - r_2) \cdots (x D - r_n),$$

where r_1, r_2, \ldots, r_n are the roots of the characteristic equation (30). In this connection it should be noticed that the operators $x^m \frac{d^m}{dx^m}$ and $\left(x\frac{d}{dx}\right)^m$ are not equivalent.

The notion of operators is useful in establishing the general validity of the method of *undetermined coefficients* described in Section 1.5. To illustrate the argument, we here consider an equation of the form $L y = a \cos qx$, where L is a linear differential operator with constant coefficients. Since the operator $D^2 + q^2$ annihilates the right-hand member, it follows that all solutions of the given equation are included in the general solution of the equation $(D^2 + q^2)L y = 0$. However, if $D^2 + q^2$ is not a factor of L, then the general solution of this equation is

$$y = y_H + A\cos qx + B\sin qx,$$

where $y = y_H$ is the general solution of L y = 0. If $D^2 + q^2$ is an unrepeated factor of L, then y_H will contain $\cos qx$ and $\sin qx$, so that $A \cos qx + B \sin qx$ then must be replaced by $Ax \cos qx + Bx \sin qx$, and so forth, in accordance with the rules set down in Section 1.5.

1.8. Simultaneous linear differential equations. Frequently two or more unknown functions are related to a single independent variable by an equal number of linear differential equations. Thus, in the case of two unknown functions x and y and the independent variable t, we may have a pair of simultaneous equations of the form

$$L_{1}x + L_{2}y = h_{1}(t) L_{3}x + L_{4}y = h_{2}(t)$$
(37a,b)

where the L's are linear differential operators in t. The unknown functions x and y are to be determined as functions of t.

When the operators involved are commutative, all unknown functions except one can be successively eliminated from the given set of equations, to give a new set of linear differential equations each involving only one unknown function. We illustrate this procedure in the case of Equations (37a,b). If Equation (37a) is operated on by L_4 and (37b) by $-L_2$, and if the resultant equations are added, there follows

$$(L_4L_1 - L_2L_3)x + (L_4L_2 - L_2L_4)y = L_4h_1 - L_2h_2,$$

or, if L_2 and L_4 are commutative,

$$(L_4L_1 - L_2L_3)x = L_4h_1 - L_2h_2.$$
(38a)

Similarly, to eliminate x we operate on Equation (37a) by $-L_3$ and on (37b) by L_1 . If L_1 and L_3 are also commutative, we then obtain, by addition,

$$(L_1L_4 - L_3L_2)y = L_1h_2 - L_3h_1.$$

Finally, if L_1, L_4 and L_2, L_3 are commutative, the operators of x and y in the last equations are identical, and we have

$$(L_4L_1 - L_2L_3)y = L_1h_2 - L_3h_1.$$
(38b)

Equations (38a) and (38b) can be written formally in the determinantal form

$$\Delta x = \begin{vmatrix} h_1 & L_2 \\ h_2 & L_4 \end{vmatrix}, \quad \Delta y = \begin{vmatrix} L_1 & h_1 \\ L_3 & h_2 \end{vmatrix}, \quad (39a,b)$$

where Δ is the operator

$$\Delta = \begin{vmatrix} L_1 & L_2 \\ L_3 & L_4 \end{vmatrix},\tag{40}$$

if it is understood that in each term of the expansion of the right-hand sides of Equations (39a,b) the operator is to be written before the function operated upon. The formal analogy with *Cramer's rule* for solving linear equations by determinants should be noticed.

Since the same operator affects x and y in Equations (39a,b), it is seen that the homogeneous solutions of these equations are linear combinations of the same functions, the number n of independent constants in each linear combination being equal to the degree of the operator Δ . Thus, the solutions of Equations (39a,b) will contain 2n independent constants.

At this stage of the solution a certain amount of care must be taken. It is clear that since Equations (37a,b) imply (39a,b), all solutions of the original simultaneous equations are contained in the solutions of the final equations. However, since differentiation is generally involved in obtaining (39) from (37), the converse is not generally true, in the sense that the solutions of (39a,b) may satisfy (37a,b) only if certain relationships exist among the 2n constants. These relationships may be determined by substituting the solutions of (39) into (37a,b) and requiring that the resultant equations be identities. However, if the coefficients are *constants*, and if in one of Equations (37a,b) the two operators involved have no common factors, the relationships are completely determined by substitution into that single equation (see Problem 25). If x_P and y_I , satisfy Equations (37), only the added x_H and y_H need be so checked. An alternate procedure consists of solving only one of Equations (39a,b) for one unknown function and of then substituting this result into whichever of (37a,b) is more convenient for the subsequent determination of the second unknown function. The expressions so obtained are then introduced into the remaining one of Equations (37a,b), to determine possible restrictions on the arbitrary constants.

The extension of this procedure to cases in which more than two unknown functions are present leads to results again completely analogous to the statement of Cramer's rule. Thus, assuming that all operators involved are commutative with each other, the solutions of the equations

$$L_{1}x + L_{2}y + L_{3}z = h_{1}(t),$$

$$L_{4}x + L_{5}y + L_{6}z = h_{2}(t),$$

$$L_{7}x + L_{8}y + L_{9}z = h_{3}(t)$$

are also solutions of three linear differential equations each involving only one dependent variable, one of which can be written formally as

$$\Delta y = \begin{vmatrix} L_1 & h_1 & L_3 \\ L_4 & h_2 & L_6 \\ L_7 & h_3 & L_9 \end{vmatrix},$$
$$\Delta = \begin{vmatrix} L_1 & L_2 & L_3 \\ L_4 & L_5 & L_6 \\ L_7 & L_8 & L_9 \end{vmatrix},$$

where

if in each term of the expansion of the first determinant the function is written after the operators. If the operator Δ is of order *n*, the solutions of the three equations so obtained involve 3n arbitrary constants, and possible restriction on these constants must be obtained by substitution into the *original* equations. This procedure may be quite laborious if several unknown functions are present. A method of solving such sets of equations with reduced labor in certain problems, in cases when the operators have constant coefficients, is given in Chapter 2.

It may be stated that if all operators involved are commutative, the total number of independent constants present in the solution of a set of linear differential equations is equal to the order of the operator Δ . This order cannot exceed the sum of the orders of the several equations and in certain cases may be less than this number. To illustrate the preceding method, we solve the equations

$$\frac{d^{2}x}{dt^{2}} - x - 2y = t$$

$$\frac{d^{2}y}{dt^{2}} - 2y - 3x = 1$$
(41a,b)

In operational form, these equations become

$$(D^{2} - 1)x - 2y = t$$

-3x + (D^{2} - 2)y = 1

Equation (40) gives

$$\Delta = \begin{vmatrix} (D^2 - 1) & -2 \\ -3 & (D^2 - 2) \end{vmatrix} = D^4 - 3D^2 - 4,$$

and Equations (39a,b) then become

$$(D^4 - 3D^2 - 4)x = \begin{vmatrix} t & -2 \\ 1 & (D^2 - 2) \end{vmatrix} = (D^2 - 2)t + 2 = 2 - 2t$$
 (42a)

and

$$(D^4 - 3D^2 - 4)y = \begin{vmatrix} (D^2 - 1) & t \\ -3 & 1 \end{vmatrix} = (D^2 - 1)1 + 3t = 3t - 1.$$
 (42b)

We notice that the characteristic equation for both x and y is obtained by formally replacing D by r in the expression for $\Delta = 0$,

$$r^4 - 3r^2 - 4 = 0,$$

from which $r = \pm 2, \pm i$. Hence we obtain

$$x_H = c_1 e^{2t} + c_2 e^{-2t} + c_3 \cos t + c_4 \sin t,$$

$$y_H = d_1 e^{2t} + d_2 e^{-2t} + d_3 \cos t + d_4 \sin t.$$

Particular solutions of Equations (42a,b) are readily found by inspection or by the method of undetermined coefficients,

$$x_P = \frac{1}{2}t - \frac{1}{2}, \quad y_P = \frac{1}{4} - \frac{3}{4}t.$$

To determine the relationships which must exist among the c's and d's, we may first verify that x_P and y_P satisfy Equations (41a,b) as well as (42a,b). Hence we then introduce the expressions for x_H and y_H into the left-hand sides of Equations (41a,b) and require the results to be identically zero. Using first Equation (41a), we find the conditions

$$d_1 = \frac{3}{2}c_1, \quad d_2 = \frac{3}{2}c_2, \quad d_3 = -c_3, \quad d_4 = -c_4.$$

The same conditions are obtained by using Equation (41b) (see Problem 25). Thus only four of the eight constants are truly arbitrary. Retaining the four c's, we write

$$\begin{aligned} x_H &= c_1 e^{2t} + c_2 e^{-2t} + c_3 \cos t + c_4 \sin t, \\ y_H &= \frac{3}{2} \left(c_1 e^{2t} + c_2 e^{-2t} \right) - \left(c_3 \cos t + c_4 \sin t \right), \end{aligned}$$

and the final solutions are

$$x = c_1 e^{2t} + c_2 e^{-2t} + c_3 \cos t + c_4 \sin t + \frac{1}{2}t - \frac{1}{2},$$

$$y = \frac{3}{2} (c_1 e^{2t} + c_2 e^{-2t}) - (c_3 \cos t + c_4 \sin t) - \frac{3}{4}t + \frac{1}{4}.$$

It may be seen that Equation (42a) could be obtained directly, in this case, by solving (41a) for y and substituting the result into (41b). In more complicated cases the present procedure is generally preferable.

If the expression for x_H were introduced into the left-hand side of Equation (41a) and the right-hand side were replaced by zero, an expression for y_H would be found directly, in this case, in terms of the constants of x_H . Substitution of these results into the left-hand side of Equation (41b) would then show that no further restrictions on the constants were necessary. If Equation (41b) were used to determine y_H in terms of x_H , the two new constants introduced would be determined in terms of the c's by substitution into (41a).

The solutions of Equations (41a,b) can also be obtained by a slightly different but equivalent method which is of some practical interest. Since the coefficients in both linear equations are constants, it can be assumed initially that homogeneous solutions exist of the form

$$x_H = c_k e^{r_k t}, \qquad y_H = d_k e^{r_k t}.$$

By introducing these assumptions into Equations (41a,b) and replacing the right-hand sides by zeros, there follows

$$(r_k^2 - 1)c_k - 2d_k = 0,$$

 $-3c_k + (r_k^2 - 2)d_k = 0.$

In order that nontrivial solutions of these equations exist, it is necessary that the determinant of the coefficients of c_k and d_k vanish, giving the characteristic equation obtained previously. If r_k satisfies this equation, the coefficient d_k can be expressed in terms of c_k by either of the two equations. Thus, using the first equation, we have

$$d_k=\frac{r_k^2-1}{2}c_k.$$

For the roots $r = \pm 2$ there follows $d_k = \frac{3}{2}c_k$, whereas for $r = \pm i$ there follows $d_k = -c_k$. These results are easily shown to lead, by superposition, to the previously obtained homogeneous solutions.

In order to obtain particular solutions directly from Equations (41a,b), the method of undetermined coefficients can be applied if all terms on the right-hand sides of the equations are taken into account in constructing the families. Thus, from Equation (41a) we have the family $\{t, 1\}$ and from (41b) the family $\{1\}$, which is contained in the former family. Since there is no representative in either homogeneous solution, we assume particular solutions of the form

$$x_P = At + B, \qquad y_P = Ct + D,$$

Substitution into Equations (41a,b) gives

$$-(A + 2C)t - (B + 2D) = t,$$

$$-(3A + 2C)t - (3B + 2D) = 1.$$

In order that these be identities, we must have

-A - 2C = 1, B + 2D = 0, 3A + 2C = 0, -3B - 2D = 1

from which there follows

$$A = \frac{1}{2}, \quad B = -\frac{1}{2}, \quad C = -\frac{3}{4}, \quad D = \frac{1}{4},$$

in accordance with the previously obtained results.

In order to illustrate a special situation, we consider also the set

$$2\frac{dx}{dt} - 3x + y = 4e^t$$

$$x + 2\frac{dy}{dt} - 3y = 0$$

$$\left. \qquad (43)$$

If Equations (39) and (40) are applied to these equations, there follows

$$\frac{d^{2}x}{dt^{2}} - 3\frac{dx}{dt} + 2x = -e^{t} \\ \frac{d^{2}y}{dt^{2}} - 3\frac{dy}{dt} + 2y = -e^{t} \end{bmatrix},$$
(44)

from which one obtains

$$\begin{array}{l} x_{H} = c_{1}e^{2t} + c_{2}e^{t} \\ y_{H} = d_{1}e^{2t} + d_{2}e^{t} \end{array} \right)$$

$$(45)$$

and

$$\begin{aligned} x_P &= t \ e^t \\ y_P &= t \ e^t \end{aligned} \right\}.$$
 (46)

But here it happens that x_P and y_P do not satisfy the original Equations (43). Thus it is necessary to substitute the sums $x_H + x_P$ and $y_H + y_P$ into Equations (43) for the purpose of obtaining conditions on the constants in Equations (45). This process gives, finally,

$$d_2 = 2 + c_2, \quad d_1 = -c_1.$$
 (47)

1.9. Particular solutions by variation of parameters. We next derive a method for determining the complete solution of any linear differential equation for which the general homogeneous solution is known.

Suppose that the general homogeneous solution of the equation

$$Ly = \frac{d^{n}y}{dx^{n}} + a_{1}(x)\frac{d^{n-1}y}{dx^{n-1}} + \dots + a_{n-1}(x)\frac{dy}{dx} + a_{n}(x) \ y = h(x)$$
(48)

has been obtained in the form

$$y_H = \sum_{k=1}^n c_k u_k(x),$$
 (49)

where the u's are n linearly independent homogeneous solutions and the c's are n arbitrary constants or "parameters." We will find that a particular solution of the complete equation can be obtained by replacing the constant parameters c_k in the solution of the associated homogeneous equation by certain functions of x. Thus, we assume that

$$y_P = \sum_{k=1}^{n} C_k(x) \, u_k(x) \tag{50}$$

is a solution of Equation (48) and attempt to choose the *n* functions C_k suitably. Since we have *n* functions to determine, and since the requirement that Equation (50) satisfy Equation (48) represents only one condition, we have n - 1 additional conditions at our disposal.

Differentiating Equation (50) and using primes to denote differentiation with respect to x, we obtain

$$\frac{dy_P}{dx} = \sum_{k=1}^n C_k u'_k + \sum_{k=1}^n C'_k u_k.$$

In order to simplify this expression, we require as our first condition that the second summation vanish,

$$\sum_{k=1}^{n} C'_{k} u_{k} = 0.$$
 (51a)

There then follows

$$\frac{dy_P}{dx} = \sum_{k=1}^n C_k u'_k$$
$$\frac{d^2 y_P}{dx^2} = \sum_{k=1}^n C_k u''_k + \sum_{k=1}^n C'_k u'_k.$$

and
As the second condition, we require again that the sum of the terms involving *derivatives* of the C's vanish,

$$\sum_{k=1}^{n} C'_{k} u'_{k} = 0.$$
 (51b)

Proceeding in this way through the (n - 1)th derivative, we have as our (n - 1)th condition the requirement

$$\sum_{k=1}^{n} C'_{k} u_{k}^{(n-2)} = 0$$
 (51c)

and the (n - 1)th derivative is

$$\frac{d^{n-1}y_P}{dx^{n-1}} = \sum_{k=1}^n C_k u_k^{(n-1)}.$$

The expression for the *n*th derivative is then

$$\frac{d^n y_P}{dx^n} = \sum_{k=1}^n C_k u_k^{(n)} + \sum_{k=1}^n C'_k u_k^{(n-1)}.$$

By introducing the expressions for y_P and its derivatives into the left-hand side of Equation (48), we find that the final condition, that Equation (50) satisfy (48), becomes

$$L y_P = \sum_{k=1}^n C_k u_k^{(n)} + a_1(x) \sum_{k=1}^n C_k u_k^{(n-1)} + \cdots + a_{n-1}(x) \sum_{k=1}^n C_k u_k' + a_n(x) \sum_{k=1}^n C_k u_k + \sum_{k=1}^n C_k' u_k^{(n-1)} = h(x).$$

Combining the first summations, we obtain

$$\sum_{k=1}^{n} C_{k} \left[\frac{d^{n} u_{k}}{dx^{n}} + a_{1}(x) \frac{d^{n-1} u_{k}}{dx^{n-1}} + \cdots + a_{n-1}(x) \frac{du_{k}}{dx} + a_{n}(x) u_{k} \right] + \sum_{k=1}^{n} C'_{k} u_{k}^{(n-1)} = h(x).$$

Now, since each function u_k satisfies Equation (48) with h(x) replaced by zero, and since each bracket in the first summation is precisely the result of replacing y in the left-hand side of Equation (48) by a function u_k , the first summation vanishes identically, and the final condition becomes merely

$$\sum_{k=1}^{n} C'_{k} u_{k}^{(n-1)} = h(x).$$
 (51d)

In summary, the *n* conditions imposed on the *n* unknown functions can be written in the expanded form

If this set of equations is solved for C'_1, C'_2, \ldots, C'_n by Cramer's rule, the common-denominator determinant is seen to be the Wronskian of u_1, u_2, \ldots, u_n .*

If the solutions C'_1, C'_2, \ldots, C'_n are integrated and the results are introduced into Equation (50), the result is a particular solution of Equation (48) for any choice of the *n* constants of integration. If the constants are left arbitrary, this procedure yields the complete solution of Equation (48).

It is important to notice that Equation (48) was written in "standard form." If the coefficient of $\frac{d^n y}{dx^n}$ in Equation (48) were $a_0(x)$, the last equation of (52) would be modified by replacing h(x) by $\frac{h(x)}{a_0(x)}$.

In particular, for a second-order linear equation of the form

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x) y = h(x),$$
(53)

there follows

$$y = C_1(x) u_1(x) + C_2(x) u_2(x),$$
 (54a)

$$C_{1}' = \frac{\begin{vmatrix} 0 & u_{2} \\ h & u_{2}' \end{vmatrix}}{\begin{vmatrix} u_{1} & u_{2} \\ u_{1}' & u_{2}' \end{vmatrix}} = -\frac{h(x) u_{2}(x)}{W[u_{1}(x), u_{2}(x)]}$$

and, similarly,

$$C'_{2} = \frac{h(x) u_{1}(x)}{W[u_{1}(x), u_{2}(x)]}.$$

Thus we can write

$$C_{1} = -\int^{x} \frac{h(x)u_{2}(x)}{W[u_{1}(x), u_{2}(x)]} dx + c_{1},$$

$$C_{2} = \int^{x} \frac{h(x)u_{1}(x)}{W[u_{1}(x), u_{2}(x)]} dx + c_{2}$$
(54b)

* It is known that if the coefficients a_1, a_2, \ldots, a_n are continuous in an interval *I*, the indicated derivatives exist in *I*, and furthermore the Wronskian of the linearly independent functions cannot vanish in *I*. Hence a unique solution always exists.

and the introduction of these results into Equation (54a) gives the required solution.

If h(x) is not given explicitly, but the general solution is required for an *arbitrary* h(x), we may combine the result of this substitution into a more compact form if before substituting Equations (54b) into (54a) we replace the dummy variable of integration by a new variable, say ξ , to distinguish it from the current variable x which appears as the limit of the integrals. Substitution of Equations (54b) into (54a) then leads to the result

$$y = \int^{x} \frac{h(\xi) [u_1(\xi) u_2(x) - u_2(\xi) u_1(x)]}{W[u_1(\xi), u_2(\xi)]} d\xi + c_1 u_1(x) + c_2 u_2(x).$$
(55)

Here x is to be held constant in the ξ integration. If h(x) is given explicitly, the direct evaluation of Equations (54b) and subsequent substitution into (54a) is usually more convenient than the use of Equation (55).

Example 10. For the differential equation

$$\frac{d^2y}{dx^2} + y = f(x)$$

two linearly independent homogeneous solutions are $u_1 = \cos x$, $\dot{u_2} = \sin x$. The Wronskian is

$$W(\cos x, \sin x) = \begin{vmatrix} \cos x & \sin x \\ -\sin x & \cos x \end{vmatrix} = \sin^2 x + \cos^2 x = 1.$$

Thus, use of Equations (54a,b) gives the solution

$$y = -\cos x \left[\int_{-\infty}^{x} f(x) \sin x \, dx + c_1 \right] + \sin x \left[\int_{-\infty}^{x} f(x) \cos x \, dx + c_2 \right].$$

This form is usually most convenient for actual evaluation of the solution when f(x) is given. The form (55), which is useful in more general considerations, here takes the form

 $y = \int^{x} f(\xi) \left[\cos \xi \sin x - \sin \xi \cos x \right] d\xi + c_1 \cos x + c_2 \sin x$ $y = \int^{x} f(\xi) \sin (x - \xi) d\xi + c_1 \cos x + c_2 \sin x.$

Example 11. For the differential equation

$$\frac{d^3y}{dx^3}-3\frac{d^2y}{dx^2}+2\frac{dy}{dx}=f(x),$$

we may take $u_1 = 1$, $u_2 = e^x$, $u_3 = e^{2x}$. Equations (52) then become

$$C'_{1} + C'_{2}e^{x} + C'_{3}e^{2x} = 0,$$

$$C'_{2}e^{x} + 2C'_{3}e^{2x} = 0,$$

$$C'_{2}e^{x} + 4C'_{3}e^{2x} = f(x).$$

For the determinant of this system we find $W(1, e^x, e^{2x}) = 2e^{3x}$. Solving the three simultaneous equations, we obtain

$$C'_1 = \frac{1}{2}f(x), \qquad C'_2 = -e^{-x}f(x), \qquad C'_3 = \frac{1}{2}e^{-2x}f(x).$$

The solution of the differential equation is then

$$y = 1 \left[\frac{1}{2} \int^{x} f(\xi) \, d\xi + c_1 \right] + e^x \left[- \int^{x} e^{-\xi} f(\xi) \, d\xi + c_2 \right] \\ + e^{2x} \left[\frac{1}{2} \int^{x} e^{-2\xi} f(\xi) \, d\xi + c_3 \right]$$

or, equivalently,

$$y = \frac{1}{2} \int^{x} f(\xi) \left[1 - 2e^{x-\xi} + e^{2(x-\xi)} \right] d\xi + c_1 + c_2 e^x + c_3 e^{2x}.$$

It will be shown in Section 1.10 that the Wronskian of two homogeneous solutions of Equation (53) is of the form

$$W(u_1, u_2) = A \ e^{-\int a_1(x) \, dx},\tag{56}$$

where A is a definite constant depending only on the choice of the arbitrary multiplicative constants involved in the homogeneous solutions u_1 and u_2 . It follows (see Problem 36) that $W(u_1, u_2)$ can be determined if only the values of u_1 and u_2 and their first derivatives are known at a single point. This fact is useful in evaluating Equation (55) if, for example, the solutions $u_1(x)$ and $u_2(x)$ are expressed in terms of power series (see Chapter 4).

It can be shown* that, more generally, the Wronskian of n homogeneous solutions of Equation (48) is also given by the right-hand member of Equation (56). The statement of this fact is known as *Abel's formula*. From the properties of the exponential function, it follows that if $a_1(x)$ is continuous in an interval I, the Wronskian cannot vanish in I unless it vanishes *identically*.

1.10. Reduction of order. One of the important properties of linear differential equations is the fact that if one homogeneous solution of an equation of order n is known, a new linear differential equation of order n - 1, determining the remainder of the solution, can be obtained. This procedure is in a sense analogous to the reduction of the degree of an algebraic equation when one solution is known.

Suppose that one homogeneous solution $u_1(x)$ is known. We next write $y = v(x) u_1(x)$ and attempt to determine the function v(x). Substituting vu_1 for y in the left-hand side of the differential equation, we obtain a new linear differential equation of order n to determine v. But since $y = c u_1(x)$ is a homogeneous solution of the original equation, v = c must be a homogeneous solution of the new equation. Hence the new equation must lack the term of zero order in v; that is, the coefficient of v must be zero. Thus the new equation dv

is of order n-1 in the variable $\frac{dv}{dx}$.

^{*} By differentiating the Wronskian determinant, one obtains the relation $\frac{dW}{dx} = -a_1 W$.

We apply this procedure to the solution of the general second-order linear equation

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x) y = h(x),$$
(57)

assuming that one homogeneous solution $u_1(x)$ is known. Writing

$$y = v(x) u_1(x),$$
 (58)

and introducing Equation (58) into (57), there follows

$$v''u_1 + 2v'u_1' + a_1v'u_1 + v(u_1'' + a_1u_1' + a_2u_1) = h.$$

But since u_1 is a homogeneous solution of Equation (57), the expression in parentheses vanishes and the differential equation determining v becomes

$$v''u_{1} + 2v'u_{1}' + a_{1}v'u_{1} = h$$

(v')' + $\left(2\frac{u_{1}'}{u_{1}} + a_{1}\right)v' = \frac{h}{u_{1}}.$ (59)

This equation is of first order in v', with an integrating factor given by the results of Section 1.4 in the form

$$e^{2\log u_1 + \int a_1 dx} = p u_1^2,$$

 $p = e^{\int a_1 dx}.$ (60)

where

Hence there follows

$$v' = \frac{1}{pu_1^2} \int^x phu_1 \, dx \, + \, \frac{c_1}{pu_1^2}$$

An integration then gives

$$v = \int^{x} \frac{\int^{x} phu_{1} dx}{pu_{1}^{2}} dx + c_{1} \int^{x} \frac{dx}{pu_{1}^{2}} + c_{2}$$
(61)

and the introduction of Equation (61) into (58) yields the general solution

$$y = u_1(x) \int^x \frac{\int^x phu_1 dx}{pu_1^2} dx + c_1 u_1(x) \int^x \frac{dx}{pu_1^2} + c_2 u_1(x).$$
 (62)

Thus, if u_1 is one homogeneous solution, the most general linearly independent solution of the associated homogeneous equation is of the form

$$u_2 = A u_1(x) \int_{-\infty}^{\infty} \frac{dx}{pu_1^2} + B u_1(x), \qquad (63)$$

or

where A and B are arbitrary constants with $A \neq 0$, and a particular solution of the complete equation is

$$y = u_1(x) \int^x \frac{\int^x phu_1 \, dx}{pu_1^2} \, dx.$$
 (64)

We remark that no constants of integration need be added in either of the integrations in Equation (64), since the additional terms thereby introduced can be absorbed into the homogeneous solution. However, if arbitrary constants *are* introduced in each of the integrations, Equation (64) represents the complete solution of Equation (57).

In view of Equation (63), the Wronskian of any two homogeneous solutions of (57) is given by

$$W(u_1, u_2) = \begin{vmatrix} u_1 & Au_1 \int^x \frac{dx}{pu_1^2} + Bu_1 \\ u_1' & \frac{A}{pu_1} + Au_1' \int^x \frac{dx}{px_1^2} + Bu_1' \end{vmatrix}$$

Expansion of this determinant gives Abel's result in this case,

$$W(u_1, u_2) = \frac{A}{p(x)} = A e^{-\int a_1(x) dx}$$
(65)

Example 12. One homogeneous solution of the equation

$$x^2\frac{d^2y}{dx^2}+(x-1)\left(x\frac{dy}{dx}-y\right)=x^2e^{-x}$$

is seen by inspection to be y = x. To find the complete solution, we first write the equation in the standard form of (57),

$$\frac{d^2y}{dx^2}+\left(1-\frac{1}{x}\right)\frac{dy}{dx}+\frac{1-x}{x^2}y=e^{-x},$$

and find

$$a_{\mathbf{I}}(x) = 1 - \frac{1}{x}, \qquad p(x) = \frac{1}{x} e^{x}.$$

With $u_1 = x$, a second linearly independent solution is obtained from Equation (63), taking A = 1 and B = 0 for convenience, in the form

$$u_2 = x \int^x \frac{e^{-x}}{x} dx.$$

The lower limit can be chosen at pleasure. This indefinite integral cannot be evaluated in terms of elementary functions. However, the function

$$Ei(-x) = \int_{\infty}^{x} \frac{e^{-x}}{x} dx = \int_{-\infty}^{-x} \frac{e^{x}}{x} dx \qquad (x > 0)$$

is a tabulated function, known as the "exponential-integral function." [See Reference 7 of Chapter 4, for example, where similar integrals involving $(\sin x)/x$ and $(\cos x)/x$, and known as "sine-integral" and "cosine-integral" functions, are also defined and tabulated.] Thus, taking the lower limit to be ∞ , the second independent homogeneous solution can be taken to be

$$u_2 = x Ei(-x),$$

and the general homogeneous solution is of the form

$$y_H = x[c_1 + c_2 Ei(-x)].$$

A particular solution is given by Equation (64),

$$y_P = x \int_{-\infty}^{x} \frac{e^{-x} \int_{-\infty}^{x} e^{-x} dx}{x} dx = x \int_{-\infty}^{x} e^{-x} dx = -x e^{-x},$$

if constants of integration are omitted. Thus the complete solution is

$$y = x[c_1 + c_2 Ei(-x) - e^{-x}]$$
 $(x > 0).$

1.11. Determination of constants. The n arbitrary constants present in the general solution of a linear differential equation of order n are to be determined by n suitably prescribed supplementary conditions.

Frequently these conditions consist of the requirement that the function and its first n - 1 derivatives take on prescribed values at a given point x = a,

$$y(a) = y_0, \quad y'(a) = y'_0, \quad \dots, \quad y^{(n-1)}(a) = y_0^{(n-1)}.$$
 (66)

When such conditions are prescribed, the problem is known as an initial-value problem. In this case it can be shown that if the point x = a is included in an interval where the coefficients $a_1(x), \ldots, a_n(x)$ and the right-hand side h(x) of the differential equation, in standard form, are continuous, there exists a unique solution satisfying Equations (66). Here, if the complete solution is written in the form

$$y = \sum_{k=1}^{n} c_k u_k(x) + y_P(x),$$
 (67)

the conditions of Equations (66) require that the constants c_k satisfy the *n* equations

$$\sum_{k=1}^{n} c_k u_k^{(m)}(a) = y_0^{(m)} - y_P^{(m)}(a) \qquad (m = 0, 1, 2, \dots, n-1).$$
(68)

We notice that the determinant of the coefficients of the constants c_k is the value of the Wronskian of the linearly independent solutions u_k at the point x = a, which (as was stated in the footnote on page 27) cannot vanish under the specified conditions. Thus a unique solution is assured.

Sets of conditions other than those of Equations (66) may, however, be prescribed. For example, values of the function and/or certain of the derivatives may be prescribed at *two* distinct points x = a, x = b. In such cases a unique solution may or may not exist.

Example 13. The general solution of the differential equation

$$\frac{d^2y}{dx^2} + y = 0$$

is $y = c_1 \cos x + c_2 \sin x$. The initial-value problem with conditions $y(0) = y_0$, $y'(0) = y'_0$ has the unique solution $y = y_0 \cos x + y'_0 \sin x$.

The conditions y(0) = 1, $y(\pi/2) = 1$ imply the unique solution $y = \cos x + \sin x$, while the only solution satisfying the conditions $y(0) = y(\pi/2) = 0$ is the trivial solution y = 0. However, the conditions $y(0) = y(\pi) = 0$ are both satisfied if we take $c_1 = 0$, and hence in this case there exist an infinite number of solutions of the form $y = A \sin x$, where A is arbitrary.

1.12. Special solvable types of nonlinear equations. Although there exist no techniques of general applicability for the purpose of obtaining solutions of nonlinear differential equations in closed form, there are several special types of equations for which such solutions can be obtained, a few of which are treated very briefly in this section. Here, instead of seeking y as a function of x, we may be led to determine x as a function of y or to accept as a solution a functional relationship involving the two variables in a less simple way.

(1) Separable equations. Separable first-order equations have already been mentioned in Section 1.1.

Example 14. The equation $(1 + x^2) dy + (1 + y^2) dx = 0$ is separable in the form

$$\frac{dx}{1+x^2} + \frac{dy}{1+y^2} = 0,$$

and integration gives the solution

$$\tan^{-1} x + \tan^{-1} y = \tan^{-1} c$$

 $x + y = c(1 - xy).$

or

Example 15. The equation

$$\left(\frac{dy}{dx}\right)^2 - 4y + 4 = 0$$

yields two separable equations when solved algebraically for $\frac{dy}{dx}$,

$$\frac{dy}{dx}=\pm 2\sqrt{y-1},$$

from which
$$\pm \frac{dy}{\sqrt{y-1}} = 2 \, dx$$

provided that the division by $\sqrt{y-1}$ is legitimate, and hence there follows $\pm \sqrt{y-1} = x - c$ or $y = 1 + (x - c)^2$.

Since the relation y - 1 = 0 has been excluded in the derivation of this solution, the possibility that y = 1 may also be a solution must be explored separately. Direct substitution into the original differential equation shows that y = 1 is indeed a solution. Furthermore, it cannot be obtained by specializing the *constant c* in the relation $y = 1 - (x - c)^2$. Here the complete solution consists of the latter one-parameter solution *together with* the "singular solution," y = 1. It can be verified, in this case, that all the curves which represent particular solutions, in correspondence with particular choices of the constant c, are *tangent* to the straight line representing the singular solution, so that this line is the *envelope* of those curves.

(2) Exact first-order equations. A first-order equation, written in the form

$$P(x,y) \, dx + Q(x,y) \, dy = 0, \tag{69}$$

where P and Q are assumed to have continuous first partial derivatives, is said to be *exact* when P and Q satisfy the condition

$$\frac{\partial P}{\partial y} = \frac{\partial Q}{\partial x}.$$
(70)

In this case, and in this case only, there exists a function u(x,y) such that

$$du = P \, dx + Q \, dy. \tag{71}$$

Thus Equation (69) then is identical with the equation du = 0, whose general solution clearly is

$$u(x,y)=c, \tag{72}$$

where c is an arbitrary constant.

Since Equation (71) implies

$$\frac{\partial u}{\partial x} = P, \qquad \frac{\partial u}{\partial y} = Q,$$
 (73)

the necessity of Equation (70) follows from the fact that

$$\frac{\partial}{\partial x}\left(\frac{\partial u}{\partial y}\right) = \frac{\partial}{\partial y}\left(\frac{\partial u}{\partial x}\right)$$

when the indicated derivatives are continuous. In order to obtain a function u satisfying the two relations of Equations (73), we may, for example, start with

the first relation, integrating with respect to x with y held constant to obtain

$$u(x,y) = \int^{x} P(x,y) \, dx + f(y). \tag{74}$$

Here f(y) is the added "constant of integration," to be determined by the second relation of Equations (73), which gives

$$\int^x \frac{\partial P}{\partial y} \, dx + f'(y) = Q$$

and hence

$$f'(y) = Q - \int^{x} \frac{\partial P}{\partial y} dx.$$
 (75)

That the right-hand member is indeed only a function of y, so that f(y) can be determined (with an irrelevant arbitrary additive constant) by direct integration, follows from the fact that its partial derivative with respect to x is zero since

$$\frac{\partial}{\partial x}\left(Q-\int^x\frac{\partial P}{\partial y}\,dx\right)=\frac{\partial Q}{\partial x}-\frac{\partial P}{\partial y}=0,$$

when Equation (70) is satisfied, so that the *sufficiency* of that condition is also established.

Example 16. The equation

$$\frac{dy}{dx} = \frac{1 + y^2 + 3x^2y}{1 - 2xy - x^3}$$

can be written in the form

$$(3x^2y + y^2 + 1) dx + (x^3 + 2xy - 1) dy = 0$$

and the condition (70) of exactness is satisfied. From the relation

$$\frac{\partial u}{\partial x} = 3x^2y + y^2 + 1$$

there follows $u = x^3y + xy^2 + x + f(y)$. The relation

$$\frac{\partial u}{\partial y} = x^3 + 2xy - 1$$

then gives $x^3 + 2xy + f'(y) = x^2 + 2xy - 1$ or f'(y) = -1, from which f(y) = -y, apart from an irrelevant arbitrary additive constant. Hence the required solution u = c is

$$x^3y + xy^2 + x - y = c.$$

(3) Homogeneous first-order equations. A function f(x,y) is said to be homogeneous of degree n if there exists a constant n such that, for every number λ ,

$$f(\lambda x,\lambda y)=\lambda^n f(x,y).$$

Thus, for example, the functions $x^2 + xy$ and $\tan^{-1}(y/x)$ are both homogeneous, the first of degree two and the second of degree zero, whereas $x^2 + y$ is not homogeneous. The first-order differential equation,

$$P(x,y) dx + Q(x,y) dy = 0,$$
 (76)

is said to be homogeneous if P and Q are both homogeneous of degree n, for some constant n.

Such an equation becomes separable upon the change of variables

$$y = vx, \quad dy = v \, dx + x \, dv. \tag{77}$$

For, since $P(x,vx) = x^n P(1,v)$ and $Q(x,vx) = x^n Q(1,v)$, this substitution reduces Equation (76) to the form

$$x^{n}P(1,v) dx + x^{n}Q(1,v)(v dx + x dv) = 0$$

[P(1,v) + v Q(1,v)] dx + x Q(1,v) dv = 0, (78)

or

which is indeed separable.

The substitution

 $x = uy, \quad dx = u \, dy + y \, du$

also is appropriate and may or may not lead to a more convenient form after a separation of variables.

Example 17. The equation

$$(3y^2 - x^2) \, dx = 2xy \, dy$$

is homogeneous (with n = 2) and, with y = vx, it becomes

 $(v^2 - 1) dx + 2xv dv = 0$

from which there follows

$$\frac{2v \, dv}{v^2 - 1} = \frac{dx}{x} \quad (x \neq 0, v^2 \neq 1)$$

 $v^2 - 1 = cx$

and hence $\log |v^2 - 1| = \log |x| + \log |c| = \log |cx|$

or

or $y^2 - x^2 = cx^3.$

The temporarily excluded relations, x = 0 and $y = \pm x$, both are seen to satisfy the given equation and hence in fact are solutions, but they need not be appended to the solution obtained since they are included in it when $c = \infty$ and c = 0.

(4) *Miscellaneous first-order equations*. Most of the other known techniques for solving special first-order equations in closed form consist of reductions to linear, separable, exact, or homogeneous forms. Some typical examples follow.

Example 18. Bernoulli's equation. The equation

$$\frac{dy}{dx} + P(x)y = Q(x)y^n \qquad (n \neq 1)$$

can be written in the form

$$y^{-n}\frac{dy}{dx}+P(x)y^{-n+1}=Q(x),$$

which clearly becomes *linear* under the substitution $v = y^{-n+1}$.

Example 19. The equation

$$\frac{dy}{dx} = \frac{ax + by + c}{Ax + By + C}$$

can be reduced to a homogeneous equation by writing x = t + h and y = s + kand determining suitable values of the constants h and k, provided that $aB \neq bA$. In that exceptional case, the substitution s = ax + by + c renders the equation separable.

Example 20. Although the equation

$$y^2 \, dx = x(x \, dy - y \, dx)$$

can be treated as a homogeneous equation or as a Bernoulli equation in y (Example 18), the combination $x \, dy - y \, dx$ tends to suggest division by y^2 or by x^2 , for the purpose of forming -d(x/y) or d(y/x). Here a division by y^2 , accompanied by a division by x, is clearly indicated, leading to the form

$$\frac{dx}{x} + \frac{y\,dx - x\,dy}{y^2} = 0$$

which is *exact*, since each term is an exact differential, and there follows

$$\log|x| + \frac{x}{y} = c$$

or

$$y=\frac{x}{c-\log|x|}.$$

In this case, multiplication of the equation by the *integrating factor* $x^{-1}y^{-2}$ makes the equation exact. In less contrived situations, the discovery of such a factor may be much less straightforward.

Example 21. The nonlinear Equation (11),

$$\left(\frac{dy}{dx}\right)^2 - 2\frac{dy}{dx} + 4y = 4x - 1,$$

yields the two first-degree equations

$$\frac{dy}{dx}=1\pm 2\sqrt{x-y}.$$

The prominence of the expression x - y may suggest the substitution

$$x - y = u$$
, $1 - \frac{dy}{dx} = \frac{du}{dx}$,

which leads to the separable equations

$$\frac{du}{dx} = \pm 2\sqrt{u}$$

If it is noticed that the process of separation necessitates the special consideration of the relation u = 0, the one-parameter solution (12) and the singular solution y = x then are easily obtained. Equally fortuitous substitutions may suggest themselves in other cases.

(5) Second-order equations lacking one variable. The general equation of second order is of the form

$$F\left(x, y, \frac{dy}{dx}, \frac{d^2y}{dx^2}\right) = 0, \qquad (79)$$

containing $\frac{d^2y}{dx^2}$ explicitly. Any such equation can be written equivalently as a pair of simultaneous first-order equations, in various ways. In particular, we may write

$$\frac{dy}{dx} = p \tag{80}$$

and

$$\frac{d^2 y}{dx^2} = \begin{cases} \frac{dp}{dx} \\ \frac{dp}{dy} \frac{dy}{dx} = p \frac{dp}{dy} \end{cases}$$
(81)

in accordance with which Equation (79) can be replaced by either the set

$$F\left(x, y, p, \frac{dp}{dx}\right) = 0$$

$$\cdot \qquad \frac{dy}{dx} = p$$
(82)

or the set

$$F\left(x, y, p, p \frac{dp}{dy}\right) = 0$$

$$\frac{dy}{dx} = p$$
(83)

In the general case, both equations of either set involve the *three* variables x, y, and p, and hence no one of the equations can be solved independently of the associated equation. However, if y is not explicitly involved in F, then the first equation of (82) involves only x and p. If it can be solved, to provide a relation between x and p, and if the result can be used to eliminate p from the second relation in Equation (82), then y can be determined in terms of x by integration of the resulting equation.

On the other hand, if x is not explicitly involved in F, then the first equation of (83) involves only y and p. If it can be solved, to provide a relation between y and p, and if the result can be used to eliminate p from the second relation in Equation (83), then a separable first-order equation in y and x results.

In either case, it may be more feasible to obtain x and y in terms of p, with p retained as a parameter, than to eliminate p by using the solution of the first equation of the pair.

Example 22. The equation

$$\frac{d^2y}{dx^2} = x \left(\frac{dy}{dx}\right)^3$$

lacks the variable y. With $\frac{dy}{dx} = p$ and $\frac{d^2y}{dx^2} = \frac{dp}{dx}$, there follows

$$\frac{dp}{dx} = xp^3$$

which separates to give

$$p=\frac{\pm 1}{\sqrt{c_1^2-x^2}}.$$

Hence

$$dy = \pm \frac{dx}{\sqrt{c_1^2 - x^2}},$$

from which there follows

$$y = \pm \sin^{-1} \frac{x}{c_1} + c_2$$
 or $x = c'_1 \sin(y - c_2)$

where $c_1' = \pm c_1$.

Example 23. The equation

$$y\frac{d^2y}{dx^2} = \left(\frac{dy}{dx}\right)^2$$

lacks the variable x. With $\frac{dy}{dx} = p$ and $\frac{d^2y}{dx^2} = p \frac{dp}{dy}$, there follows

$$p\left(y\frac{dp}{dy}-p\right)=0$$

and hence

$$y\frac{dp}{dy}=p$$
 or $p=0$.

The first alternative gives

 $p = c_1 y,$

which includes the second alternative p = 0 as a special case, and hence the general solution is

$$y = c_2 e^{c_1 x}.$$

Example 24. The equation

$$\frac{d^2y}{dx^2} - y = 0$$

lacks the variable x. Whereas it can be solved by the present method, it is *linear* in y and is much more easily solved by the methods of Section 1.5.

Example 25. The equation (

$$\left(\frac{dy}{dx}\right)^2 \frac{d^2y}{dx^2} = 1 + \left(\frac{dy}{dx}\right)^2$$

lacks both x and y. If the absence of y is exploited, there follows

$$p^2\frac{dp}{dx}=1+p^2$$

and hence

$$x=p-\tan^{-1}p+c_1.$$

This relation can be used to eliminate x (rather than p) from the relation

$$dy = p \, dx = p \, \frac{dx}{dp} \, dp,$$

to give

$$dy = p\left(1 - \frac{1}{1+p^2}\right)dp = \frac{p^3}{1+p^2}dp.$$

Hence

$$y = \frac{1}{2}p^2 - \frac{1}{2}\log(p^2 + 1) + c_2.$$

Here the solution provides a parametric representation of x and y. The fact that the parameter p happens to be identifiable with $\frac{dy}{dx}$ may afford a subsequent added convenience.

REFERENCES

- 1. Agnew, R. P., Differential Equations, 2nd ed., McGraw-Hill Book Company, Inc., New York, 1960.
- 2. Coddington, E. A., and N. Levinson, Theory of Ordinary Differential Equations, McGraw-Hill Book Company, Inc., New York, 1955.
- 3. Ince, E. L., Ordinary Differential Equations, 4th ed., Dover Publications, Inc., New York, 1953.

- 4. Martin, W. T., and E. Reissner, *Elementary Differential Equations*, 2nd ed., Addison-Wesley Publishing Co., Inc., Reading, Mass., 1961.
- 5. Piaggio, H. T. H., *Differential Equations*, Bell Mathematical Series, Open Court Publishing Company, La Salle, Ill., 1952.
- 6. Rainville, E. D., Intermediate Differential Equations, John Wiley & Sons, Inc., New York, 1943.

PROBLEMS

Section 1.1

1. Differentiate each of the following relations with respect to x and, by using the result to eliminate the constant c, obtain a first-order differential equation of which the given relation defines a one-parameter solution:

(a)
$$y = ce^{x}$$
,
(b) $y = ce^{x^{2}} + 2x$,
(c) $y = \frac{1}{c - x}$,
(d) $y = e^{cx}$,
(e) $x^{2} + y^{2} = c^{2}$,
(f) $y = cx + c^{2}x^{2}$.

2. By differentiating each of the following relations twice, and eliminating the constants c_1 and c_2 from the three resultant equations, obtain a second-order differential equation of which the given relation defines a two-parameter solution:

(a) $y = c_1 e^x + c_2 e^{-x}$, (b) $y = e^x (c_1 + c_2 x)$, (c) $y = c_1 \cos x + c_2 \sin x$, (d) $y = c_1 e^{c_2 x}$, (e) $y = \frac{c_1 + x}{c_2 + x}$, (f) $(x - c_1)^2 + (y - c_2)^2 = 1$.

3. Obtain the general solution of each of the following differential equations:

(a)
$$\frac{dy}{dx} = 2xy$$
,
(b) $y \, dy + \sqrt{1 - y^2} \, dx = 0$,
(c) $(1 - y^2) \, dx + (1 - x^2) \, dy = 0$,
(d) $\sqrt{1 - x^2} \, y \, dy = \sqrt{1 - y^2} \, x \, dx$.

Section 1.2

4. Prove that e^{r_1x} and e^{r_2x} are linearly independent over any finite interval if $r_1 \neq r_2$.

5. Prove that e^{rx} and $x e^{rx}$ are linearly independent over any finite interval.

6. If $u_1(x)$ and $u_2(x)$ are linearly independent, prove that $A_1u_1(x) + A_2u_2(x)$ and $B_1u_1(x) + B_2u_2(x)$ are also linearly independent if $A_1B_2 - A_2B_1 \neq 0$.

7. By considering the functions $u_1 = x^2$ and $u_2 = x^2 |x|$ over an interval including the origin, show that identical vanishing of the Wronskian of u_1 and u_2 over an interval does not imply linear dependence of u_1 and u_2 over that interval. (Notice that $u_2 = x^3 \operatorname{sgn} x$ and $u'_2 = 3x^2 \operatorname{sgn} x$ for all x, where $\operatorname{sgn} x = +1$ when x > 0, $\operatorname{sgn} x = -1$ when x < 0, and $\operatorname{sgn} 0 = 0$.)

8. Suppose that u_1 and u_2 both satisfy the linear differential equation

$$(py')' + qy = 0$$

where p and q are functions of the independent variable x. Show that there follows

$$u_1(pu'_2)' - u_2(pu'_1)' = 0,$$

and that this equation is equivalent to the equation (pW)' = 0, where W is the Wronskian of u_1 and u_2 . Hence deduce that W = A/p, where A is a constant.

Section 1.3

9. If $y = u_1(x)$ and $y = u_2(x)$ satisfy the homogeneous linear equation

$$\frac{d^n y}{dx^n} + a_1(x)\frac{d^{n-1}y}{dx^{n-1}} + \cdots + a_n(x)y = 0,$$

prove that $y = c_1 u_1(x) + c_2 u_2(x)$ is also a solution, for any constant values of c_1 and c_2 .

10. Verify that $u_1 = 1$ and $u_2 = \log x$ each satisfy the nonlinear equation $y'' + y'^2 = 0$, but that $y = c_1u_1 + c_2u_2$ is not a solution unless either $c_2 = 0$ or $c_2 = 1$.

11. If $y_{P}^{(1)}$ and $y_{P}^{(2)}$ are two linearly independent particular solutions of the nonhomogeneous linear equation

$$\frac{d^2y}{dx^2}+a_1(x)\frac{dy}{dx}+a_2(x)y=h(x),$$

show that the function $u_1 = y_P^{(1)} - y_P^{(2)}$ satisfies the associated homogeneous equation (in which h is replaced by 0).

12. Verify that $y = \log x$ reduces the nonlinear expression $y'' + y'^2$ to zero, and that y = x reduces it to unity. To what does the sum $y = \log x + x$ reduce it?

Section 1.4

13. Solve the following differential equations:

(a)
$$x \frac{dy}{dx} - ky = x^2$$
 (k constant), (b) $\frac{dy}{dx} - y \tan x = x$,
(c) $\frac{dy}{dx} + y \tan x = \sec x$, (d) $\frac{dy}{dx} + (y - 2\sin x)\cos x = 0$,
(e) $(x^2 - 1)\frac{dy}{dx} + 2y = (x + 1)^2$, (f) $x^2 \log x \, dy + (xy - 1) \, dx = 0$,
(g) $2xy\frac{dy}{dx} - y^2 = x^2$, (h) $y \, dx = (x + y^2) \, dy$.

14. Show that the substitution $u = y^{1-n}$ reduces the nonlinear equation

$$\frac{dy}{dx} + a_1(x)y = h(x)y^n \qquad (n \neq 1)$$

to a linear equation. (This equation is often called *Bernoulli's equation*.)

Section 1.5

15. If two roots of the characteristic equation (24) are $r = \pm a$, show that the corresponding part of the homogeneous solution can be written in the form

 $y = c_1 \cosh ax + c_2 \sinh ax,$

where sinh ax and cosh ax are the hyperbolic functions defined by the equations

$$\sinh ax = \frac{e^{ax} - e^{-ax}}{2}$$
, $\cosh ax = \frac{e^{ax} + e^{-ax}}{2}$.

16. If two roots of Equation (24) are $r = a \pm b$, show that the corresponding part of the homogeneous solution can be written in the form

$$y = e^{ax}(c_1 \cosh bx + c_2 \sinh bx)$$

17. Solve the following differential equations:

(a)
$$\frac{d^2y}{dx^2} - \frac{dy}{dx} - 2y = 0$$
, (b) $\frac{d^3y}{dx^3} - \frac{d^2y}{dx^2} - \frac{dy}{dx} + y = 0$,
(c) $\frac{d^2y}{dx^2} - 2\frac{dy}{dx} + 2y = 0$, (d) $\frac{d^4y}{dx^4} - 4\frac{d^3y}{dx^3} + 7\frac{d^2y}{dx^2} - 6\frac{dy}{dx} + 2y = 0$,
(e) $\frac{d^3y}{dx^3} - y = 0$, (f) $\frac{d^2y}{dx^2} - 2iy = 0$ ($i^2 = -1$).

18. The following differential equations arise in dealing with the problems noted. Find their complete solution, assuming that k is a nonzero constant:

(a)
$$\frac{d^4y}{dx^4} - k^4y = 0$$
 (vibration of a beam),
(b) $\frac{d^4y}{dx^4} + 4k^4y = 0$ (beam on an elastic foundation),
(c) $\frac{d^4y}{dx^4} - 2k^2\frac{d^2y}{dx^2} + k^4y = 0$ (bending of an elastic plate)

19. Use the method of undetermined coefficients to find the complete solution of each of the following differential equations:

(a)
$$\frac{d^2y}{dx^2} + k^2y = \sin x$$
 ($k^2 \neq 0, 1$), (b) $\frac{d^2y}{dx^2} + y = \sin x$,
(c) $\frac{d^2y}{dx^2} - y = \sin x$, (d) $\frac{d^2y}{dx^3} - y = e^x$,

(e)
$$\frac{d^2y}{dx^3} - y = x e^x$$
,
(f) $\frac{d^2y}{dx^2} - 2 \frac{dy}{dx} + 2y = \sin x$,
(g) $\frac{d^2y}{dx^2} - 2 \frac{dy}{dx} + 2y = e^x \sin x$,
(h) $\frac{d^2y}{dx^2} - 9 \frac{dy}{dx} + 20y = 4x^2 e^{3x}$,
(i) $\frac{d^2y}{dx^3} - \frac{dy}{dx} - 2y = 3 e^{-x} + 10 \sin x - 4x$.

Section 1.6

20. Solve the following differential equations:

(a)
$$x^{2} \frac{d^{2}y}{dx^{2}} + x \frac{dy}{dx} - k^{2}y = 0,$$

(b) $x^{2} \frac{d^{2}y}{dx^{3}} - x \frac{dy}{dx} + 2y = 0,$
(c) $x^{2} \frac{d^{2}y}{dx^{3}} - 2y = 0,$
(d) $x^{2} \frac{d^{2}y}{dx^{3}} - x \frac{dy}{dx} + y = 0,$
(e) $x^{3} \frac{d^{3}y}{dx^{3}} + 2x^{2} \frac{d^{2}y}{dx^{2}} - x \frac{dy}{dx} + y = 0,$
(f) $x^{2} \frac{d^{2}y}{dx^{3}} + 2x \frac{dy}{dx} - n(n+1)y = 0,$
(g) $x^{2} \frac{d^{2}y}{dx^{3}} + x \frac{dy}{dx} - y = x^{2},$
(h) $x^{2} \frac{d^{2}y}{dx^{3}} + x \frac{dy}{dx} - y = x,$
(i) $x^{2} \frac{d^{2}y}{dx^{3}} - 4x \frac{dy}{dx} + 6y = 6x + 12$

Section 1.7

21. Show, by direct expansion, that

$$(x^2 D^2)(x D) = (x D)(x^2 D^2)$$

 $(x D)^2 \neq x^2 D^2.$

but that

22. Use the method of Example 9 to obtain the general solution of the equation

$$(D - r_1)(D - r_2)y = f(x)$$
 $(r_1 \neq r_2),$

where r_1 and r_2 are constants, in the form

$$y = e^{r_{2}x} \int e^{-r_{2}x} \left[\int e^{-r_{1}x} f(x) \, dx \right] e^{r_{1}x} \, dx + c_{1}e^{r_{1}x} + c_{2}e^{r_{2}x}$$

23. (a) If the notation

$$y(x)=\frac{1}{D-r}f(x)$$

is used to indicate that y(x) satisfies the equation (D - r) y = f, where $D = \frac{d}{dx}$ and r is a constant, show that

$$\frac{1}{D-r}f(x)=e^{rx}\int^x e^{-rx}f(x)\,dx,$$

where an arbitrary additive constant of integration is implied in the integral.

(b) Verify that, with the notation of part (a), the expression

$$y = \frac{1}{(D - r_2)(D - r_1)} f(x) = \frac{1}{D - r_2} \left[\frac{1}{D - r_1} f(x) \right]$$

satisfies the equation $(D - r_1)(D - r_2)y = f$. Hence obtain the general solution of that equation in the form

$$y = e^{r_2 x} \int^x e^{(r_1 - r_2) x} \left[\int^x e^{-r_1 x} f(x) \, dx \right] dx,$$

where an arbitrary additive constant is implied in each integration.

24. Verify that, with the notation of Problem 23(a), the expression

$$y = \left[\frac{1}{r_1 - r_2} \left(\frac{1}{D - r_1} - \frac{1}{D - r_2}\right)\right] f(x)$$
$$= \frac{1}{r_1 - r_2} \left[\frac{1}{D - r_1} f(x) - \frac{1}{D - r_2} f(x)\right]$$

satisfies the equation $(D - r_1)(D - r_2)y = f$ when $r_1 \neq r_2$. Hence obtain the general solution of that equation in the form

$$y = \frac{1}{r_1 - r_2} \left[e^{r_1 x} \int^x e^{-r_1 x} f(x) \, dx \, - e^{r_2 x} \int^x e^{-r_2 x} f(x) \, dx \right]$$

when $r_1 \neq r_2$, where an arbitrary additive constant is implied in each integration.

Section 1.8

25. Suppose that the coefficients in Equations (37a,b) are constants. With the symbols $R_1 = L_1 x + L_2 y - h_1$ and $R_2 = L_3 x + L_4 y - h_2$, Equations (37a,b) become

$$R_1 = 0, \qquad R_2 = 0,$$
 (a,b)

and Equations (38a,b) or, equivalently, (39a,b) become

$$L_4R_1 - L_2R_2 = 0, \qquad L_1R_2 - L_3R_1 = 0.$$
 (c,d)

Show that if solutions of (c,d) also satisfy (b), so that $R_2 = 0$, there follows also $L_4R_1 = 0$, $L_3R_1 = 0$. Hence, noticing that these two equations can have a non-trivial common solution $(R_1 \neq 0)$ only if L_4 and L_3 have a common factor, deduce that if all the coefficients in Equations (37a,b) are constant and if, in either of Equations (37a,b), the two operators involved have no factors in common, the restrictions on the constants appearing in the solutions of Equations (39a,b) are completely determined by substitution into that equation. Otherwise, the solutions of Equations (37a,b).

26. Illustrate the results of Problem 25 in the case of the simultaneous equations

$$\begin{cases} \frac{d^2y}{dt^2} + \frac{dx}{dt} = 0\\ \frac{dy}{dt} + \frac{dx}{dt} + x = 0 \end{cases}$$

27. Find the solution of each of the following sets of equations:

(a)
$$\begin{cases} 2\frac{dx}{dt} = 3x - y \\ 2\frac{dy}{dt} = 3y - x, \end{cases}$$
(b)
$$\begin{cases} \frac{dx}{dt} - \frac{dy}{dt} + x = 2e^{t} \\ \frac{d^{2}x}{dt^{2}} - \frac{dy}{dt} + 3x - y = 4e^{t}, \end{cases}$$
(c)
$$\begin{cases} 2\frac{d^{2}x}{dt^{2}} - \frac{dy}{dt} - 4x = 2t \\ 2\frac{dx}{dt} + 4\frac{dy}{dt} - 3y = 0, \end{cases}$$
(d)
$$\begin{cases} \frac{d^{2}x}{dt^{2}} - \frac{1}{2}\frac{d^{2}y}{dt^{2}} + k^{2}x = 0 \\ \frac{d^{2}y}{dt^{2}} - \frac{d^{2}x}{dt^{2}} + k^{2}y = 0, \end{cases}$$
(e)
$$\begin{cases} \frac{d^{2}x}{dt^{2}} + x + 2y = 7e^{2t} - 1 \\ \frac{d^{2}x}{dt^{2}} + 3x + 2y = 9e^{2t} + 1, \end{cases}$$
(f)
$$\begin{cases} \frac{d^{2}x}{dt^{2}} + \frac{dy}{dt} + 3x + y = 0, \\ \frac{d^{2}x}{dt^{2}} + \frac{dy}{dt} + 3x + y = 0, \end{cases}$$
(g)
$$\begin{cases} \frac{dx}{dt} = 2x \\ \frac{dy}{dt} = 3x - 2y \\ \frac{dz}{dt} = 2y + 3z. \end{cases}$$

28. Find the solution of each of the following sets of equations:

(a)
$$\begin{cases} 2t \frac{dx}{dt} = 3x - y \\ 2t \frac{dy}{dt} = 3y - x, \end{cases}$$
 (b)
$$\begin{cases} t \frac{dx}{dt} = 2x \\ t \frac{dy}{dt} = 3x - 2y \\ t \frac{dz}{dt} = 2y + 3z. \end{cases}$$

Section 1.9

29. Solve by the method of variation of parameters:

(a)
$$\frac{dy}{dx} + a_1(x)y = h(x),$$

(b) $\frac{d^2y}{dx^2} + y = \cot x,$
(c) $\frac{d^2y}{dx^2} + y = \sec x,$
(d) $\frac{d^2y}{dx^2} + y = \log x,$
(e) $\frac{d^2y}{dx^2} - 2\frac{dy}{dx} + 2y = e^x \tan x,$
(f) $x^2 \frac{d^2y}{dx^2} - 4x \frac{dy}{dx} + 6y = x^2 \sin x,$
(g) $x^2 \frac{d^2y}{dx^2} - 2x \frac{dy}{dx} + 2y = x \log x.$

30. The differential equation

$$\frac{d^2y}{dx^2} + \frac{dy}{dx} - xy = 0$$

possesses solutions $u_1(x)$ and $u_2(x)$ which can be represented by series, valid near x = 0, the leading terms of which are as follows:

$$u_1(x) = 1 + \frac{1}{6}x^3 + \cdots, \quad u_2(x) = x - \frac{1}{2}x^2 + \cdots.$$

Use Abel's formula to show that $W(u_1, u_2) = e^{-x}$, and hence deduce that the general solution of the equation

$$\frac{d^2y}{dx^2} + \frac{dy}{dx} - xy = h(x)$$

is of the form

$$y = \int^x h(\xi) \left[u_1(\xi) \, u_2(x) \, - \, u_2(\xi) \, u_1(x) \right] \, e^{\xi} \, d\xi \, + \, c_1 u_1(x) \, + \, c_2 u_2(x) \, dx$$

Section 1.10

31. Verify that $y = e^x$ satisfies the homogeneous equation associated with (x - 1)y'' - xy' + y = 1, and obtain the general solution.

32. Verify that $y = \tan x$ satisfies the equation $y'' \cos^2 x = 2y$, and obtain the general solution.

33. One homogeneous solution of the equation

$$(1 - x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + 2y = 6(1 - x^2)$$

is y = x. Find the complete solution.

34. One solution of Legendre's equation,

$$(1 - x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + n(n+1)y = 0,$$

is called $P_n(x)$. Show that a second solution is of the form

$$P_n(x)\int^x\frac{dx}{(1-x^2)[P_n(x)]^2}.$$

35. One solution of Bessel's equation,

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (x^{2} - p^{2})y = 0,$$

is called $J_p(x)$. Show that a second solution is of the form

$$J_p(x) \int^x \frac{dx}{x[J_p(x)]^2}$$

36. Use Abel's formula (65) to show that if u_1 and u_2 are solutions of

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x)y = 0$$

in an interval I, and if we write

$$p=e^{\int a_1dx},$$

then the Wronskian of u_1 and u_2 is given by the expression

$$W[u_1(x), u_2(x)] = \frac{p(a)}{p(x)} \cdot \begin{vmatrix} u_1(a) & u_2(a) \\ u_1'(a) & u_2'(a) \end{vmatrix}$$

where x = a is a point in the interval *I*.

37. In the differential equation,

$$\frac{d^2y}{dx^2}+a_1(x)\frac{dy}{dx}+a_2(x)y=h(x),$$

make the substitution y = u(x) v(x) and determine v so that the coefficient of $\frac{du}{dx}$ in the resultant equation vanishes. Thus, show that, if $p = e^{\int a_1 dx}$, the substitution $y = u/\sqrt{p}$ reduces the differential equation to the form

$$\frac{d^2 u}{dx^2} - \frac{1}{4} \left(a_1^2 + 2 \frac{da_1}{dx} - 4a_2 \right) u = \sqrt{p} h(x).$$

38. Use the result of Problem 37 to show that Bessel's equation (Problem 35) takes the form

$$x^{2}\frac{d^{2}u}{dx^{2}}+\left(x^{2}-p^{2}+\frac{1}{4}\right)u=0$$

with the substitution

$$y=\frac{u}{\sqrt{x}}.$$

Section 1.11

39. A fundamental set of solutions of a linear differential equation of order n, relative to a point x = a, may be defined as a set of n solutions $u_1(x), u_2(x), \ldots, u_n(x)$ such that

(a) Deduce that the solution of the initial-value problem in which the values $y(a) = y_0, y'(a) = y'_0, \dots, y^{(n-1)}(a) = y^{(n-1)}_0$ are prescribed is then given by

$$y(x) = \sum_{k=1}^{n} y_0^{(k)} u_k(x).$$

(b) Show that the Wronskian of a fundamental set of solutions, relative to x = a, is unity at x = a.

40. Show that the functions $u_1 = 1$, $u_2 = \sin x$, and $u_3 = 1 - \cos x$ comprise a fundamental set of solutions for the equation

$$\frac{d^3y}{dx^3}+\frac{dy}{dx}=0,$$

relative to x = 0.

Problems

41. Show that the functions $\cosh x$ and $\sinh x$ constitute a fundamental set of solutions relative to x = 0 for the equation

$$\frac{d^2y}{dx^2}-y=0.$$

(See Problem 15.)

42. Obtain a fundamental set of solutions for the equation

$$\frac{d^4y}{dx^4}-y=0,$$

relative to x = 0.

43. (a) Determine those values of the constant k for which the differential equation $y'' + k^2y = 0$ possesses a nontrivial solution which vanishes when x = 0 and when x = a, where a is a given positive constant.

(b) Determine those values of k for which the differential equation $y'' + k^2y = 1$ possesses a solution which vanishes when x = 0 and when x = a.

Section 1.12

44. Solve the following:

(a)
$$(2xy^3 + y) dx + (3x^2y^2 + x - 2y) dy = 0$$
,
(b) $(3x^2y - y^3) dx - (3xy^2 - x^3) dy = 0$,
(c) $e^x(y dx + dy) + e^y(dx + x dy) = 0$.

45. Solve the following:

(a)
$$\frac{dy}{dx} = \frac{x - y}{x + y}$$
,
(b) $(x^2 + 2xy) dy = y^2 dx$,
(c) $x \frac{dy}{dx} = y - \sqrt{x^2 + y^2}$,
(d) $x dy - y dx = x \tan\left(\frac{y}{x}\right)$.

46. Solve the following:

(a)
$$x \frac{dy}{dx} + y = y \log (xy)$$
,
(b) $x dy + (y - x^2) dx = 0$,
(c) $y dx + (x - y^2) dy = 0$,
(d) $(x + y - 3) dy = (x - y + 1) dx$,
(e) $x \frac{dy}{dx} + y + x^2y^2 = 0$,
(f) $(2xy^2 - x) dx + (x^2y + y) dy = 0$.

47. Solve the following:

(a)
$$\frac{d^2y}{dx^2} + \left(\frac{dy}{dx}\right)^2 = 0,$$

(b) $\frac{d^2y}{dx^2} + 2x\left(\frac{dy}{dx}\right)^2 = 0,$
(c) $y\frac{d^2y}{dx^2} + \left(\frac{dy}{dx}\right)^2 = 0,$
(d) $\frac{d^2y}{dx^2} = \left[1 + \left(\frac{dy}{dx}\right)^2\right]^{\frac{1}{2}},$
(e) $\frac{d^2y}{dx^2} + y = 0,$
(f) $\left[6\left(\frac{dy}{dx}\right)^2 + 1\right]\frac{d^2y}{dx^2} = 1.$

48. Show that the substitution

$$y=\frac{u'}{Q\,u},$$

where $u' = \frac{du}{dx}$, reduces the nonlinear first-order equation

$$\frac{dy}{dx} + P(x)y + Q(x)y^2 = R(x)$$

to the linear second-order equation

$$\frac{d^2u}{dx^2}+\left(P-\frac{Q'}{Q}\right)\frac{du}{dx}-RQ\ u=0.$$

(The nonlinear form is known as *Riccati's equation*.)

49. Use the procedure suggested in Problem 48 to obtain the general solution of the equation

$$x^2\frac{dy}{dx} + xy + x^2y^2 = 1$$

in the form $xy = (x^2 - k)/(x^2 + k)$, where k is an arbitrary constant.

CHAPTER 2

The Laplace Transform

2.1. An introductory example. If a function f(t) is multiplied by e^{-st} and the result is integrated with respect to t from t = 0 to $t = \infty$, a new function of the variable s is obtained when the integral exists.* This function (when it exists) is called the *Laplace transform* of f(t). Before studying the properties of such transforms, we illustrate one of their most useful applications by considering a simple problem.

Suppose that we require the solution of the differential equation

$$\frac{dy}{dt} - y = e^{at},\tag{1}$$

for positive values of t, which satisfies the initial condition

$$y(0) = -1.$$
 (2)

In place of determining the general solution of Equation (1) by the methods of Chapter 1, and then determining the arbitrary constant by satisfying Equation (2), we proceed as follows.

We first take the Laplace transform of both sides of Equation (1), by multiplying both sides of the equation by e^{-st} and integrating the results with respect to t from zero to infinity, to obtain the equation

$$\int_{0}^{\infty} e^{-st} \frac{dy}{dt} dt - \int_{0}^{\infty} e^{-st} y dt = \int_{0}^{\infty} e^{-st} e^{at} dt.$$
 (3)

It is assumed, of course, that the separate integrals exist for some range of values of s.

* The variable s is considered to be a *real* variable in this chapter. However, all results which hold when s is *real* and s > a, for some real value of a, also hold when s is *complex* and (real part of s) > a.

The integral on the right is readily evaluated,

$$\int_0^\infty e^{-st} e^{at} dt = -\frac{e^{-(s-a)t}}{s-a} \Big|_0^\infty = \frac{1}{s-a}, \qquad (4)$$

the integral existing when s > a. The first integral on the left in Equation (3) can be integrated formally by parts to give

$$\int_{0}^{\infty} e^{-st} \frac{dy}{dt} dt = e^{-st} y(t) \Big|_{0}^{\infty} + s \int_{0}^{\infty} e^{-st} y dt$$
$$= -y(0) + s \int_{0}^{\infty} e^{-st} y dt$$
$$= 1 + s \int_{0}^{\infty} e^{-st} y dt, \qquad (5)$$

assuming that $e^{-st} y(t)$ approaches zero, for sufficiently large values of s, as $t \to \infty$. Thus, the transform of $\frac{dy}{dt}$ is expressed in terms of the prescribed initial value of y and in terms of the transform of y itself.

If the results of Equations (4) and (5) are introduced into (3), there then follows

$$(s-1)\int_{0}^{\infty} e^{-st} y \, dt = \frac{1}{s-a} - 1$$
$$\int_{0}^{\infty} e^{-st} y \, dt = \frac{a+1-s}{(s-1)(s-a)}.$$
(6)

or

The original problem is now apparently reduced to the problem of determining a function y(t) whose Laplace transform is given by the right-hand side of Equation (6). To determine such a function, we first expand this expression by the method of partial fractions, to obtain the equivalent form

$$\int_0^\infty e^{-st} y \, dt = \frac{1}{a-1} \frac{1}{s-a} - \frac{a}{a-1} \frac{1}{s-1} \,. \tag{6a}$$

Reference to Equation (4) then indicates that, since 1/(s - a) is the transform of e^{at} , the first term of (6a) is the transform of $e^{at}/(a - 1)$ and the second term the transform of $-ae^t/(a - 1)$. Thus, (6a) will be satisfied if we write

$$y = \frac{1}{a-1} (e^{at} - ae^{t}),$$
 (7)

when $a \neq 1$. A corresponding expression in the case a = 1 can be obtained by taking the limit as $a \rightarrow 1$, in the form $y = (t - 1)e^t$.

With this expression for y, the validity of the transition from Equations (1) and (2) to (6a) is readily established when s > a. Still, it is by no means obvious that (7) is the *only* solution of (6a). That is, while (1) and (2) are known

to have a unique solution, it is conceivable that (6a) could have several solutions, only one of which would then also satisfy (1) and (2).

However, direct substitution shows that Equation (7) actually does represent the solution of (1) and (2). Further, it can be shown also that (7) is the only continuous solution of (6a).

Although this procedure has the advantage that the *particular* solution required is obtained directly, without first obtaining the *general* solution, it is clearly desirable to simplify the procedure by eliminating the necessity of carrying out certain general integrations in each case, and to determine in what cases such a procedure is valid.

In the remainder of this chapter, certain properties of Laplace transforms are investigated and relevant formulas are tabulated in such a way that the solution of initial-value problems involving linear differential equations with constant coefficients, or sets of simultaneous equations of this type, can be conveniently obtained. Thus, for example, use of the tabulated formulas will permit immediate transition from Equations (1) and (2) to (6), and from (6a) to the solution (7). Use of the methods to be given will, in general, introduce a considerable saving in labor over the alternative procedures of Chapter 1.

Laplace transforms are also useful, for example, in connection with the solution of certain problems governed by partial differential equations (and integral equations), as will be shown in later chapters. Certain of the properties developed in this chapter are of principal use in these later applications.

The so-called *operational methods* to be developed have a close formal resemblance to earlier techniques developed by Heaviside in a very ingenious but nonrigorous way. The use of Laplace transforms is to be preferred, since the relevant derivations can be put on a rigorous mathematical basis, so that success or failure of the method in a given problem can be predicted with complete confidence. Also, this technique possesses all the useful features of the Heaviside technique, as well as certain advantages in flexibility.

2.2. Definition and existence of Laplace transforms. The Laplace transform of a function f(t), defined for positive values of t, is frequently indicated by the notation $\mathscr{L}{f(t)}$ and is defined, as a function of the variable s, by the integral*

$$\mathscr{L}{f(t)} = \int_0^\infty e^{-st} f(t) dt \tag{8}$$

over that range of values of s for which the integral exists. The notation f(s), or merely f, is often used in place of $\mathscr{L}{f(t)}$.

The integral (8) may fail to define a function of s, in particular, because of

* Some authors replace definition (8) by the definition

$$\mathscr{L}{f(t)} = s \int_0^\infty e^{-st} f(t) dt.$$

infinite discontinuities in f(t) for certain positive values of t or because of failure of f(t) to behave in a sufficiently regular way near t = 0 or for large values of t. However, the presence of a finite number of *finite* discontinuities or "jumps" will not, in itself, affect the existence of the integral.

A function f(t) is said to be *piecewise continuous* in a finite range if it is possible to divide that range into a finite number of intervals such that f(t) is continuous inside each interval and approaches finite values as either end of any interval is approached from the interior. Such functions may thus have finite jumps at points inside the range considered. At such a point, say $t = t_0$, different limits are approached by f(t) as t approaches t_0 from the right (that is, from larger values of t) and from the left (from smaller values). These two limits are called *right-hand* and *left-hand* limits, respectively, and when necessary are conveniently indicated by the respective notations $\lim_{t \to t_0^+} f(t) = f(t_0^+)$ and $\lim_{t \to t_0^-} f(t) = f(t_0^-)$.

In illustration, if f(t) is defined to be unity when 0 < t < 1 and zero elsewhere, then f(t) is piecewise continuous over any range. There follows also, for example, f(1+) = 0 and f(1-) = 1. If f(t) is $1/\sqrt{t}$ when 0 < t < 1 and zero elsewhere, then f(t) is piecewise continuous in any range not including t = 0 as an interior or end point.

In the developments of this chapter, we consider only functions which are at least piecewise continuous in every positive range not including zero as an end point. Then, if we write (8) as the sum of three integrals,

$$\mathscr{L}\lbrace f(t)\rbrace = \int_0^\infty e^{-st} f(t) \, dt = \left(\int_0^{t_1} + \int_{t_1}^T + \int_T^\infty\right) e^{-st} f(t) \, dt, \qquad (8a)$$

the second integral on the right exists for all positive finite values of t_1 and T.

If, in addition, f(t) approaches a finite limit as $t \to 0+$ or if $|f(t)| \to \infty$ as $t \to 0+$ in such a way that for some number n, n < 1, the product $t^n f(t)$ is bounded near t = 0, then the first integral of Equation (8a) exists.

Finally, a sufficient additional condition to guarantee the existence of the third integral of (8a), at least for sufficiently large values of s, is the requirement that f(t) belong to the rather extensive class of "functions of exponential order." A function f(t) is said to be of exponential order if, for some number s_0 , the product $e^{-s_0 t} | f(t) |$ is bounded for large values of t, say for t > T. If the bound is denoted by M, then there follows, when t > T,

$$e^{-s_0 t} |f(t)| < M \text{ or } |f(t)| < M e^{s_0 t}.$$
 (9)

Thus, though f(t) may become infinitely large as $t \to \infty$, we see that |f(t)| must not "grow" more rapidly than a multiple of *some* exponential function of t. We say that f(t) is "of the order" of $e^{s_0 t}$ and frequently write $f(t) = O(e^{s_0 t})$. In particular, if $\lim_{t\to\infty} e^{-s_0 t} |f(t)|$ exists (and is finite) for some $s_0 > 0$, then for sufficiently large values of t the product must be bounded, and hence f(t) is

of the order $e^{s_0 t}$. The limit may, of course, be zero, in which case we also write $f(t) = o(e^{s_0 t})$.

We note that any bounded function is of exponential order with $s_0 = 0$. Other examples are e^{at} (with $s_0 = a$), $e^{at} \sin bt$ (with $s_0 = a$), and t^n (with s_0 any positive number no matter how small). The function e^{t^2} is not of exponential order, since $e^{-s_0 t} e^{t^2} = e^{t^2 - s_0 t}$ is unbounded as $t \to \infty$ for all values of s_0 .

If f(t) is piecewise continuous and of exponential order, then its integral $\int_0^t f(u) du$ is continuous and is *also* of exponential order. Although it *cannot* be said in general that derivatives of functions of exponential order have the same property, this is true in most practical cases.

As an example of a reasonably simple exceptional case, we notice that though sin (e^{t^2}) is bounded, and hence of exponential order, its derivative $2te^{t^2} \cos(e^{t^2})$ is not of exponential order.

In case f(t) is of exponential order, and hence satisfies Equation (9) when t > T, then there follows

$$|e^{-st} f(t)| < e^{-st} \cdot M e^{s_0 t} = M e^{-(s-s_0)t}$$

Hence, since we have assumed f(t) to be piecewise continuous and since $\int_{T}^{\infty} e^{-(s-s_0)t} dt$ exists if $s > s_0$, it follows that the third integral in (8a) then also exists when $s > s_0$.

Thus, in summary, the Laplace transform of f(t) exists, when s is sufficiently large, if f(t) satisfies the following conditions:

(1) f(t) is continuous or piecewise continuous in every finite interval $t_1 \leq t \leq T$, where $t_1 > 0$.

(2)
$$t^n |f(t)|$$
 is bounded near $t = 0$ for some number n, where $n < 1$.

(3) $e^{-s_0 t} |f(t)|$ is bounded for large values of t, for some number s_0 .

Although the transform may also exist in other cases, these conditions are sufficiently weak to include most functions occurring in practice.

For reference purposes, it may be stated that whenever any integral of the form

$$\int_0^\infty e^{-st} f(t) \, dt$$

exists for $s = s_0$, it exists also for all s such that $s \ge s_0$. Also, it is then true that

$$\lim_{s \to c} \int_0^\infty e^{-st} f(t) \, dt = \int_0^\infty e^{-ct} f(t) \, dt \tag{10}$$

when $c \ge s_0$, that

$$\frac{d}{ds}\int_0^\infty e^{-st}f(t)\,dt = \int_0^\infty \left(\frac{d}{ds}\,e^{-st}\right)f(t)\,dt \tag{11}$$

when $s \ge s_0$, and that

$$\int_{\alpha}^{\beta} \left[\int_{0}^{\infty} e^{-st} f(t) dt \right] ds = \int_{0}^{\infty} \left[\int_{\alpha}^{\beta} e^{-st} f(t) ds \right] dt$$
(12)

when $s_0 \leq \alpha \leq \beta < \infty$. The remarkable fact that all these operations can be effected under the integral sign, for any convergent Laplace transform, is of frequent usefulness.

The direct calculation of Laplace transforms may be illustrated by the following simple cases:

$$\mathscr{L}\{1\} = \int_0^\infty e^{-st} dt = -\frac{e^{-st}}{s} \Big|_0^\infty = \frac{1}{s} \qquad (s > 0).$$
(13a)

$$\mathscr{L}\{e^{at}\} = \int_0^\infty e^{-(s-a)t} dt = -\frac{e^{-(s-a)t}}{s-a} \Big|_0^\infty = \frac{1}{s-a} \qquad (s>a). \quad (13b)$$

$$\mathscr{L}\{\sin at\} = \int_0^\infty e^{-st} \sin at \, dt = -\frac{e^{-st}}{s^2 + a^2} (s \sin at + a \cos at) \Big|_0^\infty$$
$$= \frac{a}{s^2 + a^2} \quad (s > 0). \tag{13c}$$

2.3. Properties of Laplace transforms. Among the most useful properties of Laplace transforms are the following:

$$\mathscr{L}{af(t) + bg(t)} = a\tilde{f}(s) + b\tilde{g}(s).$$
(14)

$$\mathscr{L}\left\{\frac{d^{n}f(t)}{dt^{n}}\right\} = s^{n}\tilde{f}(s) - \left[s^{n-1}f(0+) + s^{n-2}\frac{df(0+)}{dt} + s^{n-3}\frac{d^{2}f(0+)}{dt^{2}} + \dots + \frac{d^{n-1}f(0+)}{dt^{n-1}}\right].$$
 (15)

$$\mathscr{L}\left\{\int_{0}^{t} f(u) \, du\right\} = \frac{1}{s}\bar{f}(s). \tag{16}$$

$$\mathscr{L}\lbrace e^{at}f(t)\rbrace = \bar{f}(s-a). \tag{17}$$

If
$$f(t) = \begin{cases} 0, & t < a \\ g(t-a), & t \ge a \end{cases}$$
 with $a \ge 0$, then $\bar{f}(s) = e^{-as} \bar{g}(s)$. (18)

$$\mathscr{L}{t^n f(t)} = (-1)^n \frac{d^n f(s)}{ds^n} \,. \tag{19}$$

$$\mathscr{L}\left\{\int_0^t f(t-u) g(u) \, du\right\} = \bar{f}(s) \, \bar{g}(s). \tag{20}$$

In these equations a and b are constants, and n is a positive integer.

In all cases except Equation (15), we suppose that the functions f(t) and g(t) satisfy the conditions of page 55. In the case of (15), more stringent restrictions are imposed. These conditions are stated in connection with the proof to be given in this section (see page 58).

Equation (14) expresses the linear property of Laplace transforms. Its proof follows directly from the definition, in view of the corresponding linear properties of the integral.

Example 1. By using Equations (14) and (13b), we obtain

$$\mathscr{L}\{\sinh at\} = \mathscr{L}\left\{\frac{1}{2}e^{at} - \frac{1}{2}e^{-at}\right\} = \frac{1}{2(s-a)} - \frac{1}{2(s+a)} = \frac{a}{s^2 - a^2}.$$

Equation (15) states one of the most important properties of Laplace transforms. It expresses the transform of any derivative of a function in terms of the transform of the function itself and in terms of the values of the lowerorder derivatives of the function at t = 0 (or, more precisely, the values approached by these derivatives as $t \rightarrow 0$ from positive values). We consider first the case when n = 1, for which, from the definition,

$$\mathscr{L}\left\{\frac{df(t)}{dt}\right\} = \int_0^\infty e^{-st} \frac{df(t)}{dt} dt.$$

An integration by parts gives

$$\int_0^\infty e^{-st} \frac{df(t)}{dt} dt = e^{-st} f(t) \Big|_0^\infty + s \int_0^\infty e^{-st} f(t) dt$$

if f(t) is continuous and $\frac{df(t)}{dt}$ is piecewise continuous in every interval (0,T).* But since f(t) is of exponential order, the integrated part vanishes as $t \to \infty$ (for $s > s_0$), and there follows

$$\mathscr{L}\left\{\frac{df(t)}{dt}\right\} = s\,\bar{f}(s) - f(0+). \tag{15a}$$

Similarly, in the case n = 2, integration by parts gives

$$\mathscr{L}\left\{\frac{d^2f(t)}{dt^2}\right\} = \int_0^\infty e^{-st} \frac{d^2f(t)}{dt^2} dt$$
$$= e^{-st} \frac{df(t)}{dt} \Big|_0^\infty + s \int_0^\infty e^{-st} \frac{df(t)}{dt} dt$$
$$= e^{-st} \frac{df(t)}{dt} \Big|_0^\infty + s \mathscr{L}\left\{\frac{df(t)}{dt}\right\},$$

* Unless these conditions are satisfied, the formula for integration by parts may not be valid.

if $\frac{df}{dt}$ is continuous and $\frac{d^2f}{dt^2}$ piecewise continuous. If $\frac{df}{dt}$ is also of exponential order, the integrated part again vanishes as $t \to \infty$ and, making use of Equation (15a), there then follows

$$\mathscr{L}\left\{\frac{d^2f(t)}{dt^2}\right\} = s^2 \tilde{f}(s) - s f(0+) - \frac{df(0+)}{dt}.$$
 (15b)

Equations (15a) and (15b) are special cases of (15). The general proof of (15) follows by induction from the general result

$$\mathscr{L}\left\{\frac{d^n f(t)}{dt^n}\right\} = s \,\mathscr{L}\left\{\frac{d^{n-1} f(t)}{dt^{n-1}}\right\} - \frac{d^{n-1} f(0+)}{dt^{n-1}},$$

if $\frac{d^{n-1}f(t)}{dt^{n-1}}$ is continuous and $\frac{d^n f(t)}{dt^n}$ is piecewise continuous, and if f(t), $\frac{df(t)}{dt}, \ldots, \frac{d^n f(t)}{dt^n}$ are all of exponential order. This result is obtained, as in the special cases n = 1 and 2, by an integration by parts.

We thus obtain the following result:

Equation (15) is valid if f(t) and its first n - 1 derivatives are continuous over every interval (0,T), if $\frac{d^n f(t)}{dt^n}$ is (at least) piecewise continuous over every interval (0,T), and if f(t) and its first n derivatives are of exponential order. These conditions are satisfied by a great variety of functions of physical interest.

Example 2. If we take $f(t) = \sin at$, then, from (13c),

$$f(s) = \frac{a}{s^2 + a^2}.$$

Equation (15a) then gives

$$\mathscr{L}\{a\cos at\} = s\left(\frac{a}{s^2+a^2}\right) - \sin 0$$

or, using (14),

$$\mathscr{L}\{\cos at\} = \frac{s}{s^2 + a^2}.$$

In particular, for the class of functions considered, we see that if a function and its first n - 1 derivatives vanish at t = 0 (or as $t \rightarrow 0+$), the transform of its nth derivative is obtained by multiplying the transform of the function by s^n . Applications of this fact are closely related to the use of the operational notation $D^n f(t)$ to represent $\frac{d^n f(t)}{dt^n}$. Equation (16) is established by similar methods. Again making use of integration by parts, and recalling that

$$\frac{d}{dt}\int_0^t f(u)\ du=f(t),$$

we obtain

$$\mathscr{L}\left\{\int_{0}^{t} f(u) \, du\right\} = \int_{0}^{\infty} e^{-st} \left\{\int_{0}^{t} f(u) \, du\right\} dt$$
$$= \left[\frac{e^{-st}}{-s} \int_{0}^{t} f(u) \, du\right]_{0}^{\infty} + \frac{1}{s} \int_{0}^{\infty} e^{-st} f(t) \, dt$$
$$= \frac{1}{s} \bar{f}(s),$$

the integrated part vanishing at the upper limit (for sufficiently large values of s), since f(t), and hence $\int_0^t f(u) du$, is of exponential order. Thus, in general, if a function is integrated over (0,t), the transform of the integral is obtained by dividing the transform of the function by s.

If the lower limit differs from zero, the formula

$$\mathscr{L}\left\{\int_{a}^{t} f(u) \, du\right\} = \frac{1}{s} \bar{f}(s) - \frac{1}{s} \int_{0}^{a} f(u) \, du \tag{21}$$

is easily established.

Equations (17) and (18) express the so-called *translation properties* of the Laplace transform. The proof of the former property follows immediately from the definition, since the transform of $e^{at} f(t)$ is given by

$$\int_0^\infty e^{-st} \left[e^{at} f(t) \right] dt = \int_0^\infty e^{-(s-a)t} f(t) dt,$$

and the last expression differs from f(s) only in that s is replaced by s - a. Thus, if a function is multiplied by e^{at} , the transform of the result is obtained by replacing s by s - a in the transform of the original function. It is seen that if f(s) is plotted as a function of s, the representation of the transform of $e^{at} f(t)$ is thus obtained by shifting or "translating" the transform of f(t)through a units in the positive direction of s.

Example 3. If we take $f(t) = \sin bt$, Equation (13c) gives $f(s) = \frac{b}{s^2 + b^2}$, and (17) then gives

$$\mathscr{L}\lbrace e^{at} \sin bt \rbrace = \frac{b}{(s-a)^2 + b^2}.$$

Suppose now that a function is defined to be g(t) for $t \ge 0$ and to be zero for negative values of t. Its transform may be denoted by $\overline{g}(s)$. If the given

function is translated through a units in the positive direction of t, and so becomes g(t - a) when $t \ge a$ and zero otherwise, Equation (18) states that the transform of the translated function is obtained by multiplying the transform of the original function by e^{-as} . To establish this property, we notice that since the translated function vanishes when $0 \le t < a$, its transform is defined by the integral

$$\int_a^\infty e^{-st}\,g(t-a)\,dt.$$

If t is replaced by t + a, and the lower limit of the integral is changed accordingly, this integral becomes

$$\int_0^\infty e^{-s(t+a)} g(t) dt = e^{-as} \int_0^\infty e^{-st} g(t) dt = e^{-as} \bar{g}(s),$$

in accordance with Equation (18).

Example 4. If $f(t) = \sin(t - t_0)$ when $t \ge t_0$ and f(t) = 0 when $t < t_0$, there follows $g(t) = \sin t$, and, from (13c), $\overline{g}(s) = (s^2 + 1)^{-1}$. Thus Equation (18) gives

$$\mathscr{L}\left\{f(t)\right\} = \frac{e^{-st_0}}{s^2+1}\,.$$

To establish the property of *Equation* (19), we merely differentiate both sides of the equation

$$\bar{f}(s) = \int_0^\infty e^{-st} f(t) \, dt$$

n times with respect to *s*. Differentiation under the integral sign is valid for all values of *s* for which the transform exists, as was stated at the end of Section 2.2.

Example 5. To find the transform of t^n , Equation (19) states that we merely differentiate the transform of *unity* n times with respect to s and multiply the result by $(-1)^n$. We thus obtain

$$\mathscr{L}{t^n} = (-1)^n \frac{d^n}{ds^n} \left(\frac{1}{s}\right) = \frac{n!}{s^{n+1}}$$

where n is a positive integer or zero, with the usual convention that 0! = 1.

2.4. The inverse transform. In applications of Laplace transforms we frequently encounter the inverse problem of determining a function which has a given transform. The notation $\mathscr{L}^{-1}{F(s)}$ is conventionally used for the inverse Laplace transform of F(s); that is, if $F(s) = \mathscr{L}{f(t)}$, then we write also $f(t) = \mathscr{L}^{-1}{F(s)}$. The notation $f(t) \leftrightarrow F(s)$ is also frequently useful:

To determine the inverse transform of a given function F(s) it is thus necessary to determine a function f(t) which satisfies the equation

$$\int_0^\infty e^{-st}f(t)\,dt=F(s).$$

Since the unknown function f(t) appears under an integral sign, an equation of this type is called an *integral equation*.

In more advanced works it is proved that if this equation has a solution, then that solution is *unique*. Thus, *if one function having a given transform is* known, it is the only possible one. This result is known as Lerch's theorem.

More precisely, Lerch's theorem states that two functions having the same transform cannot differ throughout any interval of positive length. Thus, for example, Equation (13a) shows that the continuous solution of

$$\int_0^\infty e^{-st} f(t) \, dt = \frac{1}{s}$$

is f(t) = 1; that is, $\mathcal{L}^{-1}{s^{-1}} = 1$. However, it is clear that if we take f(t) to be, say, zero at t = 1 and unity elsewhere, or otherwise redefine the function f(t) at a finite number of points, the value of the integral is not changed. Hence the new function is also a solution. Such artificialities are, however, generally of no significance in applications.

Although the direct determination of inverse transforms involves methods outside the scope of this chapter,* extensive tables of corresponding functions and transforms are available in the literature, and their use (in conjunction with the use of the properties listed in Section 2.3) is sufficient for many purposes. A short table of this sort is presented on pages 74–76.

It should be pointed out that not all functions of s are transforms, but that the class of such functions is greatly restricted by requirements of continuity and satisfactory behavior as $s \to \infty$. A useful result in this connection is the following: If f(t) is piecewise continuous in every finite interval $0 \le t \le T$ and is of exponential order, then $\bar{f}(s) \to 0$ as $s \to \infty$; furthermore $s \bar{f}(s)$ is bounded as $s \to \infty$. The proof follows from the fact that in such cases

$$|f(t)| < M e^{s_0 t}$$
 and $|e^{-st} f(t)| < M e^{-(s-s_0)t}$

for some fixed constants s_0 and M. Hence we have

$$\begin{split} |\bar{f}(s)| &= \left| \int_0^\infty e^{-st} f(t) \, dt \right| \leq \int_0^\infty e^{-st} |f(t)| \, dt \\ &\leq M \int_0^\infty e^{-(s-s_0)t} \, dt \\ &\leq \frac{M}{s-s_0} \, . \end{split}$$

* See, however, Problems 87–94, Chapter 10.
The theorem stated then follows from the fact that $\frac{M}{s-s_0}$ approaches zero and $\frac{s}{s-s_0}$ M remains bounded as $s \to \infty$. Thus, such functions as 1, $\frac{s}{s+1}$, $\frac{1}{\sqrt{s}}$, and sin s cannot be transforms of functions satisfying the conditions stated.

It should also be noted that if f(t) is continuous and $\frac{df(t)}{dt}$ is piecewise continuous in every finite interval $0 \le t \le T$, and if f(t) and $\frac{df(t)}{dt}$ are of exponential order, then

$$\lim_{s \to \infty} s \, \bar{f}(s) = f(0+). \tag{22}$$

This result follows from the fact that in this case the preceding theorem states that the left-hand side of Equation (15a) vanishes as $s \rightarrow \infty$. It is useful in those cases when only the *initial value* of f(t) is required and the *transform* of f is known.

2.5. The convolution. It frequently happens that, although a given function F(s) is not the transform of a known function, it can be expressed as the product of two functions, each of which *is* the transform of a known function. Thus, it may be possible to write

$$F(s) = \bar{f}(s)\,\bar{g}(s),$$

where f(s) and g(s) are known to be the transforms of the functions f(t) and g(t), respectively. We suppose that these functions satisfy the conditions of page 55. In this case, Equation (20) states that the product $f(s)\bar{g}(s)$ is the transform of the function defined by the integral $\int_0^t f(t-u)g(u) du$. This integral is called the convolution of f and g, and may be denoted by the abbreviation f * g. It is indicated by the symmetry in \bar{f} and \bar{g} that f and g can be interchanged in the convolution, that is, that f * g = g * f. Before outlining the proof of Equation (20), we illustrate its application.

Example 6. To determine $\mathscr{L}^{-1}\left\{\frac{a}{s(s-a)}\right\}$ we refer to Equations (13a) and (13b) and write the given function of s in the form

$$\frac{a}{s}\frac{1}{s-a}=\tilde{f}(s)\,\tilde{g}(s)$$

where f(t) = a and $g(t) = e^{at}$. Equation (20) states that the product is the transform of the function

$$f \ast g = \int_0^t a \, e^{au} \, du = e^{at} - 1.$$

If the functions f and g are interchanged, there follows alternatively

$$g * f = \int_0^t e^{a(t-u)} a \, du = e^{at} \int_0^t e^{-au} a \, du = e^{at} (1 - e^{-at}) = e^{at} - 1,$$

as before. The same result is obtained without making use of the convolution, in this case, if the product is first expanded by the method of partial fractions in the form $(s - a)^{-1} - s^{-1}$, and use is then made of Equations (13a) and (13b).

Equation (20) can be obtained formally as follows. From the definition, the right-hand side of (20) can be written in the form

$$\bar{f}(s) \ \bar{g}(s) = \left[\int_0^\infty e^{-sv} f(v) \, dv\right] \left[\int_0^\infty e^{-su} g(u) \, du\right]$$
$$= \int_0^\infty \int_0^\infty e^{-s(v+u)} f(v) g(u) \, dv \, du$$
$$= \int_0^\infty g(u) \left\{\int_0^\infty e^{-s(v+u)} f(v) \, dv\right\} du$$

if different "dummy variables" of integration (v and u) are used in defining the two transforms. If, in the inner integral

of the last form, we replace v by a new variable t with the substitution

$$v = t - u, \quad dv = dt,$$

there follows

$$\int_0^\infty e^{-s(v+u)} f(v) \, dv = \int_u^\infty e^{-st} f(t-u) \, dt,$$

and hence

$$\bar{f}(s)\,\bar{g}(s)=\int_0^\infty\left\{\int_u^\infty e^{-st}f(t-u)\,dt\right\}\,du.$$

Interchanging the order of integration in

the double integral and changing the limits as indicated in Figure 2.1, we then obtain formally

$$\bar{f}(s) \ \bar{g}(s) = \int_0^\infty \left\{ \int_0^t e^{-st} f(t-u) \ g(u) \ du \right\} dt$$
$$= \int_0^\infty e^{-st} \left\{ \int_0^t f(t-u) \ g(u) \ du \right\} dt$$
$$= \mathscr{L}\left\{ \int_0^t f(t-u) \ g(u) \ du \right\}$$

in accordance with Equation (20). The interchange of order of integration can be shown to be legitimate, by using appropriate limiting processes, when fand g satisfy the assumed conditions.

2.6. Singularity functions. Consider the function f(t) which has the value $1/t_0$ when $0 < t < t_0$ and is zero elsewhere (Figure 2.2). We then have

$$\int_0^\infty f(t) \, dt = \int_0^{t_0} f(t) \, dt = 1;$$



Figure 2.1

that is, the area under the graph representing f(t) is unity. The transform of this function is found to be

$$\mathscr{L}{f(t)} = \frac{1}{t_0} \int_0^{t_0} e^{-st} dt = \frac{1 - e^{-st_0}}{st_0}$$

Now as $t_0 \rightarrow 0$, the magnitude of f(t) over $(0, t_0)$ increases without limit, while, at the same time, the length of the interval $(0, t_0)$ shrinks toward zero in such a way that the integral $\int_0^{t_0} f(t) dt$ retains the value of unity.



In the limit we have the *ideal* case of a "function" which is *infinite* at the point t = 0 and zero elsewhere, but which has the property that its integral across t = 0 is unity. Making use of L'Hospital's rule, we find

$$\lim_{t_0 \to 0} \frac{1 - e^{-st_0}}{st_0} = \lim_{t_0 \to 0} \frac{s e^{-st_0}}{s} = 1.$$

That is, the transform of f(t) approaches unity as $t_0 \rightarrow 0$.

If t represents time and f(t) force, then $\int_0^{t_0} f(t) dt$ represents the *impulse* of the force f(t) acting over the time interval $(0, t_0)$. Hence, as $t_0 \rightarrow 0$, we may speak loosely of a resulting "unit impulse" at t = 0, due to "an infinite force acting over a zero time interval." In view of this interpretation, the limiting form of f(t) is frequently called the *unit impulse function*. If we denote it by $\delta(t)$, we are led to write

$$\mathscr{L}\{\delta(t)\} = 1. \tag{23}$$

The "function" $\delta(t)$ is also often called the *Dirac delta function*. If, for example, t represented *distance* along, say, the centerline of a beam and f(t) represented the intensity of a distributed *load*, then $\delta(t)$ could be considered as the formal representation of a *concentrated unit load* applied at the *point* t = 0, and analogous interpretations in other fields are frequently useful.

In a similar way, if f(t) has the value $-1/t_0^2$ when $0 < t < t_0$, the value $+1/t_0^2$ when $t_0 < t < 2t_0$, and the value zero elsewhere (Figure 2.3), there follows

$$\int_0^\infty (t-t_0) f(t) \, dt = \int_0^{2t_0} (t-t_0) f(t) \, dt = 1.$$

That is, the *moment* of the area under the graphical representation of f(t), about the point t_0 , is unity, whereas the (signed) area is zero. The transform of f(t) is

$$\mathscr{L}{f(t)} = -\frac{1}{t_0^2} \left\{ \int_0^{t_0} e^{-st} dt - \int_{t_0}^{2t_0} e^{-st} dt \right\}$$
$$= -\frac{1}{st_0^2} (1 - e^{-st_0})^2.$$

Repeated use of L'Hospital's rule shows that, as $t_0 \rightarrow 0$, $\mathscr{L}{f(t)} \rightarrow -s$. The limiting form of f(t), as $t_0 \rightarrow 0$, is frequently referred to as the *doublet function*, because of certain interpretations relating to electric field theory and fluid flow. Denoting *the negative* of this limiting form by $\delta'(t)$, we write

$$\mathscr{L}\{\delta'(t)\} = s. \tag{24}$$

With the interpretation of t as distance and f(t) as load intensity, $\delta'(t)$ could be considered as the formal representation of a concentrated negative unit moment applied at the point t = 0.

Such functions, often called *singularity functions*, are dealt with rigorously in the branch of mathematics known as the *theory of distributions*, and are of frequent use in physical applications. Although they do not conform to the restrictions of page 55, and although, in fact, *they are not true functions*, nevertheless *if formal use of them leads to a result which is capable of physical interpretation*, *then in practical cases the result may be accepted as correct*.

If the singularity occurs at $t = t_1$ rather than at t = 0, we denote the corresponding functions by $\delta(t - t_1)$ and $\delta'(t - t_1)$ and obtain (by limiting processes analogous to those given above) the formal results

$$\mathscr{L}\left\{\delta(t-t_1)\right\} = e^{-st_1}, \qquad \mathscr{L}\left\{\delta'(t-t_1)\right\} = s \ e^{-st_1}. \tag{25a,b}$$

It may be noticed that these results are also obtained by formally applying the translation property (18) to Equations (23) and (24).

2.7. Use of table of transforms. A brief table of corresponding functions and transforms is given on pages 74–76, in order to facilitate the determination of both direct and inverse transforms. The first ten pairs represent general relationships proved in Section 2.3. In pairs (T3,4,5) the conditions of page 55 are assumed, whereas in the remaining pairs the less restrictive conditions of page 58 are implied.

Pairs (T11-30) either have been established in examples or can be easily obtained from established results by using certain of the general properties. Only frequently occurring basic forms are listed; other forms are readily deduced from those given. Pairs (T31-34) involve the singularity functions of the preceding section.

Pair (T35) was derived in Example 5 when n is a positive integer and (T36,37) follow, in this case, by virtue of (T8) and (T3), respectively. The formulas are valid also (under the given restrictions on n) if n is not an integer, if (n - 1)! is interpreted in a way to be defined in Section 2.9.

Pairs (T38-41) are included principally for future reference. In these pairs the functions J_m and I_m are certain functions known as *Bessel functions of* order m. These functions are to be treated in Chapter 4. If, in these pairs, n is zero or a positive integer, then the order of the function involved is half an odd integer. In such cases the functions in brackets, in the right-hand column, can be expressed in terms of products of polynomials and either circular or hyperbolic functions. These expressions are given for $m = -\frac{1}{2}, \frac{1}{2}, \ldots, \frac{9}{2}$ in Table 2, page 77 (where x is written for at). The expressions for $m = \frac{11}{2}, \frac{13}{2}$, and so on, can be obtained in terms of these expressions by use of the recurrence formulas listed at the foot of the table.

Although the transforms of the simpler functions of frequent occurrence in practice can be obtained directly from the table, or by direct integration, the determination of *inverse transforms* may frequently involve a certain amount of manipulation. In this connection, it should be observed that *if n is a positive integer*, all functions of t appearing in these tables (except the singularity functions), as well as all their derivatives, are continuous everywhere and are of exponential order. Hence it follows that *if n is a positive integer*, all properties (T1-10) can be applied to all succeeding pairs in the table, except for (T31-34).

Pair (T3) is particularly useful in the determination of inverse transforms, when f(0) = 0. Reference to Equation (22) shows that this is so if s f(s) tends to zero as $s \to \infty$. Hence it follows that if $\lim_{s \to \infty} s f(s) = 0$, then

$$\mathscr{L}^{-1}\{s\,\bar{f}(s)\} = \frac{df}{dt}\,.\tag{26}$$

Example 7. To determine $\mathscr{L}^{-1}\left\{\frac{s^2}{(s^2+4)^2}\right\}$, we first obtain from (T19) the result $\mathscr{L}^{-1}\left\{\frac{s}{(s^2+4)^2}\right\} = \frac{1}{4}t \sin 2t$. Hence, using Equation (26),

$$\mathscr{L}^{-1}\left\{\frac{s^2}{(s^2+4)^2}\right\} = \frac{d}{dt}\left(\frac{1}{4}t\sin 2t\right) = \frac{1}{4}\sin 2t + \frac{t}{2}\cos 2t.$$

When a given function F(s), whose inverse transform is required, is the

ratio of two polynomials in s, the method of partial fractions can be used to express F(s) as the sum of a number of terms whose inverse transforms can be determined from the table. In such cases, if the inverse transform does not involve the singularity functions of Section 2.6, the degree of the denominator must be greater than that of the numerator. In particular, if

$$F(s)=\frac{N(s)}{D(s)},$$

where D(s) is a polynomial of degree *n* with *n* distinct real zeros $s = a_1, a_2, \ldots, a_n$, and N(s) is a polynomial of degree n - 1 or less, there follows (see Problem 15)

$$\frac{N(s)}{D(s)} = \frac{N(a_1)}{D'(a_1)} \frac{1}{s - a_1} + \frac{N(a_2)}{D'(a_2)} \frac{1}{s - a_2} + \dots + \frac{N(a_n)}{D'(a_n)} \frac{1}{s - a_n}$$
$$= \sum_{m=1}^n \frac{N(a_m)}{D'(a_m)} \frac{1}{s - a_m};$$

and hence, from (T12),

$$\mathscr{L}^{-1}\left\{\frac{N(s)}{D(s)}\right\} = \sum_{m=1}^{n} \frac{N(a_m)}{D'(a_m)} e^{a_m t}.$$
(27)

If certain of the zeros of D(s) are repeated or complex, recourse may be had to conventional methods of expansion in partial fractions.

Example 8. To determine
$$\mathscr{L}^{-1}\left\{\frac{s^2+1}{s^3+3s^2+2s}\right\}$$
, we write
 $N(s) = s^2 + 1$, $D(s) = s^3 + 3s^2 + 2s = s(s+1)(s+2)$.

With $a_1 = 0$, $a_2 = -1$, $a_3 = -2$, there follows

$$N(a_1) = 1,$$
 $D'(a_1) = 2,$
 $N(a_2) = 2,$ $D'(a_2) = -1,$
 $N(a_3) = 5,$ $D'(a_3) = 2,$

and Equation (27) gives

$$\mathscr{L}^{-1}\left\{\frac{s^2+1}{s^3+3s^2+2s}\right\} = \frac{1}{2} - 2 e^{-t} + \frac{5}{2} e^{-2t}.$$

Example 9. To determine $\mathscr{L}^{-1}\left\{\frac{1}{(s+1)(s^2+1)}\right\}$, we first assume an expansion of the form

$$\frac{1}{(s+1)(s^2+1)} = \frac{A}{s+1} + \frac{Bs+C}{s^2+1}.$$

After clearing fractions, we require the equation to be an identity and obtain $A = -B = C = \frac{1}{2}$. Hence

$$\frac{1}{(s+1)(s^2+1)} = \frac{1}{2} \left[\frac{1}{s+1} + \frac{1}{s^2+1} - \frac{s}{s^2+1} \right]$$

and the use of (T12,15,16) gives the inverse transform $\frac{1}{2}(e^{-t} + \sin t - \cos t)$.

The usefulness of (T25,26) in determining inverse transforms should not be overlooked.

Example 10. To determine $\mathscr{L}^{-1}\left\{\frac{s}{s^2+4s+5}\right\}$, we first write

$$\frac{s}{s^2+4s+5}=\frac{s}{(s+2)^2+1}=\frac{(s+2)-2}{(s+2)^2+1}.$$

Pairs (T25,26) then give the required inverse transform

$$e^{-2t}\left(\cos t - 2\sin t\right).$$

2.8. Applications to linear differential equations with constant coefficients. It follows from the property (T5) and its special cases (T3,4) that any ordinary



Figure 2.4

linear differential equation with constant coefficients, with prescribed initial conditions at t = 0, can be transformed immediately to a linear algebraic equation determining the transform of the required solution, provided that the right-hand member of the equation has a transform. The solution then is to be obtained as the inverse of the transform so determined.

If the right-hand member is of exponential order, the same will be true of the solution and of those derivatives whose transforms are involved, and all the relations of Table 1 are appropriate.

We take as a simple example the case of *forced vibration* of a mass m attached to a spring with spring constant k. (That is, the force exerted on the free end of the spring is assumed to be proportional to its displacement x from the position of equilibrium, the constant of proportionality being k.) If the applied force is f(t) and if *no damping is present* (Figure 2.4), the differential equation of motion is

$$m\frac{d^2x}{dt^2} + k x = f(t).$$
 (28)

Furthermore, if the mass is assumed to be at rest at equilibrium when t = 0, the initial conditions

$$x(0) = \frac{dx(0)}{dt} = 0$$
 (29)

must be satisfied. Denoting the transforms of x(t) and f(t) by $\bar{x}(s)$ and $\bar{f}(s)$, respectively, the transform of Equation (28) becomes merely

$$ms^2 \, \bar{x} + k \, \bar{x} = \bar{f}.$$

Thus, if we write

 $\omega_0^2=\frac{k}{m},$ (30)

the transform of the required solution is

$$\bar{x} = \frac{1}{m} \frac{\bar{f}}{s^2 + \omega_0^2}.$$
 (31)

Since $\frac{1}{s^2 + \omega_0^2}$ is the transform of $\frac{\sin \omega_0 t}{\omega_0}$, this product can be considered as the product of the transforms of $\frac{\sin \omega_0 t}{m\omega_0}$ and f(t), and hence use of the con-

volution (T10) gives the solution

$$x = \frac{1}{m\omega_0} \int_0^t f(u) \sin \omega_0(t-u) \, du, \qquad (32)$$

in terms of an arbitrary force function f(t).

However, in place of specializing this general form, it is generally more convenient in specific cases to derive the required solution directly from Equation (31). We consider several cases of interest.

(1) Suppose that an instantaneous impulse of magnitude I is applied just after the time t = 0. Then $f(t) = I \cdot \delta(t)$ and $\tilde{f}(s) = I \cdot 1 = I$. Hence we have

$$\bar{x} = \frac{I}{m} \frac{1}{s^2 + \omega_0^2}, \qquad x = \frac{I}{m\omega_0} \sin \omega_0 t \qquad (t > 0).$$
 (33)

Thus the motion in this case is a sinusoidal vibration of amplitude $I/m\omega_0$ and angular frequency ω_0 , following the application of the impulse; ω_0 is known as the natural frequency of the system. It should be noticed that here the initial condition $\frac{dx(0)}{dt} = 0$ is apparently not fulfilled. However, the velocity cannot vanish when the impulse is applied, since the momentum mv = I must be imparted by the impulse, in accordance with Newton's laws of motion. Here we may suppose that x and $v = \frac{dx}{dt}$ are zero throughout an infinitesimal

interval following the time t = 0, and that on the subsequent application of the impulse the velocity abruptly takes on the value I/m and a sinusoidal motion ensues. Interpretations of this general nature are frequently necessary in dealing with the idealized "singularity functions."

(2) If a sinusoidal force $f(t) = A \sin \omega t$ is applied, there follows

$$\bar{x}=\frac{A\omega}{m}\frac{1}{(s^2+\omega_0^2)(s^2+\omega^2)},$$

or, expanding in terms of partial fractions,

$$\bar{x} = \frac{A\omega}{m(\omega^2 - \omega_0^2)} \left[\frac{1}{s^2 + \omega_0^2} - \frac{1}{s^2 + \omega^2} \right].$$

Hence $x = \frac{A\omega}{m(\omega^2 - \omega_0^2)} \left[\frac{\sin \omega_0 t}{\omega_0} - \frac{\sin \omega t}{\omega} \right]$

 $x = \frac{A}{m\omega_0(\omega^2 - \omega_0^2)} \left[\omega \sin \omega_0 t - \omega_0 \sin \omega t\right].$ (34)

Thus, if $\omega \neq \omega_0$, the motion is compounded of two modes of vibration, one (the *natural mode*) at the natural frequency ω_0 , and the other (the *forced mode*) at the frequency of the imposed force. In case the system is excited at its natural frequency ($\omega = \omega_0$), the motion can be determined by considering the limiting form of Equation (34) as $\omega \rightarrow \omega_0$, or, more easily, by noticing that in this case

$$\bar{x} = \frac{A\omega_0}{m} \frac{1}{(s^2 + \omega_0^2)^2}$$

Hence we obtain, from (T21),

$$x = \frac{A}{2m\omega_0^2} (\sin \omega_0 t - \omega_0 t \cos \omega_0 t).$$
(35)

Thus the last term of Equation (35) shows that when the exciting frequency equals the natural frequency, the amplitude of the oscillations increases indefinitely with time. This is the case of resonance.

Similarly, if $f(t) = A \cos \omega_0 t$, there follows

$$x = \frac{A}{2m\omega_0} t \sin \omega_0 t.$$

(3) If a constant force f(t) = A is applied when t > 0, there follows

$$\bar{x} = \frac{A}{m} \frac{1}{s(s^2 + \omega_0^2)} = \frac{A}{m\omega_0^2} \left(\frac{1}{s} - \frac{s}{s^2 + \omega_0^2} \right)$$
(5),

and hence, from (T11,16

$$x = \frac{A}{m\omega_0^2} (1 - \cos \omega_0 t) \tag{36}$$

or

Thus, in this case, the mass oscillates with its natural frequency between the points x = 0 and $x = \frac{2A}{m\omega_0^2} = \frac{2A}{k}$, when damping is absent.

(4) If constant force is applied only over the interval $0 < t < t_0$, and no force acts when $t > t_0$, there follows $\overline{f} = \frac{A}{s} (1 - e^{-st_0})$, and hence

$$\bar{x} = \frac{A}{m} \frac{1 - e^{-st_0}}{s(s^2 + \omega_0^2)} = \frac{A}{m} \left[\frac{1}{s(s^2 + \omega_0^2)} - \frac{e^{-st_0}}{s(s^2 + \omega_0^2)} \right].$$

The inverse transform of the first term is given by Equation (36) and, in view of (T9), the inverse of the second term is zero when $t < t_0$ and is obtained by replacing t by $t - t_0$ in (36) when $t > t_0$. Hence we have, when $0 < t < t_0$,

$$x = \frac{A}{m\omega_0^2} (1 - \cos \omega_0 t); \qquad (37a)$$

and, when $t > t_0$,

$$x = \frac{A}{m\omega_0^2} \{ (1 - \cos \omega_0 t) - [1 - \cos \omega_0 (t - t_0)] \}$$

= $\frac{A}{k} [\cos \omega_0 (t - t_0) - \cos \omega_0 t]$
= $\frac{2A}{k} (\sin \frac{1}{2} \omega_0 t_0) \sin \omega_0 (t - \frac{1}{2} t_0).$ (37b)

Thus, while the force acts $(0 < t < t_0)$, the mass oscillates at its natural frequency, with amplitude A/k, about the point x = A/k; however, after the force is removed $(t > t_0)$, the mass oscillates about the point of equilibrium (x = 0), at the same frequency, but with an amplitude $\frac{2A}{k} \sin \frac{1}{2}\omega_0 t_0$. If $t_0 = \frac{2\pi}{\omega_0} = T$, where T is the period of the natural mode of vibration, then x = 0 when $t \ge t_0$, so that the mass returns to its equilibrium position as the force is removed, and then remains at that position.

It is seen that in the preceding example, and in similar cases, the use of tables permits the determination of the transform of the solution by purely algebraical methods. This is true, however, only in cases when the coefficients of the linear differential equation are constants, and the usefulness of the present methods is mainly restricted, in such applications, to such cases

The use of Laplace transforms is particularly advantageous in the solution of initial-value problems associated with sets of simultaneous linear equations. We illustrate the procedure by considering an example. We require the solution of the simultaneous equations

$$\frac{dx}{dt} - y = e^{t}$$

$$\frac{dy}{dt} + x = \sin t$$
(38)

which satisfies the conditions

$$x(0) = 1, \quad y(0) = 0.$$
 (39)

The transforms of (38) satisfying (39) are

$$s \, \bar{x} - \bar{y} = \frac{1}{s-1} + 1$$

 $\bar{x} + s \, \bar{y} = \frac{1}{s^2 + 1}$

from which we obtain, algebraically,

$$\bar{x} = \frac{s}{(s-1)(s^2+1)} + \frac{s}{s^2+1} + \frac{1}{(s^2+1)^2},$$

$$\bar{y} = -\frac{1}{(s-1)(s^2+1)} - \frac{1}{s^2+1} + \frac{s}{(s^2+1)^2}.$$

If the first terms on the right-hand sides of these equations are expanded in partial fractions, there follows

$$\bar{x} = \frac{1}{2} \left[\frac{1}{s-1} + \frac{1}{s^2+1} + \frac{s}{s^2+1} + \frac{2}{(s^2+1)^2} \right]
\bar{y} = \frac{1}{2} \left[-\frac{1}{s-1} - \frac{1}{s^2+1} + \frac{s}{s^2+1} + \frac{2s}{(s^2+1)^2} \right]$$
(40)

and reference to Table 1 gives the required solution,

$$x = \frac{1}{2}(e^{t} + 2\sin t + \cos t - t\cos t) \\ y = \frac{1}{2}(-e^{t} - \sin t + \cos t + t\sin t)$$
(41)

To illustrate the existence of exceptional cases which may arise in connection with simultaneous differential equations, we next attempt to find a solution of the equations

$$\left. \frac{dx}{dt} + y = 0 \\
\frac{d^2x}{dt^2} + \frac{dy}{dt} + y = e^t \right\}$$
(42)

satisfying the conditions

$$x(0) = 1, \quad x'(0) = 0, \quad y(0) = 0.$$
 (43)

The transformed equations are

$$s \, \bar{x} + y = 1,$$

 $s^2 \, \bar{x} + (s+1) \, \bar{y} = s + \frac{1}{s-1}$

from which there follow

$$\bar{x} = \frac{2}{s} - \frac{1}{s-1}, \qquad \bar{y} = \frac{1}{s-1}.$$
 (44)

The inverse transforms of these expressions are then

$$x = 2 - e^t, \qquad y = e^t. \tag{45}$$

However, these solutions do not satisfy the last two of the prescribed initial conditions (43). It is readily shown, by methods of Chapter 1, that the most general solution of (42) is of the form

$$x = C - e^t$$
, $y = e^t$

where C is an arbitrary constant. Hence only the initial value of x is arbitrary, and the problem as stated does not possess a solution.

This example shows that, in the case of simultaneous equations, although the method of Laplace transforms will yield the correct solution if it exists, it may also supply an erroneous solution (which fails to satisfy certain prescribed initial conditions) if no true solution exists. Thus, in doubtful cases, the satisfaction of initial conditions should be checked.

2.9. The Gamma function. In calculating the transform of t^n , where n > -1 but n is not necessarily an integer, we encounter a function, known as the Gamma function, which also occurs frequently in many other applications. In this section we investigate certain properties of this function.

If, in the integral defining the transform of t^n ,

$$\mathscr{L}\left\{t^{n}\right\}=\int_{0}^{\infty}e^{-st}t^{n}dt \qquad (n>-1),$$

we introduce a new variable of integration by setting st = x, there follows*

$$\mathscr{L}\left\{t^{n}\right\} = \frac{1}{s^{n+1}} \int_{0}^{\infty} e^{-x} x^{n} dx \qquad (n > -1).$$
(46)

The integral appearing in Equation (46) depends only upon n. Although it cannot be expressed in terms of elementary functions of n, the same integral

* The restriction n > -1 is necessary to ensure the convergence of the integral. If $n \leq -1$, the function t^n does not have a Laplace transform, as here defined.

Function	f(t)	af(t) + bg(t)	$\frac{df(t)}{dt}$	$\frac{d^2 f(t)}{dt^2}$	$\frac{d^n f(t)}{dt^n}$	<i>n</i> times <i>n</i> times	$\int_0^t \cdots \int_0^t f(t) dt \cdots dt$	$t_u f(t)$	$e^{\alpha t}f(t)$	$\begin{cases} f(t-a) & (t>a) \\ 0 & (t$	$\int_0^t f(t-u)g(u) du = \int_0^t f(u)g(t-u) du$	1	e-al	$\frac{1}{a-b}(e^{-bt}-e^{-at})$
Transform	$f(s) = \mathcal{L}\{f(t)\} = \int_{-\infty}^{\infty} e^{-st} f(t) dt$	$af(s) + b\tilde{g}(s)$	sf(s) - f(0)	$s^2 f(s) - s f(0) - \frac{df(0)}{dt}$	$s^n f(s) - \sum_{k=1}^n s^{n-k} \frac{d^{k-1}f(0)}{dt^{k-1}}$	1	$\frac{1}{s^n}f(s)$	$(-1)^n rac{d^n f(s)}{ds^n}$	f(s-a)	$e^{-as}f(s)$ $(a \ge 0)$	f(s) g(s)		$\frac{1}{s+a}$	$\frac{1}{(s+a)(s+b)}$
	TI	T2	T3	T4	TS		T6	T7	T8	T 9	T10	T11	T12	T 13

Table 1. Laplace Transforms

$\frac{1}{b-a}(be^{-bt}-ae^{-at})$	sin at	cos at	sınh <i>at</i>	cosh at	t sin at	t cos at	$\sin at - at \cos at$	t sinh at	t cosh at	at cosh at – sınh at	e -bt sin at	e ^{-bt} cos at	sin at cosh at - cos at suh at	sın at sınh at
$\frac{s}{(s+a)(s+b)}$	$\frac{a}{s^2 + a^2}$	$\frac{s}{s^2 + a^2}$	$\frac{a}{s^2-a^2}$	$\frac{5}{5^2-a^2}$	$\frac{2as}{(s^2+a^2)^2}$	$\frac{s^2 - a^2}{(s^2 + a^2)^2}$	$\frac{2a^3}{(s^2+a^2)^2}$	$\frac{2as}{(s^2-a^2)^2}$	$rac{S^2 + a^2}{(S^2 - a^2)^2}$	$\frac{2a^3}{(s^2-a^2)^2}$	$\frac{a}{(s+b)^2+a^2}$	$\frac{s+b}{(s+b)^2+a^2}$	$\frac{4a^3}{s^4+4a^4}$	$\frac{2a^{2}s}{s^{4}+4a^{4}}$
T14	T15	T16	T17	T18	T19	T20	T21	T22	T23	T24	T25	T26	T27	T28

ued) Function	$\sinh at - \sin at$	$\cos \alpha$	$\delta(t)$ $\delta(t-t_1)$	$\delta'(t) = \delta'(t)$	$\frac{t^{n-1}}{(n-1)!} \qquad (n>0)$	t^n $(n > -1)$	$\frac{t^{n-1}e^{-at}}{(n-1)!} \qquad (n>0)$	$\frac{(n-1)-at}{(n-1)!}t^{n-2}e^{-at} \qquad (n>1)$	$\frac{1}{2^{n-1}(n-1)!} \left[\sqrt{\frac{\pi}{2}} (at)^{n-1/2} J_{n-1/2}(at) \right] (n > 0)$	$\frac{at}{2^n(n-1)!}\left[\sqrt{\frac{\pi}{2}}\left(at\right)^{n-3/2}J_{n-3/2}(at)\right] (n>\frac{1}{2})$	$\frac{1}{2^{n-1}(n-1)!} \left[\sqrt{\frac{\pi}{2}} (at)^{n-1/2} I_{n-1/2}(at) \right] (n > 0)$	$\frac{at}{2^{n}(n-1)!}\left[\sqrt{\frac{\pi}{2}}(at)^{n-3/2}I_{n-3/2}(at)\right] (n > \frac{1}{2})$
Table 1.—(Contin Transform	T29 $\frac{2a^3}{s^4 - a^4}$	T30 $\overline{s^4 - a^4}$	T31 1 1 T32 e^{-st_1}	T33 <i>S</i> T34 <i>S e^{-st}1</i>	T35 $\frac{1}{s^n}$	T35a $\frac{n!}{s^{n+1}}$	T36 $\frac{1}{(s+a)^n}$	T37 $\frac{s}{(s+a)^n}$	T38 $\frac{a^{2n-1}}{(s^2+a^2)^n}$	T39 $\frac{a^{2n-2}s}{(s^2+a^2)^n}$	T40 $\frac{a^{2n-1}}{(s^2-a^2)^n}$	T41 $\frac{a^{2n-2}S}{(S^2-a^2)^n}$

76

TADIC 2. DOSCI FUNCTIONS OF OTACL FIAN AN OWN INTEGE	$\sqrt{\frac{\pi}{2}} x^m J_m(x) \qquad \qquad$	$(\cos x)/x \qquad (\cosh x)/x$	sin x sinh x	$\sin x - x \cos x$	$(3 - x^2) \sin x - 3x \cos x$ $(3 + x^2) \sinh x - 3x \cosh x$	$(15 - 6x^2) \sin x - (15 - x^2)x \cos x$ $(15 + x^2)x \cosh x - (15 + 6x^2) \sinh x$	$(105 - 45x^2 + 1) \sin x - (105 - 10x^2)x \cos x$ $(105 + 45x^2 + 1) \sinh x - (105 + 10x^2)x \cosh x$	Recurrence Formulas	$x^{m+1}J_{m+1}(x) = 2m[x^m J_m(x)] - x^2[x^{m-1} J_{m-1}(x)] \qquad x^{m+1} I_{m+1}(x) = -2m[x^m I_m(x)] + x^2[x^{m-1} I_{m-1}(x)]$
	<i>w</i>	-1/2	1/2	32	52	72	92		

Table 2. Bessel Functions of Order Half an Odd Integer

with *n* (inconveniently) replaced by n - 1 is a tabulated function which occurs frequently in practice and is known as the *Gamma function* of *n*, written $\Gamma(n)$:

$$\Gamma(n) = \int_0^\infty e^{-x} x^{n-1} dx \qquad (n > 0).$$
 (47)

With this notation, Equation (46) can be written in the form

$$\mathscr{L}\left\{t^{n}\right\} = \frac{\Gamma(n+1)}{s^{n+1}} \qquad (n > -1).$$
(48)

A comparison of Equation (48) with the result of Example 5, Section 2.4, shows that

$$\Gamma(n+1) = n! \tag{49}$$

if n is a positive integer, and that

$$\Gamma(1) = 0! = 1.$$
 (50)

Thus it is seen that, if n > -1, $\Gamma(n + 1)$ is a continuous function of n which takes on the value n! when n is a positive integer or zero. For this reason, the Gamma function is often referred to as the generalized factorial function.

Making use of an integration by parts, we obtain the result

$$\Gamma(n+1) = \int_0^\infty e^{-x} x^n \, dx = -x^n \, e^{-x} \Big|_0^\infty + n \int_0^\infty e^{-x} x^{n-1} \, dx$$
$$= n \int_0^\infty e^{-x} x^{n-1} \, dx \qquad (n > 0),$$

from which there follows

$$\Gamma(n+1) = n \Gamma(n) \qquad (n>0). \tag{51}$$

Inductive reasoning then leads to the formula

$$\Gamma(n+N) = (n+N-1)(n+N-2)\cdots(n+1)n\ \Gamma(n) \qquad (n>0), \quad (52)$$

where N is any positive integer. Also, if n is replaced by n - 1, Equation (51) can be written in the alternative form

$$\Gamma(n-1) = \frac{\Gamma(n)}{n-1} \qquad (n>1). \tag{53}$$

Values of $\Gamma(n)$, or of $\log_{10} \Gamma(n)$, are commonly tabulated for the interval $1 \le n \le 2$.* Equation (52) can then be used to evaluate the Gamma function for arguments greater than 2, since any such argument differs from a value in the tabulated range by some positive integer N. If we write x = n + N and replace n by x_0 , where $1 \le x_0 \le 2$, Equation (52) becomes

$$\Gamma(x) = (x-1)(x-2)\cdots(x_0+1)x_0 \ \Gamma(x_0)$$
(x > 2, 1 ≤ x₀ ≤ 2). (54)

* A table of values of the Gamma function itself, in this interval, is presented in the Appendix.

Also, Equation (53) serves to determine values of the Gamma function for arguments between zero and unity.

For negative values of *n*, the function $\Gamma(n)$ is not defined by Equation (47), since the integral does not exist. However, it is conventional to extend the definition in such cases by requiring that the *recurrence formula* (51) hold also for negative values of *n*. A different method of definition yields $\Gamma(n)$ for all complex values of *n* except for zero and negative integers, at which points $\Gamma(n)$ does not remain finite. Since, for any negative value of *n* which is not an integer, there exists a positive integer *N* such that n + N is in the tabulated range (1 < n + N < 2), we may then replace n + N by x_0 and *n* by *x* in (52), where $1 < x_0 < 2$, to obtain

$$\Gamma(x_0) = (x_0 - 1)(x_0 - 2) \cdots (x + 1) x \Gamma(x)$$

or, solving for $\Gamma(x)$,

$$\Gamma(x) = \frac{\Gamma(x_0)}{x(x+1)(x+2)\cdots(x_0-2)(x_0-1)}$$
(x < 1, 1 < x₀ < 2). (55)

Equations (54) and (55) thus serve to determine values of the Gamma function for real arguments outside the tabulated range. It should be noticed, however, that since the denominator of Equation (55) vanishes when x is zero or a negative integer, the Gamma function is not defined for these values, and becomes infinite as these values are approached (see Figure 2.5).

It will be convenient in later work to use the notation n! even in cases when n is not a positive integer or zero, with the convention that in such cases n! is defined by $\Gamma(n + 1)$.

The value of $\Gamma(\frac{1}{2})$ is of particular interest. A well-known but quite indirect method of determining this value is now presented. From the definition (47), we have

$$\Gamma(\frac{1}{2})=\int_0^\infty e^{-z}\,z^{-1/2}\,dz.$$

With the change in variables $z = x^2$, this integral becomes

$$\Gamma(\frac{1}{2}) = 2 \int_0^\infty e^{-x^2} dx.$$
 (56)

If the right-hand side of Equation (56) is multiplied by itself, and if the variable of integration is replaced by y in one factor, there follows

$$\left[\Gamma(\frac{1}{2})\right]^2 = 4\left(\int_0^\infty e^{-x^2}\,dx\right)\left(\int_0^\infty e^{-y^2}\,dy\right) = 4\,\int_0^\infty\int_0^\infty e^{-(x^2+y^2)}\,dx\,dy.$$

This double integral represents the volume under the surface $z = e^{-(x^2+y^2)}$ in the first quadrant ($x \ge 0, y \ge 0$). Changing to polar coordinates, we obtain

$$\left[\Gamma\left(\frac{1}{2}\right)\right]^{2} = 4\left\{\int_{0}^{\pi/2}\int_{0}^{\infty}e^{-r^{2}} r \, dr\right\} d\theta = 4\frac{\pi}{2}\frac{e^{-r^{2}}}{-2}\Big|_{0}^{\infty} = \pi.$$

Thus, finally,

$$\Gamma(\frac{1}{2}) = \sqrt{\pi}.$$
(57)

From Equation (56) we also have the useful result

$$\int_0^\infty e^{-x^2} dx = \frac{1}{2}\sqrt{\pi}.$$
 (58)

We include without proof the formula

$$\Gamma(x)\,\Gamma(1-x) = \frac{\pi}{\sin \pi x}\,,\tag{59}$$

which can be shown to be valid for nonintegral values of x and which is of some use in applications of the Gamma function.



Figure 2.5

It can be shown that the Gamma function is defined alternatively by the limit

$$\Gamma(x) = \lim_{n \to \infty} \frac{n!}{x(x+1)\cdots(x+n)} n^x.$$

Hence there follows also

$$\lim_{n \to \infty} \frac{n! n^x}{\Gamma(x+n+1)} = 1.$$
 (60)

Thus in limiting operations involving Gamma functions, when $n \to \infty$, the expression $\Gamma(x + n + 1)$ can be replaced by its approximation $n! n^x$. We indicate this fact by the notation

$$(n+x)! = \Gamma(x+n+1) \sim n! \, n^x \qquad (n \to \infty), \tag{61}$$

where the symbol \sim is to be interpreted as indicating that (61) implies (60).

A further limit of some interest is of the form

$$\lim_{n \to \infty} \frac{\Gamma(n+1)}{\sqrt{2\pi} n^{n+(1/2)} e^{-n}} = 1$$
 (62)

which we can write symbolically

$$n! = \Gamma(n+1) \sim \sqrt{2\pi} \, n^{n+(1/2)} \, e^{-n} \qquad (n \to \infty). \tag{63}$$

If n is a positive integer, this approximation is known as the Stirling formula for the factorial.

Proof of these relations is beyond the scope of this chapter.

REFERENCES

- 1. Campbell G. A. and R. M. Foster, *Fourier Integrals for Practical Applications*, D. Van Nostrand Company, Inc., Princeton N.J., 1947.*
- 2. Carslaw, H. S., and J. C. Jaeger, *Operational Methods in Applied Mathematics*, Oxford University Press, Fair Lawn, N.J., 1941.
- 3. Churchill, R. V., Operational Mathematics, 2nd ed., McGraw-Hill Book Company, Inc., New York, 1958.
- 4. Erdélyi, A., ed., Tables of Integral Transforms, 2 vols., McGraw-Hill Book Company, Inc., New York, 1954.
- 5. Gardner, M. F., and J. L. Barnes, *Transients in Linear Systems*, John Wiley & Sons, Inc., New York, 1942.
- 6. Le Page, W. R., Complex Variables and the Laplace Transform for Engineers, McGraw-Hill Book Company, Inc., New York, 1961.
- 7. McLachlan, N. W., Complex Variable Theory and Transform Calculus, 2nd ed., Cambridge University Press, New York, 1953.
- 8. Widder, D. V., The Laplace Transform, Princeton University Press, Princeton, N.J., 1941.

* This reference contains 763 pairs, most of which can be interpreted as functions and their Laplace transforms. For this purpose, if the variable g is replaced by t, then the column headed "Coefficient G(g)" lists the function f(t) in the present notation and the column headed "Coefficient F(f)" lists the corresponding Laplace transform $\mathscr{L}{f(t)} = \tilde{f}(s)$.

PROBLEMS

Section 2.1

1. (a) Obtain the solution of the equation

$$\frac{dy}{dt} - \alpha y = e^{at}$$

for which $y(0) = y_0$, by the method of Section 2.1. Assume that $\alpha \neq a$. (b) Verify the solution so obtained.

2. (a) Obtain the solution of the equation

$$\frac{dy}{dt} - ay = e^{at}$$

for which $y(0) = y_0$, by considering the limit of the solution of Problem 1 as $\alpha \rightarrow a$.

(b) Verify the solution so obtained.

Section 2.2

3. Find the Laplace transform of each of the following functions, by direct integration:

(a)
$$e^{at} \cos kt$$
,
(b) $t^n e^{-at}$ (n a positive integer),
(c) $\begin{cases} \sin t & (0 < t < \pi) \\ 0 & (t > \pi), \end{cases}$
(d) $\begin{cases} 0 & (0 < t < a) \\ 1 & (a < t < b) \\ 0 & (t > b). \end{cases}$

Section 2.3

- 4. Find the Laplace transform of each of the following functions:
 - (a) t^3 , (b) $t^2 e^{-3t}$, (c) $\cos at \sinh at$, (d) $t e^t \sin 2t$, (e) $t^2 \sin at$, (f) $e^{at} \cosh bt$.
- 5. Find the Laplace transform of each of the following functions:

(a)
$$\frac{d^3 f(t)}{dt^3}$$
, (b) $t e^t f(t)$,
(c) $\sum_{n=0}^{N} a_n t^n$, (d) $\sum_{n=0}^{N} a_n \cos nt$

6. The Laguerre polynomial of degree n is defined by the equation

$$L_n(t) = e^t \frac{d^n}{dt^n} (t^n e^{-t}).$$

Prove that

$$\mathscr{L}\left\{L_n(t)\right\} = \frac{n!}{s} \left(\frac{s-1}{s}\right)^n.$$

7. Prove that if $f(s) = \mathcal{L}{f(t)}$, and if a > 0, then also

(a)
$$\mathscr{L}\left\{f(at)\right\} = \frac{1}{a}f\left(\frac{s}{a}\right)$$
, (b) $\mathscr{L}^{-1}\left\{f(as)\right\} = \frac{1}{a}f\left(\frac{t}{a}\right)$.

8. Let f(t) = F(t) when 0 < t < a, and let f(t) be *periodic*, of period *a*, so that f(t + a) = f(t). By writing

$$\mathscr{L} \{ f(t) \} = \int_0^a e^{-st} f(t) \, dt + \int_a^{2a} e^{-st} f(t) \, dt + \int_{2a}^{3a} e^{-st} f(t) \, dt + \cdots$$

and transforming each integral in such a way that in each case the range of integration is (0,a), show that

$$\mathscr{L}\{f(t)\} = \int_0^a e^{-st} F(t) dt [1 + e^{-sa} + e^{-2sa} + \cdots],$$

and hence, when s > 0,

$$\mathscr{L}\left\{f(t)\right\} = \frac{\int_0^a e^{-st} F(t) dt}{1 - e^{-sa}}$$

9. Apply the result of Problem 8 to the "square-wave function" for which F(t) = 1 when $0 < t < \frac{a}{2}$ and F(t) = -1 when $\frac{a}{2} < t < a$. Show that the transform of this function is

$$\frac{(1-e^{-as/2})^2}{s(1-e^{-as})} = \frac{1}{s} \frac{1-e^{-as/2}}{1+e^{-as/2}} = \frac{1}{s} \tanh \frac{as}{4}.$$

10. Show that if f(t) is the "square-wave function" of Problem 9, then $\int_0^t f(u) du$ is a "triangular-wave function." Sketch it and give the expression for its transform.

11. (a) Show that, if $f(s) = \mathcal{L} \{ f(t) \}$, and if an interchange of order of integrations is valid, then

$$\int_{s}^{\infty} f(v) \, dv = \mathscr{L}\left\{\frac{f(t)}{t}\right\}$$

[The result is valid, for s sufficiently large, whenever f(t)/t has a transform.]

(b) Use this result to deduce the transforms

$$\mathscr{L}\left\{\frac{\sin t}{t}\right\} = \cot^{-1} s, \qquad \mathscr{L}\left\{\frac{1-e^{-t}}{t}\right\} = \log\left(\frac{s}{s+1}\right).$$

12. (a) By formally setting s = 0 in the result of Problem 11(a), obtain the formula

$$\int_0^\infty f(s) \, ds = \int_0^\infty \frac{f(t)}{t} \, dt$$

[The result is valid when the integral on the right exists.]

(b) Use the result of part (a) to obtain the evaluations

$$\int_0^\infty \frac{\sin t}{t} dt = \frac{\pi}{2}, \qquad \int_0^\infty \frac{e^{-at} - e^{-bt}}{t} dt = \log \frac{b}{a} \qquad (a, b > 0).$$

(c) Show that if $f(t) = e^t$, then neither side of the relation of part (a) exists, whereas if $f(t) = e^t \sin t$, then the left-hand member exists but the right-hand member does not.

13. By applying the property of Equation (16) to the result of Problem 11, obtain the formula

$$\mathscr{L}\left\{\int_0^t \frac{f(u)}{u}\,du\right\} = \frac{1}{s}\int_s^\infty f(v)\,dv.$$

Also, assuming that the result of Problem 12 also applies, obtain the formula

$$\mathscr{L}\left\{\int_t^\infty \frac{f(u)}{u}\,du\right\} = \frac{1}{s}\int_0^s f(v)\,dv.$$

[In each case, the given formula is valid, for s sufficiently large, whenever the lefthand member exists. The integral $\int_0^\infty f(t)/t \, dt$ need not exist.]

14. Assuming the results of Problem 13, show that

$$\mathscr{L}\left\{Si(t)\right\} = \frac{1}{s}\cot^{-1}s, \qquad \mathscr{L}\left\{Ci(t)\right\} = \frac{1}{s}\log\frac{1}{\sqrt{s^2+1}},$$
$$\mathscr{L}\left\{Ei(-t)\right\} = \frac{1}{s}\log\frac{1}{s+1},$$

and

where

$$Si(t) = \int_0^t \frac{\sin u}{u} \, du, \quad Ci(t) = \int_t^\infty \frac{\cos u}{u} \, du, \quad Ei(-t) = \int_t^\infty \frac{e^{-u}}{u} \, du.$$

Section 2.4

15. Let $\frac{N(s)}{D(s)}$ denote the ratio of two polynomials, with no common factors, such that the degree *n* of D(s) is greater than that of N(s), and suppose that D(s) has *n* distinct real zeros $s = a_1, a_2, \ldots, a_n$. Show that the coefficients in the partial-fraction expansion

$$\frac{N(s)}{D(s)} = \frac{A_1}{s - a_1} + \frac{A_2}{s - a_2} + \cdots + \frac{A_n}{s - a_n} = \sum_{m=1}^n \frac{A_m}{s - a_m}$$

are determined by the equations

$$A_m = \lim_{s \to a_m} (s - a_m) \frac{N(s)}{D(s)} = \frac{N(a_m)}{D'(a_m)}.$$

Hence show that in this case

$$\mathscr{L}^{-1}\left(\frac{N(s)}{D(s)}\right) = \sum_{m=1}^{n} \frac{N(a_m)}{D'(a_m)} e^{a_m t}.$$

16. If $\mathscr{L}{f(t)} = \dot{f}(s)$ and if

$$f(t) = A_0 + A_1t + A_2t^2 + \cdots + A_nt^n + \cdots$$

Problems

(in some interval about t = 0) and

$$f(s) = \frac{B_0}{s} + \frac{B_1}{s^2} + \frac{B_2}{s^3} + \cdots + \frac{B_n}{s^{n+1}} + \cdots$$

(for sufficiently large values of s), show that

$$A_n=\frac{B_n}{n!}\,.$$

17. By using the ratio test, show that if $\lim_{n \to \infty} \left| \frac{B_{n+1}}{B_n} \right| = s_0$, the second series in Problem 16 converges when $s > s_0$. Deduce that in this case $\lim_{n \to \infty} \left| \frac{A_{n+1}}{A_n} \right| = 0$, so that the first series in Problem 16 then converges for all values of t.

18. Use the results of Problems 16 and 17 to find the inverse transform of each of the following functions as a power series in t:

19. Verify Equation (22) in each of the following cases:

(a)
$$f(t) = \cos at$$
,
(b) $f(t) = \sinh at$,
(c) $f(s) = \frac{s}{s^2 - a^2}$,
(d) $f(s) = \frac{2s + 1}{s^2 + 2s + 2}$.

20. By starting with Equation (15a) and considering the limiting form as $s \rightarrow 0$, obtain the relation

$$\lim_{s\to 0} sf(s) = f(0) + \lim_{s\to 0} \int_0^\infty e^{-st} f'(t) dt$$

and, by formally taking the limit on the right under the integral sign, obtain the result

$$\lim_{s\to 0} sf(s) = f(\infty)$$

where $f(\infty) = \lim_{t \to \infty} f(t)$. [Compare Equation (22). The result is valid when the integral $\int_0^\infty f'(t) dt$ exists. In particular, $\lim_{t \to \infty} f(t)$ must exist.]

21. (a) Show that the result of Problem 20 is *not* valid, in particular, in the cases for which f(s) is given by

$$\frac{1}{s-1}$$
, $\frac{1}{s^2+1}$, $\frac{1}{s(s^2-3s+2)}$

(b) Show that the result of Problem 20 is valid, in particular, in the cases for which f(s) is given by

$$\frac{1}{s}$$
, $\frac{1}{s+1}$, $\frac{1}{s(s^2+3s+2)}$

[The result of Problem 20 occasionally is stated without qualifying restrictions. The above examples show that some care should be taken in applying it. In practical situations, it can be used if the existence of $\lim_{t\to\infty} f(t)$ is assured, from physical considerations or otherwise.]

Section 2.5

22. Determine the convolution of each of the following pairs of functions:

(a) 1, sin at; (b) t, e^{at} ; (c) e^{at}, e^{bt} ; (d) sin at, sin bt.

23. Verify Equation (20) in each of the cases considered in Problem 22.

24. Suppose that y(t) satisfies the integral equation

$$y(t) = F(t) + \int_0^t G(t - u) y(u) du,$$

where F(t) and G(t) are known functions, with Laplace transforms $\overline{F}(s)$ and $\overline{G}(s)$, respectively.

(a) Show that then

$$\bar{y}(s) = \frac{\bar{F}(s)}{1 - \bar{G}(s)} = \bar{F}(s) + \frac{\bar{G}(s)}{1 - \bar{G}(s)} \bar{F}(s).$$

(b) Deduce that the solution of the integral equation is

$$y(t) = F(t) + \int_0^t H(t-u) F(u) \, du,$$

where H(t) is the function whose transform is given by

$$\bar{H}(s) = \frac{\bar{G}(s)}{1 - \bar{G}(s)}$$

(c) Illustrate this result in the special case when $G(t) = e^{-at}$.

25. (a) Show that $s^{-n} f(s)$ is the transform of the function

$$F(t) = \frac{1}{(n-1)!} \int_0^t (t-u)^{n-1} f(u) \, du,$$

when n is a positive integer.

(b) Deduce that

n times *n* times

$$\overbrace{\int_0^t \cdots \int_0^t f(t) \ dt \cdots dt}^{n \text{ times}} = \frac{1}{(n-1)!} \int_0^t (t-u)^{n-1} f(u) \ du.$$

Section 2.6

26. (a) Show that

$$\int_{a}^{b} \delta(u - t_{1}) f(u) \, du = f(t_{1}) \quad \text{if} \quad a < t_{1} < b,$$

Problems

if the integral is defined as the limit (as $\epsilon \to 0$) of the integral in which $\delta(u - t_1)$ is replaced by a function equal to $1/(2\epsilon)$ when $t_1 - \epsilon < u < t_1 + \epsilon$ and equal to zero elsewhere.

(b) In a similar way, obtain the relation

$$\int_{a}^{b} \delta'(u - t_{1}) f(u) \, du = -f'(t_{1}) \quad \text{if} \quad a < t_{1} < b.$$

(c) Show that the convolution of $\delta(t - t_1)$ and f(t) is given by

$$\int_0^t \delta(u-t_1) f(t-u) \, du = \begin{cases} 0 & (t < t_1) \\ f(t-t_1) & (t > t_1). \end{cases}$$

27. (a) If the Heaviside unit step function H(t) is defined such that H(t) = 1 when t > 0 and H(t) = 0 when t < 0, show that

$$\mathscr{L}{H(t)} = \frac{1}{s}, \qquad \mathscr{L}{H(t-t_1)} = \frac{e^{-st_1}}{s}$$

when $t_1 \ge 0$.

(b) Noticing that $\mathscr{L} \{\delta'(t)\} = s \mathscr{L} \{\delta(t)\} = s^2 \mathscr{L} \{H(t)\}$, indicate a sense in which, correspondingly, we may be led to think of $\delta'(t)$ as a formal derivative of $\delta(t)$, and of $\delta(t)$ as a formal derivative of H(t).

Section 2.7

28. Find the inverse Laplace transform of each of the following functions:

(a)
$$\frac{1}{s^2 - 3s + 2}$$
, (b) $\frac{s}{s^2 - 2s + 5}$. (c) $\frac{s + 1}{s^4 + 1}$,
(b) $\frac{2s + 1}{s(s + 1)(s + 2)}$, (e) $\frac{1}{s^2(s^2 + 1)}$, (f) $\frac{e^{-s}}{s + 1}$.

29. Find the inverse Laplace transform of each of the following functions:

(a)
$$\frac{s^2}{(s^2 + a^2)^2}$$
, (b) $\frac{s^2}{(s^2 - a^2)^2}$, (c) $\frac{3s + 1}{(s + 1)(s + 2)(s + 3)}$,
(d) $\frac{1}{s^3 + a^3}$, (e) $\frac{1 - e^{-s}}{s}$, (f) $\frac{1}{(s^2 + a^2)(s^2 + b^2)}$.

30. Find the inverse Laplace transform of each of the following functions:

(a)
$$\frac{s+1}{s^2+2s+2}$$
, (b) $\frac{s^2}{s^4+4a^4}$, (c) $\frac{e^{-\pi s}}{s^2+1}$,
(d) $\frac{s^2}{(s+a)^3}$, (e) $\frac{1}{(s^2+1)^3}$, (f) $\frac{1}{(s^2-1)^3}$.

Section 2.8

31. Solve the following problems by the use of Laplace transforms:

(a)
$$\frac{dy}{dt} + ky = 0$$
, $y(0) = 1$.
(b) $\frac{dy}{dt} + ky = 1$, $y(0) = 0$.
(c) $\frac{dy}{dt} + ky = \delta(t - 1)$, $y(0) = 0$.
(d) $\frac{dy}{dt} + ky = f(t)$, $y(0) = y_0$.

32. Solve the following problems by the use of Laplace transforms:

1.

(a)
$$\frac{d^2y}{dt^2} + 2\frac{dy}{dt} + 2y = 0$$
, $y(0) = 1$, $\frac{dy(0)}{dt} = -1$.
(b) $\frac{d^2y}{dt^2} + 2\frac{dy}{dt} + 2y = 2$, $y(0) = 0$, $\frac{dy(0)}{dt} = 1$.
(c) $\frac{d^2y}{dt^2} + 2\frac{dy}{dt} + 2y = \delta(t-1)$, $y(0) = 1$, $\frac{dy(0)}{dt} = -1$.
(d) $\frac{d^2y}{dt^2} + 2\frac{dy}{dt} + 2y = f(t)$, $y(0) = y_0$, $\frac{dy(0)}{dt} = y'_0$.

33. Solve the following problems by the use of Laplace transforms:

(a)
$$\frac{d^4y}{dt^4} + 4y = 0$$
, $y(0) = \frac{dy(0)}{dt} = \frac{d^2y(0)}{dt^2} = 0$, $\frac{d^3y(0)}{dt^3} = 1$.
(b) $\frac{d^4y}{dt^4} + 4y = 4$, $y(0) = \frac{dy(0)}{dt} = \frac{d^2y(0)}{dt^2} = \frac{d^3y(0)}{dt^3} = 0$.

34. Use Laplace transforms to solve the problem

$$\frac{d^2y}{dt^2} + y = \delta(t-a), \qquad y(0) = 0, \quad y(b) = 0,$$

where 0 < a < b, by first writing $\frac{dy(0)}{dt} = c$, and finally determining c so that the condition y(b) = 0 is satisfied by the inverse transform.

35. In Figure 2.6 a mass *m* is connected to an elastic spring with spring constant k, and to a dashpot which resists motion of the mass with a force numerically equal to *c* times the velocity of motion. The applied external force is indicated by f(t) and the displacement of the mass from equilibrium position by x.

Problems

(a) Show that the differential equation of motion can be put into the form

$$\frac{d^2x}{dt^2} + 2\alpha \frac{dx}{dt} + (\alpha^2 + \beta^2) x = \frac{1}{m} f(t)$$

with the abbreviations

$$\frac{m}{k}=\omega_0^2, \qquad \alpha=\frac{c}{2m}, \qquad \beta=\sqrt{\omega_0^2-\alpha^2}.$$

(b) Assuming that the mass starts from rest at its equilibrium position and

is acted on by a uniform force f_0 , find the resulting motion. Consider separately the cases when β is real and positive, $\beta = 0$, and $\beta = i\gamma$, where γ is real. Discuss the three cases and sketch typical curves representing the displacement as a function of time.

(c) Assuming that the mass starts from rest at the position x = a and that no external force acts, investigate the resulting motion as in part (b).



Figure 2.6

36. (a) Use the method of Laplace transforms to obtain the solution of the simultaneous equations

$$\begin{cases} \frac{dx}{dt} + \frac{dy}{dt} + x = -e^{-t} \\ \frac{dx}{dt} + 2\frac{dy}{dt} + 2x + 2y = 0 \end{cases}$$

which satisfies the initial conditions

$$x(0) = -1, \quad y(0) = +1$$

(b) Solve the same problem by one of the methods of Chapter 1.

37. Establish the relation

$$\mathscr{L}\left\{t^k\frac{d^n y}{dt^n}\right\} = (-1)^k\frac{d^k}{ds^k}[s^n \bar{y} - s^{n-1}y(0) - \cdots - y^{(n-1)}(0)].$$

(Notice that by use of this relation a linear differential equation in y with polynomial coefficients can be transformed into a new linear differential equation in the transform \bar{y} , which may in certain cases be more tractable than the original equation. In particular, if the coefficients are *linear* functions of t, the transformed equation is of *first order*.)

Section 2.9

38. Use tables (when necessary) to evaluate the following:

(a) $\int_0^\infty e^{-x} \sqrt{x} \, dx$, (b) $\Gamma(2.7)$, (c) $\Gamma(-1.3)$, (d) (1.6)!, (e) (-1.3)!, (f) $\Gamma(\frac{5}{2})$. 39. By making the indicated substitutions, transform the integral

$$\Gamma(n) = \int_0^\infty e^{-t} t^{n-1} dt$$

to the following equivalent forms:

(a)
$$\Gamma(n) = \left(\int_0^1 \log \frac{1}{x}\right)^{n-1} dx$$
 $\left(t = \log \frac{1}{x}\right)$,
(b) $\Gamma(n) = 2\int_0^\infty e^{-r^2} r^{2n-1} dr$ $(t = r^2)$.

40. (a) Obtain from Stirling's formula (63) the relation

$$\log_{10} n! \sim (n + \frac{1}{2}) \log_{10} n - 0.4342945n + 0.39909 \qquad (n \to \infty)$$

for use in numerical computation.

(b) Show that use of this formula gives the approximations

$$10! \approx 3.599 \times 10^{6}, \quad 100! \approx 9.325 \times 10^{157}.$$

(The true values, to four figures, are 3.629×10^6 and 9.333×10^{157} . It can be shown, more generally, that Stirling's formula for n! is accurate to within 1 per cent when n > 10 and within 0.1 per cent when n > 100.)

41. Investigate the following limits:

(a)
$$\lim_{n \to \infty} \frac{n!}{(\alpha + 1)(\alpha + 2)(\alpha + 3) \cdots (\alpha + n)},$$

(b)
$$\lim_{n \to \infty} \frac{(2n)!}{2^{2n}n!(n - \frac{1}{2})!}.$$

42. Starting with the relation

$$\frac{\Gamma(x+n+1)}{\Gamma(x+1)}=(x+1)(x+2)\cdots(x+n),$$

for any positive integer n, obtain the result

$$\frac{d}{dx}\log\Gamma(x+n+1) = \frac{\Gamma'(x+n+1)}{\Gamma(x+n+1)} = \frac{\Gamma'(x+1)}{\Gamma(x+1)} + \frac{1}{1+x} + \frac{1}{2+x} + \dots + \frac{1}{n+x}.$$

(b) If we write $\Psi(z) = \Gamma'(z+1)/\Gamma(z+1)$, and set x = 0 in the result of part (a), show that there follows

$$\Psi(n) = \Psi(0) + 1 + \frac{1}{2} + \frac{1}{3} + \cdots + \frac{1}{n}$$

[The function $\Psi(z)$ is often called the *Digamma function*. It takes on the value $\Psi(0) = -\gamma$, where $\gamma = 0.5772157...$ is a number known as *Euler's constant*.]

43. The Beta function of p and q is defined by the integral

$$\mathbf{B}(p,q) = \int_0^1 t^{p-1} (1 - t)^{q-1} dt \qquad (p,q > 0).$$

By writing $t = \sin^2 \theta$, obtain the equivalent form

$$\mathbf{B}(p,q) = 2 \int_0^{\pi/2} \sin^{2p-1}\theta \cos^{2q-1}\theta \,d\theta \qquad (p,q>0).$$

44. Make use of the results of Problems 39(b) and 43 to verify the following development:

$$B(p,q) \ \Gamma(p+q) = 4 \int_0^\infty e^{-r^2} r^{2p+2q-1} dr \int_0^{\pi/2} \sin^{2p-1} \theta \cos^{2q-1} \theta \, d\theta$$
$$= 4 \int_0^\infty \int_0^\infty e^{-(x^2+y^2)} y^{2p-1} x^{2q-1} \, dx \, dy$$
$$= \Gamma(p) \ \Gamma(q) \qquad (p,q>0).$$

Hence show that

$$\mathbf{B}(p,q) = \frac{\Gamma(p) \ \Gamma(q)}{\Gamma(p+q)} \qquad (p,q>0).$$

45. By writing t = x/(x + a) in the definition of the Beta function, and using the result of Problem 44, obtain the result

$$\int_0^\infty \frac{x^{p-1} dx}{(x+a)^{p+q}} = a^{-q} \operatorname{B}(p,q) = a^{-q} \frac{\Gamma(p) \Gamma(q)}{\Gamma(p+q)} \qquad (p,q,a>0).$$

46. Use the results of preceding problems to verify the following evaluations:

(a)
$$\int_{0}^{1} \frac{t^{p-1}}{\sqrt{1-t}} dt = B\left(p, \frac{1}{2}\right) = \sqrt{\pi} \frac{\Gamma(p)}{\Gamma(p+\frac{1}{2})}$$
 $(p > 0),$
(b) $\int_{0}^{\pi/2} \sin^{n} \theta \, d\theta = \int_{0}^{\pi/2} \cos^{n} \theta \, d\theta = \frac{\sqrt{\pi}}{2} \frac{\Gamma\left(\frac{n+1}{2}\right)}{\Gamma\left(\frac{n+2}{2}\right)}$ $(n > -1),$
(c) $\int_{0}^{1} \frac{dt}{\sqrt{1-t}} = \frac{1}{2} \int_{0}^{1} \frac{s^{\frac{1}{n}-1} ds}{\sqrt{1-t}} = \frac{\sqrt{\pi}}{2} \frac{\Gamma\left(\frac{1}{n}\right)}{\Gamma\left(\frac{1}{n}\right)}$ $(n > 0),$

(c)
$$\int_0^1 \frac{dt}{\sqrt{1-t^n}} = \frac{1}{n} \int_0^1 \frac{s^n}{\sqrt{1-s}} = \frac{\sqrt{n}}{n} \frac{(n)}{\Gamma\left(\frac{1}{n}+\frac{1}{2}\right)}$$
 $(n > 0),$

- (d) $\int_0^\infty \frac{x^c dx}{(1+x)^2} = \Gamma(1-c) \Gamma(1+c) = \frac{\pi c}{\sin \pi c}$ (-1 < c < 1),
- (e) $\int_0^\infty \frac{dx}{1+x^n} = \frac{1}{n} \int_0^\infty \frac{s^{\frac{1}{n}-1} ds}{1+s} = \frac{\pi/n}{\sin(\pi/n)}$ (n > 1),
- (f) $\int_0^{\pi/2} \tan^n \theta \, d\theta = \frac{\pi/2}{\cos(n\pi/2)}$ (0 < n < 1).

47. Verify the relationship

$$\mathbf{B}(p,p) = 2 \int_0^{1/2} [t(1-t)]^{p-1} dt = \frac{\sqrt{\pi}}{2^{2p-1}} \frac{\Gamma(p)}{\Gamma(p+\frac{1}{2})},$$

by making use of the substitution $t = \frac{1}{2}(1 - \cos \theta)$.

48. By comparing the result of Problem 47 with the expression for B(p,p) which follows from the result of Problem 44, deduce the *duplication formula* for the Gamma function:

$$\Gamma(2p) = \frac{2^{2p-1}}{\sqrt{\pi}} \Gamma(p) \Gamma(p+\frac{1}{2}).$$

Also show that this result can be written in the form

$$(2n)! = \frac{2^{2n}}{\sqrt{\pi}} n! (n - \frac{1}{2})!.$$

49. (a) By writing x = u/t in the definition

$$B(m,n) = \int_0^1 x^{m-1}(1 - x)^{n-1} dx,$$

show that

$$B(m,n) t^{m+n-1} = \int_0^t u^{m-1} (t - u)^{n-1} du.$$

(b) By noticing that the right-hand member is the convolution of t^{m-1} and t^{n-1} , deduce that

$$B(m,n) \frac{\Gamma(m+n)}{s^{m+n}} = \frac{\Gamma(m)}{s^m} \frac{\Gamma(n)}{s^n},$$

and so obtain a simple derivation of the result of Problem 44.

CHAPTER 3

Numerical Methods for Solving Ordinary Differential Equations

3.1. Introduction. In this chapter there are presented certain methods of numerically calculating particular solutions of ordinary differential equations which cannot be readily solved analytically. The methods given are, in general, *step-by-step* methods and are described initially for first-order equations; however, the extension of these methods to the solution of higher-order equations is also indicated.

Before considering these procedures, we outline a graphical method of solving first-order differential equations which is of some practical interest. If such a differential equation is solved for the derivative of the unknown function, the result consists of one or more relations of the form

$$\frac{dy}{dx} = F(x,y) \tag{1}$$

where it may be assumed that the function F(x,y) is a single-valued function of x and y. Such an equation states that at any point (x,y) for which F(x,y) is defined, the slope of any integral curve passing through that point is given by F(x,y). If we plot the family of so-called *isocline* curves, defined by the equation

$$F(x,y) = C \tag{2}$$

for a series of values of the constant C, it then follows that all integral curves of (1) intersect a particular curve of the family (2) with the same slope angle φ , where tan φ is given by the value of C specifying the isocline. Thus, if on each isocline a series of short parallel segments having the required slope is drawn, an infinite number of integral curves can be sketched by starting in each case at a given point on one isocline and sketching a curve passing through that point with the indicated slope and crossing the successive isoclines with the slopes associated with them. This method can always be used to determine graphically the particular solution of (1) which passes through a prescribed



of (1) which passes through a prescribed point (x_0, y_0) , when the function F(x, y) is single-valued and continuous. The procedure is illustrated in Figure 3.1.

3.2. Use of Taylor series. Suppose that the solution of (1) which passes through the point (x_0, y_0) is required. Knowing the value of y at $x = x_0$, we attempt, by a step-by-step method, to calculate successively approximate values of y at the points

$$x_1 = x_0 + h,$$
 $x_2 = x_0 + 2h,$
..., $x_k = x_0 + kh,$

where h is a suitably chosen spacing along

the x axis. For this purpose we now suppose that the value of y has been determined at $x = x_k$, and denote this value by y_k ; that is, we write $y_k = y(x_k)$. We then make use of the Taylor series representation in the form

$$y(x + h) = y(x) + y'(x) \frac{h}{1!} + y''(x) \frac{h^2}{2!} + \cdots$$

and, setting $x = x_k$, we obtain the formula

$$y_{k+1} = y_k + y'_k \frac{h}{1!} + y''_k \frac{h^2}{2!} + \cdots$$
 (3)

for calculating the value of y_{k+1} . Primes are used to denote differentiation with respect to x. Since y' is given by (1) in terms of x and y, the coefficients in (3) can be determined from (1) by successive differentiation. Thus we obtain

$$y' = F(x,y), \qquad y'_{k} = F(x_{k},y_{k})$$
$$y'' = \frac{\partial F}{\partial x} + \frac{\partial F}{\partial y}\frac{dy}{dx}, \qquad y''_{k} = \frac{\partial F(x_{k},y_{k})}{\partial x} + \frac{\partial F(x_{k},y_{k})}{\partial y}y'_{k}$$

and so forth, for the higher derivatives. Since y_0 is given, Equation (3) can be used to determine first y_1 , with k = 0, then y_2 with k = 1, and so on, the number of terms being retained in (3) at each step depending upon the spacing and upon the accuracy desired. It is evident that a single series may, if preferred, be used for *several* calculations in some cases, by assigning successively doubled values to h. **Example 1.** To illustrate this procedure, we consider the solution of the differential equation

$$\frac{dy}{dx} = y - x$$

with the initial condition that y = 2 at x = 0. We choose an interval h = 0.1, and hence calculate successively the approximate values of y at x = 0.1, 0.2, 0.3, and so on. By successive differentiation we obtain

$$y' = y - x$$
, $y'' = y' - 1$, $y''' = y''$,

Hence, at x = 0, we have $y_0 = 2$ and

$$y'_0 = 2, \quad y''_0 = 1, \quad y'''_0 = 1, \quad \ldots,$$

and, with k = 0, Equation (3) gives

$$y_1 = y_0 + 2h + \frac{h^2}{2} + \frac{h^3}{6} + \cdots$$

$$\simeq 2 + 0.2000 + 0.0050 + 0.0002 + \cdots = 2.2052.$$

Next, at x = 0.1, we have $y_1 \simeq 2.2052$ and

$$y'_1 \simeq 2.1052, \quad y''_1 \simeq 1.1052, \quad y''_1 \simeq 1.1052, \quad \ldots,$$

and, with k = 1, Equation (3) gives

$$y_2 \cong y_1 + 2.1052h + 0.5526h^2 + 0.1842h^3 + \cdots$$

 $\cong 2.2052 + 0.2105 + 0.0055 + 0.0002 + \cdots = 2.4214.$

The procedure may be repeated as often as is required. In this example the exact solution is readily found to be

$$y=e^x+x+1,$$

and the above results are found to be accurate to the four decimal places retained.

This procedure is readily generalized to the solution of initial-value problems involving differential equations of higher order, as may be seen from Example 2, below. For a second-order equation it is found to be necessary to calculate y'_{k+1} as well as y_{k+1} before proceeding to the calculation of y_{k+2} . For this purpose we may differentiate y(x + h) with respect to x to obtain

$$y'(x + h) = y'(x) + y''(x) \frac{h}{1!} + y'''(x) \frac{h^2}{2!} + \cdots$$

Setting $x = x_k$, there then follows

$$y'_{k+1} = y'_k + y''_k \frac{h}{1!} + y'''_k \frac{h^2}{2!} + \cdots$$
 (4)

This result can also be obtained by differentiating Equation (3) with respect to

h, as may be seen directly from the fact that the derivatives of y(x + h) with respect to x and h are identical.

Example 2. Consider the nonlinear differential equation

$$\frac{d^2y}{dx^2} - \frac{dy}{dx} + xy^2 = 0$$

with the initial conditions that y = 1 and y' = -1 when x = 0. We calculate the successive derivatives

$$y'' = y' - xy^2, \quad y''' = y'' - 2xyy' - y^2,$$

 $y^{iv} = y''' - 2xy'^2 - 2xyy'' - 4yy', \quad \dots$

Hence, at x = 0, we have $y_0 = 1$, $y'_0 = -1$, and

$$y''_0 = y'_0 = -1, \qquad y'''_0 = y''_0 - y^2_0 = -2, \qquad y^{iv}_0 = 2, \qquad \dots$$

Then, with k = 0, Equation (3) gives

$$y_1 = 1 - h - \frac{h^2}{2} - \frac{h^3}{3} + \frac{h^4}{12} + \cdots,$$

and, taking h = 0.1,

 $y_1 \simeq 1 - 0.1 - 0.005 - 0.0003 + \cdots = 0.8947.$

Now, in order to calculate y_2 it will be next necessary to calculate y_1'', y_1''' , and so on. However, the calculation of these values involves knowledge of the value y_1' in addition to the value of y_1 , which is now known. The value of y_1' can be calculated by using the series

$$y'_1 = -1 - h - h^2 + \frac{h^3}{3} + \cdots,$$

which is obtained by differentiating the series defining $y_1 = y(x_0 + h)$ with respect to h. Hence we obtain

 $y'_1 \simeq -1 - 0.1 - 0.01 + 0.0003 + \cdots = -1.1097.$

The values of y_1'' , y_1''' , and so on, can now be calculated from the forms given, and the calculation of y_2 and y_2' proceeds in the same way.

A further generalization to the solution of two simultaneous equations of the form

$$\frac{dx}{dt} = F(x,y,t)$$
 and $\frac{dy}{dt} = G(x,y,t)$,

with prescribed initial values of x and y, is readily devised.

3.3. The Adams method. Suppose again that the solution of the problem

$$\frac{dy}{dx} = F(x,y), \qquad y(x_0) = y_0 \tag{5}$$

has been determined up to the point $x_k = x_0 + kh$. If now we assume that over the interval $(x_k, x_k + h)$ the derivative $\frac{dy}{dx}$ changes so slowly that it can be approximated by its value y'_k at the point x_k , then over that interval the approximate increase in y is given by $h y'_k$, and we obtain

$$y_{k+1} \cong y_k + h \ y'_k \cong y_k + h \ F_k, \tag{6}$$

where F_k is written for $F(x_k, y_k)$. This formula obviously would give exact results if y were a linear function of x over the interval considered.

A more nearly exact formula is obtained, in general, if we assume that the derivative $\frac{dy}{dx}$ is nearly *linear* over the interval $(x_k - h, x_k + h)$, and hence that the graph of y can be approximated by a parabola over this interval. We thus assume the approximation

$$F(x,y) \cong a + b(x - x_k) \qquad (x_k - h < x < x_k + h)$$

and determine the constants a and b in such a way that the approximation takes on the calculated values of F(x,y) at the points $x_k - h$ and x_k . In this way we obtain

$$a=F_k, \qquad b=\frac{F_k-F_{k-1}}{h}$$

Hence, integrating both sides of the differential equation (5) over the interval $(x_k, x_k + h)$, we find

$$y_{k+1} - y_k \simeq \int_{x_k}^{x_k + h} \left[F_k + (F_k - F_{k-1}) \frac{x - x_k}{h} \right] dx$$

$$\simeq h F_k + \frac{h}{2} (F_k - F_{k-1})$$

$$y_{k+1} \simeq y_k + h F_k + \frac{h}{2} (F_k - F_{k-1}).$$
(7)

or

The last term of this expression is seen to be a correction to the expression given by (6) and may be appreciable if the derivative $\frac{dy}{dx}$ varies appreciably over the interval considered. It should be noticed that the first ordinate calculable by (7) is y_2 , when k = 1, and that in this case the values of y_1 and F_1 are needed in addition to the known initial value F_0 . The value of y_1 must be determined by another method, for example, by the use of Taylor series; the value of F_1 is then given as $F(x_1, y_1)$. With the notation

$$\Delta F_k = F_{k+1} - F_k,$$

Equation (7) can be written in the form

$$y_{k+1} \simeq y_k + h(F_k + \frac{1}{2}\Delta F_{k-1}).$$
 (8)
Still more accurate formulas can be obtained, in general, if the derivative of the unknown function y is approximated by a polynomial of higher degree n, taking on calculated values at n + 1 consecutive points. By an extension of the preceding method (see Section 3.7), a formula is obtained when F(x,y)is approximated by a polynomial of *fourth degree*, in the form

 $y_{k+1} \simeq y_k + h[F_k + \frac{1}{2}\Delta F_{k-1} + \frac{5}{12}\Delta^2 F_{k-2} + \frac{3}{8}\Delta^3 F_{k-3} + \frac{25}{720}\Delta^4 F_{k-4}]$ (9) with the notations

$$\Delta F_r = F_{r+1} - F_r,$$

$$\Delta^2 F_r = \Delta F_{r+1} - \Delta F_r,$$

$$\Delta^3 F_r = \Delta^2 F_{r+1} - \Delta^2 F_r,$$

$$\Delta^4 F_r = \Delta^3 F_{r+1} - \Delta^3 F_r.$$
(10)

These notations define the first, second, third, and fourth forward differences of the calculated values of F which, as will be seen, are readily evaluated if the calculations are suitably tabulated.

Formula (9), which involves fourth differences, would give exact results if, over the interval $(x_k - 4h, x_k + h)$, the unknown function y were a polynomial of *fifth degree* in x. Formula (6) is obtained from (9) by neglecting all differences, and (7) or (8) is obtained by retaining only first differences. Formulas of intermediate accuracy can be obtained from (9) by retaining only terms through the second or third differences.

It should be noticed that if, for example, it is decided that second differences are to be retained, the calculation of y_{k+1} makes use of the values of F_k , F_{k-1} , and F_{k-2} , and hence of the values y_k , y_{k-1} , and y_{k-2} . Thus, the first ordinate calculable in this case would be y_3 . Since only y_0 is prescribed at the start, the values of y_1 and y_2 must first be determined by another method, such as that of Taylor series, or the Runge-Kutta method of the following section when initial series developments are not feasible. Similarly, if third or fourth differences are retained, three or four additional ordinates, respectively, must be first calculated by another method before the present procedure can be applied.

Example 3. We apply this method to the continuation of the solution of the problem considered in Example 1, retaining *second differences*. The work is tabulated as follows:

x	у	F=y-x	ΔF	$\Delta^2 F$
0	2	2	0.1052	0.0110
0.1	2.2052	2.1052	0.1162	0.0122
0.2	2.4214	2.2214	0.1284	
0.3	2.6498	2.3498		
0.4	2.8917			

Before the Adams formula is applied to calculate y_3 , the two initial ordinates y_1 and y_2 are required in addition to the prescribed ordinate $y_0 = 2$. These ordinates are taken from the results of Example 1 and are entered into the second column of the table as shown. The corresponding values of F are then entered in the third column. Each entry in the fourth column is obtained by subtracting the corresponding entry in the F column from the succeeding entry in that column. Algebraic signs must, of course, be retained. The last column contains similar differences between successive entries in the ΔF column. In this way the entries above the division line in each column of the table are obtained. Now, to calculate the ordinate y_3 , we make use of Formula (9), neglecting the last two terms, and notice that the quantities needed are exactly those which appear immediately above the division line in each column, i.e., the last numbers entered in the columns at this stage of the computation. We thus obtain

$$y_3 \simeq 2.4214 + 0.1[2.2214 + \frac{1}{2}(0.1162) + \frac{5}{12}(0.0110)] = 2.6498.$$

It is seen that the differences needed at a given step of the calculation recede along successive columns in the table. The value of y_3 is now entered in the second column, the corresponding value of F is calculated, and additional differences are determined. The next ordinate is then calculated as before,

$$y_4 \simeq 2.6498 + 0.1[2.3498 + \frac{1}{2}(0.1284) + \frac{5}{12}(0.0122)] = 2.8917,$$

and the process is continued in the same way. A rough check on the accuracy obtained at each step can be obtained by estimating the contribution of the neglected third difference. Thus, retention of third differences in the calculation of y_4 would, in this case, increase the value obtained by ${}_8^3(0.1)(0.0012) = 0.00005$, and hence the fourth places in the results are in doubt. The correct value is $e^{0.3} + 1.3 = 2.8918$.

To apply the Adams method to a second-order equation of the form

$$\frac{d^2y}{dx^2} = F\left(x, y, \frac{dy}{dx}\right),\tag{11}$$

we may first introduce the notation $p = \frac{dy}{dx}$ and hence replace Equation (11) by the simultaneous equations

$$\frac{dy}{dx} = p$$

$$\frac{dp}{dx} = F(x,y,p)$$
(12a,b)

The initial conditions at $x = x_0$,

$$y = y_0, \qquad \frac{dy}{dx} = p = p_0$$
 (13a,b)

are to be satisfied. Then, applying (9) separately to (12a) and (12b), we obtain the two formulas $\frac{1}{2}$

$$y_{k+1} \simeq y_k + h(p_k + \frac{1}{2}\Delta p_{k-1} + \cdots) \\ p_{k+1} \simeq p_k + h(F_k + \frac{1}{2}\Delta F_{k-1} + \cdots) \}, \qquad (14a,b)$$

and proceed step by step as before.

Example 4. We apply this method to the continuation of the solution of the problem considered in Example 2, retaining (for simplicity) only first differences. The work is tabulated as follows:

x	у	р	Δp	$F = p - xy^2$	ΔF
0	1	-1	-0.110	1	-0.190
0.1	0.895	-1.110		-1.190	
0.2	0.779	-1.238			

The values of y_1 and p_1 , taken from Example 2, as well as the prescribed values of y_0 and p_0 , are first entered. Next the corresponding values of F and the differences Δp_0 and ΔF_0 are calculated and entered. Equations (14a,b) then give

$$y_2 \simeq 0.895 + 0.1(-1.110 - 0.055) = 0.779,$$

 $p_2 \simeq -1.110 + 0.1(-1.190 - 0.095) = -1.238$

At this stage a second difference $\Delta^2 p_0 = -0.018$ can be calculated. Since its contribution to the calculation of y_3 would be -0.00075, it may be presumed that the result for y_2 is also in doubt by about one unit in the third decimal place.

The generalization of these methods to the solution of more general initialvalue problems involving two simultaneous differential equations offers no difficulty.

3.4. The modified Adams method. A useful modification of the Adams method in the case of a first-order equation consists of using an appropriate truncation of the Adams formula

$$y_{k+1} \cong y_k + h \left[F_k + \frac{1}{2} \Delta F_{k-1} + \frac{5}{12} \Delta^2 F_{k-2} + \frac{3}{8} \Delta^3 F_{k-3} + \frac{251}{720} \Delta^4 F_{k-4} \right]$$
(15)

only as a "predictor," to provide a first approximation to the value of y_{k+1} , and of using a corresponding truncation of the formula

$$y_{k+1} \cong y_k + h \left[F_{k+1} - \frac{1}{2} \Delta F_k - \frac{1}{12} \Delta^2 F_{k-1} - \frac{1}{24} \Delta^3 F_{k-2} - \frac{19}{720} \Delta^4 F_{k-3} \right]$$
(16)

as a "corrector." The derivation of the latter formula is indicated in Section 3.7.

The approximation yielded by (16) usually is better than that afforded by (15), when both are terminated with differences of like order, since the coefficients of the higher differences are smaller in (16). However, this advantage is partly offset by the fact that, since the right-hand member of (16) involves $F_{k+1} \equiv F(x_{k+1}, y_{k+1})$, this equation expresses y_{k+1} (approximately) in terms of itself. Thus (16) generally cannot be solved *analytically* for y_{k+1} except in special cases, such as those in which F is a linear function of y. Fortunately, resort generally can be had to a simple iterative procedure for the solution of (16) when the spacing h is sufficiently small. For this purpose, (15) is first used to determine a "predicted" value of y_{k+1} , in correspondence with which the approximate value of F_{k+1} is calculated, together with the approximate new differences $\Delta F_k = F_{k+1} - F_k$, $\Delta^2 F_{k-1}$, etc. Then (16) is used to determine a "corrected" value of y_{k+1} . If this value differs significantly from the predicted value, the entries F_{k+1} , ΔF_k , etc., then are correspondingly corrected and (16) is used again, to provide a "recorrected" value of y_{k+1} . Generally the need for this recorrection is avoided either by retaining a sufficiently large number of differences or by taking the spacing h to be sufficiently small.

Example 5. In the case of the problem considered in Example 3, the entry 2.8917 here would be interpreted as the *predicted* value of y_4 . When corresponding approximations to F_4 , ΔF_3 , and $\Delta^2 F_2$ are calculated, the tabulation appears as follows:

x	у	F	ΔF	$\Delta^2 F$
0	2	2	0.1052	0.0110
0.1	2.2052	2.1052	0.1162	0.0122
0.2	2.4214	2.2214	0.1284	0.0135
0.3	2.6498	2.3498	0.1419	
0.4	2.8917	2.4917		

From (16), truncated also to second differences, a corrected value is obtained,

 $y_4 \simeq 2.6498 + 0.1 [2.4917 - \frac{1}{2}(0.1419) - \frac{1}{12}(0.0135)] = 2.8918,$

in correspondence with which the tabulated values of F_4 , ΔF_3 , and $\Delta^2 F_2$ each are corrected by one unit in the last place retained. Clearly, no additional corrections will result from a repetition of the process, so that an advance to x = 0.5 appears to be in order.

When both formulas are truncated with a difference of order r (r = 2 in Example 5), the error in the finally corrected value can be estimated by the formula

$$E_{k+1} \simeq \frac{y_{k+1}^{P} - y_{k+1}^{C}}{2 + 4r}, \qquad (17)$$

where y_{k+1}^{P} is the "predicted" value obtained from (15) and y_{k+1}^{C} is the final "corrected" value obtained from (16). (See Reference 2.) This estimate applies only to the error introduced, by using (16) to approximate the differential equation, in progressing from x_k to x_{k+1} . It ignores the accumulated effects of errors of the same type introduced at earlier stages, as well as the effects of "round-off" errors; also its validity essentially presumes that the spacing h is sufficiently small to obviate the need for "recorrection."

Thus, for example, when only second differences are retained, the error in the corrected value can be estimated as $(y_{k+1}^P - y_{k+1}^C)/10$, provided that this estimate is small, that the same is true of the corresponding error estimates at all preceding stages, that recorrection was not needed, and that round-off errors have been controlled by the retention of an appropriate number of significant figures in all calculations.

It happens that the rate of convergence of the iterative process, in the determination of y_{k+1} from (16), depends upon the magnitude of the quantity

$$\rho_k = \beta h \, \frac{\partial F(x_k, y_k)}{\partial y} \,, \tag{18}$$

where β is the algebraic sum of the numerical coefficients of the differences retained in (16). (See Reference 2.) Thus, when *second* differences are retained,

$$\beta = 1 - \frac{1}{2} - \frac{1}{12} = \frac{5}{12}$$

Unless $|\rho_k| < 1$, the process either will not converge or will converge so slowly that many recorrections will be needed, in general. In the case of Example 5, the "convergence factor" ρ_3 is seen to be

$$\rho_3 = \frac{5}{12}h \approx 0.04.$$

Before starting a step-by-step calculation based on (16), it is desirable to verify that the spacing h is such that the magnitude of the initial convergence factor ρ_0 is small relative to unity.

The application of the modified Adams method to the second-order equation (11), or to a higher-order equation, is perfectly straightforward. In the case of Equation (11), y'' = F(x,y,y'), the approximate "convergence factor" is found to be

$$\rho_{k} = \beta h \left[\beta h \left(\frac{\partial F}{\partial y} \right)_{k} + \left(\frac{\partial F}{\partial y'} \right)_{k} \right].$$
(19)

3.5. The Runge-Kutta method. The method associated with the names of Runge and Kutta is a step-by-step process in which an approximation to y_{k+1} is obtained from y_k in such a way that the power series expansion of the approximation would coincide, up to terms of a certain order h^N in the spacing $h = x_{k+1} - x_k$, with the actual Taylor series development of $y(x_k + h)$ in powers of h. However, no preliminary differentiation is needed, and the method also has the advantage that no initial values are needed beyond the prescribed values. Such a method is particularly useful if certain coefficients in the differential equation are empirical functions for which analytical expressions are not known, and hence for which initial series developments are not feasible. In place of using values of N derivatives of y at one point, we use values only of the first derivative at N suitably chosen points.

sec. 3.5 | The Runge-Kutta method

The derivation of the basic formulas may be illustrated by considering the special case when *second-order* accuracy in h is required. Again starting with the differential equation

$$\frac{dy}{dx} = F(x,y) \tag{20}$$

and the prescribed initial condition $y = y_0$ when $x = x_0$, the Taylor series expansion of $y_{k+1} = y(x_k + h)$ up to second powers of h is obtained, by introducing the two succeeding relations into (3), in the form

$$y_{k+1} = y_k + F_k h + \left(\frac{\partial F_k}{\partial x} + \frac{\partial F_k}{\partial y}F_k\right)\frac{h^2}{2} + \cdots$$
 (21)

with the notations

$$F_k = F(x_k, y_k), \qquad \frac{\partial F_k}{\partial x} = \frac{\partial F(x_k, y_k)}{\partial x}, \qquad \frac{\partial F_k}{\partial y} = \frac{\partial F(x_k, y_k)}{\partial y}.$$

We assume an approximation of the form

$$y_{k+1} \cong y_k + \lambda_1 h F_k + \lambda_2 h F(x_k + \mu_1 h, y_k + \mu_2 h F_k)$$
(22)

and attempt to determine the constants λ_1 , λ_2 , μ_1 , and μ_2 in such a way that the expansion of the right-hand side of (22) in powers of *h* agrees with the expression given by (21) *through terms of second order in h*. From the Taylor series expansion

$$f(x + H, y + K) = f(x,y) + H \frac{\partial f(x,y)}{\partial x} + K \frac{\partial f(x,y)}{\partial y} + \cdots$$

for small values of H and K, where the following terms are of second order in H and K, we find

$$F(x_{k} + \mu_{1}h, y_{k} + \mu_{2}h F_{k})$$

$$= F(x_{k}, y_{k}) + \mu_{1}h \frac{\partial F(x_{k}, y_{k})}{\partial x} + \mu_{2}h \frac{\partial F(x_{k}, y_{k})}{\partial y} F_{k} + \cdots$$

$$= F_{k} + h \left[\mu_{1} \frac{\partial F_{k}}{\partial x} + \mu_{2} \frac{\partial F_{k}}{\partial y} F_{k} \right] + \cdots,$$

where the omitted terms are of at least second order in h. Hence (22) becomes

$$y_{k+1} = y_k + (\lambda_1 + \lambda_2)F_kh + \lambda_2\left(\mu_1\frac{\partial F_k}{\partial x} + \mu_2\frac{\partial F_k}{\partial y}F_k\right)h^2 + \cdots .$$
(23)

The terms retained in Equations (21) and (23) are brought into agreement if we take, in particular,

$$\lambda_1=\lambda_2=\frac{1}{2}, \qquad \mu_1=\mu_2=1.$$

Thus Equation (22) becomes, in this case,

$$y_{k+1} \cong y_k + \frac{h}{2} [F(x_k, y_k) + F(x_k + h, y_k + h F_k)]$$

or, writing

$$K_1 = h F(x_k, y_k), \quad K_2 = h F(x_k + h, y_k + K_1),$$
 (24)

this result can be put in the form

$$y_{k+1} \cong y_k + \frac{1}{2}(K_1 + K_2).$$
 (25)

It should be noticed that Equation (23) can be brought into agreement with (21) in infinitely many ways, by taking

$$\lambda_2 = 1 - \lambda_1, \qquad \mu_1 = \mu_2 = \frac{1}{2(1 - \lambda_1)}$$

where $\lambda_1 \neq 1$ but is otherwise arbitrary. Thus infinitely many other forms of (22) could be obtained, in addition to the rather symmetrical form chosen here.

By methods analogous to that just given, similar formulas giving higherorder accuracy in h may be obtained. We give without derivation two such procedures:

Third-order accuracy:

$$y_{k+1} \cong y_k + \frac{1}{6}(a_1 + 4a_2 + a_3) \tag{26}$$

$$\begin{array}{l} a_{1} = h \ F(x_{k}, y_{k}) \\ a_{2} = h \ F(x_{k} + \frac{1}{2}h, y_{k} + \frac{1}{2}a_{1}) \\ a_{3} = h \ F(x_{k} + h, y_{k} + 2a_{2} - a_{1}) \end{array} \right)$$
(27)

Fourth-order accuracy:

$$y_{k+1} \simeq y_k + \frac{1}{6}(b_1 + 2b_2 + 2b_3 + b_4)$$

$$b_1 = h F(x_1, y_2)$$
(28)

$$b_{1} = h F(x_{k}, y_{k})$$

$$b_{2} = h F(x_{k} + \frac{1}{2}h, y_{k} + \frac{1}{2}b_{1})$$

$$b_{3} = h F(x_{k} + \frac{1}{2}h, y_{k} + \frac{1}{2}b_{2})$$

$$b_{4} = h F(x_{k} + h, y_{k} + b_{3})$$
(29)

The close relationship between Equations (26) and (27) and the formula of "Simpson's rule" (see Problem 25) may be noticed.

Let the error associated with using a procedure of Nth-order accuracy k times with spacing h be expressed in the form $K k h^{N+1}$; that is, suppose that the correct result is given by adding $K k h^{N+1}$ to the calculated result. Then it can be shown that in ordinary cases the quantity K is not strongly dependent on k, h, and N. Thus, if the spacing were doubled, the error associated with the corresponding new calculation would be approximately

$$K\frac{k}{2}(2h)^{N+1} = 2^{N}Kkh^{N+1}.$$

Since the result of subtracting the ordinate determined by the second method from that determined by the first would then be $(2^N - 1) Kk h^{N+1}$, it follows that this difference is approximately $(2^N - 1)$ times the error in the first (more nearly exact) calculated value. In this way we are led to the following error estimate:

If a procedure of Nth-order accuracy gives an ordinate $y^{(1)}$ with spacing h and an ordinate $y^{(2)}$ with spacing 2h, the error in $y^{(1)}$ is given approximately by $(y^{(1)} - y^{(2)})/(2^N - 1)$.

Thus, the difference between the two results is divided by 3 if (24) and (25) are used, by 7 if (26) and (27) are used, and by 15 if (28) and (29) are used, to obtain an error estimate.

Example 6. We apply (26) and (27) to integrate the equation considered in Example 1. The work can be arranged as follows:

x	0	0.1	0.2
У	2	2.20517	2.42139
<i>a</i> ₁	0.2	0.21052	
$x+\frac{1}{2}h$	0.05	0.15	(
$y+\frac{1}{2}a_1$	2.1	2.31043	
a 2	0.205	0.21604	/
x+h	0.1	0.2	(
$y+2a_2-a_1$	2.21	2.42673	/
a ₃	0.211	0.22267	(
$\frac{a_1+4a_2+a_3}{6}$	0.20517	0.21622	

If the spacing is doubled, the result $y_2 \simeq 2.42133$ is obtained directly (with h = 0.2). Application of the error estimate gives the approximate error (0.00006)/7 = 0.00001, which indicates that the first value calculated for y_2 may be one unit too small in the last place retained. The exact value is $y_2 = 2.42140$ to five decimal places.

To integrate a second-order differential equation of the form

$$\frac{d^2y}{dx^2} = F\left(x, y, \frac{dy}{dx}\right) \tag{30}$$

(33)

with the initial conditions $y = y_0$ and $y' = y'_0$ when $x = x_0$, with *third-order* accuracy in the spacing, we first calculate successively the values

$$k_{1} = h y'_{k},$$

$$k'_{1} = h F(x_{k}, y_{k}, y'_{k}),$$

$$k_{2} = h(y'_{k} + \frac{1}{2}k'_{1}),$$

$$k'_{2} = h F(x_{k} + \frac{1}{2}h, y_{k} + \frac{1}{2}k_{1}, y'_{k} + \frac{1}{2}k'_{1}),$$

$$k_{3} = h(y'_{k} + 2k'_{2} - k'_{1}),$$

$$k'_{3} = h F(x_{k} + h, y_{k} + 2k'_{2} - k'_{1}, y'_{k} + 2k'_{2} - k'_{1}).$$
(31)

The values y_{k+1} and y'_{k+1} are then given by

$$y_{k+1} \simeq y_k + \frac{1}{6}(k_1 + 4k_2 + k_3) \tag{32}$$

and

A corresponding formula giving *fourth-order accuracy* can be written down by analogy from (28) and (29).

 $y'_{k+1} \simeq y'_k + \frac{1}{6}(k'_1 + 4k'_2 + k'_3).$

Example 7. Applying this method to calculate y_1 in the case of the problem of Example 2, we obtain successively

$$k_1 = -0.1,$$
 $k_2 = -0.105,$ $k_3 = -0.11190,$
 $k'_1 = -0.1,$ $k'_2 = -0.10951,$ $k'_3 = -0.11982.$

There then follows

$$y_1 \simeq 1 - \frac{1}{6}(0.1 + 0.42 + 0.11190) = 0.89468,$$

 $y'_1 \simeq -1 - \frac{1}{6}(0.1 + 0.43804 + 0.11982) = -1.1096.$

3.6. Picard's method. In contrast with the step-by-step methods so far considered, in which successive ordinates are calculated point by point, the method of Picard is an *iterative method* which gives successive *functions* which, in favorable cases, tend as a whole (at least over a certain interval) toward the exact solution. Although the method is of limited practical usefulness, it illustrates a type of procedure which is of frequent use in other applications, and is in itself of theoretical importance.

Considering first an initial-value problem of first order,

$$\frac{dy}{dx} = F(x,y),\tag{34}$$

where $y = y_0$ when $x = x_0$, we formally integrate both sides of Equation (34) over the interval (x_0, x) to obtain an equivalent expression

$$y = y_0 + \int_{x_0}^x F(x, y) \, dx. \tag{35}$$

Now, to start the procedure, we take as an initial approximation to the function y (to be determined) a suitable function of x, say $y^{(1)}(x)$. If the general nature of the required solution of (34) is known, this initial function may be chosen on this basis. It is preferable that it satisfy the initial condition, although this is not necessary. In the absence of further information, the initial approximation function $y^{(1)}(x)$ may be taken as the constant y_0 . With this assumed approximation for y, as a function of x, the function $F(x,y^{(1)})$ becomes a known function of x, and a second approximation to the function y, say $y^{(2)}(x)$, is given by

$$y^{(2)}(x) = y_0 + \int_{x_0}^x F[x, y^{(1)}(x)] dx.$$
 (36)

A third approximation is obtained by replacing y by $y^{(2)}(x)$ in F(x,y) and using Equation (35) to give

$$y^{(3)}(x) = y_0 + \int_{x_0}^x F[x, y^{(2)}(x)] dx.$$
 (37)

In this way successive approximations are obtained as functions of x, according to the formula

$$y^{(n+1)}(x) = y_0 + \int_{x_0}^x F[x, y^{(n)}(x)] \, dx. \tag{38}$$

If F(x,y) is sufficiently regular near the point (x_0,y_0) , the successive approximations $y^{(1)}, y^{(2)}, \ldots, y^{(n)}$ will tend toward a limiting function y(x) over some interval in x about $x = x_0$, and that function will satisfy the differential equation (34) as well as the prescribed initial condition.

Example 8. Applying this method to the solution of the equation of Example 1, we write

$$y^{(n+1)} = 2 + \int_0^x (y^{(n)} - x) \, dx.$$

Taking $y^{(1)} = 2$, we obtain successively

$$y^{(2)} = 2 + \int_0^x (2 - x) \, dx = 2 + 2x - \frac{x^2}{2},$$

$$y^{(3)} = 2 + \int_0^x \left(2 + x - \frac{x^2}{2}\right) \, dx = 2 + 2x + \frac{x^2}{2} - \frac{x^3}{6},$$

$$y^{(4)} = 2 + \int_0^x \left(2 + x + \frac{x^2}{2} - \frac{x^3}{6}\right) \, dx = 2 + 2x + \frac{x^3}{2} + \frac{x^3}{6} - \frac{x^4}{24},$$

and so forth. In this case it is readily shown by induction that

$$y^{(n+1)} = 2 + 2x + \frac{x^2}{2!} + \frac{x^3}{3!} + \cdots + \frac{x^n}{n!} - \frac{x^{n+1}}{(n+1)!}$$

and as *n* becomes large we are led to a consideration of the infinite series

$$y = 1 + x + \left(1 + \frac{x}{1!} + \frac{x^2}{2!} + \cdots + \frac{x^n}{n!} + \cdots\right),$$

which may be recognized as the expansion of the known exact solution

 $y=1+x+e^x,$

the expansion converging for all values of x.

In practice it is usually not feasible to obtain an explicit expression for the nth approximation and to proceed to the exact solution by a limiting process, as in the example given. Furthermore, the successive approximations need not be polynomials, as in this case. Still, if a particular approximation is used to calculate y, the accuracy at a given point can be estimated roughly by considering the deviation between this approximation and the preceding approximation in the neighborhood of the point.

Similar procedures can be devised for dealing with equations of higher order, the convergence of the successive approximations to the true solution depending upon the regularity of the coefficients.

3.7. Extrapolation with differences. For purposes of simplicity, the details of the derivation of Equations (9) and (16), which specify the Adams method and its modification for the numerical integration of differential equations, were omitted in Sections 3.3 and 3.4. In this section we indicate this derivation and also point out the usefulness in other connections of certain intermediate and related results.

We again denote the value of a function f(x) at one of n + 1 equally spaced points,

$$x_k, \quad x_{k-1} = x_k - h, \quad x_{k-2} = x_k - 2h, \quad \dots, \quad x_{k-n} = x_k - nh,$$
 (39)
we the abbreviation $f = f(x)$. In Section 3.3 the forward differences were

by the abbreviation $f_r = f(x_r)$. In Section 3.3 the forward differences were defined by the equations

$$\Delta f_r = f_{r+1} - f_r, \qquad \Delta^2 f_r = \Delta f_{r+1} - \Delta f_r = f_{r+2} - 2f_{r+1} + f_r, \qquad (40)$$

and so forth. In some developments it is more convenient to use the so-called *backward difference* notation, according to which we write instead

$$\nabla f_r = f_r - f_{r-1}, \qquad \nabla^2 f_r = \nabla f_r - \nabla f_{r-1} = f_r - 2f_{r-1} + f_{r-2}, \qquad (41)$$

and so forth. These two notations clearly are related by the equations

$$\Delta f_r = \nabla f_{r+1}, \quad \Delta^2 f_r = \nabla^2 f_{r+2}, \quad \dots, \quad \Delta^m f_r = \nabla^m f_{r+m}. \tag{42}$$

Thus, for example, we have

$$f_3 - f_2 = \Delta f_2 = \nabla f_3, \quad f_3 - 2f_2 + f_1 = \Delta^2 f_1 = \nabla^2 f_3.$$

In particular, Equations (9) and (16) take the forms

 $y_{k+1} \cong y_k + h \left[F_k + \frac{1}{2} \nabla F_k + \frac{5}{12} \nabla^2 F_k + \frac{3}{8} \nabla^3 F_k + \frac{251}{720} \nabla^4 F_k \right]$ (43) and

$$y_{k+1} \cong y_k + h \left[F_{k+1} - \frac{1}{2} \nabla F_{k+1} - \frac{1}{12} \nabla^2 F_{k+1} - \frac{1}{24} \nabla^3 F_{k+1} - \frac{19}{720} \nabla^4 F_{k+1} \right], \quad (44)$$

with the notation (41) of backward differences, the subscripts on the right in these equations no longer varying from difference to difference.

In order to determine a *polynomial approximation* of degree n, to a function f(x), having the property that agreement is exact at the n + 1 points defined in (39), it is convenient for present purposes to write the approximation in the form

$$f(x) \cong a_0 + a_1(x - x_k) + a_2(x - x_k)(x - x_{k-1}) + a_3(x - x_k)(x - x_{k-1})(x - x_{k-2}) + \cdots + a_n(x - x_k)(x - x_{k-1}) \cdots (x - x_{k-n+1}).$$
(45)

In this form the coefficients are readily determined by forming successive backward differences at $x = x_k$, as follows. First, setting $x = x_k$, and replacing the approximation by an equality, we have

$$a_0 = f_k. \tag{46a}$$

Next, if we calculate the difference $\nabla f(x) = f(x) - f(x - h)$ and use (39), there follows

$$\nabla f(x) \simeq ha_1 + 2ha_2(x - x_k) + 3ha_3(x - x_k)(x - x_{k-1}) + \cdots + nha_n(x - x_k)(x - x_{k-1}) \cdots (x - x_{k-n+2}).$$

Thus, setting $x = x_k$ and requiring equality, we obtain

$$a_1 = \frac{1}{h} \nabla f_k. \tag{46b}$$

After calculating the second difference,

$$\nabla^2 f(x) \simeq 2 \cdot 1h^2 a_2 + 3 \cdot 2h^2 a_3(x - x_k) + \cdots + n(n-1)h^2 a_n(x - x_k)(x - x_{k-1}) \cdots (x - x_{k-n+3}),$$

we determine a_2 ,

$$a_2 = \frac{1}{2! h^2} \nabla^2 f_k.$$
 (46c)

From the way in which these results were obtained, it can be seen that the general result will be of the form

$$a_r = \frac{1}{r! \ h^r} \ \nabla^r f_k \qquad (k = 0, 1, 2, \dots, n).$$
 (47)

The resultant approximation formula (45) can be put into a conveniently compact form if we write

$$x = x_k + sh, \qquad s = \frac{x - x_k}{h}, \qquad (48)$$

so that s is distance measured from the point x_k , in units of the spacing h.

With this notation, the introduction of (47) into (45) then gives

$$f(x_{k} + sh) \cong f_{k} + s \nabla f_{k} + \frac{s(s+1)}{2!} \nabla^{2} f_{k} + \frac{s(s+1)(s+2)}{3!} \nabla^{3} f_{k} + \dots + \frac{s(s+1)\cdots(s+n-1)}{n!} \nabla^{n} f_{k}, \quad (49)$$

if use is made of (39). This result is known as Newton's (or Gregory's) backward difference formula for polynomial approximation.

Thus, if the values of a smooth function f(x) are known at n + 1 equally spaced points, we may suppose that the function is approximated by the *n*thdegree polynomial which agrees with f(x) at these points and, accordingly, use (49) to determine approximate values of f(x) at additional near-by points. This formula is particularly useful for extrapolation (prediction) beyond the point x_k , at the end of a range of tabulation, although it can also be used for interpolation, with negative values of s.

To illustrate the use of this formula in extrapolation, we consider the function $f(x) = \sqrt{x}$. Assuming known three-place values for x = 2, 3, 4, 5, and 6, we form the following difference table:

x	$\int f$	∇f	$\nabla^2 f$	$ abla^{3}f$
2.0	1.414	· · · · · · · · · · · · · · · · · · ·		
		0.318		
3.0	1.732		0.050	
	!	0.268		0.018
4.0	2.000		-0.032	
		0.236		0.009
5.0	2.236		-0.023	
		0.213		
6.0	2.449			

To extrapolate for $\sqrt{6.2}$, we set s = 0.2, since here h = 1, and retain third differences. Equation (49) then gives

$$f(6.2) \simeq 2.449 + (0.2)(0.213) + \frac{(0.2)(1.2)}{2}(-0.023) + \frac{(0.2)(1.2)(2.2)}{6}(0.009)$$

= 2.449 + 0.0426 - 0.0028 + 0.0008
= **2.490**.

Similarly, to interpolate for $\sqrt{5.8}$ we take s = -0.2 and, again retaining third differences, obtain the value 2.408. Both the values so obtained are correct to the same number of places as the given data. In general, however, extrapolation is less dependable than interpolation.

Equation (49) is also useful for *integrating* a tabular function over an interval near the end point x_k . Thus we have, for example,

$$\int_{x_k}^{x_k+h} f(x) \, dx = h \int_0^1 f(x_k + sh) \, ds$$

and (49) then gives

$$\int_{x_{k}}^{x_{k}+h} f(x) dx \simeq h [f_{k} + \frac{1}{2} \nabla f_{k} + \frac{5}{12} \nabla^{2} f_{k} + \frac{3}{8} \nabla^{3} f_{k} + \frac{251}{720} \nabla^{4} f_{k} + \cdots].$$
(50)

When f(x) is a polynomial, the sum on the right terminates, and yields an exact result. More generally, when f(x) is not a polynomial the result of retaining differences through the *n*th is identical with the result of replacing f(x) by the polynomial of degree *n* which agrees with f at the n + 1 points $x_k, x_{k-1}, \ldots, x_{k-n}$ and integrating that polynomial over the interval $(x_k, x_k + h)$.

In illustration, if we use (50) to approximate $\int_6^7 \sqrt{x} \, dx$, with third differences, we obtain

$$\int_{6}^{7} \sqrt{x} \, dx \simeq 1 \left[2.449 + \frac{1}{2} (0.213) + \frac{5}{12} (-0.023) + \frac{3}{8} (0.009) \right] = 2.549.$$

This is in agreement with the true three-place value.

Similarly, from the relation

$$\int_{x_k-h}^{x_k} f(x) \, dx = h \int_{-1}^0 f(x_k + sh) \, ds,$$

we obtain the formula

$$\int_{x_{k}-h}^{x_{k}} f(x) dx \simeq h \Big[f_{k} - \frac{1}{2} \nabla f_{k} - \frac{1}{12} \nabla^{2} f_{k} - \frac{1}{24} \nabla^{3} f_{k} - \frac{19}{720} \nabla^{4} f_{k} + \cdots \Big].$$
(51)

Equation (50), with f replaced by F, leads to the approximate relation (43) or, equivalently, (9) when dy = F dx, and hence

$$y_{k+1} - y_k = \int_{x_k}^{x_{k+1}} F \, dx.$$

Further, Equation (51) leads similarly to (44) or (16) when k is replaced by k + 1.

Finally, we may use (49) to obtain an approximate *derivative* formula. Since $f'(x) = h^{-1} \frac{df}{ds}$, we obtain the relation

$$f'(x_k + sh) \simeq \frac{1}{h} \bigg[\nabla f_k + \frac{2s+1}{2} \nabla^2 f_k + \frac{3s^2 + 6s + 2}{6} \nabla^3 f_k + \cdots \bigg].$$
(52)

Thus, to approximate $\left(\frac{d}{dx}\sqrt{x}\right)_{x=6}$ by using the tabulated data, we take s=0 and hence, again using third differences, obtain

$$f'(6) \simeq \frac{1}{1} \left[0.213 + \frac{1}{2}(-0.023) + \frac{1}{3}(0.009) \right]$$

= 0.213 - 0.0115 + 0.003 = **0.2045**.

The true result is 0.204, to three places. Although satisfactory results were obtained here, it should be noted that, *in general, approximate differentiation is inaccurate*. Its accuracy may be very seriously impaired by small inaccuracies in the tabulated data. Still, there are occasions when there is no alternative but to use an approximate method of this sort.

The formulas and methods of this section illustrate the treatment of tabulated data near the end of the range over which a tabulation exists. These data are usually the most troublesome to deal with. This section may be considered also as illustrating the more general use of finite differences in approximate analysis.

Analogous formulas for interpolation and other operations near *interior* tabular points may be found in texts treating numerical methods. (See also Problems 21–24.) In such formulas the so-called *central difference* notation is particularly useful. In this notation one writes

$$f_r - f_{r-1} = \delta f_{r-1/2}, \qquad \delta f_{r+1/2} - \delta f_{r-1/2} = \delta^2 f_r, \qquad \dots, \qquad (53)$$

so that the subscript of a difference is the mean of the subscripts of its parents.

Thus we have the general notational correspondence

$$f_{r+1} - f_r = \Delta f_r = \delta f_{r+1/2} = \nabla f_{r+1},$$

$$f_{r+2} - 2f_{r+1} + f_r = \Delta^2 f_r = \delta^2 f_{r+1} = \nabla^2 f_{r+2},$$

and so forth. The "forward" difference Δf is sometimes called a "descending" difference, whereas the "backward" difference ∇f is sometimes called an "ascending" difference. Although this practice leads to a certain amount of confusion, its motivation may be realized by an inspection of the following table of difference notations.

$$\begin{array}{cccc} f_{0} & & \Delta f_{0} = \delta f_{1/2} = \nabla f_{1} & & \Delta^{2} f_{0} = \delta^{2} f_{1} = \nabla^{2} f_{2} & & \Delta^{3} f_{0} = \delta^{3} f_{3/2} = \nabla^{3} f_{3} & & \\ f_{2} & & & \Delta^{2} f_{1} = \delta^{2} f_{2} = \nabla^{2} f_{3} & & \Delta^{3} f_{0} = \delta^{3} f_{3/2} = \nabla^{3} f_{3} & & \\ f_{2} & & & \Delta^{2} f_{1} = \delta^{2} f_{2} = \nabla^{2} f_{3} & & \Delta^{3} f_{1} = \delta^{3} f_{5/2} = \nabla^{3} f_{4} & & \\ f_{3} & & & \Delta^{2} f_{2} = \delta^{2} f_{3} = \nabla^{2} f_{4} & & \Delta^{3} f_{1} = \delta^{3} f_{5/2} = \nabla^{3} f_{4} & & \\ f_{4} & & & & \Delta^{4} f_{3} = \delta f_{7/2} = \nabla f_{4} & & & & \\ \end{array}$$

ſ

It is apparent that the differences Δf_r , $\Delta^2 f_r$, $\Delta^3 f_r$, and so forth, "descend" to the right, whereas the differences ∇f_r , $\nabla^2 f_r$, $\nabla^3 f_r$, and so forth, "ascend" to the right. The central differences $\delta^2 f_r$, $\delta^4 f_r$, and so forth, remain in the same horizontal line as the entry f_r , whereas the differences $\delta f_{r+1/2}$, $\delta^2 f_r$, $\delta^3 f_{r+1/2}$, $\delta^4 f_r$, and so forth, form a "forward zigzag" set of differences in increasing order, remaining as close as possible to a horizontal line.

More detailed treatments of the errors associated with formulas for approximate interpolation, numerical integration and differentiation, and the numerical solution of differential equations may be found in many sources, including the references listed at the end of this chapter.

REFERENCES

- 1. Hildebrand, F. B., Introduction to Numerical Analysis, McGraw-Hill Book Company, Inc., New York, 1956.
- 2. Henrici, P., Discrete Variable Methods in Ordinary Differential Equations, John Wiley & Sons, Inc., New York, 1962.
- 3. Levy, H., and E. A. Baggott, Numerical Solutions of Differential Equations, Dover Publications, Inc., New York, 1950.
- 4. Milne, W. E., Numerical Solution of Differential Equations, John Wiley & Sons, Inc., New York, 1953.
- 5. Steffensen, J. F., Interpolation, Chelsea Publishing Company, New York, 1950.

PROBLEMS

Section 3.1

1. Use the method of isoclines to sketch the integral curves of the equation

$$\frac{dy}{dx} - y = x$$

in the first quadrant.

2. Proceed as in Problem 1 with the equation

$$\frac{dy}{dx} - y^2 = x^2$$

3. Use the method of isoclines to obtain a sketch, in the first quadrant, of the integral curve of the equation

$$\frac{dy}{dx} - x^2 y = x$$

which passes through the point (0,1).

Section 3.2

4. Use the method of Taylor series to determine to four places the values of the solution of the problem

$$\frac{dy}{dx} - x^2 y = x, \qquad y(0) = 1$$

at the points x = 0.1, 0.2, and 0.3.

(The true values at x = 0.1, 0.2, 0.3, 0.4, and 0.5 round to 1.00533, 1.02270, 1.05428, 1.10260, and 1.17072.)

5. Use the method of Taylor series to determine to four places the values of the solution of the problem

$$\frac{dy}{dx} + xy^2 = 0, \qquad y(0) = 1$$

at the points x = 0.1, 0.2, and 0.3.

(The true values at x = 0.1, 0.2, 0.3, 0.4, and 0.5 round to 0.99502, 0.98039, 0.95694, 0.92593, and 0.88889.)

6. Use the method of Taylor series to determine to four places the values of the solution of the problem

$$\frac{d^2y}{dx^2} - x^2 \frac{dy}{dx} - 2xy = 1, \qquad y(0) = 1, \quad y'(0) = 0$$

at the points x = 0.1, 0.2, and 0.3.

(The true values are the same as those in Problem 4.)

Section 3.3

7. Assuming the values of y in Problem 4 for x = 0, 0.1, 0.2, and 0.3, use the Adams method to calculate the values for x = 0.4 and 0.5, using third differences.

8. Assuming the values of y in Problem 5 for x = 0, 0.1, 0.2, and 0.3, use the Adams method to calculate the values for x = 0.4 and 0.5, using third differences.

9. Assuming the results of Problem 6 for x = 0, 0.1, 0.2, and 0.3, use the Adams method to calculate the values of y for x = 0.4 and 0.5, using third differences.

Section 3.4

10. Use the modified Adams method, retaining only second differences, to effect the determinations of (a) Problem 7, (b) Problem 8, and (c) Problem 9. At each stage, use Equation (17) to estimate the error introduced.

11. If no differences beyond the *second* are retained, show that the formulas (15) and (16), of the modified Adams method, can be written in the explicit forms

$$y_{k+1} \cong y_k + \frac{h}{12} (23 F_k - 16 F_{k-1} + 5 F_{k-2})$$

 $y_{k+1} \cong y_k + \frac{h}{12} (5 F_{k+1} + 8 F_k - F_{k-1}),$

and

respectively. (These formulas are easily used, since differencing is not involved. However, the use of differences is usually preferable in questionable cases, since then gross errors may be indicated by irregularities in the columns of differences, and also warnings may be served by trends observable in these columns.)

12. Making use of the fact that, for small values of the spacing h, there follows

$$\Delta^r f \approx \nabla^r f \approx \delta^{\prime} f \approx h^r \frac{d^r f}{dx^r},$$

show that the errors introduced into the formulas of Problem 11 by neglecting the *first* omitted difference are respectively approximated by

$$\frac{3h^4}{8}\left(\frac{d^4y}{dx^4}\right)_k, \qquad -\frac{h^4}{24}\left(\frac{d^4y}{dx^4}\right)_k.$$

[It is known that the true truncation error in each of these cases is given by the result of replacing $\left(\frac{d^4y}{dx^4}\right)_k$ by the value of $\frac{d^4y}{dx^4}$ at some point between x_{k-2} and x_{k+1} in the expression so obtained.]

Section 3.5

13. Use the Runge-Kutta method, with third-order accuracy, to determine the approximate values of y at x = 0.1 and 0.2 if y satisfies the conditions of (a) Problem 4, (b) Problem 5, and (c) Problem 6.

14. A function y(x) satisfies the equation

$$\frac{d^2y}{dx^2} + y\,\varphi(x) = 0$$

and the initial conditions y(0) = 1 and y'(0) = 0. The following approximate values of the function $\varphi(x)$ are known:

x	0	0.05	0.10	0.15	0.20
$\varphi(x)$	1.000	1.032	1.115	1.249	1.434

Use the Runge-Kutta method, with third-order accuracy, to determine approximate values of y at x = 0.1 and 0.2.

Section 3.6

15. Apply Picard's method to the solution of the problem

$$\frac{dy}{dx}-x^2y=x, \qquad y(0)=1,$$

taking $y^{(1)}(x) = 1$ and making two successive substitutions, and compare the approximations with the series expansion,

 $y = 1 + \frac{1}{2}x^2 + \frac{1}{3}x^3 + \frac{1}{10}x^5 + \frac{1}{18}x^6 + \cdots,$

of the exact solution.

16. Apply Picard's method to the solution of the problem

$$\frac{dv}{dx}-y^2=x, \qquad y(0)=1,$$

taking $y^{(1)}(x) = 1$ and making two successive substitutions, and compare the approximations with the series expansion,

$$y = 1 + x + \frac{3}{2}x^2 + \frac{4}{3}x^3 + \frac{17}{12}x^4 + \cdots$$

of the exact solution.

Section 3.7

17. Suppose that the following rounded values of a certain function f(x) are known:

<i>x</i>	1.0	2.0	3.0	4.0	5.0
f(x)	1.2840	1.3499	1.4191	1.4918	1.5683

By making use of formulas derived in Section 3.7, obtain approximate values of the following quantities as accurately as possible:

(a) f(4.8), (b) f(5.2), (c) f'(5.0), (d) $\int_{5}^{6} f(x) dx$.

18. Derive from Equation (49) the formula

$$\int_{x_k}^{x_k+th} f(x) dx \simeq th \left[f_k + \frac{t}{2} \nabla f_k + \frac{t(3+2t)}{12} \nabla^2 f_k + \cdots \right].$$

19. Use the result of Problem 18 to obtain, from the data of Problem 17, approximate values of the following quantities:

(a)
$$\int_{5.0}^{5.2} f(x) dx$$
, (b) $\int_{4.8}^{5.0} f(x) dx$

20. Establish the following notational relations:

(a) $\Delta f_r = \delta f_{r+1/2}$, (b) $\nabla \Delta f_r = \Delta \nabla f_r = \delta^2 f_r$, (c) $\Delta \nabla \Delta f_r = \delta^3 f_{r+1/2}$.

21. Determine the coefficients specifying the polynomial approximation of degree n = 2k of the form

$$f(x) \simeq a_0 + a_1(x - x_0) + a_2(x - x_0)(x - x_1) + a_3(x - x_0)(x - x_1)(x - x_{-1}) + a_4(x - x_0)(x - x_1)(x - x_{-1})(x - x_2) + \cdots + a_n(x - x_0)(x - x_1)(x - x_{-1}) \cdots (x - x_k)(x - x_{-k}),$$

so that the two members are equal at the n + 1 equally spaced points

 $x_{-k} = x_0 - kh, \ldots, x_0, \ldots, x_k = x_0 + kh,$

by calculating the difference $\Delta f(x) = f(x + h) - f(x)$, then the differences

 $\nabla [\Delta f(x)], \Delta [\nabla \Delta f(x)]$, and so on, and equating the two members at $x = x_0$ after each such differencing. Thus show that $a_0 = f_0$, that

$$\Delta f(x) \simeq h[a_1 + 2a_2(x - x_0) + 3a_3(x - x_0)(x - x_{-1}) + 4a_4(x - x_0)(x - x_{-1})(x - x_1) + \cdots],$$

and hence that $a_1 = \frac{\Delta f_0}{h} = \frac{\delta f_{1/2}}{h}$, and so forth. Make use of the results of Problem 20 to express the result in the form

$$f(x) \cong f_0 + (x - x_0) \frac{\delta f_{1/2}}{1! h} + (x - x_0)(x - x_1) \frac{\delta^2 f_0}{2! h^2} + (x - x_0)(x - x_1)(x - x_{-1}) \frac{\delta^3 f_{1/2}}{3! h^3} + (x - x_0)(x - x_1)(x - x_{-1})(x - x_2) \frac{\delta^4 f_0}{4! h^4} + \cdots$$

22. By writing $x = x_0 + sh$, express the result of Problem 21 in the form

$$f(x_0 + sh) \simeq f_0 + s \,\delta f_{1/2} + \frac{s(s-1)}{2!} \,\delta^2 f_0 + \frac{s(s^2-1)}{3!} \,\delta^3 f_{1/2} \\ + \frac{s(s^2-1)(s-2)}{4!} \,\delta^4 f_0 + \cdots.$$

(This is the interpolation formula of *Gauss*. The central differences involved remain as near as possible to the horizontal line through the tabular value f_0 , comprising a "forward zigzag.")

23. By integrating the result of Problem 22 over appropriate intervals, obtain the integral formulas

(a)
$$\int_{x_0-h}^{x_0+h} f(x) dx \cong 2h \left[f_0 + \frac{1}{6} \delta^2 f_0 - \frac{1}{160} \delta^4 f_0 + \cdots \right],$$

(b) $\int_{x_0-2h}^{x_0+2h} f(x) dx \cong 4h \left[f_0 + \frac{2}{3} \delta^2 f_0 + \frac{7}{90} \delta^4 f_0 + \cdots \right].$

24. Make appropriate use of the Gauss interpolation formula of Problem 22 to obtain, from the data given in Problem 17, approximate values of each of the following quantities:

(a)
$$f(2.8)$$
, (b) $f(3.2)$, (c) $f'(3.0)$,

(d)
$$f'(2.8)$$
, (e) $\int_{2}^{4} f(x) dx$, (f) $\int_{1}^{5} f(x) dx$.

25. By retaining only second differences in the result of Problem 23(a), deduce the formula

$$\int_{x_0-h}^{x_0+h} f(x) \, dx \simeq \frac{h}{3} (f_{-1} + 4f_0 + f_1).$$

[This is the celebrated formula of Simpson's rule. Notice that since the coefficient of the third difference in the more general formula is zero, the formula is exact when f(x) is a cubic polynomial.]

26. Show (as in Problem 12) that the error associated with Simpson's rule is approximately

$$-\frac{h^5}{90}\left(\frac{d^4f}{dx^4}\right)_{x_0}.$$

[The *true* error is known to be expressible in this form as well, but with $\frac{d^4y}{dx^4}$ evaluated instead at some unknown point between $x_0 - h$ and $x_0 + h$.]

27. Show that if Simpson's rule, of Problem 25, is applied successively over the adjacent double intervals (a, a + 2h), (a + 2h, a + 4h), ..., (b - 2h, b) in the approximate evaluation of the integral $\int_{a}^{b} f(x) dx$, there follows

$$\int_{a}^{b} f(x) dx \simeq \frac{h}{3} [f_{0} + 4f_{1} + 2f_{2} + 4f_{3} + \cdots + 2f_{n-2} + 4f_{n-1} + f_{n}],$$

where h = (b - a)/n and *n* is an even integer. [This formula, often known as the *parabolic rule*, is probably the most widely used formula for numerical integration. It is *exact* when f(x) is a polynomial of degree not greater than three. As is suggested by n/2 applications of the result of Problem 26, for a given interval (a,b) the error in the approximation is nearly proportional to h^4 and hence to $1/n^4$. From this fact it follows that if two calculations are made, one (I_n) with *n* subdivisions and one (I_{2n}) with 2*n* subdivisions, the error in I_{2n} may be estimated by $(I_{2n} - I_n)/15$.]

28. Evaluate the following integrals approximately by use of the parabolic rule, Problem 27, first with n = 2 and then with n = 4, and compare the results with the given rounded true values:

(a)
$$\int_0^1 \frac{dx}{1+x^2} = 0.7854$$
, (b) $\int_0^\pi \frac{\sin x}{x} dx = 1.8518$,
(c) $\int_0^{\pi/2} \sqrt{1-\frac{1}{2}\sin^2 x} dx = 1.3506$.

CHAPTER 4

Series Solutions of Differential Equations: Special Functions

4.1. Properties of power series. A large class of ordinary differential equations possesses solutions expressible, over a certain interval, in terms of power series and related series. Before investigating methods of obtaining such solutions, we review without proof certain useful properties of power series.

An expression of the form

$$A_0 + A_1(x - x_0) + \dots + A_n(x - x_0)^n + \dots = \sum_{n=0}^{\infty} A_n(x - x_0)^n \qquad (1)$$

is called a *power series* and is defined as the limit

$$\lim_{N\to\infty}\sum_{n=0}^N A_n(x-x_0)^n$$

for those values of x for which the limit exists. For such values of x the series is said to *converge*. In this chapter we suppose that the variable x and the coefficients are *real*; *complex* power series are dealt with in Chapter 10.

To determine for what values of x the series (1) converges, we may make use of the *ratio test*, which states that if the absolute value of the ratio of the (n + 1)th term to the *n*th term in *any* infinite series approaches a limit ρ as $n \to \infty$, then the series converges when $\rho < 1$ and diverges when $\rho > 1$. The test fails if $\rho = 1$. A more delicate test states that if the absolute value of the same ratio is bounded by some number σ as $n \to \infty$, then the series converges when $\sigma < 1$. In the case of the power series (1) we obtain

$$\rho = \lim_{n \to \infty} \left| \frac{A_{n+1}}{A_n} \right| |x - x_0| = L |x - x_0|,$$

$$L = \lim_{n \to \infty} \left| \frac{A_{n+1}}{A_n} \right|, \tag{2}$$

if the last limit exists. In this case it follows that (1) converges when

$$|x-x_0| < \frac{1}{L} \tag{3}$$

and diverges when

$$|x - x_0| > \frac{1}{L}.$$
 (4)

Thus, when L exists and is finite, an interval of convergence

$$\left(x_0-\frac{1}{L}, x_0+\frac{1}{L}\right)$$

is determined symmetrically about the point x_0 , such that inside the interval the series converges and outside the interval it diverges. The distance 1/L is frequently called the *radius of convergence*.

The behavior of the series at the end points of the interval is not determined by the ratio test. Useful tests for investigating convergence of the two series of constants corresponding to the end points $x = x_0 \pm L^{-1}$ are:

(1) If, at an end point, the successive terms of the series alternate in sign for sufficiently large values of n, the series converges if after a certain stage the successive terms always decrease in magnitude and if the nth term approaches zero, and the series diverges if the nth term does not tend to zero.

(2) If, at an end point, the successive terms of the series *are of constant* sign, and if the ratio of the (n + 1)th term to the *n*th term can be written in the form

$$1-\frac{k}{n}+\frac{\theta_n}{n^2}$$
,

where k is independent of n and θ_n is bounded as $n \to \infty$, then the series converges if k > 1 and diverges if $k \le 1$. It should be noticed that this test is applicable even in the case when k = 1, so long as an expression of the indicated form can be obtained. We shall refer to this test as *Raabe's test.**

As an example, we consider the series

$$\sum_{n=1}^{\infty} \frac{[1 \cdot 2 \cdot 3 \cdots n] (x-a)^n}{(\alpha+1)(\alpha+2)(\alpha+3) \cdots (\alpha+n)}$$

where α is not a negative integer. In this case we obtain

$$L = \lim_{n \to \infty} \left| \frac{n+1}{\alpha + n + 1} \right| = 1.$$

* This very useful test is also associated with the name of Gauss The term θ_n/n^2 can in fact be replaced by θ_n/n^{1+p} , where it is required only that p > 0 and θ_n be bounded, and still more delicate modifications exist. (See, for example, Reference 8.)

where

Hence the interval of convergence is given by |x - a| < 1 or

$$a-1 < x < a+1.$$

When x = a - 1, the signs of successive terms alternate when $n > -\alpha$. Apart from algebraic sign, the *n*th term is then

$$\frac{n!}{(\alpha+1)(\alpha+2)\cdots(\alpha+n)}=\frac{n!\ \Gamma(\alpha+1)}{\Gamma(\alpha+n+1)}.$$

Reference to Equation (61) of Chapter 2 shows that this ratio is approximated by $\Gamma(\alpha + 1) n^{-\alpha}$ when *n* is large, and hence approaches zero as *n* increases only if $\alpha > 0$. Thus the series converges at x = a - 1 if $\alpha > 0$, and diverges at x = a - 1 otherwise. When x = a + 1, the terms are of constant sign when $n > -\alpha$. The ratio of consecutive terms is then

$$\frac{n+1}{n+\alpha+1} = 1 - \frac{\alpha}{n} + \frac{\alpha(\alpha+1)}{n(n+\alpha+1)}$$

Hence, by Raabe's test with $\theta_n = [\alpha(\alpha + 1)n]/(n + \alpha + 1)$, the series converges at x = a + 1 if $\alpha > 1$ and diverges at x = a + 1 otherwise.

It may be noticed that if L is zero, the interval of convergence includes all values of x. However, if L is infinite, the series converges only at the point $x = x_0$. Whether or not the limit L exists, it is known that always, for the power series (1), either the series converges only when $x = x_0$, or the series converges everywhere, or there exists a positive number R such that the series converges when $|x - x_0| < R$ and diverges when $|x - x_0| > R$.

It may happen that the series (1) contains only terms for which the subscript n is an integral multiple of an integer N > 1, and hence is of the form

$$A_0 + A_N(x - x_0)^N + A_{2N}(x - x_0)^{2N} + \cdots = \sum_{k=0}^{\infty} A_{kN}(x - x_0)^{kN}.$$
 (5)

Examples are afforded by the Maclaurin series for the functions $\cos x$ and $\log (1 - x^4)$,

$$\cos x = 1 - \frac{x^2}{2!} + \frac{x^4}{4!} + \dots = \sum_{k=0}^{\infty} (-)^k \frac{x^{2k}}{(2k)!} \qquad (|x| < \infty),$$
$$-\log (1 - x^4) = x^4 + \frac{x^8}{2} + \frac{x^{12}}{3} + \dots = \sum_{k=1}^{\infty} \frac{x^{4k}}{k} \qquad (|x| < 1),$$

for which N = 2 and 4, respectively.

In such cases, the ratio A_{n+1}/A_n is undefined for infinitely many values of n. The limiting absolute value of the ratio of successive terms is

$$\rho_N = \lim_{k \to \infty} \left| \frac{A_{(k+1)N}}{A_{kN}} \right| |x - x_0|^N = L_N |x - x_0|^N,$$

Series solutions of differential equations | chap. 4

$$L_N = \lim_{k \to \infty} \left| \frac{A_{(k+1)N}}{A_{kN}} \right|,\tag{6}$$

and the series converges when $L_N | x - x_0 |^N < 1$, or

$$|x - x_0| < \frac{1}{\sqrt[N]{L_N}}.$$
(7)

A particularly useful property of power series is the fact that convergent power series can be treated, for many purposes, in the same way as polynomials. Inside its interval of convergence, a power series represents a continuous function of x with continuous derivatives of all orders. Inside this interval, a power series can be integrated or differentiated term by term, as in the case of a polynomial, and the resultant series will converge, *in the same interval*, to the integral or derivative of the function represented by the original series. Further, two power series in $x - x_0$ can be multiplied together term by term and the resultant series will converge to the product of the functions represented by the original series, *over the common interval of convergence*. A similar statement applies to division of one series by another, provided that the denominator is not zero at x_0 . Here the resultant series will converge to the ratio over *some* subinterval of the common interval of convergence.

Now suppose that the series $\sum_{n=0}^{\infty} A_n (x - x_0)^n$ converges in a nonzero interval about $x = x_0$ and hence represents a function, say f(x), in that interval,

$$f(x) = \sum_{n=0}^{\infty} A_n (x - x_0)^n.$$
 (8)

Then, differentiating both sides of Equation (8) k times and setting $x = x_0$ in the result, we obtain

$$f^{(k)}(x_0) = k! A_k \qquad (k = 0, 1, 2, ...),$$

and hence (8) becomes

$$f(x) = \sum_{n=0}^{\infty} \frac{f^{(n)}(x_0)}{n!} (x - x_0)^n.$$
 (9)

This is the so-called *Taylor series* expansion of f(x) near $x = x_0$. It is clear that not all functions possess such expansions, since, in particular, in order that (9) be defined, *all derivatives* of f(x) must exist at $x = x_0$. A function which possesses such an expansion is said to be *regular* at $x = x_0$. The above derivation shows that if a function is regular at $x = x_0$, it has only one expansion in powers of $x - x_0$ and that that expansion is given by (9).

where

If f(x) and all its derivatives are continuous in an interval including $x = x_0$, then f(x) can be expressed as a *finite* Taylor series plus a remainder, in the form

$$f(x) = \sum_{n=0}^{N-1} \frac{f^{(n)}(x_0)}{n!} (x - x_0)^n + R_N(x).$$
(10)

Here R_N , the remainder after N terms, is given by

$$R_N(x) = \frac{f^{(N)}(\xi)}{N!} (x - x_0)^N, \qquad (11)$$

where ξ is some point in the interval (x_0, x) . To show that the expansion (9) is valid, so that f(x) is regular at x_0 , we must show that $R_N(x) \to 0$ as $N \to \infty$ for values of x in an interval including $x = x_0$. A test which is much more easily applied and which is sufficient in the case of most functions occurring in practice consists of determining whether the formal series in Equation (9) converges in an interval about $x = x_0$.

It is apparent, in particular, that any *polynomial* in x is regular for all x. Further, any *rational function* (*ratio* of polynomials) is regular for all values of x which are not zeros of the denominator.

4.2. Illustrative examples. To illustrate the use of power series in obtaining solutions of differential equations, we first consider the solution of three specific linear equations of second order.

(1) To solve the differential equation

$$Ly \equiv \frac{d^2y}{dx^2} - y = 0, \qquad (12)$$

we assume a solution in the form

$$y = A_0 + A_1 x + A_2 x^2 + A_3 x^3 + A_4 x^4 + A_5 x^5 + \cdots$$

and assume that the series converges in an interval including x = 0. Differentiating twice term by term, we then obtain

$$\frac{d^2y}{dx^2} = 2A_2 + 6A_3x + 12A_4x^2 + 20A_5x^3 + \cdots$$

With the assumed form for y there follows

$$L y \equiv (2A_2 - A_0) + (6A_3 - A_1)x + (12A_4 - A_2)x^2 + (20A_5 - A_3)x^3 + \cdots = 0.$$

In order that this equation be valid over an interval, it is necessary that the

coefficients of all powers of x vanish independently, giving the equations

 $2A_2 = A_0, \quad 6A_3 = A_1, \quad 12A_4 = A_2, \quad 20A_5 = A_3, \quad \ldots,$

from which there follows

$$A_2 = \frac{1}{2}A_0, \qquad A_3 = \frac{1}{6}A_1, \qquad A_4 = \frac{1}{12}A_2 = \frac{1}{24}A_0,$$
$$A_5 = \frac{1}{20}A_3 = \frac{1}{120}A_1, \qquad \dots$$

The solution then becomes

$$y = A_0(1 + \frac{1}{2}x^2 + \frac{1}{24}x^4 + \cdots) + A_1(x + \frac{1}{6}x^3 + \frac{1}{120}x^5 + \cdots).$$

It is seen that the coefficients A_0 and A_1 are undetermined, and hence arbitrary, but that succeeding coefficients are determined in terms of them. The general solution is thus of the form

$$y = A_0 u_1(x) + A_1 u_2(x),$$

where $u_1(x)$ and $u_2(x)$ are two linearly independent solutions, expressed in power series, of which the first three terms have been obtained. The terms found may be recognized as the first terms of the series representing the known solutions cosh x and sinh x, respectively.

A more compact and convenient procedure uses the summation notation in place of writing out a certain number of terms of the series. Thus, we write the assumed solution in the form

$$y = \sum_{k=0}^{\infty} A_k x^k$$

and obtain, by differentiation,

$$\frac{d^2y}{dx^2} = \sum_{k=0}^{\infty} k(k-1)A_k x^{k-2}.$$

There then follows

$$Ly = \sum_{k=0}^{\infty} k(k-1)A_k x^{k-2} - \sum_{k=0}^{\infty} A_k x^k = 0.$$

In order to collect the coefficients of like powers of x, we next change the indices of summation in such a way that the exponents of x in the two summations are equal. For this purpose we may, for example, replace k by k - 2 in the second summation, so that it becomes

$$\sum_{k=2=0}^{\infty} A_{k-2} x^{k-2} \equiv \sum_{k=2}^{\infty} A_{k-2} x^{k-2},$$

and hence

$$Ly \equiv \sum_{k=0}^{\infty} k(k-1)A_k x^{k-2} - \sum_{k=2}^{\infty} A_{k-2} x^{k-2} = 0.$$

Since the first two terms (k = 0, 1) of the first summation are zero, we may replace the lower limit by k = 2 and then combine the summations to obtain

$$Ly = \sum_{k=2}^{\infty} [k(k-1)A_k - A_{k-2}]x^{k-2} = 0.$$

Equating to zero the coefficients of all powers of x involved in this sum, we have the condition

$$k(k-1)A_k = A_{k-2}$$
 $(k = 2, 3, ...).$

This condition is known as the recurrence formula for A_k . It expresses each coefficient A_k for which $k \ge 2$ as a multiple of the second preceding coefficient A_{k-2} , and reduces to the previously determined conditions when k = 2, 3, 4, and 5.

(2) As a second example we consider the equation

$$Ly \equiv x^{2} \frac{d^{2}y}{dx^{2}} + (x^{2} + x) \frac{dy}{dx} - y = 0.$$
 (13)

Assuming a solution of the form

$$y = \sum_{k=0}^{\infty} A_k x^k,$$

we obtain

$$\frac{dy}{dx} = \sum_{k=0}^{\infty} k A_k x^{k-1}, \qquad \frac{d^2 y}{dx^2} = \sum_{k=0}^{\infty} k(k-1) A_k x^{k-2},$$

and hence

$$Ly = \sum_{k=0}^{\infty} k(k-1)A_k x^k + \sum_{k=0}^{\infty} k A_k x^{k+1} + \sum_{k=0}^{\infty} k A_k x^k - \sum_{k=0}^{\infty} A_k x^k.$$

The first, third, and fourth summations may be combined to give

$$\sum_{k=0}^{\infty} [k(k-1) + k - 1] A_k x^k = \sum_{k=0}^{\infty} (k^2 - 1) A_k x^k,$$

and hence there follows

$$Ly = \sum_{k=0}^{\infty} (k^2 - 1)A_k x^k + \sum_{k=0}^{\infty} k A_k x^{k+1}.$$

In order to combine these sums, we replace k by k-1 in the second, to obtain

$$Ly \equiv \sum_{k=0}^{\infty} (k^2 - 1)A_k x^k + \sum_{k=1}^{\infty} (k - 1)A_{k-1} x^k.$$

Since the ranges of summation differ, the term corresponding to k = 0 must

be extracted from the first sum, after which the remainder of the first sum can be combined with the second. In this way we find

$$Ly \equiv -A_0 + \sum_{k=1}^{\infty} \left[(k^2 - 1)A_k + (k - 1)A_{k-1} \right] x^k.$$

In order that Ly may vanish identically, the constant term, as well as the coefficients of the successive powers of x, must vanish independently, giving the condition

$$A_{0} = 0$$

and the recurrence formula

$$(k-1)[(k+1)A_k + A_{k-1}] = 0$$
 $(k = 1, 2, 3, ...).$

The recurrence formula is *automatically* satisfied when k = 1. When $k \ge 2$, it becomes

$$A_k = -\frac{A_{k-1}}{k+1}$$
 (k = 2, 3, 4, ...).

Hence, we obtain

$$A_2 = -\frac{A_1}{3}, \quad A_3 = -\frac{A_2}{4} = \frac{A_1}{3 \cdot 4}, \quad A_4 = -\frac{A_3}{5} = -\frac{A_1}{3 \cdot 4 \cdot 5}, \quad \dots$$

Thus, in this case $A_0 = 0$, A_1 is *arbitrary*, and all succeeding coefficients are determined in terms of A_1 . The solution becomes

$$y = A_1 \left(x - \frac{x^2}{3} + \frac{x^3}{3 \cdot 4} - \frac{x^4}{3 \cdot 4 \cdot 5} + \cdots \right).$$

If this solution is put in the form

$$y = \frac{2A_1}{x} \left(\frac{x^2}{2!} - \frac{x^3}{3!} + \frac{x^4}{4!} - \frac{x^5}{5!} + \cdots \right)$$
$$= \frac{2A_1}{x} \left[x - 1 + \left(1 - \frac{x}{1!} + \frac{x^2}{2!} - \frac{x^3}{3!} + \frac{x^4}{4!} - \cdots \right) \right],$$

the series in parentheses in the final form is recognized as the expansion of e^{-x} , and, writing $2A_1 = c$, the solution obtained may be put in the closed form

$$y=c\bigg(\frac{e^{-x}-1+x}{x}\bigg).$$

In this case only one solution was obtained. This fact indicates that any linearly independent solution cannot be expanded in power series near x = 0; that is, it is *not regular* at x = 0. Although a second solution could be obtained by the method of Section 1.10, an alternative procedure given in a following section is somewhat more easily applied.

(3) As a final example we consider the equation

$$Ly \equiv x^{3} \frac{d^{2}y}{dx^{2}} + y = 0.$$
 (14)

Again assuming a solution of the form

$$y = \sum_{k=0}^{\infty} A_k x^k,$$

we obtain

$$Ly = \sum_{k=0}^{\infty} k(k-1)A_{k}x^{k+1} + \sum_{k=0}^{\infty} A_{k}x^{k}.$$

Replacing k by k - 1 in the first sum, there follows

$$Ly \equiv \sum_{k=1}^{\infty} (k-1)(k-2)A_{k-1}x^{k} + \sum_{k=0}^{\infty} A_{k}x^{k}$$
$$\equiv A_{0} + \sum_{k=1}^{\infty} [A_{k} + (k-1)(k-2)A_{k-1}]x^{k}$$

The condition L y = 0 then requires that

 $A_{0} = 0$

and that the succeeding coefficients satisfy the recurrence formula

$$A_k = -(k-1)(k-2)A_{k-1}$$
 $(k = 1, 2, ...).$

For k = 1 and k = 2, the recurrence formula gives $A_1 = A_2 = 0$; and since, from this point, the remaining conditions express each A as a multiple of the preceding one, it follows that all the A's must be zero. Hence the only solution obtained is the trivial one y = 0. It thus follows that the equation possesses no nontrivial solutions which are regular at x = 0.

Next we proceed to a classification of types of linear differential equations of second order, and to a study of the basic differences among the three problems so far considered.

4.3. Singular points of linear second-order differential equations. If a homogeneous second-order linear differential equation is written in the standard form

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x) y = 0,$$
 (15)

the behavior of solutions of the equation near a point $x = x_0$ is found to depend upon the behavior of the coefficients $a_1(x)$ and $a_2(x)$ near $x = x_0$. The point $x = x_0$ is said to be an ordinary point of the differential equation if both

 $a_1(x)$ and $a_2(x)$ are regular at $x = x_0$, that is, if a_1 and a_2 can be expanded in power series in an interval including $x = x_0$. Otherwise, the point $x = x_0$ is said to be a singular point of the differential equation. In such a case, if the products $(x - x_0)a_1(x)$ and $(x - x_0)^2a_2(x)$ are both regular at $x = x_0$, the point $x = x_0$ is said to be a regular singular point; otherwise, the point is called an irregular singular point.

In illustration, we notice that for the first differential equation of the preceding section, $a_1(x) = 0$ and $a_2(x) = -1$. Thus all points are ordinary points. In the second example the coefficients $a_1(x) = 1 + x^{-1}$ and $a_2(x) = -x^{-2}$ cannot be expanded in powers of x but can be expanded in power series near any other point $x = x_0$. Thus the point x = 0 is the only singular point. Since the products $xa_1(x) = x + 1$ and $x^2a_2(x) = -1$ are regular at x = 0, it follows that the point is a regular singular point. In the third case the point x = 0 is readily seen to be an irregular singular point. Similarly, for the equation

$$x^{3}(1-x)^{2}\frac{d^{2}y}{dx^{2}}-2x^{2}(1-x)\frac{dy}{dx}+3y=0,$$

it can be verified that x = 0 is an irregular singular point and x = 1 is a regular singular point. All other points are ordinary points.

When the coefficients $a_1(x)$ and $a_2(x)$ in the standard form (15) are ratios of polynomials, singular points can occur only when a denominator is zero, so that, unless the numerator also vanishes there, the corresponding coefficient is not *finite*. Most of the equations considered in this text will be of this type.

However, a singular point also may occur when $a_1(x)$ or $a_2(x)$ becomes infinite in some other way, or even in the absence of such behavior. For example, if $a_1(x) = (x - 1)^{5/3}$, it is seen that $a''_1(x)$ becomes infinite as $x \to 1$, so that $a_1(x)$ cannot be expanded in a series of powers of x - 1. Since the function $(x - 1)a_1(x) = (x - 1)^{7/3}$ also has the same property, it follows that the differential equation (15) has, in fact, an *irregular* singular point at x = 1when $a_1(x) = (x - 1)^{5/3}$.

It will be shown that if $x = x_0$ is an ordinary point of (15), then the equation possesses two linearly independent solutions which are regular at $x = x_0$, and

hence are both expressible in the form $\sum_{k=0}^{\infty} A_k (x - x_0)^k$. If $x = x_0$ is a regular

singular point of (15), it will be shown that the equation does not necessarily possess any nontrivial solution which is regular near $x = x_0$, but that at least one solution exists of the form

$$y = (x - x_0)^s \sum_{k=0}^{\infty} A_k (x - x_0)^k$$

where s is a determinable number which may be real or complex. Such a solution is regular at $x = x_0$ only if s is zero or a positive integer. If $x = x_0$ is

an *irregular singular point* of (15), the problem is more involved; a nontrivial regular solution may or may not exist.

In illustration, the equation

$$x^2\frac{d^2y}{dx^2} + (1+2x)\frac{dy}{dx} = 0$$

has an irregular singular point at x = 0. The general solution is

$$y = c_1 e^{1/x} + c_2;$$

thus, the solution y = constant is the only solution regular at x = 0. The equation

$$x^4 \frac{d^2 y}{dx^2} + 2x^3 \frac{dy}{dx} + y = 0$$

also has an irregular singular point at x = 0 and has the general solution

$$y = c_1 \sin \frac{1}{x} + c_2 \cos \frac{1}{x}.$$

It follows from the nature of these functions that this equation has neither a nontrivial regular solution at x = 0 nor a solution expressible in the form

$$y = x^s \sum_{k=0}^{\infty} A_k x^k.$$

4.4. The method of Frobenius. In this section we restrict attention to solutions valid in the neighborhood of the point x = 0. Solutions valid near a more general point $x = x_0$ may be obtained in an analogous way, although for this purpose it is frequently more convenient first to replace $x - x_0$ by a new variable t and then to determine solutions of the transformed differential equation near the point t = 0.

In place of reducing a second-order linear equation to the standard form (15), it is frequently more convenient to use a form which is, to some extent, cleared of fractions, particularly if $a_1(x)$ or $a_2(x)$ is the ratio of two polynomials. For this reason, to investigate solutions valid near x = 0, we suppose that the equation has been put in the form

$$Ly \equiv R(x)\frac{d^2y}{dx^2} + \frac{1}{x}P(x)\frac{dy}{dx} + \frac{1}{x^2}Q(x)y = 0,$$
 (16)

where R(x) does not vanish in some interval including x = 0. We also suppose that P(x), Q(x), and R(x) are regular at x = 0. Then, with the notation of (15), the products

$$xa_1(x) = \frac{P(x)}{R(x)}$$
 and $x^2a_2(x) = \frac{Q(x)}{R(x)}$

are regular at x = 0, and this point is either an ordinary point or, at worst, a regular singular point.

It is convenient to suppose also that the original equation has been divided through by a suitable constant so that R(0) = 1. Then we may write

$$P(x) = P_0 + P_1 x + P_2 x^2 + \cdots,$$

$$Q(x) = Q_0 + Q_1 x + Q_2 x^2 + \cdots,$$

$$R(x) = 1 + R_1 x + R_2 x^2 + \cdots,$$
(17)

the series converging in some interval including x = 0.

We attempt to find nontrivial solutions which are in the form of a power series in x multiplied by a power of x,

$$-y = x^{s} \sum_{k=0}^{\infty} A_{k} x^{k} = A_{0} x^{s} + A_{1} x^{s+1} + A_{2} x^{s+2} + \cdots, \qquad (18)$$

where s is to be determined. The number A_0 is now, by assumption, the coefficient of the *first term* in the series, and hence must not vanish. Substitution into the left-hand member of (16) then gives*

$$L y \equiv (1 + R_1 x + R_2 x^2 + \cdots) \times [s(s-1)A_0 x^{s-2} + (s+1)sA_1 x^{s-1} + (s+2)(s+1)A_2 x^s + \cdots] + (P_0 + P_1 x + P_2 x^2 + \cdots) \times [sA_0 x^{s-2} + (s+1)A_1 x^{s-1} + (s+2)A_2 x^s + \cdots] + (Q_0 + Q_1 x + Q_2 x^2 + \cdots) \times [A_0 x^{s-2} + A_1 x^{s-1} + A_2 x^s + \cdots],$$

or, multiplying term by term and collecting the coefficients of successive powers of x,

$$L y \equiv [s(s-1) + P_0 s + Q_0] A_0 x^{s-2} + \{[(s+1)s + P_0(s+1) + Q_0] A_1 + [R_1 s(s-1) + P_1 s + Q_1] A_0\} x^{s-1} + \{[(s+2)(s+1) + P_0(s+2) + Q_0] A_2 + [R_1(s+1)s + P_1(s+1) + Q_1] A_1 + [R_2 s(s-1) + P_2 s + Q_2] A_0\} x^s + \cdots$$
(19)

Equation (19) can be put in a convenient form in terms of the functions

$$f(s) = s(s-1) + P_0 s + Q_0 = s^2 + (P_0 - 1)s + Q_0$$
(20)

and

$$g_n(s) = R_n(s-n)(s-n-1) + P_n(s-n) + Q_n$$

= $R_n(s-n)^2 + (P_n - R_n)(s-n) + Q_n.$ (21)

* A more compact development is obtained by using summation notation.

With the notation of (20) and (21), Equation (19) then becomes

$$Ly \equiv f(s) A_0 x^{s-2} + [f(s+1)A_1 + g_1(s+1)A_0] x^{s-1} + [f(s+2)A_2 + g_1(s+2)A_1 + g_2(s+2)A_0] x^s + \cdots + [f(s+k)A_k + \sum_{n=1}^k g_n(s+k)A_{k-n}] x^{s+k-2} + \cdots .$$
(22)

In order that (16) be satisfied in an interval including x = 0, this expression must vanish *identically*, in the sense that the coefficients of all powers of x in (22) must vanish independently. The vanishing of the coefficient of the *lowest* power x^{s-2} gives the requirement

$$f(s) = 0$$
 or $s^2 + (P_0 - 1)s + Q_0 = 0.$ (23)

This equation determines two values of s (which may however be equal) and is called the *indicial equation*. The two values of s, which give the exponents of the leading terms of two series of the form (18), are called the *exponents of* the differential equation at x = 0.

For each such value of s, the vanishing of the coefficient of x^{s-1} in (22) gives the requirement

$$f(s+1)A_1 = -g_1(s+1)A_0$$

and hence determines A_1 in terms of A_0 if $f(s + 1) \neq 0$. Next, the vanishing of the coefficient of x^s in (22) determines A_2 in terms of A_1 and A_0 ,

$$f(s+2)A_2 = -g_1(s+2)A_1 - g_2(s+2)A_0,$$

and hence in terms of A_0 , if $f(s + 2) \neq 0$.

In general, the vanishing of the coefficient of x^{s+k-2} in Equation (22) gives the *recurrence formula*

$$f(s+k)A_{k} = -\sum_{n=1}^{k} g_{n}(s+k)A_{k-n} \qquad (k \ge 1), \qquad (24)$$

which determines each A_k in terms of the preceding A's, and hence in terms of A_0 , if for each k the quantity f(s + k) is not zero.

Thus, if two distinct values of s are determined by (23), and if for each such value of s the quantity f(s + k) is never zero for any positive integer k, the coefficients of two series of the form (18) are determined and these series are solutions of (16) in their interval of convergence. We next investigate the exceptional cases.

Let the roots of Equation (23) be $s = s_1$ and $s = s_2$, where

$$s_{1} = \frac{1 - P_{0}}{2} + \frac{1}{2}\sqrt{(1 - P_{0})^{2} - 4Q_{0}}$$

$$s_{2} = \frac{1 - P_{0}}{2} - \frac{1}{2}\sqrt{(1 - P_{0})^{2} - 4Q_{0}}$$
(25)

The first exceptional case is then the case when the exponents s_1 and s_2 are equal,

$$(1 - P_0)^2 - 4Q_0 = 0. (26)$$

In this case only one solution of the form (18) can be obtained.

Now suppose that the two exponents are distinct. The second exceptional case may then arise if $f(s_1 + k)$ or $f(s_2 + k)$ vanishes for a positive integral value of k, say k = K, so that (24) cannot be solved for the coefficient A_K . With the notation of (25), we have

$$f(s) = (s - s_1)(s - s_2)$$

and hence

$$f(s + k) = (s + k - s_1)(s + k - s_2),$$

from which there follows

$$f(s_1 + k) = k[k + (s_1 - s_2)], \quad f(s_2 + k) = k[k - (s_1 - s_2)].$$
 (27)

If s_1 is imaginary and the coefficients P_k , Q_k , and R_k are all real, then s_2 is the conjugate complex number. Hence, in this case $s_1 - s_2$ is imaginary and the expressions in (27) cannot vanish for any real values of k except k = 0. Next suppose that s_1 and s_2 are real and distinct and that $s_1 > s_2$. Then, since $s_1 - s_2 > 0$, it follows that $f(s_1 + k)$ cannot vanish for $k \ge 1$ and that $f(s_2 + k)$ can vanish only when $k = s_1 - s_2$. Since k may take on only positive integral values, this condition is possible only if $s_1 - s_2$ is a positive integer. If $s_1 = s_2$, then $f(s_1 + k) = k^2$, and hence $f(s_1 + k)$ cannot vanish when $k \ge 1$.

We thus see that if the two exponents s_1 and s_2 do not differ by zero or a positive integer, two distinct solutions of type (18) are obtained. If the exponents are equal, one such solution is obtained; whereas if the exponents differ by a positive integer, a solution of type (18) corresponding to the larger exponent is obtained.

It is known (see, for example, Reference 5 of Chapter 1) that the interval of convergence of each series so obtained is at least the largest interval, centered at x = 0, inside which the expansions of $xa_1(x)$ and $x^2a_2(x)$ in powers of x both converge, with the natural understanding that the point x = 0 itself must be excluded when the exponent $(s_1 \text{ or } s_2)$ is negative or has a negative real part.

When x is complex, each infinite series converges to a solution in a circle in the complex plane, with center at the origin and radius at least the distance to the nearest singularity of $a_1(x)$ or $a_2(x)$, and with the center deleted when necessary. The solution then is said to have a *pole* at the origin when the associated exponent is a negative integer, and a *branch point* at the origin when the exponent is nonintegral, as well as in the exceptional cases (Section 4.5) when the function log x is involved (see Chapter 10).

If $s_1 - s_2 = K$, where K is a positive integer, then when k = K the recurrence formula (24) becomes

$$(s - s_2)(s - s_2 + K)A_K = -\sum_{n=1}^{K} g_n(s + K)A_{K-n}.$$
 (28)

Thus, as we have seen, the left member vanishes when $s = s_2$, and the equation cannot be satisfied by any value of A_K unless it happens that the right member is also zero, in which case the coefficient A_K is undetermined, and hence arbitrary. If this condition exists, a solution of type (18) is then obtained, corresponding to the smaller exponent s_2 , which contains two arbitrary constants A_0 and A_K , and hence is the complete solution. Thus we conclude that if the exponents differ by a positive integer, either no solution of type (18) is obtained for the smaller exponent or two independent solutions are obtained. In the latter case the two solutions so obtained must then include the solution corresponding to the larger exponent as the coefficient of A_K .

It is important to notice that this is the situation, for example, when x = 0 is an *ordinary point*. For in this case one has $P_0 = Q_0 = Q_1 = 0$, and hence $s_1 = 1$, $s_2 = 0$, and $K = s_1 - s_2 = 1$. Thus Equation (28) here becomes

$$s(s + 1)A_1 = -g_1(s + 1)A_0$$

= -[R_1s^2 + (P_1 - R_1)s]A_0
= -s[R_1s + (P_1 - R_1)]A_0,

and when $s = s_2 = 0$ the recurrence formula is identically satisfied, leaving A_1 as well as A_0 arbitrary. Thus, when x = 0 is an ordinary point, two linearly independent solutions which are regular at x = 0 are obtained.

The preceding detailed derivation was intended for the purpose of investigating the *existence* of series solutions of the assumed type. Although the formulas obtained can be used directly for the determination of the coefficients, once the functions f(s), $g_1(s)$, $g_2(s)$, ... are identified, it is suggested instead that the indicial equation and the recurrence formulas be obtained in actual practice by direct substitution of the assumed series into the differential equation, written in any convenient form.

To illustrate the application of the preceding treatment, we consider again the second example, Equation (13), of Section 4.2. We thus seek solutions of the equation

$$Ly \equiv x^{2} \frac{d^{2}y}{dx^{2}} + (x^{2} + x) \frac{dy}{dx} - y = 0$$

of the form

$$y = x^s \sum_{k=0}^{\infty} A_k x^k = \sum_{k=0}^{\infty} A_k x^{k+s}.$$
By direct substitution, there follows

$$Ly = (s^{2} - 1)A_{0}x^{s} + \sum_{k=1}^{\infty} \{ [(s+k)^{2} - 1]A_{k} + (s+k-1)A_{k-1} \} x^{k+s} \}$$

Hence the indicial equation is

$$s^2 - 1 = 0$$

and the recurrence formula is

$$[(s+k)^2 - 1]A_k + (s+k-1)A_{k-1} = 0$$

(s+k-1)[(s+k+1)A_k + A_{k-1}] = 0 (k \ge 1).

or

The exponents s_1 and s_2 are +1 and -1, respectively. Since they differ by an integer, a solution of the required type is *assured* only when s has the larger value +1.

With s = +1, the recurrence formula becomes

$$k[(k+2)A_{k}+A_{k-1}]=0,$$

or, since $k \neq 0$,

$$A_k = -\frac{A_{k-1}}{k+2} \qquad (k \ge 1).$$

Thus, one has

$$A_1 = -\frac{A_0}{3}, \quad A_2 = \frac{A_0}{3 \cdot 4}, \quad A_3 = -\frac{A_0}{3 \cdot 4 \cdot 5}, \quad \dots,$$

and hence the solution corresponding to s = 1 is

$$y = x \left[A_0 - \frac{A_0}{3} x + \frac{A_0}{3 \cdot 4} x^2 - \frac{A_0}{3 \cdot 4 \cdot 5} x^3 + \cdots \right]$$

= $A_0 \left(x - \frac{x^2}{3} + \frac{x^3}{3 \cdot 4} - \frac{x^4}{3 \cdot 4 \cdot 5} + \cdots \right)$
= $2A_0 \left(\frac{e^{-x} - 1 + x}{x} \right),$

in accordance with the result obtained previously.

With s = -1, the recurrence formula becomes

$$(k-2)(k A_k + A_{k-1}) = 0$$
 $(k \ge 1).$

It is important to notice that the factor k - 2 cannot be canceled except on the understanding that $k \neq 2$. That is, when k = 2, the correct form of the recurrence formula is 0 = 0. If k = 1, there follows

$$A_1 = -A_0.$$

If k = 2, the recurrence formula is identically satisfied, so that A_2 is *arbitrary*.

If $k \ge 3$, the recurrence formula can be written in the form

$$A_k = -\frac{A_{k-1}}{k} \qquad (k \ge 3).$$

If we take $A_2 = 0$, then there follows $A_3 = A_4 = \cdots = 0$, and the solution corresponding to s = -1 becomes simply

$$y = x^{-1}(A_0 - A_0 x) = A_0 \frac{1 - x}{x}$$

The general solution of the given equation is then a linear combination of the two solutions so obtained, and hence can be taken conveniently in the form

$$y = c_1 \frac{e^{-x} - 1 + x}{x} + c_2 \frac{1 - x}{x}$$

or, alternatively,

$$y = C_1 \frac{e^{-x}}{x} + C_2 \frac{1-x}{x},$$

where $C_1 = c_1$, and $C_2 = c_2 - c_1$. The only solution regular at x = 0 is the one formerly obtained,

$$y = c \, \frac{e^{-x} - 1 + x}{x}$$

It can be verified that if A_2 is left arbitrary, the solution corresponding to the exponent -1 is the sum of A_0 times the two-term solution obtained and A_2 times the infinite series solution corresponding to the exponent +1.

4.5. Treatment of exceptional cases.* We consider first the case of equal exponents, and attempt to determine a second solution which is independent of the one obtained by the method of Frobenius. Although this result could be accomplished by the methods of Section 1.10, the method to be given is usually more easily applied.

In place of first introducing the value of the repeated exponent s_1 into the recurrence formula (24) and determining $A_1, A_2, \ldots, A_k, \ldots$ directly in terms of A_0 , we suppose that the coefficients $A_1, A_2, \ldots, A_k, \ldots$ are expressed, by the recurrence formula, in terms of A_0 and s. We indicate this fact by writing $A_1 = A_1(s), A_2 = A_2(s), \ldots$. With these values of the A's, as functions of s, a function y depending upon s as well as x is determined and denoted by $y_s(x)$,

$$y_s(x) = x^s \sum_{k=0}^{\infty} A_k(s) x^k.$$
 (29)

* Sections 4.5 and 4.6, together with the derivation of the series for $Y_0(x)$ in Section 4.8, can be omitted without logical difficulty. However, in this event a consideration of the last paragraph of Section 4.5 is suggested.

Reference to Equation (22) shows that satisfaction of the recurrence formula, for $k \ge 1$, brings about vanishing of all terms in (22) except the first, and so there follows

$$L y_s(x) = A_0 f(s) x^{s-2}, (30)$$

or, since in the case of a repeated exponent s_1 we have $f(s) = (s - s_1)^2$,

$$L y_s(x) = A_0(s - s_1)^2 x^{s-2}.$$
 (31)

The fact that the right member of (31) vanishes when $s = s_1$ is in accordance with the known fact that (29) is a solution of (16), say $y_1(x)$, when $s = s_1$. However, since $s = s_1$ is a double root of the right member of (31), it follows also that the result of differentiating either member of (31) with respect to s (holding x constant),

$$\frac{\partial}{\partial s}Ly_s(x) = A_0[2(s-s_1)+(s-s_1)^2\log x]x^{s-2},$$

is zero when $s = s_1$. But, since the operator $\frac{\partial}{\partial s}$ and the linear operator L are commutative, there then follows also

$$\left[\frac{\partial}{\partial s}Ly_s(x)\right]_{s=s_1} = L\left[\frac{\partial y_s(x)}{\partial s}\right]_{s=s_1} = 0.$$
(32)

Hence a second solution of (16), when $s_1 = s_2$, is of the form

$$y_2(x) = \left[\frac{\partial y_s(x)}{\partial s}\right]_{s=s_1}$$
(33)

The second exceptional case is that in which the exponents differ by a positive integer K,

$$s_1-s_2=K\geq 1,$$

but where the recurrence formula is not identically satisfied when k = K and $s = s_2$, that is, when the right member of (28) does not vanish when $s = s_2$. In such a case, Equation (28) can be satisfied only if $A_0 = A_1 = \cdots = A_{K-1} = 0$, and hence Equation (16) does not possess a solution of type (18) beginning with a term of the form $A_0 x^{s_2}$.

In this case we suppose again that the recurrence formula is satisfied when $k \ge 1$ for all values of s, so that with the A's expressed in terms of A_0 and s, we again define a function $y_s(x)$ of type (29). In this case, however, it is clear from (28) and from the nature of the recurrence formula (24) that the expressions for the coefficients $A_K(s), A_{K+1}(s), \ldots$, will now all have a factor $s - s_2$ in a denominator, and hence will not approach finite limits as $s \to s_2$. If we consider the product $(s - s_2) y_s(x)$, we see that as $s \to s_2$ terms with coefficients A_k for which k < K will vanish and the remaining terms will approach finite limits, thus giving rise to an infinite series of powers of x starting with a term

involving $x^{s_2+K} = x^{s_1}$. Thus the limiting series must be proportional to the series for which $s = s_1$. In this case, however, since again satisfaction of the recurrence formula for $k \ge 1$ causes all terms of (22) except the first to vanish, we have

$$L(s - s_2) y_s(x) = A_0(s - s_2)^2(s - s_1)x^{s-2}.$$
 (34)

But since the right member has a double root $s = s_2$, the partial derivative of either member must vanish as $s \rightarrow s_2$, and, by an argument similar to that leading to (32), we conclude that

$$L\left\{\frac{\partial}{\partial s}\left[\left(s-s_{2}\right)y_{s}(x)\right]\right\}_{s=s_{2}}=0$$

so that the function

$$y_2(x) = \left\{ \frac{\partial}{\partial s} \left[(s - s_2) y_s(x) \right] \right\}_{s = s_2}$$
(35)

is a solution of (16), in addition to the solution $y_1(x) = [y_s(x)]_{s=s_1}$ corresponding to the larger exponent s_1 .

From Equations (29) and (33) it follows that when $s_2 = s_1$ the second solution y_2 is expressible as

$$y_2(x) = \left[\sum_{k=0}^{\infty} A_k(s_1) x^{k+s_1}\right] \log x + \sum_{k=0}^{\infty} A'_k(s_1) x^{k+s_1}, \quad (36)$$

and the coefficient of log x is seen to be $y_1(x)$. Further, when s_2 and s_1 differ by a positive integer but there is no Frobenius series solution with $s = s_2$, Equations (29) and (35) show that the missing solution y_2 is expressible as

$$y_{2}(x) = \left[\sum_{k=0}^{\infty} (s - s_{2})A_{k}(s)x^{k+s}\right]_{s=s_{2}} \log x + \sum_{k=0}^{\infty} \left\{\frac{d}{ds}\left[(s - s_{2})A_{k}(s)\right]\right\}_{s=s_{2}} x^{k+s_{2}}.$$
 (37)

The coefficient of log x is $\lim_{s \to s_2} [(s - s_2)y_s(x)]$, which has been seen to be proportional to $y_1(x)$.

Hence it follows that, in all cases when the differential equation, having x = 0 as a regular singular point with exponents s_1 and s_2 , possesses only one solution

$$y_1(x) = \sum_{k=0}^{\infty} A_k x^{k+s_1} \equiv A_0 u_1(x)$$

of the form (18), any independent solution is of the form

$$y_2(x) = Cu_1(x) \log x + \sum_{k=0}^{\infty} B_k x^{k+s_2}$$
 (38)

where C is a constant. Thus, in place of using the results of Equations (33) or (35) in such cases, a second solution may be obtained by directly assuming a solution of this last form and by determining the necessary relationships between the coefficients B_k and an arbitrarily chosen constant $C \neq 0$.

4.6. Example of an exceptional case. To illustrate the procedures developed in the preceding section, we consider the equation

$$L y = x \frac{d^2 y}{dx^2} - y = 0.$$
 (39)

With the notation

$$y_s(x) = \sum_{k=0}^{\infty} A_k(s) x^{k+s},$$
 (40)

there follows

$$L y_s = s(s-1)A_0 x^{s-1} + \sum_{k=1}^{\infty} [(k+s)(k+s-1)A_k - A_{k-1}] x^{k+s-1}.$$

Hence we obtain the recurrence formula

 $(k+s)(k+s-1)A_{k} = A_{k-1} \qquad (k \ge 1)$ (41)

and the two indices

$$s_1 = 1, \quad s_2 = 0.$$
 (42)

Since the indices differ by unity, a solution of the form (40) is assured only for s = 1.

From (41) there follows

$$A_1 = \frac{A_0}{(s+1)s}, \qquad A_2 = \frac{A_0}{(s+2)(s+1)^2s}, \qquad \dots,$$

and in general, by inductive reasoning,

$$A_k(s) = \frac{A_0}{(s+k)[(s+k-1)\cdots(s+1)]^2 s} \quad (k \ge 2).$$
(43)

The solution for which $s = s_1 = 1$ then becomes

$$y_1(x) = A_0 \sum_{k=0}^{\infty} \frac{x^{k+1}}{(k+1)! \, k!} \equiv A_0 u_1(x). \tag{44}$$

However, since $A_k(s) \to \infty$ when $s \to 0$, for all $k \ge 1$, there is indeed no solution of type (40) for which s = 0.

In order to obtain a second solution, we may refer to Equation (37). The coefficient of $\log x$ is seen to be

$$\sum_{k=0}^{\infty} [sA_k(s)]_{s=0} x^k = A_0 \sum_{k=1}^{\infty} \frac{x^k}{k! (k-1)!}$$
$$= A_0 \sum_{k=0}^{\infty} \frac{x^{k+1}}{(k+1)! k!} = A_0 u_1(x),$$
(45)

when account is taken of the vanishing of $(sA_0)_{s=0}$. The coefficient $\frac{d [sA_k(s)]}{ds}$, involved in the second series of Equation (37), is conveniently evaluated by logarithmic differentiation in the form

$$\frac{d}{ds}[sA_k(s)] = -\frac{\frac{1}{s+k} + 2\sum_{m=1}^{k-1} \frac{1}{s+m}}{(s+k)[(s+k-1)\cdots(s+1)]^2} A_0 \qquad (k \ge 2),$$

and hence there follows

$$\left\{\frac{d}{ds}\left[sA_{k}(s)\right]\right\}_{s=0} = -\frac{\varphi(k) + \varphi(k-1)}{k! (k-1)!} A_{0} \qquad (k \ge 2), \qquad (46)$$

with the abbreviation

-

$$\varphi(k) = \begin{cases} \sum_{m=1}^{k} \frac{1}{m} = 1 + \frac{1}{2} + \frac{1}{3} + \dots + \frac{1}{k} & (k = 1, 2, \dots) \\ 0 & (k = 0). \end{cases}$$
(47)

It is found by direct calculation that Equation (46) also holds for k = 1, whereas the right-hand member must be replaced by A_0 when k = 0. Hence (37) gives

$$y_2(x) = A_0 \left[u_1(x) \log x + 1 - \sum_{k=1}^{\infty} \frac{\varphi(k) + \varphi(k-1)}{k! (k-1)!} x^k \right]$$
(48)

or, in expanded form,

$$y_2(x) = A_0 \left[\left(x + \frac{1}{2}x^2 + \frac{1}{12}x^3 + \cdots \right) \log x + \left(1 - x - \frac{5}{4}x^2 - \frac{5}{18}x^3 - \cdots \right) \right].$$

The alternative procedure described at the end of the preceding section consists of substituting the relation of Equation (38) in the form

$$y_2(x) = C u_1(x) \log x + \sum_{k=0}^{\infty} B_k x^k$$
 (49)

directly into (39), to obtain the condition

$$C\left[\left(x\frac{d^{2}u_{1}}{dx^{2}}-u_{1}\right)\log x+2\frac{du_{1}}{dx}-\frac{1}{x}u_{1}\right]+\sum_{k=0}^{\infty}\left[(k+1)k\ B_{k+1}-B_{k}\right]x^{k}=0.$$
 (50)

Since u_1 satisfies Equation (39), the coefficient of log x in (50) vanishes, and the introduction of (45) reduces (50) to the form

$$\sum_{k=0}^{\infty} \left[(k+1)k B_{k+1} - B_k \right] x^k + C \sum_{k=0}^{\infty} \left[\frac{2}{(k!)^2} - \frac{1}{(k+1)! k!} \right] x^k = 0.$$

The requirement that the coefficient of the general power x^k vanish becomes

$$(k+1)k B_{k+1} - B_k = -\frac{2k+1}{(k+1)! k!} C \qquad (k \ge 0).$$

By setting k successively equal to $0, 1, 2, \ldots$, we obtain

$$B_0 = C$$
, $B_2 = \frac{1}{2}B_1 - \frac{3}{4}C$, $B_3 = \frac{1}{12}B_1 - \frac{7}{36}C$, ...

Here both B_1 and C are arbitrary and all the other coefficients are expressible in terms of these two constants, yielding the solution (49) in the form

$$y_{2}(x) = C \left[\left(x + \frac{1}{2}x^{2} + \frac{1}{12}x^{3} + \cdots \right) \log x + \left(1 - \frac{3}{4}x^{2} - \frac{7}{36}x^{3} + \cdots \right) \right] \\ + B_{1}\left(x + \frac{1}{2}x^{2} + \frac{1}{12}x^{3} + \cdots \right).$$
(51)

The coefficient of B_1 is seen to be $u_1(x)$. Thus, if C and B_1 are left as arbitrary constants, this expression for $y_2(x)$ in fact represents the general solution of (39). The particular expression for y_2 obtained in (48), by the first method, is obtained from (51) by choosing $C = -B_1 = A_0$.

4.7. A particular class of equations. Many important second-order equations, of frequent occurrence in practice, can be obtained by specializing the constants in the equation

$$(1 + R_M x^M) \frac{d^2 y}{dx^2} + \frac{1}{x} (P_0 + P_M x^M) \frac{dy}{dx} + \frac{1}{x^2} (Q_0 + Q_M x^M) y = 0, \quad (52)$$

where M is a positive integer. (In the case when M = 0, the equation is equidimensional.) Here the introduction of the assumption

$$y(x) = \sum_{k=0}^{\infty} A_k x^{k+s}$$
(53)

leads to the condition

$$\sum_{k=0}^{M-1} f(s+k)A_k x^{s+k-2} + \sum_{k=M}^{\infty} \left[f(s+k)A_k + g(s+k)A_{k-M} \right] x^{s+k-2} = 0,$$

where

$$f(s) = s^2 + (P_0 - 1)s + Q_0$$

 $g(s) \equiv g_M(s) = R_M(s - M)^2 + (P_M - R_M)(s - M) + Q_M.$

and

Thus, for each exponent satisfying the indicial equation f(s) = 0, the recurrence formula is

$$f(s+k)A_{k} = 0 \qquad (k = 1, 2, ..., M-1),$$

$$f(s+k)A_{k} = -g(s+k)A_{k-M} \qquad (k \ge M).$$

The first M - 1 conditions are satisfied by taking

$$A_1=A_2=\cdots=A_{M-1}=0$$

after which the recurrence formula for $k \ge M$ shows that all coefficients A_k for which k is not an integral multiple of M can be taken to be zero.

Accordingly, it is convenient to write

$$y(x) = \sum_{k=0}^{\infty} B_k x^{Mk+s}$$
(54)

when seeking a solution of a special case of Equation (52). Here k has been replaced by Mk in (53) and B_k has been written for A_{Mk} .

Further, it is seen that here an exceptional case can occur only when the exponents s_1 and s_2 are equal or when $s_1 - s_2 = KM$, where K is a positive integer. In such a situation, when only one solution of type (54) is obtained, a second solution can be found, as usual, by use of Equations (33) or (35).

Since the expansion of $(1 + R_M x^M)^{-1}$ converges when

$$|x| < |R_M|^{-1/M}, (55)$$

the solutions obtained for (52) also will converge in that interval. In particular, if $R_M = 0$ the series will converge for all finite values of x (the value x = 0 itself being excepted, as usual, when the real part of s is negative).

Among the many important specializations of Equation (52), we note Bessel's equation,

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (x^{2} - p^{2})y = 0,$$
 (56)

for which M = 2; Legendre's equation,

$$(1-x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + p(p+1)y = 0,$$
 (57)

for which M = 2; and Gauss's equation,

$$x(1-x)\frac{d^2y}{dx^2} + \left[\gamma - (\alpha + \beta + 1)x\right]\frac{dy}{dx} - \alpha\beta y = 0, \qquad (58)$$

for which M = 1. The solutions of these equations, in the neighborhood of x = 0, are studied in the following sections.

Other notable special cases may be listed as follows.

(1) The equation

$$x\frac{d^{2}y}{dx^{2}} + (c-x)\frac{dy}{dx} - ay = 0,$$
 (59)

for which M = 1, is satisfied by the confluent hypergeometric function of

Kummer, y = M(a,c,x) (see Problem 8). If c = 1 and a = -n, where n is a positive integer or zero, one solution is the nth Laguerre polynomial, $y = L_n(x)$. If c = m + 1 and a = m - n, where m and n are integral, one solution is the associated Laguerre polynomial,

$$y = L_n^m(x) = \frac{d^m}{dx^m} L_n(x),$$

if $m \leq n$.

(2) The equation

$$\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + 2ny = 0,$$
 (60)

for which M = 2, is satisfied by the *n*th Hermite polynomial, $y = H_n(x)$, when *n* is a positive integer or zero.

(3) The equation

$$(1-x^2)\frac{d^2y}{dx^2} - x\frac{dy}{dx} + n^2y = 0,$$
(61)

for which M = 2, is satisfied by the *n*th Chebyshev polynomial, $y = T_n(x)$, when *n* is a positive integer or zero.

(4) The equation

$$x(1-x)\frac{d^2y}{dx^2} + [a - (1+b)x]\frac{dy}{dx} + n(b+n)y = 0,$$
 (62)

for which M = 1, is satisfied by the *n*th Jacobi polynomial, $y = J_n(a,b,x)$, when *n* is a postive integer or zero.

The functions mentioned are useful in many applications. It is a curious fact that they all satisfy equations which are special cases of (52).

4.8. Bessel functions. Solutions of the differential equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (x^{2} - p^{2})y = 0,$$
 (63)

or, equivalently,

$$x\frac{d}{dx}\left(x\frac{dy}{dx}\right) + (x^2 - p^2)y = 0,$$
 (63a)

are known as *Bessel functions of order p*. These functions are of frequent use in the solution of many types of potential problems involving circular cylindrical boundaries, as well as in other applications, in such fields as elasticity, fluid flow, electrical field theory, and aerodynamic flutter analysis. We suppose that the constant p is *real*. Since only the quantity p^2 appears in Equation (63), we may also consider p to be *nonnegative* without loss of generality.

Since Equation (63) is of type (52), with M = 2, we may seek a solution of the form (54),

$$y(x) = \sum_{k=0}^{\infty} B_k x^{2k+s}.$$
 (64)

Substitution into (63) yields the indices

$$s_1 = p, \qquad s_2 = -p \tag{65}$$

and the recurrence formula

$$(s+2k+p)(s-2k-p)B_k = -B_{k-1}$$
 $(k \ge 1).$ (66)

The exceptional cases may arise only if $s_1 - s_2 = 2p$ is zero or an integral multiple of M = 2, that is, if p is zero or a positive integer. In such cases we can be certain only of one solution of type (64).

In correspondence with the exponent $s_1 = p$, repeated use of Equation (66) gives

$$B_{k}(p) = (-1)^{k} \frac{1}{(2+2p)(4+2p)\cdots(2k+2p)} \frac{B_{0}}{2^{k}k!}$$
$$= (-1)^{k} \frac{1}{(1+p)(2+p)\cdots(k+p)} \frac{B_{0}}{2^{2k}k!} \quad (k \ge 1), \qquad (67)$$

and we see that all coefficients B_k are defined. Thus a series of type (64) is determined, for s = p, in the form

$$y_1(x) = B_0 \left[x^p + \sum_{k=1}^{\infty} \frac{(-1)^k x^{2k+p}}{(1+p)(2+p)\cdots(k+p)2^{2k}k!} \right],$$

or, making use of Equation (52), Chapter 2,

$$y_1(x) = B_0 \Gamma(1+p) \sum_{k=0}^{\infty} \frac{(-1)^k x^{2k+p}}{2^{2k} k! \Gamma(k+p+1)}$$

This result is put into a more compact form if we use the abbreviation $\Gamma(k + p + 1) = (k + p)!$ and write

$$y_{1}(x) = 2^{p} \Gamma(1+p) B_{0} \sum_{k=0}^{\infty} \frac{(-1)^{k} \left(\frac{x}{2}\right)^{2k+p}}{k! (k+p)!} = B_{0} u_{1}(x).$$
(68)

The series multiplying $2^{p} \Gamma(1+p)B_{0}$ in Equation (68) is known as the Bessel function of the first kind, of order p, and is denoted by $J_{p}(x)$,

$$J_{p}(x) = \sum_{k=0}^{\infty} \frac{(-1)^{k} \left(\frac{x}{2}\right)^{2k+p}}{k! (k+p)!}.$$
 (69)

In particular, when p = 0 and 1 we obtain the series in the forms

$$J_0(x) = 1 - \frac{x^2}{2^2} + \frac{x^4}{2^4(2!)^2} - \frac{x^6}{2^6(3!)^2} + \dots,$$
 (70)

$$J_1(x) = \frac{x}{2} - \frac{x^3}{2^3 2!} + \frac{x^5}{2^5 2! 3!} - \frac{x^7}{2^7 3! 4!} + \dots$$
(71)

With $s = s_2 = -p$, Equation (66) yields the result of replacing p by -p in (67),

$$B_{k}(-p) = (-1)^{k} \frac{1}{(1-p)(2-p)\cdots(k-p)} \frac{B_{0}}{2^{2k}k!} \qquad (k \ge 1).$$
(72)

Thus, if p is a positive integer, all coefficients B_k for which $k \ge p$ become infinite, and no Frobenius solution is obtained, in such a case, corresponding to the exponent s = -p. However, if p is not zero or a positive integer, a second solution is obtained by replacing p by -p in the first solution, and hence may be taken in the form

$$J_{-p}(x) = \sum_{k=0}^{\infty} \frac{(-1)^k \left(\frac{x}{2}\right)^{2k-p}}{k! (k-p)!} \,. \tag{73}$$

Thus, if p is not zero or a positive integer, the complete solution of Bessel's equation (63) is a linear combination of the solutions (69) and (73), of the form

$$y = C_1 J_p(x) + C_2 J_{-p}(x).$$
(74)

If p = 0, the two solutions are identical. Moreover, if p is a positive integer, the second solution $J_{-p}(x)$ is not independent of $J_{p}(x)$. This statement is a

consequence of the fact that if p is a positive integer n, the factor $\frac{1}{(k-n)!}$ in (73) is zero when k < n, and hence (73) is then equivalent to

$$J_{-n}(x) = \sum_{k=n}^{\infty} \frac{(-1)^k \left(\frac{x}{2}\right)^{2k-n}}{(k!)(k-n)!}$$

or, replacing the index k by k + n,

$$J_{-n}(x) = \sum_{k=0}^{\infty} \frac{(-1)^{k+n} \left(\frac{x}{2}\right)^{2k+n}}{k! (k+n)!}.$$

Hence, if n is an integer, we obtain

$$J_{-n}(x) = (-1)^n J_n(x).$$
(75)

It should be noticed that although the higher coefficients in the solution $y_2(x)$ would become infinite as $p \to n$ (n = 1, 2, 3, ...) if the coefficient B_0 were held fixed, we have essentially replaced B_0 by a factor proportional to $1/\Gamma(1-p)$ which vanishes as $p \to n$ and so have obtained a solution $J_{-p}(x)$ in which the coefficients which previously became infinite as $p \to n$ now approach finite limits, and the remaining coefficients tend to zero.

To find a second solution complementing $J_p(x)$ in the exceptional cases, recourse may be had to the methods of Section 4.5. We illustrate the procedure in the case of equal exponents, p = 0. In this case we obtain a function $y_s(x)$ by determining $B_k(s)$ from (66) and introducing the result into (64). The required second solution is then

$$y_2(x) = \left[\frac{\partial y_s(x)}{\partial s}\right]_{s=0}$$

where here

$$y_s(x) = \sum_{k=0}^{\infty} B_k(s) x^{2k+s},$$

so that

$$y_2(x) = \left[\sum_{k=0}^{\infty} B_k(0) x^{2k}\right] \log x + \sum_{k=1}^{\infty} B'_k(0) x^{2k}.$$
 (76)

The recurrence formula (66) first gives

$$B_k(s) = (-1)^k \frac{B_0}{[(s+2)(s+4)\cdots(s+2k)]^2}$$

To calculate $B'_k(s)$ it is convenient first to take the logarithm of the two sides of the equation,

$$\log\left[(-1)^{k}\frac{B_{k}(s)}{B_{0}}\right] = -2\log\left[(s+2)(s+4)\cdots(s+2k)\right]$$
$$= -2\sum_{m=1}^{k}\log(s+2m).$$

Differentiation with respect to s then gives

$$\frac{B'_{k}(s)}{B_{k}(s)} = -2 \sum_{m=1}^{k} \frac{1}{s+2m}$$

and hence

$$\frac{B'_k(0)}{B_k(0)} = -2 \sum_{m=1}^k \frac{1}{2m} = -\sum_{m=1}^k \frac{1}{m}.$$

Thus, if we again introduce the abbreviation

$$\varphi(k) = \sum_{m=1}^{k} \frac{1}{m} = 1 + \frac{1}{2} + \dots + \frac{1}{k} \qquad (k \ge 1) \bigg\}, \qquad (77)$$
$$\varphi(0) = 0$$

we obtain

$$B'_{k}(0) = -\varphi(k)B_{k}(0) = -\varphi(k)\left[(-1)^{k} \frac{B_{0}}{[2 \cdot 4 \cdot 6 \cdots (2k)]^{2}}\right]$$
$$= -\frac{(-1)^{k}\varphi(k)}{2^{2k}(k!)^{2}}B_{0},$$

and (76) then gives the required second solution in the case p = 0 in the form

$$y_2(x) = B_0 \left[J_0(x) \log x + \sum_{k=1}^{\infty} (-1)^{k+1} \varphi(k) \frac{\left(\frac{x}{2}\right)^{2k}}{(k!)^2} \right].$$

The coefficient of B_0 is thus a second solution of Bessel's equation (63) when p = 0,

$$x\frac{d^2y}{dx^2} + \frac{dy}{dx} + xy = 0.$$

It was taken as the standard form of the second solution by *Neumann*, and is usually denoted by $Y^{(0)}(x)$. Thus any linear combination of $J_0(x)$ and $Y^{(0)}(x)$ is also a solution. The standard form chosen by *Weber* is defined in terms of J_0 and $Y^{(0)}$ by the equation

$$Y_0(x) = \frac{2}{\pi} \left[Y^{(0)}(x) + (\gamma - \log 2) J_0(x) \right], \tag{78}$$

where γ is *Euler's constant*, defined by the relation

$$\gamma = \lim_{k \to \infty} \left[\varphi(k) - \log k \right]$$

=
$$\lim_{k \to \infty} \left(1 + \frac{1}{2} + \frac{1}{3} + \dots + \frac{1}{k} - \log k \right) = 0.5772157 \dots (79)$$

We thus obtain a second solution in the form

$$Y_{0}(x) = \frac{2}{\pi} \left[\left(\log \frac{x}{2} + \gamma \right) J_{0}(x) + \sum_{k=0}^{\infty} (-1)^{k+1} \varphi(k) \left(\frac{\left(\frac{x}{2} \right)^{2k}}{(k!)^{2}} \right) \right]$$

$$= \frac{2}{\pi} \left[\left(\log \frac{x}{2} + \gamma \right) J_{0}(x) + \left\{ \frac{x^{2}}{2^{2}} - \frac{x^{4}}{2^{4} (2!)^{2}} \left(1 + \frac{1}{2} \right) + \frac{x^{6}}{2^{6} (3!)^{2}} \left(1 + \frac{1}{2} + \frac{1}{3} \right) - \cdots \right\} \right].$$
(80)

The function $Y_0(x)$ is known as Weber's Bessel function of the second kind, of order zero. In German texts it is frequently denoted by $N_0(x)$. The complete solution can thus be written in the form

$$y = C_1 J_0(x) + C_2 Y_0(x).$$
(81)

Weber's definition of the function of the second kind [Equation (78)] is generally more convenient than that of Neumann, and is usually preferred, because of the fact that the behavior of the function $Y_0(x)$ so defined, for large values of x, is more nearly comparable with the behavior of $J_0(x)$ [see Equation (88)].

A similar but more involved calculation leads to expressions for Weber's form of the Bessel functions of the second kind, of order n,

$$Y_{n}(x) = \frac{2}{\pi} \left[\left(\log \frac{x}{2} + \gamma \right) J_{n}(x) - \frac{1}{2} \sum_{k=0}^{n-1} \frac{(n-k-1)! \left(\frac{x}{2}\right)^{2k-n}}{k!} + \frac{1}{2} \sum_{k=0}^{\infty} (-1)^{k+1} [\varphi(k) + \varphi(k+n)] \frac{\left(\frac{x}{2}\right)^{2k+n}}{k! (n+k)!} \right]$$
(82)

when n is a positive integer. Thus, in particular,

$$Y_{1}(x) = \frac{2}{\pi} \left[\left(\log \frac{x}{2} + \gamma \right) J_{1}(x) - \frac{1}{x} - \frac{x}{4} + \left\{ \frac{1 + (1 + \frac{1}{2})}{2} \right\} \frac{x^{3}}{2^{3} 2!} - \left\{ \frac{(1 + \frac{1}{2}) + (1 + \frac{1}{2} + \frac{1}{3})}{2} \right\} \frac{x^{5}}{2^{5} 2! \ 3!} + \cdots \right].$$
(83)

It follows that if p = n, where n is zero or a positive integer, the general solution of (63) can be taken in the form

$$y = C_1 J_n(x) + C_2 Y_n(x).$$
 (84)

If p is not zero or a positive integer, the function $Y_p(x)$ is defined by the equation

$$Y_{p}(x) = \frac{(\cos p\pi) J_{p}(x) - J_{-p}(x)}{\sin p\pi}.$$
 (85)

This definition can be shown to be consistent with Equation (82) as $p \rightarrow n$, and it defines $Y_p(x)$ as a linear combination of $J_p(x)$ and $J_{-p}(x)$ otherwise. It should be emphasized, however, that the second solution $Y_p(x)$ is not needed unless p = n.

The general solution of Bessel's equation is frequently abbreviated by use of the notation

$$y = Z_p(x), \tag{86}$$

with the convention that (86) stands for (74) unless p is zero or a positive integer, and for (84) in these cases.

The transformation $y = \frac{u(x)}{\sqrt{x}}$ transforms (63) to the equation

$$\frac{d^2u}{dx^2} + \left(1 - \frac{p^2 - \frac{1}{4}}{x^2}\right)u = 0$$
(87)

(see Problem 38, Chapter 1). For large values of x the term $\frac{p^2 - \frac{1}{4}}{x^2}$ is negligible in comparison with unity. Thus it may be expected that for large values of x the behavior of solutions of (87) will be similar to that of corresponding solutions of the equation $\frac{d^2u}{dx^2} + u = 0$. Since such solutions can be written in the form $u = A \cos(x - \varphi)$, where A and φ are constants, we are led to the possibility that for large values of x any solution of (63) behaves like the function

$$\frac{u(x)}{\sqrt{x}} = \frac{A}{\sqrt{x}}\cos(x-\varphi),$$

for properly chosen values of A and φ . A rather involved analysis shows that for the function $J_p(x)$ one has

$$A_1 = \sqrt{\frac{2}{\pi}}$$
 and $\varphi_1 = (2p+1)\frac{\pi}{4}$,

whereas for $Y_{p}(x)$ there follows

$$A_2 = \sqrt{\frac{2}{\pi}}$$
 and $\varphi_2 = \frac{\pi}{2} + \varphi_1$.

Thus we may write

$$J_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \cos (x - \alpha_{p})$$

$$Y_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \sin (x - \alpha_{p})$$

$$(x \to \infty)$$
(88)

where

$$\alpha_p = (2p+1)\frac{\pi}{4}.$$
 (89)

The notation of (88) denotes that the ratio of two expressions connected by the symbol \sim approaches unity as $x \rightarrow \infty$. We say that $J_p(x)$ behaves asymptotically like $\sqrt{\frac{2}{\pi x}} \cos(x - \alpha_p)$.

It follows from (88) that the complex function $J_p(x) + i Y_p(x)$ has the asymptotic behavior

$$J_{p}(x) + i Y_{p}(x) \sim \sqrt{\frac{2}{\pi x}} e^{i(x-\alpha_{p})} \qquad (x \to \infty), \qquad (90)$$

whereas the conjugate complex function has the behavior

$$J_{p}(x) - i Y_{p}(x) \sim \sqrt{\frac{2}{\pi x}} e^{-i(x-\alpha_{p})} \qquad (x \to \infty).$$
(91)

These complex functions are known as *Bessel functions of the third kind*, or, more generally, as the *Hankel functions of the first and second kinds*, respectively, and the notations

$$H_{p}^{(1)}(x) = J_{p}(x) + i Y_{p}(x)$$

$$H_{p}^{(2)}(x) = J_{p}(x) - i Y_{p}(x)$$
(92)

are conventional. These functions are particularly useful in studying certain types of wave propagation (see Section 9.13).

The differential equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} - (p^{2} + x^{2}) y = 0, \qquad (93)$$

which differs from Bessel's equation (63) only in the sign of x^2 in the coefficient of y, is transformed by the substitution ix = t to the equation

$$t^{2}\frac{d^{2}y}{dt^{2}}+t\frac{dy}{dt}+(t^{2}-p^{2}) y=0,$$

which is in the form of Bessel's equation (63). Hence, the general solution is of the form $y = Z_p(t)$, or, in terms of the original variable x, the general solution of (93) is

$$y = Z_{p}(ix). \tag{94}$$

That is, if p is not zero or a positive integer, the general solution is of the form

$$y = c_1 J_p(ix) + c_2 J_{-p}(ix),$$

whereas otherwise it may be taken in the form

$$y = c_1 J_n(ix) + c_2 Y_n(ix).$$

From Equation (69) we have

$$J_{p}(ix) = \sum_{k=0}^{\infty} \frac{(-1)^{k} i^{2k+p} \left(\frac{x}{2}\right)^{2k+p}}{k! (k+p)!} = i^{p} \sum_{k=0}^{\infty} \frac{\left(\frac{x}{2}\right)^{2k+p}}{k! (k+p)!}.$$

In place of using this function as a fundamental solution of (93), it is preferable to use the function $I_p(x) = i^{-p} J_p(ix)$,

$$I_{p}(x) = \sum_{k=0}^{\infty} \frac{\left(\frac{x}{2}\right)^{2k+p}}{k! (k+p)!},$$
(95)

since this function is *real* for real values of p. This function is known as the *modified Bessel function of the first kind, of order p*. The terms in the series representing $I_p(x)$ differ from those in the series for $J_p(x)$ only in that the terms are all positive in the I_p series, whereas they alternate in sign in the J_p series. Thus, if p is not zero or a positive integer, the general solution of (93) can be taken in the real form

$$y = Z_{p}(ix) = c_{1} I_{p}(x) + c_{2} I_{-p}(x).$$
(96)

As a second real fundamental solution of (93), in the case when p = n, where *n* is zero or a positive integer, it is conventional to define the function $K_n(x)$ by the equation

$$K_n(x) = \frac{\pi}{2} i^{n+1} [J_n(ix) + i Y_n(ix)] = \frac{\pi}{2} i^{n+1} H_n^{(1)}(ix), \qquad (97)$$

leading to the general solution

$$y = Z_n(ix) = c_1 I_n(x) + c_2 K_n(x),$$
(98)

when n is zero or a positive integer. The function $K_n(x)$ is known as the modified Bessel function of the second kind, of order n.

If p is not zero or a positive integer, the function $K_p(x)$ is defined by the equation

$$K_{p}(x) = \frac{\pi}{2} \frac{I_{-p}(x) - I_{p}(x)}{\sin p\pi},$$
(99)

which is consistent with (97) when $p \rightarrow n$.

For large values of x the modified functions have the asymptotic behavior

$$I_{p}(x) \sim \frac{e^{x}}{\sqrt{2\pi x}}$$

$$K_{p}(x) \sim \frac{e^{-x}}{\sqrt{\frac{2}{\pi}x}}$$

$$(x \to \infty). \qquad (100)$$

It is important to notice that the right members of (100) are independent of p.

4.9. Properties of Bessel functions. It is readily verified directly that all power series involved in the definitions of all Bessel functions converge for all

finite values of x. However, in consequence of the fact that these series are in many cases multiplied by a negative power of x or by a logarithmic term, it is found that only the functions $J_{p}(x)$ and $I_{p}(x)$ are finite at x = 0 (when $p \ge 0$).

For small values of x, retention of the leading terms in the respective series leads to the approximations

$$J_{p}(x) \sim \frac{1}{2^{p}p!} x^{p}, \qquad J_{-p}(x) \sim \frac{2^{p}}{(-p)!} x^{-p},$$
 (101)

$$Y_p(x) \sim -\frac{2^p(p-1)!}{\pi} x^{-p} \quad (p \neq 0), \qquad Y_0(x) \sim -\frac{2}{\pi} \log x, \qquad (102)$$

$$I_{p}(x) \sim \frac{1}{2^{p}p!} x^{p}, \qquad I_{-p}(x) \sim \frac{2^{p}}{(-p)!} x^{-p},$$
 (103)

$$K_p(x) \sim 2^{p-1}(p-1)! x^{-p} \quad (p \neq 0), \qquad K_0(x) \sim -\log x, \qquad (104)$$

in the sense that the ratio of two quantities connected by the symbol \sim approaches unity as $x \rightarrow 0$.

For large values of $x (x \to \infty)$, we recapitulate the results listed in the preceding section:

$$J_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \cos\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right),$$

 $Y_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \sin\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right),$ (105)

$$H_p^{(1)}(x) \sim \sqrt{\frac{2}{\pi x}} e^{i\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right)}, \qquad H_p^{(2)}(x) \sim \sqrt{\frac{2}{\pi x}} e^{-i\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right)}, \quad (106)$$

$$I_{p}(x) \sim \frac{e^{x}}{\sqrt{2\pi x}}, \qquad K_{p}(x) \sim \sqrt{\frac{\pi}{2x}} e^{-x}.$$
 (107)

The following derivative formulas are of frequent use:

$$\frac{d}{dx} [x^{p} y_{p}(\alpha x)] = \begin{cases} \alpha x^{p} y_{p-1}(\alpha x) & (y = J, Y, I, H^{(1)}, H^{(2)}) \\ -\alpha x^{p} y_{p-1}(\alpha x) & (y = K); \end{cases}$$
(108)

$$\frac{d}{dx} \left[x^{-p} y_{p}(\alpha x) \right] = \begin{cases} -\alpha x^{-p} y_{p+1}(\alpha x) & (y = J, Y, K, H^{(1)}, H^{(2)}) \\ \alpha x^{-p} y_{p+1}(\alpha x) & (y = I). \end{cases}$$
(109)

These formulas are established for J_p and Y_p by considering their series definitions, and for the remaining functions by considering their definitions in

terms of J_p and Y_p . Thus, to prove (108) for J_p , we note that from the definition (69) we have

$$\frac{d}{dx} \left[x^p J_p(\alpha x) \right] = \frac{d}{dx} \sum_{k=0}^{\infty} \frac{(-1)^k \alpha^{2k+p} x^{2k+2p}}{2^{2k+p} k! (k+p)!}$$
$$= \sum_{k=0}^{\infty} \frac{(-1)^k \alpha^{2k+p} x^{2k+2p-1}}{2^{2k+p-1} k! (k+p-1)!}$$
$$= \alpha x^p \sum_{k=0}^{\infty} \frac{(-1)^k \left(\frac{\alpha x}{2}\right)^{2k+p-1}}{k! (k+p-1)!}$$
$$= \alpha x^p J_{p-1}(\alpha x).$$

From (108) there follows

$$\frac{d}{dx}y_{p}(\alpha x) = \begin{cases} \alpha y_{p-1}(\alpha x) - \frac{p}{x}y_{p}(\alpha x) & (y = J, Y, I, H^{(1)}, H^{(2)}) \\ -\alpha y_{p-1}(\alpha x) - \frac{p}{x}y_{p}(\alpha x) & (y = K), \end{cases}$$
(110)

whereas (109) becomes

$$\frac{d}{dx} y_{p}(\alpha x) = \begin{cases} -\alpha y_{p+1}(\alpha x) + \frac{p}{x} y_{p}(\alpha x) & (y = J, Y, K, H^{(1)}, H^{(2)}) \\ \alpha y_{p+1}(\alpha x) + \frac{p}{x} y_{p}(\alpha x) & (y = I). \end{cases}$$
(111)

By addition and subtraction of (110) and (111) we also obtain the relations

$$2\frac{d}{dx}y_{p}(\alpha x) = \alpha [y_{p-1}(\alpha x) - y_{p+1}(\alpha x)] \\ y_{p-1}(\alpha x) + y_{p+1}(\alpha x) = \frac{2p}{\alpha x}y_{p}(\alpha x)$$
 $(y = J, Y, H^{(1)}, H^{(2)}), (112, 113)$

$$2\frac{d}{dx}I_{p}(\alpha x) = \alpha [I_{p-1}(\alpha x) + I_{p+1}(\alpha x)], \qquad (112a)$$

$$2\frac{d}{dx}K_{p}(\alpha x) = -\alpha [K_{p-1}(\alpha x) + K_{p+1}(\alpha x)], \qquad (112b)$$

$$I_{p-1}(\alpha x) - I_{p+1}(\alpha x) = \frac{2p}{\alpha x} I_p(\alpha x), \qquad (113a)$$

$$K_{p-1}(\alpha x) - K_{p+1}(\alpha x) = -\frac{2p}{\alpha x} K_p(\alpha x).$$
(113b)

Relations (108-112) are useful in evaluating certain integrals and derivatives involving Bessel functions. In particular, setting p = 0 in (109), we obtain the relations

$$\frac{d}{dx}y_0(\alpha x) = \begin{cases} -\alpha \ y_1(\alpha x) & (y = J, Y, K, H^{(1)}, H^{(2)}) \\ \alpha \ y_1(\alpha x) & (y = I). \end{cases}$$
(114)

The relations (113) are useful in expressing Bessel functions of order p + 1 in terms of two corresponding functions of lower order p and p - 1.

The Bessel functions of order p, where p is half an odd integer, can be expressed in closed form, in terms of elementary functions. To establish this fact, we consider first the case $p = \frac{1}{2}$ and denote the general solution of (63) by $y = Z_{1/2}(x)$. If we write

$$u(x) = \sqrt{x} Z_{1/2}(x),$$

Equation (87) shows that the function u(x) satisfies the equation

$$\frac{d^2u}{dx^2}+u=0$$

and hence is of the form $u = A \cos x + B \sin x$. Taking

$$u=\sqrt{x}\,J_{1/2}(x),$$

we have also, from (108),

$$\frac{du}{dx} = \sqrt{x} J_{-1/2}(x).$$

Hence, using (101), we find that

$$u(0) = 0$$
 and $u'(0) = \frac{2^{1/2}}{(-\frac{1}{2})!} = \frac{\sqrt{2}}{\Gamma(\frac{1}{2})} = \sqrt{\frac{2}{\pi}}$

Thus we must have A = 0, $B = \sqrt{\frac{2}{\pi}}$, and there follows

$$u = \sqrt{\frac{2}{\pi}} \sin x, \qquad \frac{du}{dx} = \sqrt{\frac{2}{\pi}} \cos x,$$

or, finally,

$$J_{1/2}(x) = \sqrt{\frac{2}{\pi x}} \sin x, \qquad J_{-1/2}(x) = \sqrt{\frac{2}{\pi x}} \cos x.$$
 (115)

Also, since $I_{1/2}(x) = i^{-1/2} J_{1/2}(ix)$ and $I_{-1/2}(x) = i^{1/2} J_{-1/2}(ix)$, we have

$$I_{1/2}(x) = \sqrt{\frac{2}{\pi x}} \sinh x, \quad I_{-1/2}(x) = \sqrt{\frac{2}{\pi x}} \cosh x.$$
 (116)

From (113), with $\alpha = 1$ and $p = n - \frac{1}{2}$, we obtain the recurrence formulas

$$J_{n+1/2}(x) = \frac{2n-1}{x} J_{n-1/2}(x) - J_{n-3/2}(x)$$
(117)

and

$$I_{n+1/2}(x) = -\frac{2n-1}{x} I_{n-1/2}(x) + I_{n-3/2}(x), \qquad (118)$$

which permit the determination of $J_{n+1/2}(x)$ and $I_{n+1/2}(x)$ for all integral values of *n*, in terms of the functions in (115) and (116). Certain of those functions are listed in Table 2 of Chapter 2 (page 77).



Figure 4.1

As is indicated by the asymptotic approximations (105), the functions $J_p(x)$ and $Y_p(x)$ are oscillatory in nature, the amplitude of oscillation about a zero value tending to decrease with $\sqrt{2/\pi x}$ and the distance between successive zeros of the function decreasing toward π .* It can be shown that the zeros of $J_{p+1}(x)$ separate the zeros of $J_p(x)$; that is, that between any two consecutive zeros of $J_{p+1}(x)$ there is one and only one zero of $J_p(x)$. The same applies to the zeros of the functions of the second kind. The functions I_p and K_p are not

* A brief table of values of zeros of certain Bessel functions is presented in the Appendix.

oscillatory. The former function essentially increases exponentially with x, whereas the second essentially decreases exponentially. Sketches of these functions are presented in Figure 4.1.

4.10. Differential equations satisfied by Bessel functions. The solution of the differential equation

$$X^{2}\frac{d^{2}Y}{dX^{2}} + X\frac{dY}{dX} + (X^{2} - p^{2})Y = 0$$
 (119)

can be written in the form

$$Y = Z_{p}(X). \tag{120}$$

If we make the substitutions

$$Y = \frac{y}{g(x)}, \qquad X = f(x) \tag{121}$$

and notice that then

$$\frac{d}{dX} = \frac{dx}{dX}\frac{d}{dx} = \frac{dx}{df}\frac{d}{dx} = \frac{1}{f'(x)}\frac{d}{dx},$$
(122)

Equation (119) becomes

$$f^{2}\left\{\frac{1}{f'}\frac{d}{dx}\left[\frac{1}{f'}\frac{d}{dx}\left(\frac{y}{g}\right)\right]\right\} + f\left[\frac{1}{f'}\frac{d}{dx}\left(\frac{y}{g}\right)\right] + (f^{2} - p^{2})\frac{y}{g} = 0,$$

$$f\frac{d}{dx}\left[\frac{\frac{d}{dx}\left(\frac{y}{g}\right)}{f'}\right] + \frac{d}{dx}\left(\frac{y}{g}\right) + \frac{f'}{f}(f^{2} - p^{2})\frac{y}{g} = 0.$$
(123)

or

Reference to (120) and (121) shows that (123) is the differential equation satisfied by

$$y = g(x) Z_p[f(x)].$$
 (124)

In particular, if we set

$$g(x) = x^{A}e^{-Bx^{r}}, \quad f(x) = Cx^{s},$$
 (125)

Equation (123) can be reduced to the form

$$x^{2} \frac{d^{2}y}{dx^{2}} + x[(1-2A) + 2rBx^{r}] \frac{dy}{dx} + [A^{2} - p^{2}s^{2} + s^{2}C^{2}x^{2s} - rB(2A - r)x^{r} + r^{2}B^{2}x^{2r}] y = 0.$$

This equation is somewhat simplified if we write

$$1-2A = a$$
, $rB = b$, $A^2 - p^2 s^2 = c$, $s^2 C^2 = d$,

from which there follows

$$A = \frac{1-a}{2}, \quad B = \frac{b}{r}, \quad C = \frac{\sqrt{d}}{s}, \quad p = \frac{1}{s}\sqrt{A^2 - c}.$$
 (126)

With this notation, the differential equation takes the form

$$x^{2} \frac{d^{2}y}{dx^{2}} + x(a + 2bx^{r}) \frac{dy}{dx} + [c + dx^{2s} - b(1 - a - r)x^{r} + b^{2}x^{2r}] y = 0, \quad (127)$$

and the solution of (127), obtained by introducing (125) and (126) into (124), is of the form

$$y = x^{\frac{1-a}{2}} e^{-\frac{bx^r}{r}} Z_p\left(\frac{\sqrt{d}}{s} x^s\right), \qquad (128)$$

where

$$p = \frac{1}{s} \sqrt{\left(\frac{1-a}{2}\right)^2 - c}.$$
 (129)

If $\frac{\sqrt{d}}{s}$ is real, Z_p is to be interpreted by (74) or (84), whereas if $\frac{\sqrt{d}}{s}$ is imaginary, Z_p stands for (96) or (98), the choice in either case depending upon whether p is not or is zero or a positive integer, respectively.*

Thus, if it is possible to identify a particular differential equation with (127), by suitably choosing the constants in (127), the solution is given immediately by (128) in terms of Bessel functions of order p, where p is given by (129). We list here certain useful special forms with corresponding solutions readily obtainable in this way.

$$x^{2}y'' + xy' + (k^{2}x^{2} - p^{2})y = 0, \quad y = Z_{p}(kx) \quad (k \neq 0).$$
 (130a)

$$x^{2}y'' + xy' - (k^{2}x^{2} + p^{2})y = 0, \quad y = Z_{p}(ikx) \quad (k \neq 0).$$
 (130b)

$$x^{2}y'' + xy' - (\beta^{2} - \alpha x^{2s}) y = 0, \quad y = Z_{\beta/s}\left(\frac{\sqrt{\alpha}}{s}x^{s}\right) \quad (\alpha s \neq 0).$$
 (130c)

$$y'' + kx^{m}y = 0, \quad y = \sqrt{x} Z_{\frac{1}{m+2}} \left(\frac{2\sqrt{k}}{m+2} x^{\frac{m+2}{2}} \right) \quad (mk \neq -2k). \quad (130d)$$
$$\frac{d}{dx} \left(x^{n} \frac{dy}{dx} \right) + k x^{m}y = 0, \quad y = x^{\frac{1-n}{2}} Z_{p} \left(\frac{\sqrt{k}}{s} x^{s} \right)$$
$$m = n \qquad 1 = n$$

where

$$s = \frac{m-n}{2} + 1, \quad p = \frac{1-n}{2s} \quad (ks \neq 0).$$
 (130e)

The restricted cases in (130a-e) are all readily solvable as equidimensional equations (see Section 1.6).

4.11. Ber and bei functions. In certain problems it is convenient to obtain a desired solution as the real or imaginary part of a complex function, in terms

^{*} Because of the ambiguity of the signs of the radicals in (128) and (129), both the order p and the coefficient of x^* can be replaced by their negatives, and hence can be taken to be *nonnegative* when they are real, in (128) and in its specializations (130a-e).

of which a simplified formulation of the problem is possible. In such cases differential equations with complex coefficients may occur. In particular, equations are rather frequently obtained which are reducible to the form

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} - (p^{2} + ix^{2})y = 0, \qquad (131)$$

where p is a real constant. Since (131) is equivalent to (130b) if $k^2 = i$ or $k = i^{1/2}$, the solution of (131) can be expressed in the form

$$y = Z_p(t^{3/2}x).$$
 (132)

The solutions of (131) which are finite at x = 0 are thus multiples of the function $J_p(i^{3/2}x)$. This function is a complex function whose real and imaginary parts are written ber_p x and bei_p x, respectively,

ber
$$_{p} x + i \text{ bei}_{p} x = J_{p}(i^{3/2}x) = i^{p}I_{p}(i^{1/2}x).$$
 (133)

The polar notation

$$\operatorname{ber}_{p} x + i \operatorname{bei}_{p} x = M_{p}(x) e^{i\theta_{p}(x)}$$
(134)

is frequently used.

We consider, in illustration, the case p = 0. In this case the zero subscripts are conventionally omitted in the notation for the real and imaginary parts,

$$J_0(i^{3/2}x) = ber x + i bei x.$$
 (135)

From the definition (69) we obtain, with p = 0,

$$J_0(i^{3/2}x) = \sum_{m=0}^{\infty} \frac{(-1)^m i^{3m} \left(\frac{x}{2}\right)^{2m}}{(m!)^2} \,.$$

We now separate this series into two parts, in the first of which m takes on even values and in the second of which only odd values of m occur. If we replace m by 2k in the first series and m by 2k + 1 in the second, we obtain

$$J_0(i^{3/2}x) = \sum_{k=0}^{\infty} \frac{i^{6k} \left(\frac{x}{2}\right)^{4k}}{[(2k)!]^2} - \sum_{k=0}^{\infty} \frac{i^{6k+3} \left(\frac{x}{2}\right)^{4k+2}}{[(2k+1)!]^2}$$

Noticing that

$$i^{6k} = (-1)^{3k} = (-1)^k, \quad i^{6k+3} = (-1)^k i^3 = (-1)^{k+1} i,$$

we obtain finally

$$J_0(i^{3/2}x) = \sum_{k=0}^{\infty} (-1)^k \frac{\left(\frac{x}{2}\right)^{4k}}{\left[(2k)!\right]^2} + i \sum_{k=0}^{\infty} (-1)^k \frac{\left(\frac{x}{2}\right)^{4k+2}}{\left[(2k+1)!\right]^2}.$$
 (136)

A comparison of (135) and (136) gives the results

ber
$$x = \sum_{k=0}^{\infty} (-1)^k \frac{\left(\frac{x}{2}\right)^{4k}}{\left[(2k)!\right]^2}$$

= $1 - \frac{x^4}{2^2 4^2} + \frac{x^8}{2^2 4^2 6^2 8^2} - \cdots$ (137)

and

bei
$$x = \sum_{k=0}^{\infty} (-1)^k \frac{\left(\frac{x}{2}\right)}{\left[(2k+1)!\right]^2}$$

= $\frac{x^2}{2^2} - \frac{x^6}{2^2 4^2 6^2} + \frac{x^{10}}{2^2 4^2 6^2 8^2 10^2} - \cdots$ (138)

Analogous series expansions defining the functions $ber_p x$ and $bei_p x$ for general values of p are obtained by similar methods.

Similar functions of the second kind, which are not finite at x = 0, are defined by the relation

$$\ker_{p} x + i \ker_{p} x = i^{-p} K_{p}(i^{1/2}x).$$
(139)

The general solution of (131) can then be written in the form

$$y = [c_1 \operatorname{ber}_p x + c_2 \operatorname{ker}_p x] + i[c_1 \operatorname{bei}_p x + c_2 \operatorname{kei}_p x].$$
(140)

To illustrate the occurrence of Bessel functions of order zero in practice, we consider particular solutions of the *partial differential equations*

$$\frac{\partial^2 U}{\partial x^2} + \frac{1}{x} \frac{\partial U}{\partial x} = \lambda \frac{\partial^2 U}{\partial t^2}$$
(141)

and

$$\frac{\partial^2 U}{\partial x^2} + \frac{1}{x} \frac{\partial U}{\partial x} = \mu \frac{\partial U}{\partial t}.$$
 (142)

It is found that in many physical problems in various fields, a physical property U depending on a single distance variable x measured from a reference axis of symmetry, and on a time variable t, must satisfy one or the other of these equations, the quantities λ and μ involving known physical constants independent of time and position. It is frequently important to determine solutions of the form

 $f(x)\sin\omega t + g(x)\cos\omega t$,

where ω is a constant. However, instead of proceeding directly to such a determination, it is more convenient to consider the required solution as *the* real or imaginary part of a complex solution of the form

$$U = y(x) e^{i\omega t}, \qquad (143)$$

where y(x) is a complex function of the form y(x) = F(x) + i G(x).

Substitution of (143) into (141) and subsequent cancellation of the factor $e^{i\omega t}$ show that the function y(x) must satisfy the differential equation

$$x\frac{d^2y}{dx^2} + \frac{dy}{dx} + \lambda\omega^2 xy = 0.$$
(144)

If $\lambda > 0$, the general solution of (144) is given by (130a) in the form $y \equiv Z_0(\sqrt{\lambda} \omega x)$, whereas if $\lambda < 0$, the general solution is of the form $y = Z_0(i\sqrt{-\lambda} \omega x)$. In particular, if physical considerations require the solution to be finite when x = 0, the solutions must be of the form $C J_0(\sqrt{\lambda} \omega x)$ when $\lambda > 0$ and $C I_0(\sqrt{-\lambda} \omega x)$ when $\lambda < 0$, where in either case C is a constant (which may be complex).

In a similar way, substitution of (143) into (142) gives the equation

$$x\frac{d^2y}{dx^2} + \frac{dy}{dx} - i\,\mu\omega xy = 0 \tag{145}$$

to be satisfied by the complex amplitude function y(x). If $\mu > 0$, comparison of (145) and (131) shows that the most general solution of (145) which is finite when x = 0 is of the form

$$y = C J_0(i^{3/2}\sqrt{\mu\omega} x) = C(\text{ber }\sqrt{\mu\omega} x + i \text{ bei }\sqrt{\mu\omega} x). \quad (146)$$

If $\mu < 0$, the corresponding solution is of the form

$$y = C J_0(i^{1/2}\sqrt{-\mu\omega} x)$$

or, equivalently,

$$y = C(\operatorname{ber} \sqrt{-\mu\omega} x - i \operatorname{bei} \sqrt{-\mu\omega} x), \qquad (147)$$

as may be seen by taking the complex conjugate of the equal members of (145).

Useful tables of Bessel functions are included in *Tables of Functions*, compiled by Jahnke, Emde, and Losch (Reference 7), the notation $N_p(x)$ being used in place of $Y_p(x)$. In addition, the real and imaginary parts of the functions $J_0(i^{1/2}x)$, $H_0^{(1)}(i^{1/2}x)$, and $i^{1/2}J_1(i^{1/2}x)$, $i^{1/2}H_1^{(1)}(i^{1/2}x)$ are tabulated therein. The first two functions are independent solutions of the equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} - (p^{2} - ix^{2})y = 0$$
 (148)

when p = 0, and the last two functions are solutions when p = 1. Supplementary material, including tables of zeros of $J_p(x)$, is also included. Brief tables of the functions appearing in (140) are included in Dwight's *Tables* (Reference 3).

4.12. Legendre functions. Solutions of the differential equation

$$(1-x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + p(p+1)y = 0$$
(149)

or, equivalently,

$$\frac{d}{dx}\left[(1-x^2)\frac{dy}{dx}\right] + p(p+1)y = 0$$
(149a)

are known as Legendre functions of order p, where p is assumed here to be real and nonnegative.* They are of particular use in the solution of potential problems involving spherical boundaries.

We may notice that x = 0 is an ordinary point for (149), so that one could assume $y = \sum A_k x^k$, with the knowledge that both A_0 and A_1 will be arbitrary. However, since (149) also is of type (52), with M = 2, it is somewhat more convenient to use the method of Section 4.7. Thus the introduction of

$$y(x) = \sum_{k=0}^{\infty} B_k x^{2k+s}$$
 (150)

into Equation (149) readily yields the exponents

$$s_1 = 1, \quad s_2 = 0$$
 (151)

(as was anticipated) and the recurrence formula

$$(s+2k)(s+2k-1)B_{k} = -(p-s-2k+2)(p+s+2k-1)B_{k-1}.$$
 (152)

With s = 0, this formula yields the result

$$B_{k}(0) = (-1)^{k} \frac{\left[p(p-2)(p-4)\cdots(p-2k+2)\right]}{\left[1\cdot 3\cdot 5\cdots(2k-1)\right]} \times \frac{\left[(p+1)(p+3)(p+5)\cdots(p+2k-1)\right]}{\left[2\cdot 4\cdot 6\cdots(2k)\right]} B_{0},$$

or

$$B_{k}(0) = \frac{(-1)^{k}B_{0}}{(2k)!} \left[p(p-2)\cdots(p-2k+2) \right] \times \left[(p+1)(p+3)\cdots(p+2k-1) \right].$$
(153)

Similarly, when s = 1, Equation (152) gives

$$B_{k}(1) = \frac{(-1)^{k}B_{0}}{(2k+1)!} \left[(p-1)(p-3)\cdots(p-2k+1) \right] \times \left[(p+2)(p+4)\cdots(p+2k) \right].$$
(154)

The solutions corresponding to the exponents s = 0 and s = 1 are then of the respective forms

$$B_0 + \sum_{k=1}^{\infty} B_k(0) x^{2k}, \qquad B_0 x + \sum_{k=1}^{\infty} B_k(1) x^{2k+1}.$$

* Since p(p + 1) is unchanged when p is replaced by -(p + 1), the solutions for $p = -p_0$ are the same as those for $p = p_0 - 1$.

The coefficients of B_0 in these expressions are here denoted by $u_p(x)$ and $v_p(x)$, respectively, so that we write

$$u_{p}(x) = 1 - \frac{p(p+1)}{2!} x^{2} + \frac{p(p-2)(p+1)(p+3)}{4!} x^{4} - \frac{p(p-2)(p-4)(p+1)(p+3)(p+5)}{6!} x^{6} + \cdots$$
(155)

and

$$v_{p}(x) = x - \frac{(p-1)(p+2)}{3!} x^{3} + \frac{(p-1)(p-3)(p+2)(p+4)}{5!} x^{5} - \frac{(p-1)(p-3)(p-5)(p+2)(p+4)(p+6)}{7!} x^{7} + \cdots$$
 (156)

It may be seen that if p is an even positive integer n (or zero) the series (155) terminates with the term involving x^n , and hence is a polynomial of degree n. Similarly, if p is an odd positive integer n, the series (156) terminates with the term involving x^n . Otherwise, the expressions are infinite series. The results of Section 4.4 show that the series converge when -1 < x < 1; they diverge otherwise (unless they terminate), as can be verified directly.

Thus the general solution of Equation (149) could be expressed in the form

$$y = c_1 u_p(x) + c_2 v_p(x)$$

when -1 < x < 1. However, a different terminology is conventional, for reasons which are now to be explained.

We consider first the cases when p = n, where *n* is a positive integer or zero. These are the cases commonly arising in practice. When p = n, one of the solutions (155) and (156) is a *polynomial* of degree *n*, whereas the other is an infinite series. That multiple of the polynomial of degree *n* which has the value unity when x = 1 is called the *n*th Legendre polynomial and is denoted by $P_n(x)$.

Thus we have, when n is even,

$$P_n(x) = \frac{u_n(x)}{u_n(1)}$$
 (157a)

and, when n is odd,

$$P_n(x) = \frac{v_n(x)}{v_n(1)}$$
. (157b)

The first six Legendre polynomials are readily found to be

$$P_{0}(x) = 1, \quad P_{1}(x) = x, \quad P_{2}(x) = \frac{1}{2}(3x^{2} - 1),$$

$$P_{3}(x) = \frac{1}{2}(5x^{2} - 3x), \quad P_{4}(x) = \frac{1}{8}(35x^{4} - 30x^{2} + 3),$$

$$P_{5}(x) = \frac{1}{8}(63x^{5} - 70x^{2} + 15x). \quad (158)$$

For later reference, it is noted that the functions $u_n(x)$, with *n* even, and $v_n(x)$, with *n* odd, can be shown to have the following values at x = 1:

$$u_{0}(1) = 1, \qquad u_{n}(1) = (-1)^{\frac{n}{2}} \frac{2 \cdot 4 \cdot 6 \cdots n}{1 \cdot 3 \cdot 5 \cdots (n-1)}$$

$$(n = 2, 4, 6, \ldots), \qquad (159a)$$

$$v_1(1) = 1, \quad v_n(1) = (-1)^{\frac{n}{2}} \frac{2 \cdot 4 \cdot 6 \cdot 1 \cdot (n-1)}{1 \cdot 3 \cdot 5 \cdot 1 \cdot n}$$

(n = 3, 5, 7, ...). (159b)

When p is an even integer n, the solution $v_n(x)$ is in the form of an infinite series, whereas if p is an odd integer n, the solution $u_n(x)$ is an infinite series. Suitable multiples of these solutions are called *Legendre functions of the second* kind and are denoted by $Q_n(x)$. It is conventional to take the multiplicative factors as $(-1)^n u_n(1)$ and $(-1)^n v_n(1)$, respectively, leading to the definition

$$Q_n(x) = \begin{cases} -v_n(1) u_n(x) & (n \text{ odd}) \\ u_n(1) v_n(x) & (n \text{ even}), \end{cases}$$
(160)

where the constants $u_n(1)$, *n* even, and $v_n(1)$, *n* odd, are defined by (159a,b). However, since the series appearing in (160) converge only when |x| < 1, the functions $Q_n(x)$ are defined by (160) only inside this interval.

Thus, when p is an integer n, a certain multiple of that solution (155) or (156) which is a polynomial is written as $P_n(x)$, and a certain multiple of the other (infinite series) solution is written as $Q_n(x)$, so that the general solution of (149) in this case is written in the form

$$y = c_1 P_n(x) + c_2 Q_n(x).$$
(161)

It can be shown by direct (but involved) calculation that $P_n(x)$ and $Q_n(x)$ both satisfy the *recurrence formula*

$$n y_n(x) = (2n - 1) x y_{n-1}(x) - (n - 1) y_{n-2}(x).$$
(162)

Equation (162) permits the determination of expressions for Legendre functions in terms of corresponding functions of lower degree.

We next express $Q_0(x)$ and $Q_1(x)$ in closed form, by using the methods of Section 1.10. Reference to Problem 34, Chapter 1, shows that the functions $Q_n(x)$ are expressible in the form

$$Q_n(x) = A_n P_n(x) \int^x \frac{dx}{(1-x^2) [P_n(x)]^2} + B_n P_n(x),$$

where A_n and B_n are suitably chosen constants. In particular, since $P_0(x) = 1$, there follows

$$Q_0(x) = A_0 \int^x \frac{dx}{1-x^2} + B_0,$$

or, *if* |x| < 1,

$$Q_0(x) = \frac{A_0}{2} \log \frac{1+x}{1-x} + B_0$$

From (160) we obtain $Q_0(0) = u_0(1)v_0(0) = 0$ and $Q'_0(0) = u_0(1)v'_0(0) = 1$. Hence there follows $A_0 = 1$, $B_0 = 0$, and we have the result

$$Q_0(x) = \frac{1}{2} \log \frac{1+x}{1-x} = \tanh^{-1} x.$$
 (163)

Similarly, when n = 1, since $P_1(x) = x$ there follows

$$Q_1(x) = A_1 x \int^x \frac{dx}{x^2(1-x^2)} + B_1 x$$

or, *if* |x| < 1,

$$Q_1(x) = A_1\left(\frac{x}{2}\log\frac{1+x}{1-x}-1\right) + B_1x.$$

From (160) we have

$$Q_1(0) = -v_1(1) u_1(0) = -1$$
 and $Q'_1(0) = -v_1(1) u'_1(0) = 0.$

Hence we must take $A_1 = 1$ and $B_1 = 0$, and so obtain

$$Q_1(x) = \frac{x}{2} \log \frac{1+x}{1-x} - 1 = x Q_0(x) - 1.$$
 (164)

The series expansions of $Q_0(x)$ and $Q_1(x)$, in powers of x, are readily shown to be in agreement with the series indicated in (160).

Use of the recurrence formula (162) now permits the determination of $Q_n(x)$ for any positive integral value of n. In this way one obtains the expressions

$$\left.\begin{array}{l}
Q_{2}(x) = P_{2}(x) \ Q_{0}(x) - \frac{3}{2}x, \\
Q_{3}(x) = P_{3}(x) \ Q_{0}(x) - \frac{5}{2}x^{2} + \frac{2}{3}, \\
Q_{4}(x) = P_{4}(x) \ Q_{0}(x) - \frac{35}{8}x^{3} + \frac{55}{24}x, \\
Q_{5}(x) = P_{5}(x) \ Q_{0}(x) - \frac{63}{8}x^{4} + \frac{49}{8}x^{2} - \frac{8}{15}
\end{array}\right\}.$$
(165)

For values of x such that |x| > 1, the integral $\int^x \frac{dx}{1-x^2}$ takes the form

$$\frac{1}{2}\log\frac{x+1}{x-1} + C = \coth^{-1}x + C.$$

Thus, if |x| > 1, the function

$$Q_0(x) = \frac{1}{2} \log \frac{x+1}{x-1} = \coth^{-1} x \tag{166}$$

is a solution of (149) which complements the polynomial solution $P_0(x) = 1$ outside the interval (-1,1). Corresponding solutions $Q_n(x)$ for integral values

of *n*, in the range |x| > 1, are obtained by using the notation of (166) [in place of (163)] in Equations (164) and (165).

If p is not an integer, a certain combination of the series (155) and (156) can be determined so as to remain finite and take on the value unity at x = 1 (see Problem 43). This function is called $P_{p}(x)$, and a second independent combination is denoted by $Q_{p}(x)$, so that the general solution of (149) in the general case is written in the form

$$y = c_1 P_p(x) + c_2 Q_p(x).$$
(167)

The function $P_p(x)$ so defined, however, will not also remain finite at the point x = -1 unless p is integral, and the function $Q_p(x)$ cannot be finite at x = 1. Thus, the only Legendre functions which are finite at both x = 1 and x = -1 are the Legendre polynomials $P_p(x)$, for which p is integral.

Rodrigues' formula, which expresses $P_n(x)$ in the alternative form

$$P_n(x) = \frac{1}{2^n n!} \frac{d^n (x^2 - 1)^n}{dx^n},$$
 (168)

is particularly useful in dealing with certain integrals involving Legendre polynomials. Proof that (168) is indeed consistent with (157) for all positive integral values of n is omitted here (see Problem 41), but it is readily verified that (168) reduces, in the special cases of (158), to the forms given.

If |x| < 1, the substitution $x = \cos \theta$ transforms Legendre's equation from the form (149a) into the form

$$\frac{1}{\sin\theta}\frac{d}{d\theta}\left(\sin\theta\frac{dy}{d\theta}\right) + n(n+1)y = 0$$
(169)

or, equivalently,

$$\frac{d^2y}{d\theta^2} + \frac{dy}{d\theta}\cot\theta + n(n+1) y = 0, \qquad (169a)$$

when p = n, and hence (169) has the general solution

$$y = c_1 P_n(\cos \theta) + c_2 Q_n(\cos \theta). \tag{170}$$

Equations of such a form frequently arise in connection with solutions of various types of potential problems using spherical coordinates. It may be remarked that $Q_n(\cos \theta)$ is not finite when $\cos \theta = \pm 1$, that is, when $\theta = k\pi$, whereas $P_n(\cos \theta)$ is merely a polynomial of degree *n* in $\cos \theta$. In particular, we have the expressions

$$P_{0}(\cos \theta) = 1, \qquad P_{1}(\cos \theta) = \cos \theta,$$

$$P_{2}(\cos \theta) = \frac{1}{2}(3\cos^{2} \theta - 1) = \frac{1}{4}(3\cos 2\theta + 1),$$

$$P_{3}(\cos \theta) = \frac{1}{2}(5\cos^{3} \theta - 3\cos \theta) = \frac{1}{8}(5\cos 3\theta + 3\cos \theta)$$

$$\left. (171)\right.$$

When |x| < 1, the functions

$$P_n^m(x) = (1 - x^2)^{m/2} \frac{d^m P_n(x)}{dx^m}, \qquad Q_n^m = (1 - x^2)^{m/2} \frac{d^m Q_n(x)}{dx^m} \quad (172)$$

are called the associated Legendre functions of degree n and order m, of the first and second kinds, respectively. They can be shown to satisfy the differential equation

$$(1-x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + \left[n(n+1) - \frac{m^2}{1-x^2}\right]y = 0, \qquad (173)$$

which differs from (149) only in the presence of the term involving m, and reduces to (149) when m = 0. When |x| > 1, the definitions (172) are modified by replacing $(1 - x^2)$ by $(x^2 - 1)$.

The substitution $x = \cos \theta$ transforms (173) into the equation

$$\frac{d^2y}{d\theta^2} + \frac{dy}{d\theta}\cot\theta + [n(n+1) - m^2\csc^2\theta]y = 0, \qquad (174)$$

which thus is satisfied by $P_n^m(\cos \theta)$ and $Q_n^m(\cos \theta)$.

4.13. The hypergeometric function. Solutions of the differential equation

$$x(1-x)\frac{d^2y}{dx^2} + \left[\gamma - (\alpha + \beta + 1)x\right]\frac{dy}{dx} - \alpha\beta y = 0$$
(175)

are generally called hypergeometric functions, since their series representations are, in a sense, generalizations of the elementary geometric series. Since (175) is of type (52), with M = 1, the series (54) reduces here to the usual form

$$y(x) = \sum_{k=0}^{\infty} B_k x^{k+s}.$$
 (176)

The exponents are found to be

$$s=0, 1-\gamma, \tag{177}$$

so that only one solution of the assumed form can be expected when γ is integral. The recurrence formula is obtained in the form

$$(s+k)(s+k+\gamma-1)B_k = (s+k+\alpha-1)(s+k+\beta-1)B_{k-1} \qquad (k \ge 1), \quad (178)$$

from which there follows

$$B_{k}(s) = \frac{[(s+\alpha)(s+\alpha+1)\cdots(s+\alpha+k-1)]}{[(s+1)(s+2)\cdots(s+k)]} \times \frac{[(s+\beta)(s+\beta+1)\cdots(s+\beta+k-1)]}{[(s+\gamma)(s+\gamma+1)\cdots(s+\gamma+k-1)]} B_{0} \quad (k \ge 1).$$
(179)

Corresponding to the exponent s = 0, we thus obtain the solution

$$y = B_0 \left\{ 1 + \sum_{k=1}^{\infty} \frac{\left[\alpha(\alpha+1) \cdots (\alpha+k-1) \right] \left[\beta(\beta+1) \cdots (\beta+k-1) \right]}{\left[1 \cdot 2 \cdots k \right] \left[\gamma(\gamma+1) \cdots (\gamma+k-1) \right]} x^k \right\}.$$
(180)

The coefficient of B_0 in (180) is written as $F(\alpha, \beta; \gamma; x)$, and the series is known as the hypergeometric series or function,

$$F(\alpha,\beta;\gamma;x) = 1 + \frac{\alpha \cdot \beta}{1 \cdot \gamma} x + \frac{\alpha(\alpha+1)\beta(\beta+1)}{1 \cdot 2 \cdot \gamma(\gamma+1)} x^2 + \cdots$$
 (181)

It is found that the series (181) converges in the interval |x| < 1 and also that, when x = +1, the series converges only if $\gamma - \alpha - \beta > 0$ and, when x = -1, the series converges only if $\gamma - \alpha - \beta + 1 > 0$.

It may be noticed that if $\alpha = 1$ and $\beta = \gamma$, the series becomes the elementary geometric series

$$F(1, \beta; \beta; x) = 1 + x + x^{2} + \dots + x^{n} + \dots$$
$$= \frac{1}{1 - x} \quad (|x| < 1). \tag{182}$$

It is seen also that because of the symmetry in α and β , these parameters are interchangeable,

$$F(\alpha, \beta; \gamma; x) = F(\beta, \alpha; \gamma; x).$$
(183)

The solution (181) does not exist (in general) when γ is zero or a negative integer.

Corresponding to the exponent $s = 1 - \gamma$, we obtain the solution

$$y = B_0 x^{1-\gamma} \left\{ 1 + \sum_{k=1}^{\infty} \frac{\left[(\alpha - \gamma + 1)(\alpha - \gamma + 2) \cdots (\alpha - \gamma + k) \right]}{\left[1 \cdot 2 \cdots k \right]} \times \frac{\left[(\beta - \gamma + 1)(\beta - \gamma + 2) \cdots (\beta - \gamma + k) \right]}{\left[(2 - \gamma)(3 - \gamma) \cdots (k + 1 - \gamma) \right]} x^k \right\}.$$
 (184)

The series in braces in (184) is seen to differ from that in (180) only in that α , β , and γ in (180) are replaced by $(\alpha - \gamma + 1)$, $(\beta - \gamma + 1)$ and $(2 - \gamma)$, respectively, in (184). Hence (184) can be written in the form

$$y = B_0 x^{1-\gamma} F(\alpha - \gamma + 1, \beta - \gamma + 1; 2 - \gamma; x).$$
 (185)

The solution (185) does not exist (in general) when γ is a positive integer greater than unity. When $\gamma = 1$, the solution (185) becomes identical with (181).

Thus, if γ is not zero or an integer, the general solution of (175) can be expressed in the form

 $y = c_1 F(\alpha, \beta; \gamma; x) + c_2 x^{1-\gamma} F(\alpha - \gamma + 1, \beta - \gamma + 1; 2 - \gamma; x)$, (186) when |x| < 1. The exceptional cases can be discussed by the methods of Section 4.5.

Many elementary functions are expressible in terms of the hypergeometric function (181). In particular, the following examples may be given:

$$(1-x)^{-\alpha} = F(\alpha,\beta;\beta;x),$$

$$(1+x)^{-k} + (1-x)^{-k} = 2F\left(\frac{k}{2},\frac{k+1}{2};\frac{1}{2};x^2\right),$$

$$[1+\sqrt{1-x}]^{-k} = 2^{-k}F\left(\frac{k}{2},\frac{k+1}{2};k+1;x\right),$$

$$\log(1-x) = xF(1,1;2;x),$$

$$\log\frac{1+x}{1-x} = 2xF\left(\frac{1}{2},1;\frac{3}{2};x^2\right).$$

4.14. Series solutions valid for large values of x. In the preceding sections we have considered series solutions valid in an interval centered at the point x = 0, and have noticed that if solutions valid near a point $x = x_0$ were desired, such solutions could be conveniently obtained by first replacing $x - x_0$ by a new independent variable t and then seeking series solutions of the form

$$\sum A_n t^{n+s} = \sum A_n (x-x_0)^{n+s}$$

from the new equation. In such cases, the point $x = x_0$, naturally should be not worse than a regular singular point.

Thus, if series solutions of Bessel's equation of order zero in powers of x - 1 were required, we could set t = x - 1 and thus transform that equation to the form

$$(1 + t)\frac{d^2y}{dt^2} + \frac{dy}{dt} + (1 + t)y = 0.$$

Since the point t = 0 is an ordinary point, two solutions of the form $\sum A_n t^n$ can be obtained and rewritten finally in the desired form $\sum A_n (x - 1)^n$.

In order to investigate the behavior of solutions for large values of x, we are led to the possibility of replacing 1/x by a new independent variable t, and then of studying the behavior of solutions of the new equation for *small* values

of t, since $t \to 0$ as $|x| \to \infty$. If the new equation has the point t = 0 as an ordinary point, we obtain two solutions of the form $\sum A_n t^n = \sum A_n x^{-n}$, whereas for a regular singular point at least one solution $\sum A_n t^{n+s} = \sum A_n x^{-n-s}$ is obtained.

With the substitution x = 1/t, the equation

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x) y = 0$$
(187)

becomes

$$\frac{d^2 y}{dt^2} + \frac{1}{t^2} \left[2t - a_1 \left(\frac{1}{t} \right) \right] \frac{dy}{dt} + \frac{1}{t^4} a_2 \left(\frac{1}{t} \right) y = 0.$$
 (188)

If the point t = 0 is an ordinary point (or a singular point) of (188), it is conventional to say that "the point $x = \infty$ is an ordinary point (or a singular point) of (187)." The use of such a phrase is motivated by the fact that if t = 0is an ordinary point (or *regular* singular point) of (188), then (187) possesses solutions of the form $\sum A_n x^{-n}$ (or $\sum A_n x^{-n-s}$).

Equation (188) shows that in order that $x = \infty$ be an ordinary point of (187), the functions

$$\frac{1}{t^2} \left[2t - a_1 \left(\frac{1}{t} \right) \right] \quad \text{and} \quad \frac{1}{t^4} a_2 \left(\frac{1}{t} \right) \tag{189}$$

must be regular at t = 0, whereas in order that $x = \infty$ be a regular singular point of (187), the functions

$$\frac{1}{t}a_1\left(\frac{1}{t}\right) \quad \text{and} \quad \frac{1}{t^2}a_2\left(\frac{1}{t}\right) \tag{190}$$

must be regular at t = 0.

To illustrate, we notice that if a_1 and a_2 are constants in (187), the point $x = \infty$ is an *irregular singular point* unless $a_1 = a_2 = 0$, since the functions a_1/t and a_2/t^2 are not regular at t = 0.

Bessel's equation (63) also has an irregular singular point at $x = \infty$, since the function

$$\frac{1}{t^2}a_2\left(\frac{1}{t}\right) = \frac{1}{t^2} - p^2$$

is not regular at t = 0. Thus, Bessel's equation has x = 0 as a regular singular point and $x = \infty$ as an irregular singular point. All other points are ordinary.

Legendre's equation (149), however, has a regular singular point at $x = \infty$, since

$$\frac{1}{t}a_1\left(\frac{1}{t}\right) = \frac{2}{1-t^2}$$
 and $\frac{1}{t^2}a_2\left(\frac{1}{t}\right) = \frac{p(p+1)}{t^2-1}$

are regular at t = 0, whereas the functions (189) are not regular at t = 0. Thus, it is seen that Legendre's equation has regular singular points at $x = \pm 1$ and at $x = \infty$ and ordinary points elsewhere.

The expansion of $Q_n(x)$ in inverse powers of x can be expressed in terms of a hypergeometric series in the form

$$Q_n(x) = \frac{n!\sqrt{\pi}}{(n+\frac{1}{2})! (2x)^{n+1}} F\left(\frac{n+1}{2}, \frac{n+1}{2}; \frac{n+3}{2}; \frac{1}{x^2}\right).$$

For the hypergeometric equation (175) we find that $x = \infty$ is an ordinary point only if either $\alpha = 0$, $\beta = 1$ or $\alpha = 1$, $\beta = 0$, and is a regular singular point otherwise. Thus, except for the cases noted, the hypergeometric equation has regular singular points at x = 0, x = 1, and $x = \infty$ and ordinary points elsewhere. The general solution, valid when |x| > 1, is of the form

$$y = c_1 x^{-\alpha} F\left(\alpha, \alpha - \gamma + 1; \alpha - \beta + 1; \frac{1}{x}\right) + c_2 x^{-\beta} F\left(\beta, \beta - \gamma + 1; \beta - \alpha + 1; \frac{1}{x}\right),$$

provided that $\beta - \alpha$ is nonintegral. (See Problem 52.)

If, in the hypergeometric equation (175), we replace x by a new independent variable x/β , we obtain the equation

$$x\left(1-\frac{x}{\beta}\right)\frac{d^2y}{dx^2} + \left[(\gamma-x)-(1+\alpha)\frac{x}{\beta}\right]\frac{dy}{dx} - \alpha y = 0, \quad (191)$$

which has regular singular points at x = 0, $x = \beta$, and $x = \infty$, and whose general solution, when |x| < 1, is given by (186) with x replaced by x/β . If now we let $\beta \to \infty$, Equation (191) formally becomes

$$x\frac{d^2y}{dx^2} + (\gamma - x)\frac{dy}{dx} - \alpha y = 0.$$
(192)

In the transition from (191) to (192), we have moved the singular point at $x = \beta$ in (191) into coincidence or "confluence" with the second singular point at $x = \infty$. For this reason, (192) is known as the *confluent hypergeometric equation*. This equation is a special case of (52), as was pointed out at the end of Section 4.7. It is of interest to notice that $x = \infty$ is an *irregular* singular point of (192), formed by the confluence of two originally distinct regular singular singular points.

In cases where $x = \infty$ is an *irregular* singular point, it is frequently possible to obtain series of the type

$$y(x) \sim e^{rx} \sum_{k=0}^{\infty} \frac{A_k}{x^{k+s}}, \qquad (193)$$
such that *formal* substitution of the series into the differential equation reduces the equation to an identity. However, the series so obtained generally do not converge for any finite values of x. Still, they generally do have the property that if a finite number, N, of terms is retained, the sum, $S_N(x)$, of these terms approximates a solution y(x) in such a way that not only the difference $y(x) - S_N(x)$ but also the product $x^N[y(x) - S_N(x)]$ approaches zero as $|x| \rightarrow \infty$. Such series, called asymptotic expansions of a solution y(x), are of use not only in studying the *nature* of solutions for large values of |x| but also in actually calculating values of such solutions to within a predictable accuracy. The reason is that in such cases the error associated with calculating y(x) by using N terms is of the order of magnitude of the next following (neglected) term, provided that that term is numerically smaller than the last term retained. [This, of course, is not a general property of all divergent (or convergent) series.] Although this error eventually increases without limit as N increases, the first few successive terms frequently decrease rapidly in magnitude when xis large, so that it may be possible to stop with a term preceding a term of the order of magnitude of the tolerable error.*

The asymptotic approximations given in Equations (105-107) for the Bessel functions represent in each case the leading term of asymptotic expansions of this sort, which may be listed as follows:

$$J_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \left[U_{p}(x) \cos\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right) - V_{p}(x) \sin\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right) \right], \quad (194)$$

$$Y_{p}(x) \sim \sqrt{\frac{2}{\pi x}} \left[U_{p}(x) \sin\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right) + V_{p}(x) \cos\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right) \right],$$
 (195)

$$H_{p}^{(1)}(x) \sim \sqrt{\frac{2}{\pi x}} \left[U_{p}(x) + i V_{p}(x) \right] e^{i\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right)}, \tag{196}$$

$$H_{p}^{(2)}(x) \sim \sqrt{\frac{2}{\pi x}} \left[U_{p}(x) - i V_{p}(x) \right] e^{-i\left(x - \frac{\pi}{4} - \frac{p\pi}{2}\right)},$$
(197)

$$I_{p}(x) \sim \frac{e^{x}}{\sqrt{2\pi x}} W_{p}(x), \qquad (198)$$

$$K_{p}(x) \sim \frac{e^{-x}}{\sqrt{\frac{2}{\pi}x}} W_{p}(-x),$$
 (199)

* For further information on asymptotic expansions, see H. Jeffreys and B. S. Jeffreys, Methods of Mathematical Physics, Chap. 17, Cambridge University Press, London, 1956. References

where the asymptotic series for U_p , V_p , and W_p are of the form

$$U_{p}(x) = 1 - \frac{(4p^{2} - 1^{2})(4p^{2} - 3^{2})}{2! (8x)^{2}} + \frac{(4p^{2} - 1^{2})(4p^{2} - 3^{2})(4p^{2} - 5^{2})(4p^{2} - 7^{2})}{4! (8x)^{4}} - \cdots,$$
(200)

$$V_p(x) = \frac{4p^2 - 1^2}{1! 8x} - \frac{(4p^2 - 1^2)(4p^2 - 3^2)(4p^2 - 5^2)}{3! (8x)^3} + \cdots, \quad (201)$$

$$W_{p}(x) = U_{p}(ix) - iV_{p}(ix)$$

= $1 - \frac{4p^{2} - 1^{2}}{1! 8x} + \frac{(4p^{2} - 1^{2})(4p^{2} - 3^{2})}{2! (8x)^{2}} - \cdots$ (202)

These expansions are frequently useful in the solution of problems involving Bessel functions of large argument, when the accuracy afforded by the leading term is insufficient or subject to question.

Asymptotic series, as defined above, are sometimes also called *semi*convergent series.

REFERENCES

- 1. References at end of Chapter 1.
- 2. Davis, H. T., Tables of Higher Mathematical Functions, Principia Press, Bloomington, Ind., 1960.
- 3. Dwight, H. B., Tables of Integrals and Other Mathematical Data, 3rd ed., Dover Publications, Inc., New York, 1958.
- 4. Fletcher, A., J. C. P. Miller, and L. Rosenhead, An Index of Mathematical Tables, McGraw-Hill Book Company, Inc., New York, 1946.
- 5. Gray, A., G. R. Mathews, and T. M. MacRobert, A Treatise on Bessel Functions, 2nd ed., St. Martin's Press, New York, 1952.
- 6. Hobson, E. W., Theory of Spherical and Ellipsoidal Harmonics, Chelsea Publishing Company, New York, 1955.
- 7. Jahnke, E., F. Emde, and F. Losch, *Tables of Higher Functions*, McGraw-Hill Book Company, Inc., New York, 1960.
- 8. Knopp, K., Infinite Sequences and Series, Dover Publications, Inc., New York, 1955.
- 9. MacRobert, T. M., Spherical Harmonics, 2nd rev. ed., Dover Publications, Inc., New York, 1948.
- 10. Magnus, W., and F. Oberhettinger, Special Functions of Mathematical Physics, Chelsea Publishing Company, New York, 1949.

- 11. McLachlan, N. W., Bessel Functions for Engineers, 2nd ed., Oxford University Press, New York, 1955.
- 12. Watson, G. N., A Treatise on the Theory of Bessel Functions, The Macmillan Company, New York, 1945.
- 13. Whittaker, E. T., and G. N. Watson, *Modern Analysis*, Cambridge University Press, New York, 1958.

PROBLEMS

Section 4.1

1. Determine the interval of convergence for each of the following series, including consideration of the behavior of the series at each end point:

(a)
$$\sum_{n=0}^{\infty} \frac{2^n x^n}{n!}$$
, (b) $\sum_{n=1}^{\infty} (-1)^n \frac{n(x+1)^n}{2^n}$,
(c) $\sum_{n=1}^{\infty} (-1)^n \frac{(x-1)^{2n}}{3^n n}$, (d) $\sum_{n=1}^{\infty} \frac{k(k+1)\cdots(k+n-1)}{n!} x^n$,
(e) $\sum_{n=1}^{\infty} \frac{x^n}{n^n}$, (f) $\sum_{n=1}^{\infty} n^n x^n$,
(g) $\sum_{n=1}^{\infty} \frac{(n-\alpha)!}{n! \alpha!} (x-a)^n$, (h) $\sum_{n=0}^{\infty} \frac{(n!)^2}{(2n)!} x^n$,
(i) $\sum_{n=1}^{\infty} \frac{1}{n x^n}$, (j) $\sum_{n=0}^{\infty} \left(\frac{x}{x+2}\right)^n$.

2. Show that

(a)
$$e^x = 1 + \frac{x}{1!} + \frac{x^2}{2!} + \dots + \frac{x^n}{n!} + \frac{e^{\theta_1 x}}{(n+1)!} x^{n+1}$$
,
(b) $\cos x = 1 - \frac{x^2}{2!} + \frac{x^4}{4!} - \dots + (-1)^m \frac{x^{2m}}{(2m)!} + (-1)^{m+1} \frac{\cos(\theta_2 x)}{(2m+2)!} x^{2m+2}$,
where $0 < \theta_{1,2} < 1$.

Section 4.2

3. Obtain the general solution of each of the following differential equations in terms of Maclaurin series:

(a)
$$\frac{d^2y}{dx^2} = xy$$
,
(b) $\frac{d^2y}{dx^2} + x\frac{dy}{dx} - y = 0$,
(c) $x\frac{d^2y}{dx^2} - \frac{dy}{dx} - 4x^3y = 0$.

Problems

4. For each of the following differential equations, obtain the most general solution which is of the form $y = \sum A_k x^k$:

(a)
$$\frac{d^2y}{dx^2} + y = 0$$
,
(b) $\frac{d^2y}{dx^2} - (x - 3)y = 0$,
(c) $\left(1 - \frac{1}{2}x^2\right)\frac{d^2y}{dx^2} + x\frac{dy}{dx} - y = 0$,
(d) $x^2\frac{d^2y}{dx^2} - \frac{dy}{dx} + y = 0$,
(e) $(x^2 + x)\frac{d^2y}{dx^2} - (x^2 - 2)\frac{dy}{dx} - (x + 2)y = 0$,
(f) $x^2\frac{d^2y}{dx^2} - \frac{dy}{dx} = 0$.

Obtain three nonvanishing terms in each infinite series involved.

Section 4.3

5. (a-f) Locate and classify the singular points of the differential equations of Problem 4.

Section 4.4

6. Use the method of Frobenius to obtain the general solution of each of the following differential equations, valid near x = 0:

(a)
$$2x \frac{d^2 y}{dx^2} + (1 - 2x) \frac{dy}{dx} - y = 0,$$

(b) $x^2 \frac{d^2 y}{dx^2} + x \frac{dy}{dx} + \left(x^2 - \frac{1}{4}\right) y = 0,$
(c) $x \frac{d^2 y}{dx^2} + 2 \frac{dy}{dx} + xy = 0,$
(d) $x(1 - x) \frac{d^2 y}{dx^2} - 2 \frac{dy}{dx} + 2y = 0.$

7. Use the method of Frobenius to obtain the general solution of each of the following differential equations, valid near x = 0:

(a)
$$x^2 \frac{d^2 y}{dx^2} - 2x \frac{dy}{dx} + (2 - x^2) y = 0$$
,
(b) $(x - 1) \frac{d^2 y}{dx^2} - x \frac{dy}{dx} + y = 0$,
(c) $x \frac{d^2 y}{dx^2} - \frac{dy}{dx} + 4x^2 y = 0$,
(d) $(1 - \cos x) \frac{d^2 y}{dx^2} - \frac{dy}{dx} \sin x + y = 0$.

Obtain three nonvanishing terms in each infinite series of parts (a-c). In part (d), obtain two such terms in each series.

8. Find the general solution of the differential equation

$$x\frac{d^2y}{dx^2}+(c-x)\frac{dy}{dx}-ay=0,$$

valid near x = 0, assuming that c is nonintegral. The solution which is regular at x = 0 and which is unity at that point is called the *confluent hypergeometric* function and is usually denoted by M(a, c; x). Show then that if c is nonintegral, the general solution is of the form

$$y = c_1 M(a, c; x) + c_2 x^{1-c} M(1 + a - c, 2 - c; x).$$

9. (a) Show that the equation

$$x\frac{d^2y}{dx^2} + \frac{dy}{dx} - y = 0$$

possesses equal exponents $s_1 = s_2 = 0$ at x = 0.

(b) Obtain the regular solution, and denote by $u_1(x)$ the result of setting the leading coefficient A_0 equal to unity.

(c) Assume a second solution of the form

$$y_2(x) = C u_1(x) \log x + v(x),$$

where $C \neq 0$, and show that v(x) must satisfy the equation

$$x\frac{d^2v}{dx^2}+\frac{dv}{dx}-v=-2C\frac{du_1}{dx}$$

(d) Obtain one solution of this equation in the form

$$v(x) = \sum_{k=0}^{\infty} B_k x^{k+s_2} = \sum_{k=0}^{\infty} B_k x^k,$$

showing that C and B_0 are arbitrary, but taking C = 1 and $B_0 = 0$ for convenience. Hence obtain the general solution of the original equation in the form

$$y = c_1 u_1(x) + c_2[u_1(x) \log x + v(x)].$$

10. (a) Show that the equation

$$x\frac{d^2y}{dx^2} - y = 0$$

possesses exponents $s_1 = 1$ and $s_2 = 0$ at x = 0.

(b) Obtain the regular solution, corresponding to $s_1 = 1$, and denote by $u_1(x)$ the result of setting the leading coefficient equal to unity.

(c) Assume a second solution in the form

$$y_2(x) = C u_1(x) \log x + v(x),$$

where $C \neq 0$, and show that v(x) must satisfy the equation

$$x\frac{d^2v}{dx^2}-v=C\left(\frac{u_1}{x}-2\frac{du_1}{dx}\right).$$

(d) Obtain one solution of this equation in the form

$$v(x) = \sum_{k=0}^{\infty} B_k x^{k+s_2} = \sum_{k=0}^{\infty} B_k x^k,$$

showing that C and B_1 are arbitrary, but taking C = 1 and $B_1 = -1$ for convenience. Hence obtain the general solution of the original equation in the form

$$y = c_1 u_1(x) + c_2[u_1(x) \log x + v(x)].$$

11. (a) Show that a particular solution of the nonhomogeneous linear differential equation

$$x^{2}(1 + R_{1}x + \cdots)\frac{d^{2}y}{dx^{2}} + x(P_{0} + P_{1}x + \cdots)\frac{dy}{dx} + (Q_{0} + Q_{1}x + \cdots)y = x^{r}(d_{0} + d_{1}x + \cdots),$$

where r is any constant, and where all series converge in some interval about x = 0 or terminate, can be obtained in the form

$$y = x^r(A_0 + A_1x + \cdots) = \sum_{k=0}^{\infty} A_k x^{k+r},$$

if neither of the exponents s_1 and s_2 , for which

$$s^2 + (P_0 - 1)s + Q_0 = 0,$$

equals r or exceeds r by a positive integer.

(b) For an equation of the form

$$(1 + a_1x + \cdots)\frac{d^2y}{dx^2} + (b_0 + b_1x + \cdots)\frac{dy}{dx} + (c_0 + c_1x + \cdots)y = d_0 + d_1x + \cdots,$$

which hence possesses an ordinary point at x = 0 and a regular right-hand member, deduce the existence of a particular solution of the form

$$y = v(x) = x^2(A_0 + A_1x + \cdots) = \sum_{k=0}^{\infty} A_k x^{k+2}.$$

[If the equation is written in the form $y'' + a_1y' + a_2y = x^{r-2}d$, the series solutions will converge at least inside the largest interval in which the series representing the resultant functions xa_1 , x^2a_2 , and d would all converge. In the exceptional cases noted in part (a), logarithmic terms may be involved.]

12. Determine a particular solution of each of the following equations, in the form of a series valid near x = 0, by the method of Problem 11. In each case, obtain four nonvanishing terms.

(a)
$$\frac{d^2y}{dx^2} + y = e^x$$
,
(b) $\frac{d^2y}{dx^2} + xy = 1$,
(c) $x\frac{d^2y}{dx^2} - y = x$,
(d) $x^2\frac{d^2y}{dx^2} + y = \frac{e^x}{\sqrt{x}}$.

Sections 4.5 and 4.6

13. Obtain the general solution of the equation

$$x\frac{d^2y}{dx^2}+\frac{dy}{dx}-y=0$$

by the method of Section 4.5. (Compare Problem 9.)

14. Obtain the general solution of the equation

$$x\frac{d^2y}{dx^2}-y=0$$

by the method of Section 4.5. (Compare Problem 10.)

15. Obtain the general solution of the equation of Problem 8 when a = c = 1 by the methods of Section 4.5.

16. Obtain the general solution of the equation of Problem 8 when a = 1 and c = 0 by the methods of Section 4.5.

Section 4.7

17. The differential equation

$$x\frac{d^2y}{dx^2}+(1-x)\frac{dy}{dx}+ny=0$$

is known as Laguerre's equation.

(a) Verify that this equation is a special case of (52), with M = 1, and show that the exponents at x = 0 are both zero.

(b) Obtain the regular solution in the form

$$y_1(x) = B_0 \left[1 + \sum_{k=1}^{\infty} (-1)^k \frac{n(n-1)(n-2)\cdots(n-k+1)}{(k!)^2} x^k \right].$$

(c) Show that this solution is a polynomial of degree *n* when *n* is a nonnegative integer, and verify that the choice $B_0 = 1$ leads to the Laguerre polynomial of degree *n*, with the definition

$$L_n(x) = 1 - {n \choose 1} \frac{x}{1!} + {n \choose 2} \frac{x^2}{2!} - \cdots + \frac{(-x)^n}{n!},$$

where $\binom{n}{k}$ represents the binomial coefficient n!/[(n-k)!k!].

18. The differential equation

$$\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + 2ny = 0$$

is known as *Hermite's equation*. Verify that this equation is a special case of (52), with M = 2, and obtain the general solution in the form

$$y(x) = c_1 u_1(x) + c_2 u_2(x)$$

Problems

where

$$u_1(x) = 1 - \frac{n}{1} \frac{x^2}{1!} + \frac{n(n-2)}{1 \cdot 3} \frac{x^4}{2!} - \frac{n(n-2)(n-4)}{1 \cdot 3 \cdot 5} \frac{x^6}{3!} + \cdots$$

and

$$u_2(x) = x - \frac{n-1}{3} \frac{x^3}{1!} + \frac{(n-1)(n-3)}{3 \cdot 5} \frac{x^5}{2!} - \frac{(n-1)(n-3)(n-5)}{3 \cdot 5 \cdot 7} \frac{x^7}{3!} + \cdots$$

[Hence verify that the solution $u_1(x)$ is a polynomial of degree *n* when *n* is a positive even integer or zero, whereas $u_2(x)$ is a polynomial of degree *n* when *n* is a positive odd integer. Certain multiples of these polynomials are called *Hermite polynomials*.]

Section 4.8

19. Evaluate the following quantities, from the series definitions, to three-place accuracy:

(a)
$$J_1(0.3)$$
, (b) $Y_0(0.2)$, (c) $J_{0.75}(0.2)$,
(d) $I_2(1)$, (e) $H_0^{(1)}(0.2)$, (f) $J_1'(0.5)$.

20. Find the general solution of the simultaneous equations

$$\begin{cases} y + t \frac{dx}{dt} = 0 \\ \frac{dy}{dt} - tx = 0. \end{cases}$$

21. By making an appropriate change of variables, obtain the general solution of the differential equation

$$(Ax + B)\frac{d^2y}{dx^2} + A\frac{dy}{dx} + A^2(Ax + B)y = 0.$$

Section 4.9

22. Use Equations (101–107) to evaluate the following limits:

(a)
$$\lim_{x \to 0} x^{1/3} J_{-1/3}(x)$$
, (b) $\lim_{x \to 0} x Y_1(x)$,
(c) $\lim_{x \to 0} x^2 K_2(x)$, (d) $\lim_{x \to 0} \frac{J_n(x)}{x^n}$,
(e) $\lim_{x \to \infty} x \{ [J_p(x)]^2 + [Y_p(x)]^2 \}$.

23. Prove from the series definition that

$$\frac{d}{dx}[x^{-p}J_p(\alpha x)] = -\alpha x^{-p}J_{p+1}(\alpha x).$$

24. By using Equations (108) and (109), together with integration by parts, deduce the following reduction formulas:

(a)
$$\int x^m J_n(x) dx = x^m J_{n+1}(x) - (m - n - 1) \int x^{m-1} J_{n+1}(x) dx$$
,
(b) $\int x^m J_n(x) dx = -x^m J_{n-1}(x) + (m + n - 1) \int x^{m-1} J_{n-1}(x) dx$.

[Notice that the first reduction eventually yields a closed form, by iteration, when m - n is an odd positive integer, whereas the second reduction does so when m + n is an odd positive integer. When m and n are both even or both odd integers, appropriate use of one or both of the reductions will yield the sum of a closed form and a multiple of the integral $\int J_0(x) dx$. Whereas this integral cannot be further simplified, the function $\int_0^x J_0(t) dt$ is a *tabulated* function.]

25. Use the results of Problem 24 to deduce the following formulas:

(a)
$$\int x^{p+1} J_p(\alpha x) dx = \frac{1}{\alpha} x^{p+1} J_{p+1}(\alpha x) + C$$
,
(b) $\int x^{1-p} J_p(\alpha x) dx = -\frac{1}{\alpha} x^{1-p} J_{p-1}(\alpha x) + C$,
(c) $\int x^3 J_0(x) dx = x^3 J_1(x) - 2x^2 J_2(x) + C$,
(d) $\int x^6 J_1(x) dx = x^6 J_2(x) - 4x^5 J_3(x) + 8x^4 J_4(x) + C$,
(e) $\int J_3(x) dx = -J_2(x) - \frac{2}{x} J_1(x) + C$,
(f) $\int x J_1(x) dx = -x J_0(x) + \int J_0(x) dx + C$,
(g) $\int x^{-1} J_1(x) dx = -J_1(x) + \int J_0(x) dx + C$,
(h) $\int J_2(x) dx = -2 J_1(x) + \int J_0(x) dx + C$.

26. Establish the relation

$$J_0(x) = \frac{2}{\pi} \int_0^{\pi/2} \cos(x \sin \theta) \, d\theta$$

by verifying that the right-hand member satisfies Bessel's equation of order zero and investigating its value when x = 0. (The other Bessel functions have similar integral representations. See Problem 27, below, and Problem 45 of Chapter 5.)

27. (a) Deduce from Problem 26 that

$$J_1(x) = \frac{2}{\pi} \int_0^{\pi/2} \sin(x \sin \theta) \sin \theta \, d\theta.$$

(b) By integrating the right-hand member of this relation by parts, show that

$$\frac{1}{x}J_1(x) = \frac{2}{\pi}\int_0^{\pi/2} \cos(x\sin\theta)\cos^2\theta \,d\theta.$$

(c) Deduce from Problems 26 and 27(b) that

$$J_2(x) = \frac{2}{\pi} \int_0^{\pi/2} \cos(x \sin \theta) \cos 2\theta \ d\theta.$$

28. Establish the relation

$$e^{\frac{x}{2}\left(r-\frac{1}{r}\right)} = \sum_{n=-\infty}^{\infty} r^n J_n(x)$$

by first considering the expansion

$$e^{\frac{x}{2^{r}}}e^{-\frac{x}{2^{r}}} = \left[\sum_{j=0}^{\infty}\frac{(xr/2)^{j}}{j!}\right]\left[\sum_{k=0}^{\infty}(-1)^{k}\frac{(x/2r)^{k}}{k!}\right] = \sum_{j=0}^{\infty}\sum_{k=0}^{\infty}(-1)^{k}\frac{(x/2)^{j+k}}{j!\,k!}r^{j-k},$$

replacing j by n + k, where n is a new index of summation, and then identifying the coefficient of r^n in the resultant series. [The function $e^{\frac{x}{2}\left(r-\frac{1}{r}\right)}$ is known as the generating function for $J_n(x)$.]

29. (a) Use Abel's formula, Equation (65) of Chapter 1, to show that the Wronskian of any two solutions of Bessel's equation (63) or of the modified equation (93) is always of the form A/x, where A is an appropriate constant.

(b) From this result deduce that

$$J_{p}(x) Y'_{p}(x) - Y_{p}(x) J'_{p}(x) = \frac{C}{x}$$

and, by considering the limiting form as $x \to 0$, show that $C = 2/\pi$ and hence

$$J_{p}(x) Y'_{p}(x) - Y_{p}(x) J'_{\nu}(x) = \frac{2}{\pi x}$$

(c) In a similar way, show that

$$J_{p}(x) J'_{-p}(x) - J_{-p}(x) J'_{p}(x) = -\frac{2}{\pi x} \sin p\pi,$$
$$I_{p}(x) K'_{p}(x) - K_{p}(x) I'_{p}(x) = -\frac{1}{x},$$
$$I_{p}(x) I'_{-p}(x) - I_{-p}(x) I'_{p}(x) = -\frac{2}{\pi x} \sin p\pi.$$

[Make use of Equation (59), Chapter 2.]

30. Use the result of Problem 29(b), and the methods of Section 1.9, to obtain the general solution of the equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (x^{2} - p^{2})y = xf(x)$$

in the form

$$y = \frac{\pi}{2} \left[Y_{p}(x) \int^{x} f(\xi) J_{p}(\xi) d\xi - J_{p}(x) \int^{x} f(\xi) Y_{p}(\xi) d\xi \right] + c_{1} J_{p}(x) + c_{2} Y_{p}(x).$$

Section 4.10

31. Obtain the general solution of each of the following equations in terms of Bessel functions or, if possible, in terms of elementary functions:

(a)
$$x \frac{d^2 y}{dx^2} - 3 \frac{dy}{dx} + xy = 0$$
,
(b) $x \frac{d^2 y}{dx^2} - \frac{dy}{dx} + 4x^3 y = 0$,
(c) $x^2 \frac{d^2 y}{dx^2} + x \frac{dy}{dx} - (x^2 + \frac{1}{4})y = 0$,
(d) $x \frac{d^2 y}{dx^2} + (2x + 1)(\frac{dy}{dx} + y) = 0$,
(e) $x \frac{d^2 y}{dx^2} - \frac{dy}{dx} - xy = 0$,
(f) $x^4 \frac{d^2 y}{dx^2} + a^2 y = 0$,
(g) $\frac{d^2 y}{dx^2} - x^2 y = 0$,
(h) $x \frac{d^2 y}{dx^2} + (1 + 2x)\frac{dy}{dx} + y = 0$,
(i) $x \frac{d^2 y}{dx^2} + (1 + 4x^2)\frac{dy}{dx} + x(5 + 4x^2)y = 0$

32. Show that for the differential equation

$$x\frac{d^2y}{dx^2} + 3\frac{dy}{dx} + 4xy = 0$$

0.

the condition y(0) = 1 determines a unique solution, and hence that y'(0) cannot also be prescribed. Determine this solution.

33. Find the most general solution of the equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (x^{2} - 1)y = 0$$

for which

$$\lim_{x\to 0} 2\pi x \ y(x) = P,$$

where *P* is a given constant.

34. The differential equation for small deflections of a rotating string is of the form

$$\frac{d}{dx}\left(T\frac{dy}{dx}\right)+\rho\omega^2 y=0.$$

Obtain the general solution of this equation under the following assumptions:

(a)
$$T = T_0 x^n$$
, $\rho = \rho_0 x^n$; $T_0 = L^2 \rho_0 \omega^2$.
(b) $T = T_0 x^n$, $\rho = \rho_0$, $n \neq 2$; $T_0 = L^2 \rho_0 \omega^2$.
(c) $T = T_0 x^2$, $\rho = \rho_0$; $T_0 = 4\rho_0 \omega^2$.

Section 4.11

35. From the series definitions (137) and (138), and the definition (134), calculate the value of each of the following quantities to four significant figures:

(a) ber 1, (b) bei 1, (c) $M_0(1)$, (d) $\theta_0(1)$.

36. Establish the following relations, assuming that $i^{3/2} = e^{3\pi i/4}$:

(a) ber' x + i bei' $x = -i^{3/2} J_1(i^{3/2} x) = M_1(x) e^{i[\theta_1(x) - \pi/4]}$, (b) ber₁ $x = \frac{\sqrt{2}}{2}$ (ber' x - bei' x),

(c)
$$bei_1 x = \frac{\sqrt{2}}{2} (ber' x + bei' x).$$

37. (a) Use Equations (133) and (107) to obtain the results

$$\operatorname{ber}_{p} x \sim \frac{e^{x/\sqrt{2}}}{\sqrt{2\pi x}} \cos \varphi_{p}, \qquad \operatorname{bei}_{p} x \sim \frac{e^{x/\sqrt{2}}}{\sqrt{2\pi x}} \sin \varphi_{p},$$

as $x \to \infty$, where

$$\varphi_{p}=\frac{x}{\sqrt{2}}-\frac{\pi}{8}+\frac{p\pi}{2}.$$

(b) Use Equations (139) and (107) to obtain the results

$$\ker_{p} x \sim \frac{e^{-x/\sqrt{2}}}{\sqrt{\frac{2}{\pi}x}} \cos \psi_{p}, \qquad \ker_{p} x \sim -\frac{e^{-x/\sqrt{2}}}{\sqrt{\frac{2}{\pi}x}} \sin \psi_{p},$$

as $x \to \infty$, where

$$\psi_p = \frac{x}{\sqrt{2}} + \frac{\pi}{8} + \frac{p\pi}{2}.$$

38. (a) Show that Equation (131) can be put into the form

$$\frac{d}{dx}(xy') = \left(\frac{p^2}{x} + ix\right)y.$$

(b) Noticing that this equation is satisfied by the complex function $y = ber_p x + i bei_p x$, deduce the results

$$\frac{d}{dx}(x \operatorname{ber}'_{p} x) = -x \operatorname{bei}_{p} x + p^{2} \frac{\operatorname{ber}_{p} x}{x},$$
$$\frac{d}{dx}(x \operatorname{bei}'_{p} x) = x \operatorname{ber}_{p} x + p^{2} \frac{\operatorname{bei}_{p} x}{x}.$$

(c) Show similarly that the functions $\ker_p x$ and $\ker_p x$ satisfy completely analogous equations.

Section 4.12

39. Determine particular solutions of the partial differential equation

$$\frac{1}{\sin\theta}\frac{\partial}{\partial\theta}\left(\sin\theta\frac{\partial U}{\partial\theta}\right) + \frac{\partial}{\partial r}\left(r^2\frac{\partial U}{\partial r}\right) = 0$$

of the form $U(r,\theta) = r^n y(\theta)$, where n is a constant and y is independent of r.

40. (a) Use Equations (157) and (159) to obtain an expression for $P_n(x)$ in descending powers of x in the form

$$P_n(x) = \frac{(2n-1)(2n-3)\cdots 3\cdot 1}{n!} \left[x^n - \frac{n(n-1)}{2(1)(2n-1)} x^{n-2} + \frac{n(n-1)(n-2)(n-3)}{2^2(1\cdot 2)(2n-1)(2n-3)} x^{n-4} - \cdots \right].$$

(b) Verify that the coefficient of x^{n-2k} in the expression for $P_n(x)$ is

$$c_{nk} = (-1)^k \frac{(2n-2k)!}{2^n k! (n-k)! (n-2k)!}$$

where k may vary from 0 to n/2 or (n - 1)/2, according as n is even or odd.

41. (a) Use the binomial expansion to obtain the result

$$(x^2 - 1)^n = \sum_{k=0}^n (-1)^k \frac{n!}{k! (n-k)!} x^{2n-2k}.$$

(b) Hence show that, with the notation of Problem 40(b),

$$\frac{d^n}{dx^n}(x^2-1)^n = 2^n n! \sum_{k=0}^N c_{nk} x^{n-2k} = 2^n n! P_n(x),$$

where N = n/2 or (n - 1)/2, according as n is even or odd. [This establishes *Rodrigues' formula* (168).]

42. Establish the relation

$$(1 - 2rx + r^2)^{-1/2} = \sum_{n=0}^{\infty} r^n P_n(x)$$
 $(|x|, |r| < 1)$

by first obtaining the binomial expansion

$$[1 - r(2x - r)]^{-1/2} = 1 + \frac{r}{2}(2x - r) + \frac{1 \cdot 3}{2^2 2!}r^2(2x - r)^2 + \cdots$$
$$+ \frac{1 \cdot 3 \cdots (2n - 3)}{2^{n-1}(n - 1)!}r^{n-1}(2x - r)^{n-1}$$
$$+ \frac{1 \cdot 3 \cdots (2n - 1)}{2^n n!}r^n(2x - r)^n + \cdots,$$

then picking out the coefficient of r^n in this expression, and using the result of Problem 40(a). [Show that the coefficient of r^{λ} in $(2x - r)^m$ is

$$(-1)^k \frac{m(m-1)\cdots(m-k)}{k!} (2x)^{m-k}$$

when $k \ge 1$ and is $(2x)^m$ when k = 0, and notice that no terms beyond those written can involve r^n . The function $(1 - 2rx + r^2)^{-1/2}$ is called the *generating function* for $P_n(x)$.]

43. (a) Show that the substitution t = 1 - x transforms Legendre's equation to the form

$$t(2 - t)\frac{d^2y}{dt^2} + 2(1 - t)\frac{dy}{dt} + p(p + 1)y = 0,$$

that this equation has one solution regular at t = 0, with exponent zero, and that all other solutions become logarithmically infinite at t = 0.

(b) Determine the regular solution, and hence show that the solution $P_p(x)$ of Legendre's equation which is finite at x = 1 and which is unity at that point is given by

$$P_{p}(x) = 1 + \sum_{k=1}^{\infty} [(p+1)(p+2)\cdots(p+k)] \\ \times [(-p)(1-p)\cdots(k-1-p)] \frac{(1-x)^{k}}{2^{k}(k!)^{2}}$$

near x = 1.

(c) Verify that this result is in accordance with Equations (158) when p = 0, 1, and 2 by writing out the terms in the series.

Section 4.13

44. (a) Show that the change in variables x = 1 - t transforms the hypergeometric equation (175) to the form

$$t(1-t)\frac{d^2y}{dt^2} + \left[(\alpha + \beta - \gamma + 1) - (\alpha + \beta + 1)t\right]\frac{dy}{dt} - \alpha\beta y = 0.$$

(b) By comparing the coefficients in this equation with those in Equation (175), deduce that the general solution of (175) can be written in the form

$$y = c_1 F(\alpha, \beta; \alpha + \beta - \gamma + 1; 1 - x) + c_2 x^{\gamma - \alpha - \beta} F(\gamma - \beta, \gamma - \alpha; 1 - \alpha - \beta + \gamma; 1 - x)$$

near the second singular point at x = 1, if $\alpha + \beta - \gamma$ is nonintegral.

45. (a) Show that Jacobi's equation,

$$x(1-x)\frac{d^2y}{dx^2} + [a - (1+b)x]\frac{dy}{dx} + n(b+n)y = 0,$$

is a hypergeometric equation.

(b) Deduce that the general solution, near x = 0, can be expressed in the form

$$y = c_1 F(n + b, -n; a; x) + c_2 x^{1-a} F(n + b - a + 1, 1 - n - a; 2 - a; x),$$

if a is nonintegral.

46. Show that a solution of Jacobi's equation (Problem 45) which is regular at x = 0 can be written in the explicit form

$$J_n(a, b, x) = 1 - \frac{(n+b)n}{a} \frac{x}{1!} + \frac{(n+b)(n+b+1)n(n-1)}{a(a+1)} \frac{x^2}{2!} + \cdots,$$

if a is not zero or a negative integer, where

$$J_n(a, b, x) = F(n + b, -n; a; x).$$

(Notice that this solution is a polynomial of degree n when n is a positive integer or zero.)

Section 4.14

47. Show that the following differential equations each have an ordinary point at $x = \infty$ and in each case obtain two independent solutions expressed in the form $y = \sum A_k x^{-k}$:

(a)
$$x^2 \frac{d^2 y}{dx^3} + (1 + 2x) \frac{dy}{dx} = 0$$
, (b) $x^4 \frac{d^2 y}{dx^2} + 2x^3 \frac{dy}{dx} + y = 0$.

48. (a) Show that the differential equation

$$x^4\frac{d^2y}{dx^2}+y=0$$

has a regular singular point at $x = \infty$ and obtain two independent solutions directly in the form $y = \sum A_k x^{-k-s}$.

(b) Obtain the same solutions by first making the change in variables x = 1/t.

Problems

49. Verify that the substitution x = 1/t transforms the hypergeometric equation (175) to the form

$$t(1-t)\frac{d^2y}{dt^2}+\left[(1-\alpha-\beta)-(2-\gamma)t\right]\frac{dy}{dt}+\frac{\alpha\beta}{t}y=0,$$

and that this equation possesses a regular singular point at t = 0 unless $\alpha \beta = 0$ and $\alpha + \beta = 1$, in which case t = 0 is an ordinary point.

50. Show that the exponents of the differential equation obtained in Problem 49 are α and β at t = 0 and deduce that the transformed equation possesses solutions of the forms $t^{\alpha} u(t)$ and $t^{\beta} v(t)$, where u(t) and v(t) are regular at t = 0, unless α and β are equal or differ by an integer.

51. Verify that the substitution $y = t^{\alpha}u(t)$ transforms the equation of Problem 49 to the form

$$t(1-t)\frac{d^2u}{dt^2}+\left[(1+\alpha-\beta)-(2\alpha-\gamma+2)t\right]\frac{du}{dt}-\alpha(1+\alpha-\gamma)u=0.$$

52. Show that the equation obtained in Problem 51 is of the form of (175) if β is replaced by $1 + \alpha - \gamma$ and γ is replaced by $1 + \alpha - \beta$ in (175), and hence deduce that the functions

$$x^{-\alpha}F\left(\alpha, 1 + \alpha - \gamma; 1 + \alpha - \beta; \frac{1}{x}\right), \qquad x^{-\beta}F\left(\beta, 1 + \beta - \gamma; 1 - \alpha + \beta; \frac{1}{x}\right)$$

are independent solutions of the hypergeometric equation (175) for large values of x unless either α and β differ by an integer, in which case one of these expressions is undefined, or $\alpha = \beta$, in which case the two expressions are identical.

53. Use Equations (194) and (195) to verify the three-place values

- (a) $J_0(10) = -0.246$, (b) $Y_0(10) = 0.056$.
- 54. Show that the relations (194–199) are equalities for $p = \frac{1}{2}$.

55. Assume as a formal solution of the modified Bessel equation of order zero,

$$\frac{d^2y}{dx^2}+\frac{1}{x}\frac{dy}{dx}-y=0,$$

an expression of the form

$$y \sim e^{rx} \sum_{k=0}^{\infty} A_k x^{-(k+s)}$$

and obtain the formal requirement

$$(r^{2}-1)A_{0}x^{-s} + [(r^{2}-1)A_{1} + r(1-2s)A_{0}]x^{-s-1} + \sum_{k=0}^{\infty} [(r^{2}-1)A_{k+2} - r(1+2k+2s)A_{k+1} + (k+s)^{2}A_{k}]x^{-s-k-2} = 0.$$

Hence deduce that, if $A_0 \neq 0$, there must follow either

$$r = 1, \qquad s = \frac{1}{2}, \qquad A_{k+1} = \frac{(2k+1)^2}{8(k+1)}A_k \qquad (k \ge 0)$$

or

$$r = -1, \quad s = \frac{1}{2}, \qquad A_{k+1} = -\frac{(2k+1)^2}{8(k+1)}A_k \qquad (k \ge 0)$$

Thus obtain the formal solutions

$$y_{1}(x) \sim c_{1} \frac{e^{x}}{\sqrt{x}} \left[1 + \frac{1^{2}}{1! (8x)} + \frac{1^{2} \cdot 3^{2}}{2! (8x)^{2}} - \cdots \right]$$

$$y_{2}(x) \sim c_{2} \frac{e^{-x}}{\sqrt{x}} \left[1 - \frac{1^{2}}{1! (8x)} + \frac{1^{2} \cdot 3^{2}}{2! (8x)^{2}} - \cdots \right] \right\} \qquad (x \to \infty).$$

[With $c_1 = 1/\sqrt{2\pi}$, $c_2 = \sqrt{\pi/2}$, these results are respectively the asymptotic (but *divergent*) expansions of $I_0(x)$ and $K_0(x)$, the first terms of which are given in Equation (107). In accordance with the statement at the end of Section 4.14, retention of only the first term would introduce an error of less than approximately 1 per cent for values of x larger than about 12.]

CHAPTER 5

Boundary-Value Problems and Characteristic-Function Representations

5.1. Introduction. In many problems the solution of an ordinary differential equation must satisfy certain conditions which are specified for *two or more* values of the independent variable. Such problems are called *boundary-value problems*, in distinction with *initial-value problems*, wherein all conditions are specified at one point.

A condition or equation is said to be homogeneous if, when it is satisfied by a particular function y(x), it is also satisfied by c y(x), where c is an arbitrary constant. For example, the requirement that a function or one of its derivatives (or some linear combination of the function and/or certain of its derivatives) vanish at a point is a homogeneous condition. In the present chapter we are mainly concerned with homogeneous linear differential equations and associated homogeneous boundary conditions.

In illustration, we may require a solution of a homogeneous linear equation of second order, of the form

$$\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + a_2(x) y = 0$$
(1)

which vanishes at the two points x = a and x = b,

$$y(a) = 0, \quad y(b) = 0.$$
 (2)

Since the general solution of the differential equation is of the form

$$y = c_1 u_1(x) + c_2 u_2(x),$$
187
(3)

where u_1 and u_2 are linearly independent solutions and c_1 and c_2 are constants, the boundary conditions constitute the requirements

$$c_1 u_1(a) + c_2 u_2(a) = 0 \\ c_1 u_1(b) + c_2 u_2(b) = 0$$
 (4)

One solution of these equations is $c_1 = c_2 = 0$, leading to the *trivial solution* $y \equiv 0$. If the determinant of the coefficients of c_1 and c_2 does not vanish, then (by Cramer's rule) this is *the only solution*. Hence, in order that nontrivial solutions exist, it is necessary that the determinant of coefficients vanish,

$$\begin{vmatrix} u_1(a) & u_2(a) \\ u_1(b) & u_2(b) \end{vmatrix} = 0.$$
 (5)

If this condition exists, the two equations in (4) are in general equivalent and one constant can be expressed as a multiple of the second by use of either equation, the second constant then being arbitrary. Thus, if (5) is satisfied, the second equation of (4) may be discarded and the first equation then gives $c_2 u_2(a) = -c_1 u_1(a)$. If we write $c_1 = C u_2(a)$, there follows $c_2 = -C u_1(a)$, and the solution (3) becomes

$$y = C [u_2(a) u_1(x) - u_1(a) u_2(x)],$$
 (6)

with C arbitrary.

It should be noticed that (6) is a nontrivial solution only if $u_1(a)$ and $u_2(a)$ are not both zero. If $u_1(a) = u_2(a) = 0$, the first equation of (4) is the trivial identity, and the *second* equation must be used to relate c_1 and c_2 . This process leads to a nontrivial solution of the form

$$y = C [u_2(b) u_1(x) - u_1(b) u_2(x)],$$

assuming that $u_1(b)$ and $u_2(b)$ are not also both zero. If $u_1(x)$ and $u_2(x)$ should both vanish at x = a and at x = b, then (3) would satisfy (2) for arbitrary values of both c_1 and c_2 .

In many cases one or both of the coefficients $a_1(x)$ and $a_2(x)$ in (1), and hence the solutions $u_1(x)$ and $u_2(x)$, depend upon a constant parameter λ which may take on various constant values in a particular discussion. In such cases the determinant (5) may vanish for certain definite values of λ , say $\lambda = \lambda_1$, λ_2, \ldots . For each such value of λ a solution of type (6), involving an arbitrary multiplicative factor, is then obtained. Problems of this sort are known as *characteristic-value problems*; the values of λ for which nontrivial solutions exist are called the *characteristic values* of λ , and the corresponding solutions (with convenient choices of the arbitrary multiplicative constants) are called the *characteristic functions* of the problem. The terms "eigenvalues" and "eigenfunctions" are also frequently used. In the remainder of this chapter we first consider certain examples of such problems, and then establish and investigate certain rather general properties of characteristic functions which are extremely useful in a wide class of related problems.

5.2. The rotating string. We consider the problem of determining the form assumed by a tightly stretched flexible string of length L and linear density ρ , rotating with uniform angular velocity ω about its equilibrium position along the x axis. It is assumed that the initial tension in the string is so large that additional nonuniform stress introduced by the curvature of the string is relatively negligible. Denoting the displacement from the axis of rotation by y(x) and the uniform tensile force by T, and considering only small displacements and slopes, the condition of force equilibrium is of the form

$$T\frac{d^2y}{dx^2} + \rho\omega^2 y = 0. \tag{7}$$

If we consider a small element of the deformed string, projecting into the interval $(x, x + \Delta x)$, the y component of the tensile force is given by $-T\frac{dy}{ds}$ at the end (x,y) and by $\left(T\frac{dy}{ds}\right)_{x+\Delta x}$ at the end $(x + \Delta x, y + \Delta y)$, where s is arc length measured along the string. The differential resultant force on the element is thus given by $\frac{d}{dx}\left(T\frac{dy}{ds}\right)\Delta x$. If the distributed external force, per unit distance along the string, in the y direction is denoted by Y, then the requirement of differential force equilibrium is

$$\frac{d}{dx}\left(T\frac{dy}{ds}\right)\Delta x + Y\Delta s = 0$$

By dividing by Δx , letting $\Delta x \to 0$, and replacing Y in the present case by the inertia force $\rho \omega^2 y$, we thus obtain the equation

$$\frac{d}{ds}\left(T\frac{dy}{ds}\right) + \rho\omega^2 y = 0.$$

For small slopes we have

$$\frac{d(\cdots)}{ds} = \frac{\frac{d(\cdots)}{dx}}{\sqrt{1 + \left(\frac{dy}{dx}\right)^2}} \approx \frac{d(\cdots)}{dx} \qquad \left(\left| \frac{dy}{dx} \right| \ll 1 \right).$$

With this approximation and the assumption of nearly uniform tensile force, we obtain Equation (7).

Restricting attention to the case of a string of *uniform* density, we define the constant parameter

$$\lambda = \frac{\rho \omega^2}{T} \tag{8}$$

and write (7) in the form

$$\frac{d^2y}{dx^2} + \lambda y = 0. \tag{9}$$

If one end of the string is attached to the axis of rotation at the point x = 0, the end condition

$$y(0) = 0 \tag{10}$$

must be satisfied. If the other end is also attached to the axis, the second end condition is

$$y(L) = 0.$$
 (11a)

If, however, the second end is attached to a yielding support equivalent to an elastic spring which exerts a restoring force proportional to its stretch, toward

the axis of rotation, the second end condition is of the form $T\frac{dy}{dx} = -ky$, where k is the "spring constant." With the abbreviation

$$\alpha=\frac{T}{kL},$$

where α is a *dimensionless* constant inversely proportional to the elastic modulus of the spring support, this end condition can be written in the more convenient form

$$\alpha L y'(L) = -y(L).$$
 (11b)

In the limiting case where $\alpha = 0$, Equation (11b) reduces to the condition of fixity (11a); in the limiting case $\alpha = \infty$, the end at x = L is not restrained from motion normal to the axis of rotation, and we are led to the "free-end" condition

$$y'(L) = 0.$$
 (11c)

In order that the hypothesis of nearly uniform tension be realized, it must be supposed that the "free end" is, however, restricted from appreciable movement in the direction of the axis of rotation.

The most general solution of Equation (9) satisfying (10) is of the form

$$y = C \sin \sqrt{\lambda} x. \tag{12}$$

If the end x = L is *fixed*, the condition of (11a) becomes

$$C\sin\sqrt{\lambda} L = 0. \tag{13}$$

Hence, nontrivial solutions of this problem exist only if $\sqrt{\lambda}$ has a value such that

$$\sqrt{\lambda} L = n\pi \qquad (n = 1, 2, \ldots). \tag{14}$$

If these values of λ are ordered as $\lambda_1, \lambda_2, \ldots, \lambda_n, \ldots$, with respect to their magnitude, we can then write

$$\lambda_n = \frac{n^2 \pi^2}{L^2}, \qquad (15)$$

where *n* is an integer. For each such value of λ , the solution (12) can be written in the form

$$y = C \varphi_n(x),$$

where $\varphi_n(x) = \sin \frac{n\pi x}{L}.$ (16)

It is clear that only *positive* integral values of *n* need be considered.

Thus the boundary-value problem consisting of the differential equation (9) and the homogeneous boundary conditions (10) and (11a) has no solution other than the trivial solution $y \equiv 0$, unless λ has one of the *characteristic values* given by (15). Corresponding to each characteristic value of λ , there exists a characteristic function $\varphi_n(x)$, given by (16), such that any constant multiple of this function is a solution of the problem.

Each characteristic value λ_n corresponds to a definite value of the angular velocity ω (for given T and ρ), according to the notation of (8),

$$\omega_n = \frac{n\pi}{L} \sqrt{\frac{T}{\rho}} \qquad (n = 1, 2, \ldots). \tag{17}$$

These velocities are known as the critical speeds of the rotating string.

The present analysis indicates that for speeds smaller than ω_1 ,

$$0<\omega<rac{\pi}{L}\sqrt{rac{T}{
ho}}$$
 ,

the only stable position of the string is its undeformed position along the axis of rotation. However, if the speed is continuously increased until the first critical speed is attained ($\omega = \omega_1$), a possible new equilibrium form or deflection mode

$$y = C\sin\pi\frac{x}{L}$$

may exist. Corresponding to the second critical speed ($\omega = \omega_2 = 2\omega_1$) another mode

$$y = C \sin 2\pi \frac{x}{L}$$

may exist, and so on.

It may be seen that only the *shapes* of the string modes are determined, the amplitudes being apparently arbitrary. This indeterminacy is a result of the approximations made in deriving the linearized formulation of the problem and can be removed by a more nearly exact analysis of the true nonlinear problem.

Thus, if use is made of the additional relation $T\frac{dx}{ds} = T_0 = \text{constant}$,

which follows from the condition of force equilibrium in the x direction (in the absence of *external* force components in that direction), the tension T can be eliminated from the relation

$$\frac{d}{ds}\left(T\frac{dy}{ds}\right)+\rho\omega^2 y=0$$

to yield the nonlinear equation

$$T_0\frac{d^2y}{dx^2}+\rho\omega^2 y\sqrt{1+\left(\frac{dy}{dx}\right)^2}=0.$$

When ρ is constant, the appropriate solution of this equation can be obtained by use of so-called *elliptic integrals*,* after which T is given by the relation

$$T = T_0 \sqrt{1 + \left(\frac{dy}{dx}\right)^2}.$$

If the end x = L is attached to the axis of rotation by an elastic spring, the requirement that (12) satisfy (11b) becomes

$$-\alpha \sqrt{\lambda} L \cos \sqrt{\lambda} L = \sin \sqrt{\lambda} L.$$

With the introduction of a dimensionless parameter μ , of the form

$$\mu = \sqrt{\lambda} L, \tag{18}$$

this condition can be written

$$\tan \mu = -\alpha \mu. \tag{19}$$

That there are an infinite number of values of μ which satisfy (19), for a given value of α , is readily seen if the two curves $y = \tan \mu$ and $y = -\alpha \mu$ are plotted together and the desired roots are recognized as the values of μ corresponding

* The integrals

$$F(k,\varphi) = \int_0^{\varphi} \frac{d\varphi}{\sqrt{1-k^2 \sin^2 \varphi}} \quad \text{and} \quad E(k,\varphi) = \int_0^{\varphi} \sqrt{1-k^2 \sin^2 \varphi} \, d\varphi,$$

where 0 < k < 1, are known as *elliptic integrals of the first and second kind*, respectively, and are tabulated. For reductions expressing related integrals in terms of these integrals, see P. Franklin, *Treatise on Advanced Calculus*, pp. 218–228, John Wiley & Sons, Inc., New York, 1940.

to the intersections.* If we order the positive roots with respect to magnitude as $\mu_1, \mu_2, \ldots, \mu_n, \ldots$, the corresponding *characteristic values* of λ are given by

$$\lambda_n = \frac{\mu_n^2}{L^2} \tag{20}$$

and the corresponding deflection modes are of the form

$$\varphi_n(x) = \sin \mu_n \frac{x}{L}.$$
 (21)

Finally, the corresponding *critical speeds* are given by

$$\omega_n = \frac{\mu_n}{L} \sqrt{\frac{T}{\rho}}.$$
 (22)

In the limiting case $\alpha = \infty$, when the end x = L is not restrained from moving normal to the axis of rotation, Equation (19) gives

$$\cos \mu = 0, \qquad \mu_n = \frac{2n-1}{2}\pi.$$
 (23)

The characteristic values of λ are thus

$$\lambda_n = \left(\frac{2n-1}{2}\right)^2 \frac{\pi^2}{L^2}$$
 (*n* = 1, 2, ...), (24)

with corresponding characteristic modes

$$\varphi_n(x) = \sin \frac{2n-1}{2} \pi \frac{x}{L}, \qquad (25)$$

and the critical speeds are given by

$$\omega_n = \frac{2n-1}{2} \frac{\pi}{L} \sqrt{\frac{T}{\rho}}.$$
 (26)

5.3. The rotating shaft. We next consider the determination of the possible deflection modes of an originally straight shaft of length L, rotating with uniform angular velocity ω about its equilibrium position along the x axis. According to the elementary theory of bending of beams, the deflection y

* Once approximate values of roots of transcendental equations such as (19) have been obtained graphically, improved values often can be obtained by a "feedback" method, in which an approximation is inserted in *one side* of the equality and the resultant equation is then solved for the next approximation. *Newton's method* (see Section 7.10) is also useful.

from the straight form is determined (approximately) by the differential equation

$$\frac{d^2}{dx^2} \left(EI \frac{d^2 y}{dx^2} \right) - \rho \omega^2 y = 0, \qquad (27)$$

where E is Young's modulus, I is the moment of inertia of a cross section about an axis perpendicular to the xy plane, and ρ is the linear density of the shaft material.

If we consider the portion of the shaft to the left of an arbitrary cross section (x positive to the right, y positive upward), the influence of the right-hand portion of the shaft on the portion considered can be resolved into a



Figure 5.1

vertical shearing force S and a bending moment M. We choose the convention that S is then positive downward and M is positive when counterclockwise (see Figure 5.1). Thus, if the portion considered is to be in equilibrium, S must equal the algebraic sum of the actual upward forces (distributed and concentrated) acting to the left of the section considered, and M must equal

the resultant *clockwise* moment, about the section chosen, of all forces acting on the portion of the shaft to the left of the section. With these conventions, there follows also

$$M = EI \frac{d^2 y}{dx^2} \tag{28}$$

$$S = \frac{d}{dx} \left(EI \frac{d^2 y}{dx^2} \right).$$
 (29)

If we consider a small element of the shaft, in the interval $(x, x + \Delta x)$, and assume small deflections and slopes, the differential unbalanced vertical force acting on the element is given by $-\frac{dS}{dx}\Delta x$ and the differential unbalanced counterclockwise moment is $\frac{dM}{dx}\Delta x - S\Delta x$. With the distributed external load intensity (positive in the positive y direction) denoted by Y(x), the conditions of differential force and moment equilibrium become

$$-\frac{dS}{dx}\Delta x + Y\Delta x = 0, \qquad \frac{dM}{dx}\Delta x - S\Delta x = 0,$$

and there follows

$$\frac{dM}{dx} = S, \qquad \frac{dS}{dx} = Y.$$

194

and

The elementary theory of bending leads to the approximate relationship EI/R = M, where 1/R is the curvature of the shaft and is approximated by $\frac{d^2y}{dx^2}$ if the slope of the shaft is small. We thus obtain the relations (28) and (29), in addition to the equation

$$\frac{d^2}{dx^2}\left(EI\frac{d^2y}{dx^2}\right) = Y(x),$$

which reduces to (27) if Y is replaced by the inertia loading $\rho \omega^2 y$.

In addition to the differential equation (27), four boundary conditions must be prescribed, two conditions being imposed at each end of the shaft. Types of end conditions of particular interest are the following:

(1) Hinged end. In this case the lateral displacement y and the moment M must vanish, at the end considered, giving the two conditions

$$y = 0, \qquad EI \frac{d^2 y}{dx^2} = 0.$$
 (30a)

(2) Fixed end. Here the displacement y and the slope $\frac{dy}{dx}$ must vanish,

$$y = 0, \qquad \frac{dy}{dx} = 0. \tag{30b}$$

(3) Free end. If no moment or shearing force is applied at an end, there follows

$$EI\frac{d^2y}{dx^2} = 0, \qquad \frac{d}{dx}\left(EI\frac{d^2y}{dx^2}\right) = 0. \tag{30c}$$

(4) Sliding clamped end. If one end of the shaft is constrained to retain zero slope but is completely free to move in the y direction, the slope $\frac{dy}{dx}$ and the shear S must vanish at that end,

$$\frac{dy}{dx} = 0, \qquad \frac{d}{dx} \left(EI \frac{d^2 y}{dx^2} \right) = 0.$$
 (30d)

(5) Elastically supported end. If the motion of one end in the y direction is partially restrained as by an elastic spring, with modulus k_1 , the magnitude of the shearing force must be k_1 times the displacement,

$$\frac{d}{dx}\left(EI\frac{d^2y}{dx^2}\right) = \pm k_1 y. \tag{30e}$$

If the change in slope of the end is also partially restrained as by a spring system such that the restraining moment is proportional to the slope, the second condition at that end is of the form

$$EI\frac{d^2y}{dx^2} = \pm k_2 \frac{dy}{dx}.$$
 (30f)

By taking the four possible combinations of the limiting forms of these conditions when k_1 and k_2 are zero or infinitely large, the four preceding cases are attained. Thus $k_1 = \infty$, $k_2 = 0$ gives (30a); $k_1 = k_2 = \infty$ gives (30b); $k_1 = k_2 = 0$ gives (30c); and $k_1 = 0$, $k_2 = \infty$ gives (30d).

We restrict attention to the case of a *uniform* shaft, for which EI and ρ are constant, and introduce the constant parameter

$$\lambda = \frac{\rho \omega^2}{EI}.$$
 (31)

The differential equation (27) then becomes

$$\frac{d^4y}{dx^4} - \lambda y = 0, \tag{32}$$

with general solution of the form

$$y = c_1 \sinh \sqrt[4]{\lambda} x + c_2 \cosh \sqrt[4]{\lambda} x + c_3 \sin \sqrt[4]{\lambda} x + c_4 \cos \sqrt[4]{\lambda} x.$$
(33)

In the special case of a rotating shaft *hinged at both ends*, we take the origin at one end of the shaft and impose the boundary conditions

$$y(0) = y''(0) = 0, \quad y(L) = y''(L) = 0.$$
 (34)

From the conditions at x = 0 we obtain

$$c_2 + c_4 = 0, \qquad c_2 - c_4 = 0,$$

from which there follows

$$c_2 = c_4 = 0.$$
 (35)

To simplify the notation in the remaining work, it is convenient to introduce the dimensionless parameter μ , where

$$\mu = \sqrt[4]{\lambda} L, \qquad \mu^2 = \omega L^2 \sqrt{\frac{\rho}{EI}}.$$
 (36)

Making use of (35) and (36), we then write (33) in the form

$$y = c_1 \sinh \mu \frac{x}{L} + c_3 \sin \mu \frac{x}{L}.$$
 (37)

The last two conditions of (34) then become

$$c_{1} \sinh \mu + c_{3} \sin \mu = 0 \\c_{1} \sinh \mu - c_{3} \sin \mu = 0$$
(38a,b)

These equations are compatible only if the determinant of coefficients vanishes,

$$\begin{vmatrix} \sinh \mu & \sin \mu \\ \sinh \mu & -\sin \mu \end{vmatrix} = -2 \sinh \mu \sin \mu = 0.$$

Since the value $\mu = 0$ leads to a trivial solution y = 0, the only permissible values of μ which are of interest are the positive roots of the equation $\sin \mu = 0$,

$$\mu_n = n\pi$$
 (*n* = 1, 2, ...). (39)

For such values of μ , Equations (38a,b) require that

$$c_1 = 0.$$
 (40)

Hence the characteristic functions corresponding to permissible values μ_n are of the form $y = c_3 \varphi_n(x)$, where

$$\varphi_n(x) = \sin \frac{n\pi x}{L} \,. \tag{41}$$

These deflection modes are of the same form as the modes of a rotating flexible string with fixed ends. The characteristic values of λ in the present case are given by (36),

$$\lambda_n = \frac{n^4 \pi^4}{L^4}, \qquad (42)$$

and the critical speeds are given by

$$\omega_n = \frac{n^2 \pi^2}{L^2} \sqrt{\frac{EI}{\rho}}$$
 (n = 1, 2, ...). (43)

The smallest critical speed corresponds to n = 1. If we denote this *fundamental* critical speed by Ω ,

$$\Omega = \frac{\pi^2}{L^2} \sqrt{\frac{EI}{\rho}}, \qquad (44)$$

then the *n*th critical speed is given by

$$\omega_n = n^2 \Omega. \tag{45}$$

Thus, whereas Equation (17) shows that for a rotating string with fixed ends the successive critical speeds increase in proportion to the integers, Equation (45) shows that for a shaft hinged at both ends the speeds increase in proportion to the *squares* of the integers. 5.4. Buckling of long columns under axial loads. As a final example, we next consider a long shaft or column subjected to an axial compressive force P applied at an end and investigate possible modes of lateral deflection from the initial equilibrium position along the x axis. The differential equation determining the deflection y, according to the elementary theory, is then of the form

$$\frac{d^2}{dx^2}\left(EI\frac{d^2y}{dx^2}\right) + P\frac{d^2y}{dx^2} = 0.$$
(46)

The bending moment M_P at an arbitrary section x, due to a compressive force P applied in the negative x direction at the end x = L, is given by P[y(L) - y(x)]. Hence (see Section 5.3) the effect of the axial load can be taken into account if a fictitious distributed lateral loading of intensity

$$Y_P(x) = \frac{d^2 M_P}{dx^2} = -P \frac{d^2 y}{dx^2}$$

is introduced. If the *actual* distributed lateral load intensity is denoted by Y(x), the differential equation governing lateral deflection of the column is then

$$\frac{d^2}{dx^2}\left(EI\frac{d^2y}{dx^2}\right) = Y(x) - P\frac{d^2y}{dx^2}.$$

This equation reduces to (46) in the absence of distributed lateral loading.

The boundary conditions to be imposed depend upon the nature of the end restraints, as was outlined in the preceding section.

Restricting attention to a uniform column, and writing

$$\lambda = \frac{P}{EI}, \qquad (47)$$

Equation (46) then becomes

$$\frac{d^4y}{dx^4} + \lambda \frac{d^2y}{dx^2} = 0, \qquad (48)$$

with general solution of the form

$$y = c_1 \sin \sqrt{\lambda} x + c_2 \cos \sqrt{\lambda} x + c_3 x + c_4.$$
 (49)

In the special case when the two ends x = 0 and x = L are both hinged, the conditions

$$y(0) = y''(0) = 0$$

at the end x = 0 give $c_2 + c_4 = 0$ and $c_2 = 0$, from which there follows

$$c_2 = c_4 = 0,$$
 (50)

and Equation (49) takes the form

$$y = c_1 \sin \sqrt{\lambda} x + c_3 x. \tag{51}$$

The conditions

$$y(L) = y''(L) = 0$$

at the end x = L then give

$$c_1 \sin \sqrt{\lambda} L + c_3 L = 0, \quad c_1 \sin \sqrt{\lambda} L = 0$$
 (52)

so that nontrivial solutions of the problem exist only if $\sin \sqrt{\lambda} L = 0$ and $c_3 = 0$. Thus the characteristic values of λ are of the form

$$\lambda_n = \frac{n^2 \pi^2}{L^2}$$
 (*n* = 1, 2, ...), (53)

and the corresponding characteristic functions are the deflection modes

$$\varphi_n(x) = \sin \frac{n\pi x}{L} \,. \tag{54}$$

The values of the axial load P corresponding to these modes are known as the *critical buckling loads*, and are given by (47) and (53) in the form

$$P_n = n^2 \pi^2 \frac{EI}{L^2}$$
 (n = 1, 2, ...). (55)

For axial loads smaller in magnitude than the smallest critical value

$$P_1 = \pi^2 \frac{EI}{L^2},\tag{56}$$

the column is stable only in its unbent position. That is, if the column is artificially imparted a small amount of bending, it will tend to return to its initial straight form. When $P = P_1$, however, this theory predicts that under a small disturbance it will assume a form

$$y = C \sin \frac{\pi x}{L}, \qquad (57)$$

known as the fundamental buckling mode. Although the linear theory does not predict the amplitude C of the sinusoidal deformation, a more nearly exact analysis shows that the amplitude will increase with a small increase in the axial load P above the critical value P_1 , until bending failure of the column occurs. The critical value (56) is known as the *Euler load* of a column hinged at both ends.

In practice, failure is not caused entirely by bending except for very long columns, and the structure may fail before the Euler load is attained. However, the Euler load does give an *upper limit* of stability in the undeformed position. The higher modes cannot be attained in practice unless additional constraints are introduced.

If the end x = 0 is hinged and the end x = L is fixed, we find, by imposing the conditions

$$y(L) = y'(L) = 0$$

on Equation (51) and introducing the dimensionless parameter

$$\mu = \sqrt{\lambda} L = \sqrt{\frac{P}{EI}} L, \qquad (58)$$

that the critical loads are then of the form

$$P_n = \mu_n^2 \frac{EI}{L^2} , \qquad (59)$$

where μ_n is a positive solution of the transcendental equation

$$\tan \mu = \mu. \tag{60}$$

The corresponding deflection modes are of the form

$$\varphi_n(x) = \frac{\sin \mu_n \frac{x}{L}}{\sin \mu_n} - \frac{x}{L}.$$
(61)

Since the smallest positive root of (60) is approximately

$$\mu_1 = 1.43\pi$$
,

the Euler load in this case is approximately

$$P_1 = 2.05\pi^2 \frac{EI}{L^2}.$$
 (62)

5.5. The method of Stodola and Vianello. A certain iterative procedure, which often is useful for the approximate determination of the characteristic numbers and functions of a boundary-value problem, is sometimes known as the method of Stodola and Vianello. Before outlining the application of this procedure to problems of a general type, we first illustrate the method by considering a previously solved problem.

In the case of a rotating string fixed at its ends (Section 5.2), the boundary value problem consisting of the differential equation

$$\frac{d^2y}{dx^2} + \lambda y = 0 \tag{63}$$

and the end conditions

$$y(0) = 0, \quad y(L) = 0,$$
 (64)

was found to have the characteristic values

$$\lambda_n = \frac{n^2 \pi^2}{L^2}$$
 (*n* = 1, 2, ...), (65)

with corresponding characteristic functions

$$\varphi_n = \sin \frac{n\pi x}{L} \,. \tag{66}$$

The method to be presented is a method of successive approximations leading to approximate expressions for the *smallest* characteristic value and the corresponding characteristic function, although the method can be modified in such a way that the higher modes are obtained.*

We first transpose the term involving λ to the right in Equation (63), to obtain

$$\frac{d^2y}{dx^2} = -\lambda y. \tag{67}$$

Next, we replace the unknown function y on the right-hand side of (67) by a conveniently chosen first approximation $y_1(x)$. It is preferable that this function satisfy the prescribed end conditions although this is not necessary. The resulting equation,

$$\frac{d^2y}{dx^2} = -\lambda y_1(x),\tag{68}$$

then can be solved directly and the two constants of integration can be determined so that the solution satisfies the prescribed end conditions. It is clear that the solution so obtained will contain the parameter λ as a factor, and hence can be written in the form

$$y = \lambda f_1(x). \tag{69}$$

Now, if the originally assumed function $y_1(x)$ were actually a characteristic function of the problem, the function y(x) given by (69) would necessarily become identical with the assumed function $y_1(x)$ if λ were assigned the corresponding characteristic value. That is, the corresponding characteristic value of λ would be given by the constant ratio $y_1(x)/f_1(x)$.

Thus, if we take

$$y_1(x) = \sin \frac{\pi x}{L},$$

* See Problem 32. Also see, for example, L. Beskin and R. M. Rosenberg, "Higher Modes of Vibration by a Method of Sweeping," *Journal of the Aeronautical Sciences*, Vol. 13, No. 11, pp. 597-604 (1946).

the solution of (68) for which y(0) = y(L) = 0 is readily found to be

$$y(x) = \lambda \frac{L^2}{\pi^2} \sin \frac{\pi x}{L}.$$

In order that $y_1(x) = y(x)$, we must then have $\lambda = \pi^2/L^2$, in accordance with (65) and (66).

However, since $y_1(x)$ is generally not a characteristic function, the functions $y_1(x)$ and $f_1(x)$ will in general not be in a constant ratio. Still, as will be shown in a later section, if a convenient multiple of $f_1(x)$ is taken as a new approximation $y_2(x)$ and the process is repeated indefinitely, the ratio $y_n(x)/f_n(x)$ tends toward a constant value over the interval (0,L) as the number *n* of cycles is increased, and this constant value is exactly the *smallest* characteristic value λ_1 . Also, the successive functions $y_1(x), y_2(x), \ldots, y_n(x)$, converge to the corresponding characteristic function.

Successive estimates of the characteristic value λ_1 may be obtained after each cycle by requiring that the functions $y_n(x)$ and $y(x) = \lambda f_n(x)$ agree as well as possible (in some sense) over the interval (0,L). In particular, a simple procedure consists of determining λ so that the integral, over (0,L), of the difference between the functions be zero. Thus the *n*th approximation to the smallest permissible value of λ is given by

$$\lambda_{1}^{(n)} = \frac{\int_{0}^{L} y_{n}(x) \, dx}{\int_{0}^{L} f_{n}(x) \, dx} \,. \tag{70}$$

To illustrate this procedure in the present problem, we choose as our first approximation the function

$$y_1(x)=x(L-x),$$

which satisfies the end conditions (64). If y is replaced by $y_1(x)$ on the righthand side of (67), there results

$$\frac{d^2y}{dx^2} = \lambda(x^2 - Lx),$$

and the solution of this equation satisfying (64) is found by direct integration in the form

$$y=\frac{\lambda}{12}(x^4-2Lx^3+L^3x)\equiv\lambda f_1(x).$$

If the integrals of $y_1(x)$ and $y(x) = \lambda f_1(x)$, over (0,L), are equated, there follows

$$\frac{L^3}{6} = \lambda \frac{L^5}{60}$$

from which the first approximation to the smallest characteristic value of λ is

$$\lambda_1^{(1)} = \frac{10}{L^2}$$

Comparison with the exact value $\pi^2/L^2 = 9.870/L^2$ shows that the error is about 1.3 per cent.

If now we take

$$y_2(x) = x^4 - 2Lx^3 + L^3x$$

as the initial approximation of the next cycle, we obtain

$$y = \lambda f_2(x) = -\frac{\lambda}{30} (x^6 - 3Lx^5 + 5L^3x^3 - 3L^5x),$$

and Equation (70) gives the second approximation to λ_1 ,

$$\lambda_1^{(2)} = rac{9.882}{L^2}$$
 ,

which differs from the exact value by about 0.12 per cent.

The procedure as illustrated in this problem applies without essential modification to somewhat more involved boundary-value problems. Thus, for example, we may have differential equations of the form

$$\frac{d}{dx}\left[p(x)\frac{dy}{dx}\right] + \lambda r(x) y = 0$$
(71a)

$$\frac{d^2}{dx^2} \left[p(x) \frac{d^2 y}{dx^2} \right] + \lambda r(x) y = 0$$
(71b)

with appropriate homogeneous end conditions. If the term involving λ is transposed to the right and if the factor y in its coefficient is then replaced by a suitable approximation $y_1(x)$, the resultant differential equation can be solved by direct integration for a new function of the form $y = \lambda f_1(x)$ and the remaining procedure outlined above is directly applicable.

In place of using Equation (70) to determine successive approximations to λ_1 , one may make use of either of the following formulas (see Section 5.9):

$$\lambda_{1}^{(n)} = \frac{\int_{a}^{b} r(x) \left[y_{n}(x) \right]^{2} dx}{\int_{a}^{b} r(x) f_{n}(x) y_{n}(x) dx}$$
(72a)

$$\lambda_1^{(n)} = \frac{\int_a^b r(x) f_n(x) y_n(x) dx}{\int_a^b r(x) [f_n(x)]^2 dx}.$$
 (72b)

or

Either of these formulas will in general give a better approximation to the true value of λ_1 , the last form being in general the most accurate.* In illustration, we list the results of using the three formulas to estimate $L^2\lambda_1$ in the preceding example.

n	(70)	(72a)	(72b)
1 2 ∞	10.00000 9.88235 9.86960	9.88235 9.86975	9.87096

The tabulated values of $L^2 \lambda_1^{(n)}$ show that in this case only one approximation, using (72b), is needed to obtain an approximate value of λ_1 correct to within 2 parts in 10,000.

As a further application we investigate the Euler load of a column of length L, so constructed that its bending stiffness factor EI is of the form

$$EI(x) = C\frac{x}{L},\tag{73}$$

where C is the maximum stiffness factor (at x = L). We suppose that both ends of the column are *hinged*. Then the bending moment M(x) at a section x is due entirely to the axial load P and is given by M = -Py(x). Hence, from Equation (28) of Section 5.3, the relevant differential equation is of the form

$$C\frac{x}{L}\frac{d^2y}{dx^2} + Py = 0.$$
 (74)

(See also Problem 13.) If we introduce the dimensionless parameter

$$\mu^2 = \frac{PL^2}{C},\tag{75}$$

Equation (74) becomes

$$x\frac{d^2y}{dx^2} + \frac{\mu^2}{L}y = 0.$$
 (76)

The end conditions are of the form

$$y(0) = 0, \quad y(L) = 0.$$
 (77)

* It can be shown that, in the general case of (71a) or (71b), the result of using (72a) in the *n*th cycle is comparable with the result of using (70) in the (2*n*)th cycle, while the result of using (72b) in the *n*th cycle is comparable with that of using (70) in the (2*n* + 1)th cycle. If the initial approximation is such that $Ly_1 = \text{constant}$, these pairs of approximations are respectively equal.

According to Equation (130d) of Chapter 4, the general solution of (76) is of the form

$$y = \sqrt{x} Z_1 \left(2\mu \sqrt{\frac{x}{L}} \right)$$
$$y = c_1 \sqrt{x} J_1 \left(2\mu \sqrt{\frac{x}{L}} \right) + c_2 \sqrt{x} Y_1 \left(2\mu \sqrt{\frac{x}{L}} \right).$$

or

Since, for small values of x, we have \cdot

$$\sqrt{x} J_1\left(2\mu\sqrt{\frac{x}{L}}\right) \sim \frac{\mu}{\sqrt{L}} x, \quad \sqrt{x} Y_1\left(2\mu\sqrt{\frac{x}{L}}\right) \sim -\frac{\sqrt{L}}{\pi\mu}$$

the condition y(0) = 0 requires that

 $c_2 = 0,$

and the remaining condition y(L) = 0 requires that μ be a solution of the equation

$$J_1(2\mu) = 0. (78)$$

A rounded value of the smallest positive root of this equation (see Appendix B) is

$$\mu_1 = 1.916.$$
 (79)

Correspondingly, the Euler load is given by

$$P_1 = \mu_1^2 \frac{C}{L^2} = 3.670 \frac{C}{L^2},$$
(80)

and the fundamental buckling mode is given by

$$\varphi(x) = \sqrt{x} J_1\left(3.832\sqrt{\frac{x}{L}}\right). \tag{81}$$

To find an *approximate* value of μ , we take as our first approximation the function

$$y_1(x) = x(L-x).$$
 (82)

From (76) we then have

$$\frac{d^2y}{dx^2} = \frac{\mu^2}{L} (x-L),$$

from which there follows

$$y = \frac{\mu^2}{6L}(x^3 - 3Lx^2 + 2L^2x) \equiv \mu^2 f_1(x).$$
(83)

The condition

$$\int_{0}^{L} y_{1}(x) \, dx = \mu^{2} \int_{0}^{L} f_{1}(x) \, dx \tag{84}$$
then gives, as a first approximation,

$$\mu_1^{(1)} = 2, \tag{85}$$

leading to an approximation

$$P_1^{(1)} = 4 \frac{C}{L^2} \tag{86}$$

for the Euler load. A second approximation gives

$$\mu_1^{(2)} = 1.936, \quad P_1^{(2)} = 3.750 \, \frac{C}{L^2}.$$
 (87)

If the condition

with

$$\int_{0}^{L} r(x) [y_{n}(x)]^{2} dx = \mu^{2} \int_{0}^{L} r(x) f_{n}(x) y_{n}(x) dx$$
$$r(x) = \frac{1}{x},$$

corresponding to (72a), is used in place of (84) to determine approximations to μ_1 , the improved values $\mu_1^{(1)} = 1.936$ and $\mu_1^{(2)} = 1.917$ are obtained. Use of (72b) leads to the successive approximations 1.922 and 1.916.

An investigation of the convergence of the iterative procedure to the lowest characteristic mode, as well as an indication of modifications leading to the determination of higher modes, is presented in Section 5.9.

5.6. Orthogonality of characteristic functions. Two functions $\varphi_m(x)$ and $\varphi_n(x)$ are said to be orthogonal over an interval (a,b) if the integral of the product $\varphi_m \varphi_n$ over that interval vanishes:

$$\int_a^b \varphi_m(x) \varphi_n(x) \, dx = 0. \tag{88}$$

More generally, the functions $\varphi_m(x)$ and $\varphi_n(x)$ are said to be orthogonal with respect to a weighting function r(x), over an interval (a,b), if

$$\int_a^b r(x) \varphi_m(x) \varphi_n(x) dx = 0.$$
(89)

Finally, a set of functions is said to be orthogonal in (a,b) if all pairs of distinct functions in the set are orthogonal in (a,b).

An extremely useful property of boundary-value problems of a rather general type consists of the fact that the sets of characteristic functions corresponding to such problems are orthogonal with respect to a weighting function. In order to establish this fact, we consider first the boundary-value problem consisting of the linear homogeneous second-order differential equation

$$\frac{d}{dx}\left[p(x)\frac{dy}{dx}\right] + \left[q(x) + \lambda r(x)\right]y = 0$$
(90)

and suitably prescribed homogeneous boundary conditions, to be specified presently, at the ends of an interval (a,b). The functions p, q, and r are assumed to be *real*. In terms of the operator

$$L = \frac{d}{dx} \left(p \frac{d}{dx} \right) + q = p \frac{d^2}{dx^2} + \frac{dp}{dx} \frac{d}{dx} + q, \qquad (91)$$

Equation (90) can be written in the abbreviated form

$$L y + \lambda r(x) y = 0. \tag{92}$$

We notice that any equation of the form

$$a_0(x)\frac{d^2y}{dx^2} + a_1(x)\frac{dy}{dx} + [a_2(x) + \lambda a_3(x)] y = 0$$
(93)

can be written in the form of (90) by setting

$$p = e^{\int \frac{a_1}{a_0} dx}, \quad q = \frac{a_2}{a_0} p, \quad r = \frac{a_3}{a_0} p.$$
 (94)

Suppose now that λ_1 and λ_2 are any two different characteristic values of the problem considered and that the corresponding characteristic functions are $y = \varphi_1(x)$ and $y = \varphi_2(x)$, respectively. There then follows

$$\frac{d}{dx}\left(p\frac{d\varphi_{1}}{dx}\right) + (q + \lambda_{1}r)\varphi_{1} = 0$$

$$\frac{d}{dx}\left(p\frac{d\varphi_{2}}{dx}\right) + (q + \lambda_{2}r)\varphi_{2} = 0$$
(95)

If the first of these equations is multiplied by $\varphi_2(x)$ and the second by $\varphi_1(x)$, and the resultant equations are subtracted from each other, there follows

$$\varphi_2 \frac{d}{dx} \left(p \frac{d\varphi_1}{dx} \right) - \varphi_1 \frac{d}{dx} \left(p \frac{d\varphi_2}{dx} \right) + (\lambda_1 - \lambda_2) r \varphi_1 \varphi_2 = 0,$$

and hence

$$(\lambda_2 - \lambda_1) \int_a^b r \varphi_1 \varphi_2 \, dx = \int_a^b \left[\varphi_2 \frac{d}{dx} \left(p \frac{d\varphi_1}{dx} \right) - \varphi_1 \frac{d}{dx} \left(p \frac{d\varphi_2}{dx} \right) \right] \, dx. \tag{96}$$

Integrating the right member by parts, we obtain

$$(\lambda_2 - \lambda_1) \int_a^b r \varphi_1 \varphi_2 \, dx = \left[\varphi_2 \left(p \, \frac{d\varphi_1}{dx} \right) - \varphi_1 \left(p \, \frac{d\varphi_2}{dx} \right) \right]_a^b$$
$$- \int_a^b \left[\frac{d\varphi_2}{dx} \left(p \, \frac{d\varphi_1}{dx} \right) - \frac{d\varphi_1}{dx} \left(p \, \frac{d\varphi_2}{dx} \right) \right] \, dx.$$

Since the last integrand vanishes identically, there follows finally

$$(\lambda_2 - \lambda_1) \int_a^b r \varphi_1 \varphi_2 \, dx = \left[p(x) \left\{ \varphi_2(x) \, \frac{d\varphi_1(x)}{dx} - \varphi_1(x) \, \frac{d\varphi_2(x)}{dx} \right\} \right]_a^b. \tag{97}$$

In view of the fact that both $\varphi_1(x)$ and $\varphi_2(x)$ satisfy the conditions prescribed in connection with (90) at the points x = a and x = b, the right-hand member of (97) clearly vanishes if at *each* end point a prescribed condition is of *one* of the following forms:

$$\begin{cases} y = 0 \\ \frac{dy}{dx} = 0 \\ y + \alpha \frac{dy}{dx} = 0 \end{cases}$$
 (when $x = a$ or $x = b$). (98a,b,c)

The fact that the right-hand member of (97) vanishes at an end point for which $\varphi_1(x)$ and $\varphi_2(x)$ satisfy (98c), for arbitrary values of α , follows from the identity

$$\varphi_2\varphi_1'-\varphi_1\varphi_2' \equiv (\varphi_2+\alpha\varphi_2')\varphi_1'-(\varphi_1+\alpha\varphi_1')\varphi_2'.$$

Further, if it happens that

$$p(x) = 0$$
 when $x = a$ or $x = b$, (99)

then vanishing of the right-hand member of (97) at x = a or x = b is assured dy = dy

if only y is required to be finite at that point and if either $\frac{dy}{dx}$ is finite or $p\frac{dy}{dx}$ tends to zero at that point.

Finally, if

$$p(b) = p(a), \tag{100}$$

the right-hand member of (97) vanishes if the conditions

$$y(b) = y(a), \quad y'(b) = y'(a)$$
 (101)

are satisfied. It should be noticed that (101) will be satisfied, in particular, if the solutions $\varphi_1(x)$ and $\varphi_2(x)$ are required to be *periodic*, of period b - a.

In the cases listed, there follows

$$\int_{a}^{b} r(x) \varphi_{1}(x) \varphi_{2}(x) dx = 0 \quad \text{if } \lambda_{2} \neq \lambda_{1}; \quad (102)$$

that is, if the characteristic functions correspond to different characteristic numbers, then they are orthogonal with respect to the function r(x).

A boundary-value problem consisting of a differential equation of type (90), together with two homogeneous conditions of the types noted, is called a *Sturm-Liouville problem*. It can be shown that, except in the case of the periodicity

conditions (100) and (101), to each characteristic number λ_n there corresponds only one characteristic function $\varphi_n(x)$. From the homogeneity of the problem it is clear, however, that each such function involves an arbitrary multiplicative factor.

As an example, the problem

$$\frac{d^2y}{dx^2} + \lambda y = 0, \qquad y(0) = y(L) = 0$$

has the characteristic values $\lambda_n = n^2 \pi^2 / L^2$ with corresponding characteristic functions proportional to the functions $y_n = \sin(n\pi x/L)$. Since in this case r(x) = 1, there follows from (102)

$$\int_{0}^{L} y_{m}(x) y_{n}(x) dx = \int_{0}^{L} \sin \frac{m \pi x}{L} \sin \frac{n \pi x}{L} dx = 0 \qquad (m \neq n),$$

when m and n are positive integers. This fact is readily verified independently by direct integration.

In most applications, the functions p(x) and r(x) in Equation (90) are *positive* throughout the interval (a,b), including the end points, while the function q(x) is *nonpositive* in that interval. Furthermore, when one or both of the end conditions is of type (98c), with $\alpha \neq 0$, normally α is *negative* when the condition is imposed at the lower limit x = a but is *positive* when the condition is imposed at the upper limit x = b. We will speak of a Sturm-Liouville problem satisfying these restrictions as a *proper* problem.

For a proper Sturm-Liouville problem, it is known that all characteristic numbers are real and nonnegative, and that the corresponding characteristic functions are real (or can be made real by rejecting a possible complex constant multiplicative factor).

Furthermore, the weighted integral of the square of a characteristic function $\varphi_n(x)$,

$$C_{n} = \int_{a}^{b} r(x) [\varphi_{n}(x)]^{2} dx, \qquad (103)$$

then has a positive numerical value. If the arbitrary multiplicative factor involved in the definition of $\varphi_n(x)$ is so chosen that this integral has the value *unity*, the function $\varphi_n(x)$ is said to be *normalized* with respect to the weighting function r(x). A set of normalized orthogonal functions is said to be *orthonormal*.

Thus, in the above example, we find by direct integration that

$$C_n = \int_0^L y_n^2 \, dx = \int_0^L \sin^2 \frac{n \pi x}{L} \, dx = \frac{L}{2} \, .$$

Hence, in order to normalize the functions $\sin(n\pi x/L)$ over (0,L) we would divide them by the normalizing factor $\sqrt{L/2}$. The set of functions

$$\varphi_n(x) = \sqrt{\frac{2}{L}} \sin \frac{n\pi x}{L}$$

is thus an orthonormal set in (0,L).

Orthogonality properties of sets of characteristic functions generated by certain boundary-value problems of higher order are readily established. Thus, in the case of a problem involving the *fourth-order* differential equation

$$\frac{d^2}{dx^2} \left[s(x) \frac{d^2 y}{dx^2} \right] + \frac{d}{dx} \left[p(x) \frac{d y}{dx} \right] + \left[q(x) + \lambda r(x) \right] y = 0$$
(104)

and appropriate homogeneous boundary conditions, the methods used above lead to the equation

$$(\lambda_2 - \lambda_1) \int_a^b r\varphi_1 \varphi_2 \, dx = \left[\{ \varphi_2(s\varphi_1'')' - \varphi_1(s\varphi_2'')' \} - s(\varphi_2'\varphi_1'' - \varphi_1'\varphi_2'') + p(\varphi_2\varphi_1' - \varphi_1\varphi_2') \right]_a^b, \quad (105)$$

where $\varphi_1(x)$ and $\varphi_2(x)$ are characteristic functions corresponding to distinct characteristic numbers λ_1 and λ_2 , and primes indicate differentiation with respect to x. It follows, in particular, that the functions φ_1 and φ_2 are orthogonal with respect to r(x) over (a,b) if at *each* end of the interval there is prescribed one of the following pairs of conditions:

$$y = 0, \qquad \frac{dy}{dx} = 0$$

$$y = 0, \qquad s(x)\frac{d^2y}{dx^2} = 0$$

$$\frac{dy}{dx} = 0, \qquad \frac{d}{dx}\left[s(x)\frac{d^2y}{dx^2}\right] = 0$$
when $x = a$ or $x = b$. (106a,b,c)

5.7. Expansion of arbitrary functions in series of orthogonal functions. Suppose that we have a set of functions $\{\varphi_n(x)\}$, orthogonal in a given interval (a,b) with respect to a certain known weighting function r(x), and desire to expand a given function f(x) in terms of a series of these functions, of the form

$$f(x) = A_0 \varphi_0(x) + A_1 \varphi_1(x) + A_2 \varphi_2(x) + \cdots$$

= $\sum_{n=0}^{\infty} A_n \varphi_n(x).$ (107)

If we assume that such an expansion exists, and multiply both sides of (107) by $r(x) \varphi_k(x)$, where $\varphi_k(x)$ is the kth function in the set, we have

$$\mathbf{r}(x)f(x) \varphi_k(x) = \sum_{n=0}^{\infty} A_n \mathbf{r}(x) \varphi_n(x) \varphi_k(x).$$

Next, if we integrate both sides of this last equation over the interval (a,b) and assume that the integral of the infinite sum is equivalent to the sum of the integrals,* there follows formally

$$\int_{a}^{b} r(x) f(x) \varphi_{k}(x) dx = \sum_{n=0}^{\infty} A_{n} \int_{a}^{b} r(x) \varphi_{n}(x) \varphi_{k}(x) dx.$$
(108)

But by virtue of the orthogonality of the set $\{\varphi_n(x)\}$, all terms in the sum on the right are zero except that one for which n = k, and hence (108) reduces to the equation

$$A_n \int_a^b r(x) [\varphi_n(x)]^2 \, dx = \int_a^b r(x) f(x) \, \varphi_n(x) \, dx.$$
 (109)

Thus Equation (109) determines the constants involved in the required series (107) as the ratio of integrals of known functions.

With these values of the constants a formal series $\sum A_n \varphi_n(x)$ is determined. It should be emphasized, however, that we have not established the fact that this series actually does *represent* the function f(x) in the interval (a,b). In fact, we have not even shown that the series *converges* in (a,b) and hence represents any function in that interval. We may, however, speak of the series so obtained as the formal representation of the function f(x).

In illustration, we have seen that the set of functions $\{\sin(n\pi x/L)\}$ is orthogonal over (0,L), with the weighting function r(x) = 1. In the formal representation of the function f(x) = x in a series of these functions, the constants A_n are given, in accordance with (109), by the equation

or
$$A_n \int_0^L \sin^2 \frac{n\pi x}{L} \, dx = \int_0^L x \sin \frac{n\pi x}{L} \, dx$$
$$A_n = -\frac{2L}{n\pi} \cos n\pi = \frac{2L}{\pi} \frac{(-1)^{n+1}}{n}.$$

0

Hence we obtain the formal expression

$$x = \frac{2L}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n} \sin \frac{n\pi x}{L} = \frac{2L}{\pi} \left(\frac{1}{1} \sin \frac{\pi x}{L} - \frac{1}{2} \sin \frac{2\pi x}{L} + \cdots \right).$$

The general problem of determining whether the formal expansion obtained actually represents the given function in the interval involved is a difficult one

* This assumption is justified if the series (107) is uniformly convergent in (a,b).

efficients in the formal expansion would vanish and no expansion would be obtained. However, it can be shown that, except for artificial functions of no physical interest, there are no functions which are orthogonal to *all* members of any set generated as characteristic functions of a proper Sturm-Liouville problem. In this sense we say that the sets so generated are *complete*.

If the Sturm-Liouville problem is proper, and if also the relevant functions p(x), q(x), and r(x) are regular in the interval (a,b), then it is known that the formal representation of a bounded piecewise differentiable function f(x) in a series of the generated characteristic functions converges to f(x) inside (a,b) at all points where f(x) is continuous, and converges to the mean value $\frac{1}{2}[f(x+) + f(x-)]$ at points where finite jumps occur. Certain cases in which one or both of the functions p and r vanish at an end point of the interval are also of importance, as has been noted; although they require individual treatment, it has been shown that in the special cases to be treated here the convergence of the formal representation is again described by the preceding statement. The series may or may not converge to the value of the given function f(x) at the end points of the interval.

Thus, in the preceding illustration, the theorem stated shows that the expansion obtained converges to the function f(x) = x at all points *inside* the interval (0,L). Since all terms in the series vanish at *both* end points x = 0 and x = L, whereas f(x) vanishes only at x = 0, it is seen that the series also represents f(x) = x at x = 0, but *does not* represent it at the second end point x = L. From the convergence at the point x = L/2 we obtain, in particular,

$$\frac{L}{2} = \frac{2L}{\pi} \left(1 - \frac{1}{3} + \frac{1}{5} - \frac{1}{7} + \cdots \right)$$

and hence establish the useful result

$$1 - \frac{1}{3} + \frac{1}{5} - \frac{1}{7} + \cdots = \frac{\pi}{4}$$
.

It is important to notice the very great generality of such expansions. In the case of the power series expansions of Taylor and Maclaurin, a function cannot be representable over an interval unless *that function and all its derivatives* are continuous throughout that interval, and even these stringent conditions are not sufficient to ensure representation. However, in the present case of expansions in series of characteristic functions, a representable function may itself possess a finite number of finite discontinuities, and may even be defined by different analytical expressions over different parts of the interval of representation. Expansions of the type considered will be of particular usefulness in the solution of boundary-value problems involving *partial differential equations*, as will be seen in Chapter 9. We next point out certain other useful applications of the theory.

5.8. Boundary-value problems involving nonhomogeneous differential equations. Suppose that we require the solution of the nonhomogeneous differential equation

$$\left[\frac{d}{dx}\left(p\frac{dy}{dx}\right) + qy\right] + \lambda ry = F(x)$$
(110)

which satisfies prescribed *homogeneous* boundary conditions of the type listed in (98). Using the operational notation of (91), we may write (110) in the form

$$L y + \lambda r y = F(x). \tag{111}$$

With the given problem we may associate the boundary-value problem consisting of the *homogeneous* equation

$$L y + \lambda r y = 0 \tag{112}$$

and the prescribed boundary conditions. This problem determines a set of orthogonal characteristic functions $\{\varphi_n(x)\}$, such that the *n*th function $\varphi_n(x)$ corresponds to a characteristic number λ_n ,

$$L \varphi_n(x) + \lambda_n r(x) \varphi_n(x) = 0.$$
(113)

Assuming that the required solution of Equation (111) exists, we express it as a series of the form

$$y = \sum a_n \varphi_n(x) \tag{114}$$

and attempt to determine the coefficients a_n . For this purpose we introduce (114) into the differential equation (111). Then, since (113) gives

$$L\varphi_n=-\lambda_n r\varphi_n,$$

Equation (111) becomes formally

$$r(x)\sum (\lambda - \lambda_n) a_n \varphi_n(x) = F(x).$$
(115)

Hence, if we use (109) to calculate the coefficients A_n in the expansion

$$f(x) = \frac{F(x)}{r(x)} = \sum A_n \varphi_n(x), \qquad (116)$$

assuming that F(x)/r(x) is bounded and piecewise differentiable, a comparison of Equations (115) and (116) determines the required coefficients a_n as solutions of the equations

$$(\lambda - \lambda_n) a_n = A_n, \tag{117}$$

so that, provided that λ is not a characteristic number, the solution (114) of the nonhomogeneous problem becomes simply

$$y = \sum \frac{A_n}{\lambda - \lambda_n} \varphi_n(x) = \frac{A_0}{\lambda - \lambda_0} \varphi_0(x) + \frac{A_1}{\lambda - \lambda_1} \varphi_1(x) + \cdots$$
 (118)

It is immediately verified that the above derivation, beginning with Equation (111), is unchanged if the operator L in (111) is interpreted as the *fourth-order operator*,

$$L = \frac{d^2}{dx^2} \left[s(x) \frac{d^2}{dx^2} \right] + \frac{d}{dx} \left[p(x) \frac{d}{dx} \right] + q(x)$$
(119)

in Equation (104), and if boundary conditions of the type listed in (106) are then assumed.

From these results one may draw certain important conclusions. We have seen that if F(x) is identically zero, the problem consisting of (110) and the prescribed homogeneous boundary conditions has nontrivial solutions only if λ has a characteristic value $\lambda = \lambda_k$. Equation (118) shows that if F(x) is not identically zero, the corresponding problem has a solution in general only if λ does not take on one of the characteristic values of the homogeneous problem. More specifically, the solution (118) becomes nonexistent as $\lambda \to \lambda_k$ unless it happens that $A_k = 0$, that is, unless the integral

$$\int_a^b r(x) f(x) \varphi_k(x) dx = \int_a^b F(x) \varphi_k(x) dx$$

vanishes, so that F(x) is simply orthogonal to $\varphi_k(x)$. In that special case, Equation (117) shows that the coefficient of $\varphi_k(x)$ in (114) is *arbitrary*.

In connection with the physical problems of rotating strings and shafts (Sections 5.2 and 5.3), the presence of a prescribed function of the axial distance x on the right-hand side of the relevant differential equation would correspond to the presence of a distribution of transverse load. The above conclusions indicate that, for noncritical rotation speeds, definite deflection shapes are determined but that (according to the linearized theory) as a critical speed is approached, the amplitude of the deflection will in general increase without limit.

5.9. Convergence of the method of Stodola and Vianello. We may also make use of these developments in investigating the convergence of the iterative methods of Stodola and Vianello (Section 5.5). Considering the problem consisting of Equation (71),

$$\frac{d}{dx}\left[p(x)\frac{dy}{dx}\right] + \lambda r(x) y = 0, \qquad (120)$$

and appropriate homogeneous boundary conditions, the first cycle in the procedure consists of assuming an initial approximation $y_1(x)$ and determining the solution of the differential equation

$$\frac{d}{dx}\left[p(x)\frac{dy}{dx}\right] = -\lambda r(x) y_1(x), \qquad (121)$$

which satisfies the boundary conditions. The initial approximation may be imagined as expressed in terms of a series of the exact characteristic functions of the problem, say

$$y_1(x) = \sum_{n=1}^{\infty} A_n \varphi_n(x), \qquad (122)$$

where

$$\frac{d}{dx}\left[p(x)\frac{d\varphi_n}{dx}\right] = -\lambda_n r(x) \varphi_n(x) \qquad (n = 1, 2, 3, \ldots) \qquad (123)$$

and where $0 < |\lambda_1| < |\lambda_2| < \cdots$. The solution of (121) can also be

The solution of (121) can also be written in the form

$$y(x) = \sum_{n=1}^{\infty} a_n \varphi_n(x).$$
 (124)

To determine the coefficients in (124), we introduce (124) into (121) and obtain

$$\sum_{n=1}^{\infty} a_n \frac{d}{dx} \left[p(x) \frac{d\varphi_n}{dx} \right] = -\lambda r(x) \sum_{n=1}^{\infty} A_n \varphi_n(x),$$

or, using (123),

$$-\sum_{n=1}^{\infty} a_n \lambda_n r(x) \varphi_n(x) = -\lambda r(x) \sum_{n=1}^{\infty} A_n \varphi_n(x).$$

Hence, the coefficients in (124) are determined in the form

$$a_n = \frac{\lambda}{\lambda_n} A_n \tag{125}$$

and (124) becomes

$$y(x) = \lambda \sum_{n=1}^{\infty} \frac{A_n}{\lambda_n} \varphi_n(x)$$

or

$$y(x) = \lambda \left[\frac{A_1}{\lambda_1} \varphi_1(x) + \frac{A_2}{\lambda_2} \varphi_2(x) + \frac{A_3}{\lambda_3} \varphi_3(x) + \cdots \right]$$
$$= \frac{\lambda}{\lambda_1} \left[A_1 \varphi_1(x) + \frac{\lambda_1}{\lambda_2} A_2 \varphi_2(x) + \frac{\lambda_1}{\lambda_3} A_3 \varphi_3(x) + \cdots \right]. \quad (126)$$

If the last expression in brackets is taken as the initial approximation $y_2(x)$ in the second cycle, the resulting solution is found, by a repetition of the above process, in the form

$$y(x) = \frac{\lambda}{\lambda_1} \bigg[A_1 \varphi_1(x) + \bigg(\frac{\lambda_1}{\lambda_2} \bigg)^2 A_2 \varphi_2(x) + \bigg(\frac{\lambda_1}{\lambda_3} \bigg)^2 A_3 \varphi_3(x) + \cdots \bigg],$$

and, similarly, in the Nth cycle we have the initial approximation

$$y_N(x) = A_1 \varphi_1(x) + \left(\frac{\lambda_1}{\lambda_2}\right)^{N-1} A_2 \varphi_2(x) + \left(\frac{\lambda_1}{\lambda_3}\right)^{N-1} A_3 \varphi_3(x) + \cdots \quad (127)$$

and the corresponding solution

$$y(x) = \frac{\lambda}{\lambda_1} \left[A_1 \varphi_1(x) + \left(\frac{\lambda_1}{\lambda_2}\right)^N A_2 \varphi_2(x) + \left(\frac{\lambda_1}{\lambda_3}\right)^N A_3 \varphi_3(x) + \cdots \right]. \quad (128)$$

Since λ_1 is the *smallest* characteristic number, the ratios λ_1/λ_2 , λ_1/λ_3 , ..., are smaller than unity and increasing powers of these ratios tend to zero as $N \rightarrow \infty$, showing that successive approximations always tend to a multiple of the characteristic function corresponding to the smallest characteristic number λ_1 unless $A_1 = 0$, that is, unless the initial approximation $y_1(x)$ is orthogonal to $\varphi_1(x)$ with respect to r(x). It is also clear that the ratio $y_N(x)/f_N(x)$, where $f_{\Lambda}(x) = y(x)/\lambda$, tends to the limit λ_1 , as was stated in Section 5.5.

It is seen also that if the condition

. **A**

$$\int_{a}^{b} r(x) \varphi_{1}(x) y(x) dx = \int_{a}^{b} r(x) \varphi_{1}(x) y_{N}(x) dx \qquad (129)$$

is imposed in the Nth cycle, there follows from the orthogonality of the system

$$\frac{\lambda}{\lambda_1} A_1 \int_a^b r(x) [\varphi_1(x)]^2 dx = A_1 \int_a^b r(x) [\varphi_1(x)]^2 dx$$
$$\lambda = \lambda_1.$$

or

Hence, if the characteristic function $\varphi_1(x)$ were known exactly in any cycle of the calculation, Equation (129) would identify λ with the exact value of λ_1 . Although $\varphi_1(x)$ is not known exactly, an approximate determination of λ_1 may be accomplished by replacing $\varphi_1(x)$ by its approximation $y_N(x)$ or $f_N(x)$ in (129). This consideration is the motivation of Equations (72a) and (72b).

Modifications of the iterative procedure which permit numerical determination of the higher characteristic modes depend essentially on approximate methods of "subtracting off" the terms involving the lower modes in successive cycles.

5.10. Fourier sine series and cosine series. Since the boundary-value problem

$$\frac{d^2y}{dx^2} + \lambda y = 0; \qquad y(0) = 0, \quad y(L) = 0$$
(130)

has the characteristic functions

$$\varphi_n(x) = \sin \frac{n \pi x}{L}$$
 (n = 1, 2, 3, ...) (131)

corresponding to the characteristic numbers $\lambda_n = n^2 \pi^2 / L^2$, the results of the preceding section state that if a function f(x) can be represented by a series of the form

$$f(x) = A_1 \sin \frac{\pi x}{L} + A_2 \sin \frac{2\pi x}{L} + \cdots$$
$$= \sum_{n=1}^{\infty} A_n \sin \frac{n\pi x}{L} \qquad (0 < x < L), \qquad (132)$$

then the coefficients A_n are given, in accordance with (109) with r(x) = 1, by

$$A_n \int_0^L \sin^2 \frac{n\pi x}{L} \, dx = \int_0^L f(x) \sin \frac{n\pi x}{L} \, dx.$$
 (133)

This formula is a consequence of the orthogonality of the characteristic functions $\varphi_n = \sin(n\pi x/L)$ with respect to the weighting function r(x) = 1,

$$\int_0^L \sin \frac{m\pi x}{L} \sin \frac{n\pi x}{L} \, dx = 0 \qquad (m \neq n). \tag{134}$$

Since, by direct integration,

$$\int_{0}^{L} \sin^{2} \frac{n \pi x}{L} \, dx = \frac{L}{2} \qquad (n = 1, 2, 3, \ldots), \tag{135}$$

there follows

$$A_{n} = \frac{2}{L} \int_{0}^{L} f(x) \sin \frac{n\pi x}{L} \, dx.$$
 (136)

It may be noticed that the coefficient A_n in (136) is given by twice the average value of the product $f(x) \sin(n\pi x/L)$ in the interval (0,L).

The average value of a function F(x) in (a,b) is defined by the integral

$$\bar{F}=\frac{1}{b-a}\int_a^b F(x)\,dx.$$

If f(x) is piecewise differentiable in the interval (0,L), the series (132) converges to f(x) at points of continuity, and to the mean of the two values approached from the right and left at a finite discontinuity.

The series (132) is known as the Fourier sine series representation of f(x) in the interval (0,L). It is seen that all terms or "harmonics" in (132) are periodic and have the common period 2L; that is, the common period is twice the length

of the interval of representation (0, L). Also, if x is replaced by -x, the algebraic sign of each harmonic is merely reversed. Functions having this last property are known as *odd* functions of x.

In general, a function F(x) is said to be an *odd* function if F(-x) = -F(x) and an *even* function if F(-x) = F(x). In a graphical representation, y = F(x), an even function is *symmetrical* with respect to the y axis, whereas an odd function is *antisymmetrical*; that is, an odd function is symmetrical with respect to the origin of the coordinate system. Thus the functions sin mx and x^k , where k is an odd integer, are odd functions, and the functions as e^x and $\log x$ are *neither even nor odd*. It is clear that the product of two even functions or of two odd function is an odd function, whereas the product of an even function and an odd function, whereas the integral of an odd function is an even function, whereas the integral of an odd function is an even function, whereas the integral of an odd function is an even function, whereas the integral of an odd function is an even function, whereas the integral of an odd function is an odd function.

It follows that in the interval (-L,0) the series (132) represents the function -f(-x). Since all terms have the common period 2L, the behavior of the series in (-L,L) is repeated periodically for all values of x.

If f(x) is an odd function of x, the series (132) accordingly represents f(x) not only in the interval (0,L) but also in the larger interval (-L,L). If, in addition, f(x) is periodic, of period 2L, the series represents f(x) everywhere (with the usual convention regarding points of discontinuity).

Thus, the sine series representation of the odd function f(x) = x, given as an illustration in the preceding section, represents x in the larger interval (-L,L) except at the end points $x = \pm L$, and repeats this behavior *periodically* for all values of x (see Figure 5.2).

To obtain the Fourier sine series representing $f(x) = e^x$ in the interval $(0,\pi)$, we set $L = \pi$ in Equations (132) and (136) and obtain

$$A_n = \frac{2}{\pi} \int_0^{\pi} e^x \sin nx \, dx = \frac{2}{\pi} \frac{n}{n^2 + 1} (1 - e^x \cos n\pi).$$

Thus the desired expansion is of the form

$$e^{x} = \frac{2}{\pi} \left[\frac{e^{x} + 1}{2} \sin x - \frac{2(e^{x} - 1)}{5} \sin 2x + \frac{3(e^{x} + 1)}{10} \sin 3x + \cdots \right] \quad (0 < x < \pi). \quad (137)$$

The series (137) represents e^x when $0 < x < \pi$, and vanishes at the end points x = 0 and $x = \pi$. In the interval $-\pi < x < 0$ the series represents the function $-e^{-x}$, and, in general, the series represents an *odd* periodic function which



Figure 5.2

coincides with e^x when $0 < x < \pi$, with $-e^{-x}$ when $-\pi < x < 0$, which is zero when $x = -\pi$, 0, π , and which is of period 2π .

Similar series developments involving cosine terms, rather than sine terms, may be obtained by considering the boundary-value problem

$$\frac{d^2y}{dx^2} + \lambda y = 0; \qquad y'(0) = 0, \quad y'(L) = 0, \tag{138}$$

which has the characteristic functions

$$\varphi_n(x) = \cos \frac{n \pi x}{L}$$
 (*n* = 0, 1, 2, ...) (139)

corresponding to the characteristic numbers $\lambda_n = n^2 \pi^2 / L^2$. It is important to notice that $\varphi_0(x) = 1$ is a member of the set (139), corresponding to $\lambda_0 = 0$. The functions (139) possess the orthogonality property

$$\int_0^L \cos \frac{m\pi x}{L} \cos \frac{n\pi x}{L} \, dx = 0 \qquad (m \neq n), \tag{140}$$

as can be independently verified by direct integration. The constants in a series representation of the form

$$f(x) = A_0 + A_1 \cos \frac{x\pi}{L} + A_2 \cos \frac{2\pi x}{L} + \cdots$$

= $A_0 + \sum_{n=1}^{\infty} A_n \cos \frac{n\pi x}{L}$ (0 < x < L) (141)

are given by (109), with r(x) = 1, in the forms

$$A_0 \int_0^L dx = \int_0^L f(x) \, dx, \qquad A_n \int_0^L \cos^2 \frac{n\pi x}{L} \, dx = \int_0^L f(x) \cos \frac{n\pi x}{L} \, dx.$$

Since
$$\int_{0}^{L} \cos^{2} \frac{n\pi x}{L} dx = \begin{cases} L/2 & (n = 1, 2, 3, \ldots) \\ L & (n = 0), \end{cases}$$
(142)

there follows

$$A_{0} = \frac{1}{L} \int_{0}^{L} f(x) dx$$

$$A_{n} = \frac{2}{L} \int_{0}^{L} f(x) \cos \frac{n\pi x}{L} dx \qquad (n = 1, 2, 3, ...)$$
(143)

Hence the coefficient A_0 in (141) is the average value of f(x) in (0,L), whereas the coefficient A_n , when $n \neq 0$, is twice the average value of the product $f(x) \cos(n\pi x/L)$ in (0,L).

With these values of the coefficients, the series (141) is known as the *Fourier* cosine series representation of f(x) in (0,L). It is seen that all harmonics in



Figure 5.3

(141) are even functions of x and are again periodic, of period 2L. Although representation of a piecewise differentiable function f(x), in the sense described previously, is in general valid only when 0 < x < L, the series will also represent the function f(-x) in the interval -L < x < 0 because of the evenness of the harmonics as functions of x. Hence if f(x) is an even function, that is, if f(-x) = f(x), the series will represent f(x) in the double interval -L < x < L. If, in addition, f(x) is periodic, of period 2L, the series will represent f(x) everywhere (with the usual convention regarding points of discontinuity).

Thus the cosine series representation of the *odd* function f(x) = x in (0,L) would represent x only in (0,L), and would represent f(-x) = -x in (-L,0) (see Figure 5.3).

To obtain the Fourier cosine series representing $f(x) = e^x$ in the interval $(0,\pi)$, we set $L = \pi$ in (141) and (143) and obtain

$$A_0 = \frac{1}{\pi} \int_0^\pi e^x \, dx = \frac{e^\pi - 1}{\pi} \,,$$
$$A_n = \frac{2}{\pi} \int_0^\pi e^x \cos nx \, dx = \frac{2}{\pi} \frac{1}{n^2 + 1} \left(e^\pi \cos n\pi - 1 \right) \qquad (n \neq 0).$$

Thus the desired expansion is of the form

$$e^{x} = \frac{2}{\pi} \left(\frac{e^{x} - 1}{2} - \frac{e^{x} + 1}{2} \cos x + \frac{e^{x} - 1}{5} \cos 2x - \frac{e^{x} + 1}{10} \cos 3x + \cdots \right) \qquad (0 \le x \le \pi).$$
(144)

The series (144) must represent e^x when $0 < x < \pi$, and must represent e^{-x} when $-\pi < x < 0$. In general, the series represents an *even* periodic function which coincides with e^x when $0 < x < \pi$ and with e^{-x} when $-\pi < x < 0$, and which is of period 2π . Since this periodic function is *continuous* everywhere when made so at x = 0 and $x = \pi$, the series must converge to the values of e^x at these points, as well as in the interior of $(0,\pi)$.

We notice that the two expansions (137) and (144) both represent e^x inside the interval $(0,\pi)$ but represent entirely distinct functions outside this interval. It may be remarked that half the sum of the two series will represent a periodic function which is zero when $-\pi < x < 0$ and which coincides with e^x when $0 < x < \pi$. At x = 0 this composite series, consisting of both sine and cosine terms, converges to the value $\frac{1}{2}$, whereas at the points $x = \pm L$ the series converges to the value $\frac{1}{2}e^{\pi}$.

5.11. Complete Fourier series. It has now been seen that any reasonably well-behaved function f(x) can be represented in the interval (0,L) by a series consisting either of sines or of cosines with common period P = 2L, and that for an odd function f(x) the sine series representation is valid in (-L,L), whereas for an even function f(x) the cosine series representation is valid in (-L,L). We next show that if both sines and cosines of common period 2L are used, a representation valid in the interval (-L,L) can be obtained for any function f(x) which is piecewise differentiable in (-L,L).

It is clear that any function f(x) can be expressed as the sum of an even function $f_e(x)$ and an odd function $f_o(x)$,

$$f(x) = f_e(x) + f_o(x),$$
 (145)

since if we write

$$f(x) = \frac{1}{2}[f(x) + f(-x)] + \frac{1}{2}[f(x) - f(-x)], \quad (146)$$

the first bracket is an even function of x, whereas the second bracket is an odd function. The preceding theory states that the expansions

$$f_{e}(x) = A_{0} + \sum_{n=1}^{\infty} A_{n} \cos \frac{n\pi x}{L}$$

$$f_{o}(x) = \sum_{n=1}^{\infty} B_{n} \sin \frac{n\pi x}{L}$$
(147)

are valid in the interval (-L,L), if the coefficients are calculated by the formulas

$$A_0 = \frac{1}{L} \int_0^L f_e(x) \, dx,$$
$$A_n = \frac{2}{L} \int_0^L f_e(x) \cos \frac{n\pi x}{L} \, dx,$$
$$B_n = \frac{2}{L} \int_0^L f_o(x) \sin \frac{n\pi x}{L} \, dx.$$

Since all integrands are *even* functions of x, we clearly may replace \int_0^L by $\frac{1}{2} \int_{-L}^L$ in each case. Finally, if we use (145) to replace $f_e(x)$ by $f(x) - f_o(x)$ in the first two integrals and replace $f_o(x)$ by $f(x) - f_e(x)$ in the third integral, and notice that the integral of an odd function over a symmetrical interval (-L,L) vanishes, there follows

$$A_{0} = \frac{1}{2L} \int_{-L}^{L} f(x) \, dx - \frac{1}{2L} \int_{-L}^{L} f_{o}(x) \, dx = \frac{1}{2L} \int_{-L}^{L} f(x) \, dx,$$

$$A_{n} = \frac{1}{L} \int_{-L}^{L} f(x) \cos \frac{n\pi x}{L} \, dx - \frac{1}{L} \int_{-L}^{L} f_{o}(x) \cos \frac{n\pi x}{L} \, dx$$

$$= \frac{1}{L} \int_{-L}^{L} f(x) \cos \frac{n\pi x}{L} \, dx,$$

$$B_{n} = \frac{1}{L} \int_{-L}^{L} f(x) \sin \frac{n\pi x}{L} \, dx - \frac{1}{L} \int_{-L}^{L} f_{e}(x) \sin \frac{n\pi x}{L} \, dx$$

$$= \frac{1}{L} \int_{-L}^{L} f(x) \sin \frac{n\pi x}{L} \, dx.$$

Hence, introducing (147) into (145), there follows

$$f(x) = A_0 + \sum_{n=1}^{\infty} \left(A_n \cos \frac{n\pi x}{L} + B_n \sin \frac{n\pi x}{L} \right) \qquad (-L < x < L), \qquad (148)$$

where
$$A_0 = \frac{1}{2L} \int_{-L}^{L} f(x) \, dx, \qquad A_n = \frac{1}{L} \int_{-L}^{L} f(x) \cos \frac{n\pi x}{L} \, dx,$$

 $B_n = \frac{1}{L} \int_{-L}^{L} f(x) \sin \frac{n\pi x}{L} \, dx.$ (149)

The series (148) is the complete Fourier series representation of f(x) in the interval (-L,L). It is seen that again A_0 is the average value of f(x), the average now being calculated over the interval of representation (-L,L), whereas A_n and B_n are twice the average values of $f(x) \cos(n\pi x/L)$ and $f(x) \sin(n\pi x/L)$, respectively, over the same interval. For an even function there follows $B_n = 0$ and the cosine series previously obtained results, whereas for an odd function $A_0 = A_n = 0$ and the previously obtained sine series results. If f(x) is neither even nor odd, the series (148) will contain both sines and cosines of period 2L and will represent f(x), in the sense described, in the interval (-L,L) of length 2L. Because of the periodicity of all terms in the series, this representation is repeated periodically for all values of x.

To obtain a Fourier series of period 2π , representing $f(x) = e^x$ in the interval $(-\pi,\pi)$, we set $L = \pi$ in (148) and (149) and obtain

$$A_{0} = \frac{1}{2\pi} \int_{-\pi}^{\pi} e^{\pi} dx = \frac{1}{2\pi} (e^{\pi} - e^{-\pi}) = \frac{1}{\pi} \sinh \pi,$$

$$A_{n} = \frac{1}{\pi} \int_{-\pi}^{\pi} e^{x} \cos nx \, dx = \frac{2}{\pi} \frac{\cos n\pi}{n^{2} + 1} \sinh \pi \qquad (n \neq 0),$$

$$B_{n} = \frac{1}{\pi} \int_{-\pi}^{\pi} e^{x} \sin nx \, dx = -\frac{2}{\pi} \frac{n \cos n\pi}{n^{2} + 1} \sinh \pi.$$

The desired expansion is thus of the form

$$e^{x} = \frac{\sinh \pi}{\pi} \left[1 - \cos x + \frac{2}{5} \cos 2x - \frac{1}{5} \cos 3x + \cdots + \sin x - \frac{4}{5} \sin 2x + \frac{3}{5} \sin 3x + \cdots \right] \quad (-\pi < x < \pi)$$
(150)

or

$$e^{x} = \frac{\sinh \pi}{\pi} \bigg[1 - 2 \sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n^{2} + 1} (\cos nx - n \sin nx) \bigg]$$

$$(-\pi < x < \pi). \quad (151)$$

The series (150) converges to e^x at all points inside the interval $(-\pi,\pi)$. Since $e^x = \cosh x + \sinh x$, it follows that the even terms in (150) represent the function $\cosh x$ in $(-\pi,\pi)$, whereas the odd terms represent sinh x:

$$\cosh x = \frac{\sinh \pi}{\pi} \left[1 - 2 \sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n^2 + 1} \cos nx \right] \qquad (-\pi \le x \le \pi),$$
$$\sinh x = 2 \frac{\sinh \pi}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^{n+1}n}{n^2 + 1} \sin nx \qquad (-\pi < x < \pi).$$

If we think of x as measured as counterclockwise distance along the circumference of a circle of unit radius, then the interval $-\pi < x < \pi$ corresponds to the entire circumference of the circle. As $x \to \pi$ from smaller values or as $x \to -\pi$ from larger values, the *same* point on the circle is approached from different directions and the same values are assumed by any term in the series. This fact suggests that at the two end points $x = \pm \pi$ the series (150) converges to the mean of the two values of f(x) at these points, that is, to the value $\frac{1}{2}(e^{\pi} + e^{-\pi}) = \cosh \pi$. This can be shown to be actually the case.

In general, it can be shown that the series (148) converges at both end points $x = \pm L$ to the value $\frac{1}{2}[f(L) + f(-L)]$, that is, to the mean of the two values of f(x) at the end points. Hence, if f(L) = f(-L), the series converges to f(x) at the end points. The sine series (132) vanishes identically at the end points of the interval (0,L) and hence converges to f(x) at these points only if f(x) vanishes. The cosine series (141) converges to f(x) at both end points of the interval (0,L).

We next show that the constants in (148) can be determined in such a way that the given function f(x) is represented in *any* specified interval of length 2L, that is, of length equal to the common period of the trigonometric functions involved. Suppose that a function f(x) is specified in an interval (a, a + 2L)and that it is required that the coefficients in (148) be determined so that (148) represents f(x) in the usual way in that interval. We consider a new function $f^*(x)$ defined for all values of x in such a way that $f^*(x)$ coincides with f(x)in the given interval, and furthermore that $f^*(x)$ is a periodic function, of period 2L. Then if f(x) is replaced by $f^*(x)$ in the definitions of the constants (149), the corresponding series (148) clearly will represent the periodic function $f^*(x)$ for all values of x (in the usual sense), and hence, in particular, will represent f(x) in the interval (a, a + 2L). But since the integrands in (149) are then periodic functions, of period 2L, the range of integration may equally well be taken as *any* interval of length 2L. In particular, we may choose the interval over which representation is desired, and over which $f^*(x) = f(x)$.

It follows that if a representation of the form

$$f(x) = A_0 + \sum_{n=1}^{\infty} \left(A_n \cos \frac{2n\pi x}{P} + B_n \sin \frac{2n\pi x}{P} \right)$$
(152)

is required over the interval (a, a + P), the coefficients are to be determined by the equations

$$A_{0} = \frac{1}{P} \int_{a}^{a+P} f(x) dx, \qquad A_{n} = \frac{2}{P} \int_{a}^{a+P} f(x) \cos \frac{2n\pi x}{P} dx,$$
$$B_{n} = \frac{2}{P} \int_{a}^{a+P} f(x) \sin \frac{2n\pi x}{P} dx. \qquad (153)$$

In particular, representation in the interval (0, P) is obtained by setting a = 0.

These general results are also obtainable directly by considering the boundary-value problem

$$\frac{d^2y}{dx^2} + \lambda y = 0;$$
 $y(a) = y(a + P), y'(a) = y'(a + P)$

for which the characteristic values $\lambda_n = 4n^2\pi^2/P^2$ correspond to the twoparameter characteristic functions

$$\varphi_n = c_n \sin \frac{2n\pi x}{P} + d_n \cos \frac{2n\pi x}{P}.$$

The orthogonality of these functions for arbitrarily assigned values of both c_n and d_n , in the interval (a, a + P), corresponds to the relations

$$\int_{a}^{a+P} \sin \frac{2m\pi x}{P} \sin \frac{2n\pi x}{P} dx = 0 \qquad (m \neq n),$$

$$\int_{a}^{a+P} \cos \frac{2m\pi x}{P} \cos \frac{2n\pi x}{P} dx = 0 \qquad (m \neq n),$$

$$\int_{a}^{a+P} \sin \frac{2m\pi x}{P} \cos \frac{2n\pi x}{P} dx = 0$$
(154)

These relations are readily established by direct integration.

As a final example, we suppose that it is required to find a Fourier series of period 6 which in the interval (1,7) represents a function f(x) taking on the constant value +1 when 1 < x < 4 and the constant value -1 when 4 < x < 7. With a = 1, P = 6, the series (152) becomes

$$f(x) = A_0 + \sum_{n=1}^{\infty} \left(A_n \cos \frac{n\pi x}{3} + B_n \sin \frac{n\pi x}{3} \right),$$
(155)

where, in accordance with (153),

$$\begin{aligned} A_0 &= \frac{1}{6} \int_1^7 f(x) \, dx = \frac{1}{6} \left[\int_1^4 1 \, dx + \int_4^7 (-1) \, dx \right] = 0, \\ A_n &= \frac{1}{3} \int_1^7 f(x) \cos \frac{n\pi x}{3} \, dx = \frac{1}{3} \left[\int_1^4 \cos \frac{n\pi x}{3} \, dx - \int_4^7 \cos \frac{n\pi x}{3} \, dx \right] \\ &= \frac{1}{n\pi} \left[2 \sin \frac{4n\pi}{3} - \sin \frac{n\pi}{3} - \sin \frac{7n\pi}{3} \right] \quad (n \neq 0), \\ B_n &= \frac{1}{3} \int_1^7 f(x) \sin \frac{n\pi x}{3} \, dx = \frac{1}{3} \left[\int_1^4 \sin \frac{n\pi x}{3} \, dx - \int_4^7 \sin \frac{n\pi x}{3} \, dx \right] \\ &= -\frac{1}{n\pi} \left[2 \cos \frac{4n\pi}{3} - \cos \frac{n\pi}{3} - \cos \frac{7n\pi}{3} \right]. \end{aligned}$$

By virtue of the relations

$$\sin \frac{4n\pi}{3} = \cos n\pi \sin \frac{n\pi}{3}, \qquad \sin \frac{7n\pi}{3} = \sin \frac{n\pi}{3}, \\ \cos \frac{4n\pi}{3} = \cos n\pi \cos \frac{n\pi}{3}, \qquad \cos \frac{7n\pi}{3} = \cos \frac{n\pi}{3},$$

these results take the form

$$A_{0} = 0, \qquad A_{n} = -2 \frac{1 - \cos n\pi}{n\pi} \sin \frac{n\pi}{3} \\ B_{n} = 2 \frac{1 - \cos n\pi}{n\pi} \cos \frac{n\pi}{3} \end{bmatrix}.$$
(156)

Since the factor $1 - \cos n\pi$ is zero when *n* is even and 2 when *n* is odd, it follows that the required series contains only harmonics corresponding to the *odd* integers. The series can be written in the form*

$$f(x) = \frac{4}{\pi} \sum_{n \text{ odd}} \frac{1}{n} \left(\cos \frac{n\pi}{3} \sin \frac{n\pi x}{3} - \sin \frac{n\pi}{3} \cos \frac{n\pi x}{3} \right).$$
(157)

We notice that this series is equivalent to the form

$$f(x) = \frac{4}{\pi} \sum_{n \text{ odd}} \frac{1}{n} \sin \frac{n\pi(x-1)}{3}$$

This form could have been obtained more directly, in this case, by first transforming the variable by moving the origin to the point x = 1.

Any function f(x) of period P which also satisfies the relation

$$f\left(x+\frac{P}{2}\right) = -f(x) \tag{158}$$

is called an *odd-harmonic function*, since it is readily shown that the Fourier series representation of such a function possesses only harmonics corresponding to odd integers. The function in the preceding example is of this type.

5.12. Fourier-Bessel series. Expansions in terms of Bessel functions arise most frequently in connection with a boundary-value problem involving the differential equation

$$x^{2}\frac{d^{2}y}{dx^{2}} + x\frac{dy}{dx} + (\mu^{2}x^{2} - p^{2})y = 0$$
(159)

* The phrase "*n* odd" is used in summations to indicate that *n* is to take on the values $n = 1, 3, 5, \ldots$

or, equivalently,

$$\frac{d}{dx}\left(x\frac{dy}{dx}\right) + \left(-\frac{p^2}{x} + \mu^2 x\right)y = 0.$$
 (160)

Reference to Equation (130a), Chapter 4, shows that the general solution of this equation is of the form $y = Z_p(\mu x)$ or

$$y = \begin{cases} c_1 J_p(\mu x) + c_2 J_{-p}(\mu x) & (p \text{ not an integer}), \\ c_1 J_p(\mu x) + c_2 Y_p(\mu x) & (p \text{ an integer}). \end{cases}$$
(161)

Equation (160) is identified with (90) if we write

$$p(x) = x, \quad q(x) = -\frac{p^2}{x}, \quad r(x) = x, \quad \lambda = \mu^2.$$
 (162)

Restricting attention to the interval (0,L), we see that since p(0) = 0, the results of Section 5.6 show that the characteristic functions of the problem are orthogonal in (0,L), with respect to the weighting function r(x) = x, if at x = 0 it is required merely that y be finite and $x \frac{dy}{dx}$ be zero and if at x = L a homogeneous condition is prescribed.

The conditions of *finiteness* at x = 0 require that we take

$$c_2 = 0 \tag{163}$$

in (161). If at x = L the condition

$$v(L) = 0 \tag{164a}$$

is prescribed, the characteristic values of μ are determined as the roots of the equation

$$J_p(\mu_n L) = 0.$$
 (164b)

Similarly, the end condition

$$y'(L) = 0 \tag{165a}$$

requires that the characteristic numbers μ_n satisfy

$$J'_{p}(\mu_{n}L) = 0,$$
 (165b)

whereas the more general condition

$$k y(L) + y'(L) = 0$$
 $(k \ge 0)$ (166a)

determines μ_n as a solution of the equation

$$k J_{p}(\mu_{n}L) + \mu_{n} J_{p}'(\mu_{n}L) = 0.$$
 (166b)

In all these cases the characteristic functions are of the form

$$\varphi_n(x) = J_p(\mu_n x), \tag{167}$$

where μ_n is a solution of (164b), (165b), or (166b), and the results of Section 5.6 show that these functions are orthogonal in (0,L) with the weighting function r(x) = x,

$$\int_0^L x J_p(\mu_m x) J_p(\mu_n x) \, dx = 0 \qquad (m \neq n). \tag{168}$$

Although Equation (160) is one of the exceptional cases described in Section 5.7 in that p(x) and r(x) vanish at the end point x = 0, and also q(x)is not regular at x = 0 unless the order p of the Bessel functions is zero, still it can be shown that the formal series (107), with coefficients determined by (109), will converge to f(x) inside the interval (0, L) under the same conditions as those stated with reference to the Fourier series.

We must, of course, be careful to include *all* the characteristic functions of the problem involved. From the definition of the Bessel functions $J_p(x)$, it is clear that the relation

$$J_p(-x) = (-1)^p J_p(x)$$

is satisfied. Thus, roots of (164b), (165b), and (166b) exist in pairs, symmetrically located with respect to the point x = 0. However, since replacing μ_n by $-\mu_n$ in (167) either does not change $\varphi_n(x)$ or merely multiplies it by a numerical factor, the negative values of μ_n need not be considered. If $\mu_0 = 0$ were a characteristic number, the corresponding function $\varphi_0(x)$, as given by (167), would be identically zero (and hence not a characteristic function) except in the case when the order p of the relevant Bessel functions is zero. When p = 0, only (165b) permits the solution $\mu_0 = 0$. Thus, it is necessary to consider only the set of functions { $\varphi_n(x)$ } corresponding to positive characteristic numbers $\mu_n(n = 1, 2, 3, ...)$ in all cases except in the case of (165b) when p = 0, in which case the function $\varphi_0 = 1$ corresponding to $\mu_0 = 0$ must be added to the set.

Temporarily excluding this exceptional case, the coefficients in the series

$$f(x) = A_1 J_p(\mu_1 x) + A_2 J_p(\mu_2 x) + \cdots$$

= $\sum_{n=1}^{\infty} A_n J_p(\mu_n x)$ (0 < x < L), (169)

where the numbers μ_n are the positive solutions of one of Equations (164b), (165b), and (166b), are given, by virtue of (109), by the equation

$$A_n \int_0^L x [J_p(\mu_n x)]^2 \, dx = \int_0^L x f(x) \, J_p(\mu_n x) \, dx$$

The coefficient of A_n is independent of f(x) and can be determined once and for all. Denoting it by C_n , we then have

$$A_{n} = \frac{1}{C_{n}} \int_{0}^{L} x f(x) J_{p}(\mu_{n} x) dx, \qquad (170)$$

$$C_n = \int_0^L x [J_p(\mu_n x)]^2 \, dx. \tag{171}$$

ı.

where

sec. 5.12 | Fourier-Bessel series

To determine the factors C_n , we proceed by an indirect method as follows. With the notation of (167) we have

$$\frac{d}{dx}\left(x\frac{d\varphi_n}{dx}\right) + \left(\mu_n^2 x - \frac{p^2}{x}\right)\varphi_n = 0, \qquad (172)$$

since the function $\varphi_n(x)$ satisfies (160) when $\mu = \mu_n$. If both sides of (172) are multiplied by $2x \frac{d\varphi_n}{dx}$, the resulting equation can be written in the form

$$(\mu_n^2 x^2 - p^2) \frac{d}{dx} (\varphi_n^2) = -\frac{d}{dx} \left(x \frac{d\varphi_n}{dx} \right)^2$$

If both sides of this equation are integrated over (0, L), and the resulting lefthand side is simplified by an integration by parts, there follows

$$\left[(\mu_n^2 x^2 - p^2)\varphi_n^2\right]_0^L - 2\mu_n^2 \int_0^L x \varphi_n^2 \, dx = -\left[x^2 \left(\frac{d\varphi_n}{dx}\right)^2\right]_0^L.$$

If we notice that $\varphi_n(0) = J_p(0) = 0$ when p > 0, that $J_0(0) = 1$, and that $x J'_p(x) = 0$ at x = 0, it is readily seen that the integrated terms vanish at the lower limit. Thus we obtain the important relation

$$C_{n} = \int_{0}^{L} x [J_{p}(\mu_{n}x)]^{2} dx$$

= $\frac{1}{2\mu_{n}^{2}} \left\{ (\mu_{n}^{2}L^{2} - p^{2})[\varphi_{n}(L)]^{2} + L^{2} \left[\frac{d\varphi_{n}(L)}{dx} \right]^{2} \right\}$ (173)
 $\varphi_{n}(x) = J_{p}(\mu_{n}x).$ (174)

where

We also have

 $\varphi_n(L) = J_p(\mu_n L)$

and, making use of Equation (111), Section 4.9,

$$\frac{d\varphi_n}{dx} = \frac{d}{dx}J_p(\mu_n x) = -\mu_n J_{p+1}(\mu_n x) + \frac{p}{x}J_p(\mu_n x),$$

so that there follows

$$\frac{d\varphi_n(L)}{dx} = -\mu_n J_{p+1}(\mu_n L) + \frac{p}{L}\varphi_n(L).$$

Thus, if μ_n satisfies (164b), there follows $\varphi_n(L) = 0$ and

$$C_n = \frac{L^2}{2} \left[J_{p+1}(\mu_n L) \right]^2.$$
(175a)

If μ_n satisfies (165b), there follows $\frac{d\varphi_n(L)}{dx} = 0$ and

$$C_n = \frac{\mu_n^2 L^2 - p^2}{2\mu_n^2} \left[J_p(\mu_n L) \right]^2.$$
(175b)

Finally, if
$$\mu_n$$
 satisfies (166b), there follows $\frac{d\varphi_n(L)}{dx} = -k\varphi_n(L)$ and

$$C_n = \frac{\mu_n^2 L^2 - p^2 + k^2 L^2}{2\mu_n^2} \left[J_p(\mu_n L)\right]^2. \quad (175c)$$

In the exceptional case noted above, when expansion in terms of the functions $J_0(\mu_n x)$ is required, where μ_n is a solution of the equation

$$J'_0(\mu_n L) = 0 \quad \text{or} \quad \mu_n J_1(\mu_n L) = 0,$$
 (176)

the presence of the additional characteristic number $\mu_0 = 0$ introduces a constant term A_0 , leading to the series

$$f(x) = A_0 + \sum_{n=1}^{\infty} A_n J_0(\mu_n x), \qquad (177)$$

where

$$A_0 = \frac{2}{L^2} \int_0^L x \ f(x) \ dx \tag{178}$$

and the remaining coefficients are given by Equations (170) and (175b) with p = 0.

Although the explicit evaluation of the integral appearing in (170) is usually not feasible, even when f(x) is of a simple form, the integral can be evaluated approximately for as many values of n as are required, by numerical or graphical methods.

Expansions of the types considered are of frequent use in connection with the solution of certain boundary-value problems involving partial differential equations, the particular type of expansion to be used being determined by the nature of the problem.

As an application, we determine the expansion of the function f(x) = 1in the interval (0,L), in a series of the form

$$1 = A_1 J_0(\mu_1 x) + A_2 J_0(\mu_2 x) + \cdots$$

= $\sum_{n=1}^{\infty} A_n J_0(\mu_n x)$ (0 < x < L), (179)

where μ_n is a solution of

$$J_0(\mu_n L) = 0. (180)$$

In this case Equations (170) and (175a) give

$$A_n = \frac{1}{C_n} \int_0^L x \, J_0(\mu_n x) \, dx, \qquad (181)$$

$$C_n = \frac{L^2}{2} \left[J_1(\mu_n L) \right]^2.$$
 (182)

where

Also, making use of Equation (108), Section 4.9, we have

$$\mu_n x J_0(\mu_n x) = \frac{d}{dx} \left[x J_1(\mu_n x) \right]$$

and hence

$$\int_{0}^{L} x J_{0}(\mu_{n}x) dx = \frac{1}{\mu_{n}} \left[x J_{1}(\mu_{n}x) \right]_{0}^{L} = \frac{L}{\mu_{n}} J_{1}(\mu_{n}L).$$
(183)

With the results of (182) and (183), Equation (181) becomes

$$A_n = \frac{2}{(\mu_n L) J_1(\mu_n L)},$$
 (184)

and the desired series is of the form

$$1 = \frac{2}{\mu_1 L} \frac{J_0(\mu_1 x)}{J_1(\mu_1 L)} + \frac{2}{\mu_2 L} \frac{J_0(\mu_2 x)}{J_1(\mu_2 L)} + \dots = \sum_{n=1}^{\infty} \frac{2}{\mu_n L} \frac{J_0(\mu_n x)}{J_1(\mu_n L)}.$$
 (185)

If we write

$$\mu_n L = \alpha_n, \tag{186}$$

so that α_n is the *n*th zero of $J_0(x)$, the expansion takes the form

$$1 = \frac{2}{\alpha_1} \frac{J_0\left(\alpha_1 \frac{x}{L}\right)}{J_1(\alpha_1)} + \frac{2}{\alpha_2} \frac{J_0\left(\alpha_2 \frac{x}{L}\right)}{J_1(\alpha_2)} + \cdots .$$
(187)

A table of the first 40 zeros α_n and of the corresponding values of $J_1(\alpha_n)$ is given in Jahnke-Emde-Losch, *Tables of Higher Functions*. For the first three zeros, the rounded values

$$egin{aligned} & lpha_1 = 2.4048, & J_1(lpha_1) = 0.5191, \ & lpha_2 = 5.5201, & J_1(lpha_2) = -0.3403, \ & lpha_3 = 8.6537, & J_1(lpha_3) = 0.2715 \end{aligned}$$

are given. Hence the leading terms of (187) become

$$1 = 1.602 J_0 \left(2.405 \frac{x}{L} \right) - 1.065 J_0 \left(5.520 \frac{x}{L} \right) + 0.8512 J_0 \left(8.654 \frac{x}{L} \right) + \cdots$$

Since the Bessel functions involved are *even* functions of x, the series must represent 1 not only in (0,L) but also in the symmetrical interval (-L,L). At the end points $x = \pm L$ the series clearly does not converge to the given function, since all terms in the series vanish.

5.13. Legendre series. Legendre's differential equation,

$$(1-x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + p(p+1)y = 0, \qquad (188)$$

or, equivalently,

$$\frac{d}{dx}\left[(1-x^2)\frac{dy}{dx}\right] + p(p+1) y = 0,$$
 (189)

can be identified with (90) if we write

$$p(x) = 1 - x^2$$
, $q(x) = 0$, $r(x) = 1$, $\lambda = p(p + 1)$. (190)

Since the function p(x) vanishes at the points $x = \pm 1$, the results of Section 5.6 state that any two distinct solutions of (188) which are *finite* and have a finite derivative at $x = \pm 1$ are orthogonal, with respect to the weighting function r(x) = 1, in the interval (-1,1). Since (188) possesses solutions which are finite at $x = \pm 1$ only if p is a positive integer or zero (see Section 4.12), this condition of finiteness determines permissible values of p in the form

$$p = n$$
 $(n = 0, 1, 2, 3, ...).$ (191)

The corresponding solutions which are finite when $x = \pm 1$ are proportional to the Legendre polynomials

$$\varphi_n = P_n(x), \tag{192}$$

and the above reasoning leads to the conclusion

$$\int_{-1}^{1} P_m(x) P_n(x) \, dx = 0 \qquad (m \neq n). \tag{193}$$

Just as in the preceding cases, a function f(x) which is piecewise differentiable in the interval (-1,1) can be represented by a series of the functions φ_n ,

$$f(x) = A_0 P_0(x) + A_1 P_1(x) + A_2 P_2(x) + \cdots$$

= $\sum_{n=0}^{\infty} A_n P_n(x)$ (-1 < x < 1). (194)

The coefficients A_n are given by the equation

$$A_n \int_{-1}^{1} \left[P_n(x) \right]^2 dx = \int_{-1}^{1} f(x) P_n(x) dx.$$
 (195)

In order to evaluate the integrals appearing in (195), it is frequently useful to express $P_n(x)$ by *Rodrigues' formula* [Equation (168), Section 4.12],

$$P_n(x) = \frac{1}{2^n n!} \frac{d^n}{dx^n} (x^2 - 1)^n.$$
(196)

Hence we may write

$$\int_{-1}^{1} f(x) P_n(x) dx = \frac{1}{2^n n!} \int_{-1}^{1} f(x) \frac{d^n}{dx^n} (x^2 - 1)^n dx.$$
(197)

If the right-hand member of (197) is integrated by parts N times, where $N \leq n$, assuming that the first N derivatives of f(x) are continuous in (-1,1),* and if it

* Otherwise, integration by parts may not be valid.

is noticed that the first n - 1 derivatives of $(x^2 - 1)^n$ vanish when $x = \pm 1$, Equation (197) becomes

$$\int_{-1}^{1} f(x) P_n(x) dx = \frac{(-1)^N}{2^n n!} \int_{-1}^{1} \left[\frac{d^N}{dx^N} f(x) \right] \left[\frac{d^{n-N}}{dx^{n-N}} (x^2 - 1)^n \right] dx$$

$$(N \le n). \quad (198)$$

In particular, when N = n there follows

$$\int_{-1}^{1} f(x) P_n(x) dx = \frac{(-1)^n}{2^n n!} \int_{-1}^{1} (x^2 - 1)^n \frac{d^n f(x)}{dx^n} dx$$
(199)

if f(x) and its first n derivatives are continuous throughout the interval (-1,1). Replacing f(x) by $P_n(x)$ in (199) and noticing that, from (196),

$$\frac{d^{n}P_{n}(x)}{dx^{n}} = \frac{1}{2^{n}n!} \frac{d^{2n}}{dx^{2n}} (x^{2} - 1)^{n}$$
$$= \frac{1}{2^{n}n!} \frac{d^{2n}}{dx^{2n}} (x^{2n} - nx^{2n-2} + \cdots) = \frac{(2n)!}{2^{n}n!}, \qquad (200)$$

we obtain, from (199),

$$\int_{-1}^{1} \left[P_n(x) \right]^2 dx = \frac{(2n)!}{2^{2n} (n!)^2} \int_{-1}^{1} (1 - x^2)^n \, dx. \tag{201}$$

The integral on the right can be evaluated by successive reductions (involving integration by parts) in the form

$$\int_{-1}^{1} (1-x^2)^n dx = 2 \frac{(2n)(2n-2)\cdots 4\cdot 2}{(2n+1)(2n-1)\cdots 5\cdot 3} = \frac{2^{2n+1}(n!)^2}{(2n+1)!}.$$
 (202)

Thus Equation (201) takes the form

$$\int_{-1}^{1} \left[P_n(x) \right]^2 dx = \frac{2}{2n+1}$$
(203)

and Equations (195) and (199) give for the coefficients in the expansion

$$f(x) = A_0 P_0(x) + A_1 P_1(x) + \cdots$$

= $\sum_{n=0}^{\infty} A_n P_n(x)$ (-1 < x < 1) (204a)

the alternative forms

$$A_{n} = \begin{cases} \frac{2n+1}{2} \int_{-1}^{1} f(x) P_{n}(x) dx \\ \frac{2n+1}{2^{n+1}n!} \int_{-1}^{1} (1-x^{2})^{n} \frac{d^{n}f(x)}{dx^{n}} dx, \end{cases}$$
(204b)

where, again, the second form in (204b) is necessarily correct only if f(x) and its first n derivatives are continuous in (-1,1).

Since $P_n(x)$ is an even function of x when n is even, and an odd function when n is odd, it follows that if f(x) is an even function of x, the coefficients A_n

will vanish when n is odd; whereas if f(x) is an odd function of x, the coefficients A_n will vanish when n is even.

Thus, for an even function f(x), there follows

$$A_{n} = \begin{cases} 0 & (n \text{ odd}) \\ (2n+1) \int_{0}^{1} f(x) P_{n}(x) dx & (n \text{ even}); \end{cases}$$
(205)

whereas for an *odd* function f(x) we have

$$A_{n} = \begin{cases} (2n+1) \int_{0}^{1} f(x) P_{n}(x) dx & (n \text{ odd}) \\ 0 & (n \text{ even}). \end{cases}$$
(206)

If we are interested in representing a given function f(x) merely over the interval (0,1), it is then clear that a series containing only *even* polynomials can be obtained by using (205), or, if preferred, a series containing only *odd* polynomials can be obtained by using (206). Either of these series will then represent f(x) inside the interval (0,1), and the first series will represent f(-x) and the second -f(-x) in the interval (-1,0).

Finally, we remark that expansions valid in the more general interval (-L,L) are readily obtained by replacing x by x/L in the preceding developments. This procedure leads to the expansion

$$f(x) = A_0 P_0 \left(\frac{x}{L}\right) + A_1 P_1 \left(\frac{x}{L}\right) + \cdots$$
$$= \sum_{n=0}^{\infty} A_n P_n \left(\frac{x}{L}\right) \qquad (-L < x < L), \qquad (207)$$

$$A_{n} = \begin{cases} \frac{2n+1}{2L} \int_{-L}^{L} f(x) P_{n}\left(\frac{x}{L}\right) dx \\ \frac{2n+1}{2^{n+1}n! L^{n+1}} \int_{-L}^{L} (L^{2}-x^{2})^{n} \frac{d^{n}f(x)}{dx^{n}} dx, \end{cases}$$
(208)

where

which generalizes (204).

It is clear that if f(x) is a polynomial of degree k, all derivatives of f(x) of order n vanish identically when n > k. Hence it follows from (204) or (208) that any polynomial of degree k can be expressed as a linear combination of the first k + 1 Legendre polynomials.

As an application, we determine A_0 , A_1 , and A_2 so that

$$x^2 = A_0 P_0(x) + A_1 P_1(x) + A_2 P_2(x),$$

the higher coefficients vanishing by virtue of the above remark. Since x^2 is an even function of x, the remaining odd coefficient A_1 must also vanish:

$$A_1 = 0.$$

To calculate A_0 and A_2 , we use the second form of (204b) and obtain

$$A_0 = \frac{1}{2} \int_{-1}^{1} 1 \cdot x^2 \, dx = \frac{1}{3}, \qquad A_2 = \frac{5}{8 \cdot 2} \int_{-1}^{1} (1 - x^2)^2 \cdot 2 \, dx = \frac{2}{3},$$

and hence

$$x^{2} = \frac{1}{3}[P_{0}(x) + 2P_{2}(x)].$$
(209)

The truth of this identity is verified by reference to the expressions for $P_0(x)$ and $P_2(x)$ in Section 4.12 [Equation (158)].

As a second application, we determine the Legendre expansion in (-1,1) of a function f(x) which vanishes when -1 < x < 0 and which takes on the value unity when 0 < x < 1,

$$f(\mathbf{x}) = \begin{cases} 0 & (-1 < \mathbf{x} < 0) \\ 1 & (0 < \mathbf{x} < 1). \end{cases}$$
(210)

Since f(x) is discontinuous in (-1,1), the second expression in (204b) cannot be used. The basic relation becomes, in this case,

$$A_n = \frac{2n+1}{2} \int_0^1 P_n(x) \, dx.$$

Hence, referring to Equation (158), Section 4.12, there follows

$$A_{0} = \frac{1}{2} \int_{0}^{1} 1 \, dx = \frac{1}{2},$$

$$A_{1} = \frac{3}{2} \int_{0}^{1} x \, dx = \frac{3}{4},$$

$$A_{2} = \frac{5}{2} \cdot \frac{1}{2} \int_{0}^{1} (3x^{2} - 1) \, dx = 0,$$

$$A_{3} = \frac{7}{2} \cdot \frac{1}{2} \int_{0}^{1} (5x^{3} - 3x) \, dx = -\frac{7}{16},$$

and so forth. Since f(x) can be considered as the sum of the constant $\frac{1}{2}$ and an odd function of x, it is seen that A_0 is the only nonvanishing *even* coefficient. The first terms of the desired series are of the form

$$f(x) = \frac{1}{2}P_0(x) + \frac{3}{4}P_1(x) - \frac{7}{16}P_3(x) + \frac{11}{32}P_5(x) + \cdots$$

$$(-1 < x < 1). \quad (211)$$

In general, we find

$$\int_{0}^{1} P_{n}(x) dx = \begin{cases} 1 & (n = 0) \\ 0 & (n = 2, 4, 6, \ldots) \\ (-1)^{\frac{n-1}{2}} \frac{1}{n(n+1)} \frac{(1 \cdot 3 \cdot 5 \cdots n)^{2}}{n!} & (n = 1, 3, 5, \ldots). \end{cases}$$
(212)

For since (189) gives

$$\left[\frac{d}{dx}(1-x^2)\frac{dP_n}{dx}\right]+n(n+1)P_n=0,$$

we obtain

$$n(n+1)\int_0^1 P_n\,dx = -\left[(1-x^2)\frac{dP_n}{dx}\right]_0^1 = P'_n(0).$$

If n is even, we have $P'_n(0) = 0$, whereas if n is odd, Equations (156), (157b), and (159b) of Section 4.12 give

$$P'_{n}(0) = (-1)^{\frac{n-1}{2}} \frac{(1 \cdot 3 \cdot 5 \cdots n)^{2}}{n!}$$

Thus, the expansion (211) can be written in the form

$$f(x) = \frac{1}{2} + \sum_{n \text{ odd}} (-1)^{\frac{n-1}{2}} \frac{2n+1}{2n+2} \frac{(1 \cdot 3 \cdot 5 \cdots n)^2}{n \cdot n!} P_n(x)$$

$$(-1 < x < 1).$$
(213)

5.14. The Fourier integral. We have seen in Section 5.10 that the functions

$$\varphi_n(x) = \sin \frac{n\pi}{L} x \qquad (n = 1, 2, 3, \ldots)$$

form an orthogonal set in the interval 0 < x < L and that a function f(x) can, in general, be represented as a series of these functions in the stated interval. The question arises as to the possibility of a similar representation in the complete *semi-infinite* interval x > 0. However, since the functions $\varphi_n(x)$ vanish when $L \to \infty$, it is clear that such a representation cannot be obtained by merely replacing 1/L by zero in the series (132).

If we consider the relation

$$\frac{2}{L}\int_0^L \sin u_1 x \sin u_2 x \, dx = \frac{\sin (u_2 - u_1)L}{(u_2 - u_1)L} - \frac{\sin (u_2 + u_1)L}{(u_2 + u_1)L}, \quad (214)$$

we notice that the expression on the right vanishes if u_1 and u_2 are distinct integral multiples of π/L and is equal to unity (in the limit) if u_1 and u_2 both take on the same integral multiple of π/L . This is, of course, in accordance with the results of Section 5.10. In addition, however, we notice that as $L \to \infty$ the right-hand member of (214) approaches zero for any positive values of u_1 and u_2 so long as $u_2 \neq u_1$, but that if $u_2 = u_1$, the right-hand member approaches unity. That is, if we write

$$\varphi_u(x) = \sin ux, \tag{215}$$

we obtain

$$\lim_{L \to \infty} \frac{2}{L} \int_0^L \varphi_{u_1}(x) \, \varphi_{u_2}(x) \, dx = \begin{cases} 0 & (u_1 \neq u_2) \\ 1 & (u_1 = u_2). \end{cases}$$
(216)

Thus two functions of form (215), corresponding to any two positive values of u, have a sort of orthogonality property in the semi-infinite interval $(0,\infty)$, and we are led to the conclusion that a representation of a function f(x) in a semi-infinite interval must involve all possible functions of type (215), where uis any positive number and is not restricted, as in the finite case, to a set of discrete values.

In the finite case f(x) is expressed as a linear combination of functions of the latter type, that is, as an *infinite series*,

$$f(x) = \sum_{n=1}^{\infty} A_n \sin \frac{n\pi x}{L} \qquad (0 < x < L),$$

where the constants A_n specify the contributions of the successive harmonics in the series. In the limiting case $L \to \infty$, if we write the contribution of sin uxto the representation of f(x) in the form $A(u) \sin ux$, the complete representation of f(x) should be obtained by superimposing these contributions by means of an *infinite integral*,

$$f(x) = \int_0^\infty A(u) \sin ux \, du \qquad (0 < x < \infty). \tag{217}$$

Assuming that such a representation exists, for a sufficiently well-behaved function f(x), it remains to determine A(u) for all positive values of u, that is, as a function of u. In the following developments we indicate a formal method of determining A(u), and afterward state conditions under which the result obtained is correct.

In determining the function A(u), we are guided by the analogous procedure of Section 5.10. Thus, we first multiply both sides of (217) by $\sin u_0 x$, where u_0 is any positive value of u,

$$f(x)\sin u_0 x = \sin u_0 x \int_0^\infty A(u) \sin ux \, du.$$

We next integrate both sides of the result with respect to x over the interval (0,L), where L is destined to become infinite, and so obtain

$$\int_0^L f(x) \sin u_0 x \, dx = \int_0^L \sin u_0 x \left\{ \int_0^\infty A(u) \sin u x \, du \right\} dx$$

or, assuming that the order of integration may be interchanged,

$$\int_{0}^{L} f(x) \sin u_{0}x \, dx = \int_{0}^{\infty} A(u) \left\{ \int_{0}^{L} \sin ux \sin u_{0}x \, dx \right\} du.$$
(218)

We denote the right-hand member of (218) by R_L ; then, making use of (214), we obtain

$$R_{L} = \frac{1}{2} \int_{0}^{\infty} \frac{A(u)}{u - u_{0}} \sin L(u - u_{0}) du - \frac{1}{2} \int_{0}^{\infty} \frac{A(u)}{u + u_{0}} \sin L(u + u_{0}) du. \quad (219)$$

With the substitution $L(u - u_0) = t$ in the first integral of (219) and the substitution $L(u + u_0) = t$ in the second integral, this expression takes the form

$$R_{L} = \frac{1}{2} \int_{-Lu_{0}}^{\infty} A\left(u_{0} + \frac{t}{L}\right) \frac{\sin t}{t} dt - \frac{1}{2} \int_{Lu_{0}}^{\infty} A\left(-u_{0} + \frac{t}{L}\right) \frac{\sin t}{t} dt. \quad (220)$$

Now, if $u_0 > 0$, we see that as $L \to \infty$ the second integral *formally* approaches zero, whereas the first integral takes the value

$$\frac{1}{2}\int_{-\infty}^{\infty}A(u_0)\frac{\sin t}{t}\,dt=\frac{A(u_0)}{2}\int_{-\infty}^{\infty}\frac{\sin t}{t}\,dt=\frac{\pi}{2}A(u_0).$$

Thus, as $L \rightarrow \infty$, Equation (218) gives formally

$$\int_0^\infty f(x) \sin u_0 x \, dx = \lim_{L \to \infty} R_L = \frac{\pi}{2} A(u_0) \qquad (u_0 > 0). \tag{221}$$

Replacing u_0 by u in (221), we have determined the function A(u) required in (217) in the form

$$A(u) = \frac{2}{\pi} \int_0^\infty f(x) \sin ux \, dx.$$
 (222)

To distinguish the dummy variable x in (222) from the current variable x in (217), we replace x by t in (222) before introducing (222) into (217), and so obtain

$$f(x) = \frac{2}{\pi} \int_0^\infty \sin ux \int_0^\infty f(t) \sin ut \, dt \, du \qquad (0 < x < \infty). \tag{223}$$

This expression is known as the Fourier sine integral representation of f(x) and can be shown, by more rigorous analysis, to be valid when x > 0 if f(x) is piecewise differentiable in every positive interval and if the integral $\int_0^\infty |f(x)| dx$ exists. As in the Fourier series theory, the integral represents f(x) at points of continuity, and represents $\frac{1}{2}[f(x+) + f(x-)]$ at finite jumps.

In a similar way, the Fourier cosine integral representation

$$f(x) = \frac{2}{\pi} \int_0^\infty \cos ux \int_0^\infty f(t) \cos ut \, dt \, du \qquad (0 < x < \infty) \quad (224)$$

can be obtained, subject to the same restrictions.

It is clear that (223) represents -f(-x) when x < 0, whereas (224) represents f(-x) when x < 0. Thus, if f(x) is an odd function, the representation in (223) includes all values of x; whereas if f(x) is an even function, the same is true of (224). As in the finite case, we can make use of these facts to obtain a representation valid for all values of x by using both sine and cosine harmonics.

* See Problem 12(b), Chapter 2.

Thus, if we write $f(x) = f_e(x) + f_o(x)$, where f_e is even and f_o is odd, we verify that

$$\frac{1}{\pi} \int_0^\infty \cos ux \int_{-\infty}^\infty f(t) \cos ut \, dt \, du = \frac{1}{\pi} \int_0^\infty \cos ux \int_{-\infty}^\infty f_e(t) \cos ut \, dt \, du$$
$$= \frac{2}{\pi} \int_0^\infty \cos ux \int_0^\infty f_e(t) \cos ut \, dt \, du$$
$$= f_e(x) \qquad (-\infty < x < \infty), \qquad (225a)$$

and, similarly,

$$\frac{1}{\pi}\int_0^\infty \sin ux \int_{-\infty}^\infty f(t) \sin ut \, dt \, du = f_o(x) \qquad (-\infty < x < \infty).$$
 (225b)

Hence, adding (225a) and (225b), we obtain the representation

$$f(x) = \int_0^\infty \left[A(u) \cos ux + B(u) \sin ux \right] du \qquad (-\infty < x < \infty), \qquad (226)$$
where $A(u)$ and $B(u)$ are defined by the expressions

where A(u) and B(u) are defined by the expressions

$$A(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \cos ut \, dt, \qquad B(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(t) \sin ut \, dt. \quad (227)$$

By introducing (227) into (226), we obtain also

$$f(x) = \frac{1}{\pi} \int_0^\infty \left[\int_{-\infty}^\infty f(t) \cos u(t-x) dt \right] du \qquad (-\infty < x < \infty), \qquad (228)$$

or, equivalently,

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} f(t) \cos u(t-x) dt du \qquad (-\infty < x < \infty).$$
 (228a)

This expression is known as the complete Fourier integral representation of f(x) and represents f(x) for all values of x in the usual sense if f(x) is piecewise differentiable in every finite interval and if the integral $\int_{-\infty}^{\infty} |f(x)| dx$ exists.

By writing

$$\cos u(t-x) = \frac{1}{2}e^{iu(t-x)} + \frac{1}{2}e^{-iu(t-x)},$$

we can show that (228) also can be written in the alternative complex form

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} f(t) e^{-iu(t-x)} dt du \qquad (-\infty < x < \infty), \quad (229)$$

or, equivalently,

$$f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{iux} \int_{-\infty}^{\infty} e^{-iut} f(t) dt du \qquad (-\infty < x < \infty). \quad (229a)$$

Finally, if we write

$$u=2\pi\omega$$
,

Equation (229a) becomes

$$f(x) = \int_{-\infty}^{\infty} e^{2\pi i \, \omega x} \int_{-\infty}^{\infty} e^{-2\pi i \, \omega t} f(t) \, dt \, d\omega \qquad (-\infty < x < \infty). \tag{230}$$

Equations (226), (228), (229), and (230) are equivalent forms of the Fourier integral.

If we denote the inner integral in (229a) by $\mathscr{F}(u)$, there follows

$$\mathscr{F}(u) = \int_{-\infty}^{\infty} e^{-iut} f(t) dt \qquad (-\infty < u < \infty), \tag{231a}$$

$$f(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{iut} \mathscr{F}(u) \, du \qquad (-\infty < t < \infty). \tag{231b}$$

The function $\mathscr{F}(u)$ defined by (231a) is called the *Fourier transform* of f. For a function f(t) which vanishes when t < 0, the Fourier transform becomes formally identical with the Laplace transform (Chapter 2) if we replace *iu* by s. If the Fourier transform of f is known, Equation (231b) permits the determination of the function f in terms of an integral involving the transform \mathscr{F} .

In a similar way we obtain from (223) and (224) the inversion formulas

$$\mathscr{S}(u) = \int_0^\infty f(t) \sin ut \, dt \qquad (0 < u < \infty), \tag{232a}$$

$$f(t) = \frac{2}{\pi} \int_0^\infty \mathscr{S}(u) \sin ut \, du \qquad (0 < t < \infty)$$
(232b)

and

$$C(u) = \int_0^\infty f(t) \cos ut \, dt \qquad (0 < u < \infty), \tag{233a}$$

$$f(t) = \frac{2}{\pi} \int_0^\infty C(u) \cos ut \, du \qquad (0 < t < \infty).$$
 (233b)

The functions defined by (232a) and (233a) are called the Fourier sine and cosine transforms of f, respectively.

To illustrate these definitions, we consider the function

$$f(x) = \begin{cases} 0 & (x < 0) \\ 1 & (0 < x < a) \\ 0 & (x > a). \end{cases}$$
(234)

From (223) we obtain the Fourier sine integral representation,

$$f(x) = \frac{2}{\pi} \int_0^\infty \frac{1 - \cos au}{u} \sin ux \, du \qquad (0 < x < \infty); \qquad (235a)$$

Equation (224) gives the Fourier cosine integral representation,

$$f(x) = \frac{2}{\pi} \int_0^\infty \frac{\sin au \cos ux}{u} du \qquad (0 < x < \infty); \qquad (235b)$$

and Equations (226), (229a), and (230) give the alternative complete Fourier integral representations,

$$f(x) = \frac{1}{\pi} \int_0^\infty \left(\frac{\sin au}{u} \cos xu + \frac{1 - \cos au}{u} \sin xu \right) du$$

or
$$(-\infty < x < \infty) \qquad (236a)$$

OI

$$f(x) = \begin{cases} \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{1 - e^{-iau}}{u} e^{iux} du \\ \frac{1}{2\pi i} \int_{-\infty}^{\infty} \frac{1 - e^{-2\pi ia\omega}}{\omega} e^{2\pi ix\omega} d\omega \end{cases} \qquad (-\infty < x < \infty). \quad (236b,c)$$

When x < 0, the integrals appearing in (235a) and (235b) represent -f(-x) and f(-x), respectively. Thus, for example, taking into account the discontinuities in f(x), we have from (235b) the useful result

$$\int_{0}^{\infty} \frac{\sin au \cos ux}{u} du = \begin{cases} \frac{\pi}{2} & (|x| < a) \\ \frac{\pi}{4} & (|x| = a) \\ 0 & (|x| > a). \end{cases}$$
(237)

Also, from (231a), (232a), and (233a) we obtain the transforms of f(x),

$$\mathscr{F}(u) = \frac{1 - e^{-iau}}{iu}, \qquad (238a)$$

$$\mathscr{S}(u) = \frac{1 - \cos au}{u} \qquad (u > 0), \tag{238b}$$

$$C(u) = \frac{\sin au}{u} \qquad (u > 0). \tag{238c}$$

We notice that in this case there follows $\mathscr{F}(u) = C(u) - i\mathscr{G}(u)$. This relationship between the transforms is readily shown to be true if and only if the function f(x) vanishes when x < 0.

Analogous integral representations involving Bessel functions can be obtained. In particular, it can be shown* that the equation

$$f(x) = \int_0^\infty \int_0^\infty ut \, f(t) \, J_p(ut) \, J_p(ux) \, dt \, du \qquad (0 < x < \infty)$$
(239)

* See A. Gray, G. B. Mathews, and T. M. MacRobert, A Treatise on Bessel Functions, St. Martin's Press, New York, 1952.
is valid when p > -1, with the usual convention regarding discontinuities, if f(x) is piecewise differentiable in every finite interval and if the integral $\int_0^\infty x |f(x)| dx$ exists. This expression is known as the Fourier-Bessel integral representation of f(x), of order p.

The Fourier-Bessel transform, of order p, of a function f(x) may be defined by the relation

$$\mathscr{B}_{p}(u) = \int_{0}^{\infty} t f(t) J_{p}(ut) dt \qquad (0 < u < \infty).$$
 (240a)

The function f(x) is then expressible in terms of its transform by the equation

$$f(x) = \int_0^\infty u \, \mathscr{B}_p(u) \, J_p(ux) \, du \qquad (0 < x < \infty). \tag{240b}$$

REFERENCES

- 1. Carslaw, H. S., Introduction to the Theory of Fourier's Series and Integrals, 3rd Ed. Dover Publications, Inc., New York, 1952.
- 2. Churchill, R. V., Fourier Series and Boundary Value Problems, McGraw-Hill Book Company, Inc., New York, 1941.
- 3. Jackson, D., Fourier Series and Orthogonal Polynomials, Sixth Carus Mathematical Monograph, Open Court Publishing Co., La Salle, Ill., 1941.
- 4. Sneddon, I. N., Fourier Transforms, McGraw-Hill Book Company, Inc., New York, 1951.
- 5. Szego, G., Orthogonal Polynomials, Am. Math. Soc. Coll. Pub., Vol. 23, New York, 1939.
- 6. Titchmarsh, E. C., Eigenfunction Expansions Associated with Second-Order Differential Equations, Oxford University Press, Fair Lawn, N.J., 1946.
- 7. Whittaker, E. T., and G. N. Watson, *Modern Analysis*, Cambridge University Press, New York, 1958.

PROBLEMS

Section 5.1

1. Show that the boundary-value problem

$$\frac{d^2y}{dx^2} - \lambda^2 y = 0, \qquad y(0) = y(L) = 0$$

cannot have a nontrivial solution for real values of λ .

2. Determine those values of k for which the partial differential equation

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} = 0$$

possesses nontrivial solutions of the form $T(x,y) = f(x) \sinh ky$ which vanish when x = 0 and when x = L.

3. Show that the equation

$$\frac{d^2y}{dx^2} + \lambda x^2 y = 0$$

can possess a nontrivial solution which vanishes at x = 0 and at x = 1 only if λ is such that

$$J_{1/4}(\frac{1}{2}\lambda^{1/2}) = 0$$

and that, corresponding to such a characteristic number λ_k , any multiple of the function

$$\varphi_k(x) = x^{1/2} J_{1/4}(\frac{1}{2}\lambda_k^{1/2} x^2)$$

is a solution with the required properties.

4. Show that the equation

$$x\frac{d^2y}{dx^2}+\frac{dy}{dx}+\lambda xy=0$$

can possess a nontrivial solution which is *finite* at x = 0 and which vanishes when x = a only if λ is such that $J_0(\lambda^{1/2} a) = 0$, and obtain the form of the solution in this case.

Section 5.2

5. Suppose that the uniform string considered in Section 5.2 is unrestrained from transverse motion at the end x = 0, and is attached at the end x = L to a yielding support of modulus k, the ends of the string being constrained against appreciable movement parallel to the axis of rotation. Show that the *n*th critical speed ω_n is given by

$$\omega_n = \frac{\mu_n}{L} \sqrt{\frac{T}{\rho}},$$

where $\cot \mu_n = \alpha \mu_n$ and $\alpha = T/kL$.

6. The linear density of a flexible string of length L varies according to the law $\rho = \rho_0(1 + x/L)^2$, where ρ_0 is a constant and x is measured from one end. The ends of the string are attached to an axis rotating with angular velocity ω .

(a) Show that the governing differential equation can be written in the form

$$\frac{d^2y}{dr^2}+4\mu^2r^2y=0,$$

where r = 1 + x/L and $\mu^2 = (\rho_0 \omega^2 L^2)/(4T)$, and that the end conditions then require that y vanish when r = 1 and when r = 2.

(b) Show that the *n*th critical speed ω_n is given by

$$\omega_n=2\sqrt{\frac{\overline{T}}{\rho_0}}\frac{\mu_n}{L},$$

where μ_n is the *n*th solution of the equation

$$J_{1/4}(\mu) Y_{1/4}(4\mu) - J_{1/4}(4\mu) Y_{1/4}(\mu) = 0.$$

7. (a) If a flexible string is caused to vibrate transversely in a plane with respect to a straight form along the x axis, the displacement w of a point is a function of x and of time t. Noticing that the inertia force per unit length is then given by $-\rho \frac{\partial^2 w}{\partial t^2}$, and denoting the actual transverse external force by f(x,t), show that w satisfies the partial differential equation

$$\frac{\partial}{\partial x}\left(T\frac{\partial w}{\partial x}\right) - \rho \frac{\partial^2 w}{\partial t^2} + f = 0$$

when small slopes and displacements are considered.

(b) In the case of free periodic vibrations of circular frequency ω , with f = 0, we must have

$$w(x,t) = y(x)\sin(\omega t + \varphi),$$

where φ is constant and the amplitude y depends only on x. Show that y(x) then satisfies the same ordinary differential equation as the displacement of a string rotating about the x axis with angular velocity ω (Section 5.2), and hence that the results of Section 5.2 can be interpreted in terms of natural vibration modes and natural frequencies in the analysis of periodic vibrations.

Section 5.3

8. Determine expressions for the critical speeds and deflection modes of a rotating uniform shaft of length L fixed at both ends, taking the origin *at one end* of the shaft.

9. Solve the preceding problem, taking the origin at the center of the shaft. (Express the general solution in terms of circular and hyperbolic functions and, by suitably combining the four equations determining the constants of integration, obtain an equivalent set consisting of two pairs of equations each involving coefficients of either only even or only odd functions. Notice that symmetrical and antisymmetrical modes are thus readily distinguished.)

10. Proceeding as in Problem 7, show that the displacement w(x,t) of a shaft or beam executing small transverse vibrations in a plane satisfies the equation

$$\frac{\partial^2}{\partial x^2}\left(EI\frac{\partial^2 w}{\partial x^2}\right) + \rho \frac{\partial^2 w}{\partial t^2} - f = 0.$$

Show also that in the case of free periodic vibration the amplitude function satisfies Equation (27).

Problems

11. A uniform beam is hinged at the end x = 0, whereas the end x = L is free. Show that the natural frequencies of free transverse vibrations of the beam are given by $\omega_n = \mu_n^2 L^2 \sqrt{\rho/EI}$, where μ_n is a root of the equation $\tanh \mu = \tan \mu$, and the deflection in a natural mode is given by

$$w_n(x,t) = C_n \left[\frac{\sinh \mu_n \frac{x}{L}}{\cosh \mu_n} + \frac{\sin \mu_n \frac{x}{L}}{\cos \mu_n} \right] \sin \omega_n t.$$

(See Problem 10.) Notice that the problem also admits the solution w = Cx when $\omega = 0$, corresponding to a rigid-body rotation about the end x = 0.

12. Suppose that a mass $k_{\rho}L$, equal to k times the mass of the entire beam, is attached to the free end of the beam of Problem 11.

(a) Verify that the condition of vanishing shear at x = L then is replaced by the requirement $L^3y''(L) + k\mu^4 y(L) = 0$, where $\mu^4 = \rho \omega^2 L^4 / EI$, and where y(x)is the amplitude function.

(b) Show that μ then must satisfy the equation

$$tanh \mu = tan \mu + 2k\mu tan \mu tanh \mu$$

or

$$\cot \mu = \coth \mu + 2k\mu.$$

(Notice that this condition reduces to that obtained in Problem 11 when k = 0.)

Section 5.4

13. By integrating Equation (46) twice and determining the two constants of integration, show that the differential equation for buckling of a long column *hinged at both ends* can be taken in the form

$$EI\frac{d^2y}{dx^2}+Py=0,$$

with the two end conditions y(0) = y(L) = 0, if no transverse loads are acting.

14. An axial load P is applied to a column of circular cross section with linear taper, so that $I(x) = I_0(x/b)^4$, where x is measured from the point at which the column would taper to a point if it were extended, and I_0 is the value of I at the end x = b.

(a) If the column is hinged at the ends x = a and x = b, show that the governing differential equation can be put in the form

$$x^4\frac{d^2y}{dx^2}+\mu^2 y=0,$$

where $\mu^2 = Pb^4/EI_0$, and where the conditions y(a) = y(b) = 0 are to be satisfied. (b) Show that the general solution of this equation can be expressed first in

(b) Show that the general solution of this equation can be expressed first in terms of Bessel functions and finally in the elementary form

$$y = x \left(c_1 \cos \frac{\mu}{x} + c_2 \sin \frac{\mu}{x} \right).$$

(c) Show that the critical loads are of the form

$$P_n = n^2 \pi^2 \left(\frac{a}{b}\right)^2 \frac{EI_0}{L^2},$$

where L = b - a is the length of the column, and that the buckling modes are given by

$$\varphi_n(x) = x \sin \left[n \pi \frac{b}{L} \left(1 - \frac{a}{x} \right) \right].$$

15. An originally vertical column of length L is assumed to bend under its own weight. If its mass per unit length is denoted by $\rho(x)$, where x is measured from the top of the column downward, show that the shearing force S at distance x from the top is given by the component of the load $w(x) = g \int_0^x \rho dx$ in the direction normal to the deflected axis of the column, and hence is of the form $S = -w \tan \theta$, where θ is the angle of deflection from the vertical. Noticing that the bending moment is given by $M = EI \frac{d\theta}{dx}$, and that $S = \frac{dM}{dx}$, obtain the governing differential equation in the form

$$\frac{d}{dx}\left(EI\frac{d\theta}{dx}\right) + w(x)\theta = 0,$$

when small deflections and slopes are assumed, where, if the end x = 0 is unrestrained whereas the end x = L is built in, the end conditions $EI\theta'(0) = 0$ and $\theta(L) = 0$ are to be satisfied.

16. In the case of a *uniform* column, for which EI = constant and $w' = g\delta A = \text{constant}$, where δ is the volume mass density and A the cross-sectional area, show that the differential equation of Problem 15 takes the form

$$\frac{d^2\theta}{dx^2}+\frac{g\delta A}{EI}x\ \theta=0,$$

and that the solution of this equation for which $\theta'(0) = 0$ is of the form

$$\theta(x) = C x^{1/2} J_{-1/3} \left(\frac{2}{3} \sqrt{\frac{g \delta A}{EI}} x^{3/2}\right),$$

where C is an arbitrary constant. By imposing the condition $\theta(L) = 0$, and making use of the fact that the smallest zero of $J_{-1/3}(z)$ is $z \doteq 1.866$, deduce that the smallest value of L for which buckling may occur (the "critical length" of the column) is given by

$$L_{\rm cr} \doteq 1.986 \left(\frac{EI}{g\delta A}\right)^{1/3}$$

Section 5.5

17. Determine an approximation to the Euler load P_1 of a column of length L, hinged at both ends, with bending stiffness

$$EI(x)=C\frac{L}{L+x},$$

where C is a constant, by the method of Stodola and Vianello. Take Equation (82) as a first approximation to the deflection curve and compare the results of using Equations (70) and (72a) or (72b) in calculating $\lambda_1^{(1)} = L^2 P_1^{(1)}/C$.

18. Find approximately the smallest characteristic value of λ for the problem

$$\frac{d^2y}{dx^2} + \lambda y = 0, \qquad y'(0) = y(1) = 0$$

by using the method of Vianello and Stodola as follows:

(a) Determine a first approximation, by first replacing the term λy by a convenient constant in the given differential equation, in the form $y_1 = 1 - x^2$.

(b) Show that there follows $f_1 = \frac{1}{12}(5 - 6x^2 + x^4)$.

(c) Show that the use of (70) gives $\lambda_1 \approx 2.5$, whereas the use of (72a) gives $\lambda_1 \approx \frac{42}{17} \doteq 2.471$.

(d) Show that the true value of λ_1 is $\pi^2/4 \doteq 2.467$.

19. Find approximately the smallest value of μ for which the problem

$$x \frac{d^2y}{dx^2} - \frac{dy}{dx} + \mu^2 xy = 0, \qquad y(0) = y(1) = 0$$

possesses a nontrivial solution, by using the method of Stodola and Vianello as follows:

(a) Show that the differential equation can be written in the form

$$\frac{d}{dx}\left(\frac{1}{x}\frac{dy}{dx}\right) + \frac{\mu^2}{x}y = 0.$$

(b) Determine the first approximation $y_1(x)$ in such a way that

$$\frac{d}{dx}\left(\frac{1}{x}\frac{dy_1}{dx}\right) = \text{constant},$$

and also $y_1(0) = y_1(1) = 0$, and show that there follows $y_1 = x^2(1 - x)$, with a convenient choice of the constant.

(c) Show that then $f_1(x) = \frac{1}{120}(7x^2 - 15x^4 + 8x^5)$.

(d) Verify that the use of Equation (70), with $\lambda = \mu^2$, gives

$$\mu_1 \approx \sqrt{15} \doteq 3.873$$
,

whereas the use of (72a), with p = 1/x, gives

$$\mu_1 \approx \sqrt{\frac{280}{19}} \doteq 3.839.$$

(e) Show that the true value of μ_1 is the smallest zero of $J_1(x)$ and hence (see Appendix B) is given by $\mu_1 \doteq 3.832$.

Section 5.6

20. Reduce each of the following differential equations to the standard form

$$\frac{d}{dx}\left(p\frac{dy}{dx}\right) + (q + \lambda r)y = 0:$$
(a) $x\frac{d^2y}{dx^2} + 2\frac{dy}{dx} + (x + \lambda)y = 0$
(b) $\frac{d^2y}{dx^2} + \frac{dy}{dx}\cot x + \lambda y = 0,$
(c) $\frac{d^2y}{dx^2} + a\frac{dy}{dx} + (b + \lambda)y = 0$ (*a*, *b* constants),
(d) $x\frac{d^2y}{dx^2} + (c - x)\frac{dy}{dx} - ay + \lambda y = 0$ (*a*, *c* constants)

21. By considering the characteristic functions of the problem

$$\frac{d^2y}{dx^2} + \mu^2 y = 0, \qquad y(0) = 0, \quad \alpha y'(L) + y(L) = 0,$$

with $\alpha \ge 0$, and using the results of Section 5.6, show that

$$\int_0^L \sin \mu_1 x \sin \mu_2 x \, dx = 0$$

when μ_1 and μ_2 are distinct positive solutions of the equation $\tan \mu L + \alpha \mu = 0$.

22. By considering the characteristic functions of the problem

$$x\frac{d^2y}{dx^2} + \frac{dy}{dx} + \mu^2 xy = 0, \qquad y(0) \text{ finite,} \quad y(L) = 0,$$

and using the results of Section 5.6, show that

$$\int_0^L x J_0(\mu_1 x) J_0(\mu_2 x) dx = 0$$

when μ_1 and μ_2 are distinct positive roots of the equation $J_0(\mu L) = 0$.

23. By considering the characteristic functions of the problem

$$(1 - x^2)\frac{d^2y}{dx^2} - 2x\frac{dy}{dx} + \lambda y = 0, \qquad y(\pm 1) \text{ finite,}$$

and using the results of Section 5.6, show that

$$\int_{-1}^{1} P_r(x) P_s(x) \, dx = 0$$

when r and s are distinct nonnegative integers. [Notice that the equation is Legendre's equation, and that λ must be of the form n(n + 1), where n is an integer, in order that solutions finite at both x = 1 and x = -1 exist, according to Section 4.12.]

24. Derive Equation (105) from (104).

Problems

Section 5.7

25. Determine the coefficients in the representation

$$f(x) = \sum_{n=1}^{\infty} A_n \sin nx \qquad (0 < x < \pi)$$

in the following cases:

(a)
$$f(x) = 1$$
, (b) $f(x) = x$, (c) $f(x) = \begin{cases} \frac{1}{2} & (x = \pi/2) \\ 0 & (x > \pi/2). \end{cases}$

26. (a) If the expansion $f(x) = \sum_{n=1}^{\infty} A_n \sin nx$ is valid in $(0,\pi)$, show formally that

$$\int_0^{\pi} [f(x)]^2 dx = \frac{\pi}{2} \sum_{n=1}^{\infty} A_n^2$$

(b) From this result, and from the results of Problem 25(a,b), deduce the following relations:

$$\frac{1}{1^2} + \frac{1}{3^2} + \frac{1}{5^2} + \cdots = \frac{\pi^2}{8}, \qquad \frac{1}{1^2} + \frac{1}{2^2} + \frac{1}{3^2} + \cdots = \frac{\pi^2}{6}.$$

27. Expand the function f(x) = 1 in a series of the characteristic functions of the boundary-value problem

$$\frac{d^2y}{dx^2} + \lambda y = 0,$$

$$y(0) = 0, \quad \alpha L y'(L) + y(L) = 0 \quad (\alpha \ge 0),$$

over the interval (0,L).

Section 5.8

28. Obtain the solution of the problem

$$\frac{d^2y}{dx^2} + \lambda y = F(x), \qquad y(0) = y(L) = 0$$

in the form

$$y(x) = \sum_{n=1}^{\infty} \frac{A_n}{\lambda - n^2 \pi^2 / L^2} \sin \frac{n \pi x}{L} \qquad (0 < x < L)$$

when $\lambda \neq n\pi/L$ (n = 1, 2, ...), where A_n is the *n*th coefficient in the expansion

$$F(x) = \sum_{n=1}^{\infty} A_n \sin \frac{n \pi x}{L} \qquad (0 < x < L)$$

29. Obtain the solution of the problem

$$\frac{d}{dx}\left(x\frac{dy}{dx}\right) + \lambda xy = F(x), \qquad y(0) \text{ finite,} \quad y(L) = 0$$

 $\int 1 \qquad (x < \pi/2)$

in the form

$$y(x) = \sum_{n=1}^{\infty} \frac{A_n}{\lambda - \mu_n^2} J_0(\mu_n x)$$
 (0 < x < L),

when $\lambda \neq \mu_1^2, \mu_2^2, \ldots$, where μ_n is the *n*th root of the equation $J_0(\mu L) = 0$, and where A_n is the *n*th coefficient in the expansion

$$\frac{F(x)}{x} = \sum_{n=1}^{\infty} A_n J_0(\mu_n x) \qquad (0 < x < L).$$

Show also that the coefficient A_n is to be determined by the equation

$$A_n \int_0^L x [J_0(\mu_n x)]^2 \, dx = \int_0^L F(x) \, J_0(\mu_n x) \, dx.$$

Section 5.9

30. If the initial approximation $y_1(x)$ is orthogonal to the first characteristic function $\varphi_1(x)$ with respect to r(x), show that the method of Stodola and Vianello leads to the *second* characteristic quantities λ_2 and $\varphi_2(x)$ unless $y_1(x)$ is also orthogonal to $\varphi_2(x)$.

31. Show that zero is a characteristic number of the problem

$$\frac{d^2y}{dx^2} + \lambda y = 0, \qquad y'(0) = y'(1) = 0,$$

corresponding to the characteristic function $\varphi(x) = 1$, in addition to the characteristic numbers $\lambda_n = n^2 \pi^2$ (n = 1, 2, ...), corresponding to the respective characteristic functions $\varphi_n(x) = \cos n\pi x$. Hence deduce that the method of Stodola and Vianello would lead to $\lambda = 0$ and y(x) = 1 in this case unless the initial approximation $y_1(x)$ were orthogonal to the function y = 1 over (0,1), that is, unless

$$\int_0^1 y_1(x) dx = 0.$$

32. Apply the method of Stodola and Vianello to the approximate determination of the smallest nonzero characteristic number of the problem considered in Problem 31, assuming $y_1(x)$ in the form

$$y_1(x) = c_0 + c_1 x + c_2 x^2 + c_2 x^2,$$

and determining the constants such that

$$y'_1(0) = 0, \quad y'_1(1) = 0,$$

and

$$\int_0^1 y_1(x)\,dx\,=0.$$

Show that the resultant initial approximation $y_1 = 1 - 6x^2 + 4x^3$, corresponding to a convenient choice of the arbitrary multiplicative factor, leads to the result

$$f_1 = \frac{1}{10}(1 - 5x^3 + 5x^4 - 2x^5)$$

if the same three conditions are imposed on f_1 . Notice that Equation (70) cannot be applied here, but that the simple process of equating $y_1(x)$ to $\lambda f_1(x)$ at either x = 0 or x = 1 gives $\lambda_1 \approx 10$, whereas the use of (72a) gives $\lambda_1 \approx \frac{306}{31} \doteq 9.871$. (The exact result is $\lambda_1 = \pi^2 \doteq 9.870$.)

Section 5.10

33. Establish, by direct integration, the truth of (a) Equation (134), (b) Equation (140), and (c) Equations (154).

34. Expand each of the following functions in a Fourier sine series of period 2L, over the interval (0,L), and in each case sketch the function represented by the series in the interval (-3L,3L):

(a)
$$f(x) = \begin{cases} 0 & (x < 0) \\ x(L - x) & (x > 0), \end{cases}$$
 (b) $f(x) = \begin{cases} 0 & (x < 0) \\ x & (0 < x < L/2) \\ L - x & (x > L/2), \end{cases}$
(c) $f(x) = \begin{cases} 1 & (x < L/2) \\ 0 & (x > L/2), \end{cases}$ (d) $f(x) = \sin \frac{\pi x}{2L}, \end{cases}$
(e) $f(x) = \begin{cases} \sin \frac{\pi x}{L} & (0 < x < L/2) \\ 0 & (\text{otherwise}), \end{cases}$ (f) $f(x) = \begin{cases} \frac{1}{\epsilon} & (0 < x < \epsilon < L) \\ 0 & (\text{otherwise}). \end{cases}$

35. (a-f) Expand each of the functions listed in Problem 34 in a Fourier cosine series of period 2L, over the interval (0,L), and in each case sketch the function represented by the series in the interval (-3L,3L).

36. (a) Obtain the expansion

$$\cos \alpha x = \frac{\sin \pi \alpha}{\pi \alpha} + \sum_{n=1}^{\infty} (-1)^n \frac{2\alpha \sin \pi \alpha}{\pi (\alpha^2 - n^2)} \cos nx \qquad (-\pi \leq x \leq \pi),$$

when α is nonintegral.

(b) Deduce from this result that

$$\cot \pi \alpha = \frac{1}{\pi} \left(\frac{1}{\alpha} - \sum_{n=1}^{\infty} \frac{2\alpha}{n^2 - \alpha^2} \right),$$

when α is nonintegral.

37. Plot the first three partial sums $s_n(x)$, for which

$$s_1(x) = \sin x$$
, $s_2(x) = \sin x - \frac{1}{2} \sin 2x$

and

$$s_3(x) = \sin x - \frac{1}{2} \sin 2x + \frac{1}{3} \sin 3x$$

of the right-hand member of the relation

$$\frac{x}{2} = \sum_{k=1}^{\infty} \frac{(-1)^{k+1}}{k} \sin kx \qquad (0 < x < \pi),$$

over the interval $(0,\pi)$, and compare these plots with a plot of the limit function.

38. Plot the first three distinct partial sums of the series representation

$$x(\pi - x) = \frac{8}{\pi} \left(\sin x + \frac{1}{3^3} \sin 3x + \frac{1}{5^3} \sin 5x + \cdots \right) \qquad (0 < x < \pi),$$

over the interval $(0,\pi)$, and compare these plots with a plot of the limit function.

39. Show that, if we determine the coefficients in an N-term approximation of the form

$$f(x) \approx \sum_{n=1}^{N} a_n \sin \frac{n \pi x}{L} \qquad (0 < x < L)$$

in such a way that

$$\int_0^L \left[f(x) - \sum_{n=1}^N a_n \sin \frac{n \pi x}{L} \right]^2 dx = \min$$

(that is, by the method of *least squares*), the coefficients are determined in the form

$$a_k = \frac{2}{L} \int_0^L f(x) \sin \frac{k \pi x}{L} dx$$
 $(k = 1, 2, ..., N),$

that is, as the first N coefficients in the Fourier sine series representation. (Differentiate the quantity to be minimized with respect to a general coefficient a_k and equate the result to zero.)

Section 5.11

40. (a-f) Expand each of the functions listed in Problem 34 in a Fourier series of period 2L, over the interval (-L,L), and in each case sketch the function represented by the series in the interval (-3L,3L).

41. Expand each of the following functions in a Fourier series of period equal to the length of the indicated interval of representation:

(a)
$$f(x) = a + bx$$
 [0,P],
(b) $f(x) = \begin{cases} 0 & (x < 0) \\ 1 & (x > 0) \end{cases}$ [-1,1],
(c) $f(x) = \sin x$ [0, π],
(d) $f(x) = x$ [1,2].

42. (a) If the representation

$$f(x) = A_0 + \sum_{n=1}^{\infty} A_n \cos \frac{n \pi x}{L}$$
 (0 < x < L)

is valid, show formally that

$$\frac{2}{L}\int_0^L [f(x)]^2 dx = 2A_0^2 + \sum_{n=1}^\infty A_n^2.$$

(b) If the representation

$$f(x) = \sum_{n=1}^{\infty} B_n \sin \frac{n\pi x}{L} \qquad (0 < x < L)$$

is valid, show formally that

$$\frac{2}{L}\int_0^L [f(x)]^2 \, dx = \sum_{n=1}^\infty B_n^2.$$

(c) If the representation

$$f(x) = A_0 + \sum_{n=1}^{\infty} \left(A_n \cos \frac{n \pi x}{L} + B_n \sin \frac{n \pi x}{L} \right) \qquad (-L < x < L)$$

is valid, show formally that

$$\frac{1}{L}\int_{-L}^{L} [f(x)]^2 dx = 2A_0^2 + \sum_{n=1}^{\infty} (A_n^2 + B_n^2).$$

[It is a remarkable fact that the *results* of parts (a), (b), and (c) are valid whenever f(x) is bounded and integrable over the relevant interval, whether or not the associated Fourier representations referred to are valid. These results are often called the *Parseval equalities*.]

43. (a) Show that the Fourier series expansion described by Equations (148) and (149) can be written in the form

$$f(x) = \frac{1}{2L} \int_{-L}^{L} f(t) dt + \sum_{n=1}^{\infty} \frac{1}{L} \int_{-L}^{L} f(t) \left[\cos \frac{n\pi x}{L} \cos \frac{n\pi t}{L} + \sin \frac{n\pi x}{L} \sin \frac{n\pi t}{L} \right] dt$$
$$= \frac{1}{2L} \int_{-L}^{L} f(t) dt + \sum_{n=1}^{\infty} \frac{1}{L} \int_{-L}^{L} f(t) \cos \frac{n\pi (x-t)}{L} dt$$

when -L < x < L.

(b) By making use of Euler's formula

$$\cos\frac{n\pi(x-t)}{L}=\frac{1}{2}\left[e^{\frac{in\pi(x-t)}{L}}+e^{\frac{-in\pi(x-t)}{L}}\right],$$

show that the last form can be transformed to

$$f(x) = \sum_{n=-\infty}^{\infty} \frac{1}{2L} \int_{-L}^{L} f(t) e^{\frac{in\pi(x-t)}{L}} dt$$

or, equivalently,

$$f(x) = \sum_{n=-\infty}^{\infty} a_n e^{\frac{in\pi x}{L}} \qquad (-L < x < L)$$
$$a_n = \frac{1}{2L} \int_{-L}^{L} f(x) e^{\frac{-in\pi x}{L}} dx.$$

where

[This is known as the complex form of the Fourier series.]

44. (a) Assuming the relation

$$e^{\frac{x}{2}\left(r-\frac{1}{r}\right)} = \sum_{n=-\infty}^{\infty} r^n J_n(x)$$

(see Problem 28, Chapter 4), replace r by $e^{i\theta}$ and deduce the Fourier expansion

$$e^{ix\sin\theta} = \sum_{n=-\infty}^{\infty} J_n(x) e^{in\theta}.$$

(b) By equating separately the real and imaginary parts of the two sides of this equation, and recalling that $J_{-n}(x) = (-1)^n J_n(x)$, show that

$$\cos(x\sin\theta) = J_0(x) + 2[J_2(x)\cos 2\theta + J_4(x)\cos 4\theta + \cdots]$$

and

$$\sin(x\sin\theta) = 2[J_1(x)\sin\theta + J_3(x)\sin 3\theta + \cdots]$$

45. (a) From the result of Problem 44(a) deduce the integral formula

$$J_n(x) = \frac{1}{2\pi} \int_{-\pi}^{\pi} e^{i(x \sin \theta - n\theta)} d\theta$$
$$= \frac{1}{\pi} \int_{0}^{\pi} \cos(x \sin \theta - n\theta) d\theta,$$

when *n* is an integer.

(b) From this result or from the result of Problem 44(b), also obtain the relations

$$J_n(x) = \frac{1}{\pi} \int_0^{\pi} \cos(x \sin \theta) \cos n\theta \, d\theta$$
$$= \frac{2}{\pi} \int_0^{\pi/2} \cos(x \sin \theta) \cos n\theta \, d\theta \qquad (n \text{ even});$$
$$J_n(x) = \frac{1}{\pi} \int_0^{\pi} \sin(x \sin \theta) \sin n\theta \, d\theta$$
$$= \frac{2}{\pi} \int_0^{\pi/2} \sin(x \sin \theta) \sin n\theta \, d\theta \qquad (n \text{ odd}).$$

[Notice that the integrands in part (b) are symmetrical with respect to $\theta = \pi/2$, with the stated restrictions on *n*.]

(c) In particular, show that

$$J_0(x) = \frac{2}{\pi} \int_0^{\pi/2} \cos(x \sin \theta) d\theta$$
$$= \frac{2}{\pi} \int_0^1 \frac{\cos xt}{\sqrt{1-t^2}} dt.$$

Section 5.12

46. Expand the function $f(x) = x^p$ in a series of the characteristic functions of the boundary-value problem

$$x\frac{d}{dx}\left(x\frac{dy}{dx}\right) + (\mu^2 x^2 - p^2)y = 0$$

y(0) finite, $y(L) = 0$

over the interval 0 < x < L, where p is a given nonnegative constant. [Make use of Equation (108), Section 4.9.]

47. Expand $f(x) = x^p$ in a series of the characteristic functions of the boundaryvalue problem

$$x \frac{d}{dx} \left(x \frac{dy}{dx} \right) + (\mu^2 x^2 - p^2) y = 0$$

y(0) finite, y'(L) = 0

over the interval 0 < x < L, where p is a given nonnegative constant. (Consider the case p = 0 separately.)

48. Obtain the solution of the problem

$$\frac{d}{dx}\left(x\frac{dy}{dx}\right) + \mu^2 xy = x, \qquad y(0) \text{ finite, } y(L) = 0$$

in the form

$$y = \sum_{n=1}^{\infty} \frac{2}{\mu^2 - \mu_n^2} \frac{J_0(\mu_n x)}{\mu_n L J_1(\mu_n L)} \qquad (0 < x < L),$$

where $J_0(\mu_n L) = 0$ (n = 1, 2, ...), if $J_0(\mu L) \neq 0$.

49. If $\mu = 0$ in Problem 48, show that the solution is of the form $y = \frac{1}{4}(x^2 - L^2)$. Hence deduce the representation

$$1 - \left(\frac{x}{L}\right)^2 = \sum_{n=1}^{\infty} \frac{8}{\alpha_n^3} \frac{J_0\left(\alpha_n \frac{x}{L}\right)}{J_1(\alpha_n)} \qquad (0 < x < L),$$

,

where α_n is the *n*th zero of $J_0(x)$.

Section 5.13

50. Show that the coefficients in the expansion

$$F(\theta) = \sum_{n=0}^{\infty} A_n P_n(\cos \theta) \qquad (0 < \theta < \pi)$$

are of the form

$$A_n = \frac{2n+1}{2} \int_0^{\pi} F(\theta) P_n(\cos \theta) \sin \theta \, d\theta \qquad (n = 0, 1, \ldots).$$

51. Find the first three coefficients in the expansion of the function

$$f(x) = \begin{cases} 0 & (-1 < x < 0) \\ x & (0 < x < 1) \end{cases}$$

in a series of Legendre polynomials over the interval (-1,1).

52. Find the first three coefficients in the expansion of the function

$$f(\theta) = \begin{cases} \cos \theta & (0 < \theta < \pi/2) \\ 0 & (\pi/2 < \theta < \pi) \end{cases}$$

in a series of the form

$$f(\theta) = \sum_{n=0}^{\infty} A_n P_n(\cos \theta) \qquad (0 < \theta < \pi).$$

53. Show that if the coefficients A_0, A_1, \ldots, A_n are determined in such a way that

$$I = \int_{-1}^{1} \{f(x) - [A_0 P_0(x) + A_1 P_1(x) + \cdots + A_n P_n(x)]\}^2 dx = \text{minimum},$$

so that the integral of the squared error in the approximation

$$f(x) \approx \sum_{k=0}^{n} A_k P_k(x) \qquad (-1 \le x \le 1)$$

is least, the coefficients are obtained (by requiring that $\partial I/\partial A_k = 0$ for k = 0, 1, ..., *n* and making use of the relevant orthogonality) in the form

$$A_k \int_{-1}^{1} [P_k(x)]^2 dx = \int_{-1}^{1} f(x) P_k(x) dx \qquad (k = 0, 1, ..., n).$$

Hence, recalling that any polynomial of degree *n* can be expressed as a linear combination of $P_0(x), P_1(x), \ldots, P_n(x)$, deduce that the polynomial of degree *n* which best approximates a function over (-1,1) in the least-squares sense consists of the sum of the terms of degree not greater than *n* in the Legendre expansion of that function over that interval.

54. Determine the first three nonvanishing terms in the Legendre expansion, over the interval (-1,1), of the function

$$F(x) = \begin{cases} \frac{1}{2\epsilon} & (|x| < \epsilon) \\ 0 & (\epsilon < |x| < 1), \end{cases}$$

in the form

where

$$F(x) = A_0 P_0(x) + A_2 P_2(x) + A_4 P_4(x) + \cdots \quad (-1 < x < 1)$$
$$A_0 = \frac{1}{2}, \quad A_2 = -\frac{5}{4}(1 - \epsilon^2), \quad A_4 = \frac{9}{16}(3 - 10\epsilon^2 + 7\epsilon^4), \quad \ldots$$

55. Use the results of Section 5.8 to obtain the solution of the problem

$$\frac{d}{dx}\left[(1-x^3)\frac{dy}{dx}\right] + ky = F(x),$$

y(±1) finite,

where F(x) is defined in Problem 54, in the form

$$y = \sum_{n=0}^{\infty} \frac{A_n}{k - n(n+1)} P_n(x),$$

if $k \neq 0, 6, 20, ...$

56. As $\epsilon \to 0$ in Problems 54 and 55, the function F(x) tends toward the "unit impulse function." It can be shown that the series in Problem 54 does not converge when $\epsilon = 0$. However, the series solution in Problem 55 *does* then converge.

(a) Show that if F(x) tends toward the unit impulse function $\delta(x)$, the solution of Problem 55 becomes

$$y = \frac{\frac{1}{2}}{k} P_0(x) - \frac{\frac{5}{4}}{k-6} P_2(x) + \frac{\frac{27}{16}}{k-20} P_4(x) + \cdots$$

(b) Show that the same result is obtained if the *formal* expansion

$$\delta(x) \sim \frac{1}{2} P_0(x) - \frac{5}{4} P_2(x) + \frac{27}{16} P_4(x) + \cdots$$

is dealt with as though it were truly a convergent representation.

[In the procedure of part (a), the limit is taken *after* the coefficients of the expansion have been subsituted into the general solution, whereas in that of part (b) the limit is taken formally *before* substitution. The former is mathematically sound; the second, strictly speaking, is not. However, both give the same final result. Since in many similar problems the second procedure may be much more simply applied than the first, it is rather frequently used in practice and can be justified from a practical point of view by an analysis similar to the one just considered.]

57. (a) With the notation of Section 4.12, show that the solution of Legendre's equation of order p which vanishes when x = 0 must be a multiple of $v_p(x)$. Making use of the fact that $v_p(x)$ is not finite at x = 1 unless p is an odd integer, deduce that the problem

$$\frac{d}{dx}\left[(1-x^2)\frac{dy}{dx}\right] + \lambda y = 0, \quad y(0) = 0, \quad y(1) \text{ finite}$$

possesses the characteristic numbers $\lambda_n = 2n(2n - 1)$ and the corresponding characteristic functions $\varphi_n(x) = P_{2n-1}(x)$, where n = 1, 2, ...

(b) In a similar way, deduce that the problem

$$\frac{d}{dx}\left[(1-x^2)\frac{dy}{dx}\right] + \lambda y = 0, \quad y'(0) = 0, \quad y(1) \text{ finite}$$

possesses the characteristic numbers $\lambda_n = 2n(2n + 1)$ and the corresponding characteristic functions $\varphi_n(x) = P_{2n}(x)$, where n = 0, 1, 2, ...

58. (a) Obtain the solution of the problem

$$\frac{d}{dx}\left[(1-x^2)\frac{dy}{dx}\right] + \lambda y = F(x), \qquad y(0) = 0, \quad y(1) \text{ finite}$$

in the form

$$y = \sum_{n=1}^{\infty} \frac{C_n}{\lambda - 2n(2n-1)} P_{2n-1}(x), \quad C_n = (4n-1) \int_0^1 F(x) P_{2n-1}(x) dx,$$

if $\lambda \neq 2, 12, 30, ...$

(b) If the condition at x = 0 in part (a) is replaced by the condition y'(0) = 0, obtain the solution in the form

$$y = \sum_{n=0}^{\infty} \frac{C_n}{\lambda - 2n(2n+1)} P_{2n}(x), \quad C_n = (4n+1) \int_0^1 F(x) P_{2n}(x) dx,$$

if $\lambda \neq 0, 6, 20, \ldots$

Section 5.14

59. With the symbols

$$u_n = \frac{n\pi}{L} = n \Delta u, \qquad \Delta u = \frac{\pi}{L}$$

where u_n is a function of the integer *n*, increasing by jumps of magnitude Δu , show that the complex form of the Fourier series representation of f(x), as obtained in Problem 43, can be expressed in the form

$$f(x) = \frac{1}{2\pi} \sum_{n=-\infty}^{\infty} \Delta u \int_{-L}^{L} f(t) e^{-iu_n(t-x)} dt.$$

[Notice the formal similarity between this form (as $L \rightarrow \infty$) and the Fourier integral representation expressed in the form (229).]

60. Solve the problem

$$\frac{d^2y}{dx^2} - k^2y + f(x) = 0 \quad (x > 0), \qquad y(0) = 0, \quad \lim_{x \to \infty} y(x) \text{ finite}$$

formally, by the following steps:

(a) Write

$$y(x) = \frac{2}{\pi} \int_0^\infty Y(u) \sin ux \, du \quad \text{and} \quad f(x) = \frac{2}{\pi} \int_0^\infty F(u) \sin ux \, du$$

and, by substitution into the differential equation and formal differentiation under the integral sign, obtain the condition

$$\int_0^\infty [(u^2 + k^2) Y(u) - F(u)] \sin ux \, du = 0.$$

Problems

Hence deduce that

$$Y(u)=\frac{F(u)}{u^2+k^2}.$$

(b) By expressing F in terms of f, obtain the formal solution

$$y(x) = \frac{2}{\pi} \int_0^\infty \frac{F(u)}{u^2 + k^2} \sin ux \, du$$
$$F(u) = \int_0^\infty f(t) \sin ut \, dt.$$

where

61. By proceeding as in Problem 60, obtain a formal solution of the problem

$$\frac{d^2y}{dx^2} - k^2y + f(x) = 0 \quad (x > 0), \qquad y'(0) = 0, \quad \lim_{x \to \infty} y(x) \text{ finite}$$

in the form

$$y(x) = \frac{2}{\pi} \int_0^\infty \frac{F(u)}{u^2 + k^2} \cos ux \, du$$

where

$$F(u) = \int_0^\infty f(t) \cos ut \, dt.$$

62. (a) By assuming

$$y(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{iux} Y(u) du \text{ and } f(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{iux} F(u) du,$$

obtain a formal solution of the problem

$$\frac{d^2y}{dx^2} - k^2y + f(x) = 0 \qquad (-\infty < x < \infty),$$
$$\lim_{x \to -\infty} y(x) \text{ finite,} \qquad \lim_{x \to +\infty} y(x) \text{ finite}$$

in the form

$$y(x) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \frac{F(u)}{u^2 + k^2} e^{iux} du \quad \text{where} \quad F(u) = \int_{-\infty}^{\infty} e^{-iut} f(t) dt.$$

(b) Show that this result also can be written in the real form

$$y(x)=\frac{1}{2\pi}\int_{-\infty}^{\infty}\frac{1}{u^2+k^2}\left[\int_{-\infty}^{\infty}f(t)\cos u(t-x)\,dt\right]du.$$

63. Obtain a formal solution of the problem

$$\frac{d}{dx}\left(x\frac{dy}{dx}\right) - k^2xy + f(x) = 0, \qquad y(0) \text{ finite, } \lim_{x \to \infty} y(x) \text{ finite}$$

in the form

$$y(x) = \int_0^\infty \frac{uF(u)}{u^2 + k^2} J_0(ux) \, du \quad \text{where} \quad F(u) = \int_0^\infty f(t) \, J_0(ut) \, dt.$$

64. Show that, if use is made of the formula

$$\int_0^\infty \frac{\cos \alpha u}{u^2 + k^2} \, du = \frac{\pi}{2k} \, e^{-k|\alpha|} \qquad (k > 0)$$

[see Equation (155a), Chapter 10], and if an interchange of order of integrations is permissible, the solution of Problem 62 can be expressed in the form

$$y(x) = \frac{1}{2k} \int_{-\infty}^{\infty} e^{-k|t-x|} f(t) dt.$$

(Similar simplifications of the solutions of Problems 60, 61, and 63 are possible.)

65. It is required to determine C(u) in such a way that the representation

$$f(x) = \frac{2}{\pi} \int_0^\infty C(u) (\alpha \sin ux + \beta u \cos ux) \, du$$

is valid when $0 < x < \infty$, where α and β are prescribed constants.

(a) Assuming that such a representation exists, and that differentiation under the integral sign is permissible, show that, if we write

$$F(x)=\frac{2}{\pi}\int_0^\infty C(u)\sin ux\,du,$$

there follows

$$\beta F'(x) + \alpha F(x) = f(x), \qquad F(0) = 0$$

and

$$C(u) = \int_0^\infty F(x) \sin ux \, dx.$$

(b) Deduce that

$$F(x) = \frac{1}{\beta} \int_0^x f(t) e^{-(\alpha/\beta)(x-t)} dt$$

and, by formally interchanging the order of integration in the result of introducing this expression into the expression for C(u), obtain the evaluation

$$C(u) = \int_0^\infty f(t) \frac{\alpha \sin ut + \beta u \cos ut}{\alpha^2 + \beta^2 u^2} dt.$$

(c) Verify that the result obtained reduces to the Fourier cosine and sine representations when $\alpha = 0$ and when $\beta = 0$, respectively. [The representation can be shown to be valid under the same restrictions on f(x) as those which apply in those special cases.]

(d) Show that the representation obtained expresses f(x) as a superposition of the characteristic functions of the problem

$$\frac{d^2y}{dx^2} + \lambda y = 0, \qquad \beta y'(0) = \alpha y(0), \quad y(\infty) \text{ finite.}$$

Problems

66. By proceeding as in Problem 60 and using the results of Problem 65 obtain a formal solution of the problem

$$\frac{d^2y}{dx^2} - k^2y + f(x) = 0 \qquad (x > 0),$$

$$\beta y'(0) = \alpha y(0), \qquad \lim_{x \to \infty} y(x) \text{ finite}$$

in the form

$$y(x) = \frac{2}{\pi} \int_0^\infty G(u) \frac{\alpha \sin ux + \beta u \cos ux}{(u^2 + k^2)(\alpha^2 + \beta^2 u^2)} du$$

where

$$G(u) = \int_0^\infty f(t)(\alpha \sin ut + \beta u \cos ut) dt.$$

67. Show formally that the function

$$T(x,y) = \int_0^\infty e^{-\nu u} A(u) \sin ux \, du \qquad (y>0)$$

satisfies the partial differential equation

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} = 0$$

and vanishes when x = 0 and when $y \to \infty$. Determine A(u) so that T(x,y) formally reduces to a prescribed function f(x) when y = 0 and x is positive.

68. Show formally that the function

$$T(x,y) = \frac{1}{\pi} \int_0^\infty \int_{-\infty}^\infty e^{-yu} f(t) \cos u(t-x) dt du \qquad (y>0)$$

satisfies the partial differential equation

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} = 0,$$

vanishes when $y \to \infty$, and reduces to f(x) when y = 0.

CHAPTER 6

Vector Analysis

6.1. Elementary properties of vectors. A vector quantity is distinguished from a scalar quantity by the fact that a scalar quantity possesses only magnitude, whereas a vector quantity possesses both magnitude and (except for the



zero vector) direction. It is conventional to represent a vector geometrically as an arrow, pointing in the direction associated with the vector, and having a length proportional to the associated magnitude. Thus, in particular, the position of a point *B* relative to a point *A* can be completely described by a vector \rightarrow

AB from A to B, in the sense that the vector

AB specifies both the distance and direction from A to B and, in fact, indicates the displacement from A to B. Since motion from A to a third point C can be accomplished

along the vector \overrightarrow{AC} or, alternatively, along the vector \overrightarrow{AB} to \overrightarrow{B} and thence along \overrightarrow{BC} to C (Figure 6.1), it is natural to extend the concept of *addition* to vector quantities by writing

$$\overrightarrow{AC} = \overrightarrow{AB} + \overrightarrow{BC}$$

We say that two vectors are *equal* if they have the same direction and the same magnitude. In accordance with this definition a vector is unchanged if it is moved parallel to itself in any way; that is, the actual *position* of the vector in space can be assigned at pleasure. It should be mentioned, however, that in certain applications it may be *necessary* to specify the position of a vector. With this convention, * any two vectors **a** and **b** can be represented as arrows so placed that the terminal point of **a** coincides with the initial point of **b**. The sum of **a** and **b**,

$$\mathbf{c} = \mathbf{a} + \mathbf{b},$$

is then defined as the vector extending from the initial point of \mathbf{a} to the terminal point of \mathbf{b} . From the definition it follows readily that vector addition is *commutative*,

$$\mathbf{a} + \mathbf{b} = \mathbf{b} + \mathbf{a} \tag{1}$$

and also that the associative law

$$(a + b) + c = a + (b + c)$$
 (2)

is satisfied. The common value of the two expressions in (2) is written as the sum $\mathbf{a} + \mathbf{b} + \mathbf{c}$ of the three vectors, and the definition is readily extended to the sum of any number of vectors.

If the initial and terminal points of the arrow representing a vector are interchanged, the resulting vector is called the *negative* of the given vector. Thus, for example, $\overrightarrow{BA} = -\overrightarrow{AB}$. To *subtract* a vector **b** from a vector **a**, we add the negative of **b** to **a**,

$$\mathbf{a} - \mathbf{b} = \mathbf{a} + (-\mathbf{b}). \tag{3}$$

If a vector **a** is multiplied by a scalar *m*, the result is a vector *m***a** whose magnitude is the arithmetic product of *m* and the magnitude of **a** and whose direction is that of **a** if *m* is positive or that of $-\mathbf{a}$ if *m* is negative.

A vector of unit length is called a *unit vector*. It is clear that any vector can be written as the product of its length and a unit vector.

A vector of zero length is called a *zero vector* and is denoted by 0. The difference between two equal vectors is a zero vector.

We now consider a right-handed rectangular coordinate system, in which the coordinates x, y, and z are measured along three mutually perpendicular axes such that rotation of the x axis into the y axis about the z axis is accomplished by the right-hand rule, and define *unit vectors* i, j, and k having the directions of the positive x, y, and z axes, respectively (Figure 6.2). Then it is readily verified that any vector v whose projections on these axes are v_x , v_y , and v_z , respectively, can be written as the vector sum

$$\mathbf{v} = v_x \mathbf{i} + v_y \mathbf{j} + v_z \mathbf{k}. \tag{4}$$

The lengths v_x , v_y , and v_z are called the scalar *components* of v in the x, y, and z directions (Figure 6.3). If, when the initial point of v coincides with the

* In printed work it is customary to denote a vector quantity by a boldface letter. Thus, a represents a vector, and *a* represents a scalar quantity. origin of the system, the angles measured to v from the positive x, y, and z axes are denoted by α , β , and γ , respectively, there follows

$$v_x = v \cos \alpha, \quad v_y = v \cos \beta, \quad v_z = v \cos \gamma,$$
 (5)

where v is written for the length or magnitude of the vector v. The notation |v| is also frequently used in place of v. The numbers $\cos \alpha$, $\cos \beta$, and $\cos \gamma$ are called the *direction cosines* of v and are frequently denoted by *l*, *m*, and *n*,



Figure 6.2

Figure 6.3

respectively. Any three numbers (A,B,C) proportional to (l,m,n) are known as "direction ratios." From the geometrical relationship

$$v = \sqrt{v_x^2 + v_y^2 + v_z^2}$$
(6)

relating v to the scalar components of v, it follows that the direction cosines satisfy the equation

$$l^{2} + m^{2} + n^{2} \equiv \cos^{2} \alpha + \cos^{2} \beta + \cos^{2} \gamma = 1.$$
 (7)

We see also that if we write Equation (4) in the form

$$\mathbf{v} = v \left(\mathbf{i} \cos \alpha + \mathbf{j} \cos \beta + \mathbf{k} \cos \gamma\right), \tag{8}$$

the coefficient of v is a unit vector, and hence the vector

$$\mathbf{v}_1 = \mathbf{i} \cos \alpha + \mathbf{j} \cos \beta + \mathbf{k} \cos \gamma \tag{9}$$

is a unit vector in the direction of v. Combining Equations (5) and (6), we have

$$\cos \alpha = \frac{v_x}{\sqrt{v_x^2 + v_y^2 + v_z^2}}, \quad \cos \beta = \frac{v_y}{\sqrt{v_x^2 + v_y^2 + v_z^2}}, \\ \cos \gamma = \frac{v_z}{\sqrt{v_x^2 + v_y^2 + v_z^2}}.$$
(10)

6.2. The scalar product of two vectors. Two types of products of two vectors **a** and **b** are conventionally defined. The first type of product is called the scalar, dot, or inner product and is written as $\mathbf{a} \cdot \mathbf{b}$ or $(\mathbf{a} \mathbf{b})$. This product is defined to be a scalar equal to the product of the lengths of the two vectors

and the cosine of the angle θ between the *positive* directions of the two vectors (see Figure 6.4), and hence is given by the equation

$$\mathbf{a} \cdot \mathbf{b} = ab \cos \theta. \tag{11}$$

Since $b \cos \theta$ is the projection of **b** on **a**, whereas $a \cos \theta$ is the projection of **a** on **b**, it follows that $\mathbf{a} \cdot \mathbf{b}$ is numerically equal to the length of **b** times the projection of **a** on **b**, and also is equal to the length of **a** times

the projection of **b** on **a**. If θ is obtuse, the projections considered are to be taken as negative. In particular, *if two vectors are perpendicular*, *their dot product is zero*. From the geometrical definition it is readily seen that the dot product is commutative,

$$\mathbf{a} \cdot \mathbf{b} = \mathbf{b} \cdot \mathbf{a},\tag{12}$$

and distributive,

$$\mathbf{a} \cdot (\mathbf{b} + \mathbf{c}) = \mathbf{a} \cdot \mathbf{b} + \mathbf{a} \cdot \mathbf{c}. \tag{13}$$

For the unit vectors i, j, and k there follows immediately

$$\mathbf{i} \cdot \mathbf{i} = \mathbf{j} \cdot \mathbf{j} = \mathbf{k} \cdot \mathbf{k} = 1$$

$$\mathbf{j} = \mathbf{j} \cdot \mathbf{i} = \mathbf{j} \cdot \mathbf{k} = \mathbf{k} \cdot \mathbf{j} = \mathbf{i} \cdot \mathbf{k} = \mathbf{k} \cdot \mathbf{i} = 0$$
, (14)

and hence, if

$$\mathbf{a} = a_x \mathbf{i} + a_y \mathbf{j} + a_z \mathbf{k}, \qquad \mathbf{b} = b_x \mathbf{i} + b_y \mathbf{j} + b_z \mathbf{k}, \tag{15}$$

we have, using Equation (13),

i٠

$$\mathbf{a} \cdot \mathbf{b} = a_x b_x + a_y b_y + a_z b_z. \tag{16}$$

The angle θ between **a** and **b** is given by Equations (11) and (16) in the form

$$\cos\theta = \frac{a_x b_x + a_y b_y + a_z b_z}{\sqrt{a_x^2 + a_y^2 + a_z^2} \sqrt{b_x^2 + b_y^2 + b_z^2}},$$
(17)

or, if we denote the direction cosines of **a** and **b** by (l_1, m_1, n_1) and (l_2, m_2, n_2) ,

$$\cos \theta = l_1 l_2 + m_1 m_2 + n_1 n_2. \tag{18}$$

The abbreviation a^2 is sometimes used to indicate $a \cdot a$. Then we also have

$$\mathbf{a}^2 = \mathbf{a} \cdot \mathbf{a} = |\mathbf{a}|^2 = a^2. \tag{19}$$



The dot product is particularly useful in expressing a given vector \mathbf{v} as a linear combination of three mutually perpendicular unit vectors \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 . For if we write

$$\mathbf{v} = c_1 \mathbf{u}_1 + c_2 \mathbf{u}_2 + c_3 \mathbf{u}_3, \tag{20}$$

we may successively take the dot products of \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 into both sides of (20) and so obtain

$$c_1 = \mathbf{v} \cdot \mathbf{u}_1, \quad c_2 = \mathbf{v} \cdot \mathbf{u}_2, \quad c_3 = \mathbf{v} \cdot \mathbf{u}_3.$$

Thus Equation (20) becomes

$$\mathbf{v} = (\mathbf{v} \cdot \mathbf{u}_1)\mathbf{u}_1 + (\mathbf{v} \cdot \mathbf{u}_2)\mathbf{u}_2 + (\mathbf{v} \cdot \mathbf{u}_3)\mathbf{u}_3. \tag{21}$$

It is seen that this representation could fail only if a component of v were perpendicular to \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 . This condition, however, is clearly impossible in space of three dimensions, since \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 are assumed to be mutually perpendicular.

6.3. The vector product of two vectors. The second conventional type of product of two vectors **a** and **b** is called the vector, cross, or outer product and is written as $\mathbf{a} \times \mathbf{b}$ or $[\mathbf{a} \mathbf{b}]$. It is defined to be a vector having the properties that (1) the length of $\mathbf{a} \times \mathbf{b}$ is the product of the lengths of **a** and **b** and the numerical value of the sine of the angle θ between the vectors and (2) the vector $\mathbf{a} \times \mathbf{b}$ is perpendicular to the plane of **a** and **b** and is so oriented that **a** is rotated into **b** about $\mathbf{a} \times \mathbf{b}$ by the right-hand rule, through not more than 180°.

According to the definition, there follows (see Figure 6.5)

$$|\mathbf{a} \times \mathbf{b}| = ab |\sin \theta|. \tag{22}$$

Thus the length of $\mathbf{a} \times \mathbf{b}$ is equal to twice the area of the triangle of which \mathbf{a} and \mathbf{b} form coterminous sides. It is seen that the cross product is not commutative, since, from the definition,

$$\mathbf{b} \times \mathbf{a} = -(\mathbf{a} \times \mathbf{b}). \tag{23}$$

However, the cross product is distributive, so that

$$\mathbf{a} \times (\mathbf{b} + \mathbf{c}) = \mathbf{a} \times \mathbf{b} + \mathbf{a} \times \mathbf{c}.$$
 (24)

This relation can be established by geometrical considerations. It is seen that the cross product of two parallel vectors is zero.

For the unit vectors i, j, and k there follows, from the definition,

$$\mathbf{i} \times = \mathbf{j} \times \mathbf{j} = \mathbf{k} \times \mathbf{k} = \mathbf{0}$$

$$\mathbf{i} \times \mathbf{j} = -\mathbf{j} \times \mathbf{i} = \mathbf{k}, \quad \mathbf{j} \times \mathbf{k} = -\mathbf{k} \times \mathbf{j} = \mathbf{i}, \quad \mathbf{k} \times \mathbf{i} = -\mathbf{i} \times \mathbf{k} = \mathbf{j}$$
(25)

These relations are easily remembered in terms of the cyclic arrangement

... **i j k i j k** ...

if we notice that the cross product of a vector into its neighbor is the following vector when reading to the right and is the negative of the following vector when reading to the left.



To calculate the cross product of two vectors \mathbf{a} and \mathbf{b} in terms of their components as given in (15), we make use of Equations (24) and (25) and obtain

$$\mathbf{a} \times \mathbf{b} = (a_y b_z - a_z b_y) \mathbf{i} + (a_z b_x - a_x b_z) \mathbf{j} + (a_x b_y - a_y b_x) \mathbf{k}.$$
(26)

This result can be written conveniently as the determinant

$$\mathbf{a} \times \mathbf{b} = \begin{vmatrix} \mathbf{i} & \mathbf{j} & \mathbf{k} \\ a_x & a_y & a_z \\ b_x & b_y & b_z \end{vmatrix}.$$
 (27)

The usefulness of the cross product may be illustrated by two physical applications. First, we consider a point P in a rigid body rotating with angular velocity of magnitude ω about a fixed axis. Let O be a point on the axis of rotation and represent the angular velocity by a vector $\boldsymbol{\omega}$ of length ω , extending along the axis of rotation in the sense determined by the right-hand rule (Figure 6.6). Then, if the position vector from O to P is written as **r**, the velocity vector associated with the point P is given by the equation

$$\mathbf{v} = \mathbf{\omega} \times \mathbf{r},\tag{28}$$

since the magnitude of this vector is $\omega r \sin \theta$, where $r \sin \theta$ is the distance from the axis of rotation to the point P, and the direction of the vector is as indicated in the figure.

As a second application, we consider the moment vector \mathbf{M} at a point O, associated with a force \mathbf{F} acting at a point P (Figure 6.7). If the vector from O to P is denoted by \mathbf{r} , the moment of \mathbf{F} about the *point* O is defined as the vector

$$\mathbf{M} = \mathbf{r} \times \mathbf{F}.\tag{29}$$

We see that the vector \mathbf{M} is perpendicular to the plane of \overline{OP} and the vector \mathbf{F} and has as its magnitude

$$|\mathbf{M}| = M = Fr \sin \theta,$$

where θ is the angle between r and F, and hence $r \sin \theta$ is the perpendicular distance from O to the line of action of F. We may thus speak of M as the scalar moment of the force about the axis of M. Geometrically, M is seen to be twice the area of the triangle determined by r and F. Further, the scalar



moment M_{OA} of the force about any axis OA through O is seen to be twice the area of the projection of this triangle on the plane through O perpendicular to OA. But since the angle between the two planes considered is the same as the angle between the corresponding axes, the projected area must be equal to the scalar projection of M on OA. That is, the scalar moment of F about any axis through O is numerically equal to the length of the projection of M on that axis.

Figure 6.7

6.4. Multiple products. Three types of products involving three vectors are of importance, namely, those of the respective forms $(\mathbf{a} \cdot \mathbf{b})\mathbf{c}$, The first type

 $(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c}$, and $(\mathbf{a} \times \mathbf{b}) \times \mathbf{c}$. The first type,

 $(\mathbf{a} \cdot \mathbf{b})\mathbf{c}$,

is merely the product of the scalar $(\mathbf{a} \cdot \mathbf{b})$ and the vector \mathbf{c} .

The second type,

$$(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c},$$

which is called the *triple scalar product*, is seen to be the dot product of the vector $\mathbf{a} \times \mathbf{b}$ and the vector \mathbf{c} , and hence is a scalar quantity. The value of this product is given by the product of the length of $\mathbf{a} \times \mathbf{b}$ and the projection of \mathbf{c}

on $\mathbf{a} \times \mathbf{b}$ (Figure 6.8). But since $\mathbf{a} \times \mathbf{b}$ is a vector perpendicular to the plane of \mathbf{a} and \mathbf{b} , having a length numerically equal to the area of the parallelogram of which \mathbf{a} and \mathbf{b} form coterminous sides, and since the projection of \mathbf{c} on $\mathbf{a} \times \mathbf{b}$ is the altitude of the parallelepiped with \mathbf{a} , \mathbf{b} , and \mathbf{c} as coterminous edges, it follows that $(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c}$ is numerically equal to the volume of this parallelepiped. Alternatively, we see that $(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c}$ is numerically equal to six times the volume of the tetrahedron determined by \mathbf{a} , \mathbf{b} , and \mathbf{c} as edges. The sign of the product depends



Figure 6.8

upon the relative orientation of the three vectors, and is positive if and only if **a**, **b**, and **c** form a right-handed system, in the sense that **c** and **a** \times **b** lie on the same side of the plane determined by **a** and **b**. From these facts it follows easily that

$$(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c} = (\mathbf{b} \times \mathbf{c}) \cdot \mathbf{a} = (\mathbf{c} \times \mathbf{a}) \cdot \mathbf{b}$$
(30)

and that the other products

$$(\mathbf{a} \times \mathbf{c}) \cdot \mathbf{b} = (\mathbf{b} \times \mathbf{a}) \cdot \mathbf{c} = (\mathbf{c} \times \mathbf{b}) \cdot \mathbf{a}$$
 (31)

have the opposite algebraic sign. We see that the triple scalar product is not changed by a cyclic permutation of the three elements. Since also, from Equations (30) and (12),

$$(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c} = (\mathbf{b} \times \mathbf{c}) \cdot \mathbf{a} = \mathbf{a} \cdot (\mathbf{b} \times \mathbf{c}),$$

it follows that the dot and cross can be interchanged in a triple scalar product. For this reason, the notation (a b c) is frequently used to indicate the common value of the products listed in Equation (30).

If we write

$$\mathbf{a} = a_x \mathbf{i} + a_y \mathbf{j} + a_z \mathbf{k}, \quad \mathbf{b} = b_x \mathbf{i} + b_y \mathbf{j} + b_z \mathbf{k}, \quad \mathbf{c} = c_x \mathbf{i} + c_y \mathbf{j} + c_z \mathbf{k},$$

we find that

$$(\mathbf{a} \mathbf{b} \mathbf{c}) = \mathbf{a} \cdot (\mathbf{b} \times \mathbf{c}) = \mathbf{a} \cdot \begin{vmatrix} \mathbf{i} & \mathbf{j} & \mathbf{k} \\ b_x & b_y & b_z \\ c_x & c_y & c_z \end{vmatrix}$$
$$= a_x (b_y c_z - b_z c_y) + a_y (b_z c_x - b_x c_z) + a_z (b_x c_y - b_y c_x),$$

or, in determinant form,

$$(\mathbf{a} \mathbf{b} \mathbf{c}) = \begin{vmatrix} a_x & a_y & a_z \\ b_x & b_y & b_z \\ c_x & c_y & c_z \end{vmatrix}.$$
 (32)

From Equation (32) we see that if two vectors in the product (a b c) are parallel, then the product vanishes, since in this case the corresponding elements in two rows of the determinant (32) are proportional.

The third type of triple product,

$$(\mathbf{a} \times \mathbf{b}) \times \mathbf{c},$$

is clearly a vector. Since it is perpendicular to $\mathbf{a} \times \mathbf{b}$, which is itself perpendicular to the plane of \mathbf{a} and \mathbf{b} , and is also perpendicular to \mathbf{c} , it follows that $(\mathbf{a} \times \mathbf{b}) \times \mathbf{c}$ is a vector which is in the plane of \mathbf{a} and \mathbf{b} and perpendicular to \mathbf{c} . Thus this product must be expressible as a linear combination of \mathbf{a} and \mathbf{b} ,

$$(\mathbf{a} \times \mathbf{b}) \times \mathbf{c} = m\mathbf{a} + n\mathbf{b}.$$

A rather involved calculation shows that $m = -(\mathbf{b} \cdot \mathbf{c})$ and $n = \mathbf{a} \cdot \mathbf{c}$ and hence there follows

$$(\mathbf{a} \times \mathbf{b}) \times \mathbf{c} = (\mathbf{a} \cdot \mathbf{c})\mathbf{b} - (\mathbf{b} \cdot \mathbf{c})\mathbf{a}.$$
 (33)

In a similar way, the vector $\mathbf{a} \times (\mathbf{b} \times \mathbf{c})$ is seen to be expressible as a linear combination of \mathbf{b} and \mathbf{c} , and the identity

$$\mathbf{a} \times (\mathbf{b} \times \mathbf{c}) = (\mathbf{a} \cdot \mathbf{c})\mathbf{b} - (\mathbf{b} \cdot \mathbf{a})\mathbf{c}$$
 (34)

can be established. The two formulas (33) and (34) are easily remembered if it is noticed that in each expansion the *middle factor* on the left is multiplied by the dot product of the other two factors, whereas in the term with the negative sign the *other factor in parentheses* on the left is multiplied by the dot product of the remaining factors.

It should be noticed that such combinations as ab, $(a \times b)c$, and $a \times b \times c$ are here left undefined.

Vector products involving more than three vectors are readily evaluated in terms of the products considered above. For example, the product

$$(\mathbf{a} \times \mathbf{b}) \cdot (\mathbf{c} \times \mathbf{d})$$

can be considered as the triple scalar product of **a**, **b**, and $(\mathbf{c} \times \mathbf{d})$ and hence if we write temporarily $\mathbf{u} = \mathbf{c} \times \mathbf{d}$ there follows

$$\mathbf{a} \times \mathbf{b} \cdot \mathbf{u} = \mathbf{a} \cdot \mathbf{b} \times \mathbf{u}$$

= $\mathbf{a} \cdot [\mathbf{b} \times (\mathbf{c} \times \mathbf{d})]$
= $\mathbf{a} \cdot [(\mathbf{b} \cdot \mathbf{d})\mathbf{c} - (\mathbf{b} \cdot \mathbf{c})\mathbf{d}],$

and hence

$$(\mathbf{a} \times \mathbf{b}) \cdot (\mathbf{c} \times \mathbf{d}) = (\mathbf{a} \cdot \mathbf{c})(\mathbf{b} \cdot \mathbf{d}) - (\mathbf{a} \cdot \mathbf{d})(\mathbf{b} \cdot \mathbf{c}).$$
 (35)

In particular, if we take

 $\mathbf{c} = \mathbf{a}, \quad \mathbf{d} = \mathbf{b},$

Equation (35) becomes

$$(\mathbf{a} \times \mathbf{b}) \cdot (\mathbf{a} \times \mathbf{b}) = (\mathbf{a} \cdot \mathbf{a})(\mathbf{b} \cdot \mathbf{b}) - (\mathbf{a} \cdot \mathbf{b})^2.$$
 (36)

This relationship is known as the *identity of Lagrange*. The truth of Equation (36) follows also from the fact that it can be written in the form

$$|\mathbf{a} \times \mathbf{b}|^2 = a^2 b^2 - a^2 b^2 \cos^2 \theta = (ab \sin \theta)^2$$
,

which result is in agreement with the definition of $\mathbf{a} \times \mathbf{b}$.

6.5. Differentiation of vectors. If the physical properties of a vector quantity v are functions of a parameter t, the derivative of the vector v(t) with respect to t is defined by the equation

$$\frac{d\mathbf{v}(t)}{dt} = \lim_{\Delta t \to 0} \frac{\mathbf{v}(t + \Delta t) - \mathbf{v}(t)}{\Delta t}.$$
(37)

From this definition it follows that the derivative of the product of a scalar s(t) and a vector $\mathbf{v}(t)$ is given by the familiar product law

$$\frac{d}{dt}s\mathbf{v} = s\frac{d\mathbf{v}}{dt} + \frac{ds}{dt}\mathbf{v}.$$
(38)

Hence, if a vector is expressed in terms of its components along the fixed coordinate axes,

$$\mathbf{v} = f(t)\,\mathbf{i} + g(t)\,\mathbf{j} + h(t)\,\mathbf{k},$$

there follows

$$\frac{d\mathbf{v}}{dt} = \frac{df}{dt}\mathbf{i} + \frac{dg}{dt}\mathbf{j} + \frac{dh}{dt}\mathbf{k},$$
(39)

since i, j, and k are constant vectors.

It follows also, from the definition, that the derivative of a product involving two or more vectors is defined as in the corresponding scalar case if the order of the factors is retained. Thus, for example, we obtain the formulas

$$\frac{d}{dt}(\mathbf{a} \cdot \mathbf{b}) = \mathbf{a} \cdot \frac{d\mathbf{b}}{dt} + \frac{d\mathbf{a}}{dt} \cdot \mathbf{b},$$
$$\frac{d}{dt}(\mathbf{a} \times \mathbf{b}) = \mathbf{a} \times \frac{d\mathbf{b}}{dt} + \frac{d\mathbf{a}}{dt} \times \mathbf{b},$$
$$\frac{d}{dt}(\mathbf{a} \cdot \mathbf{b} \times \mathbf{c}) = \frac{d\mathbf{a}}{dt} \cdot \mathbf{b} \times \mathbf{c} + \mathbf{a} \cdot \frac{d\mathbf{b}}{dt} \times \mathbf{c} + \mathbf{a} \cdot \mathbf{b} \times \frac{d\mathbf{c}}{dt}.$$

In the first case the order of factors in the separate terms is irrelevant. This is, however, not true in the second and third cases.

The derivative of a vector of constant length, but changing direction, is perpendicular to the vector. This may be seen by noticing that if \mathbf{a} has constant length there follows

$$\frac{d}{dt}(\mathbf{a}\cdot\mathbf{a})=\frac{d}{dt}a^2=0$$

and also

$$\frac{d}{dt}(\mathbf{a}\cdot\mathbf{a})=2\mathbf{a}\cdot\frac{d\mathbf{a}}{dt}.$$

These results are compatible only if either $\frac{d\mathbf{a}}{dt}$ is zero or $\frac{d\mathbf{a}}{dt}$ is perpendicular to \mathbf{a} .

6.6. Geometry of a space curve. The equations

$$x = x(t), \quad y = y(t), \quad z = z(t),$$
 (40)

define a curve in space as the parameter t varies over a specified range. We denote by **r** the *position vector* from the origin O to the point P(x,y,z), corresponding to a specified value of t,

$$\mathbf{r} = x \, \mathbf{i} + y \, \mathbf{j} + z \, \mathbf{k}. \tag{41}$$

If t is increased by Δt , and x,y,z increase accordingly, the vector from O to the new point Q is given by



Figure 6.9



and hence there follows

$$\Delta \mathbf{r} = (\Delta x) \,\mathbf{i} + (\Delta y) \,\mathbf{j} + (\Delta z) \,\mathbf{k}. \quad (42)$$

This vector is clearly the vector \overrightarrow{PQ} , and hence is of length equal to the chord \overrightarrow{PQ} . (See Figure 6.9.) If both sides of Equation (42) are divided by the increment Δt , there follows

$$\frac{\Delta \mathbf{r}}{\Delta t} = \frac{\Delta x}{\Delta t} \mathbf{i} + \frac{\Delta y}{\Delta t} \mathbf{j} + \frac{\Delta z}{\Delta t} \mathbf{k}.$$

If s represents arc length along the curve, we can artificially rewrite this equation in the form

$$\frac{\Delta \mathbf{r}}{\Delta t} = \left(\frac{\Delta x}{\Delta s}\,\mathbf{i} + \frac{\Delta y}{\Delta s}\,\mathbf{j} + \frac{\Delta z}{\Delta s}\,\mathbf{k}\right)\frac{\Delta s}{\Delta t}\,.\tag{43}$$

The vector $\overrightarrow{\Delta \mathbf{r}}$ in parentheses is in the direction \overrightarrow{PQ} , and has a length equal to the ratio of chord length to arc length. Thus, as $\Delta t \rightarrow 0$, we have in the limit

$$\frac{d\mathbf{r}}{dt} = \left(\frac{dx}{ds}\,\mathbf{i} + \frac{dy}{ds}\,\mathbf{j} + \frac{dz}{ds}\,\mathbf{k}\right)\frac{ds}{dt},\tag{44}$$

assuming that the derivatives exist, where the expression in parentheses is clearly a unit vector tangent to the space curve at the point *P*. Since this expression is equivalent to $\frac{d\mathbf{r}}{ds}$, we have the result that the derivative of a position vector to a space curve, with respect to arc length along the curve, is a unit vector tangent to the curve, pointing in the direction of increasing arc length. We denote this unit tangent vector by \mathbf{u} ,

$$\mathbf{u} = \frac{d\mathbf{r}}{ds} = \frac{dx}{ds}\mathbf{i} + \frac{dy}{ds}\mathbf{j} + \frac{dz}{ds}\mathbf{k}.$$
 (45)

Since **u** is a unit vector, there follows

$$\left(\frac{dx}{ds}\right)^2 + \left(\frac{dy}{ds}\right)^2 + \left(\frac{dz}{ds}\right)^2 = 1.$$

This equation can be written in the differential form

$$ds = \sqrt{dx^2 + dy^2 + dz^2} = \sqrt{\left(\frac{dx}{dt}\right)^2 + \left(\frac{dy}{dt}\right)^2 + \left(\frac{dz}{dt}\right)^2} dt, \quad (46)$$

giving the element of arc length for a space curve.

If t represents time, the vector $\frac{d\mathbf{r}}{dt}$ is the velocity vector associated with a point moving with speed $\frac{ds}{dt}$ along the curve. Thus Equation (44) becomes

$$\mathbf{v} = \mathbf{u} \frac{ds}{dt} = v\mathbf{u} \tag{47}$$

where $v = \frac{ds}{dt}$.

If Equation (47) is differentiated with respect to t, the acceleration vector **a** is obtained in the form

$$\mathbf{a} = \frac{dv}{dt}\mathbf{u} + v\frac{d\mathbf{u}}{dt} = \frac{dv}{dt}\mathbf{u} + v^2\frac{d\mathbf{u}}{ds}.$$
 (48)

Since **u** is a unit vector, the derivative $\frac{d\mathbf{u}}{ds}$ is perpendicular to the tangent vector.

Making use of Equation (45), we obtain

$$\frac{d\mathbf{u}}{ds} = \frac{d^2\mathbf{r}}{ds^2} = \frac{d^2x}{ds^2}\,\mathbf{i} + \frac{d^2y}{ds^2}\,\mathbf{j} + \frac{d^2z}{ds^2}\,\mathbf{k}.$$

The length of this vector is called the *curvature* of the curve and is a measure of the rate at which the tangent vector changes its direction with distance along the curve. The reciprocal of this value is the *radius of curvature*, denoted by ρ . Hence we may write

$$\frac{d\mathbf{u}}{ds} = \frac{1}{\rho} \,\mathbf{n},\tag{49}$$

where **n** is a unit vector, perpendicular to the tangent vector **u** at P, and known as the *principal normal vector*.* There follows also

$$\frac{1}{\rho} = \sqrt{\left(\frac{d^2x}{ds^2}\right)^2 + \left(\frac{d^2y}{ds^2}\right)^2 + \left(\frac{d^2z}{ds^2}\right)^2}.$$
(50)

* Equation (49) states that **n** and $\frac{d\mathbf{u}}{ds}$ point either in the same direction or in opposite directions, depending upon the sign of ρ . We here define ρ to be always positive, so that **n** and $\frac{d\mathbf{u}}{ds}$ always point in the same direction. However, other conventions also appear in the literature.

By introducing Equation (49) into (48), the acceleration vector is expressed in the form

$$\mathbf{a} = \frac{v^2}{\rho} \mathbf{n} + \frac{dv}{dt} \mathbf{u}.$$
 (51)

Thus if a particle moves along a space curve with speed v, it experiences a component acceleration $\frac{dv}{dt}$ along the curve and a component acceleration v^2/ρ along the principal normal to the curve.

In addition to the two perpendicular vectors \mathbf{u} and \mathbf{n} at a point P of a curve, a third vector, known as the *binormal vector*, is defined by the equation

$$\mathbf{b} = \mathbf{u} \times \mathbf{n},\tag{52}$$

and hence is a unit vector perpendicular to both \mathbf{u} and \mathbf{n} . Thus with each point P moving along a space curve we may associate a moving and rotating triad of mutually orthogonal unit vectors. For a *plane curve*, the tangent vector \mathbf{u} and the principal normal vector \mathbf{n} lie in the plane of the curve, whereas the binormal vector is a constant unit vector perpendicular to that plane.

If we differentiate (52) and make use of (49), there follows

$$\frac{d\mathbf{b}}{ds} = \frac{d\mathbf{u}}{ds} \times \mathbf{n} + \mathbf{u} \times \frac{d\mathbf{n}}{ds} = \mathbf{u} \times \frac{d\mathbf{n}}{ds}.$$

Thus $\frac{d\mathbf{b}}{ds}$ is perpendicular to \mathbf{u} and, since \mathbf{b} is a unit vector, $\frac{d\mathbf{b}}{ds}$ is also perpendicular to \mathbf{b} . Hence $\frac{d\mathbf{b}}{ds}$ is a scalar multiple of \mathbf{n} and can be written in the form

$$\frac{d\mathbf{b}}{ds} = -\frac{1}{\tau}\,\mathbf{n}.\tag{53}$$

The scalar $1/\tau$ is called the *torsion* of the curve, the negative sign having been introduced so that the torsion is positive when the vector triad rotates in a right-handed sense about the tangent as it progresses along the curve. The length $|\tau|$ is called the *radius of torsion*.

Equations (49) and (53) give the derivatives of **u** and **b** with respect to arc length. To calculate $\frac{d\mathbf{n}}{ds}$, we write $\mathbf{n} = \mathbf{b} \times \mathbf{u}$ and differentiate, making use of (49) and (53), to obtain

$$\frac{d\mathbf{n}}{ds} = \frac{d\mathbf{b}}{ds} \times \mathbf{u} + \mathbf{b} \times \frac{d\mathbf{u}}{ds} = -\frac{1}{\tau} (\mathbf{n} \times \mathbf{u}) + \frac{1}{\rho} (\mathbf{b} \times \mathbf{n}),$$
$$\frac{d\mathbf{n}}{ds} = \frac{1}{\tau} \mathbf{b} - \frac{1}{\rho} \mathbf{u}.$$
(54)

or

Equations (49), (53), and (54) are known as Frenet's formulas.

sec. 6.7 | The gradient vector

If we form the dot product of **b** with the two sides of (54) and use (52) and (49), we obtain

$$\frac{1}{\tau} = \mathbf{b} \cdot \frac{d\mathbf{n}}{ds} = \mathbf{u} \times \mathbf{n} \cdot \frac{d\mathbf{n}}{ds} = \rho^2 \left(\mathbf{u} \times \frac{d\mathbf{u}}{ds} \cdot \frac{d^2\mathbf{u}}{ds^2} \right).$$

This result can be written in the determinant form

$$\frac{1}{\tau} = \rho^2 \begin{vmatrix} \frac{dx}{ds} & \frac{dy}{ds} & \frac{dz}{ds} \\ \frac{d^2x}{ds^2} & \frac{d^2y}{ds^2} & \frac{d^2z}{ds^2} \\ \frac{d^3x}{ds^3} & \frac{d^3y}{ds^3} & \frac{d^3z}{ds^3} \end{vmatrix},$$
(55)

where ρ is defined by Equation (50).

Formulas expressing ρ and τ in terms of a general parameter *t*, rather than arc length *s*, are deduced in Problems 33 to 36.

The plane determined at a point P_0 on a curve by **u** and **n** is called the *osculating plane*, that determined by **n** and **b** the *normal plane*, and that determined by **u** and **b** the *rectifying plane*. Hence, if \mathbf{r}_0 is the position vector to P_0 and $\mathbf{r} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$, the equations of these planes are

$$\mathbf{b}_0 \cdot (\mathbf{r} - \mathbf{r}_0) = 0, \qquad \mathbf{u}_0 \cdot (\mathbf{r} - \mathbf{r}_0) = 0, \qquad \mathbf{n}_0 \cdot (\mathbf{r} - \mathbf{r}_0) = 0,$$

respectively, where \mathbf{b}_0 , \mathbf{u}_0 , and \mathbf{n}_0 are evaluated at P_0 .

6.7. The gradient vector. If φ is a scalar function of the coordinates x, y, and z, the values of the partial derivatives $\frac{\partial \varphi}{\partial x}$, $\frac{\partial \varphi}{\partial y}$, and $\frac{\partial \varphi}{\partial z}$ at a point represent the rates of change of φ with respect to distance in the x, y, and z directions, respectively, at that point. If we consider the vector

$$\mathbf{V} = \mathbf{i} \, \frac{\partial \varphi}{\partial x} + \mathbf{j} \, \frac{\partial \varphi}{\partial y} + \mathbf{k} \, \frac{\partial \varphi}{\partial z} \,, \tag{56}$$

we see that the scalar components of this vector in the x, y, and z directions at any point P are then exactly the respective rates at which φ is changing with respect to distance in those directions at P. To determine the component of V in any direction at P, we consider the position vector **r** from the origin to P and indicate a differential displacement from P in any chosen direction by d**r**,

$$d\mathbf{r} = \mathbf{i} \, dx + \mathbf{j} \, dy + \mathbf{k} \, dz. \tag{57}$$

The scalar component of V in this direction is then obtained as the dot product of V and the unit vector $\mathbf{u} = \frac{d\mathbf{r}}{ds}$, where $ds = |d\mathbf{r}|$, and hence has the value

$$\mathbf{V} \cdot \mathbf{u} = \frac{\partial \varphi}{\partial x} \frac{dx}{ds} + \frac{\partial \varphi}{\partial y} \frac{dy}{ds} + \frac{\partial \varphi}{\partial z} \frac{dz}{ds} = \frac{d\varphi}{ds} \,. \tag{58}$$

Thus we see that the component of V in *any* direction is the rate of change of φ with respect to distance in that direction. The vector V is called the *gradient* of φ , and we write

grad
$$\varphi = \mathbf{i} \frac{\partial \varphi}{\partial x} + \mathbf{j} \frac{\partial \varphi}{\partial y} + \mathbf{k} \frac{\partial \varphi}{\partial z}$$
. (59)

Hence, if r represents the position vector to a point P and if dr represents differential displacement of length ds from P, Equation (58) gives

$$\frac{d\mathbf{r}}{ds} \cdot \operatorname{grad} \varphi = \frac{d\varphi}{ds}, \qquad (60a)$$

or, in terms of differentials,

$$d\mathbf{r} \cdot \operatorname{grad} \varphi = d\varphi. \tag{60b}$$

Since the component of the vector grad φ in any direction is the derivative of φ in that direction, it follows that grad φ must point in the direction in which the derivative of φ is numerically greatest, and must have a length numerically equal to that maximum derivative. In particular, grad φ must have no component in the directions in which the derivative of φ is zero, and hence must be perpendicular to the surfaces $\varphi(x,y,z) = \text{constant}$.

In illustration, if the temperature at any point P(x,y,z) is given by

$$T = C(x^2 + y^2 + z^2),$$

the gradient vector has the form

grad
$$T = 2C(x \mathbf{i} + y \mathbf{j} + z \mathbf{k}).$$

This vector is clearly a radial vector pointed outward from the origin, and hence is perpendicular to the spherical equithermal surfaces

 $x^2 + y^2 + z^2 = \text{constant.}$

The magnitude of grad T is given by

|grad T| = 2Cr,

where $r = \sqrt{x^2 + y^2 + z^2}$, and, since $T = Cr^2$, we have

$$|\text{grad } T| = \frac{dT}{dr}.$$

This result is, of course, in accordance with the fact that the prescribed temperature changes most rapidly in the r direction. The rate of change of the

temperature at P with respect to distance in a direction specified by the direction cosines (l,m,n) is given by

$$(l \mathbf{i} + m \mathbf{j} + n \mathbf{k}) \cdot \text{grad } T = 2C(lx + my + nz).$$

6.8. The vector operator ∇ . Equation (59) is conveniently abbreviated in the form

$$\operatorname{grad} \varphi = \nabla \varphi, \tag{61}$$

where the symbol ∇ , called *del*, is the vector operator

$$\nabla = \mathbf{i} \frac{\partial}{\partial x} + \mathbf{j} \frac{\partial}{\partial y} + \mathbf{k} \frac{\partial}{\partial z}.$$
 (62)

That is, we adopt the convention

$$\nabla \varphi = \left(\mathbf{i}\frac{\partial}{\partial x} + \mathbf{j}\frac{\partial}{\partial y} + \mathbf{k}\frac{\partial}{\partial z}\right)\varphi = \mathbf{i}\frac{\partial \varphi}{\partial x} + \mathbf{j}\frac{\partial \varphi}{\partial y} + \mathbf{k}\frac{\partial \varphi}{\partial z}, \quad (63)$$

where φ is any differentiable scalar function of x, y, and z. In this notation, Equations (60) become

$$\frac{d\mathbf{r}}{ds} \cdot \nabla \varphi = \frac{d\varphi}{ds}, \qquad d\mathbf{r} \cdot \nabla \varphi = d\varphi.$$
(64)

If F is a vector function of x, y, and z, we may define the dot and cross products of the operator ∇ into F. These products are called the *divergence* and *curl* of F, respectively, and are of the form

$$\nabla \cdot \mathbf{F} = \mathbf{i} \cdot \frac{\partial \mathbf{F}}{\partial x} + \mathbf{j} \cdot \frac{\partial \mathbf{F}}{\partial y} + \mathbf{k} \cdot \frac{\partial \mathbf{F}}{\partial z} = \operatorname{div} \mathbf{F},$$
 (65)

$$\nabla \times \mathbf{F} = \mathbf{i} \times \frac{\partial \mathbf{F}}{\partial x} + \mathbf{j} \times \frac{\partial \mathbf{F}}{\partial y} + \mathbf{k} \times \frac{\partial \mathbf{F}}{\partial z} = \operatorname{curl} \mathbf{F}.$$
 (66)

If $\mathbf{F} = F_x \mathbf{i} + F_y \mathbf{j} + F_z \mathbf{k}$, we then have

$$\nabla \cdot \mathbf{F} = \operatorname{div} \mathbf{F} = \frac{\partial F_x}{\partial x} + \frac{\partial F_y}{\partial y} + \frac{\partial F_z}{\partial z}$$
(67)

$$\nabla \times \mathbf{F} = \operatorname{curl} \mathbf{F} = \begin{vmatrix} \mathbf{i} & \mathbf{j} & \mathbf{k} \\ \frac{\partial}{\partial x} & \frac{\partial}{\partial y} & \frac{\partial}{\partial z} \\ F_x & F_y & F_z \end{vmatrix}$$
 (68a)

and

or

$$\nabla \times \mathbf{F} = \mathbf{i} \left(\frac{\partial F_z}{\partial y} - \frac{\partial F_y}{\partial z} \right) + \mathbf{j} \left(\frac{\partial F_x}{\partial z} - \frac{\partial F_z}{\partial x} \right) + \mathbf{k} \left(\frac{\partial F_y}{\partial x} - \frac{\partial F_x}{\partial y} \right).$$
(68b)

Since both ∇ and F are vectorial, the combination ∇ F is here left undefined.
The total differential of a vector function \mathbf{F} is given by

$$d\mathbf{F} = \frac{\partial \mathbf{F}}{\partial x} \, dx + \frac{\partial \mathbf{F}}{\partial y} \, dy + \frac{\partial \mathbf{F}}{\partial z} \, dz$$

or, in operational form,

$$d\mathbf{F} = \left(dx \frac{\partial}{\partial x} + dy \frac{\partial}{\partial y} + dz \frac{\partial}{\partial z}\right)\mathbf{F}.$$

Since the operator is the dot product of the differential vector $d\mathbf{r} = \mathbf{i} \, dx + \mathbf{j} \, dy$ + $\mathbf{k} \, dz$ and the operator ∇ , we may write this relation in the form

$$d\mathbf{F} = (d\mathbf{r} \cdot \nabla)\mathbf{F}.$$
 (69)

The derivative of \mathbf{F} in the direction of $d\mathbf{r}$ is then

$$\frac{d\mathbf{F}}{ds} = \left(\frac{d\mathbf{r}}{ds} \cdot \boldsymbol{\nabla}\right) \mathbf{F}.$$
(70)

More generally, if \mathbf{u} is any unit vector, then the derivative of \mathbf{F} in the direction of \mathbf{u} is given by

$$\frac{d\mathbf{F}}{ds} = (\mathbf{u} \cdot \nabla)\mathbf{F}.$$
(71)

Since ∇F is undefined, the parentheses in Equation (71) are generally omitted and we follow the convention

$$\mathbf{v} \cdot \boldsymbol{\nabla} \mathbf{F} \equiv (\mathbf{v} \cdot \boldsymbol{\nabla}) \mathbf{F},\tag{72}$$

where, if v is any vector with components v_x , v_y , and v_z , the product $\mathbf{v} \cdot \nabla$ is the scalar operator

$$\mathbf{v} \cdot \nabla = v_x \frac{\partial}{\partial x} + v_y \frac{\partial}{\partial y} + v_z \frac{\partial}{\partial z}.$$
 (73)

6.9. Differentiation formulas. The following identities are of frequent use

$$\nabla \cdot \varphi \mathbf{u} = \varphi \nabla \cdot \mathbf{u} + \mathbf{u} \cdot \nabla \varphi, \tag{74a}$$

$$\boldsymbol{\nabla} \times \boldsymbol{\varphi} \mathbf{u} = \boldsymbol{\varphi} \boldsymbol{\nabla} \times \mathbf{u} + \boldsymbol{\nabla} \boldsymbol{\varphi} \times \mathbf{u}, \tag{74b}$$

$$\nabla \cdot \mathbf{u} \times \mathbf{v} = \mathbf{v} \cdot \nabla \times \mathbf{u} - \mathbf{u} \cdot \nabla \times \mathbf{v}, \tag{74c}$$

$$\nabla \times (\mathbf{u} \times \mathbf{v}) = \mathbf{v} \cdot \nabla \mathbf{u} - \mathbf{u} \cdot \nabla \mathbf{v} + \mathbf{u} (\nabla \cdot \mathbf{v}) - \mathbf{v} (\nabla \cdot \mathbf{u}), \tag{74d}$$

$$\nabla(\mathbf{u} \cdot \mathbf{v}) = \mathbf{u} \cdot \nabla \mathbf{v} + \mathbf{v} \cdot \nabla \mathbf{u} + \mathbf{u} \times (\nabla \times \mathbf{v}) + \mathbf{v} \times (\nabla \times \mathbf{u}), \quad (74e)$$

$$\nabla \times (\nabla \varphi) = \operatorname{curl} \operatorname{grad} \varphi = \mathbf{0}, \tag{74f}$$

$$\nabla \cdot (\nabla \times \mathbf{u}) = \operatorname{div} \operatorname{curl} \mathbf{u} = 0, \tag{74g}$$

$$\nabla \times (\nabla \times \mathbf{u}) = \operatorname{curl} \operatorname{curl} \mathbf{u} = \nabla (\nabla \cdot \mathbf{u}) - \nabla \cdot \nabla \mathbf{u}$$

$$= \operatorname{grad} \operatorname{div} \mathbf{u} - \nabla^2 \mathbf{u}, \tag{74h}$$

$$\nabla \cdot (\nabla \varphi_1 \times \nabla \varphi_2) = 0. \tag{74i}$$

In these formulas **u** and **v** are arbitrary vectors and φ , φ_1 , and φ_2 arbitrary scalars for which the indicated derivatives exist. In formulas (74f) and (74g), we assume in addition that φ and **u** have continuous partial derivatives of the second order.

These identities can all be verified by direct expansion. In some cases, however, the proof can be considerably shortened by making use of the fact that the operator ∇ may be substituted for a vector in any vector identity provided that it operates on the same factors in all terms.

It should be noticed that the operator ∇ is a *distributive* operator, in the sense that the equations

$$\nabla(\varphi_1 + \varphi_2) = (\nabla\varphi_1) + (\nabla\varphi_2), \qquad \nabla \cdot (\mathbf{u} + \mathbf{v}) = (\nabla \cdot \mathbf{u}) + (\nabla \cdot \mathbf{v}),$$
$$\nabla \times (\mathbf{u} + \mathbf{v}) = (\nabla \times \mathbf{u}) + (\nabla \times \mathbf{v})$$

are true.

Also, direct expansion shows that the identity

$$(\mathbf{u} \cdot \nabla)\varphi = \mathbf{u} \cdot (\nabla \varphi) \tag{75}$$

is valid. For this reason the parentheses are conventionally omitted in this case and the product is written in the form $\mathbf{u} \cdot \nabla \varphi$. Further, as has been noted, no parentheses are needed in a product such as $\mathbf{u} \cdot \nabla \mathbf{v}$, since $\nabla \mathbf{v}$ is undefined and hence this notation can only mean $(\mathbf{u} \cdot \nabla)\mathbf{v}$.

In general, we adopt the convention that in a multiple product ∇ operates on all terms to its right which are not separated from it by intervening parentheses or brackets. If this is not convenient we will here indicate a term which is to be treated as a constant in the differentiation, and hence is *not* to be operated on by ∇ , by underlining that term. Thus, we use the notations $\nabla \cdot (\varphi \mathbf{u})$ and $\nabla \times (\varphi \mathbf{u})$ to indicate that \mathbf{u} is to be treated as a constant vector in the differentiation implied in ∇ , so that ∇ operates only on φ . It is readily verified that we then have

 $\nabla \cdot (\varphi \underline{\mathbf{u}}) = \mathbf{u} \cdot \nabla \varphi, \qquad \nabla \times (\varphi \underline{\mathbf{u}}) = -\mathbf{u} \times \nabla \varphi. \tag{76}$

From the nature of ∇ we see that

$$\nabla \cdot (\varphi \mathbf{u}) = \nabla \cdot (\varphi \mathbf{u}) + \nabla \cdot (\varphi \underline{\mathbf{u}}),$$
$$\nabla \times (\varphi \mathbf{u}) = \nabla \times (\varphi \mathbf{u}) + \nabla \times (\varphi \underline{\mathbf{u}}).$$

That is, the derivative of the product is the sum of the two derivatives obtained by holding one of the factors constant and allowing the other to be operated on by ∇ . These considerations lead directly to formulas (74a) and (74b).

Formula (74c) is obtained in a similar way if we first notice that

 $\nabla \cdot \mathbf{u} \times \mathbf{c} = -\nabla \cdot \mathbf{c} \times \mathbf{u} = \mathbf{c} \cdot \nabla \times \mathbf{u}$

when c is a constant vector.

We see that if the product

$$\nabla \times (\mathbf{u} \times \mathbf{v})$$

is formally expanded by substituting ∇ , u, and v for a, b, and c in Equation (34), the result

$$(\nabla \cdot \mathbf{v})\mathbf{u} - (\mathbf{u} \cdot \nabla)\mathbf{v}$$

is not necessarily equivalent to the given product, since there ∇ operates on both **u** and **v**, whereas in the two terms of the formal expansion, ∇ operates only on **v**. However, if **u** is treated as a constant vector in the given product, the expansion is valid. That is, the expansion represents $\nabla \times (\underline{\mathbf{u}} \times \mathbf{v})$. By adding to this expansion the corresponding expression for $\nabla \times (\mathbf{u} \times \underline{\mathbf{v}})$, we obtain formula (74d).

Formula (74e) may be established by an indirect method in which we first obtain the results

$$\mathbf{u} \times (\nabla \times \mathbf{v}) = \nabla (\underline{\mathbf{u}} \cdot \mathbf{v}) - \mathbf{u} \cdot \nabla \mathbf{v},$$
$$\mathbf{v} \times (\nabla \times \mathbf{u}) = \nabla (\mathbf{u} \cdot \underline{\mathbf{v}}) - \mathbf{v} \cdot \nabla \mathbf{u},$$

and then form the sum of these equations.

Equations (74f) and (74g) are easily established by direct calculation. Thus, to establish (74f), we write

$$\nabla \times \nabla \varphi = \begin{vmatrix} \mathbf{i} & \mathbf{j} & \mathbf{k} \\ \frac{\partial}{\partial x} & \frac{\partial}{\partial y} & \frac{\partial}{\partial z} \\ \frac{\partial}{\partial x} & \frac{\partial}{\partial y} & \frac{\partial}{\partial z} \\ \frac{\partial}{\partial x} & \frac{\partial}{\partial y} & \frac{\partial}{\partial z} \end{vmatrix} = \mathbf{i} \left(\frac{\partial^2 \varphi}{\partial y \partial z} - \frac{\partial^2 \varphi}{\partial z \partial y} \right) + \mathbf{j} \left(\frac{\partial^2 \varphi}{\partial z \partial x} - \frac{\partial^2 \varphi}{\partial x \partial z} \right) \\ + \mathbf{k} \left(\frac{\partial^2 \varphi}{\partial x \partial y} - \frac{\partial^2 \varphi}{\partial y \partial x} \right).$$

If the order of differentiation is immaterial, all components on the right vanish. This situation exists, in particular, *if the second crossed partial deriva*tives of φ are continuous. Similarly, Equation (74g) is readily shown to be valid if the corresponding derivatives of the components of **u** are continuous.

Equation (74h) is obtained by replacing $\mathbf{a}, \mathbf{b}, \mathbf{c}$, by $\nabla, \nabla, \mathbf{u}$ in Equation (34). The scalar operator

$$\boldsymbol{\nabla}^2 = \boldsymbol{\nabla} \cdot \boldsymbol{\nabla} = \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2}$$
(77)

is of frequent occurrence, and is known as the *Laplacian* operator. Equation (74i) is established by making use of (74c) and (74f).

If \mathbf{r} is the position vector

$$\mathbf{r} = x \, \mathbf{i} + y \, \mathbf{j} + z \, \mathbf{k},$$

we find by direct calculation that

$$\nabla \cdot \mathbf{r} = 3, \qquad \nabla \times \mathbf{r} = \mathbf{0}, \tag{78a,b}$$

and also that

$$\mathbf{u} \cdot \nabla \mathbf{r} = \mathbf{u},\tag{79}$$

where **u** is any vector.

6.10. Line integrals. Let F represent a vector function of position in a space of three dimensions and let C represent a space curve, each point P of which is specified by its position vector r from an origin O to the point P. With each point P of the curve we associate a differential distance vector dr. Then, since

$$d\mathbf{r} = \frac{d\mathbf{r}}{ds}\,ds = \mathbf{u}\,ds,\tag{80}$$

where **u** is a unit tangent vector to C at P, the differential vector d**r** has as its length the differential arc length ds along C at P and as its direction the direction of the curve at P. The scalar differential $\mathbf{F} \cdot d\mathbf{r} = (\mathbf{F} \cdot \mathbf{u}) ds$ is then numerically equal to the product of the component of **F** in the direction of C at P and the differential length ds. Finally, the integral

$$\int_{C} \mathbf{F} \cdot d\mathbf{r} = \int_{C} \left(\mathbf{F} \cdot \mathbf{u} \right) ds \tag{81}$$

taken along the curve between two specified points P_0 and P_1 , is known as the *line integral* of F along C.

In particular, if F represents *force* acting on a particle, the line integral (81) clearly represents the *work* done by the force in moving the particle along C from P_0 to P_1 .

We assume here, and in the sequel, that any curve C along which a line integral is to be evaluated is made up of a finite number of arcs, along each of which the tangent vector **u** not only exists, but also varies continuously with s, so that the integrand $\mathbf{F} \cdot \mathbf{u}$ is continuous when **F** is continuous. Such a curve is said to be *piecewise smooth*. In particular, when **u** is continuous over all of C, we say that C is a smooth curve.

If we write

$$\mathbf{F} = P(x,y,z) \mathbf{i} + Q(x,y,z) \mathbf{j} + R(x,y,z) \mathbf{k}$$

$$\mathbf{r} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$$
(82)

the line integral (81) takes the form

$$\int_C \mathbf{F} \cdot d\mathbf{r} = \int_C (P \, dx + Q \, dy + R \, dz). \tag{83}$$

On each arc of the curve C, the variables x, y, and z, as well as the corresponding differentials, are to be expressed in terms of an appropriate single variable, and the limits of the integral must be determined accordingly in terms of the initial and terminal points of C. It should be pointed out that the line integral $\int_C \mathbf{F} \cdot d\mathbf{r}$, taken along the portion of a curve between two points A and B, may be defined more fundamentally as the limit of a sum of the form

$$\sum_{k=0}^{n} \mathbf{F}_{k} \cdot (\overrightarrow{\Delta r})_{k} = \mathbf{F}_{0} \cdot (\overrightarrow{\Delta r})_{0} + \mathbf{F}_{1} \cdot (\overrightarrow{\Delta r})_{1} + \cdots + \mathbf{F}_{n} \cdot (\overrightarrow{\Delta r})_{n},$$

where F_0, F_1, \ldots, F_n , are the values of F at A and at n points P_1, P_2, \ldots, P_n , arbitrarily chosen along the arc of C between A and B, and where $(\Delta r)_0 = \overrightarrow{AP_1}$, $(\overrightarrow{\Delta r})_1 = \overrightarrow{P_1P_2}, \ldots, (\overrightarrow{\Delta r})_n = \overrightarrow{P_nB}$. The limit is taken as $n \to \infty$ in such a way that all the chords tend to zero. This definition is analogous to the definition of the ordinary integral. When F is continuous and C is piecewise smooth, the integral so defined can be evaluated by the method described above.

As an illustration, we take

$$\mathbf{F} = yz \, \mathbf{i} + xy \, \mathbf{j} + xz \, \mathbf{k}$$

and calculate the line integral of F from (0,0,0) to (1,1,1) along the path C consisting of the curve $x = y^2$, z = 0 in the xy plane from (0,0,0) to (1,1,0) and the line x = 1, y = 1 perpendicular to the xy plane from (1,1,0) to (1,1,1). In the first part of the path, C_1 , we have

$$x = y^2$$
, $z = 0$; $dx = 2y dy$, $dz = 0$,

and hence

$$\mathbf{F} \cdot d\mathbf{r} = yz \, dx + xy \, dy + xz \, dz = y^3 \, dy.$$

Thus there follows

$$\int_{C_1} \mathbf{F} \cdot d\mathbf{r} = \int_0^1 y^3 \, dy = \frac{1}{4}.$$

In the second part of the path, C_2 , we have

$$x = 1, y = 1; dx = dy = 0,$$

and hence

$$\mathbf{F} \boldsymbol{\cdot} d\mathbf{r} = z \, dz.$$

Thus there follows

$$\int_{C_2} \mathbf{F} \cdot d\mathbf{r} = \int_0^1 z \, dz = \frac{1}{2} \, ,$$

and, finally,

$$\int_C \mathbf{F} \cdot d\mathbf{r} = \frac{1}{4} + \frac{1}{2} = \frac{3}{4}.$$

If we integrate along the path C' consisting of the straight line x = y = z, directly from (0,0,0) to (1,1,1), we have, on C',

$$x = z, y = z;$$
 $dx = dz, dy = dz$

and hence

$$\int_{C'} \mathbf{F} \cdot d\mathbf{r} = \int_0^1 3z^2 \, dz = 1.$$

In this example the value of the line integral between the given points depends upon the *path* chosen. However, in certain cases this is not so. For if the expression P dx + Q dy + R dz is the differential of a function $\varphi(x,y,z)$,

$$P \, dx + Q \, dy + R \, dz = d\varphi, \tag{84}$$

then the integral (83) becomes merely

$$\int_{C} \mathbf{F} \cdot d\mathbf{r} = \int_{C} d\varphi = \varphi(P_{1}) - \varphi(P_{0}), \qquad (85)$$

where P_1 and P_0 are the end points of C. Thus the line integral depends only on the location of the end points *if the function* φ *is single-valued*, so that the value approached by φ at a point in space does not depend upon the manner of approach.

Since the differential of φ is also given by

$$d\varphi = \frac{\partial \varphi}{\partial x} dx + \frac{\partial \varphi}{\partial y} dy + \frac{\partial \varphi}{\partial z} dz, \qquad (86)$$

there follows, by comparing (84) and (86) and noticing that x, y, and z are independent variables,

$$\frac{\partial \varphi}{\partial x} = P, \qquad \frac{\partial \varphi}{\partial y} = Q, \qquad \frac{\partial \varphi}{\partial z} = R.$$
 (87)

By appropriately differentiating the first two equations, we find also that

$$\frac{\partial^2 \varphi}{\partial y \, \partial x} = \frac{\partial P}{\partial y}, \qquad \frac{\partial^2 \varphi}{\partial x \, \partial y} = \frac{\partial Q}{\partial x},$$

and hence, if the derivatives involved are continuous, we conclude that P and Q must satisfy the condition

$$\frac{\partial P}{\partial y} = \frac{\partial Q}{\partial x} \,. \tag{88a}$$

In a similar way we find that the two conditions

$$\frac{\partial P}{\partial z} = \frac{\partial R}{\partial x}, \qquad \frac{\partial Q}{\partial z} = \frac{\partial R}{\partial y}$$
 (88b,c)

must also be satisfied.

Hence, if a function φ exists so that Equation (84) is true, that is, if $\mathbf{F} \cdot d\mathbf{r} = P \, dx + Q \, dy + R \, dz$ is an exact differential, the functions P, Q, and R must satisfy (88a,b,c). Conversely, when P, Q, and R and their first partial derivatives are continuous in a region, it can be shown that if these conditions are satisfied, a function φ exists for which (84) is true in that region.

We notice next that since

$$\nabla \times \mathbf{F} = \mathbf{i} \left(\frac{\partial R}{\partial y} - \frac{\partial Q}{\partial z} \right) + \mathbf{j} \left(\frac{\partial P}{\partial z} - \frac{\partial R}{\partial x} \right) + \mathbf{k} \left(\frac{\partial Q}{\partial x} - \frac{\partial P}{\partial y} \right),$$
 (89)

the satisfaction of (88a,b,c) is equivalent to the vanishing of $\nabla \times \mathbf{F}$. Thus we conclude that if \mathbf{F} is continuously differentiable in a region \mathscr{R} and if $\nabla \times \mathbf{F} = \mathbf{0}$ everywhere in \mathscr{R} , then a scalar function φ exists such that $d\varphi = \mathbf{F} \cdot d\mathbf{r}$. Further, if C is any curve lying in \mathscr{R} and joining the points P_0 and P_1 and if φ is single-valued, then

$$\int_C \mathbf{F} \cdot d\mathbf{r} = \varphi(P_1) - \varphi(P_0).$$

To exclude the possibility that φ be multiple-valued, we henceforth require that the region \mathscr{R} have the property that any closed curve or surface lying in \mathscr{R} can be shrunk continuously to a point without passing outside of \mathscr{R} . Such a region is called a *simply connected region* or, more briefly, a *simple region*. Thus the plane annular region included between two concentric circles is *not* a simple region, since a curve in the annulus surrounding the inner circle cannot be shrunk to a point without passing outside the annulus. To show rigorously the relationship between simplicity of \mathscr{R} and single-valuedness of φ is beyond the scope of this work. (See also Section 6.16.) It may be remarked, however, that the proof depends upon the fact that by restricting \mathscr{R} to be a simple region we *exclude* the possibility that a closed curve or surface lying in \mathscr{R} may surround points or curves not included in \mathscr{R} .

For any curve C joining the points P_0 and P_1 in a simple region \mathcal{R} throughout which F is continuously differentiable and also $\nabla \times F = 0$, we may write

$$\int_C \mathbf{F} \cdot d\mathbf{r} = \int_{P_0}^{P_1} \mathbf{F} \cdot d\mathbf{r} = \int_{P_0}^{P_1} d\varphi = \varphi(P_1) - \varphi(P_0). \tag{90}$$

That is, in such a case the line integral from P_0 to P_1 is independent of the path. In particular, for a closed path the initial and terminal points coincide and the line integral vanishes. We denote the line integral around a closed curve by the symbol \oint and write

$$\oint_C \mathbf{F} \cdot d\mathbf{r} = 0. \tag{91}$$

If the conditions stated are not satisfied, the integral around C may or may not vanish. The value of that integral is called the *circulation* of F around C.

When a closed curve C lies on a surface S, one side of which is regarded as its positive side, it is conventional to define the *positive direction* on C (that is, the direction of $\mathbf{u} = \frac{d\mathbf{r}}{ds}$) as the direction along which an observer, traveling on the positive side of S, would move while keeping the area enclosed by C to his *left*.

6.11. The potential function. Since the existence of a function φ such that

$$d\varphi = \mathbf{F} \cdot d\mathbf{r} \tag{92}$$

at all points of a region is guaranteed by the condition $\nabla \times \mathbf{F} = \mathbf{0}$, and since Equation (64) gives also

$$d\varphi = \nabla \varphi \cdot d\mathbf{r},$$

there follows by subtraction

$$(\mathbf{F} - \nabla \varphi) \cdot d\mathbf{r} = 0.$$

Hence the vector $\mathbf{F} - \nabla \varphi$ must be perpendicular to $d\mathbf{r}$; but since the direction of dr is arbitrary, we conclude that $\mathbf{F} = \nabla \varphi$ must vanish,

$$\mathbf{F} = \mathbf{V}\boldsymbol{\varphi}.\tag{93}$$

Thus, if $\nabla \times \mathbf{F} = \mathbf{0}$ in a region, then \mathbf{F} is the gradient of a scalar function φ in that region. Conversely, Equation (74f) states that if $\mathbf{F} = \nabla \varphi$ and if φ has continuous second partial derivatives, then $\nabla \times \mathbf{F} = \mathbf{0}$.

The function φ defined by Equation (93) is known as the *potential* of **F**. If F represents force, the *negative* of φ is called the *potential energy* associated with F. When such a function φ exists, the force F is said to be *conservative*, since in this case the total work done in moving a particle around a closed contour in an appropriate region is zero.

To illustrate the determination of the potential, we consider the force field

$$\mathbf{F} = y^2 \,\mathbf{i} + 2(xy+z) \,\mathbf{j} + 2y \,\mathbf{k}.$$
 (94)

It is readily verified that $\nabla \times \mathbf{F} = \mathbf{0}$ everywhere. Hence the force is conservative and the work done by F in moving a particle between two points is independent of the path. Further, a scalar potential function φ exists such that

$$\mathbf{F} \cdot d\mathbf{r} = d\varphi \tag{95}$$

and also

$$\mathbf{F} = \boldsymbol{\nabla} \boldsymbol{\varphi}.\tag{96}$$

To determine φ , we write Equation (95) in the form

- - -

$$d\varphi = y^2 \, dx + 2(xy+z) \, dy + 2y \, dz.$$

Comparison of this equation with (86) shows that φ must satisfy the three conditions

$$\frac{\partial \varphi}{\partial x} = y^2, \quad \frac{\partial \varphi}{\partial y} = 2(xy+z), \quad \frac{\partial \varphi}{\partial z} = 2y.$$
 (97a,b,c)

If we integrate (97a), holding y and z constant, we obtain

$$\varphi = xy^2 + f(y,z), \tag{98}$$

- - -

where f(y,z) is an arbitrary function of y and z. In order that (98) satisfy (97b), we must then have

$$2xy + \frac{\partial f(y,z)}{\partial y} = 2xy + 2z$$
 or $\frac{\partial f(y,z)}{\partial y} = 2z$.

Integration gives

$$f(y,z) = 2yz + g(z),$$
 (99)

where g(z) is an arbitrary function of z. Introducing (98) and (99) into (97c), we then obtain the final condition

$$2y + \frac{dg(z)}{dz} = 2y,$$

$$g(z) = c,$$
 (100)

from which there follows

where e is an arbitrary constant. Thus, we have finally

$$\varphi = xy^2 + 2yz + c, \tag{101}$$

where the constant c can be chosen arbitrarily, so that the potential at a convenient reference point is zero. The surfaces $\varphi = \text{constant}$ are called *equipotential surfaces*. We see that the work done in moving a particle from a point on the surface $\varphi = c_1$ to a point on the surface $\varphi = c_2$ is merely $c_2 - c_1$.

To illustrate the existence of unusual cases, we notice that if

$$\mathbf{F} = -\frac{y}{x^2 + y^2}\mathbf{i} + \frac{x}{x^2 + y^2}\mathbf{j},$$

there then follows

$$\mathbf{F} \cdot d\mathbf{r} = \frac{x \, dy - y \, dx}{x^2 + y^2} = d \tan^{-1} \frac{y}{x} = d\theta,$$

where θ is the angle in cylindrical coordinates (r, θ, z) . Thus we have

 $\mathbf{F} = \nabla \theta$,

and hence, from (74f), there follows

 $\nabla \times \mathbf{F} = \mathbf{0}.$

Hence it might appear that the circulation of F around any closed curve C would be zero. However, we notice that F is discontinuous at points on the z axis, in the sense that the magnitude of F,

$$|\mathbf{F}| = \frac{1}{\sqrt{x^2 + y^2}},$$

becomes infinite as $(x,y) \rightarrow (0,0)$. Thus $\nabla \times F$ is not defined along the z axis. At all other points, however, $\nabla \times F$ is defined and is zero. Any closed curve which does not surround the z axis can be included in a simple region where F is continuously differentiable and $\nabla \times F = 0$ and hence the circulation around any such path must vanish. To verify this statement, we consider an arbitrary closed path C beginning and ending at a point P. Since

$$\int_C \mathbf{F} \cdot d\mathbf{r} = \int_C d\theta,$$

we see that if C does not enclose the z axis, the angle θ may alternately increase and decrease along C until a maximum value is reached, after which θ eventually returns to its original value at P and the total net increase in θ is zero. That is, over such a closed path the circulation is zero.

However, if the path C surrounds the z axis once, and the path is described in the positive (counterclockwise) direction, the angle θ experiences a net increase of 2π when the complete path is described, and the circulation is 2π . Similarly, the circulation along a closed curve encircling the z axis k times in the positive direction is $2k\pi$.

If we delete the region in the neighborhood of the z axis by cutting out an infinitely extended right circular cylinder of arbitrarily small radius, with its axis along the z axis, the remainder of space is a region \mathscr{R}' inside which F and $\nabla \times F$ possess the required properties. However, the region \mathscr{R}' is clearly not a *simple* region. Those closed curves in \mathscr{R}' which *can* be shrunk to a point without passing outside \mathscr{R}' are said to be *reducible*. Thus any closed curve in \mathscr{R}' which does not enclose the z axis is reducible, and we see that in the present example the circulation around any *reducible* closed curve in \mathscr{R}' is zero.

It can be shown, more generally, that if **F** is piecewise differentiable and $\nabla \times \mathbf{F} = \mathbf{0}$ in *any* region, whether simply connected or not, the circulation of **F** around any *reducible* closed curve in that region is zero.

6.12. Surface integrals. A surface S in three dimensions is usually specified by an equation of the form

$$g(x,y,z) = 0.$$
 (102)

If we think of S as one of the family of surfaces g(x,y,z) = c, as c takes on all possible values, we see that at any point the vector ∇g is normal to the particular surface of the family which passes through that point, and hence, in particular, ∇g is normal to S at points of S. At such points ∇g can then be expressed as the product of a scalar and a *unit vector* **n** normal to S. We can thus write

$$\mathbf{u} = \pm \frac{\nabla g}{|\nabla g|},\tag{103}$$

where the choice of the ambiguous sign depends upon the desired orientation of **n**.

For a closed surface, it is conventional to define the orientation of n at any point as outward. In any case, we suppose that S actually possesses two sides, which can be distinguished from each other, and that one side has been selected as the positive side, from which n is to point.

Further, we suppose here, and in the remainder of the text, that any surface S over which an integration is to be effected can be subdivided, by smooth curves, into a finite number of parts such that on each part the normal **n** exists and also varies continuously with position. Such a surface is said to be *piecewise smooth*. When **n** is continuous over *all* of S, we say that S is a *smooth surface*. In the cases when S is described by an equation of form (102), the preceding requirement is merely that $\frac{\partial g}{\partial x}$, $\frac{\partial g}{\partial y}$, and $\frac{\partial g}{\partial z}$ exist and be continuous over S, or

over each of a finite number of parts into which S is divided.

Thus, for the unit sphere $x^2 + y^2 + z^2 = 1$, we can write

$$g(x,y,z) = x^2 + y^2 + z^2 - 1$$

and hence

$$\nabla g = 2x \mathbf{i} + 2y \mathbf{j} + 2z \mathbf{k}$$

It follows that at any point (x_0, y_0, z_0) on the sphere we have

$$\mathbf{n} = \pm \frac{x_0 \mathbf{i} + y_0 \mathbf{j} + z_0 \mathbf{k}}{\sqrt{x_0^2 + y_0^2 + z_0^2}} = \pm (x_0 \mathbf{i} + y_0 \mathbf{j} + z_0 \mathbf{k}).$$

The *positive* sign must be chosen if \mathbf{n} is to point outward from the surface of the sphere.

At a point P(x,y,z) on S we define a differential surface area vector do whose length is numerically equal to the element of surface area do associated with P, and whose direction coincides with the normal vector **n** at P. If the element do is constructed by projecting a rectangular element dA = dx dy in the xy plane upward onto the surface S at P, it is seen that the projection of do on the xy plane is numerically equal to the projection of **n** do on the vector which is normal to the xy plane. Hence we have

$$|\mathbf{n} \cdot \mathbf{k}| \ d\sigma = dx \ dy.$$

If the direction cosines of **n** are denoted by (cos α , cos β , cos γ), this equation becomes

$$d\sigma = |\sec \gamma| \, dx \, dy. \tag{104}$$

From Equation (103) we obtain

$$\cos \gamma = \mathbf{n} \cdot \mathbf{k} = \pm \frac{\frac{\partial g}{\partial z}}{\sqrt{\left(\frac{\partial g}{\partial x}\right)^2 + \left(\frac{\partial g}{\partial y}\right)^2 + \left(\frac{\partial g}{\partial z}\right)^2}}.$$
 (105)

In particular, if the surface is given in the special form

$$z = f(x, y), \tag{106}$$

there follows g(x,y,z) = z - f(x,y), and Equation (105) becomes

$$\cos \gamma = \pm \frac{1}{\sqrt{1 + \left(\frac{\partial f}{\partial x}\right)^2 + \left(\frac{\partial f}{\partial y}\right)^2}}.$$
 (107)

With the definition $d\sigma = \mathbf{n} d\sigma$, the integral

$$\iint_{S} \mathbf{F} \cdot d\mathbf{\sigma} = \iint_{S} \mathbf{F} \cdot \mathbf{n} \, d\sigma, \tag{108}$$

carried out over a surface S, is called the *surface integral* of \mathbf{F} over S. We notice that the element of integration is the product of the component of \mathbf{F} normal to the surface at a point P and the scalar element of surface area associated with P.

In illustration, we calculate the surface integral of the vector function

$$\mathbf{F} = x \mathbf{i} + y \mathbf{j}$$

over the portion of the surface of the unit sphere

$$S: \ x^2 + y^2 + z^2 = 1$$

above the xy plane, $z \ge 0$. The unit normal at a point of S is given by

$$\mathbf{n} = x \, \mathbf{i} + y \, \mathbf{j} + z \, \mathbf{k}.$$

Also, we obtain

$$\cos \gamma = \mathbf{n} \cdot \mathbf{k} = z,$$

and hence Equation (104) gives

$$d\sigma = \frac{dx \ dy}{z}$$

Thus the required surface integral is of the form

$$\iint_{S} \mathbf{F} \cdot d\mathbf{\sigma} = \iint_{A} (x^{2} + y^{2}) \frac{dx \, dy}{z},$$

where A denotes the projection of S onto the xy plane, and hence A is the interior of the unit circle $x^2 + y^2 = 1$ in the xy plane. In this integral, the variable z must be expressed in terms of x and y by using the equation of S. Thus we obtain the form

$$\iint_{S} \mathbf{F} \cdot d\mathbf{\sigma} = 4 \int_{0}^{1} \int_{0}^{\sqrt{1-x^{2}}} \frac{x^{2}+y^{2}}{\sqrt{1-(x^{2}+y^{2})}} \, dx \, dy$$

or, changing for convenience to polar coordinates and evaluating the result as a repeated integral,

$$\iint_{S} \mathbf{F} \cdot d\mathbf{\sigma} = 4 \int_{0}^{\pi/2} \int_{0}^{1} \frac{r^{2}}{\sqrt{1-r^{2}}} r \, d\theta \, dr = \frac{4}{3} \pi.$$

It should be noticed that if lines parallel to the z axis intersect the surface S in more than one point, the surface must be considered in two or more parts, the formulas resulting from the use of (104) being applied separately to the individual parts. In such cases it may be more convenient to project the element $d\sigma$ onto the xz or yz plane, or to proceed by a more direct method.

6.13. The divergence theorem. Let V represent the velocity of flow of a fluid in three dimensions, and denote the mass density of the fluid at a point (x,y,z) by $\rho(x,y,z)$. Then the vector $\mathbf{Q} = \rho \mathbf{V}$ points in the direction of flow and has a magnitude Q numerically equal to the rate of flow of fluid mass through unit area perpendicular to the direction of flow. The differential rate of flow through a directed element of surface area $d\sigma = \mathbf{n} d\sigma$ is then given by $\mathbf{Q} \cdot d\sigma = \mathbf{Q} \cdot \mathbf{n} d\sigma$, this quantity being positive if the projection of \mathbf{Q} on the vector \mathbf{n} is positive. In particular, if $d\sigma$ is an element of a closed surface, then $\mathbf{Q} \cdot d\sigma$ is positive if the flow is *outward* from the surface. We write

$$\mathbf{Q} = Q_x \mathbf{i} + Q_y \mathbf{j} + Q_z \mathbf{k}. \tag{109}$$

Now consider a small closed differential element of volume consisting of a rectangular parallelepiped with one vertex at P(x,y,z) and with edges dx, dy, dz parallel to the coordinate axes. Then the left-hand face is represented by the differential surface vector $-\mathbf{j} dx dz$ and the differential rate of flow through this face is given by

$$\mathbf{Q} \cdot (-\mathbf{j} \, dx \, dz) = - Q_y \, dx \, dz,$$

the negative sign indicating that if Q_v is positive, the flow through this face is *into* the volume element. Similarly, the differential rate of flow through the right-hand face is given by

$$(\mathbf{Q})_{\mathbf{y}+d\mathbf{y}} \cdot (\mathbf{j} \, dx \, dz) = (Q_{\mathbf{y}})_{\mathbf{y}+d\mathbf{y}} \, dx \, dz = \left(Q_{\mathbf{y}} + \frac{\partial Q_{\mathbf{y}}}{\partial y} \, dy\right) \, dx \, dz$$

within infinitesimals of higher order. If the remaining four faces are treated similarly, we find that the *net* differential rate of flow dF outward from the volume element $d\tau = dx dy dz$ is given by

$$dF = \left[\left(Q_x + \frac{\partial Q_x}{\partial x} \, dx \right) - Q_x \right] dy \, dz + \left[\left(Q_y + \frac{\partial Q_y}{\partial y} \, dy \right) - Q_y \right] dx \, dz \\ + \left[\left(Q_z + \frac{\partial Q_z}{\partial z} \, dz \right) - Q_z \right] dx \, dy \\ = \left(\frac{\partial Q_x}{\partial x} + \frac{\partial Q_y}{\partial y} + \frac{\partial Q_z}{\partial z} \right) dx \, dy \, dz, \\ dF = (\nabla \cdot \mathbf{Q}) \, d\tau.$$
(110)

or

Thus, we may say that the divergence of Q at a point P represents the rate of fluid flow, per unit volume, outward from a differential volume associated

with P, or that the divergence of Q is the rate of decrease of mass per unit volume in the neighborhood of the point. Thus, if no mass is added to or withdrawn from the element $d\tau$, we have the relation

$$\nabla \cdot \mathbf{Q} = -\frac{\partial \rho}{\partial t}, \qquad (111)$$

which can also be written in the form

$$\nabla \cdot (\rho \mathbf{V}) + \frac{\partial \rho}{\partial t} = 0$$

or, using Equation (74a),

$$\rho \nabla \cdot \mathbf{V} = -\left(\mathbf{V} \cdot \nabla \rho + \frac{\partial \rho}{\partial t}\right). \tag{112}$$

For an *incompressible* fluid, $\rho = \text{constant}$ and there follows

$$\nabla \cdot \mathbf{Q} = \rho \, \nabla \cdot \mathbf{V} = \mathbf{0}. \tag{113}$$

Thus the velocity of an incompressible fluid has zero divergence.

It has been assumed here that no mass is introduced into the system or taken from the system, that is, that there are no points in the element $d\tau$ where fluid is added to or withdrawn from the system. A vector V with nonzero divergence can be considered as a velocity vector of an incompressible fluid in a region only if such points are assumed to be present. We speak of points where fluid is added to or taken from the system as *sources* and *sinks*, respectively. Thus the presence of a *source* distribution is associated with a *positive* divergence of V.

Consider now a closed region \mathscr{R} in three dimensions and suppose that in each elementary volume $d\tau$ of this region there are present sources through which an amount $dF = \rho \nabla \cdot \nabla d\tau$ of incompressible fluid is introduced into the region per unit time. We may think of a sink as a negative source. Then the net mass of fluid introduced into \mathscr{R} in unit time through sources and sinks is given by the integral

$$\rho \iiint_{\mathscr{R}} \nabla \cdot \mathbf{V} \, d\tau, \tag{114}$$

where the integration is carried out over the volume of the region. If the total mass in \mathscr{R} is conserved, this fluid clearly must escape from the region through the surface S which bounds it. If we consider a vector surface element $d\sigma = \mathbf{n} d\sigma$, where **n** is the *outward* unit normal, the rate of mass flow outward through the element $d\sigma$ is given by $\rho \mathbf{V} \cdot \mathbf{n} d\sigma = \rho \mathbf{V} \cdot d\sigma$. Thus the total rate of flow outward through through S is given by the surface integral of V over S,

$$\rho \oint_{S} \mathbf{V} \cdot d\mathbf{\sigma} = \rho \oint_{S} \mathbf{V} \cdot \mathbf{n} \, d\sigma, \qquad (115)$$

where the symbol \bigoplus indicates that the integration is carried out over a closed surface.

The statement that (114) and (115) must be equivalent,

$$\iiint_{\mathscr{R}} \nabla \cdot \mathbf{V} \, d\tau = \bigoplus_{\mathcal{S}} \mathbf{V} \cdot \mathbf{n} \, d\sigma, \tag{116}$$

that is, the consequence of conservation of mass in \mathcal{R} , is known as the divergence theorem. This theorem can be established without reference to the physical considerations used here,* and is true if V and its partial derivatives are continuous in \mathcal{R} and on S and if S is piecewise smooth. The region \mathcal{R} need not be simply connected, so long as its complete (inner and outer) boundary is taken into account in the surface integral.

If we write

$$\mathbf{V} = P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}, \quad \mathbf{n} = \mathbf{i} \cos \alpha + \mathbf{j} \cos \beta + \mathbf{k} \cos \gamma,$$

Equation (116) takes the form

$$\iiint_{\mathscr{R}} \left(\frac{\partial P}{\partial x} + \frac{\partial Q}{\partial y} + \frac{\partial R}{\partial z} \right) dx dy dz$$
$$= \oiint_{\mathscr{S}} (P \cos \alpha + Q \cos \beta + R \cos \gamma) d\sigma. \quad (117)$$

This theorem is useful in many applications. As one illustration, we use the theorem in obtaining the solution of the illustrative problem of Section 6.12. There we calculated directly the surface integral of $\mathbf{F} = x \mathbf{i} + y \mathbf{j}$ over the upper half of the unit sphere. If we close the surface of integration by adding the portion of the xy plane which spans the hemisphere, we notice that the surface integral of \mathbf{F} over the added surface is zero, since $\mathbf{F} \cdot \mathbf{n} = \mathbf{F} \cdot (-\mathbf{k}) = 0$ over this area. Thus the divergence theorem states that we may calculate the required surface integral of \mathbf{F} by evaluating $\iiint \nabla \cdot \mathbf{F} d\tau$ throughout the interior of the hemisphere. Since $\nabla \cdot \mathbf{F} = 2$, the result is merely twice the volume of the unit hemisphere, or $4\pi/3$, as was obtained by direct calculation.

The surface integral of a vector \mathbf{F} over a closed surface S is called the *flux* of \mathbf{F} through S. Thus the divergence theorem states that the flux of \mathbf{F} through S is a measure of the divergence of \mathbf{F} inside S. In particular, if \mathbf{F} is continuously differentiable and $\nabla \cdot \mathbf{F} = 0$ in the region bounded by S, the flux of \mathbf{F} through S is zero,

$$\oint_{S} \mathbf{F} \cdot d\mathbf{\sigma} = \oint_{S} \mathbf{F} \cdot \mathbf{n} \, d\sigma = 0$$

Thus, remembering the result of Section 6.10, we see that, for a continuously differentiable function \mathbf{F} , the circulation $\oint \mathbf{F} \cdot d\mathbf{r}$ of \mathbf{F} around a closed curve C is zero if C is in a simple region throughout which $\nabla \times \mathbf{F} = \mathbf{0}$, and also the flux $\oint \mathbf{F} \cdot d\mathbf{\sigma}$ of \mathbf{F} through a closed surface S is zero if $\nabla \cdot \mathbf{F} = 0$ in the region bounded by S.

* For an indication of the analytical proof, see Problem 65.

We may expect that the circulation of F around C will be in some sense the measure of the curl of F at points on a surface of which C is a boundary, in analogy with the relation (116) which connects flux through a surface to divergence inside the surface. A relationship of this sort, known as *Stokes's theorem*, is considered in Section 6.16.

6.14. Green's theorem. If, in the divergence theorem (116), we write

$$\mathbf{V} = \varphi_1 \, \mathbf{\nabla} \varphi_2$$
,

where φ_1 and φ_2 are scalar functions of position, we obtain the result

$$\iiint_{\mathscr{R}} \nabla \cdot \varphi_1 \nabla \varphi_2 \, d\tau = \bigoplus_{S} \mathbf{n} \cdot \varphi_1 \nabla \varphi_2 \, d\sigma.$$

By making use of (74a), this result takes the form

$$\iiint_{\mathscr{R}} \left[\varphi_1 \nabla^2 \varphi_2 + (\nabla \varphi_1) \cdot (\nabla \varphi_2) \right] d\tau = \bigoplus_{\mathscr{S}} \mathbf{n} \cdot \varphi_1 \nabla \varphi_2 \, d\sigma. \tag{118}$$

This equation is known as the *first form of Green's theorem*. A more symmetrical form is obtained if φ_1 and φ_2 are interchanged in Equation (118) and the resultant equation is subtracted from (118) to give

$$\iiint_{\mathscr{R}} \left[\varphi_1 \nabla^2 \varphi_2 - \varphi_2 \nabla^2 \varphi_1 \right] d\tau = \bigoplus_{\mathscr{S}} \mathbf{n} \cdot \left(\varphi_1 \nabla \varphi_2 - \varphi_2 \nabla \varphi_1 \right) d\sigma.$$
(119)

This equation is known as the second form of Green's theorem, or frequently merely as Green's theorem, and is frequently useful in applications. In both (118) and (119), the region \mathscr{R} is the region bounded by the closed surface S. It is assumed that φ_1 and φ_2 are twice continuously differentiable in \mathscr{R} .

Two special cases of these theorems are of particular interest. If we take $\varphi_1 = \varphi_2 = \varphi$ in (118), there follows

$$\iint \iint_{\mathscr{R}} \left[\varphi \nabla^2 \varphi + (\nabla \varphi)^2 \right] d\tau = \oint_{\mathcal{S}} \varphi \, \mathbf{n} \cdot \nabla \varphi \, d\sigma, \tag{120}$$

where $(\nabla \varphi)^2 = (\nabla \varphi) \cdot (\nabla \varphi)$. We notice that the product $\mathbf{n} \cdot \nabla \varphi$ is, according to (64), the derivative of φ in the direction of \mathbf{n} , that is, in the direction of the outward normal to S at a point on S. Hence we may write

$$\frac{\partial \varphi}{\partial n} = \mathbf{n} \cdot \nabla \varphi \tag{121}$$

and speak of this quantity as the normal derivative of φ at a point on S. With this notation, Equation (120) becomes

$$\iiint_{\mathscr{R}} \left[\varphi \nabla^2 \varphi + (\nabla \varphi)^2 \right] d\tau = \oiint_S \varphi \frac{\partial \varphi}{\partial n} d\sigma, \qquad (122)$$

and Equations (118) and (119) can be rewritten in a similar way.

A second important special case of Green's theorem is obtained by taking $\varphi_1 = \varphi$ and $\varphi_2 = 1$ in (119). Since then $\nabla \varphi_2 = 0$ and $\nabla^2 \varphi_2 = 0$, there follows, with the notation of (121),

$$\iiint_{\mathscr{R}} \nabla^2 \varphi \ d\tau = \oiint_S \frac{\partial \varphi}{\partial n} \ d\sigma. \tag{123}$$

The relations (122) and (123) will be useful in certain applications which follow (Section 9.2).

6.15. Interpretation of curl. Laplace's equation. Suppose that the motion of a fluid is simply a rotation about a given axis fixed in space. Then we may represent the angular velocity by a constant vector $\boldsymbol{\omega}$, as in Section 6.3, where the length $\boldsymbol{\omega}$ is the scalar angular velocity and the direction of $\boldsymbol{\omega}$ is along the direction of the axis of rotation in accordance with the right-hand rule. If we take the origin of a rectangular coordinate system on this axis and denote the position vector to a point P(x,y,z) by **r**, then the results of Section 6.3 show that the velocity of P is given by the vector

$$\mathbf{V} = \mathbf{\omega} \times \mathbf{r}.\tag{124}$$

We have then, using Equation (74c),

$$\nabla \cdot \mathbf{V} = \nabla \cdot \boldsymbol{\omega} \times \mathbf{r} = \mathbf{r} \cdot (\nabla \times \boldsymbol{\omega}) - \boldsymbol{\omega} \cdot (\nabla \times \mathbf{r}),$$

and hence, since $\boldsymbol{\omega}$ is constant and $\boldsymbol{\nabla} \times \mathbf{r} = \mathbf{0}$, from (78b), there follows

$$\nabla \cdot \mathbf{V} = \mathbf{0}.\tag{125}$$

We also have, using Equation (74d) and again noticing that ω is constant,

$$abla imes \mathbf{V} = \mathbf{
abla} imes (\mathbf{\omega} imes \mathbf{r}) = \mathbf{\omega} (\mathbf{
abla} \cdot \mathbf{r}) - (\mathbf{\omega} \cdot \mathbf{
abla})\mathbf{r},$$

or, by virtue of (78a) and (79),

$$\nabla \times \mathbf{V} = 2\boldsymbol{\omega}. \tag{126}$$

Thus, if a fluid experiences a pure rotation, the divergence of its velocity vector is zero, and the curl of the velocity vector is twice the angular velocity vector.

If $\nabla \times V = 0$ in a region, we say that the flow is *irrotational* in that region. The results of Section 6.10 state that the circulation around a closed curve in a simple region where the flow is irrotational is zero. If the fluid is *incompressible* and there is no distribution of sources or sinks in the region, we have also $\nabla \cdot V = 0$. Since the condition $\nabla \times V = 0$ implies the existence of a potential φ such that

$$\mathbf{V} = \boldsymbol{\nabla} \boldsymbol{\varphi},\tag{127}$$

we see that if also $\nabla \cdot \mathbf{V} = 0$ there follows $\nabla \cdot \nabla \varphi = \nabla^2 \varphi = 0$. That is, in the flow of an incompressible irrotational fluid without distributed sources or sinks,

the velocity vector is the gradient of a potential φ which satisfies the equation

$$\nabla^2 \varphi = 0 \quad or \quad \frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} + \frac{\partial^2 \varphi}{\partial z^2} = 0.$$
 (128)

This equation is known as Laplace's equation.

We see, more generally, that in *any* continuously differentiable vector field \mathbf{F} with zero divergence and curl in a simple region, the vector \mathbf{F} is the gradient of a solution of Laplace's equation. Solutions of this equation are called *harmonic functions*, and their determination is studied in a later chapter.

6.16. Stokes's theorem. We next indicate that if S is a two-sided surface in three dimensions having a closed curve C as its boundary, then the circulation of a vector V around C is equal to the flux of the curl of V over S. It is assumed throughout that S and C are piecewise smooth.

The result to be established is thus of the form

$$\iint_{S} \mathbf{n} \cdot (\nabla \times \mathbf{V}) \, d\sigma = \oint_{C} \mathbf{V} \cdot d\mathbf{r}, \qquad (129)$$

where **n** is the unit vector normal to S on that side of S which is arbitrarily taken as the *positive side*. The positive direction along C is then defined as the direction along which an observer, traveling on the positive side of S, would proceed in keeping the enclosed area to his left. Equation (129) is known as *Stokes's theorem*, and is true if S is contained in a simple region where V is continuously differentiable. In the following considerations, however, we assume also the existence of continuous partial derivatives of the *second* order.

Suppose that Equation (129) has been established for one surface S_1 having C as its boundary, and consider any other surface S_2 having the same boundary. If we denote the closed region included between S_1 and S_2 by \mathcal{R} , the divergence theorem states that

$$\int \int_{S_1} \mathbf{n}_1 \cdot (\nabla \times \mathbf{V}) \, d\sigma + \int \int_{S_2} \mathbf{n}_2 \cdot (\nabla \times \mathbf{V}) \, d\sigma$$
$$= \int \int \int_{\mathscr{H}} \nabla \cdot \nabla \times \mathbf{V} \, d\tau = 0, \quad (130)$$

if V is twice continuously differentiable in \mathscr{R} , according to Equation (74g). In this equation the vectors \mathbf{n}_1 and \mathbf{n}_2 are headed *outward* from the enclosed region \mathscr{R} . Hence if we choose a positive direction around the common boundary C so that \mathbf{n}_1 is on the positive side of S_1 , and hence $\mathbf{n}_1 = \mathbf{n}$ on S_1 , it follows that \mathbf{n}_2 is then on the negative side of S_2 , and hence $\mathbf{n}_2 = -\mathbf{n}$ on S_2 . Thus (130) becomes

$$\iint_{S_1} \mathbf{n} \cdot (\nabla \times \mathbf{V}) \, d\sigma = \iint_{S_2} \mathbf{n} \cdot (\nabla \times \mathbf{V}) \, d\sigma. \tag{131}$$

Thus, if Equation (129) is true for one surface having C as its boundary, it is also true for any other such surface, if both surfaces lie in a simple region where V is twice continuously differentiable.

We prove (129) in the special case when C is a plane curve by taking S as the plane area A bounded by C in its plane. The vector **n** normal to A is then a constant vector. The proof is obtained most readily by an indirect method in which we apply the divergence theorem to a vector $\mathbf{V} \times \mathbf{n}$, which is thus



Figure 6.10

eorem to a vector $\mathbf{V} \times \mathbf{n}$, which is thus parallel to the plane of A, in a closed region \mathscr{R} which is the interior of a right cylinder of constant height h having A as its lower base (Figure 6.10). We then have

$$\int \int \int_{\mathscr{R}} \nabla \cdot (\mathbf{V} \times \mathbf{n}) \, d\tau = \int \int_{\mathcal{S}_{\mathscr{R}}} \mathbf{n}_{\mathscr{R}} \cdot (\mathbf{V} \times \mathbf{n}) \, d\sigma_{\mathscr{R}}, \quad (132)$$

where the surface integral is extended over the complete boundary $S_{\mathcal{R}}$ of \mathcal{R} , and $\mathbf{n}_{\mathcal{R}}$ is the outward unit normal vector on $S_{\mathcal{R}}$. On the upper and lower faces of

 $S_{\mathscr{R}}$ we have $\mathbf{n}_{\mathscr{R}} = \pm \mathbf{n}$, and hence the integrand in the surface integral vanishes. On the lateral boundary of $S_{\mathscr{R}}$ we have $\mathbf{n}_{\mathscr{R}} = \mathbf{u} \times \mathbf{n}$, where \mathbf{u} is the unit tangent vector to the curve C which bounds A, and hence here the integrand becomes

$$(\mathbf{u} \times \mathbf{n}) \cdot (\mathbf{V} \times \mathbf{n}) = (\mathbf{V} \cdot \mathbf{u})(\mathbf{n} \cdot \mathbf{n}) - (\mathbf{u} \cdot \mathbf{n})(\mathbf{n} \cdot \mathbf{V}) = \mathbf{V} \cdot \mathbf{u}$$

by virtue of (35) and the fact that $\mathbf{u} \cdot \mathbf{n} = 0$. Also, since \mathbf{n} is constant, we have $\nabla \cdot (\nabla \times \mathbf{n}) = \mathbf{n} \cdot \nabla \times \nabla$, from (74c), and hence Equation (132) becomes

$$\int_0^h \left(\int \int_A \mathbf{n} \cdot \nabla \times \mathbf{V} \, d\sigma \right) dh = \int_0^h \left(\oint_C \mathbf{V} \cdot \mathbf{u} \, ds \right) dh$$

Since this relationship is true for arbitrary values of h, we conclude that

$$\iint_{A} \mathbf{n} \cdot \nabla \times \mathbf{V} \, d\sigma = \oint_{C} \mathbf{V} \cdot d\mathbf{r}, \tag{133}$$

as was to be shown.

To extend this result to the case of an arbitrary closed curve C in space, we first approximate C by a space polygon C_n whose n sides $\overline{P_1P_2}, \overline{P_2P_3}, \ldots, \overline{P_nP_1}$ are formed by joining n successive points P_1, P_2, \ldots, P_n , on C by straight lines. A surface S_n having C_n as its boundary can be defined as the polyhedron whose triangular faces are determined by the n sides of C_n and the n-3 straight lines joining P_1 to the points P_3, \ldots, P_{n-1} . If we apply Equation (133) to each of the triangular faces of S_n and add the results, we notice that the line integrals along the lines $\overline{P_1P_3}, \overline{P_1P_4}, \ldots, \overline{P_1P_{n-1}}$, are taken twice in opposite directions and hence cancel, leaving only the line integral around the polygon C_n . Thus (133) is also true for S_n and its boundary C_n . As n becomes infinite in such a way that all the chords $P_k P_{k+1}$ approach zero, the polygon C_n

approaches C, and S_n approaches a surface S for which the relation (129) is true.

From (131) we conclude finally that (129) is true for an arbitrary piecewise smooth surface S with C as its boundary if S is in a simple region where V is twice continuously differentiable.

If we write

$$\mathbf{V} = P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}, \quad \mathbf{n} = \mathbf{i} \cos \alpha + \mathbf{j} \cos \beta + \mathbf{k} \cos \gamma,$$

Stokes's theorem takes the form

$$\oint_{C} (P \, dx + Q \, dy + R \, dz) = \iint_{S} \left[\left(\frac{\partial R}{\partial y} - \frac{\partial Q}{\partial z} \right) \cos \alpha + \left(\frac{\partial P}{\partial z} - \frac{\partial R}{\partial x} \right) \cos \beta + \left(\frac{\partial Q}{\partial x} - \frac{\partial P}{\partial y} \right) \cos \gamma \right] d\sigma. \quad (134)$$

In particular, for a plane area A in the xy plane, there follows dz = 0, $\cos \alpha = \cos \beta = 0$, $\cos \gamma = 1$, $d\sigma = dx dy$, and hence

$$\oint_C (P \, dx + Q \, dy) = \iint_A \left(\frac{\partial Q}{\partial x} - \frac{\partial P}{\partial y} \right) \, dx \, dy. \tag{135}$$

It may be mentioned that Equation (135) is also frequently referred to as *Green's theorem*, although we reserve this name for the theorem of Section 6.14. As a special case of (135) we take P = -y and Q = x. The result,

$$\frac{1}{2}\oint_C (x \, dy - y \, dx) = A, \tag{136}$$

is useful in finding the area enclosed by a plane curve whose equation is given in parametric form.

Thus for the ellipse $\frac{x^2}{a^2} + \frac{y^2}{b^2} = 1$ we can write

$$x = a \cos \theta, \quad y = b \sin \theta \qquad (0 \le \theta \le 2\pi),$$

and Equation (136) gives for its area

$$A = \frac{1}{2} \int_0^{2\pi} ab \left(\cos^2 \theta + \sin^2 \theta \right) d\theta = \pi ab.$$

The rigorous proof of Stokes's theorem (129) in full generality involves subtleties which are beyond the level of the present treatments. However, it is of some importance here to notice that from that theorem one may deduce, in particular, that if V is continuously differentiable in a simply connected region \mathcal{R} , and if $\nabla \times V = 0$ everywhere in \mathcal{R} , then

$$\int_C \mathbf{V} \cdot d\mathbf{r} = \mathbf{0}$$

for any piecewise smooth closed curve C in \mathcal{R} . This is a result made plausible in Section 6.10 [Equation (91)].

From that result, in turn, one can reason in a direction opposite to that followed in Section 6.10, to deduce that, under the conditions just stated, the line integral $\int_C \mathbf{V} \cdot d\mathbf{r}$ between two points in \mathcal{R} is independent of the path C joining those points, provided only that C is piecewise smooth and lies in \mathcal{R} , and it then follows also that $\mathbf{V} \cdot d\mathbf{r} = d\varphi$, where φ is single-valued in \mathcal{R} .

The specialization of these results to the case when \mathcal{R} is a twodimensional region, in the xy plane, and in which Stokes's theorem consequently reduces to (135), will be of particular significance in Section 10.5.

6.17. Orthogonal curvilinear coordinates. Suppose that the rectangular coordinates x, y, z are expressed in terms of new coordinates u_1, u_2, u_3 by the equations

$$x = x(u_1, u_2, u_3), \quad y = y(u_1, u_2, u_3), \quad z = z(u_1, u_2, u_3),$$
 (137)

and that, conversely, these relations can be inverted to express u_1 , u_2 , u_3 in terms of x, y, z, when x, y, z and/or u_1 , u_2 , u_3 are suitably restricted. Then, at least in some region, any point with coordinates (x,y,z) has corresponding coordinates (u_1, u_2, u_3) . We assume that the correspondence is unique. If a particle moves from a point P in such a way that u_2 and u_3 are held constant and only u_1 varies, a curve in space is generated. We speak of this curve as the u_1 curve. Similarly, two other coordinate curves, the u_2 and u_3 curves, are determined at each point (Figure 6.11). Further, if only one coordinate is held constant, we determine successively three surfaces passing through a point of space, these surfaces intersecting in the coordinate curves. It is often convenient to choose the new coordinates in such a way that the coordinate curves are mutually perpendicular at each point in space. Such coordinates are called *orthogonal curvilinear coordinates*.

As an example (see Figure 6.12), in circular cylindrical coordinates (r, θ, z) we have

 $x = r \cos \theta$, $y = r \sin \theta$, z = z.

If we take

$$u_1=r, \quad u_2=\theta, \quad u_3=z,$$

then at any point P the u_2 curve is a circle of radius r and the u_1 and u_3 curves are straight lines.

Let r represent the position vector of a point P in space,

$$\mathbf{r} = x \, \mathbf{i} + y \, \mathbf{j} + z \, \mathbf{k}. \tag{138}$$

Then a tangent vector to the u_1 curve at P is given by

$$\mathbf{U}_{1} = \frac{\partial \mathbf{r}}{\partial u_{1}} = \frac{\partial \mathbf{r}}{\partial s_{1}} \frac{ds_{1}}{du_{1}}, \qquad (139)$$



where s_1 is arc length along the u_1 curve. Since $\frac{\partial \mathbf{r}}{\partial s_1}$ is a *unit vector*, we can write $\mathbf{U}_1 = h_1 \mathbf{u}_1$, (140)

where \mathbf{u}_1 is the unit vector tangent to the u_1 curve in the direction of increasing arc length and $h_1 = \frac{ds_1}{du_1}$ is the length of \mathbf{U}_1 . Considering the other coordinate curves similarly, we thus write

$$U_1 = h_1 u_1, \quad U_2 = h_2 u_2, \quad U_3 = h_3 u_3,$$
 (141)

where \mathbf{u}_k (k = 1, 2, 3) is the unit vector tangent to the u_k curve, and

$$h_1 = \frac{ds_1}{du_1} = \left| \frac{\partial \mathbf{r}}{\partial u_1} \right|, \qquad h_2 = \frac{ds_2}{du_2} = \left| \frac{\partial \mathbf{r}}{\partial u_2} \right|, \qquad h_3 = \frac{ds_3}{du_3} = \left| \frac{\partial \mathbf{r}}{\partial u_3} \right|.$$
 (142)

These equations can be written in the differential form

$$ds_1 = h_1 du_1, \quad ds_2 = h_2 du_2, \quad ds_3 = h_3 du_3.$$
 (143)

We thus see that h_1 , h_2 , h_3 are of the nature of *scale factors*, giving the ratios of differential distances to the differentials of the coordinate parameters. It should be noticed that the calculation of U_1 , from Equation (139), determines both the scale factor $h_1 = |U_1|$ and the unit vector $\mathbf{u}_1 = \mathbf{U}_1/h_1$.

We adopt the convention that \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 form a right-handed system in the order written, in consequence of an appropriate numbering of the co-ordinates.

If s is arc length along a curve in any direction, the vector

$$\frac{d\mathbf{r}}{ds} = \frac{\partial \mathbf{r}}{\partial u_1} \frac{du_1}{ds} + \frac{\partial \mathbf{r}}{\partial u_2} \frac{du_2}{ds} + \frac{\partial \mathbf{r}}{\partial u_3} \frac{du_3}{ds}$$
$$= \mathbf{U}_1 \frac{du_1}{ds} + \mathbf{U}_2 \frac{du_2}{ds} + \mathbf{U}_3 \frac{du_3}{ds}$$
(144)

has unit length. Thus, if the coordinate curves are orthogonal, so that

$$\mathbf{U}_1 \cdot \mathbf{U}_2 = \mathbf{U}_2 \cdot \mathbf{U}_3 = \mathbf{U}_3 \cdot \mathbf{U}_1 = \mathbf{0}, \tag{145}$$

there follows

$$U_1 \cdot U_1 \, du_1^2 + U_2 \cdot U_2 \, du_2^2 + U_3 \cdot U_3 \, du_3^2 = ds^2$$

or, using Equation (140),

$$ds^{2} = h_{1}^{2} du_{1}^{2} + h_{2}^{2} du_{2}^{2} + h_{3}^{2} du_{3}^{2}.$$
 (146)

To find the element of volume, we notice that the vectors $U_1 du_1$, $U_2 du_2$, and $U_3 du_3$ are mutually perpendicular vectors having as their lengths the arclength differentials ds_1 , ds_2 , and ds_3 . Thus the element of volume $d\tau$ is given by the volume of the rectangular parallelepiped determined by these vectors,

$$d\tau = |\mathbf{U}_1 \, du_1 \times \mathbf{U}_2 \, du_2 \cdot \mathbf{U}_3 \, du_3| = |\mathbf{u}_1 \times \mathbf{u}_2 \cdot \mathbf{u}_3| \, h_1 h_2 h_3 \, du_1 \, du_2 \, du_3,$$

hence

and

$$d\tau = h_1 h_2 h_3 \, du_1 \, du_2 \, du_3. \tag{147}$$

On the surface $u_1 = \text{constant}$, the vector element of surface area $d\sigma_1$ is, in a similar way, given by the vector product $U_2 du_2 \times U_3 du_3$, and hence

$$d\mathbf{\sigma}_1 = \mathbf{u}_1 \, h_2 h_3 \, du_2 \, du_3. \tag{148}$$

Analogous expressions are obtained on the other coordinate surfaces. In particular, the scalar surface elements on the coordinate surfaces are of the form

$$d\sigma_1 = h_2 h_3 \, du_2 \, du_3, \quad d\sigma_2 = h_3 h_1 \, du_3 \, du_1, \quad d\sigma_3 = h_1 h_2 \, du_1 \, du_2. \tag{149}$$

We next proceed to the determination of expressions for the gradient, divergence, and curl, and for the Laplacian operator ∇^2 in the present coordinates. To determine the gradient of a scalar function f, we make use of the definition (64),

$$df = \nabla f \cdot d\mathbf{r},\tag{150}$$

which states the basic geometrical property of the gradient. In terms of the coordinates u_1, u_2, u_3 we have

$$df = \frac{\partial f}{\partial u_1} du_1 + \frac{\partial f}{\partial u_2} du_2 + \frac{\partial f}{\partial u_3} du_3$$
$$d\mathbf{r} = \mathbf{U}_1 du_1 + \mathbf{U}_2 du_2 + \mathbf{U}_3 du_3$$
$$= h_1 \mathbf{u}_1 du_1 + h_2 \mathbf{u}_2 du_2 + h_2 \mathbf{u}_3 du_3.$$

and

$$= n_1 \mathbf{u}_1 \ u u_1 + n_2 \mathbf{u}_2 \ u u_2 + n_2 \mathbf{u}_3 \ u u_3.$$

If we now write

$$\nabla f = \lambda_1 \mathbf{u}_1 + \lambda_2 \mathbf{u}_2 + \lambda_3 \mathbf{u}_3, \qquad (151)$$

where the λ 's are to be determined, the substitution of these expressions into (150) gives

$$\frac{\partial f}{\partial u_1} du_1 + \frac{\partial f}{\partial u_2} du_2 + \frac{\partial f}{\partial u_3} du_3 = h_1 \lambda_1 du_1 + h_2 \lambda_2 du_2 + h_3 \lambda_3 du_3,$$

and hence, since the variables u_1 , u_2 , u_3 are independent, we have

$$\lambda_k = \frac{1}{h_k} \frac{\partial f}{\partial u_k} \qquad (k = 1, 2, 3). \tag{152}$$

Thus Equation (151) becomes

$$\nabla f = \frac{\mathbf{u}_1}{h_1} \frac{\partial f}{\partial u_1} + \frac{\mathbf{u}_2}{h_2} \frac{\partial f}{\partial u_2} + \frac{\mathbf{u}_3}{h_3} \frac{\partial f}{\partial u_3}$$
(153)

and the operator ∇ accordingly has the form

$$\nabla = \frac{\mathbf{u}_1}{h_1} \frac{\partial}{\partial u_1} + \frac{\mathbf{u}_2}{h_2} \frac{\partial}{\partial u_2} + \frac{\mathbf{u}_3}{h_3} \frac{\partial}{\partial u_3}.$$
 (154)

Equations (153) and (154) can be written in the equivalent forms

$$\nabla f = \mathbf{u}_1 \frac{\partial f}{\partial s_1} + \mathbf{u}_2 \frac{\partial f}{\partial s_2} + \mathbf{u}_3 \frac{\partial f}{\partial s_3},$$
 (153a)

$$\boldsymbol{\nabla} = \mathbf{u}_1 \frac{\partial}{\partial s_1} + \mathbf{u}_2 \frac{\partial}{\partial s_2} + \mathbf{u}_3 \frac{\partial}{\partial s_3}, \qquad (154a)$$

to show more clearly the intrinsic nature of the operator ∇ .

In particular, there follows

$$\nabla u_1 = \frac{\mathbf{u}_1}{h_1}, \quad \nabla u_2 = \frac{\mathbf{u}_2}{h_2}, \quad \nabla u_3 = \frac{\mathbf{u}_3}{h_3}, \quad (155)$$

for the gradients of the independent variables u_1, u_2, u_3 .

From these special results we now obtain two further results which are useful in the following work. First, since $\nabla \times \nabla u_k = 0$, from (74f), we conclude that the expressions \mathbf{u}_1/h_1 , \mathbf{u}_2/h_2 , and \mathbf{u}_3/h_3 have zero curl,

$$\nabla \times \frac{\mathbf{u}_1}{h_1} = \nabla \times \frac{\mathbf{u}_2}{h_2} = \nabla \times \frac{\mathbf{u}_3}{h_3} = \mathbf{0}.$$
 (156)

Second, since $\mathbf{u}_1 = \mathbf{u}_2 \times \mathbf{u}_3$, there follows

$$\frac{\mathbf{u}_1}{h_2h_3} = \frac{\mathbf{u}_2}{h_2} \times \frac{\mathbf{u}_3}{h_3} = (\nabla u_2) \times (\nabla u_3),$$

and two analogous expressions are similarly obtained. Reference to Equation (74i) then shows that the expressions $\mathbf{u}_1/(h_2h_3)$, $\mathbf{u}_2/(h_3h_1)$, and $\mathbf{u}_3/(h_1h_2)$ have zero divergence,

$$\nabla \cdot \frac{\mathbf{u}_1}{h_2 h_3} = \nabla \cdot \frac{\mathbf{u}_2}{h_3 h_1} = \nabla \cdot \frac{\mathbf{u}_3}{h_1 h_2} = 0.$$
(157)

It should be kept in mind that while the vectors \mathbf{u}_1 , \mathbf{u}_2 , and \mathbf{u}_3 are of constant length unity, the directions of these vectors will in general change with position in space, and hence their divergence and curl will not, in general, vanish.

To find an expression for the divergence of a vector F in u_1 , u_2 , u_3 coordinates,

$$\mathbf{F} = F_1 \mathbf{u}_1 + F_2 \mathbf{u}_2 + F_3 \mathbf{u}_3, \tag{158}$$

we first write

$$\nabla \cdot \mathbf{F} = \nabla \cdot (F_1 \mathbf{u}_1) + \nabla \cdot (F_2 \mathbf{u}_2) + \nabla \cdot (F_3 \mathbf{u}_3).$$

If we now write the first term in the form

$$\nabla \cdot (F_1 \mathbf{u}_1) = \nabla \cdot \left[(h_2 h_3 F_1) \left(\frac{\mathbf{u}_1}{h_2 h_3} \right) \right]$$

and notice that, according to Equation (157), the second factor in brackets has zero divergence, we obtain from (74a) the result

$$\nabla \cdot (F_1 \mathbf{u}_1) = \frac{\mathbf{u}_1}{h_2 h_3} \cdot \nabla (h_2 h_3 F_1) = \frac{1}{h_1 h_2 h_3} \frac{\partial}{\partial u_1} (h_2 h_3 F_1).$$

Treating the other terms in a similar way, we obtain finally

$$\nabla \cdot \mathbf{F} = \frac{1}{h_1 h_2 h_3} \left[\frac{\partial}{\partial u_1} (h_2 h_3 F_1) + \frac{\partial}{\partial u_2} (h_3 h_1 F_2) + \frac{\partial}{\partial u_3} (h_1 h_2 F_3) \right].$$
(159)

In particular, by combining Equations (154) and (159) we find the expression for the Laplacian operator,

$$\nabla^{2} = \nabla \cdot \nabla = \frac{1}{h_{1}h_{2}h_{3}} \left[\frac{\partial}{\partial u_{1}} \left(\frac{h_{2}h_{3}}{h_{1}} \frac{\partial}{\partial u_{1}} \right) + \frac{\partial}{\partial u_{2}} \left(\frac{h_{3}h_{1}}{h_{2}} \frac{\partial}{\partial u_{2}} \right) + \frac{\partial}{\partial u_{3}} \left(\frac{h_{1}h_{2}}{h_{3}} \frac{\partial}{\partial u_{3}} \right) \right].$$
(160)

In a similar way, to find an expression for the curl of F,

$$\nabla \times \mathbf{F} = \nabla \times (F_1 \mathbf{u}_1) + \nabla \times (F_2 \mathbf{u}_2) + \nabla \times (F_3 \mathbf{u}_3),$$

we write the first term in the form

$$\nabla \times (F_1 \mathbf{u}_1) = \nabla \times \left[(h_1 F_1) \left(\frac{\mathbf{u}_1}{h_1} \right) \right]$$

so that, according to Equation (156), the second factor in brackets has zero curl. Then, from (74b), we obtain the result

$$\nabla \times (F_1 \mathbf{u}_1) = -\left(\frac{\mathbf{u}_1}{h_1} \times \nabla\right)(h_1 F_1)$$

= $-\frac{\mathbf{u}_3}{h_1 h_2} \frac{\partial}{\partial u_2}(h_1 F_1) + \frac{\mathbf{u}_2}{h_3 h_1} \frac{\partial}{\partial u_3}(h_1 F_1)$
= $\frac{1}{h_1 h_2 h_3} \left(h_2 \mathbf{u}_2 \frac{\partial}{\partial u_3} - h_3 \mathbf{u}_3 \frac{\partial}{\partial u_2}\right)(h_1 F_1).$

If the other terms are treated in a similar way, the result can be written in the form of a determinant

$$\nabla \times \mathbf{F} = \frac{1}{h_1 h_2 h_3} \begin{vmatrix} h_1 \mathbf{u}_1 & h_2 \mathbf{u}_2 & h_3 \mathbf{u}_3 \\ \frac{\partial}{\partial u_1} & \frac{\partial}{\partial u_2} & \frac{\partial}{\partial u_3} \\ h_1 F_1 & h_2 F_2 & h_3 F_3 \end{vmatrix}.$$
 (161)

For the rectangular coordinates $(u_1, u_2, u_3) \equiv (x, y, z)$ we have $h_1 = h_2 = h_3 = 1$, and all the preceding results are seen to reduce to the forms originally given in that case.

6.18. Special coordinate systems. In circular cylindrical coordinates (r,θ,z) we have*

$$x = r \cos \theta$$
, $y = r \sin \theta$, $z = z$. (162a)

Since r, θ , z in that order form a right-handed system, we may take

$$(r,\theta,z)\equiv(u_1,u_2,u_3).$$

The position vector **r** has the form

$$\mathbf{r} = \mathbf{i} \, \mathbf{r} \cos \theta + \mathbf{j} \, \mathbf{r} \sin \theta + \mathbf{k} \, \mathbf{z}.$$

Thus we have

$$\mathbf{U}_r = \frac{\partial \mathbf{r}}{\partial r} = \mathbf{i} \cos \theta + \mathbf{j} \sin \theta, \qquad \mathbf{U}_{\theta} = \frac{\partial \mathbf{r}}{\partial \theta} = -\mathbf{i} r \sin \theta + \mathbf{j} r \cos \theta,$$

 $\mathbf{U}_z = \mathbf{k}.$

From Equation (142), there follows

$$h_r = 1, \quad h_\theta = r, \quad h_z = 1,$$
 (162b)

and hence also

 $\mathbf{u}_r = \mathbf{i} \cos \theta + \mathbf{j} \sin \theta$, $\mathbf{u}_{\theta} = -\mathbf{i} \sin \theta + \mathbf{j} \cos \theta$, $\mathbf{u}_z = \mathbf{k}$. (162c) Equations (146), (147), and (149) give

$$ds = \sqrt{dr^{2} + r^{2} d\theta^{2} + dz^{2}} d\tau = r d\theta dr dz d\sigma_{r} = r d\theta dz, \quad d\sigma_{\theta} = dr dz, \quad d\sigma_{z} = r d\theta dr$$
(162d).

* We notice that the coordinate r in this system is distance from the z axis. It must not be confused with $|\mathbf{r}|$, where \mathbf{r} is the position vector and $|\mathbf{r}|$ is distance from the origin. Finally, Equations (153), (159), (160), and (161) give the results

$$\nabla f = \mathbf{u}_{r} \frac{\partial f}{\partial r} + \mathbf{u}_{\theta} \frac{1}{r} \frac{\partial f}{\partial \theta} + \mathbf{u}_{z} \frac{\partial f}{\partial z}$$

$$\nabla \cdot \mathbf{F} = \frac{1}{r} \frac{\partial}{\partial r} (rF_{r}) + \frac{1}{r} \frac{\partial F_{\theta}}{\partial \theta} + \frac{\partial F_{z}}{\partial z}$$

$$\nabla^{2} f = \frac{1}{r} \frac{\partial}{\partial r} \left(r \frac{\partial f}{\partial r} \right) + \frac{1}{r^{2}} \frac{\partial^{2} f}{\partial \theta^{2}} + \frac{\partial^{2} f}{\partial z^{2}}$$

$$(162e)$$

$$\nabla \times \mathbf{F} = \frac{1}{r} \begin{vmatrix} \mathbf{u}_{r} & r\mathbf{u}_{\theta} & \mathbf{u}_{z} \\ \frac{\partial}{\partial r} & \frac{\partial}{\partial \theta} & \frac{\partial}{\partial z} \\ F_{r} & rF_{\theta} & F_{z} \end{vmatrix}$$

In spherical coordinates (r, φ, θ) , where r is distance from the origin, θ is the polar angle measured from the xz plane, and φ is the "cone angle" measured



Figure 6-13

from the z axis (Figure 6.13), there follows*

 $x = r \cos \theta \sin \varphi$, $y = r \sin \theta \sin \varphi$, $z = r \cos \varphi$, (163a) and the following results are obtained:

$$h_r = 1, \quad h_{\varphi} = r, \quad h_{\theta} = r \sin \varphi, \quad (163b)$$
$$\mathbf{u}_r = \mathbf{i} \cos \theta \sin \varphi + \mathbf{j} \sin \theta \sin \varphi + \mathbf{k} \cos \varphi$$

$$\mathbf{u}_{\varphi} = \mathbf{i} \cos \theta \cos \varphi + \mathbf{j} \sin \theta \cos \varphi - \mathbf{k} \sin \varphi \quad \Big\} , \qquad (163c)$$

* The angles φ and θ are defined in different ways by different writers. The present convention is consistent with that used for circular cylindrical coordinates in the plane z = 0.

$$\nabla f = \mathbf{u}_{r} \frac{\partial f}{\partial r} + \frac{\mathbf{u}_{\varphi}}{r} \frac{\partial f}{\partial \varphi} + \frac{\mathbf{u}_{\theta}}{r \sin \varphi} \frac{\partial f}{\partial \theta}$$

$$\nabla \cdot \mathbf{F} = \frac{1}{r^{2}} \frac{\partial}{\partial r} (r^{2} F_{r}) + \frac{1}{r \sin \varphi} \frac{\partial}{\partial \varphi} (F_{\varphi} \sin \varphi) + \frac{1}{r \sin \varphi} \frac{\partial F_{\theta}}{\partial \theta}$$

$$\nabla^{2} f = \frac{1}{r^{2}} \frac{\partial}{\partial r} \left(r^{2} \frac{\partial f}{\partial r} \right) + \frac{1}{r^{2} \sin \varphi} \frac{\partial}{\partial \varphi} \left(\sin \varphi \frac{\partial f}{\partial \varphi} \right)$$

$$+ \frac{1}{r^{2} \sin^{2} \varphi} \frac{\partial^{2} f}{\partial \theta^{2}}$$

$$r = \frac{1}{r^{2} \sin \varphi} \begin{vmatrix} \mathbf{u}_{r} & r \mathbf{u}_{\varphi} & r \sin \varphi \mathbf{u}_{\theta} \\ \frac{\partial}{\partial r} & \frac{\partial}{\partial \varphi} & \frac{\partial}{\partial \theta} \\ F_{r} & r F_{\varphi} & r \sin \varphi F_{\theta} \end{vmatrix}$$

$$(163e)$$

6.19. Application to two-dimensional incompressible fluid flow. In certain types of fluid flow it is possible to choose a system of orthogonal curvilinear coordinates in such a way that the flow depends only upon two of the coordinates and is independent of the third. The simplest example is that in which flow is parallel to a plane, say the xy plane, and is such that its properties everywhere are then specified by its properties on the xy plane. In the more general case, we assume that the flow is independent of u_3 and is then specified by its properties on any surface for which u_3 is constant. We suppose also that the coordinate system is such that the scale factors h_1 , h_2 , and h_3 are independent of u_3 .

If the flow is divergenceless and irrotational, that is, if there are no distributions of sources or sinks or of vortices, about which the fluid tends to rotate, and if also the fluid is assumed to be incompressible, we have seen that the velocity vector V is the gradient of a function φ , called the *velocity potential*, and that φ satisfies Laplace's equation.

Since in the present case V has no component in the u_3 direction, we may write

$$\mathbf{V} = V_1 \mathbf{u}_1 + V_2 \mathbf{u}_2, \tag{164}$$

where V_1 and V_2 are the components of V in the directions of \mathbf{u}_1 and \mathbf{u}_2 , respectively. Since the velocity potential is also to be independent of u_3 , the relation $\mathbf{V} = \nabla \varphi$ becomes, by virtue of (154),

$$\mathbf{V} = \frac{\mathbf{u}_1}{h_1} \frac{\partial \varphi}{\partial u_1} + \frac{\mathbf{u}_2}{h_2} \frac{\partial \varphi}{\partial u_2}, \qquad (165)$$

and hence we have

$$V_1 = \frac{1}{h_1} \frac{\partial \varphi}{\partial u_1}, \qquad V_2 = \frac{1}{h_2} \frac{\partial \varphi}{\partial u_2}.$$
 (166)

Also, the equation $\nabla^2 \varphi = 0$ becomes

$$\frac{\partial}{\partial u_1} \left(\frac{h_2 h_3}{h_1} \frac{\partial \varphi}{\partial u_1} \right) + \frac{\partial}{\partial u_2} \left(\frac{h_3 h_1}{h_2} \frac{\partial \varphi}{\partial u_2} \right) = 0.$$
(167)

When the solution φ of the partial differential equation (167), satisfying appropriate conditions on the boundary of the region considered, has been obtained, the *equipotential lines* in the u_3 surfaces are curves given by equations of the form

$$\varphi(u_1, u_2) = c_1 \tag{168}$$

as c_1 takes on successive constant values. The velocity vectors, and hence the corresponding lines of flow or *streamlines*, are normal to these curves. If we express the streamlines by equations of the form

$$\psi(u_1, u_2) = c_2,$$
 (169)

then, since we must have

$$\nabla \varphi \cdot \nabla \psi = 0, \tag{170}$$

there follows

$$\begin{pmatrix} \frac{1}{h_1} \frac{\partial \varphi}{\partial u_1} \end{pmatrix} \left(\frac{1}{h_1} \frac{\partial \psi}{\partial u_1} \right) + \left(\frac{1}{h_2} \frac{\partial \varphi}{\partial u_2} \right) \left(\frac{1}{h_2} \frac{\partial \psi}{\partial u_2} \right) = 0,$$

$$\frac{\frac{\partial \psi}{\partial u_2}}{\frac{h_2}{h_1} \frac{\partial \varphi}{\partial u_1}} = -\frac{\frac{\partial \psi}{\partial u_1}}{\frac{h_1}{h_2} \frac{\partial \varphi}{\partial u_2}}.$$

or

If we denote the common value of these two ratios by μ , where μ is an unknown function of u_1 and u_2 , there follows

$$\frac{\partial \psi}{\partial u_1} = -\mu \frac{h_1}{h_2} \frac{\partial \varphi}{\partial u_2}, \qquad \frac{\partial \psi}{\partial u_2} = \mu \frac{h_2}{h_1} \frac{\partial \varphi}{\partial u_1}. \tag{171}$$

To determine μ , we make use of the equation

$$\frac{\partial^2 \psi}{\partial u_2 \partial u_1} = \frac{\partial^2 \psi}{\partial u_1 \partial u_2},$$

assuming appropriate continuity, and so obtain from (171)

$$\frac{\partial}{\partial u_1}\left(\frac{\mu h_2}{h_1}\frac{\partial \varphi}{\partial u_1}\right) + \frac{\partial}{\partial u_2}\left(\frac{\mu h_1}{h_2}\frac{\partial \varphi}{\partial u_2}\right) = 0.$$

306

If this equation is compared with (167) one obtains the result $\mu = ch_3$, where c is an arbitrary constant. As will be seen, it is convenient to take c = 1, so that*

$$\mu = h_3. \tag{172}$$

Thus, from Equations (171) and (172) we obtain the relationship

$$d\psi = -\frac{h_3h_1}{h_2}\frac{\partial\varphi}{\partial u_2}du_1 + \frac{h_2h_3}{h_1}\frac{\partial\varphi}{\partial u_1}du_2, \qquad (173)$$

from which ψ can be determined by integration when ϕ is known. By making use of (166), Equation (173) can also be written in the form

$$d\psi = h_3(-h_1V_2 \, du_1 + h_2V_1 \, du_2). \tag{174}$$

By using Equations (154), (171), and (166), we obtain the relation

$$\nabla \psi = -\frac{h_3}{h_2} \frac{\partial \varphi}{\partial u_2} \mathbf{u}_1 + \frac{h_3}{h_1} \frac{\partial \varphi}{\partial u_1} \mathbf{u}_2 = h_3 (-V_2 \mathbf{u}_1 + V_1 \mathbf{u}_2)$$
(175)

which can also be written in the form

$$\nabla \psi = h_3 \mathbf{u}_3 \times \mathbf{V}. \tag{176}$$

Hence there follows also

$$\nabla^{2} \psi = \nabla \cdot (h_{3} \mathbf{u}_{3} \times \mathbf{V}) = \mathbf{V} \cdot \nabla \times (h_{3} \mathbf{u}_{3}) - h_{3} \mathbf{u}_{3} \cdot \nabla \times \mathbf{V}$$
$$= \mathbf{V} \cdot \nabla \times \left(h_{3}^{2} \frac{\mathbf{u}_{3}}{h_{3}}\right) = \mathbf{V} \cdot \left(\nabla h_{3}^{2} \times \frac{\mathbf{u}_{3}}{h_{3}}\right) = \frac{\nabla h_{3}^{2}}{h_{3}^{2}} \cdot h_{3} \mathbf{u}_{3} \times \mathbf{V}$$
r

or

$$\nabla^2 \psi = 2(\nabla \log h_3^2) \cdot \nabla \psi, \qquad (177)$$

when use is made of Equations (74c), (74b), and (156), and of the fact that $\nabla \times V = 0$ in consequence of the assumed irrotational nature of the flow.

Thus we conclude that if h_3 is constant, then the function ψ in the flow under consideration also satisfies Laplace's equation, and hence may be considered as the velocity potential corresponding to a second flow having the curves $\varphi = c_1$ as streamlines. The two flows so related are said to be *conjugate*.

If h_3 is not constant, the function ψ does not satisfy Laplace's equation, and hence no conjugate (nondivergent) flow exists.

In any case, the function ψ is known as the *stream function* of the flow for which φ is the velocity potential. We notice from (173) that the expression for ψ involves an arbitrary additive constant (as well as an arbitrary multiplicative constant c). The additive constant can be chosen so that along a particular reference streamline we have $\psi = 0$.

* Equation (172) defines a *particular* solution of the equation determining μ . The most general solution is $\mu = h_3 f(\psi)$, where f is an arbitrary function of the expression ψ being determined. The more general solution leads only to the obvious fact that any function of the ψ determined by Equation (173) will also be constant along the streamlines.

In order to establish another useful property of the stream function ψ , we consider a curve C in a u_3 surface, joining two points P_0 and P_1 . Then the rate of mass flow, in the direction of the normal **n**, through a "rectangular element" based on a linear element $d\mathbf{r}$ of this curve, and extending in the u_3 direction a distance corresponding to a *unit* increment in u_3 , is given by

$$df = \rho \mathbf{V} \cdot \mathbf{n} \ h_3 ds = \rho \mathbf{V} \cdot \left(\frac{d\mathbf{r}}{ds} \times \mathbf{u}_3\right) h_3 ds = \rho h_3 \ \mathbf{u}_3 \times \mathbf{V} \cdot d\mathbf{r}$$

If use is made of Equation (176), this relation takes the form

$$df = \rho \, \nabla \psi \cdot d\mathbf{r} = \rho \, d\psi. \tag{178}$$

Thus, since here ρ is constant, we obtain by integration from P_0 to P_1 the result

$$\psi(P_1) - \psi(P_0) = \frac{1}{\rho} f(P) \Big|_{P_0}^{P_1},$$

so that the difference between the values of ψ at two points in a u_3 surface is numerically equal to the rate of mass flow of a fluid with unit density "across any curve joining these points."

As an example, we verify that the function

$$\varphi = x^2 - y^2$$

is a solution of Laplace's equation in the xy plane,

$$\nabla^2 \varphi = rac{\partial^2 \varphi}{\partial x^2} + rac{\partial^2 \varphi}{\partial y^2} = 0,$$

and hence is the potential function of a flow of the type considered. In this case we have

$$u_1 = x$$
, $u_2 = y$, $h_1 = h_2 = h_3 = 1$.

The velocity vector $\mathbf{V} = \nabla \varphi$ becomes

$$\mathbf{V} = 2x \, \mathbf{i} - 2y \, \mathbf{j},$$

and the equipotential curves in the xy plane are the hyperbolas

$$x^2 - y^2 = c_1$$

The stream function ψ is determined from Equation (173),

$$d\psi = 2y \, dx + 2x \, dy = 2 \, d(xy),$$

and hence the streamlines are the hyperbolas

$$\psi=2xy=c_2$$

Since, in particular, the lines x = 0, y = 0 are streamlines, the flow in the first quadrant is a plane flow around a right-angled corner (Figure 6.14). Also, since $\psi = 2xy$ also satisfies Laplace's equation, ψ can be considered as the

velocity potential in a conjugate flow where the curves $x^2 - y^2 = c_1$ are streamlines, and the function $\varphi = x^2 - y^2$ is the stream function.

6.20. Compressible ideal fluid flow. To further illustrate the use of vector methods, we now treat briefly the basic equations governing the flow of nonviscous (ideal) compressible fluids.

We consider first an element of fluid mass dm in the form of a rectangular parallelepiped with edges dx, dy, dz parallel to the coordinate axes at the time t



Figure 6.14

and moving with velocity V at that instant. If we denote the pressure by p, the differential force exerted by the pressure distributions on the two faces parallel to the yz plane is readily found to be $-\left(\mathbf{i}\frac{\partial p}{\partial x}dx\right)dy dz$. Considering the forces acting on the other four faces similarly, we then find that the resultant differential force due to fluid pressure on the element dm is given by $-\nabla p \, dx \, dy \, dz$ or, equivalently, by $-\frac{\nabla p}{\rho} dm$. The same expression can be shown (by use of the divergence theorem) to be valid for the differential force due to pressure acting on an element of volume $\frac{dm}{\rho}$ of any shape.

As an element moves in space, its shape and volume generally will change. However, its mass dm must be conserved. For an element of constant mass dm the rate of change of momentum is given by $\frac{dV}{dt}$ dm. Hence, if external forces are omitted. Newton's second law of motion leads to the equation

$$\rho \, \frac{d\mathbf{V}}{dt} + \boldsymbol{\nabla} p = \boldsymbol{0}. \tag{179}$$

The velocity V is a function of position x, y, z and time t, whereas the position coordinates x, y, z depend not only upon time t but also upon the initial values x_0 , y_0 , z_0 at a reference time, say t = 0. Thus the time derivative $\frac{d}{dt}$ in (179) is actually not a total derivative in the strict sense, but is calculated only for fixed initial coordinates. That is, the time differentiation follows a

given particle in its motion. For this reason this derivative is frequently termed the "particle," "substantial," or "material" derivative of V, and is often indicated by the special notation $\frac{DV}{Dt}$.

For fixed initial coordinates, V depends directly on x,y,z and t, and x, y, and z themselves depend upon the time t, in such a way that $\frac{dx}{dt} = V_x$, and so on. Hence we may write the substantial derivative of V in the expanded form (see Section 7.1),

$$\frac{d\mathbf{V}}{dt} = \frac{\partial \mathbf{V}}{\partial x}\frac{dx}{dt} + \frac{\partial \mathbf{V}}{\partial y}\frac{dy}{dt} + \frac{\partial \mathbf{V}}{\partial z}\frac{dz}{dt} + \frac{\partial \mathbf{V}}{\partial t}$$
$$= V_x\frac{\partial \mathbf{V}}{\partial x} + V_y\frac{\partial \mathbf{V}}{\partial y} + V_z\frac{\partial \mathbf{V}}{\partial z} + \frac{\partial \mathbf{V}}{\partial t}$$
$$\frac{d\mathbf{V}}{dt} = (\mathbf{V}\cdot\mathbf{\nabla})\mathbf{V} + \frac{\partial \mathbf{V}}{\partial t}.$$
(180)

or

In terms of differential operators we may write also

$$\frac{d}{dt} = \mathbf{V} \cdot \mathbf{\nabla} + \frac{\partial}{\partial t} = V_x \frac{\partial}{\partial x} + V_y \frac{\partial}{\partial y} + V_z \frac{\partial}{\partial z} + \frac{\partial}{\partial t}.$$
 (181)

Thus Equation (179) can be written in the form

$$\rho\left(\mathbf{V}\cdot\nabla\mathbf{V}+\frac{\partial\mathbf{V}}{\partial t}\right)+\nabla p=\mathbf{0}$$
(182)

and is equivalent to three scalar equations, the first of which is of the form

$$\rho\left(V_x\frac{\partial V_x}{\partial x}+V_y\frac{\partial V_x}{\partial y}+V_z\frac{\partial V_x}{\partial z}+\frac{\partial V_x}{\partial t}\right)+\frac{\partial p}{\partial x}=0.$$

These equations of motion are known as Euler's equations.

In addition to the vector equation (182), the equation of conservation of mass [Equation (112)],

$$\rho \nabla \cdot \mathbf{V} = -\left(\mathbf{V} \cdot \nabla \rho + \frac{\partial \rho}{\partial t}\right) = -\frac{d\rho}{dt}, \qquad (183)$$

must be satisfied. This equation is known as the equation of continuity.

Finally, to complete the system of basic equations, we may take the effect of compressibility into account by assuming a suitable relationship between pressure p and density ρ . (For an incompressible fluid, ρ is constant.) Assuming such a relationship, we may then write

$$\nabla p = \mathbf{i} \frac{\partial p}{\partial x} + \mathbf{j} \frac{\partial p}{\partial y} + \mathbf{k} \frac{\partial p}{\partial z} = \frac{dp}{d\rho} \left(\mathbf{i} \frac{\partial \rho}{\partial x} + \mathbf{j} \frac{\partial \rho}{\partial y} + \mathbf{k} \frac{\partial \rho}{\partial z} \right) = \frac{dp}{d\rho} \nabla \rho.$$

The quantity

$$\sqrt{\frac{dp}{d\rho}} = V_S \tag{184}$$

is known as the *local sonic velocity*, and is seen to vary with ρ from point to point in the fluid, and with time.

With these results Euler's equations in the vector form (182) become

$$\rho \mathbf{V} \cdot \nabla \mathbf{V} + \rho \, \frac{\partial \mathbf{V}}{\partial t} + V_S^2 \, \nabla \rho = \mathbf{0}. \tag{185}$$

Equations (183) and (185) are sufficient to determine ρ and V if suitable boundary conditions and/or initial conditions are prescribed.

In many problems the flow is nearly uniform. That is, the vector V and the scalar functions ρ and p are nearly constant. In such cases, Equations (183) and (185) can be *linearized* by first writing

$$V = U + u, \quad \rho = \rho_0 + \delta, \quad V_S = V_{S0} + v_S$$
 (186)

where U, ρ_0 , and V_{S0} are the constant values corresponding to the uniform flow and the quantities **u**, δ , and v_S are considered as small deviations. With this notation Equations (183) and (185) become

$$(\rho_0 + \delta) \nabla \cdot \mathbf{u} + (\mathbf{U} + \mathbf{u}) \cdot \nabla \delta + \frac{\partial \delta}{\partial t} = 0,$$

$$(\rho_0 + \delta) (\mathbf{U} + \mathbf{u}) \cdot \nabla \mathbf{u} + (\rho_0 + \delta) \frac{\partial \mathbf{u}}{\partial t} + (V_{S0} + v_S)^2 \nabla \delta = 0.$$

If products of small deviations are assumed to be relatively negligible, these equations are reduced to the linear forms

$$\rho_0 \nabla \cdot \mathbf{u} + \left(\mathbf{U} \cdot \nabla + \frac{\partial}{\partial t} \right) \delta = 0, \qquad (187)$$

$$\left(\mathbf{U}\cdot\mathbf{\nabla}+\frac{\partial}{\partial t}\right)\mathbf{u}+\frac{V_{S0}^{2}}{\rho_{0}}\,\mathbf{\nabla}\delta=\mathbf{0}.$$
(188)

For an *irrotational* flow (see Section 6.15), the velocity V, and hence also u, can be expressed as the gradient of a scalar function. Hence, if we write

$$\mathbf{u} = \boldsymbol{\nabla} \boldsymbol{\varphi},\tag{189}$$

Equation (188) becomes

$$abla \left[\delta + rac{
ho_0}{V_{S0}^2} \left(\mathbf{U} \cdot \mathbf{\nabla} + rac{\partial}{\partial t}
ight) \varphi
ight] = \mathbf{0},$$

from which the density deviation δ is expressed in terms of the velocity deviation potential φ in the form

$$\delta = -\frac{\rho_0}{V_{S0}^2} \left(\mathbf{U} \cdot \nabla + \frac{\partial}{\partial t} \right) \varphi, \qquad (190)$$

except for an additive function of time only, which can be considered as incorporated in φ . It should be noticed that all the preceding relations involve only the invariant operator ∇ , insofar as space differentiation is concerned, and hence may be readily expressed in terms of any convenient coordinate system.

We now suppose that the uniform flow is parallel to the x axis, and so write

$$\mathbf{U} = U \mathbf{i},\tag{191}$$

in which case Equation (190) becomes

$$\delta = -\frac{\rho_0}{V_{s0}^2} \left(U \frac{\partial \varphi}{\partial x} + \frac{\partial \varphi}{\partial t} \right).$$
(192)

If Equations (189), (191), and (192) are introduced into (187), we obtain the equation

$$\nabla^2 \varphi - \frac{1}{V_{S0}^2} \left(U \frac{\partial}{\partial x} + \frac{\partial}{\partial t} \right)^2 \varphi = 0, \qquad (193)$$

which must be satisfied by the potential function φ . For an *incompressible* fluid $V_S = \infty$, and the equation reduces to Laplace's equation, in accordance with the results of Section 6.15.

In the special case of steady flow in two dimensions, where the flow is parallel to the xy plane and the velocity and density do not vary with time, Equation (193) reduces to the form

$$(1 - M^2)\frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} = 0, \qquad (194)$$

where M is the so-called local Mach number,

$$M = \frac{U}{V_{S0}},\tag{195}$$

and is the ratio of the speed of the uniform flow to the sonic velocity corresponding to this flow. A flow is said to be *subsonic* when M < 1 and *supersonic* when M > 1. It is important to notice that the coefficient of $\frac{\partial^2 \varphi}{\partial x^2}$ in Equation (194) changes sign when the sonic velocity is passed.

In the special case of *nonsteady one-dimensional flow*, when the velocities in the y and z directions are assumed to be negligible, Equation (193) becomes

$$(U^2 - V_{S0}^2)\frac{\partial^2 \varphi}{\partial x^2} + 2U\frac{\partial^2 \varphi}{\partial x \partial t} + \frac{\partial^2 \varphi}{\partial t^2} = 0.$$
(196)

If the above developments are still assumed to be satisfactorily approximate when *small deviations* are measured from a *state of rest*, so that U = 0, we may replace u by V and rewrite (192) and (193) in the form

$$\delta = \rho - \rho_0 = -\frac{\rho_0}{V_{s0}^2} \frac{\partial \varphi}{\partial t}$$
(197)

and

$$\nabla^2 \varphi = \frac{1}{V_{S0}^2} \frac{\partial^2 \varphi}{\partial t^2} , \qquad (198)$$

where

Here V_{S0}^2 is essentially an effective "mean value" of $\frac{dp}{d\rho}$. Thus we have also, approximately,

 $\mathbf{V} = \nabla \varphi$.

$$p - p_0 = V_{S0}^2(\rho - \rho_0),$$

so that the pressure change due to the disturbing flow is given approximately by

$$p - p_0 = -\rho_0 \frac{\partial \varphi}{\partial t}.$$
 (200)

To conclude this section, we obtain an explicit exact integral of Euler's equation, when external forces are absent, in the case of irrotational flow $(\nabla \times V = 0)$. From the identity

$$\nabla (\mathbf{V} \cdot \mathbf{V}) = 2\mathbf{V} \cdot \nabla \mathbf{V} + 2\mathbf{V} \times (\nabla \times \mathbf{V})$$

[see Equation (74e), Section 6.9], we here obtain the result

$$\mathbf{V} \cdot \boldsymbol{\nabla} \mathbf{V} = \boldsymbol{\nabla} (\frac{1}{2} V^2),$$

where $V = |\mathbf{V}| = \sqrt{\mathbf{V} \cdot \mathbf{V}}$. Thus Euler's equation (182) can be written in the form

$$\nabla\left(\frac{1}{2}V^2\right) + \frac{\partial \mathbf{V}}{\partial t} + \frac{\nabla p}{\rho} = \mathbf{0}.$$
 (201)

If we now write

$$\mathbf{V} = \mathbf{\nabla}\varphi \tag{202}$$

and define a function P by the relation

$$\nabla P = \frac{1}{\rho} \nabla p \quad \text{or} \quad P = \int \frac{dp}{\rho},$$
 (203)

Equation (201) takes the form

$$\nabla\left(\frac{1}{2}V^{2}\right) + \frac{\partial}{\partial t}\nabla\varphi + \nabla P = \mathbf{0}$$

$$\nabla\left(\frac{1}{2}V^{2} + \frac{\partial\varphi}{\partial t} + P\right) = \mathbf{0}.$$
(204)

or

(199)
Hence the quantity $\frac{1}{2}V^2 + \frac{\partial \varphi}{\partial t} + P$ must be independent of the space coordinates and, accordingly, a function of time only,

$$\frac{1}{2}V^2 + \frac{\partial\varphi}{\partial t} + P = f(t).$$
(205)

This is one form of *Bernoulli's equation*. For *steady flow* (in which the flow is independent of time), this equation becomes

$$\frac{1}{2}V^2 + P = \text{constant.} \tag{206}$$

In particular, for an *incompressible* fluid ($\rho = \text{constant}$) we may take $P = p/\rho$, in accordance with (203), and hence (206) can be written in the form

$$\frac{1}{2}\rho V^2 + p = \text{constant} \tag{207}$$

in this case.

In illustration, for the example at the end of the preceding section involving flow around a right-angled corner, Equation (207) determines the fluid pressure p in the form

$$p = p_c - 2\rho(x^2 + y^2),$$

where p_c is the pressure at the corner.

REFERENCES

- 1. Brand, L., Vector Analysis, John Wiley & Sons, Inc., New York, 1957.
- 2. Kellogg, O. D., Foundations of Potential Theory, Dover Publications, Inc., New York, 1953.
- 3. Phillips, H. B., Vector Analysis, John Wiley & Sons, Inc., New York, 1933.
- 4. Struik, D. J., Lectures on Classical Differential Geometry, Addison-Wesley Publishing Co., Inc., Reading, Mass., 1950.

PROBLEMS

Section 6.1

1. Find the length and direction cosines of the vector **a** from the point (1, -1, 3) to the mid-point of the line segment from the origin to the point (6, -6, 4).

2. The vectors **a** and **b** extend from the origin O to the points A and B. Determine the vector **c** which extends from O to the point C which divides the line segment from A to B in the ratio m : n.

3. (a) If θ denotes the angle between the vectors **a** and **b**, use a theorem of elementary geometry to show that

$$|\mathbf{a} + \mathbf{b}|^2 = |\mathbf{a}|^2 + |\mathbf{b}|^2 + 2 |\mathbf{a}| |\mathbf{b}| \cos \theta$$
.

(b) If $\mathbf{a} = a_x \mathbf{i} + a_y \mathbf{j} + a_z \mathbf{k}$ and $\mathbf{b} = b_x \mathbf{i} + b_y \mathbf{j} + b_z \mathbf{k}$, use the preceding result to show that

$$\cos\theta = \frac{a_x b_x + a_y b_y + a_z b_z}{|\mathbf{a}| |\mathbf{b}|}.$$

4. Prove that $|a + b| \le |a| + |b|$ and $|a - b| \ge |a| - |b|$.

5. (a) Show that a straight line with direction angles α , β , γ can be specified by the equations

$$\frac{x-x_0}{\cos\alpha} = \frac{y-y_0}{\cos\beta} = \frac{z-z_0}{\cos\gamma}$$

where (x_0, y_0, z_0) is a point on the line.

(b) Show that the above equations can also be written in the parametric form

$$x = x_0 + t \cos \alpha$$
, $y = y_0 + t \cos \beta$, $z = z_0 + t \cos \gamma$,

where t is a parameter.

(Notice that in either set of equations $\cos \alpha$, $\cos \beta$, and $\cos \gamma$ can be replaced by *direction ratios A*, *B*, and *C* which are proportional to them.)

6. The equations of a straight line are of the form

$$\frac{x-1}{1} = \frac{y+2}{2} = \frac{z}{-2}.$$

(a) Show that the line includes the point (1, -2, 0) and determine its direction cosines.

(b) Determine a unit vector in the direction of the line.

Section 6.2

7. Prove geometrically, from the definition of the scalar product, that the distributive law $\mathbf{a} \cdot (\mathbf{b} + \mathbf{c}) = \mathbf{a} \cdot \mathbf{b} + \mathbf{a} \cdot \mathbf{c}$ is valid.

8. The vectors **a** and **b** are defined as follows:

a = 3i - 4k, b = 2i - 2j + k.

- (a) Find the scalar projection of a on b.
- (b) Find the angle between the positive directions of the vectors.

9. Find the magnitude of the scalar component of the force vector

$$\mathbf{F} = \mathbf{i} + 2\mathbf{j} + 2\mathbf{k}$$

in the direction of the straight line with equations x = y = 2z.

10. A plane is determined by a point $P_0(x_0, y_0, z_0)$ on it and by a vector

$$\mathbf{N} = A \mathbf{i} + B \mathbf{j} + C \mathbf{k}$$

normal to it. Show that the requirement that the vector from P_0 to a point P(x,y,z) in the plane be perpendicular to N determines the equation of the plane in the form

$$A(x - x_0) + B(y - y_0) + C(z - z_0) = 0.$$

11. (a) Show that the vector $A \mathbf{i} + B \mathbf{j} + C \mathbf{k}$ is normal to the plane

$$Ax + By + Cz = D.$$

(b) Prove that the shortest distance from the point $P_0(x_0, y_0, z_0)$ to the plane Ax + By + Cz = D is given by

$$d = \frac{|Ax_0 + By_0 + Cz_0 - D|}{\sqrt{A^2 + B^2 + C^2}}$$

[Let $P_1(x_1, y_1, z_1)$ be any point on the plane and determine the projection of the vector from P_0 to P_1 on the normal to the plane.]

12. Find the angle between the planes 2x - y + 2z = 1 and x - y = 2.

Section 6.3

13. (a) Determine a unit vector perpendicular to the plane of the vectors $\mathbf{a} = \mathbf{i} + 3\mathbf{j} - \mathbf{k}$, $\mathbf{b} = 2\mathbf{i} + \mathbf{j} + \mathbf{k}$.

(b) Find the area of the triangle of which these two vectors form coterminous sides.

14. (a) Determine a unit vector normal to the plane determined by the points (0,0,0), (1,1,1), and (2,1,3).

(b) Find the area of the triangle with vertices at the points defined in part (a).

15. A rigid body rotates with angular velocity ω about the line x = y = z. Find the speed of a particle at the point (1,2,2).

16. Find the scalar moment of the force $\mathbf{F} = \mathbf{i} - 3\mathbf{j} + 2\mathbf{k}$, acting at the point (1,2,1), about the z axis.

17. Show that the shortest distance from a point P_0 to the line joining the points P_1 and P_2 is given by

$$d=\frac{|\mathbf{v}_1\times\mathbf{v}_2|}{|\mathbf{v}_3|},$$

where $\mathbf{v}_1 = \overrightarrow{P_0P_1}$, $\mathbf{v}_2 = \overrightarrow{P_0P_2}$, and $\mathbf{v}_3 = \overrightarrow{P_1P_2}$. (Notice that $|\mathbf{v}_1 \times \mathbf{v}_2|$ is twice the area of the triangle $P_0P_1P_2$, and that the desired distance is the altitude of that triangle normal to the base P_1P_2 .)

18. Show that the shortest distance between the lines AB and CD is the projection of \overrightarrow{AC} (or of the vector joining any point of AB with any point of CD) on the vector $\overrightarrow{AB} \times \overrightarrow{CD}$.

Section 6.4

19. From the last result of Section 6.3, deduce that, if **r** denotes the vector from a point O to the point of application of a force **F**, then the scalar moment of **F** about an axis OA is given by $M_{OA} = (\mathbf{r} \mathbf{F} \mathbf{u})$, where **u** is a unit vector in the direction of OA.

20. Find the volume of the tetrahedron with vertices at the points (0,0,0), (1,1,1), (2,1,1), and (1,2,1).

21. Determine a unit vector in the plane of the vectors $\mathbf{i} + \mathbf{j}$ and $\mathbf{j} + \mathbf{k}$ and perpendicular to the vector $\mathbf{i} + \mathbf{j} + \mathbf{k}$.

22. If u is a unit vector, prove that $\mathbf{u} \times (\mathbf{a} \times \mathbf{u})$ is the vector projection of a on a plane perpendicular to u.

23. Prove that $\mathbf{a} \times (\mathbf{b} \times \mathbf{c}) + \mathbf{b} \times (\mathbf{c} \times \mathbf{a}) + \mathbf{c} \times (\mathbf{a} \times \mathbf{b}) = \mathbf{0}$.

24. (a) Prove that

$$(\mathbf{a} \times \mathbf{b}) \times (\mathbf{c} \times \mathbf{d}) = \begin{cases} (\mathbf{a} \mathbf{b} \mathbf{d})\mathbf{c} - (\mathbf{a} \mathbf{b} \mathbf{c})\mathbf{d}, \\ (\mathbf{a} \mathbf{c} \mathbf{d})\mathbf{b} - (\mathbf{b} \mathbf{c} \mathbf{d})\mathbf{a}. \end{cases}$$

(To obtain the first form, write temporarily $\mathbf{u} = \mathbf{a} \times \mathbf{b}$.)

(b) Show that this vector is in the direction of the intersection of a plane including the vectors \mathbf{a} and \mathbf{b} with one including \mathbf{c} and \mathbf{d} .

Section 6.5

25. If F is a function of t, find the derivative of

$$\mathbf{F} \cdot \frac{d\mathbf{F}}{dt} \times \frac{d^2\mathbf{F}}{dt^2} \, .$$

26. At time t, the vector from the origin to a moving point is

$$\mathbf{r} = \mathbf{a}\cos\omega t + \mathbf{b}\sin\omega t,$$

where **a**, **b**, and ω are constants.

(a) Find the velocity $\mathbf{v} = \frac{d\mathbf{r}}{dt}$ and prove that $\mathbf{r} \times \mathbf{v}$ is constant, so that the curve traced out lies in a plane.

(b) Show that the acceleration is directed toward the origin and is proportional to the distance from the origin.

27. Let r represent the vector from a fixed origin O to a moving particle of mass m, subject to a force F.

(a) If **H** denotes the moment of the momentum vector $m\mathbf{v} = m\frac{d\mathbf{r}}{dt}$ about O, prove that

$$\frac{d\mathbf{H}}{dt} = m \frac{d}{dt} (\mathbf{r} \times \mathbf{v}) = \mathbf{r} \times \mathbf{F} = \mathbf{M},$$

where M is the moment of the force F about O.

(b) If the force F always passes through the fixed point O, show that

$$\frac{d}{dt}(\mathbf{r}\times\mathbf{v})=\mathbf{0},$$

and hence deduce that $\mathbf{r} \times \mathbf{v} = \mathbf{h}$, where **h** is a constant vector. Deduce also that the motion is in a plane.

(c) Show that $|\mathbf{r} \times \mathbf{v}|$ is twice the area over which the vector \mathbf{r} would sweep in unit time if \mathbf{v} were constant, and hence deduce that, when a mass at P is subject to a "central" force, which always passes through a fixed point O, the vector \overrightarrow{OP}

28. If **u** is a unit vector originating at a fixed point O, and rotating about a fixed vector ω through O, with angular velocity of constant magnitude ω , show that

$$\frac{d\mathbf{u}}{dt} = \mathbf{\omega} \times \mathbf{u}$$

29. Let $\mathbf{r} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$ represent the position vector from a fixed origin O to a point P, and suppose that the xyz axis system is rotating about a fixed vector $\boldsymbol{\omega}$ through O, with angular velocity of constant magnitude $\boldsymbol{\omega}$.

(a) By calculating $\frac{d\mathbf{r}}{dt}$, and noticing that $\frac{d\mathbf{i}}{dt} = \mathbf{\omega} \times \mathbf{i}$, and so forth, obtain the velocity vector in the form

$$\mathbf{v} = \mathbf{v}_0 + \boldsymbol{\omega} \times \mathbf{r},$$

where the vector

$$\mathbf{v_0} = \frac{dx}{dt} \mathbf{i} + \frac{dy}{dt} \mathbf{j} + \frac{dz}{dt} \mathbf{k}$$

is the velocity vector which would be obtained if the axes were fixed.

(b) Obtain the acceleration vector in the form

moves in a plane and sweeps over equal areas in equal times.

$$\mathbf{a} = \mathbf{a}_0 + 2\boldsymbol{\omega} \times \mathbf{v}_0 + \boldsymbol{\omega} \times (\boldsymbol{\omega} \times \mathbf{r}),$$

where v_0 is defined in part (a), and where

$$\mathbf{a}_0 = \frac{d^2x}{dt^2}\mathbf{i} + \frac{d^2y}{dt^2}\mathbf{j} + \frac{d^2z}{dt^2}\mathbf{k}.$$

30. If the system of Problem 29 is rotating with angular velocity of constant magnitude ω about the z axis, the z axis being fixed, show that the equations of motion for a point mass m are of the form

$$m\left(\frac{d^2x}{dt^2} - 2\omega \frac{dy}{dt} - \omega^2 x\right) = F_x,$$
$$m\left(\frac{d^2y}{dt^2} + 2\omega \frac{dx}{dt} - \omega^2 y\right) = F_y,$$
$$m\frac{d^2z}{dt^2} = F_z$$

where F_x , F_y , and F_z are the components of the external force along the respective rotating axes. (Notice that the mass hence behaves as though the axes were fixed, with an additional force $2m\omega \frac{dy}{dt} + m\omega^2 x$ acting in the positive x direction and an additional force $-2m\omega \frac{dx}{dt} + m\omega^2 y$ acting in the positive y direction.)

Section 6.6

31. (a) If θ is polar angle, show that the vectors

$$\mathbf{u}_1 = \mathbf{i} \cos \theta + \mathbf{j} \sin \theta, \quad \mathbf{u}_2 = -\mathbf{i} \sin \theta + \mathbf{j} \cos \theta$$

are perpendicular unit vectors in the radial and circumferential directions, respectively, in the xy plane, and that

$$rac{d {f u}_1}{d heta} = {f u}_2, \qquad rac{d {f u}_2}{d heta} = -{f u}_1.$$

(b) For points on a plane curve in polar coordinates, the position vector is of the form $\mathbf{r} = r \mathbf{u}_1$. By differentiation with respect to time *t*, obtain expressions for the vectors of velocity and acceleration of a point moving along the curve and show that the radial and circumferential components are of the form

$$v_r = \dot{r},$$
 $v_{\theta} = r\theta,$
 $a_r = \ddot{r} - r\dot{\theta}^2,$ $a_{\theta} = r\ddot{\theta} + 2\dot{r}\dot{\theta},$

where a dot denotes time differentiation.

(Notice, for example, that
$$\frac{d\mathbf{u_1}}{dt} = \frac{d\mathbf{u_1}}{d\theta}\frac{d\theta}{dt} = \theta \mathbf{u_2}$$
.)

32. (a) Verify that the parametric equations

 $x = a \cos t$, $y = a \sin t$, z = ct

specify a right circular helix in space.

(b) Show that for this curve there follows

$$ds = \sqrt{a^2 + c^2} \, dt,$$

where s is arc length.

(c) Determine the unit tangent, principal normal, and binormal vectors, and show that the radii of curvature and torsion are given by

$$ho=rac{a^2+c^2}{a},\quad au=rac{a^2+c^2}{c}.$$

(d) Show that the osculating, normal, and rectifying planes at the point $(a, 0, 2\pi c)$ are specified by the respective equations

$$az - cy = 2\pi ca$$
, $ay = 2\pi c^2$, $x = a$.

In Problems 33–36, the position vector to a curve C is assumed to be expressed in the form $\mathbf{r} = x(t)\mathbf{i} + y(t)\mathbf{j} + z(t)\mathbf{k}$, and a prime is used to denote differentiation with respect to the parameter t, whereas s denotes arc length along C.

33. Establish the relations

(a)
$$s' = |\mathbf{r}'| = \sqrt{\mathbf{r}' \cdot \mathbf{r}'}$$
, (b) $\mathbf{u} = \frac{\mathbf{r}'}{|\mathbf{r}'|} = \frac{\mathbf{r}'}{\sqrt{\mathbf{r}' \cdot \mathbf{r}'}}$.

34. (a) Show that

$$\frac{1}{\rho}\mathbf{n} = \frac{d\mathbf{u}}{ds} = \frac{\mathbf{u}'}{s'} = \frac{(\mathbf{r}'\cdot\mathbf{r}')\mathbf{r}'' - (\mathbf{r}'\cdot\mathbf{r}'')\mathbf{r}'}{|\mathbf{r}'|^4} = \frac{(\mathbf{r}'\times\mathbf{r}'')\times\mathbf{r}'}{|\mathbf{r}'|^4}.$$

(b) Use the identity of Lagrange [Equation (36)] to obtain the result

$$|(\mathbf{r}' \times \mathbf{r}'') \times \mathbf{r}'| = |\mathbf{r}' \times \mathbf{r}''| |\mathbf{r}'|.$$

(c) Deduce the following results:

$$\frac{1}{\rho} = \frac{|\mathbf{r}' \times \mathbf{r}''|}{|\mathbf{r}'|^3}, \qquad \mathbf{n} = \frac{(\mathbf{r}' \times \mathbf{r}'') \times \mathbf{r}'}{|\mathbf{r}' \times \mathbf{r}''| |\mathbf{r}'|}$$

35. Use Equation (52) and the results of Problems 33(b) and 34(c) to obtain the result

$$\mathbf{b} = \frac{\mathbf{r'} \times \mathbf{r''}}{|\mathbf{r'} \times \mathbf{r''}|} \,.$$

36. (a) Verify that

$$\mathbf{u} \times \frac{d\mathbf{u}}{ds} \cdot \frac{d^2\mathbf{u}}{ds^2} = \begin{pmatrix} \mathbf{r}' \\ s' \end{pmatrix} \times \begin{pmatrix} s'\mathbf{r}'' - s''\mathbf{r}' \\ s'^3 \end{pmatrix} \cdot \begin{pmatrix} \mathbf{r}'' + \cdots \\ s'^3 \end{pmatrix} = \frac{(\mathbf{r}' \mathbf{r}'' \mathbf{r}'')}{s'^6},$$

where the omitted terms in the third factor of the second expression involve \mathbf{r}' and \mathbf{r}'' linearly, and hence do not affect the value of the product.

(b) Using the relation

$$\frac{1}{\tau} = \rho^2 \left(\mathbf{u} \times \frac{d\mathbf{u}}{ds} \cdot \frac{d^2\mathbf{u}}{ds^2} \right)$$

and the results of Problems 34(c) and 36(a), obtain the result

$$\frac{1}{\tau} = \frac{(\mathbf{r}' \, \mathbf{r}'' \, \mathbf{r}'')}{|\mathbf{r}' \times \mathbf{r}''|^2}.$$

37. Determine u, n, b, ρ , and τ for the right circular helix

 $x = a \cos t$, $y = a \sin t$, z = ct

by using preceding results, and compare with the results of Problem 32. (The results of Problems 33-36 are particularly useful when t cannot be simply expressed in terms of the arc length s.)

38. Bending of a rod. Suppose that a thin rod of uniform circular cross section is bent and twisted in such a way that its axis coincides with a space curve C. Let the intensity of the applied load per unit distance along C have components p_u , p_n , and p_b along the tangent, principal normal, and binormal, respectively, of the deformed rod and write $\mathbf{p} = p_u \mathbf{u} + p_n \mathbf{n} + p_b \mathbf{b}$. Similarly, write $\mathbf{m} = m_u \mathbf{u} + m_n \mathbf{n} + m_b \mathbf{b}$ for the vector whose components are intensities of distributed applied couples along the deformed rod. Let the influence of that part of the rod beyond a section at distance s from one end upon the remaining part be resolved into a force vector $\mathbf{F} = T \mathbf{u} + Q \mathbf{n} + R \mathbf{b}$ and a moment vector $\mathbf{M} = H \mathbf{u} + L \mathbf{n} + M \mathbf{b}$, so that T is tension, Q and R shear forces, H a twisting moment, and L and M bending moments.

Problems

(a) By considering static equilibrium of a section between s and $s + \Delta s$, and proceeding to the limit as $\Delta s \rightarrow 0$, show that the conditions

$$\frac{d\mathbf{F}}{ds} + \mathbf{p} = \mathbf{0}, \quad \frac{d\mathbf{M}}{ds} + \mathbf{u} \times \mathbf{F} + \mathbf{m} = \mathbf{0}$$

must be satisfied.

(b) Show that these two vector equations imply the following six equations of equilibrium:

$$\frac{dT}{ds} - \frac{1}{\rho}Q + p_u = 0, \quad \frac{dQ}{ds} + \frac{1}{\rho}T - \frac{1}{\tau}R + p_n = 0, \quad \frac{dR}{ds} + \frac{1}{\tau}Q + p_b = 0,$$
$$\frac{dH}{ds} - \frac{1}{\rho}L + m_u = 0, \quad \frac{dL}{ds} + \frac{1}{\rho}H - \frac{1}{\tau}M - R + m_n = 0,$$
$$\frac{dM}{ds} + \frac{1}{\tau}L + Q + m_b = 0.$$

[Notice that this set comprises six equations in the eight unknown quantities T, Q, R, H, L, M, ρ , and τ . For an *elastic* rod, these equations are conventionally supplemented by the two equations

$$M = A\left(\frac{1}{\rho}-\frac{1}{\rho_0}\right), \quad H = B\left(\frac{1}{\tau}-\frac{1}{\tau_0}\right),$$

where ρ_0 and τ_0 are the radii of curvature and torsion of the undeformed rod and A and B are bending and torsional stiffnesses associated with the material and cross section of the rod.]

(c) Suppose that the rod is originally straight and untwisted, so that $1/\rho_0$ and $1/\tau_0 = 0$, and is subjected only to a transverse load distribution of intensity p_n (and to end constraints). If, in addition, small deformations are assumed, so that $1/\rho$ and $1/\tau$ are assumed to be negligible in the *equilibrium* equations for the deformed rod, show that the equations of part (b) are satisfied by T = R = H = L = 0 if M and Q satisfy the equations

$$\frac{dQ}{ds}+p_n=0, \quad \frac{dM}{ds}+Q=0.$$

[With the supplementary equation $M = EI/\rho$, where EI is the bending stiffness, these are the basic equations of the elementary theory of small deflections of laterally loaded originally straight rods or beams. (See Section 5.3, where $S \equiv -Q$ and where s is replaced by its projection x along the undeformed axis.)]

Section 6.7

39. The temperature at any point in space is given by

$$T = xy + yz + zx$$

(a) Find the direction cosines of the direction in which the temperature changes most rapidly with distance from the point (1,1,1), and determine the maximum rate of change.

(b) Find the derivative of T in the direction of the vector 3i - 4k at the point (1,1,1).

40. If r and θ are polar coordinates in the xy plane, determine grad r and grad θ .

41. Determine the direction cosines of grad $\varphi(x,y,z)$.

42. Determine a unit vector normal to the surface

$$x^3 - xyz + z^3 = 1$$

at the point (1,1,1).

Section 6.8

43. Prove that $\mathbf{u} \times \nabla \cdot \mathbf{v} = \mathbf{u} \cdot \nabla \times \mathbf{v}$.

44. Prove that $(\mathbf{u} \cdot \nabla)\varphi = \mathbf{u} \cdot (\nabla \varphi)$.

45. (a) Determine the derivative of the function $x^2 + y^2 + z^2$ in the direction specified by the direction cosines *l*, *m*, and *n*.

(b) Determine the derivative of the vector $x^2\mathbf{i} + y^2\mathbf{j} + z^2\mathbf{k}$ in the direction specified by the direction cosines l, m, and n.

46. Prove that $\nabla \times F$ is not necessarily perpendicular to F, by giving a suitable example.

Section 6.9

47. Prove that $\nabla \cdot (\nabla \varphi) = (\nabla \cdot \nabla) \varphi \equiv \nabla^2 \varphi$.

48. (a) Show that

$$\nabla (\mathbf{v} \cdot \mathbf{v}) = 2\mathbf{v} \cdot \nabla \mathbf{v} + 2\mathbf{v} \times (\nabla \times \mathbf{v}),$$

and deduce that the relation

$$\mathbf{v} \cdot \nabla \mathbf{v} = \frac{1}{2} \nabla v^2$$

is true when $\nabla \times \mathbf{v} = \mathbf{0}$ or, more generally, when $\nabla \times \mathbf{v}$ is perpendicular to \mathbf{v} .

(b) Verify the general result of part (a) when $\mathbf{v} = xy \mathbf{i} + y^2 \mathbf{j}$.

49. Show that $\nabla^2 \mathbf{u} = \mathbf{0}$ if $\nabla \times \mathbf{u} = \nabla \varphi$ and $\nabla \cdot \mathbf{u} = \mathbf{0}$, where φ is a scalar function.

50. If $\mathbf{v} = \varphi_1 \nabla \varphi_2$, prove that $\nabla \times \mathbf{v}$ is perpendicular to \mathbf{v} .

51. If there exists a family of surfaces $\varphi(x,y,z) = \text{constant}$ such that $\mathbf{v}(x,y,z)$ is perpendicular at every point to the surface $\varphi = \text{constant}$ which passes through that point, show that $\mathbf{v} \cdot (\nabla \times \mathbf{v}) = \mathbf{0}$. (Use the result of Problem 50).

Sections 6.10, 6.11

52. For each of the following vector functions, determine whether the equation $\nabla \varphi = \mathbf{F}$ possesses a solution, and determine that solution if it exists:

- (a) $\mathbf{F} = 2xyz^3 \mathbf{i} (x^2z^3 + 2y)\mathbf{j} + 3x^2yz^2\mathbf{k}$,
- (b) $\mathbf{F} = 2xy \mathbf{i} + (x^2 + 2yz) \mathbf{j} + (y^2 + 1) \mathbf{k}$.

Problems

53. For each vector function F defined in Problem 52, determine the value of the integral $\int_C \mathbf{F} \cdot d\mathbf{r}$ from the origin to the point (1,1,1) along the curve specified by the simultaneous equations $y = x^2$, $z = x^3$.

54. For each vector function defined in Problem 52, determine the value of the integral $\oint_C \mathbf{F} \cdot d\mathbf{r}$ around the unit circle with center at the origin, in the xy plane.

55. If $P dx + Q dy + R dz = d\varphi$, where φ is single-valued, show that

$$\varphi(x,y,z) = \int_0^x P(t,0,0) \, dt \, + \, \int_0^y Q(x,t,0) \, dt \, + \, \int_0^z R(x,y,t) \, dt \, + \, \text{constant},$$

where x is held constant in the second integral and both x and y are held constant in the third integral. [Integrate $d\varphi$ along straight line segments from (0,0,0) to (x,0,0) to (x,y,0) to (x,y,z).]

56. Show that the use of the formula obtained in Problem 55, for the determination of φ such that $d\varphi = y^2 dx + 2(xy + z) dy + 2y dz$, leads to the result

$$\varphi(x,y,z) = \int_0^x 0 \, dt + \int_0^y 2xt \, dt + \int_0^z 2y \, dt + c = xy^2 + 2yz + c.$$

(Compare the determination in the text, page 285.)

Section 6.12

57. (a) If S is a portion of a surface specified by an equation of the form z = f(x,y), show that the unit normal vector n which points *upward* (in the positive z direction) from S is given by

$$\mathbf{n} = \frac{-\mathbf{i}\frac{\partial f}{\partial x} - \mathbf{j}\frac{\partial f}{\partial y} + \mathbf{k}}{\sqrt{1 + \left(\frac{\partial f}{\partial x}\right)^2 + \left(\frac{\partial f}{\partial y}\right)^2}}$$

(b) Deduce that, in this case, if $\mathbf{F} = P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}$, there follows

$$\iint_{S} \mathbf{F} \cdot d\boldsymbol{\sigma} = \iint_{A} \left(R - P \frac{\partial f}{\partial x} - Q \frac{\partial f}{\partial y} \right) dx dy$$

where A is the projection of S on the xy plane, and where z is to be replaced by f(x,y) in the expressions for P, Q, and R.

58. Evaluate the surface integral of the vector $\mathbf{F} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$ over that portion of the surface z = xy + 1 which covers the square $0 \le x \le 1, 0 \le y \le 1$ in the xy plane.

59. Evaluate the surface integral of the vector $\mathbf{F} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$ over that portion of the paraboloid $z = x^2 + y^2$ which is inside the cylinder $x^2 + y^2 = 1$.

60. Evaluate the surface integral of the vector $\mathbf{F} = x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$ over the closed surface of the cube bounded by the planes $x = \pm 1$, $y = \pm 1$, $z = \pm 1$.

61. Evaluate the surface integral of $\mathbf{F} = yz \mathbf{i} + xz \mathbf{j} + xy \mathbf{k}$ over the closed boundary of the region bounded below by $z = x^2 + y^2$ and above by z = 1.

Section 6.13

62. Evaluate the integral of Problem 60 by using the divergence theorem.

63. If r is the position vector $x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$, show that

(a)
$$\oiint_{S} \mathbf{r} \cdot \mathbf{n} \, d\sigma = 3V$$
, (b) $\oiint_{S} x \, \mathbf{r} \cdot \mathbf{n} \, d\sigma = 4V\bar{x}$,

where V is the volume enclosed by S and \bar{x} is the x coordinate of its center of gravity.

64. By using the physical argument of Section 6.13, show that, if \mathscr{R} is the region inside a closed surface S and outside a closed surface Σ , then the divergence theorem takes the form

$$\iint \iint_{\mathscr{R}} \nabla \cdot \mathbf{V} \, d\tau = \oiint_{S} \mathbf{V} \cdot \mathbf{n} \, d\sigma + \oiint_{\Sigma} \mathbf{V} \cdot \mathbf{n} \, d\sigma,$$

where **n** points outward from \mathscr{R} along both S and Σ , so that n points into the deleted region enclosed by Σ in the last integral.

65. Suppose that the closed boundary S of the region \mathscr{R} is cut by lines in the z direction in not more than two points. Denote the upper part of S by S_+ and the lower part by S_- , and denote the projection of either part on the xy plane by A.

(a) Show that

$$\int\!\!\int\!\!\int_{\mathscr{R}} \frac{\partial R}{\partial z} \, dx \, dy \, dz = \int\!\!\int_{A} R_{S_{+}} \, dx \, dy - \int\!\!\int_{A} R_{S_{-}} \, dx \, dy,$$

where $R_{S_{+}}$ and $R_{S_{-}}$ denote the value of R(x,y,z) on S_{+} and on S_{-} , respectively. (b) Noticing that

$$dx dy = d\sigma \cos \gamma$$

on S_+ and

$$dx\,dy=-d\sigma\cos\gamma$$

on S₋, where $\cos \gamma$ is the z direction cosine of the outward normal, deduce the relation

in this case. [Corresponding results in the x and y directions establish the divergence theorem (117) analytically when \mathcal{R} is a convex region. Proofs for other regions essentially depend upon dividing them into convex regions.]

66. (a) If V is a two-dimensional vector in the xy plane, show that

$$\int \int_A \nabla \cdot \mathbf{V} \, dx \, dy = \oint_C \mathbf{V} \cdot \mathbf{n} \, ds,$$

where A is a simply connected region in the xy plane with closed boundary C, and **n** is the unit outward normal vector along the curve C. (Apply the divergence theorem to the vector V over a three-dimensional region which consists of the interior of a right cylinder of unit height having the region A as its lower base.)

(b) If A is the region inside a closed curve C but outside a closed interior curve Γ in the xy plane, show that there follows

$$\iint_A \nabla \cdot \mathbf{V} \, dx \, dy = \oint_C \mathbf{V} \cdot \mathbf{n} \, ds + \oint_\Gamma \mathbf{V} \cdot \mathbf{n} \, ds,$$

where **n** points outward from A along both C and Γ , so that **n** points into the deleted region enclosed by Γ in the last integral.

67. Let \mathscr{R} be a region bounded by a closed surface S.

(a) Establish the "gradient theorem,"

$$\iint \iint_{\mathscr{R}} \nabla \varphi \ d\tau = \oiint_{S} \varphi \ d\sigma,$$

in rectangular coordinates by applying the divergence theorem (117) to each component.

(b) In a similar way, establish the "curl theorem,"

$$\iint \iint_{\mathscr{R}} \nabla \times \mathbf{F} \, d\tau = \oiint_{S} \, d\sigma \times \mathbf{F}.$$

(c) Show that

$$\iint \iint_{\mathscr{R}} (\varphi \, \nabla \cdot \mathbf{F} \,+\, \mathbf{F} \cdot \nabla \varphi) \, d\tau \,=\, \oiint_{\mathcal{S}} \varphi \, \mathbf{F} \cdot d\sigma,$$

by applying (116) to the vector $\mathbf{V} = \varphi \mathbf{F}$.

Section 6.14

68. Show that Equations (119), (122), and (123) imply the relations

$$\iint_{A} \left[\varphi_{1} \nabla^{2} \varphi_{2} - \varphi_{2} \nabla^{2} \varphi_{1}\right] dx dy = \oint_{C} \left(\varphi_{1} \frac{\partial \varphi_{2}}{\partial n} - \varphi_{2} \frac{\partial \varphi_{1}}{\partial n}\right) ds,$$
$$\iint_{A} \left[\varphi \nabla^{2} \varphi + (\nabla \varphi)^{2}\right] dx dy = \oint_{C} \varphi \frac{\partial \varphi}{\partial n} ds,$$
$$\iint_{A} \nabla^{2} \varphi dx dy = \oint_{C} \frac{\partial \varphi}{\partial n} ds,$$

where $\varphi = \varphi(x,y)$ and $\nabla^2 = \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2}$, and where A is an area in the xy plane bounded by C.

69. Verify the validity of the second equation of Problem 68 when $\varphi = x$ and A is the interior of the circle of radius a with center at the origin. Use polar coordinates in the right-hand member and notice that, on the boundary C, there follows

$$\frac{\partial x}{\partial n} = \frac{\partial x}{\partial r} = \cos \theta.$$

70. Suppose that $\nabla^2 \varphi = 0$ everywhere in a region \mathscr{R} bounded by a closed surface S. Establish the following results in that case:

(a)
$$\int \int_{S} \frac{\partial \varphi}{\partial n} d\sigma = 0$$
, (b) $\iint_{S} \varphi \frac{\partial \varphi}{\partial n} d\sigma = \int \int \int_{\mathscr{R}} (\nabla \varphi)^{2} d\tau$.

71. By writing $\varphi_1 = \varphi$ and taking $\varphi_2 = \nabla^2 \varphi$ in (119), obtain the relation

Section 6.15

72. Suppose that φ satisfies Laplace's equation everywhere in a region \mathscr{R} bounded by a closed surface S.

(a) If $\frac{\partial \varphi}{\partial n}$ vanishes everywhere on S, deduce from Problem 70(b) that φ must have a constant value in \mathcal{R} .

(b) If φ vanishes everywhere on S, deduce from Problem 70(b) that φ must vanish everywhere in \mathcal{R} .

73. Suppose that φ satisfies the equation

$$abla^4 arphi = rac{\partial^4 arphi}{\partial x^4} + 2 \, rac{\partial^4 arphi}{\partial x^2 \, \partial y^2} + rac{\partial^4 arphi}{\partial y^4} = 0,$$

known as the *bi-Laplacian equation*, everywhere in a region \mathscr{R} bounded by a closed surface S, and that both φ and $\frac{\partial \varphi}{\partial n}$ vanish everywhere on S. Deduce from Problem 71 that φ then must also satisfy Laplace's equation everywhere in \mathscr{R} and hence, by virtue of the result of Problem 72(b), must vanish everywhere in \mathscr{R} .

Section 6.16

74. If S is a closed surface in a region \mathcal{R} where the vector V is continuously differentiable, show that

$$\oint S_{S} \mathbf{n} \cdot \nabla \times \mathbf{V} \, d\sigma = \mathbf{0}.$$

75. Verify the truth of Stokes's theorem, as given in Equation (129), in the case when $V = y \mathbf{i} + 2x \mathbf{j} + z \mathbf{k}$, if C is the circle $x^2 + y^2 = 1$ (or $x = \cos t$, $y = \sin t$) in the xy plane, and S is the plane area bounded by C.

76. Evaluate the integral $\iint_S \mathbf{n} \cdot \nabla \times \mathbf{V} \, d\sigma$ over the part of the unit sphere $x^2 + y^2 + z^2 = 1$ above the xy plane, where $\mathbf{V} = y$ i.

77. For any closed circuit C bounding any open surface S in a region \mathcal{R} , the electric intensity vector E and the magnetic intensity vector H satisfy the relations

$$\oint_C \mathbf{E} \cdot d\mathbf{r} = -\alpha \frac{\partial}{\partial t} \iint_S \mathbf{H} \cdot d\sigma, \qquad \oint_C \mathbf{H} \cdot d\mathbf{r} = \beta \frac{\partial}{\partial t} \iint_S \mathbf{E} \cdot d\sigma,$$

where α and β are certain constants. Use Stokes's theorem to transform these relations to the form

$$\iint_{S} \left(\nabla \times \mathbf{E} + \alpha \, \frac{\partial \mathbf{H}}{\partial t} \right) \cdot d\sigma = 0, \qquad \iint_{S} \left(\nabla \times \mathbf{H} - \beta \, \frac{\partial \mathbf{E}}{\partial t} \right) \cdot d\sigma = 0.$$

From the arbitrariness of S, deduce Maxwell's equations in the form

$$\nabla \times \mathbf{E} = -\alpha \frac{\partial \mathbf{H}}{\partial t}, \qquad \nabla \times \mathbf{H} = \beta \frac{\partial \mathbf{E}}{\partial t}.$$

78. In a region free of electric and magnetic charges, it is true that $\nabla \cdot \mathbf{E} = 0$ and $\nabla \cdot \mathbf{H} = 0$. By eliminating E and H successively between the equations of Problem 77, and using this fact, deduce that E and H both satisfy the vector wave equation

$$\nabla^2 \mathbf{V} = \alpha \beta \, \frac{\partial^2 \mathbf{V}}{\partial t^2}$$

in such a region.

79. Obtain Equation (135) from the two-dimensional specialization of the divergence theorem (see Problem 66) by taking V = Q i - P j. [Notice that

$$\frac{dx}{ds} = \cos \varphi$$
 and $\frac{dy}{ds} = \sin \varphi$,

where φ is the angle between the tangent (pointing in the positive s direction) and the positive x axis, and that the direction angles α and β of n are related to φ by the equations $\alpha = \pi/2 - \beta = \varphi - \pi/2$.]

80. Show that

(a)
$$\frac{1}{3} \oint_C (x^2 dy - xy dx) = A\bar{x}$$
, (b) $\frac{1}{4} \oint_C (x^3 dy - x^2y dx) = I_y$,

where A is the area bounded by C, (\bar{x}, \bar{y}) is its center of gravity, and I_y its moment of inertia about the y axis.

81. Let S be an open surface bounded by a closed curve C.

(a) Establish the relation

$$\int \int_{S} d\mathbf{\sigma} \times \nabla \varphi = \oint_{C} \varphi \, d\mathbf{r}$$

in rectangular coordinates by applying Stokes's theorem (134) to each component.

(b) Show that

$$\oint_C \varphi_1 \, \nabla \varphi_2 \cdot d\mathbf{r} = - \oint_C \varphi_2 \, \nabla \varphi_1 \cdot d\mathbf{r}$$

by applying (129) to the vector $\mathbf{V} = \nabla(\varphi_1 \varphi_2)$.

(c) Show that

$$\int \int_{S} (\nabla \varphi \times \mathbf{F} + \varphi \nabla \times \mathbf{F}) \cdot d\boldsymbol{\sigma} = \oint_{C} \varphi \mathbf{F} \cdot d\mathbf{r},$$

by applying (129) to the vector $\mathbf{V} = \varphi \mathbf{F}$.

Section 6.17

82. Elliptical cylindrical coordinates may be defined by the equations

 $x = a \cosh u \cos v,$ $y = a \sinh u \sin v,$ z = zwhere $u \ge 0$ and $0 \le v < 2\pi$. (a) Show that this system of coordinates (u,v,z) is orthogonal [by verifying that Equation (145) is satisfied].

(b) Show that in the xy plane a curve u = constant is an ellipse with semiaxes $a \cosh u$, in the x direction, and $a \sinh u$, in the y direction; also that a curve v = constant is half of one branch of an hyperbola with semi-axes $a \cos v$ and $a \sin v$. In particular, show that the locus u = 0 degenerates into the segment (-a,a) of the x axis, while the loci v = 0 and $v = \pi$ are respectively the positive and negative exteriors of this segment; also that the loci $v = \pi/2$ and $v = 3\pi/2$ are respectively the positive and negative portions of the y axis. Sketch and label in a single diagram the curves u = 0, 1 and $v = 0, \pi/4, \pi/2, 3\pi/4, \pi, 5\pi/4, 3\pi/2, and <math>7\pi/4$.

83. For the coordinates of Problem 82 derive the relations analogous to those of Equations (162b-e) for circular cylindrical coordinates. In particular, verify that

$$h_{u} = h_{v} = a\sqrt{\cosh^{2} u - \cos^{2} v}, \qquad h_{z} = 1,$$
$$\mathbf{u}_{1} = \frac{\mathbf{i} \sinh u \cos v + \mathbf{j} \cosh u \sin v}{\sqrt{\cosh^{2} u - \cos^{2} v}},$$
$$\mathbf{u}_{2} = \frac{-\mathbf{i} \cosh u \sin v + \mathbf{j} \sinh u \cos v}{\sqrt{\cosh^{2} u - \cos^{2} v}},$$
$$\nabla f = \frac{1}{a\sqrt{\cosh^{2} u - \cos^{2} v}} \left(\mathbf{u}_{1} \frac{\partial f}{\partial u} + \mathbf{u}_{2} \frac{\partial f}{\partial v}\right) + \mathbf{k} \frac{\partial f}{\partial z},$$
$$\nabla^{2} f = \frac{1}{a^{2} (\cosh^{2} u - \cos^{2} v)} \left(\frac{\partial^{2} f}{\partial u^{2}} + \frac{\partial^{2} f}{\partial v^{2}}\right) + \frac{\partial^{2} f}{\partial z^{2}}.$$

Show also that for large values of u there follows

 $\mathbf{u}_1 \sim \mathbf{i} \cos v + \mathbf{j} \sin v, \quad \mathbf{u}_2 \sim -\mathbf{i} \sin v + \mathbf{j} \cos v,$

and

 $\mathbf{u}_1 \cos v - \mathbf{u}_2 \sin v \sim \mathbf{i}, \quad \mathbf{u}_1 \sin v + \mathbf{u}_2 \cos v \sim \mathbf{j}.$

84. Parabolic cylindrical coordinates may be defined by the equations

$$x = \frac{1}{2}(u^2 - v^2), \quad y = uv, \quad z = z,$$

where $-\infty < u < \infty$ and $v \ge 0$.

(a) Show that this system is orthogonal.

(b) Show that in the xy plane a curve v = constant is a parabola symmetrical about the x axis and opening to the right, while a curve u = constant is one half of a similar parabola opening to the left. In particular, show that the locus v = 0 is the positive x axis, while the locus u = 0 is the negative x axis, and that the positive y axis is given by u = v and the negative y axis by u = -v. Sketch and label in a single diagram the curves $u = 0, \pm 1$ and v = 0, 1.

(c) Perform the calculations necessary to show that the Laplacian is of the form

$$\nabla^2 f = rac{1}{u^2 + v^2} \left(rac{\partial^2 f}{\partial u^2} + rac{\partial^2 f}{\partial v^2}
ight) + rac{\partial^2 f}{\partial z^2}.$$

Problems

85. Paraboloidal coordinates correspond to a system in which the plane configuration of Problem 84 is rotated about the axis of symmetry of the two sets of parabolas. The axis of rotation is then conventionally taken as the z axis.

(a) In Problem 84 replace x by z and y by $\sqrt{x^2 + y^2}$, the distance from the z axis; then write $y = x \tan \theta$, so that θ is the circumferential angle, and show that the coordinate transformation becomes

 $x = uv \cos \theta$, $y = uv \sin \theta$, $z = \frac{1}{2}(u^2 - v^2)$,

where now $u \ge 0$ and $v \ge 0$.

(b) Obtain the Laplacian in the form

$$\nabla^2 f = \frac{1}{u^2 + v^2} \left[\frac{1}{u} \frac{\partial}{\partial u} \left(u \frac{\partial f}{\partial u} \right) + \frac{1}{v} \frac{\partial}{\partial v} \left(v \frac{\partial f}{\partial v} \right) \right] + \frac{1}{u^2 v^2} \frac{\partial^2 f}{\partial \theta^2}.$$

86. In a *translation* and *rotation* of axes, in which the origin in the new x'y'z' plane is taken at the point (a,b,c) in the xyz plane, and in which the directions of the x', y', and z' axes are specified by the direction cosines (l_1,m_1,n_1) , (l_2,m_2,n_2) , and (l_3,m_3,n_3) relative to the original axes, the transformation of coordinates is of the form

$$x = a + l_1 x' + l_2 y' + l_3 z',$$

$$y = b + m_1 x' + m_2 y' + m_3 z',$$

$$z = c + n_1 x' + n_2 y' + n_3 z'.$$

Show that Laplace's equation is of the form

$$\frac{\partial^2 f}{\partial x'^2} + \frac{\partial^2 f}{\partial y'^2} + \frac{\partial^2 f}{\partial z'^2} = 0$$

in terms of the new variables.

87. Suppose that the coordinates u_1 and u_2 are related to x and y by an equation of the form

$$x + iy = F(u_1 + iu_2),$$

where $i^2 = -1$, so that x is the real part of $F(u_1 + iu_2)$ and y the imaginary part, and that u_1, u_2 , and z are chosen as curvilinear coordinates in space.

(a) Show that, if F is a differentiable function of the argument $u_1 + iu_2$, there follows

$$\frac{\partial x}{\partial u_1} + i \frac{\partial y}{\partial u_1} = F'(u_1 + iu_2), \qquad \frac{\partial x}{\partial u_2} + i \frac{\partial y}{\partial u_2} = i F'(u_1 + iu_2),$$

where a prime denotes differentiation with respect to the complete argument $u_1 + iu_2$, and hence that

$$\sqrt{\left(\frac{\partial x}{\partial u_1}\right)^2 + \left(\frac{\partial y}{\partial u_1}\right)^2} = \sqrt{\left(\frac{\partial x}{\partial u_2}\right)^2 + \left(\frac{\partial y}{\partial u_2}\right)^2} = |F'(u_1 + iu_2)|.$$

(b) Deduce also that

$$\frac{\partial x}{\partial u_1} + i \frac{\partial y}{\partial u_1} = -i \frac{\partial x}{\partial u_2} + \frac{\partial y}{\partial u_2},$$

and hence, by equating real and imaginary parts, that

$$\frac{\partial x}{\partial u_1} = \frac{\partial y}{\partial u_2}, \qquad \frac{\partial y}{\partial u_1} = -\frac{\partial x}{\partial u_2}.$$

(c) With the notation of Section 6.17, deduce that the vectors U_1 , U_2 , and U_3 are mutually orthogonal, and that

$$h_1 = h_2 = |F'(u_1 + iu_2)|, \quad h_3 = 1.$$

Hence show that Laplace's equation is of the form

$$\frac{\partial^2 f}{\partial u_1^2} + \frac{\partial^2 f}{\partial u_2^2} + h^2 \frac{\partial^2 f}{\partial z^2} = 0,$$

where $h = |F'(u_1 + iu_2)|$.

88. (a) If x and y are related to u_1 and u_2 by the equation

$$x + iy = \frac{1}{2}(u_1 + iu_2)^2$$
,

so that $x = \frac{1}{2}(u_1^2 - u_2^2)$ and $y = u_1u_2$, use the result of Problem 87 to obtain Laplace's equation in u_1u_2z coordinates in the form

$$\frac{\partial^2 f}{\partial u_1^2} + \frac{\partial^2 f}{\partial u_2^2} + (u_1^2 + u_2^2) \frac{\partial^2 f}{\partial z^2} = 0.$$

(b) Obtain the same result by using the formulas of Section 6.17. (See also Problem 84.)

Section 6.18

89. If u_r and u_{θ} are the unit tangent vectors in the r and θ directions, in circular cylindrical coordinates, show that

 $\mathbf{i} = \mathbf{u}_r \cos \theta - \mathbf{u}_\theta \sin \theta, \quad \mathbf{j} = \mathbf{u}_r \sin \theta + \mathbf{u}_\theta \cos \theta.$

90. If r, θ , and z are circular cylindrical coordinates, evaluate the following quantities:

(a) $\nabla \theta$, (b) ∇r^n , (c) $\nabla \times u_{\theta}$, (d) $\nabla \cdot [r^{n-1}(u_r \sin n\theta + u_{\theta} \cos n\theta)]$, (e) $\nabla^2(r^2 \cos \theta)$, (f) $\nabla^2(r^n \cos n\theta)$.

91. If \mathbf{u}_r , \mathbf{u}_{φ} , and \mathbf{u}_{θ} are the unit vectors tangent to the coordinates curves in spherical coordinates, show that

$$\mathbf{i} = (\mathbf{u}_r \sin \varphi + \mathbf{u}_\varphi \cos \varphi) \cos \theta - \mathbf{u}_\theta \sin \theta,$$

$$\mathbf{j} = (\mathbf{u}_r \sin \varphi + \mathbf{u}_\varphi \cos \varphi) \sin \theta + \mathbf{u}_\theta \cos \theta,$$

$$\mathbf{k} = \mathbf{u}_r \cos \varphi - \mathbf{u}_\varphi \sin \varphi.$$

92. If r, φ , and θ are spherical coordinates, evaluate the following quantities: (a) $\nabla \varphi$, (b) $\nabla \theta$,

(c)
$$\nabla \cdot [\mathbf{u}_r \cot \varphi - 2 \, \mathbf{u}_{\varphi}],$$
 (d) $\nabla^2 \left[\left(r + \frac{1}{r^2} \right) \cos \varphi \right].$

Problems

93. Show that the unit tangent vectors in spherical coordinates satisfy the following relations:

$$\frac{\partial \mathbf{u}_r}{\partial \varphi} = \mathbf{u}_{\varphi}, \qquad \frac{\partial \mathbf{u}_r}{\partial \theta} = \mathbf{u}_{\theta} \sin \varphi, \qquad \frac{\partial \mathbf{u}_{\varphi}}{\partial \varphi} = -\mathbf{u}_r, \qquad \frac{\partial \mathbf{u}_{\varphi}}{\partial \theta} = \mathbf{u}_{\theta} \cos \varphi,$$
$$\frac{\partial \mathbf{u}_{\theta}}{\partial \varphi} = \mathbf{0}, \qquad \frac{\partial \mathbf{u}_{\theta}}{\partial \theta} = -\mathbf{u}_r \sin \varphi - \mathbf{u}_{\varphi} \cos \varphi.$$

94. By writing the position vector in the form $\mathbf{r} = r \mathbf{u}_r$, and using the results of Problem 93, obtain expressions for components of acceleration along the coordinate curves in spherical coordinates as follows:

$$a_r = \ddot{r} - r\varphi^2 - r\dot{\theta}^2 \sin^2 \varphi,$$

$$a_{\varphi} = 2\dot{r}\dot{\varphi} + r\ddot{\varphi} - r\dot{\theta}^2 \sin \varphi \cos \varphi,$$

$$a_{\theta} = 2\dot{r}\dot{\theta} \sin \varphi + r\ddot{\theta} \sin \varphi + 2r\dot{\varphi}\dot{\theta} \cos \varphi.$$

95. Prove that

$$\nabla^2[r^n P_n(\cos\varphi)] = 0,$$

where r and φ are spherical coordinates and P_n is the Legendre polynomial of order n. [See Equation (169) of Chapter 4.]

96. (a) Evaluate the surface integral

$$\int \int_{S} \mathbf{F} \cdot d\boldsymbol{\sigma},$$

where $\mathbf{F} = x \mathbf{i} - y \mathbf{j} + z \mathbf{k}$ and where S is the lateral surface of the cylinder $x^2 + y^2 = 1$ between the planes z = 0 and z = 1, using right circular cylindrical coordinates.

(b) Check the result by use of the divergence theorem.

97. (a) Evaluate the surface integral of $\mathbf{F} = x \mathbf{i} - y \mathbf{j} + z \mathbf{k}$ over the closed surface of the sphere $x^2 + y^2 + z^2 = 1$, using spherical coordinates.

(b) Check the result by use of the divergence theorem.

Section 6.19

98. Suppose that a flow of an ideal incompressible fluid is free of sources and sinks, and of vortices, and that it takes place parallel to the xy plane.

(a) Show that the velocity potential $\varphi(x,y)$ and the stream function $\psi(x,y)$ are such that $\frac{\partial \varphi}{\partial x} = \frac{\partial \varphi}{\partial x}$

$$V_x = \frac{\partial \varphi}{\partial x} = \frac{\partial \psi}{\partial y}, \qquad V_y = \frac{\partial \varphi}{\partial y} = -\frac{\partial \psi}{\partial x},$$

and that φ and ψ satisfy Laplace's equation.

(b) Show that

$$\varphi(x,y) = \int^{(x,y)} (V_x \, dx \, + \, V_y \, dy), \qquad \psi(x,y) = \int^{(x,y)} (-V_y \, dx \, + \, V_x \, dy),$$

where the line integrals are each evaluated along an arbitrary path from a fixed point to the variable point (x,y).

99. (a) Verify that the function $\varphi(x,y) = x^3 - 3xy^2$ is a velocity potential function.

(b) Determine the velocity vector \mathbf{V} , and its magnitude V.

(c) Determine the stream function $\psi(x,y)$, subject to the condition $\psi(0,0) = 0$, and obtain the equation of the streamlines.

(d) If the uniform density of the fluid is ρ , determine the rate of flow across an arc joining the points (1,1) and (2,2) (i.e., through a cylindrical surface of unit height having this arc as its base).

100. (a) If elliptical cylindrical coordinates (Problems 82 and 83) are used to describe the flow of an ideal fluid parallel to the xy plane and independent of z, show that the velocity potential satisfies the equation

$$\frac{\partial^2 \varphi}{\partial \mu^2} + \frac{\partial^2 \varphi}{\partial v^2} = 0.$$

(b) Show that the corresponding stream function ψ satisfies an equation of the same form and that ψ may be related to φ by the equation

$$d\psi = -\frac{\partial\varphi}{\partial v}\,du + \frac{\partial\varphi}{\partial u}\,dv.$$

101. (a) Show that

$$\varphi = C \cosh u \sin v$$

is a permissible potential function in Problem 100 and that the corresponding stream function is then of the form

$$\psi = -C \sinh u \cos v.$$

if an arbitrary additive constant is discarded.

(b) Show that the streamline $\psi = 0$ consists of the combination of the loci u = 0 and both $v = \pi/2$ and $v = 3\pi/2$, and hence deduce from continuity considerations that neighboring streamlines follow near the y axis toward the x axis, thence around a perpendicular barrier (plate) of breadth 2a back toward the y axis and onward nearly parallel to the y axis.

(c) Show that the flow velocity is given by

$$\mathbf{V} = \frac{C}{a\sqrt{\cosh^2 u - \cos^2 v}} (\mathbf{u}_1 \sinh u \sin v + \mathbf{u}_2 \cosh u \cos v),$$

and that at large distances from the barrier $(u \rightarrow \infty)$ the velocity vector tends to the constant value

$$\mathbf{V} \sim \frac{C}{a} (\mathbf{u}_1 \sin v + \mathbf{u}_2 \cos v) \sim \frac{C}{a} \mathbf{j} \qquad (u \to \infty).$$

Deduce that the potential

$$\varphi = aV_0 \cosh u \sin v$$

corresponds to a flow, around a plate of breadth 2a, which tends to a uniform flow, with velocity V_0 at right angles to the plate, at large distances from it.

(d) Show that at a point (u,v,z) the flow velocity is of magnitude

$$V = V_0 \sqrt{\frac{\cosh^2 u - \sin^2 v}{\cosh^2 u - \cos^2 v}}.$$

In particular, verify that at points on the plate (u = 0) there follows

$$V = V_0 \frac{|x|}{\sqrt{a^2 - x^2}}$$
 $(|x| < a),$

while for points on the x axis outside the plate ($v = 0, \pi$)

$$V = V_0 \frac{|x|}{\sqrt{x^2 - a^2}}$$
 $(|x| > a),$

and for points on the y axis ($v = \pi/2, 3\pi/2$)

$$V = V_0 \frac{|y|}{\sqrt{a^2 + y^2}}.$$

102. Suppose that a source-free and vortex-free flow of an incompressible fluid possesses axial symmetry about the z axis, so that its velocity can be expressed in the form

$$\mathbf{V} = V_z \, \mathbf{k} + V_r \, \mathbf{u}_r$$

in terms of circular cylindrical coordinates.

(a) With $(u_1, u_2, u_3) = (z, r, \theta)$, show that $h_1 = h_2 = 1$ and $h_3 = r$, and that

$$V_z = \frac{\partial \varphi}{\partial z}, \qquad V_r = \frac{\partial \varphi}{\partial r}$$

where $\varphi(z,r)$ satisfies the equation

$$\frac{1}{r}\frac{\partial}{\partial r}\left(r\frac{\partial\varphi}{\partial r}\right)+\frac{\partial^{2}\varphi}{\partial z^{2}}=0.$$

(b) Show that the stream function ψ is determined by the relation

$$d\psi = r(-V_r\,dz + V_z\,dr)$$

and verify that the right-hand member is an exact differential.

(c) Show that the rate of flow through the surface obtained by rotating about the z axis an arc C joining two points P_0 and P_1 in a plane θ = constant is given by

$$2\pi\int_C d\psi = 2\pi[\psi(P_1) - \psi(P_0)].$$

103. (a) Verify that the function $\varphi(z,r) = 2z^2 - r^2$ is a potential function corresponding to the type of flow considered in Problem 102.

(b) Determine the velocity vector V, and its magnitude V.

(c) Determine the stream function ψ , and show that the streamlines are the curves in the planes θ = constant for which $r^2 z$ = constant.

(d) Show that the rate of flow through the surface obtained by rotating about the z axis an arc joining P_0 and P_1 in a plane θ = constant is given by $4\pi\rho(r_1^2z_1 - r_0^2z_0)$.

Section 6.20

104. If, in the developments of Section 6.20, a *body force* F per unit mass is assumed to be active, show that ρ F must be added to the right-hand members of Equations (179) and (182).

105. If a fluid is viscous, the force on a surface element $d\sigma$, due to internal friction, is usually assumed to have a component in any direction equal to μ times the product of $d\sigma$ and the derivative, normal to $d\sigma$, of the velocity component in that direction, where μ is a constant known as the coefficient of viscosity.

(a) Show that the viscous force in the x direction on an element $d\sigma$ then is of the magnitude

$$\mu \mathbf{n} \cdot \nabla V_x \, d\sigma,$$

and that the net viscous force, in the x direction, on the closed boundary S of a region \mathcal{A} is of the magnitude

$$\mu \oint S_{S} \mathbf{n} \cdot \nabla V_{x} \, d\sigma = \mu \iiint_{\mathscr{R}} \nabla^{2} V_{x} \, d\tau.$$

Hence deduce that the viscous force on an element $d\tau$ is equivalent to a body force with component $\mu \nabla^2 V_x d\tau$ in the x direction, and analogous components in the y and z directions, and hence to a body-force vector $\mu \nabla^2 V d\tau$.

(b) Deduce that effects of viscosity then may be taken into account by adding the term $\mu \nabla^2 V$ to the right-hand members of (179) and (182).

106. If a body force \mathbf{F} per unit mass is conservative, and hence can be written in the form

$$\mathbf{F} = -\nabla U,$$

where U is a potential energy function, use the result of Problem 104 to show that Equation (205) is replaced by the equation

$$\frac{1}{2}V^2+\frac{\partial\varphi}{\partial t}+P+U=f(t),$$

when F is present, but viscosity effects are neglected.

107. If the Mach number M is greater than unity, show that any expression of the form

$$\varphi(x,y) = f(x + \alpha t) + g(x - \alpha t) \qquad (\alpha = \sqrt{M^2 - 1})$$

satisfies Equation (194), where f and g are any twice-differentiable functions.

108. Verify that the expression

$$\varphi(x,t) = F[x - (U + V_{S0})t] + G[x - (U - V_{S0})t]$$

satisfies Equation (196), where F and G are any twice-differentiable functions.

Topics in Higher-Dimensional Calculus

7.1. Partial differentiation. Chain rules. In this section we review and discuss certain notations and relations involving partial derivatives which will be needed in the sequel.

The more general case may be illustrated here by considering a function f of three variables x, y, and z,

$$f = f(x, y, z). \tag{1}$$

If y and z are held constant and only x is allowed to vary, the partial derivative with respect to x is denoted by $\frac{\partial f}{\partial x}$ and is defined as the limit

$$\frac{\partial f(x,y,z)}{\partial x} = \lim_{\Delta x \to 0} \frac{f(x + \Delta x, y, z) - f(x,y,z)}{\Delta x}.$$
 (2)

Similarly we define the functions $\frac{\partial f}{\partial y}$ and $\frac{\partial f}{\partial z}$. In all cases two of the three variables explicitly appearing in the definition of f are held constant, and f is differentiated with respect to the third variable. The total differential of f is defined by the equation

$$df = \frac{\partial f}{\partial x} dx + \frac{\partial f}{\partial y} dy + \frac{\partial f}{\partial z} dz, \qquad (3)$$

whether or not x, y, and z are independent of each other, provided only that the partial derivatives involved are continuous. Several types of dependence among x, y, and z are now considered. In each of the formulas to be obtained, the continuity of all derivatives appearing in the right-hand member is to be assumed. (1) If x, y, and z are all functions of a single variable, say t, then the dependent variable f may also be considered as truly a function of the one independent variable t, and we may conveniently speak of x, y, and z as intermediate variables. Since only one independent variable is present, $\frac{df}{dt}$ has a meaning and it can be shown, by appropriate limiting processes, that

$$\frac{df}{dt} = \frac{\partial f}{\partial x}\frac{dx}{dt} + \frac{\partial f}{\partial y}\frac{dy}{dt} + \frac{\partial f}{\partial z}\frac{dz}{dt}.$$
(4)

This result is formally obtained by dividing the expression for df by dt. We notice that $\frac{df}{dt}$ is the sum of three terms, each of which represents the contribution of the change in t through the corresponding change in one of the *intermediate* variables.

(2) More generally, the intermediate variables x, y, and z may be functions of two (or more) independent variables, say s and t. Then if we consider f as a function of s and t, we may investigate the *partial* derivative of f with respect

to t when s is held constant. If we denote this function by the notation $\left(\frac{df}{dt}\right)_{s}$, then Equation (4) must be modified to read

$$\left(\frac{\partial f}{\partial t}\right)_{s} = \frac{\partial f}{\partial x} \left(\frac{\partial x}{\partial t}\right)_{s} + \frac{\partial f}{\partial y} \left(\frac{\partial y}{\partial t}\right)_{s} + \frac{\partial f}{\partial z} \left(\frac{\partial z}{\partial t}\right)_{s}.$$
 (5a)

The derivatives with respect to t are now not total but partial. The subscript indicates the variable held constant. With this convention we should perhaps also write $\left(\frac{\partial f}{\partial x}\right)_{y,z}$ in place of $\frac{\partial f}{\partial x}$, and so on, in (4) and (5a). However, we will follow the convention that $\frac{\partial f}{\partial x}$, without subscripts, indicates the result of differentiating f with respect to the explicitly appearing variable x, holding all other explicitly appearing variables (here y and z) constant. Frequently we also omit the subscript s in (5a) and write merely

$$\frac{\partial f}{\partial t} = \frac{\partial f}{\partial x}\frac{\partial x}{\partial t} + \frac{\partial f}{\partial y}\frac{\partial y}{\partial t} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial t},$$
(5b)

if it is clear from the context that s and t are to be associated with each other as the independent variables, with x, y, and z as the (explicit) intermediate variables. Alternatively, we could write F(s,t) for the result of replacing x, y, and z by their equivalents in f(x,y,z), so that

$$f[x(s,t), y(s,t), z(s,t)] = F(s,t),$$

in accordance with which (5a,b) could be written in the form

$$\frac{\partial F}{\partial t} = \frac{\partial f}{\partial x}\frac{\partial x}{\partial t} + \frac{\partial f}{\partial y}\frac{\partial y}{\partial t} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial t},$$
(5c)

without any possible ambiguity. Whereas this is the most elegant way of proceeding in such situations, it is often inconvenient in practice to use two different symbols (here f and F) to represent the same physical or geometrical quantity.

In the preceding cases the independent variable t does not appear explicitly in f, and its changes are reflected in f only through the intermediate changes in x, y, and z. However, it may be convenient to take an explicitly appearing variable as an independent variable. We again distinguish two cases.

(3) If we suppose that y and z are functions of x, then f is a function of the one independent variable x, and y and z are intermediate. Also, identifying t with x in Equation (4), we obtain

$$\frac{df}{dx} = \frac{\partial f}{\partial x} + \frac{\partial f}{\partial y}\frac{dy}{dx} + \frac{\partial f}{\partial z}\frac{dz}{dx}.$$
(6)

The term $\frac{\partial f}{\partial x}$ in (6) is obtained, as before, by holding the other two explicit variables (y and z) constant, and it represents the contribution of the explicit variation of x. The other terms add the contributions of the intermediate variations in y and z.

(4) If we suppose that x and y are independent but that z is a function of both x and y, then f can be considered as depending upon x and y directly and also *intermediately* through z. Also, identifying t and s with x and y in (5a), we obtain

$$\left(\frac{\partial f}{\partial x}\right)_{\nu} = \frac{\partial f}{\partial x} + \frac{\partial f}{\partial z} \left(\frac{\partial z}{\partial x}\right)_{\nu}.$$
 (7a)

Here the notation is rather treacherous. On the left-hand side of (7a) we think of f as being actually expressed in terms of x and y, the variable z having been replaced by its equivalent in terms of these variables. Then we imagine that yis held constant in the x differentiation. On the right-hand side f is expressed in its original form, in terms of x, y, and z. The first term on the right *is again* calculated with the other explicit variables y and z held constant, and it represents the contribution due to the explicit variation of x. The other term adds the contribution of the only intermediate variable z. Since z depends only on xand y, the last subscript may be omitted without confusion. However, the subscript on the left is clearly essential. Thus we may write Equation (7a) in the form

$$\left(\frac{\partial f}{\partial x}\right)_{\nu} = \frac{\partial f}{\partial x} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial x}.$$
 (7b)

These formulas are useful when we deal with a function f in abstract terms. If f is given as a concrete function and the dependencies are concretely stated, such formulas usually are not needed. In illustration we consider the function

$$f(x,y,z) = x^2 + xz + 2y^2.$$

If we consider x, y, and z as functions of t, and possibly other independent variables, we merely differentiate term by term and obtain

$$\frac{\partial f}{\partial t} = 2x \frac{\partial x}{\partial t} + x \frac{\partial z}{\partial t} + z \frac{\partial x}{\partial t} + 4y \frac{\partial y}{\partial t}$$
$$= (2x + z) \frac{\partial x}{\partial t} + 4y \frac{\partial y}{\partial t} + x \frac{\partial z}{\partial t}.$$

This result is the same as that given by Equation (5b). If we consider x as independent and assume that y and z are given by other equations as functions of x, we again differentiate term by term and obtain

$$\frac{df}{dx} = 2x + x\frac{dz}{dx} + z + 4y\frac{dy}{dx}$$
$$= 2x + z + 4y\frac{dy}{dx} + x\frac{dz}{dx},$$

in accordance with the result of using Equation (6). If z is given by another equation in terms of x and y, and if x and y are independent, then we obtain directly, holding y constant,

$$\left(\frac{\partial f}{\partial x}\right)_{\nu} = 2x + z + x \frac{\partial z}{\partial x},$$

in accordance with the result of using (7b). The term 2x + z is equivalent to $\frac{\partial f}{\partial x}$, where y and z are held constant. The term $x \frac{\partial z}{\partial x}$ corrects for the fact that here z cannot *actually* be held constant but must vary also with x.

As a further example, suppose that we have the relation

$$x^2 + xz + 2y^2 = 0.$$

We again denote the function of x, y, and z on the left by f,

$$f = x^2 + xz + 2y^2.$$

We may consider the given relation as determining, say, z in terms of x and y, both of which may then be taken as independent. Then, holding y constant and differentiating with respect to x, we obtain

$$2x+z+x\frac{\partial z}{\partial x}=0.$$

Hence we must have

$$\frac{\partial z}{\partial x} = -\frac{2x+z}{x}$$

We may also arrive at this result by considering the left-hand member of the given equation as a function of the independent variables x and y and the intermediate variable z. Then, since f constantly satisfies the equation

$$f(x,y,z) = 0, \tag{8}$$

the partial derivative of f with respect to either *independent* variable must vanish. But Equation (7b) then gives

$$\left(\frac{\partial f}{\partial x}\right)_{\mu} = \frac{\partial f}{\partial x} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial x} = 0.$$

This equation states that the contributions of the explicit variation of x and the intermediate variation of z must cancel. In order that this be so, we must then have

20

$$\frac{\partial z}{\partial x} = -\frac{\frac{\partial f}{\partial x}}{\frac{\partial f}{\partial z}},$$
(9)

in accordance with the result obtained above.*

Partial derivatives of higher order, of a function f(x,y,z), are calculated by successive differentiation. Thus we write, for example,

$$\frac{\partial^2 f}{\partial y \,\partial x} = \frac{\partial}{\partial y} \left(\frac{\partial f}{\partial x} \right),$$

and so forth. In this connection, we review the important fact that the crossed partial derivatives are equal,

$$\frac{\partial^2 f}{\partial v \,\partial x} = \frac{\partial^2 f}{\partial x \,\partial y},\tag{10}$$

that is, the order of differentiation is immaterial, if the derivatives involved are continuous. This statement is true for derivatives of any order if they are continuous, but it may not be true otherwise.

For the purpose of obtaining analytical formulas for higher-order partial derivatives, it is often convenient to use *operational* notation (see also Problems 3, 4, and 5). As an example, we suppose that f is a function of x, y, and z, with x and y independent and z a function of x and y, so that Equation (7) applies.

* Equation (9) tends to illustrate the dangers associated with routine symbolic manipulations, since a formal (but unjustified) inversion and cancellation in the right-hand member might suggest that the prefixed sign is incorrect. Here, in order to avoid the complexities involved in unambiguously generalizing the notation $\left(\frac{\partial f}{\partial x}\right)_{\nu}$ to higher-order derivatives, it is particularly desirable to introduce the special notation

$$f[x, y, z(x,y)] = F(x,y),$$
 (11)

so that F(x,y) is the result of replacing z by its equivalent in terms of x and y in the expression for f(x,y,z). The formulas (7a,b) then can be rewritten in the form

$$\frac{\partial F}{\partial x} = \frac{\partial f}{\partial x} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial x} = \left(\frac{\partial}{\partial x} + \frac{\partial z}{\partial x}\frac{\partial}{\partial z}\right)f,$$
(12)

from which there follows, by iteration,

$$\frac{\partial^2 F}{\partial x^2} = \left(\frac{\partial}{\partial x} + \frac{\partial z}{\partial x}\frac{\partial}{\partial z}\right) \left(\frac{\partial f}{\partial x} + \frac{\partial z}{\partial x}\frac{\partial f}{\partial z}\right)$$

$$= \left(\frac{\partial^2 f}{\partial x^2} + \frac{\partial^2 z}{\partial x^2}\frac{\partial f}{\partial z} + \frac{\partial z}{\partial x}\frac{\partial^2 f}{\partial x\partial z}\right) + \frac{\partial z}{\partial x}\left(\frac{\partial^2 f}{\partial x\partial z} + \frac{\partial z}{\partial x}\frac{\partial^2 f}{\partial z^2}\right)$$

$$= \frac{\partial^2 f}{\partial x^2} + 2\frac{\partial z}{\partial x}\frac{\partial^2 f}{\partial x\partial z} + \left(\frac{\partial z}{\partial x}\right)^2\frac{\partial^2 f}{\partial z^2} + \frac{\partial^2 z}{\partial x^2}\frac{\partial f}{\partial z}.$$
(13)

Here, as before, the convention is that, in each of the indicated partial derivatives, all variables explicitly involved in the function being differentiated are held constant except the one with respect to which the differentiation is being effected. Thus, since F = F(x,y) and z = z(x,y), there follows

$$\frac{\partial F}{\partial x} \equiv \left(\frac{\partial F}{\partial x}\right)_{y}$$
 and $\frac{\partial z}{\partial x} \equiv \left(\frac{\partial z}{\partial x}\right)_{y}$.

On the other hand, since f = f(x,y,z), there follows

$$\frac{\partial f}{\partial x} \equiv \left(\frac{\partial f}{\partial x}\right)_{y,z}$$

7.2. Implicit functions. Jacobian determinants. An equation of the form

$$f(x,y,z,\ldots)=0, \qquad (14)$$

involving any finite number of variables, where f possesses continuous first partial derivatives, can be considered as determining *one* of the variables, say z, as a function of the remaining variables, say

$$z = \varphi(x, y, \ldots), \tag{15}$$

in some region about any point where Equation (14) is satisfied and where the partial derivative of f with respect to that variable is not zero,

$$\frac{\partial f}{\partial z} \neq 0.$$
 (16)

In such a case we say that Equation (14) defines z as an *implicit* function of the other variables, in the neighborhood of that point. If we consider all the other variables as independent, we can determine the partial derivative of z with respect to any one of them, without solving *explicitly* for z, by differentiating

(14) partially with respect to that variable. Thus, to determine $\frac{\partial z}{\partial x}$, we obtain from (14)

$$\frac{\partial f}{\partial x} + \frac{\partial f}{\partial z}\frac{\partial z}{\partial x} = 0, \qquad \frac{\partial z}{\partial x} = -\frac{\frac{\partial f}{\partial x}}{\frac{\partial f}{\partial z}}, \qquad (17)$$

the denominator differing from zero by virtue of Equation (16).

If n + k variables are related by *n* equations, it is *usually* possible to consider *n* of the variables as functions of the remaining *k* variables. However, this is not *always* possible.

As an illustration, suppose that x, y, u, and v are related by two equations of the form

$$\begin{cases} f(x,y,u,v) = 0\\ g(x,y,u,v) = 0 \end{cases}.$$
(18)

If these equations determine u and v as differentiable functions of the variables x and y, we may differentiate the system with respect to x and y, considering these two variables to be independent, and so obtain the four relations

$$\frac{\partial f}{\partial x} + \frac{\partial f}{\partial u} \frac{\partial u}{\partial x} + \frac{\partial f}{\partial v} \frac{\partial v}{\partial x} = 0$$

$$\frac{\partial g}{\partial x} + \frac{\partial g}{\partial u} \frac{\partial u}{\partial x} + \frac{\partial g}{\partial v} \frac{\partial v}{\partial x} = 0$$

$$\frac{\partial f}{\partial y} + \frac{\partial f}{\partial u} \frac{\partial u}{\partial y} + \frac{\partial f}{\partial v} \frac{\partial v}{\partial y} = 0$$

$$\frac{\partial g}{\partial y} + \frac{\partial g}{\partial u} \frac{\partial u}{\partial y} + \frac{\partial g}{\partial v} \frac{\partial v}{\partial y} = 0$$
(19)
(19)
(20)

For brevity, we use the conventional subscript notation for partial derivatives so that, for example, u_x is written for $\frac{\partial u}{\partial x}$. Then if (19) is solved for $\frac{\partial u}{\partial x}$ and $\frac{dv}{\partial x}$ and (20) is solved for $\frac{\partial u}{\partial y}$ and $\frac{\partial v}{\partial y}$, the expressions for these partial derivatives can be written in terms of determinants as follows:

$$u_{x} = -\frac{\begin{vmatrix} f_{x} & f_{v} \\ g_{x} & g_{v} \\ f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}{\begin{vmatrix} f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}, \qquad v_{x} = -\frac{\begin{vmatrix} f_{u} & f_{x} \\ g_{u} & g_{x} \\ f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}{\begin{vmatrix} f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}, \qquad (21)$$

$$u_{y} = -\frac{\begin{vmatrix} f_{y} & f_{v} \\ g_{y} & g_{v} \\ f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}{\begin{vmatrix} f_{u} & f_{v} \\ g_{u} & g_{v} \end{vmatrix}}.$$

It must be assumed, however, that the common denominator in (21) does not vanish, that is, that

$$\frac{\partial f}{\partial u} \quad \frac{\partial f}{\partial v} \\
\frac{\partial g}{\partial u} \quad \frac{\partial g}{\partial v} \neq 0.$$
(22)

Unless (22) is satisfied, the desired partial derivatives cannot exist uniquely, so that u and v cannot be differentiable functions of x and y. However, if (18) and (22) are satisfied at a point, and if the first partial derivatives of f and gare continuous at and near that point, it can be shown that Equations (18) determine u and v as implicit functions of x and y in *some* region including that point, with partial derivatives given by (21).

The determinant in (22) is known as the Jacobian of f and g with respect to u and v and the notation

$$\frac{\partial(f,g)}{\partial(u,v)} = \begin{vmatrix} \frac{\partial f}{\partial u} & \frac{\partial f}{\partial v} \\ \frac{\partial g}{\partial u} & \frac{\partial g}{\partial v} \end{vmatrix} = \frac{\partial f}{\partial u} \frac{\partial g}{\partial v} - \frac{\partial f}{\partial v} \frac{\partial g}{\partial u}$$
(23)

is frequently used. In a similar way we write, for example,

$$\frac{\partial(f,g,h)}{\partial(u,v,w)} = \begin{vmatrix} \frac{\partial f}{\partial u} & \frac{\partial f}{\partial v} & \frac{\partial f}{\partial w} \\ \frac{\partial g}{\partial u} & \frac{\partial g}{\partial v} & \frac{\partial g}{\partial w} \\ \frac{\partial h}{\partial u} & \frac{\partial h}{\partial v} & \frac{\partial h}{\partial w} \end{vmatrix}$$
(24)

and proceed in the same way to define the Jacobian of any n functions with respect to n variables.

In this notation, if

$$\frac{\partial(f,g)}{\partial(u,v)} \neq 0, \tag{25}$$

the first equation of (21) becomes, for example,

$$\frac{\partial u}{\partial x} = -\frac{\frac{\partial (f,g)}{\partial (x,v)}}{\frac{\partial (f,g)}{\partial (u,v)}}.$$
(26)

More generally, if n + k variables are related by n equations of the form $f_1 = 0, f_2 = 0, \ldots, f_n = 0$, where the functions f_k each have continuous first partial derivatives, then any set of n variables may be considered as functions of the remaining k variables, in *some* neighborhood of a point where the n equations are satisfied, if the Jacobian of the f's with respect to the n dependent variables is not zero at that point.

In illustration, we consider the system

$$x + y + z = 0
 x2 + y2 + z2 + 2xz - 1 = 0
 .
 (27)$$

To investigate whether x and y can be considered as functions of z, we denote the left-hand members by f and g, respectively, and calculate the Jacobian

$$\frac{\partial(f,g)}{\partial(x,y)} = \begin{vmatrix} 1 & 1 \\ 2x + 2z & 2y \end{vmatrix} = -2(x + z - y). \tag{28}$$

Thus, except on the surface x + z - y = 0, x and y can be considered as functions of z. That is, z can be taken as the independent variable. When y = x + z, the equations become 2(x + z) = 0 and $2(x + z)^2 = 1$ and are hence incompatible. To investigate whether x and z can be taken as the dependent variables we calculate the Jacobian

$$\frac{\partial(f,g)}{\partial(x,z)} = \begin{vmatrix} 1 & 1 \\ 2x + 2z & 2x + 2z \end{vmatrix} = 0.$$
 (29)

Since this determinant is identically zero, we see that x and z cannot be taken as the dependent variables. It is readily verified *directly* that the system (27) cannot be solved for x and z in terms of y. This situation follows from the fact that both equations involve only y and the *combination* x + z, and hence cannot be solved for x and z separately.

By direct expansion we can verify that if u and v are functions of r and s, and also r and s are functions of x and y, then the relevant Jacobians satisfy the equation

$$\frac{\partial(u,v)}{\partial(r,s)}\frac{\partial(r,s)}{\partial(x,y)} = \frac{\partial(u,v)}{\partial(x,y)}.$$
(30)

As a special case of this result, we find that if u and v are functions of x and y, and conversely, then

$$\frac{\partial(u,v)}{\partial(x,y)}\frac{\partial(x,y)}{\partial(u,v)} = 1.$$
(31)

Analogous identities hold for Jacobians of any order. Thus Jacobians behave in certain ways like derivatives, as is suggested by the notation used.

The Jacobian notation is useful in many other applications. In this connection, the identity

$$(\nabla u) \cdot (\nabla v) \times (\nabla w) = \begin{vmatrix} \frac{\partial u}{\partial x} & \frac{\partial u}{\partial y} & \frac{\partial u}{\partial z} \\ \frac{\partial v}{\partial x} & \frac{\partial v}{\partial y} & \frac{\partial v}{\partial z} \\ \frac{\partial w}{\partial x} & \frac{\partial w}{\partial y} & \frac{\partial w}{\partial z} \end{vmatrix} = \frac{\partial(u, v, w)}{\partial(x, y, z)}$$
(32)

may be noted. From this relation we conclude that if $\frac{\partial(u,v,w)}{\partial(x,y,z)} = 0$ at a point, then the surfaces $u = c_1$, $v = c_2$, $w = c_3$ which pass through this point have coplanar normals.

7.3. Functional dependence. The general solutions of certain types of partial differential equations, to be dealt with in Chapter 8, are of the form

$$z = f[u_1(x,y)] + g[u_2(x,y)],$$
(33)

where u_1 and u_2 are *independent* particular solutions and f and g are arbitrary functions of these expressions. If u_2 were a function of u_1 , both terms would then be functions merely of u_1 , and (33) would not be the required general solution. Thus, in the expression

$$z = f(x + y) + g(x^2 + 2xy + y^2 + 1) \equiv f(u_1) + g(u_2)$$

we have $u_2 = u_1^2 + 1$ and hence both terms are functions of the same combination $u_1 = x + y$. It is thus important in more involved cases to have a criterion for determining whether one function $u_1(x,y)$ is a function of a second function $u_2(x,y)$. If such a functional relationship does exist the two functions are said to be *functionally dependent*.

Suppose that such a relationship *does* exist, so that, for some F, not *identi*cally zero, it is true that

$$F(u_1, u_2) \equiv 0, \tag{34}$$

where u_1 and u_2 are functions of the independent variables x and y. Then, if we calculate the partial derivatives of (34), we obtain

$$\frac{\partial F}{\partial u_1} \frac{\partial u_1}{\partial x} + \frac{\partial F}{\partial u_2} \frac{\partial u_2}{\partial x} = 0$$

$$\frac{\partial F}{\partial u_1} \frac{\partial u_1}{\partial y} + \frac{\partial F}{\partial u_2} \frac{\partial u_2}{\partial y} = 0$$
(35)

These two linear equations in the quantities $\frac{\partial F}{\partial u_1}$ and $\frac{\partial F}{\partial u_2}$ can have nontrivial solutions only if the determinant of their coefficients vanishes. But this determinant is precisely the Jacobian of u_1 and u_2 with respect to x and y. Hence

if $u_1(x,y)$ and $u_2(x,y)$ are functionally dependent in a region, their Jacobian must vanish identically,

$$\frac{\partial(u_1,u_2)}{\partial(x,y)} \equiv 0. \tag{36}$$

Conversely, it can be shown that if the partial derivatives are continuous and if the Jacobian vanishes identically in a region, the two functions are functionally dependent in that region.

Completely analogous statements apply in the more general case of n functions of n variables. Thus if the functions u_1, u_2, \ldots, u_n , are functions of n variables, and if their partial derivatives are continuous, then the functions are functionally dependent, that is, there exists a nontrivial F such that

$$F(u_1,u_2,\ldots,u_n)\equiv 0$$

in a region, if and only if the Jacobian of these functions with respect to the n variables is identically zero in that region.

As a simple illustration we consider the relations

$$u_1 = ax + by + c \\ u_2 = dx + ey + f$$

The Jacobian is

$$\frac{\partial(u_1,u_2)}{\partial(x,y)} = \begin{vmatrix} a & b \\ d & e \end{vmatrix} = ae - bd.$$

Thus u_1 and u_2 are functionally independent unless ae = bd. If ae = bd, the functional relationship

$$eu_1 - bu_2 = ec - bf$$

exists between u_1 and u_2 .

When there are fewer functions than variables, *several* relations of form (36) must hold. For example, in the case of two functions of three variables,

 $u_1(x,y,z)$ and $u_2(x,y,z)$, the assumption

$$F(u_1, u_2) \equiv 0 \tag{37}$$

leads to the three equations

$$\frac{\partial F}{\partial u_1} \frac{\partial u_1}{\partial x} + \frac{\partial F}{\partial u_2} \frac{\partial u_2}{\partial x} = 0,$$
$$\frac{\partial F}{\partial u_1} \frac{\partial u_1}{\partial y} + \frac{\partial F}{\partial u_2} \frac{\partial u_2}{\partial y} = 0,$$
$$\frac{\partial F}{\partial u_1} \frac{\partial u_1}{\partial z} + \frac{\partial F}{\partial u_2} \frac{\partial u_2}{\partial z} = 0,$$

from which, by considering the equations in pairs, we may deduce that the *three* conditions

$$\frac{\partial(u_1, u_2)}{\partial(x, y)} \equiv 0, \qquad \frac{\partial(u_1, u_2)}{\partial(y, z)} \equiv 0, \qquad \frac{\partial(u_1, u_2)}{\partial(z, x)} \equiv 0$$
(38)

must be satisfied, by the same argument used in deriving (36). Conversely, if the partial derivatives are continuous and if the three conditions of (38) are satisfied identically in a region, then the functions $u_1(x,y,z)$ and $u_2(x,y,z)$ are functionally dependent in that region.

The generalization to *m* functions of *n* variables, when m < n, is straightforward. When m > n, the *m* functions are *always* functionally dependent.

7.4. Jacobians and curvilinear coordinates. Change of variables in integrals. If the equations

$$x = x(u_1, u_2, u_3), \quad y = y(u_1, u_2, u_3), \quad z = z(u_1, u_2, u_3)$$
 (39)

are interpreted as defining curvilinear coordinates u_1 , u_2 , and u_3 in space and if we write

$$\mathbf{U}_{k} = \mathbf{i} \frac{\partial x}{\partial u_{k}} + \mathbf{j} \frac{\partial y}{\partial u_{k}} + \mathbf{k} \frac{\partial z}{\partial u_{k}} \qquad (k = 1, 2, 3), \tag{40}$$

then, as has been shown in Section 6.17, the vectors U_1 , U_2 , and U_3 are vectors tangent to the three coordinate curves at any point, with lengths given by $\frac{ds_1}{du_1}$, $\frac{ds_2}{du_2}$, and $\frac{ds_3}{du_3}$, where s_1 , s_2 , and s_3 represent arc length along the coordinate curves. Then [compare Equation (148), Section 6.17] the element of volume in the new coordinate system, whether or not the system is orthogonal, is seen to be given by

$$d\tau = (\mathbf{U}_1 \cdot \mathbf{U}_2 \times \mathbf{U}_3) \, du_1 \, du_2 \, du_3,$$

if the coordinates are so ordered that the right-hand member is positive. But from (40) we obtain

$$\mathbf{U}_{1} \cdot \mathbf{U}_{2} \times \mathbf{U}_{3} = \begin{vmatrix} \frac{\partial x}{\partial u_{1}} & \frac{\partial y}{\partial u_{1}} & \frac{\partial z}{\partial u_{1}} \\ \frac{\partial x}{\partial u_{2}} & \frac{\partial y}{\partial u_{2}} & \frac{\partial z}{\partial u_{2}} \\ \frac{\partial x}{\partial u_{3}} & \frac{\partial y}{\partial u_{3}} & \frac{\partial z}{\partial u_{3}} \end{vmatrix} = \frac{\partial(x, y, z)}{\partial(u_{1}, u_{2}, u_{3})},$$

since the determinant is unchanged if rows and columns are interchanged. Thus we may write

$$d\tau = \left| \frac{\partial(x, y, z)}{\partial(u_1, u_2, u_3)} \right| du_1 \, du_2 \, du_3. \tag{41}$$

It is seen that the requirement that $U_1 \cdot U_2 \times U_3$ be different from zero is necessary in order that (39) be solvable for u_1 , u_2 , and u_3 . In the special case of orthogonal coordinates, $U_1 \cdot U_2 \times U_3$ has the value $h_1h_2h_3$, with the notation of Section 6.17.

Accordingly, we have the change-of-variables formula

$$\iiint_{\mathscr{R}} f(x,y,z) \, dx \, dy \, dz$$
$$= \iiint_{\mathscr{R}^*} F(u_1,u_2,u_3) \left| \frac{\partial(x,y,z)}{\partial(u_1,u_2,u_3)} \right| \, du_1 \, du_2 \, du_3, \qquad (42)$$

where

$$F(u_1, u_2, u_3) = f[x(u_1, u_2, u_3), y(u_1, u_2, u_3), z(u_1, u_2, u_3)]$$

and where \mathscr{R}^* is the $u_1 u_2 u_3$ region into which (39) transforms the xyz region \mathscr{R} . Here it is assumed that the Jacobian $\frac{\partial(x,y,z)}{\partial(u_1,u_2,u_3)}$ is continuous and nonzero in \mathscr{R}^* .

In a similar way, the equations

$$x = x(u_1, u_2), \quad y = y(u_1, u_2)$$
 (43)

can be interpreted as defining curvilinear coordinates u_1 and u_2 in the xy plane. The vectors

$$\mathbf{U}_1 = \mathbf{i} \frac{\partial x}{\partial u_1} + \mathbf{j} \frac{\partial y}{\partial u_1}, \qquad \mathbf{U}_2 = \mathbf{i} \frac{\partial x}{\partial u_2} + \mathbf{j} \frac{\partial y}{\partial u_2}$$
(44)

are then tangent to the coordinate curves, with lengths $\frac{ds_1}{du_1}$ and $\frac{ds_2}{du_2}$. The vector element of plane area is then given by

$$d\mathbf{A} = (\mathbf{U}_1 \times \mathbf{U}_2) \, du_1 \, du_2 = \begin{vmatrix} \mathbf{i} & \mathbf{j} & \mathbf{k} \\ \frac{\partial x}{\partial u_1} & \frac{\partial y}{\partial u_1} & \mathbf{0} \\ \frac{\partial x}{\partial u_2} & \frac{\partial y}{\partial u_2} & \mathbf{0} \end{vmatrix} \, du_1 \, du_2$$

and this relation gives the result

$$dA = |d\mathbf{A}| = \left| \frac{\partial(x, y)}{\partial(u_1, u_2)} \right| du_1 du_2.$$
(45)

Accordingly, we have

$$\iint_{\mathcal{A}} f(x,y) \, dx \, dy = \iint_{\mathcal{A}^*} F(u_1,u_2) \left| \frac{\partial(x,y)}{\partial(u_1,u_2)} \right| \, du_1 \, du_2, \tag{46}$$

where $F(u_1, u_2) = f[x(u_1, u_2), y(u_1, u_2)]$ and where (43) transforms A into A*, if $\frac{\partial(x, y)}{\partial(u_1, u_2)}$ is continuous and nonzero in A*.

As an example of the use of Equation (45), we consider the coordinates u and φ defined by the equations

$$x = au \cos \varphi, \quad y = bu \sin \varphi \qquad (u \ge 0, 0 \le \varphi < 2\pi).$$
 (47)

The curves u = constant are the ellipses $\frac{x^2}{(au)^2} + \frac{y^2}{(bu)^2} = 1$ with semiaxes au and bu, whereas a curve $\varphi = \text{constant}$ is the portion of the straight line $y = \frac{b}{a}x \tan \varphi$ in the quadrant determined by φ . The element of area in $u\varphi$ coordinates is given by (45),

$$dA = \left| \frac{\partial(x,y)}{\partial(u,\varphi)} \right| du \, d\varphi = abu \, du \, d\varphi. \tag{48}$$

Thus, for example, any integral of the form $\iint_A f(x,y) dx dy$, where the integration is carried out over the area of the ellipse

$$\frac{x^2}{a^2} + \frac{y^2}{b^2} = 1$$

corresponding to u = 1, can be written in the form

$$\int \int_{A} f(x,y) \, dx \, dy = ab \int_{0}^{1} \int_{0}^{2\pi} f(au \cos \varphi, bu \sin \varphi) \, u \, du \, d\varphi. \tag{49}$$

In particular, to calculate the moment of inertia I_x of the area about the x axis we write

$$f(x,y)=y^2=b^2u^2\sin^2\varphi,$$

and Equation (49) gives

$$I_x = ab^3 \int_0^1 \int_0^{2\pi} u^3 \sin^2 \varphi \, du \, d\varphi = \frac{\pi a b^3}{4} \, .$$

7.5. Taylor series. Functions of two or more variables often can be expanded in power series which generalize the familiar one-dimensional expansions. The more general situation may be illustrated here by a consideration of the two-variable case.

For this purpose, we begin by defining a function F(t), such that

$$f(x + ht, y + kt) = F(t),$$
 (50)

where x, y, h, and k are temporarily to be held fixed and, in any case, are to be independent of t. Then, if F(t) has a continuous Nth derivative in some interval about t = 0, we may write

$$F(t) = \sum_{n=0}^{N-1} \frac{F^{(n)}(0)}{n!} t^n + \frac{F^{(N)}(\tau)}{N!} t^N,$$
 (51)

for some value of τ between 0 and t, by virtue of Equations (10) and (11) of Chapter 4.

Now, since

$$\frac{d}{dt}F(t) = h \frac{\partial f(x+ht, y+kt)}{\partial x} + k \frac{\partial f(x+ht, y+kt)}{\partial y}$$
$$= \left(h \frac{\partial}{\partial x} + k \frac{\partial}{\partial y}\right) f(x+h, y+kt),$$

there follows also

$$\frac{d^n}{dt^n}F(t)=\left(h\frac{\partial}{\partial x}+k\frac{\partial}{\partial y}\right)^nf(x+ht,y+kt) \qquad (n=0,1,\ldots).$$

Hence we have the results

$$F^{(n)}(0) = \left(h\frac{\partial}{\partial x} + k\frac{\partial}{\partial y}\right)^n f(x,y)$$
(52)

$$F^{(N)}(\tau) = \left(h\frac{\partial}{\partial x} + k\frac{\partial}{\partial y}\right)^N f(x + \tau h, y + \tau k).$$
 (53)

If we introduce Equations (51), (52), and (53) into (50), and specialize the result by taking t = 1, we thus obtain the form

$$f(x+h, y+k) = \sum_{n=0}^{N-1} \left(h \frac{\partial}{\partial x} + k \frac{\partial}{\partial y} \right)^n f(x,y) + R_N$$
(54)

where R_N , the "remainder after N terms," is given by

$$R_N = \frac{1}{N!} \left(h \frac{\partial}{\partial x} + k \frac{\partial}{\partial y} \right)^N f(x + \tau h, y + \tau k) \qquad (0 < \tau < 1), \quad (55)$$

for some τ between 0 and 1.

For example, when N = 3 this result becomes

$$f(x + h, y + k) = f(x,y) + [h f_x(x,y) + k f_y(x,y)] + \frac{1}{2!} [h^2 f_{xx}(x,y) + 2hk f_{xy}(x,y) + k^2 f_{yy}(x,y)] + R_3$$
(56)
where

$$R_{3} = \frac{1}{3!} \left[h^{3} f_{xxx} + 3h^{2} k f_{xxy} + 3hk^{2} f_{xyy} + k^{3} f_{yyy} \right]_{(x+rh, y+rk)}.$$
 (57)

It is seen that the contents of the brackets in R_3 are evaluated at *some* point on the straight line segment between the point (x, y) and the point (x + h, y + k).

More generally, Equation (54) represents an expansion of f(x + h, y + k)in powers of h and k through the (N - 1)th, with an error term, and is one form of the two-dimensional *Taylor formula*. Although the form given is perhaps the most compact one, a form which more closely resembles the most familiar one-dimensional form can be obtained from Equation (54) by first replacing (x,y) by (x_0,y_0) and then replacing h and k by $x - x_0$ and $y - y_0$, respectively. When N = 2, for example, there then follows

$$f(x,y) = f(x_0,y_0) + (x - x_0)f_x(x_0,y_0) + (y - y_0)f_y(x_0,y_0) + R_2$$
(58)

with

$$R_{2} = \frac{1}{2!} \left[(x - x_{0})^{2} f_{xx}(\xi, \eta) + 2(x - x_{0})(y - y_{0}) f_{xy}(\xi, \eta) + (y - y_{0})^{2} f_{yy}(\xi, \eta) \right], \quad (59)$$

where the point (ξ,η) is somewhere on the line segment joining the points (x_0,y_0) and (x,y).

When f(x,y) is sufficiently well behaved, the remainder R_N tends to zero for sufficiently small values of the increments, yielding a (generally infinite) power series of the form

$$f(x,y) = f(x_0,y_0) + [(x - x_0)f_x(x_0,y_0) + (y - y_0)f_y(x_0,y_0)] + \frac{1}{2!}[(x - x_0)^2 f_{xx}(x_0,y_0) + 2(x - x_0)(y - y_0)f_{xy}(x_0,y_0) + (y - y_0)^2 f_{yy}(x_0,y_0)] + \cdots, \quad (60)$$

which converges when $|x - x_0|$ and $|y - y_0|$ are sufficiently small. Within its region of convergence, the series can be differentiated or integrated term by term and the result will converge to the derivative or integral of f inside the same region.

It can be proved that the expansion (60) is *unique*, in the sense that if an expansion of the form

$$f(x,y) = a_0 + b_1(x - x_0) + b_2(y - y_0) + c_1(x - x_0)^2 + \cdots,$$

which converges to f near (x_0, y_0) , can be obtained by any method, it is necessarily the same as that defined by (60). For the elementary functions, alternative methods which are preferable to the use of (60) are usually evident. 7.6. Maxima and minima. The developments of the preceding section are helpful in studying maxima and minima of functions of several variables. We again restrict attention here to the two-dimensional case.

If we write

$$\Delta f(x,y) = f(x+h, y+k) - f(x,y)$$
(61)

for the increment of f, corresponding to the increments h and k in x and y, respectively, we say that f has a *relative minimum* at $P(x_0, y_0)$ if $\Delta f(x_0, y_0) \ge 0$ for all sufficiently small permissible increments h and k, and that f has a *relative maximum* at P if instead $\Delta f(x_0, y_0) \le 0$ for all such increments in x and y.

If the point P is an interior point of a region in which f, $\frac{\partial f}{\partial x}$, and $\frac{\partial f}{\partial y}$ exist,

Equation (56) shows that a necessary condition that f assume a relative maximum or a relative minimum at (x_0, y_0) is that

$$f_x = f_y = 0$$
 at (x_0, y_0) . (62)

For, when h and k are sufficiently small, the sign of $\Delta f(x_0, y_0)$ will be the same as the sign of $h f_x(x_0, y_0) + k f_y(x_0, y_0)$ when this quantity is not zero, and clearly the sign of this quantity will change as the signs of h and/or k change unless Equation (62) holds.

Suppose now that the condition of (62) is satisfied at a certain point P. Then, from (56), there follows

$$\operatorname{sign} \left[\Delta f(x_0, y_0)\right] = \operatorname{sign} \left[h^2 f_{xx}(x_0, y_0) + 2hk f_{xy}(x_0, y_0) + k^2 f_{yy}(x_0, y_0)\right] \quad (63)$$

when h and k are sufficiently small, unless the bracketed quantity on the right is zero. That quantity is a quadratic expression in h and k, of the form $Ah^2 + 2Bhk + Ck^2$. When the discriminant $B^2 - AC$ is positive, and only in that case, there will be two distinct values of the ratio k/h for which the expression is zero, the expression having one sign for intermediate values of k/h and the opposite sign for all other values. Hence a necessary condition that f have either a relative maximum or a relative minimum at $P(x_0, y_0)$ is that

$$\delta \equiv f_{xx} f_{yy} - f_{xy}^2 \ge 0 \quad \text{at} (x_0, y_0). \tag{64}$$

If $\delta < 0$ at a point $P(x_0, y_0)$ where Equation (62) is satisfied, then Δf is positive for some increments in x and y and negative for others, and the point P is said to be a saddle point or a minimax.

If $\delta = 0$ at a point $P(x_0, y_0)$, then the bracketed expression on the right in (63) is a perfect square, of the form $(\alpha h - \beta k)^2$, and hence either is identically zero or is zero along a line

$$\frac{k}{h} \equiv \frac{y - y_0}{x - x_0} = \frac{\alpha}{\beta}$$

passing through P. Thus, in this case the sign of $\Delta f(x_0, y_0)$ may not be completely determined near (x_0, y_0) by the quadratic terms when (62) is satisfied, and the terms involving higher-order partial derivatives must be considered.

If $\delta > 0$ at (x_0, y_0) , then clearly f_{xx} and f_{yy} must be either both positive or both negative at that point. Since $\Delta f(x_0, y_0)$ is of constant sign in either case, when h and k are sufficiently small, it follows from (63) that the former case corresponds to a relative minimum ($\Delta \ge 0$) and the latter to a relative maximum ($\Delta \le 0$).

Thus, in summary, if $f_x = 0$ and $f_y = 0$ at a point P, then at that point f has

- (a) a relative maximum if $f_{xx} < 0$, $f_{xx}f_{yy} > f_{xy}^2$ at P,
- (b) a relative minimum if $f_{xx} > 0$, $f_{xx}f_{yy} > f_{xy}^2$ at P,
- (c) a saddle point if $f_{xx}f_{yy} < f_{xy}^2$ at P.

When $f_{xx}f_{yy} = f_{xy}^2$ at P, further investigation is necessary.

Cases (a), (b), and (c) are illustrated by the functions $1 - x^2 - y^2$, $x^2 + y^2$, and xy, respectively, at (0,0). For the functions $1 - x^2y^2$, x^2y^2 , and x^3y^2 the point (0,0) is exceptional, since for each not only $f_x = f_y = 0$ but also $f_{xx} = f_{xy} = f_{yy} = 0$ at that point. However, it is obvious, by inspection, that these functions have a maximum, a minimum, and a saddle point, respectively, at the origin.

Very frequently, in practice, the use of the preceding criteria is too involved to be feasible, and a direct study of the behavior of $\Delta f(x_0, y_0)$ for small h and k may be necessary. Often physical or geometrical considerations make such investigations unnecessary.

It should be noted that a relative maximum or minimum may also be attained at a point where f_x and/or f_y fail to exist and that, when attention is restricted to a region \mathcal{R} with a finite boundary, it may happen that an extreme value is taken on at a *boundary point*, at which f_x and f_y may or may not exist and may or may not differ from zero. In order to locate an *absolute* maximum or minimum (that is, the largest or smallest value taken on) in a region, it is necessary to explore all these possibilities.

7.7. Constraints and Lagrange multipliers. Situations also may occur in which a function f, to be maximized or minimized, depends upon variables which are not independent, but are interrelated by one or more constraint conditions. The more general situation may be illustrated by the problem of maximizing or minimizing a function f(x,y,z),

$$f(x,y,z) =$$
 relative max or min, (65)

subject to two constraints of the form

$$g(x,y,z) = 0,$$
 $h(x,y,z) = 0,$ (66a,b)

where g and h are not functionally dependent, so that the constraints are neither equivalent nor incompatible. We suppose that the functions f, g, and hhave first partial derivatives everywhere in a region which includes the desired point.

An obvious procedure consists of using Equations (66a,b) to eliminate two of the variables from f, leading to a problem of maximizing or minimizing a function of only one variable, without constraints. However, this elimination, by analytical methods, will be feasible only if the functions g and h are of relatively simple form.

Alternatively, we may notice that f can have an extreme value at $P(x_0, y_0, z_0)$ only if the linear terms in the Taylor expansion of Δf about P are zero. This condition can be written in the form

$$f_x dx + f_y dy + f_z dz = 0$$
 at (x_0, y_0, z_0) . (67)

Here, however, the increments dx, dy, and dz are not independent, so that here we *cannot* conclude that f_x , f_y , and f_z must vanish separately at *P*. Further, since g and h are each constant, the differential of each must be zero, so that we must have

$$g_x dx + g_y dy + g_z dz = 0$$
 at (x_0, y_0, z_0) , (68a)

$$h_x dx + h_y dy + h_z dz = 0$$
 at (x_0, y_0, z_0) . (68b)

Now Equations (68a,b) are *linear* in dx, dy, and dz. They can be solved uniquely for two of those differentials in terms of the third when and only when g and h are functionally independent (see Section 7.3). Thus two of the differentials (say dx and dy) can be eliminated from (67), leaving an equation of the form F(x,y,z) dz = 0, in which the one remaining differential *can* be arbitrarily assigned. The satisfaction of the condition F(x,y,z) = 0, together with the conditions (66a,b), at (x_0,y_0,z_0) , constitute three equations in the three unknowns x_0 , y_0 , and z_0 .

A third alternative, of frequent usefulness, is based on the observation that one may multiply the equal members of (68a) and of (68b) by any constants λ_1 and λ_2 , respectively, and add the results to (67) to yield the requirement

$$(f_x + \lambda_1 g_x + \lambda_2 h_x) dx + (f_y + \lambda_1 g_y + \lambda_2 h_y) dy + (f_z + \lambda_1 g_z + \lambda_2 h_z) dz = 0$$

at (x_0, y_0, z_0) for any values of λ_1 and λ_2 . Now it is possible to determine λ_1 and λ_2 so that the coefficients of two of the differentials are zero. For if this were not so it would follow that

$$\begin{vmatrix} g_x & h_x \\ g_y & h_y \end{vmatrix} = \begin{vmatrix} g_y & h_y \\ g_z & h_z \end{vmatrix} = \begin{vmatrix} g_z & h_z \\ g_x & h_x \end{vmatrix} = 0,$$

and hence g and h would be functionally dependent, so that the two constraints would be either equivalent or inconsistent. If we imagine that λ_1 and λ_2 have

been so determined, then the remaining differential can be arbitrarily assigned, so that *its* coefficient *also* must vanish. Hence we obtain the three equations

$$\begin{cases} f_x + \lambda_1 g_x + \lambda_2 h_x = 0 \\ f_y + \lambda_1 g_y + \lambda_2 h_y = 0 \\ f_z + \lambda_1 g_z + \lambda_2 h_z = 0 \end{cases} \quad \text{at } (x_0, y_0, z_0), \tag{69}$$

which, together with the conditions

$$g = 0, \quad h = 0 \qquad \text{at } (x_0, y_0, z_0),$$
 (70)

comprise five equations in the five unknown quantities x_0 , y_0 , z_0 , λ_1 , and λ_2 .

The parameters λ_1 and λ_2 are called Lagrange multipliers. If they were eliminated from (69), the result would be the relation F = 0 obtained by eliminating dx, dy, and dz from (67) and (68a,b). However, the new system of five equations may be preferable because of the additional flexibility which it affords. For example, it may be more convenient to solve (69) for x, y, and z in terms of λ_1 and λ_2 and to introduce the results into (70) for the determination of λ_1 and λ_2 .

The equations (69) are easily remembered if one notices that they are the necessary conditions $\varphi_x = 0$, $\varphi_y = 0$, $\varphi_z = 0$ that the "auxiliary function"

$$\varphi = f + \lambda_1 g + \lambda_2 h \tag{71}$$

attain a relative maximum or minimum at (x_0, y_0, z_0) when no constraints are imposed.

More generally, if f were to be maximized subject to n independent constraints $g_1 = 0, g_2 = 0, \ldots, g_n = 0$, one would form the auxiliary function

$$\varphi = f + \lambda_1 g_1 + \lambda_2 g_2 + \cdots + \lambda_n g_n, \qquad (72)$$

where $\lambda_1, \ldots, \lambda_n$ are unknown constants, and write down the necessary conditions for rendering φ a relative maximum or minimum with no constraints.

As a very simple illustration, we seek the point on the plane Ax + By + Cz = D which is nearest the origin. Thus, we are to minimize $x^2 + y^2 + z^2$, subject to the single constraint Ax + By + Cz - D = 0. With

$$\varphi = (x^2 + y^2 + z^2) + \lambda(Ax + By + Cz - D),$$

the conditions $\varphi_x = \varphi_y = \varphi_z = 0$ at (x_0, y_0, z_0) become

$$2x_0 + \lambda A = 0, \qquad 2y_0 + \lambda B = 0, \qquad 2z_0 + \lambda C = 0,$$

from which there follows

$$x_0 = -\frac{1}{2}\lambda A$$
, $y_0 = -\frac{1}{2}\lambda B$, $z_0 = -\frac{1}{2}\lambda C$

and substitution into the constraint condition determines λ ,

$$-\frac{1}{2}\lambda(A^2 + B^2 + C^2) = D.$$

Thus, finally, the coordinates of the desired point are found to be

$$x_0 = \frac{AD}{A^2 + B^2 + C^2}, \quad y_0 = \frac{BD}{A^2 + B^2 + C^2}, \quad z_0 = \frac{CD}{A^2 + B^2 + C^2}$$

The corresponding minimum distance from the origin is

$$\sqrt{x_0^2 + y_0^2 + z_0^2} = \frac{|D|}{\sqrt{A^2 + B^2 + C^2}}.$$

7.8. Calculus of variations. An important class of problems involves the determination of one or more *functions*, subject to certain conditions, so as to maximize or minimize a certain *definite integral*, whose integrand depends upon the unknown function or functions and/or certain of their derivatives.

For example, to find the equation y = u(x) of the curve along which the distance from (0,0) to (1,1) in the xy plane is least, we would seek u(x) such that

$$I \equiv \int_0^1 \sqrt{1 + {u'}^2} \, dx = \min$$

with

This section presents a brief treatment of some of the simpler aspects of such problems.

 $u(0) = 0, \quad u(1) = 1.$

We consider first the case when we are to attempt to maximize or minimize an integral of the form

$$I = \int_a^b F(x, u, u') \, dx, \tag{73}$$

subject to the conditions

$$u(a) = A, \qquad u(b) = B, \tag{74}$$

where a, b, A, and B are given constants. We suppose that F has continuous second-order derivatives with respect to its three arguments and require that the unknown function u(x) possess two derivatives everywhere in (a,b). To fix ideas, we suppose that I is to be maximized.

We thus visualize a competition, to which only functions which have two derivatives in (a,b) and which take on the prescribed end values are admissible. The problem is that of selecting, from *all* admissible competing functions, the function (or functions) for which I is largest.

Under the assumption that there is indeed a function u(x) having this property, we next consider a one-parameter family of admissible functions which includes u(x), namely, the set of all functions of the form

$$u(x) + \epsilon \eta(x)$$

where $\eta(x)$ is any arbitrarily chosen twice-differentiable function which vanishes at the end points of the interval (a,b),

$$\eta(a) = \eta(b) = 0, \tag{75}$$

and where ϵ is a parameter which is constant for any one function in the set but which varies from one function to another. The increment $\epsilon \eta(x)$, representing the difference between the varied function and the actual solution function, is often called a *variation* of u(x).

If the result of replacing u(x) by $u(x) + \epsilon \eta(x)$ in I is denoted by $I(\epsilon)$,

$$I(\epsilon) = \int_a^b F(x, u + \epsilon \eta, u' + \epsilon \eta') \, dx, \qquad (76)$$

it then follows that $I(\epsilon)$ takes on its maximum value when $\epsilon = 0$, that is, when the variation of u is zero. Hence it must follow that

$$\frac{dI(\epsilon)}{d\epsilon} = 0 \quad \text{when } \epsilon = 0. \tag{77}$$

The assumed continuity of the partial derivatives of F with respect to its three arguments implies the continuity of $\frac{dF}{d\epsilon}$, so that we may differentiate $I(\epsilon)$ under the integral sign (see Section 7.9) to obtain

$$\frac{dI(\epsilon)}{d\epsilon} = \int_a^b \left[\frac{\partial F(x, u + \epsilon \eta, u' + \epsilon \eta')}{\partial (u + \epsilon \eta)} \eta + \frac{\partial F(x, u + \epsilon \eta, u' + \epsilon \eta')}{\partial (u' + \epsilon \eta')} \eta' \right] dx.$$

Hence, by setting $\epsilon = 0$, we obtain an expression for the condition (77) in the form

$$I'(0) = \int_{a}^{b} \left[\frac{\partial F}{\partial u} \eta(x) + \frac{\partial F}{\partial u'} \eta'(x) \right] dx = 0.$$
 (78)

Here we write $F \equiv F(x,u,u')$, noticing that the partial derivatives $\frac{\partial F}{\partial u}$ and $\frac{\partial F}{\partial u'}$ have been formed with x, u, and u' treated as independent variables

have been formed with x, u, and u' treated as independent variables.

The next step consists of transforming the integral of the second product in (78) by an integration by parts, to give

$$\int_{a}^{b} \frac{\partial F}{\partial u'} \eta'(x) \, dx = \left[\frac{\partial F}{\partial u'} \eta(x) \right]_{a}^{b} - \int_{a}^{b} \frac{d}{dx} \left(\frac{\partial F}{\partial u'} \right) \eta(x) \, dx$$
$$= -\int_{a}^{b} \frac{d}{dx} \left(\frac{\partial F}{\partial u'} \right) \eta(x) \, dx,$$

in consequence of (75). Hence Equation (78) becomes

$$\int_{a}^{b} \left[\frac{d}{dx} \left(\frac{\partial F}{\partial u'} \right) - \frac{\partial F}{\partial u} \right] \eta(x) \, dx = 0.$$
 (79)

It is possible to prove rigorously that, since (79) is true for any function $\eta(x)$ which is twice differentiable in (a,b) and zero at the ends of that interval, consequently the coefficient of $\eta(x)$ in the integrand must be zero everywhere in (a,b), so that the condition

$$\frac{d}{dx}\left(\frac{\partial F}{\partial u'}\right) - \frac{\partial F}{\partial u} = 0 \tag{80}$$

must be satisfied. This is the so-called *Euler equation* associated with the problem of maximizing (or minimizing) the integral (73), subject to (74).

If we recall that F, and hence also $\frac{\partial F}{\partial u'}$, may depend upon x both directly and indirectly, through the intermediate variables u(x) and u'(x), we deduce that

$$\frac{d}{dx}\left(\frac{\partial F}{\partial u'}\right) = \frac{\partial}{\partial x}\left(\frac{\partial F}{\partial u'}\right) + \left[\frac{\partial}{\partial u}\left(\frac{\partial F}{\partial u'}\right)\right]\frac{du}{dx} + \left[\frac{\partial}{\partial u'}\left(\frac{\partial F}{\partial u'}\right)\right]\frac{du'}{dx}$$

so that the Euler equation (80) also can be written in the expanded form

$$F_{u'u'}\frac{d^2u}{dx^2} + F_{uu'}\frac{du}{dx} + (F_{xu'} - F_u) = 0.$$
 (81)

Although the form (80) frequently is more convenient in practice, the expanded form shows that, except in the special cases when $F_{u'u'} \equiv \frac{\partial^2 F}{\partial u'^2}$ is zero, the equation is in fact a differential equation of second order in u, subject to the two boundary conditions u(a) = A and u(b) = B.

Since the coefficients in Equation (81) may depend not only on x but also on u and $\frac{du}{dx}$, the equation is not necessarily *linear*. However, it involves the *highest*-order derivative $\frac{d^2u}{dx^2}$ in a linear way and hence (see Section 8.4) it may be described as a *quasi-linear* equation in those cases when it is not in fact linear.

It is of some importance to notice that we have not shown that (80) has a solution satisfying (74) or, if it has such a solution, that this solution does indeed maximize or minimize I. We have indicated only that (80) is a necessary condition, which must be satisfied by u if u is to qualify. The formulation of sufficient conditions, which ensure that a function u(x) so obtained truly maximizes or minimizes I (or makes it a relative maximum or minimum in some sense), is much more difficult.

Not infrequently, in practice, one can be certain in advance that an admissible maximizing (or minimizing) function *exists*. In such a case that function necessarily will be obtained as the solution of (80) which satisfies (74), or will be one such solution if there are several. Solutions of (80) are

often called *extremals* of the variational problem, whether or not they satisfy (74) and maximize or minimize *I*.

In the case of the example cited at the beginning of this section, where $F = (1 + u'^2)^{1/2}$, the fact that F depends only upon u' shows that the relevant Euler equation (81) reduces to the form u'' = 0, so that (as was to be expected) u must be a linear function, $u = c_1 x + c_2$. The given end conditions yield $c_1 = 1$ and $c_2 = 0$ and hence u(x) = x.

As a second illustration, we seek to minimize the integral

$$I = \int_0^{\pi/2} \left[\left(\frac{dy}{dt} \right)^2 - y^2 + 2ty \right] dt$$

y(0) = 0, y(\pi/2) = 0.

with

The Euler equation (80), with u and x replaced by y and t, respectively, becomes

$$\frac{d}{dt}\left(2\frac{dy}{dt}\right) - (-2y + 2t) = 0 \quad \text{or} \quad \frac{d^2y}{dt^2} + y = t,$$

from which there follows $y = c_1 \cos t + c_2 \sin t + t$. The end conditions then give $c_1 = 0$, $c_2 = -\pi/2$, and hence

$$y=t-\frac{\pi}{2}\sin t,$$

in correspondence with which

$$I_{\min} = -\frac{\pi}{2} \left(1 - \frac{\pi^2}{12} \right).$$

Generalizations, in which more dependent and/or independent variables are involved or which involve other modifications, as well as formulations of sufficiency conditions, may be found in the literature.

Two such generalizations, which are particularly straightforward, may be described here:

(a) If (73) is replaced by the integral

$$I = \int_{a}^{b} F(x; u_{1}, \ldots, u_{n}; u'_{1}, \ldots, u'_{n}) dx, \qquad (82)$$

where values of the *n* independent unknown functions $u_1(x), \ldots, u_n(x)$ are each given at the end points x = a and x = b, we obtain an Euler equation similar to (80) in correspondence with each u_r ,

$$\frac{d}{dx}\left(\frac{\partial F}{\partial u'_r}\right) - \frac{\partial F}{\partial u_r} = 0 \qquad (r = 1, 2, \dots, n). \tag{83}$$

Thus, for example, the Euler equations associated with the integral

$$\int_a^b (u_1'^2 + u_2'^2 - 2u_1u_2 + 2xu_1) \, dx$$

are found to be

$$\frac{d}{dx}(2u'_1) - (-2u_2 + 2x) = 0 \quad and \quad \frac{d}{dx}(2u'_2) - (-2u_1) = 0$$
or
$$u''_1 + u_2 = x \quad and \quad u''_2 + u_1 = 0.$$

(b) Suppose that we are to maximize or minimize (73),

$$\int_{a}^{b} F(x,u,u') \, dx = \max \text{ or min}, \tag{84}$$

where u(x) is to satisfy the prescribed end conditions

$$u(a) = A, \qquad u(b) = B, \tag{85}$$

as before, but that also a *constraint* condition is imposed in the form

$$\int_a^b G(x,u,u') \, dx = K,\tag{86}$$

where K is a prescribed constant. In this case, the appropriate Euler equation is found to be the result of replacing F in (80) by the auxiliary function

$$H = F + \lambda G, \tag{87}$$

where λ is an unknown constant. This constant, which is of the nature of a Lagrange multiplier (Section 7.7), thus generally will appear in the Euler equation and in its solution, and is to be determined together with the two constants of integration in such a way that the *three* conditions of (85) and (86) are satisfied.

In illustration, to minimize the integral $\int_0^1 y'^2 dx$, subject to the end conditions y(0) = 0 and y(1) = 0 and to the constraint $\int_0^1 y dx = 1$, we write $H = y'^2 + \lambda y$, in correspondence with which the Euler equation is $2y'' - \lambda = 0$. Hence y must be of the form $y = \frac{1}{4}\lambda x^2 + c_1x + c_2$. The end conditions and the constraint condition yield $c_1 = 6$, $c_2 = 0$, and $\lambda = -24$, and hence there follows y = 6x(1 - x).

7.9. Differentiation of integrals involving a parameter. Rather frequently it is necessary to deal with a function $\varphi(x)$ defined by an integral of the form

$$\varphi(x) = \int_{A(x)}^{B(x)} f(x,t) \, dt, \tag{88}$$

where f is such that the integration cannot be effected analytically. In particular, an expression for the derivative $\varphi'(x)$ is often required.

If the limits A and B are finite constants, differentiation with respect to x under the integral sign can be justified for all x in an interval (a,b) when f and the resultant integrand $\frac{\partial f}{\partial x}$ are continuous for $a \leq x \leq b$ and $A \leq t \leq B$.

More generally, when the limits are not constant we can think of φ as a function of x directly and also indirectly, through the intermediate variables A and B, and hence write $\varphi = \varphi(x, A, B)$. It then follows as an application of (6) that

$$\frac{d\varphi}{dx} = \frac{\partial\varphi}{\partial x} + \frac{\partial\varphi}{\partial B}\frac{dB}{dx} + \frac{\partial\varphi}{\partial A}\frac{dA}{dx},$$
(89)

if the derivatives on the right are continuous, where $\frac{\partial \varphi}{\partial x}$ is to be calculated by treating A and B as constants, and hence by merely differentiating with respect to x under the integral sign (when f and $\frac{\partial f}{\partial x}$ are continuous). To evaluate the other partial derivatives of φ in (89), let F(x,t) be a function such that

$$f(x,t) = \frac{\partial F(x,t)}{\partial t}$$

There then follows

$$\varphi(x,A,B) = \int_{A}^{B} \frac{\partial F}{\partial t} dt = F(x,B) - F(x,A)$$

and hence, when x is held constant as A and B are imagined to vary, there follows

$$\frac{\partial \varphi}{\partial B} = \frac{\partial F(x,B)}{\partial B} = f(x,B), \qquad \frac{\partial \varphi}{\partial A} = -\frac{\partial F(x,A)}{\partial A} = -f(x,A).$$

By introducing these results into Equation (89), we thus obtain the useful formula

$$\frac{d}{dx}\int_{A}^{B}f(x,t)\,dt = \int_{A}^{B}\frac{\partial f(x,t)}{\partial x}\,dt + f(x,B)\frac{dB}{dx} - f(x,A)\frac{dA}{dx}\,,\qquad(90)$$

which is valid for all values of x in an interval (a,b) when f and $\frac{\partial f}{\partial x}$ are continuous for $a \leq x \leq b$ and $A \leq t \leq B$, and when A'(x) and B'(x) are continuous in (a,b). This formula is often known as Leibnitz's rule.

In illustration, if

$$y(x) = \int_{a}^{x} h(t) \sin(x-t) dt,$$
 (91)

then the repeated use of Equation (90) yields

$$y'(x) = \int_a^x h(t) \cos(x-t) dt$$

and

$$y''(x) = -\int_a^x h(t) \sin(x-t) dt + h(x).$$

Hence it follows that y(x) satisfies the differential equation

$$y''(x) + y(x) = h(x).$$
 (92)

In fact, by setting x = a in the expressions for y and $\frac{dy}{dx}$, we find that (91) defines that solution of (92) which satisfies the initial conditions

$$y(a) = 0, \quad y'(a) = 0.$$
 (93)

In the case of a function defined by an *improper* integral,

$$\varphi(x) = \int_{A(x)}^{\infty} f(x,t) dt, \qquad (94)$$

where the range of integration is infinite, the situation is somewhat more complicated. Here, assuming that the integral (94) in fact converges, it is known that the formula

$$\frac{d}{dx}\int_{A(x)}^{\infty}f(x,t)\,dt=\int_{A}^{\infty}\frac{\partial f(x,t)}{\partial x}\,dt-f(x,A)\frac{dA}{dx}$$
(95)

is valid for x in (a,b) when f and $\frac{\partial f}{\partial x}$ are continuous for all $t \ge A$ and $a \le x \le b$,

and A'(x) is continuous in (a,b), and when also there exists a function M(t), independent of x, such that

$$\left|\frac{\partial f(x,t)}{\partial t}\right| \leq M(t)$$

for all $t \ge A$ and $a \le x \le b$, and such that the integral

$$\int_A^\infty M(t)\,dt$$

converges. The situations in which $A = -\infty$ in (94) are included as special cases.

In illustration, we consider the integral

$$\varphi(x) = \int_0^\infty e^{-t^2} \cos\left(2tx\right) dt.$$
(96)

Formally, Equation (95) gives

$$\frac{d\varphi}{dx} = -2 \int_0^\infty t e^{-t^2} \sin(2tx) dt, \qquad (97)$$

and the validity of the use of that formula is established for all real x when we notice that $|t e^{-t^2} \sin (2tx)| \leq t e^{-t^2}$ for all real x and further that the integral $\int_0^\infty t e^{-t^2} dt$ converges.

Further, if Equation (97) is integrated by parts there follows

$$\frac{d\varphi}{dx} = \left[e^{-t^2}\sin(2tx)\right]_{t=0}^{t=\infty} - 2x\int_0^\infty e^{-t^2}\cos(2tx)\,dt$$
$$= -2x\int_0^\infty e^{-t^2}\cos(2tx)\,dt.$$

Hence φ satisfies the differential equation $\frac{d\varphi}{dx} + 2x\varphi = 0$, and therefore is of the form $\varphi(x) = c e^{-x^2}$. But when x = 0 Equation (96) gives

$$\varphi(0) = \int_0^\infty e^{-t^2} \, dt = \frac{1}{2} \sqrt{\pi},$$

by Equation (58) of Chapter 2, so that the constant c is determined and there follows

$$\int_0^\infty e^{-t^2} \cos(2tx) \, dt = \frac{1}{2} \sqrt{\pi} \, e^{-x^2}. \tag{98}$$

If we introduce the change in variables t = au, where a is real and positive, and write x = b/(2a), this result takes the useful form

$$\int_0^\infty e^{-a^2u^2} \cos bu \, du = \frac{\sqrt{\pi}}{2a} e^{-b^2/4a^2} \qquad (a > 0). \tag{99}$$

7.10. Newton's iterative method. The use of the procedure known as "Newton's method" (or as the "Newton-Raphson method") in obtaining successive approximations to roots of algebraic and transcendental equations in one variable is introduced in courses in elementary calculus. In this section we rederive the basic equation and indicate the role of Jacobian determinants in the generalization to the solution of simultaneous equations in several variables.

Suppose that we have obtained a first approximation to a certain root of an equation of the form f(x) = 0 and require a more nearly accurate value. If we denote the first approximation by x_0 , we then attempt to determine h so that $f(x_0 + h) = 0$. But for small values of h we have the Taylor series expansion

$$f(x_0 + h) = f(x_0) + h f'(x_0) + \frac{h^2}{2} f''(x_0) + \cdots .$$
 (100)

If the initial approximation is sufficiently accurate and if the higher derivatives of f are not excessively large at x_0 , we may neglect terms involving higher powers of h in the equation $f(x_0 + h) = 0$ and hence obtain for h the approximation

$$h_0 = -\frac{f(x_0)}{f'(x_0)} . \tag{101}$$

The next approximation for x is then taken as $x_1 = x_0 + h_0$, and the process is repeated. Geometrically, this procedure consists of approximating the curve representing y = f(x) by its tangent line at $x = x_0$, and of determining the intersection of the tangent line with the x axis. In unfavorable cases the process may not converge.

To examine the procedure more closely, we may consider the effect of retaining the h^2 term in the series expansion. Rather than solve the resultant quadratic equation for h_0 , we write the approximate relation in the form

$$h_0 \approx -\left(\frac{f_0}{f'_0} + \frac{h_0^2 f''_0}{2 f'_0}\right) = -\frac{f_0}{f'_0} \left(1 + \frac{h_0^2 f''_0}{2 f'_0}\right), \tag{102}$$

with an obvious abbreviated notation. From the first form it follows that the most favorable situation is that in which $|f''(x)| \ll |f'(x)|$ near the desired root. This is geometrically evident. From the second form, assuming the remaining terms to be negligible, we see that the magnitude of the correction h_0 will be *underestimated* if f''_0 and f_0 have the same sign and will be *overestimated* otherwise. In general the process will be more likely to converge if the process does not "overshoot" the true value. Thus it is preferable that the initial point be taken at that end of an interval, inside which the root is known to lie, for which f and f'' have the same sign (if this is possible or convenient).

It can be proved that, when f''(x) is continuous, the error e_k in the kth approximation x_k satisfies the equation

$$e_{k} = -e_{k-1}^{2} \frac{f''(t_{k-1})}{2f'(x_{k-1})}$$
(103)

where t_{k-1} is some number between x_{k-1} and the true root.

To illustrate Newton's method, we seek the real root of the equation

$$f(x) = x^3 + 3x - 1 = 0.$$

Since f(0) = -1 and f(1) = 3, and since f'(x) > 0 for all x, there is only one real root and it lies in (0,1). Further, linear interpolation (graphical or analytical) based on these two values of f suggests the starting value $x_0 = \frac{1}{3}$, for which it is also found that the desirable condition ff'' > 0 is satisfied. The formula

$$h_k = -\frac{f(x_k)}{f'(x_k)} = -\frac{x_k^3 + 3x_k - 1}{3(x_k^2 + 1)}$$

then yields $h_0 = -\frac{1}{90}$ when $x_0 = \frac{1}{3}$, so that $x_1 = 0.3333 - 0.0111 = 0.3222$ when four significant figures are retained. Next, with k = 1, there follows $h_1 = -0.00001465$, to four significant figures, and hence $x_2 = 0.32218535$, to eight places.

Equation (103) here is of the form

$$e_k = -\frac{t_{k-1}}{1+x_{k-1}^{*}}e_{k-1}^{*}$$

and, since here t_{k-1} certainly is in (0,1) when x_{k-1} is in that interval, it follows that certainly $|e_k| < e_{k-1}^2$ in the present case. If we estimate e_1 as approximately $x_0 - x_1 \approx 2 \times 10^{-5}$, we deduce that $|e_2| < 4 \times 10^{-10}$, indicating that the eight digits in x_2 are correct and, indeed, that a ninth digit might have been properly retained in its calculation.

A sometimes useful higher-order procedure consists of first calculating an approximation to h_0 by neglecting h_0^2 in (102) and of then introducing this value into the right-hand member of that equation to obtain an "improved" approximation.

Now suppose that a certain solution of the simultaneous equations

$$f(x,y) = 0, \qquad g(x,y) = 0$$
 (104)

is required, and that a reasonably accurate initial approximation (x_0, y_0) has been obtained by some method. We next attempt to determine values of h and k such that the equations

$$f(x_0 + h, y_0 + k) = 0, \quad g(x_0 + h, y_0 + k) = 0$$

are simultaneously satisfied. If the left-hand members are expanded in Taylor series about the initial point and if only linear terms are retained, these equations become

$$f_0 + h f_{x0} + k f_{y0} = 0,$$

$$g_0 + h g_{x0} + k g_{y0} = 0,$$
(105)

where the zero subscripts indicate that the functions involved are evaluated at the point (x_0, y_0) . Thus approximate corrections h_0 and k_0 are given by the solution of these equations, in the form

$$h_{0} = -\frac{\begin{vmatrix} f_{0} & f_{v0} \\ g_{0} & g_{v0} \end{vmatrix}}{\frac{\partial(f,g)_{0}}{\partial(x,y)}}, \qquad k_{0} = -\frac{\begin{vmatrix} f_{x0} & f_{0} \\ g_{x0} & g_{0} \end{vmatrix}}{\frac{\partial(f,g)_{0}}{\partial(x,y)}}.$$
 (106)

It is seen that the success of this method depends in part upon the magnitude of the Jacobian determinant in the neighborhood of the desired solution.

To see the geometrical significance of this dependence, we recall that the vectors

$$\nabla f = f_x \mathbf{i} + f_y \mathbf{j}, \quad \nabla g = g_x \mathbf{i} + g_y \mathbf{j}$$

are normal to the curves representing f = 0 and g = 0 in the xy plane. Further, we obtain the result

$$(\nabla f) \times (\nabla g) = \mathbf{k} \left(f_x g_y - f_y g_x \right) = \mathbf{k} \frac{\partial(f,g)}{\partial(x,y)},$$

and it follows from the definition of the cross product (Section 6.3) that the numerical value of the Jacobian is proportional to the sine of the angle between the normals (and hence also between the tangents) to the curves at points of intersection. Thus, small values of the Jacobian may be expected to correspond to cases in which the intersecting curves have nearly equal slope at the intersection, in which cases slow convergence (or divergence) of the iterative process is to be anticipated. Although procedures of higher-order accuracy can be derived by taking second-order terms into account, such procedures are rather cumbersome.

To illustrate the procedure in a favorable case, we consider the simultaneous equations

$$f = x^3 - y^2 - 3x + y + 2 = 0, \quad g = x^2 + y^2 - 4 = 0.$$

By rough graphical methods it is found that the circle g = 0 intersects the cubic curve f = 0 at a point with *approximate* coordinates (1.5, 1.5). Taking $x_0 = y_0 = 1.5$, Equations (106) give $h_0 = -0.080$ and $k_0 = -0.087$, leading to the first improved coordinates $x_1 = 1.420$, $y_1 = 1.413$. The exact values are $x = y = \sqrt{2} = 1.414...$

The methods of this section clearly can be generalized to the solution of n equations in n variables.

REFERENCES

- 1. Apostle, T. M., Mathematical Analysis, Addison-Wesley Publishing Co., Inc., Reading, Mass., 1957.
- 2. Bliss, G. A., Calculus of Variations, First Carus Mathematical Monograph, The Open Court Publishing Co., La Salle, Ill., 1925.
- 3. Buck, R. C., Advanced Calculus, McGraw-Hill Book Company, Inc., New York, 1956.
- 4. Franklin, P., A Treatise on Advanced Calculus, John Wiley & Sons, Inc., New York, 1940.
- 5. Hardy, G. H., A Course of Pure Mathematics, Cambridge University Press, Cambridge, 1959.
- Kaplan, W., Advanced Calculus, Addison-Wesley Publishing Co., Inc., Reading, Mass., 1952.
- 7. Taylor, A. E., Advanced Calculus, Ginn and Company, New York, 1955.
- 8. Weinstock, R., Calculus of Variations: With Applications to Physics and Engineering, McGraw-Hill Book Company, Inc., New York, 1952.
- 9. Widder, D. V., Advanced Calculus, 2nd ed., Prentice-Hall, Inc., Englewood Cliffs, N.J., 1961.

PROBLEMS

Section 7.1

1. If $x = r \cos \theta$ and $y = r \sin \theta$, determine expressions for each of the following and express each result as a function of r and θ :

(a)
$$\left(\frac{\partial x}{\partial r}\right)_{\theta}$$
, (b) $\left(\frac{\partial x}{\partial \theta}\right)_{r}$, (c) $\left(\frac{\partial y}{\partial r}\right)_{\theta}$, (d) $\left(\frac{\partial y}{\partial \theta}\right)_{r}$,
(e) $\left(\frac{\partial r}{\partial x}\right)_{y}$, (f) $\left(\frac{\partial r}{\partial y}\right)_{x}$, (g) $\left(\frac{\partial \theta}{\partial x}\right)_{y}$, (h) $\left(\frac{\partial \theta}{\partial y}\right)_{x}$.

2. If $x = r \cos \theta$ and $y = r \sin \theta$, express each of the following as a function of r and θ :

(a)
$$\left(\frac{\partial y}{\partial x}\right)_r$$
, (b) $\left(\frac{\partial x}{\partial y}\right)_r$, (c) $\left(\frac{\partial y}{\partial x}\right)_{\theta}$, (d) $\left(\frac{\partial x}{\partial y}\right)_{\theta}$,
(e) $\left(\frac{\partial r}{\partial \theta}\right)_x$, (f) $\left(\frac{\partial \theta}{\partial r}\right)_x$, (g) $\left(\frac{\partial r}{\partial \theta}\right)_y$, (h) $\left(\frac{\partial \theta}{\partial r}\right)_y$.

3. If s and t are functions of x and y, say s = f(x,y) and t = g(x,y), show that

$$\frac{\partial F}{\partial x} = \left(s_x \frac{\partial}{\partial s} + t_x \frac{\partial}{\partial t}\right) F, \quad \frac{\partial F}{\partial y} = \left(s_y \frac{\partial}{\partial s} + t_y \frac{\partial}{\partial t}\right) F.$$

4. With the notation of Problem 1, show that

$$\frac{\partial^2 F}{\partial x \ \partial y} = \left(s_x \frac{\partial}{\partial s} + t_x \frac{\partial}{\partial t}\right) \left(s_y \frac{\partial F}{\partial s} + t_y \frac{\partial F}{\partial t}\right)$$
$$= s_x s_y \frac{\partial^2 F}{\partial s^2} + (s_x t_y + s_y t_x) \frac{\partial^2 F}{\partial s \ \partial t} + t_x t_y \frac{\partial^2 F}{\partial t^2}$$
$$+ \left[\left(s_x \frac{\partial}{\partial s} + t_x \frac{\partial}{\partial t}\right) s_y\right] \frac{\partial F}{\partial s} + \left[\left(s_x \frac{\partial}{\partial s} + t_x \frac{\partial}{\partial t}\right) t_y\right] \frac{\partial F}{\partial t},$$

and hence obtain the result

$$\frac{\partial^2 F}{\partial x \partial y} = s_x s_y \frac{\partial^2 F}{\partial s^2} + (s_x t_y + s_y t_x) \frac{\partial^2 F}{\partial s \partial t} + t_x t_y \frac{\partial^2 F}{\partial t^2} + s_{xy} \frac{\partial F}{\partial s} + t_{xy} \frac{\partial F}{\partial t}$$

[Notice that the variables (x, y) and (s, t) may be interchanged throughout.]

5. (a) By identifying x and y in the result of Problem 4, obtain the relation

$$\frac{\partial^2 F}{\partial x^2} = s_x^2 \frac{\partial^2 F}{\partial s^2} + 2s_x t_x \frac{\partial^2 F}{\partial s \partial t} + t_x^2 \frac{\partial^2 F}{\partial t^2} + s_{xx} \frac{\partial F}{\partial s} + t_{xx} \frac{\partial F}{\partial t},$$

as well as an analogous expression for $\frac{\partial^2 F}{\partial v^2}$.

(b) If $t_y = s_x$ and $t_x = -s_y$, show that

$$\nabla^2 F \equiv \frac{\partial^2 F}{\partial x^2} + \frac{\partial^2 F}{\partial y^2} = (s_x^2 + s_y^2) \left(\frac{\partial^2 F}{\partial s^2} + \frac{\partial^2 F}{\partial t^2} \right).$$

6. If f(p,v,T) = 0, show that

$$\left(\frac{\partial v}{\partial T}\right)_p = -\frac{f_T}{f_v}, \qquad \left(\frac{\partial p}{\partial v}\right)_T = -\frac{f_v}{f_p}, \qquad \left(\frac{\partial p}{\partial T}\right)_v = -\frac{f_T}{f_p},$$

and deduce that

$$\left(\frac{\partial p}{\partial T}\right)_{v} = -\left(\frac{\partial v}{\partial T}\right)_{p}\left(\frac{\partial p}{\partial v}\right)_{T}.$$

7. If E = f(p,T) and T = g(p,v), show that

(a)
$$\left(\frac{\partial E}{\partial v}\right)_{p} = f_{T}g_{v} \equiv \left(\frac{\partial E}{\partial T}\right)_{p} \left(\frac{\partial T}{\partial v}\right)_{p},$$

(b) $\left(\frac{\partial E}{\partial p}\right)_{v} = f_{p} + f_{T}g_{p} \equiv \left(\frac{\partial E}{\partial p}\right)_{T} + \left(\frac{\partial E}{\partial T}\right)_{p} \left(\frac{\partial T}{\partial p}\right)_{v}.$

Section 7.2

8. If u and v are functions of r and s, and also r and s are functions of x and y, prove that

$$\frac{\partial(u,v)}{\partial(r,s)}\frac{\partial(r,s)}{\partial(x,y)}=\frac{\partial(u,v)}{\partial(x,y)}.$$

9. The variables x and y are expressed in terms of the variables u and v by the equations

 $x = F(u,v), \quad y = G(u,v).$

(a) By writing

$$f(x,y,u,v) = x - F(u,v), \quad g(x,y,u,v) = y - G(u,v)$$

and thus reducing the given equations to a special case of Equations (18), show that the statement following Equation (22) implies that u and v may be considered as

functions of x and y if the Jacobian $J = \frac{\partial(F,G)}{\partial(u,v)}$ does not vanish, and that this condition is equivalent to the requirement

$$J=\frac{\partial(x,y)}{\partial(u,v)}\neq 0$$

when x and y are considered as functions of u and v.

(b) Assuming that $J \neq 0$, show that Equations (21) then become

$$u_x=\frac{y_v}{J}, \quad u_y=-\frac{x_v}{J}; \quad v_x=-\frac{y_u}{J}, \quad v_y=\frac{x_u}{J}.$$

(c) Obtain the same results directly by differentiating the original equations partially with respect to x and with respect to y, in each case holding the second variable constant, to obtain the relations

$$1 = F_u u_x + F_v v_x, \qquad 0 = F_u u_y + F_v v_y, \\ 0 = G_u u_x + G_v v_x, \qquad 1 = G_u u_y + G_v v_y,$$

and by solving these equations for the desired quantities.

10. The five variables x, y, z, u, and v are related by three equations of the form

$$f(x,u,v) = 0, \quad g(y,u,v) = 0, \quad h(z,u,v) = 0.$$

(a) State conditions under which the first two equations determine u and v as functions of x and y and the third equation determines z as a function of u and v, and hence then as a function of x and y.

(b) Considering z, u, and v as functions of the independent variables x and y, differentiate the three equations partially with respect to x, holding y constant, and hence show that u_x , v_x , and z_x satisfy the equations

$$f_u u_x + f_v v_x = -f_x,$$

$$g_u u_x + g_v v_x = 0,$$

$$h_z z_x + h_u u_x + h_v v_x = 0.$$

(c) By solving these equations for z_x (by determinants), obtain the result

$$z_x = \frac{f_x}{h_z} \cdot \frac{\frac{\partial(h,g)}{\partial(u,v)}}{\frac{\partial(f,g)}{\partial(u,v)}}$$

(d) Show also, by symmetry or otherwise, that

$$z_{\mathbf{y}} = -\frac{g_{\mathbf{y}}}{h_z} \cdot \frac{\frac{\partial(h,f)}{\partial(u,v)}}{\frac{\partial(f,g)}{\partial(u,v)}}.$$

(e) Verify these results in the case when

$$2x^2 - u + v = 0$$
, $2y^3 - u - v = 0$, $z + u - v^2 = 0$

by first obtaining z explicitly as a function of x and y and differentiating, and also using the results of parts (c) and (d), to determine z_x and z_y .

11. Show that the equations

$$x = F(u,v),$$
 $y = G(u,v),$ $z = H(u,v)$

determine z as a function of x and y such that

$$z_x = \frac{\frac{\partial(z,y)}{\partial(u,v)}}{\frac{\partial(x,y)}{\partial(u,v)}}, \qquad z_y = -\frac{\frac{\partial(z,x)}{\partial(u,v)}}{\frac{\partial(x,y)}{\partial(u,v)}},$$

if $\frac{\partial(x,y)}{\partial(u,v)} \neq 0$. (Write $f = x - F, g = y - G, h = z - H$ in Problem 10.)

12. If f(x,y,z) = 0 and g(x,y,z) = 0, we may in general consider any two of the variables as functions of the third. Show that

$$\frac{dx}{\frac{\partial(f,g)}{\partial(y,z)}} = \frac{dy}{\frac{\partial(f,g)}{\partial(z,x)}} = \frac{dz}{\frac{\partial(f,g)}{\partial(x,y)}}$$

if no denominator vanishes.

13. Show that the tangent line, at a point (x_0, y_0, z_0) , to the curve of intersection of the surfaces f(x, y, z) = 0 and g(x, y, z) = 0 is specified by the equations

$$\frac{x - x_0}{\left[\frac{\partial(f,g)}{\partial(y,z)}\right]_0} = \frac{y - y_0}{\left[\frac{\partial(f,g)}{\partial(z,x)}\right]_0} = \frac{z - z_0}{\left[\frac{\partial(f,g)}{\partial(x,y)}\right]_0}$$

Section 7.3

14. Prove that the functions

$$u_1 = \frac{x+y}{x-y}, \qquad u_2 = \frac{xy}{(x-y)^2}$$

are functionally dependent.

15. Prove that the functions $u_1 = y + z$, $u_2 = x + 2z^2$, and $u_3 = x - 4yz - 2y^2$ are functionally dependent.

16. Determine whether the functions

$$u_1 = \frac{x-y}{x+z}$$
 and $u_2 = \frac{x+z}{y+z}$

are functionally dependent.

17. Determine whether the functions

$$u_1 = \frac{yz - x}{x}, \quad u_2 = \frac{xyz - y^2z^2 + x^2}{xyz + y^2z^2}$$

are functionally dependent.

Section 7.4

18. The coordinates u, φ , and θ are defined by the equations

$$x = au \cos \theta \sin \varphi, \quad y = bu \sin \theta \sin \varphi, \quad z = cu \cos \varphi$$

where $u \ge 0, \quad 0 \le \varphi \le \pi$, and $0 \le \theta < 2\pi$.

(a) Show that the surfaces u = constant are the ellipsoids

$$\left(\frac{x}{au}\right)^2 + \left(\frac{y}{bu}\right)^2 + \left(\frac{z}{cu}\right)^2 = 1$$

whereas the surfaces $\varphi = \text{constant}$ and $\theta = \text{constant}$ are elliptical cones and planes, respectively.

(b) Show that the element of volume is of the form

$$d au = abc \ u^2 \sin \varphi \ du \ d\varphi \ d\theta.$$

19. Use the result of Problem 18 to show that the z coordinate of the center of gravity of that half of the ellipsoid

$$\frac{x^2}{a^2} + \frac{y^2}{b^2} + \frac{z^2}{c^2} = 1$$

which lies above the xy plane is given by

$$V\bar{z} = abc^2 \int_0^1 u^3 du \int_0^{\pi/2} \sin\varphi \cos\varphi d\varphi \int_0^{2\pi} d\theta = \frac{\pi}{4} abc^2,$$

where

$$V = abc \int_0^1 u^2 du \int_0^{\pi/2} \sin \varphi \, d\varphi \int_0^{2\pi} d\theta = \frac{2\pi}{3} \, abc,$$

and hence has the value $\bar{z} = 3c/8$.

20. Suppose that a closed curve C, surrounding the origin in the xy plane, is specified by the parametric equations

$$x = f(\theta), \quad y = g(\theta) \quad (0 \le \theta < 2\pi)$$

(a) Show that, if coordinates u and θ are defined by the equations

$$x = u f(\theta), \qquad y = u g(\theta),$$

then the area A enclosed by C is specified by the ranges $0 \le u \le 1$ and $0 \le \theta < 2\pi$.

(b) Show that the element of area in $u\theta$ coordinates is given by

$$dA = |W(\theta)| u \, du \, d\theta,$$

where $W(\theta)$ is the Wronskian (see Section 1.2) of $f(\theta)$ and $g(\theta)$.

$$W(\theta) \equiv f(\theta) g'(\theta) - f'(\theta) g(\theta).$$

Deduce that

$$\iint_A F(x,y) \, dx \, dy = \int_0^{2\pi} \int_0^1 F[uf(\theta), ug(\theta)] \, | \, W(\theta) | \, u \, du \, d\theta$$

21. (a) If a surface is defined by the equations

$$x = x(u_1, u_2), \quad y = y(u_1, u_2), \quad z = z(u_1, u_2),$$

show that the directed element of *surface area* on that surface is given by

 $d\sigma = \mathbf{U}_1 \times \mathbf{U}_2 \, du_1 \, du_2,$

where $U_1 = \frac{\partial \mathbf{r}}{\partial u_1}$ and $U_2 = \frac{\partial \mathbf{r}}{\partial u_2}$, and where **r** is the position vector $x \mathbf{i} + y \mathbf{j} + z \mathbf{k}$.

Problems

(b) Use Lagrange's identity [Equation (36) of Section 6.4] to show that the scalar element of surface area is of the form

$$d\sigma = \sqrt{EG - F^2} \, du_1 \, du_2$$

where

$$E = \frac{\partial \mathbf{r}}{\partial u_1} \cdot \frac{\partial \mathbf{r}}{\partial u_1}, \qquad F = \frac{\partial \mathbf{r}}{\partial u_1} \cdot \frac{\partial \mathbf{r}}{\partial u_2}, \qquad G = \frac{\partial \mathbf{r}}{\partial u_2} \cdot \frac{\partial \mathbf{r}}{\partial u_2}.$$

22. Show that the element of arc length ds along any curve C on the surface considered in Problem 21 can be obtained, from the fact that

$$\frac{d\mathbf{r}}{ds} = \frac{\partial \mathbf{r}}{\partial u_1} \frac{du_1}{ds} + \frac{\partial \mathbf{r}}{\partial u_2} \frac{du_2}{ds}$$

is a unit vector tangent to the curve C, in the form

$$ds = \sqrt{E^2 \, du_1^2 + 2F \, du_1 \, du_2 + G^2 \, du_2^2},$$

where E, F, and G are defined in Problem 21.

Section 7.5

23. Obtain the terms of degree two or less in the Taylor expansion

$$e^{x+y} = a_0 + b_1 x + b_2 y + c_1 x^2 + c_2 x y + c_3 y^2 + \cdots$$

- (a) by direct use of the two-dimensional Taylor formula,
- (b) by first expanding e^{x+y} in a series of powers of x + y,
- (c) by multiplying together appropriate expansions of e^x and e^y .
- 24. Obtain the terms of degree two or less in the Taylor expansion

$$\frac{1}{1+x-y} = a_0 + b_1(x-1) + b_2(y-1) + c_1(x-1)^2 + c_2(x-1)(y-1) + c_3(y-1)^2 + \cdots$$

- (a) by direct use of the two-dimensional Taylor formula,
- (b) by any shorter method.

25. For a certain function f(x,y), it is known that

$$f(0,0) = 1, \quad f_x(0,0) = 0.25, \quad f_y(0,0) = 0.50$$

and that $|f_{xx}| \leq 0.15$, $|f_{xy}| \leq 0.05$, and $|f_{yy}| \leq 0.05$ everywhere along that segment of the line y = x which joins (0,0) and (0.1,0.1). Show that

$$1.0735 \leq f(0.1,0.1) \leq 1.0765.$$

26. Obtain a three-decimal place approximation to the value of

$$\int_0^1 \int_0^1 e^{-x^2 y^2} \, dx \, dy$$

by use of Taylor series.

Section 7.6

27. Locate and identify all relative maxima and minima and saddle points (if such exist) of the following functions:

(a)
$$x^2 + 2bxy + y^2$$
,
(b) $x^2 - xy + y^2 - 2x + y$,
(c) $x^2y^2 - x^2 - y^2$,
(d) $x^2y^3 - 3x - 3y$.

28. Locate and identify all points at which absolute maxima or minima are assumed by the following functions:

(a)
$$x^{2/3} + y^{2/3}$$
,
(b) $(x^2 + y^2 - 2x + 2y + 2)^{1/2}$,
(c) $(y^2 - x^2)^{1/2}$,
(b) $(2x - 2y - x^2 - y^2)^{1/2}$.

29. Locate and classify all maxima and minima and determine any saddle points for the following functions:

(a)
$$2x^4 - x^2y^2 + 2y^4$$
,
(b) $x^4 - 2x^2y^2 + y^4$,
(c) $x^4 - 4x^2y^2 + y^4$,
(d) $x^4 + y^4$.

30. Find the absolute maximum and minimum of the function $x^2 - y^2 - 2x$ in the region $x^2 + y^2 \le 1$. (Write $x = \cos \theta$, $y = \sin \theta$ on the boundary.)

Section 7.7

31. Determine when, if ever, $x^2 + y^2$ takes on a minimum or a maximum value under the constraint xy = 1.

32. Determine when, if ever, xy takes on a maximum or a minimum value under the constraint $x^2 + y^2 = 2$.

33. Minimize $x^2 + 4y^2 + 16z^2$ under each of the following constraints:

(a)
$$xyz = 1$$
, (b) $xy = 1$, (c) $x = 1$

34. Determine the points on the ellipse, defined by the intersection of the surfaces x + y = 1 and $x^2 + 2y^2 + z^2 = 1$, which are nearest to and furthest from the origin.

Section 7.8

35. Obtain the Euler equation associated with maximizing or minimizing $\int_{a}^{b} F dx$ in the following cases:

(a)
$$F = p(x)y'^2 - q(x)y^2 + 2f(x)y$$
,
(b) $F = A(x)y'^2 + 2B(x)yy' + C(x)y^2 + 2D(x)y' + 2E(x)y$.

36. Obtain the two Euler equations associated with maximizing or minimizing $\int_{a}^{b} F dx$ in the following cases:

(a)
$$F = A(x) u_1'^2 + 2B(x)u_1'u_2' + C(x)u_2'^2$$
,
(b) $F = a_{11}(x)u_1'^2 + 2a_{12}(x)u_1'u_2' + a_{22}(x)u_2'^2 + b_{11}(x)u_1^2 + 2b_{12}(x)u_1u_2 + b_{22}(x)u_2^2$.

Problems

37. Geodesics on a surface are the extremals of the problem of minimizing the distance, along that surface, between two points on the surface. Determine the geodesics on the right circular cylinder $x^2 + y^2 = a^2$, by writing $x = a \cos \theta$ and $y = a \sin \theta$ and proceeding by each of the following methods:

(a) Assume $z = u(\theta)$, so that $ds = \sqrt{a^2 + u'^2} d\theta$,

(b) Assume
$$\theta = v(z)$$
, so that $ds = \sqrt{a^2 v'^2 + 1} dz$,

(c) Assume z = f(t) and $\theta = g(t)$, where t is a parameter, so that

$$ds = \sqrt{a^2 f'^2 + g'^2} \, dt.$$

[Notice that in (a) we assume that z can be expressed as a single-valued differentiable function of θ , whereas in (b) the roles of the variables are reversed. In (c) neither assumption is made.]

38. Determine the geodesics on the sphere $x^2 + y^2 + z^2 = a^2$, using the spherical coordinates defined by Equation (163) of Chapter 6 and assuming that the required equation can be written in the form $\theta = u(\varphi)$. (Take $\cot \varphi$ as a new variable in evaluating a relevant integral.)

39. Determine the minimum and maximum values of the integral

$$\int_0^1 yy'\,dx,$$

subject to the end conditions y(0) = 0, y(1) = 1. Also account for the degenerate form of the Euler equation in this case.

40. Show that the Euler equation corresponding to minimizing the integral

$$I \equiv \int_a^b [p(x)y'^2 - q(x)y^2] dx,$$

subject to the end conditions y(a) = 0, y(b) = 0 and to the constraint

$$\int_a^b r(x)y^2\,dx\,=1,$$

is

$$\frac{d}{dx}\left(p\frac{dy}{dx}\right) + (q + \lambda r)y = 0,$$

when the Lagrange multiplier is denoted by $-\lambda$. [Notice that here λ is a characteristic number associated with a Sturm-Liouville problem (see Section 5.6).]

41. Find the minimum value of $\int_0^1 y'^2 dx$, subject to the conditions

$$y(0) = y(1) = 0$$
 and $\int_0^1 y^2 dx = 1$.

42. If the length of a curvilinear arc joining the points (0,0) and (1,0) is prescribed as L (with $L < \pi/2$) and if the area bounded above by this arc and below by the x axis is to be maximized, show that the arc must be circular. Assume that we may write y = u(x) along the arc.

Section 7.9

43. If

$$I(a) = \int_0^1 \frac{x^a - x^b}{\log x} \, dx,$$

where a > b > -1, show that

$$I'(a) = \int_0^1 x^a \, dx = \frac{1}{a+1} \, ,$$

and hence, noticing that I(b) = 0, obtain the result

$$\int_0^1 \frac{x^a - x^b}{\log x} \, dx = \int_b^a \frac{da}{a+1} = \log \frac{a+1}{b+1} \qquad (a > b > -1)$$
44. (a) If

$$I(a) = \int_0^\infty \frac{e^{-ax} \sin x}{x} \, dx,$$

where a > 0, show that $I'(a) = -1/(a^2 + 1)$. Hence, noticing that $I(\infty) = 0$, obtain the result

$$\int_0^\infty \frac{e^{-ax} \sin x}{x} \, dx = \frac{\pi}{2} - \tan^{-1} a \qquad (a > 0).$$

(b) By considering the limit of this result as $a \rightarrow 0$, and making the justifiable assumption that the limit can be taken under the integral sign, deduce also that

$$\int_0^\infty \frac{\sin x}{x}\,dx\,=\frac{\pi}{2}\,.$$

45. (a) By replacing x by ax in the result of Problem 44(b), obtain the result

$$I(a) \equiv \int_0^\infty \frac{\sin ax}{x} \, dx = \frac{\pi}{2} \qquad (a > 0).$$

(b) Show that, although hence I'(a) = 0 when a > 0, the derivative cannot be determined by differentiating under the integral sign.

(c) Show that I(0) = 0 and also that I(-a), = -I(a) so that I(a) must be given by $-\pi/2$ when a is negative.

46. Given the evaluation

$$\int_{-\infty}^{\infty} e^{-x^2} dx = \sqrt{\pi}.$$

[see Equation (58) of Chapter 2], determine the value of the integral

$$\int_{-\infty}^{\infty} x^2 e^{-x^2} dx$$

by replacing x by ux in the original integral, dividing by u, differentiating with respect to u, and finally setting u = 1.

47. Obtain a differential equation, together with appropriate initial conditions, satisfied by the function

$$y(x) = \frac{1}{2!} \int_a^x (x - t)^2 h(t) dt.$$

Section 7.10

48. Determine to five places the real root of the equation $x^3 - 2x - 5 = 0$.

49. Determine to three places the smallest positive root of the equation

 $\tan x = x$

by Newton's method. [Suggestions: Notice that $x = 3\pi/2$ is a fair approximation, obtained by sketching the curves $y = \tan x$ and y = x on the same graph and estimating the abscissa of the intersection. Take $f(x) = \sin x - x \cos x$, rather than $f(x) = \tan x - x$, to avoid the infinity of the latter function.]

50. Determine to three places the smallest positive root of the equation tanh x = tan x by Newton's method.

51. Determine to three places the real solution of the simultaneous equations

$$4x^3 - 27xy^2 + 25 = 0, \qquad 4x^2y - 3y^2 - 1 = 0$$

in the first quadrant.

52. The path of a projectile moving in the xy plane is specified by the parametric equations x = f(t), y = g(t) where t is time. It is required to determine the time at which its trajectory will intercept a curve specified by the equation $\varphi(x,y) = 0$. If the approximate time is t_0 and the approximate coordinates of the interception are (x_0, y_0) , show that Newton's method yields a new estimate $t = t_0 + \tau$, where

$$\tau = -\frac{\varphi_0 + (f_0 - x_0)\varphi_{x0} + (g_0 - y_0)\varphi_{y0}}{f_0'\varphi_{x0} + g_0'\varphi_{y0}},$$

and where the zero subscripts indicate evaluation for $t = t_0$, $x = x_0$, and $y = y_0$.

Partial Differential Equations

8.1. Definitions and examples. A partial differential equation is said to be *linear* if, when the equation has been rationalized and cleared of fractions, no powers or products of the unknown function or its partial derivatives are present. If it is true only that no powers or products of the partial derivatives of highest order are present, the equation is said to be quasi-linear. Thus, for example, the equation

$$x\frac{\partial z}{\partial x} + y\frac{\partial z}{\partial y} = z$$

is a linear equation in z, of first order, whereas the equation

$$z\frac{\partial^2 z}{\partial x^2} + \left(\frac{\partial z}{\partial y}\right)^2 = 0$$

is a quasi-linear equation of second order.

In this work we deal principally with linear equations, although certain procedures to be developed can also be applied to the solution of certain quasi-linear equations.

We recall that in the case of an ordinary differential equation of order n, a general solution involves n independent arbitrary constants. In particular, the general solution of a *linear* ordinary differential equation expresses the unknown variable as a linear combination of n independent functions, the narbitrary constants appearing as the coefficients of the n functions in the linear combination. For *nonlinear* ordinary equations the constants in general appear in a more complicated way in the solution, and there may exist so-called "singular solutions" which do not involve the arbitrary constants and cannot be obtained from the first solution by specializing these constants. In the case of *partial* differential equations, the most general solutions are found to involve *arbitrary functions* of *specific functions*. As a simple example, we readily verify that the equation

$$2\frac{\partial z}{\partial x} - \frac{\partial z}{\partial y} = 0 \tag{1}$$

is satisfied by the expression

$$z = f(x + 2y), \tag{2}$$

no matter what functional relationship is indicated by f (so long as f is differentiable). For from Equation (2) we obtain

$$\frac{\partial z}{\partial x} = \frac{df(x+2y)}{d(x+2y)} \frac{\partial (x+2y)}{\partial x} = f'(x+2y),$$
$$\frac{\partial z}{\partial y} = \frac{df(x+2y)}{d(x+2y)} \frac{\partial (x+2y)}{\partial y} = 2f'(x+2y) = 2\frac{\partial z}{\partial x},$$

and hence (2) implies (1).

To see that (2) is indeed the *most general* solution of (1), we introduce the new variables

$$t = x + 2y, \qquad s = x. \tag{3}$$

There then follows

$$\frac{\partial z}{\partial x} = \frac{\partial z}{\partial t}\frac{\partial t}{\partial x} + \frac{\partial z}{\partial s}\frac{\partial s}{\partial x} = \frac{\partial z}{\partial t} + \frac{\partial z}{\partial s},$$
$$\frac{\partial z}{\partial y} = \frac{\partial z}{\partial t}\frac{\partial t}{\partial y} + \frac{\partial z}{\partial s}\frac{\partial s}{\partial y} = 2\frac{\partial z}{\partial t},$$

and (1) becomes merely

$$2\frac{\partial z(s,t)}{\partial s} = 0. \tag{4}$$

The general solution of this equation is clearly z = f(t) = f(x + 2y), where f is an arbitrary function, in accordance with (2). We see that such expressions as z = x + 2y + 1, $z = \sin(x + 2y)$, $z = 4\sqrt{x + 2y} + \cos(x + 2y)$, ..., are all particular solutions of the partial differential equation (1).

As a further example, we determine a partial differential equation which has the expression

$$z = f(2x + y) + g(x - y) - xy$$
 (5)

as its general solution. The procedure consists of attempting to obtain, by differentiation, sufficiently many relations to permit elimination of the arbitrary functions. With the notations

$$f'=\frac{df(2x+y)}{d(2x+y)}, \qquad g'=\frac{dg(x-y)}{d(x-y)}, \qquad \ldots,$$

the first and second derivatives of (5) are obtained in the form

$$\frac{\partial z}{\partial x} = 2f' + g' - y, \qquad \frac{\partial z}{\partial y} = f' - g' - x,$$
 (6a)

$$\frac{\partial^2 z}{\partial x^2} = 4f'' + g'', \quad \frac{\partial^2 z}{\partial x \partial y} = 2f'' - g'' - 1, \quad \frac{\partial^2 z}{\partial y^2} = f'' + g''.$$
(6b)

Equations (6b) constitute three relations involving the two arbitrary functions f'' and g''. If two of these relations are used to determine f'' and g'', and if the results are introduced into the third relation, there follows

$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial x \partial y} - 2 \frac{\partial^2 z}{\partial y^2} = 1.$$
 (7)

This is the differential equation of lowest order satisfied by (5), with f and g arbitrary functions.

We may verify that, in fact, Equation (5) is the most general solution of (7) by first noticing that since z = -xy satisfies (7), the function

$$w = z + xy \tag{8}$$

must be the general solution of the equation

$$\frac{\partial^2 w}{\partial x^2} - \frac{\partial^2 w}{\partial x \partial y} - 2 \frac{\partial^2 w}{\partial y^2} = 0.$$
(9)

Now, if we make the substitutions

$$2x + y = s, \quad x - y = t \tag{10}$$

suggested by (5), there follows

$$\frac{\partial}{\partial x} = \frac{\partial s}{\partial x}\frac{\partial}{\partial s} + \frac{\partial t}{\partial x}\frac{\partial}{\partial t} = 2\frac{\partial}{\partial s} + \frac{\partial}{\partial t},$$
$$\frac{\partial}{\partial y} = \frac{\partial s}{\partial y}\frac{\partial}{\partial s} + \frac{\partial t}{\partial y}\frac{\partial}{\partial t} = \frac{\partial}{\partial s} - \frac{\partial}{\partial t},$$

and (9) takes the form

$$\frac{\partial^2 w}{\partial s \ \partial t} = 0 \tag{11}$$

from which the general solution

$$w = f(s) + g(t) = f(2x + y) + g(x - y)$$
(12)

is obtained by direct integration. The introduction of (12) into (8) leads to (5).

In a similar way, the expression

$$z = f(ax + by) + g(cx + dy),$$
 (13)

where a, b, c, and d are constants such that $ad \neq bc$, is readily shown to be the general solution of a linear partial differential equation of the form

$$A\frac{\partial^2 z}{\partial x^2} + B\frac{\partial^2 z}{\partial x \partial y} + C\frac{\partial^2 z}{\partial y^2} = 0, \qquad (14)$$

where A, B, and C are constants depending upon a, b, c, and d.

In more involved partial differential equations, the arbitrary functional relations may enter into a general solution in much more complicated ways, and additional solutions may exist which cannot be obtained by specializing the general solution so obtained. Because of the fact that "general" solutions of partial differential equations involve arbitrary functions, the specialization of such solutions to particular forms which satisfy prescribed boundary conditions involves the determination of functional relations, rather than merely the determination of constants, and is usually not feasible. For this reason, we generally prefer to determine a set of particular solutions directly, and attempt to combine these solutions in such a way that the prescribed boundary conditions are satisfied. The development of such procedures, in cases of linear equations, forms the basis of most of Chapter 9. However, since certain general properties of solutions of certain equations are most readily obtained by studying the general solutions, we show in Sections 8.2 through 8.5 in what way such solutions can be obtained in certain simple cases of frequent occurrence.

8.2. The quasi-linear equation of first order. The most general quasi-linear partial differential equation of first order can be written in the form

$$P(x,y,z)\frac{\partial z}{\partial x} + Q(x,y,z)\frac{\partial z}{\partial y} = R(x,y,z), \qquad (15)$$

where z is the dependent variable and x and y are independent. If (15) is to be truly *linear*, then P and Q must be independent of z and R must be a linear function of z, say $R = R_1 z + R_2$, so that (15) becomes

$$P(x,y)\frac{\partial z}{\partial x} + Q(x,y)\frac{\partial z}{\partial y} = R_1(x,y) z + R_2(x,y)$$
(16)

in this special case.

Suppose that the equation

$$u(x,y,z) = c \tag{17}$$

is an *integral* of (15), in the sense that (17) determines z as a function of x and y which satisfies (15). Then, by partial differentiation we obtain the two results

$$\frac{\partial u}{\partial x} + \frac{\partial u}{\partial z}\frac{\partial z}{\partial x} = 0, \qquad \frac{\partial u}{\partial y} + \frac{\partial u}{\partial z}\frac{\partial z}{\partial y} = 0,$$
 (18)

where in the derivatives $\frac{\partial u}{\partial x}$, $\frac{\partial u}{\partial y}$, and $\frac{\partial u}{\partial z}$ the variables x, y, and z are considered as independent. Thus we may write

$$\frac{\partial z}{\partial x} = -\frac{\frac{\partial u}{\partial x}}{\frac{\partial u}{\partial z}}, \qquad \frac{\partial z}{\partial y} = -\frac{\frac{\partial u}{\partial y}}{\frac{\partial u}{\partial z}}, \qquad (19)$$

assuming, of course, that $\frac{\partial u}{\partial z} \neq 0$. If these expressions are introduced into (15), an equivalent equation is obtained in the form

$$P\frac{\partial u}{\partial x} + Q\frac{\partial u}{\partial y} + R\frac{\partial u}{\partial z} = 0.$$
 (20)

This form has the advantage that in it the variables x, y, and z play completely symmetrical roles, and we are readily led to a geometrical interpretation of the equation. Equation (20) obviously can be written in terms of a dot product,

$$(P\mathbf{i} + Q\mathbf{j} + R\mathbf{k}) \cdot \nabla u = 0.$$
(21)

Since ∇u is a vector normal to the surface u = c, Equation (21) states that the vector $P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}$ is perpendicular to the normal to that surface at any point on the surface, and hence lies in the tangent plane. Thus we see that at any point the vector $P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}$ is tangent to a curve in the integral surface u = c which passes through that point.

Thus the differential equation in any of its related forms (15), (20), and (21) may be considered as determining at any point in some three-dimensional region a *direction*, specified by the vector $\mathbf{V} = P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}$. If a particle moves from a given initial point in such a way that its direction at any point coincides with the direction of the vector \mathbf{V} at that point, a space curve is traced out. Such curves are called the *characteristic curves* of the differential equation.

In a similar way, the ordinary differential equation of first order,

$$\frac{dy}{dx} = f(x,y),$$

defines an angle $\varphi = \tan^{-1} f(x,y)$ at any point in some region in the xy plane, which the tangent to an integral curve must make with the x axis, and the general solution of the equation consists of the set of curves having the prescribed direction at any point in the two-dimensional region of definition.

We see next that any surface which is built up from such characteristic curves in space will have the property that the tangent plane at any point contains the vector $\mathbf{V} = P \mathbf{i} + Q \mathbf{j} + R \mathbf{k}$, and hence the normal vector at any point of such a surface is perpendicular to \mathbf{V} , as is required by (21). That is, any surface built up from characteristic curves is an integral surface of the differential equation.

If **r** is the position vector to a point of a characteristic curve, and if s represents arc length along the curve, then the unit tangent vector to the curve at that point is given by

$$\frac{d\mathbf{r}}{ds} = \frac{dx}{ds}\mathbf{i} + \frac{dy}{ds}\mathbf{j} + \frac{dz}{ds}\mathbf{k}.$$

The requirement that this vector have the same direction as the vector $P \mathbf{i} + Q + R \mathbf{k}$ becomes

$$P = \mu \frac{dx}{ds}, \qquad Q = \mu \frac{dy}{ds}, \qquad R = \mu \frac{dz}{ds},$$
 (22)

where μ may be a function of x, y, and z. These relations can be written in the differential form

$$\frac{dx}{P} = \frac{dy}{Q} = \frac{dz}{R},$$
(23)

and hence are equivalent to two ordinary differential equations.

Let the solutions of two independent equations which imply (23) be denoted by

$$u_1(x,y,z) = c_1, \quad u_2(x,y,z) = c_2,$$
 (24)

where c_1 and c_2 are independent constants. Then these equations represent two families of integral surfaces of Equation (20). A surface of one family intersects a surface of the second family in a characteristic curve. If we consider the characteristic curves which are the intersections of those surfaces for which c_1 and c_2 are related by an equation of the general form $F(c_1, c_2) = 0$, the locus of these intersections will determine a surface which is an integral surface, no matter what functional relationship is indicated by F. That is, any surface given by an equation of the form

$$F[u_1(x,y,z), u_2(x,y,z)] = 0, \qquad (25a)$$

or, equivalently,

$$u_2(x,y,z) = f[u_1(x,y,z)]$$
 (25b)

will be an integral surface of the given partial differential equation.

The general result can be summarized as follows:

The general solution of the equation

$$P\frac{\partial z}{\partial x} + Q\frac{\partial z}{\partial y} = R$$
(26)

is of the form

$$F(u_1, u_2) = 0 \quad or \quad u_2 = f(u_1),$$
 (27)

where $u_1(x,y,z) = c_1$ and $u_2(x,y,z) = c_2$ are solutions of any two independent differential equations which imply the relationships

$$\frac{dx}{P} = \frac{dy}{Q} = \frac{dz}{R}.$$
(28)

The intersection of any two of the surfaces $u_1 = c_1$ and $u_2 = c_2$ is a characteristic curve whose tangent at any point has the direction ratios (P,Q,R).

We notice that, since (28) is a consequence of (22), in case P, Q, or R is identically zero we must take dx, dy, or dz, respectively, equal to zero. Thus, the relations $\frac{dx}{P} = \frac{dy}{Q} = \frac{dz}{0}$ imply the two equations $Q \, dx = P \, dy$ and dz = 0.

We may now establish the validity of the general result by an argument which is independent of geometrical considerations. Let $u_1(x,y,z) = c_1$ and $u_2(x,y,z) = c_2$ be solutions of two independent equations which imply (23). Then since (23) implies (22), we have

$$0 = du_1 = \frac{\partial u_1}{\partial x} dx + \frac{\partial u_1}{\partial y} dy + \frac{\partial u_1}{\partial z} dz = \left(P \frac{\partial u_1}{\partial x} + Q \frac{\partial u_1}{\partial y} + R \frac{\partial u_1}{\partial z} \right) \frac{ds}{\mu}$$

Hence $u = u_1$ and (similarly) $u = u_2$ are solutions of (20). It follows also that if $u_1 = c_1$ and $u_2 = c_2$ determine z as functions of x and y, these functions are solutions of (15). We next replace the independent variables x, y, and z in (20) by the new variables $r = u_1(x, y, z)$, $s = u_2(x, y, z)$, and $t = \varphi(x, y, z)$, where φ is any function which is independent of u_1 and u_2 and which does *not* satisfy (20). There then follows

$$\frac{\partial u}{\partial x} = \frac{\partial u}{\partial r}\frac{\partial r}{\partial x} + \frac{\partial u}{\partial s}\frac{\partial s}{\partial x} + \frac{\partial u}{\partial t}\frac{\partial t}{\partial x} = \frac{\partial u}{\partial r}\frac{\partial u_1}{\partial x} + \frac{\partial u}{\partial s}\frac{\partial u_2}{\partial x} + \frac{\partial u}{\partial t}\frac{\partial \varphi}{\partial x},$$

together with similar expressions for $\frac{\partial u}{\partial y}$ and $\frac{\partial u}{\partial z}$. With these substitutions (20) takes the form

$$\left(P\frac{\partial u_1}{\partial x} + Q\frac{\partial u_1}{\partial y} + R\frac{\partial u_1}{\partial z}\right)\frac{\partial u}{\partial r} + \left(P\frac{\partial u_2}{\partial x} + Q\frac{\partial u_2}{\partial y} + R\frac{\partial u_2}{\partial z}\right)\frac{\partial u}{\partial s} + \left(P\frac{\partial \varphi}{\partial x} + Q\frac{\partial \varphi}{\partial y} + R\frac{\partial \varphi}{\partial z}\right)\frac{\partial u}{\partial t} = 0,$$

where *u* is now considered as a function of the independent variables $r = u_1$, $s = u_2$, and $t = \varphi$. But since u_1 and u_2 satisfy (20), the coefficients of $\frac{\partial u}{\partial r}$ and $\frac{\partial u}{\partial s}$ vanish, and since φ does not satisfy (20), Equation (20) is thus equivalent to the equation $\frac{\partial u}{\partial t} = 0$. Hence the most general solution of (20) is $u = F(r,s) = F(u_1,u_2)$, where F is an arbitrary differentiable function.

Thus finally, if $\frac{\partial u}{\partial z} \neq 0$, the most general solution of (15) is u = c or $F(u_1, u_2) = c$, where c can be incorporated into F, and hence can be replaced by zero, as was to be shown.

As a first example, we consider Equation (1),

$$2\frac{\partial z}{\partial x} - \frac{\partial z}{\partial y} = 0, \qquad (29)$$

and deduce the solution (2) by the method derived above. In this case (28) becomes

$$\frac{dx}{2} = \frac{dy}{-1} = \frac{dz}{0},$$
$$dx + 2 dy = 0, \qquad dz = 0.$$

or

The solutions of these equations are

$$x+2y=c_1, \qquad z=c_2,$$

and hence, with $u_1 = x + 2y$, $u_2 = z$, Equation (27) gives the general solution of (29) in the alternative forms

$$F(x + 2y, z) = 0$$
 or $z = f(x + 2y)$, (30)

in accordance with (2). In this case the surfaces $u_1 = c_1$ and $u_2 = c_2$ are planes, and the characteristic curves are the straight lines parallel to the xy plane, given by the intersections of the planes $x + 2y = c_1$ with the planes $z = c_2$. The direction cosines of these lines are proportional to the coefficients P,Q,Rin (29), that is, to the values (2, -1, 0). In this case the characteristic "curves" coincide with their tangents. Any surface built up from straight lines of the type noted is of the form (30), for *some* choice of f or F, and is an integral surface of (29). It should be noticed that any solution of (29) is constant along any line x + 2y = constant in the xy plane.

More generally, we verify readily that the general solution of the equation

$$a\frac{\partial z}{\partial x} + b\frac{\partial z}{\partial y} = 0, \qquad (31)$$

where a and b are constants, includes all cylindrical surfaces with elements parallel to the direction (a,b,0), and is of the form

$$z = f(bx - ay). \tag{32}$$

It should be noticed that this solution can also be expressed in other related forms such as

$$z = F(ay - bx),$$
 $z = g\left(\frac{y}{b} - \frac{x}{a}\right),$ $z = G\left(y - \frac{b}{a}x\right),$

and so on, if exceptions when a or b vanish are taken into account.

As a second example, for the equation

$$x\frac{\partial z}{\partial x} + y\frac{\partial z}{\partial y} = z$$
(33)

the associated equations become

$$\frac{dx}{x} = \frac{dy}{y} = \frac{dz}{z},$$

for two of which solutions are obtained in the form

$$\frac{y}{x} = c_1, \qquad \frac{z}{x} = c_2.$$

The general solution of (33) can then be written in the form

$$F\left(\frac{y}{x},\frac{z}{x}\right) = 0 \quad \text{or} \quad z = x f\left(\frac{y}{x}\right)$$
 (34)

or in several other equivalent forms, such as

$$z = y g\left(\frac{y}{x}\right), \qquad z = x F\left(\frac{x}{y}\right), \qquad z = y G\left(\frac{x}{y}\right).$$

Since the characteristic curves are straight lines with the direction ratios (x, y, z), it follows that any surface (34) contains the straight lines from the origin to points on the surface, and hence is a *conical surface* with vertex at the origin.

8.3. Special devices. Initial conditions. Although the geometric interpretation of the general solution of an equation of type (26) usually is readily obtained, the determination of explicit solutions of two of the associated equations (28) may present difficulties. In the *linear* case,

$$P(x,y)\frac{\partial z}{\partial x} + Q(x,y)\frac{\partial z}{\partial y} = R_1(x,y) z + R_2(x,y), \qquad (35)$$

since P and Q are independent of z, the equation $\frac{dy}{dx} = \frac{Q}{P}$ involves only x and y. If it can be solved in the form $u_1(x,y) = c_1$, then this result may be used to express, say, y in terms of x and c_1 . Then the equation

$$\frac{dz}{dx} = \frac{R_1 z + R_2}{P}$$

becomes a linear ordinary differential equation in z. The solution of this equation can be written in the form

$$u_2 = z \alpha_1(x,c_1) + \alpha_2(x,c_1) = c_2.$$

If c_1 is then replaced by its equivalent in terms of x and y, the result is of the general form

$$u_2 \equiv z \beta_1(x,y) + \beta_2(x,y) = c_2.$$

The general solution of the partial differential equation then becomes $u_2 = f(u_1)$ or

$$z=\frac{1}{\beta_1(x,y)}f[u_1(x,y)]-\frac{\beta_2(x,y)}{\beta_1(x,y)}.$$

Thus, the general solution of a linear equation of first order can be put in the form

$$z = s_1(x,y) f[s_2(x,y)] + s_3(x,y),$$
(36)

where s_1 , s_2 , and s_3 are specific functions and f is an arbitrary function.

For example, in the case of the equation

$$y\frac{\partial z}{\partial x} + x\frac{\partial z}{\partial y} = z - 1, \qquad (37)$$

Equations (28) become

$$\frac{dx}{y} = \frac{dy}{x} = \frac{dz}{z-1} \, .$$

Equality of the first two members gives

$$y \, dy - x \, dx = 0, \qquad y^2 - x^2 = c_1$$

Equality of the first and third members then gives

$$\frac{dz}{z-1} = \frac{dx}{\sqrt{x^2 + c_1}}; \qquad \frac{z-1}{x + \sqrt{x^2 + c_1}} = c_2 \quad \text{or} \quad \frac{z-1}{x+y} = c_2.$$

Thus, with

$$u_1 = y^2 - x^2$$
, $u_2 = \frac{z - 1}{x + y}$

the general solution of (37) becomes $u_2 = f(u_1)$ or

$$z = (x + y)f(y^2 - x^2) + 1.$$
 (38)

In other cases, if one integrable equation can be obtained by suitably rearranging (28), a similar procedure can be followed. In particular, since the equal ratios in (28) are also equal to the ratio

$$\frac{k_1 \, dx + k_2 \, dy + k_3 \, dz}{k_1 P + k_2 Q + k_3 R},\tag{39}$$

where k_1 , k_2 , and k_3 are entirely arbitrary, any one of the given ratios can be replaced by a new ratio of this form. If k_1 , k_2 , and k_3 can be chosen in such a way that the denominator of the new ratio vanishes and the numerator is an
exact differential du_1 , then the corresponding vanishing of the numerator leads to the fact that one integral is $u_1 = c_1$.

As an example, we consider the equation

$$(mz - ny)\frac{\partial z}{\partial x} + (nx - lz)\frac{\partial z}{\partial y} = ly - mx.$$
(40)

The associated equations are

$$\frac{dx}{mz-ny}=\frac{dy}{nx-lz}=\frac{dz}{ly-mx},$$

and no pairs are immediately integrable. We next determine k_1 , k_2 , and k_3 so that the ratio (39) has a zero denominator and hence also a vanishing numerator, and so write

$$k_1(mz - ny) + k_2(nx - lz) + k_3(ly - mx) = 0.$$

If we require the coefficients of l, m, and n to vanish independently, we obtain

$$k_1 = x, \qquad k_2 = y, \qquad k_3 = z,$$

and if we require the coefficients of x, y, and z to vanish independently, we obtain

$$k_1 = l, \qquad k_2 = m, \qquad k_3 = n.$$

With these particular choices of the k's, we obtain the two additional equivalent ratios

$$\frac{x\,dx+y\,dy+z\,dz}{0}\,,\qquad \frac{l\,dx+m\,dy+n\,dz}{0}\,.$$

Since the numerators happen to be the exact differentials $\frac{1}{2}d(x^2 + y^2 + z^2)$ and d(lx + my + nz), their vanishing leads immediately to the integrals

$$u_1 \equiv x^2 + y^2 + z^2 = c_1, \quad u_2 \equiv lx + my + nz = c_2,$$

and hence the general solution of (40) can be written in the form

$$F(x^{2} + y^{2} + z^{2}, lx + my + nz) = 0.$$
 (41)

The characteristic curves in this case are the circles in space determined as the intersections of the spheres $u_1 = c_1$ and the planes $u_2 = c_2$.

In the case of an *ordinary* differential equation of first order, the arbitrary constant can, in general, be determined so that the integral curve passes through a specified *point* in the *xy* plane. In a similar way, the arbitrary function in the general solution of a first-order *partial* differential equation can, in general, be determined so that the integral surface includes a specified *curve* in *xyz* space. If the equation of the curve is given by the pair of equations

$$\varphi_1(x,y,z) = 0, \qquad \varphi_2(x,y,z) = 0,$$
 (42)

that is, as the intersection of the two surfaces $\varphi_1 = 0$ and $\varphi_2 = 0$, and if the solutions of two of the associated equations (28) are written in the form

$$u_1(x,y,z) = c_1, \quad u_2(x,y,z) = c_2,$$
 (43)

the determination of the function F in the general solution (27) is equivalent to the determination of a functional relationship between c_1 and c_2 in (43) such that (42) and (43) are compatible. Thus, if the elimination of x, y, and z from the four equations involved in (42) and (43) leads to an equation of the form $F(c_1, c_2) = 0$, the required solution is given by $F(u_1, u_2) = 0$.

For example, if we require the solution of (37) which passes through the curve

$$z = x^2 + y + 1, \quad y = 2x$$
 (44)

we must eliminate x, y, and z between these equations and the equations $u_1 = c_1, u_2 = c_2$,

$$y^2 - x^2 = c_1, \qquad \frac{z-1}{x+y} = c_2.$$

The result of the elimination is

$$c_2 = \frac{\sqrt{c_1}}{3\sqrt{3}} + \frac{2}{3},$$

and hence the desired solution is

$$z = \frac{x+y}{3} \left(\frac{\sqrt{y^2 - x^2}}{\sqrt{3}} + 2 \right) + 1.$$
 (45)

Alternatively, if we take x as the independent variable along the curve, we have, from (44), y = 2x, $z = (x + 1)^2$, and the introduction of these results directly into the general solution (38) gives

$$x^{2} + 2x = 3x f(3x^{2}), \quad f(3x^{2}) = \frac{x}{3} + \frac{2}{3}.$$

If we write $u = 3x^2$, this equation becomes

$$f(u) = \frac{\sqrt{u}}{3\sqrt{3}} + \frac{2}{3}$$

and so determines the function f in (38) in accordance with (45).

It should be noticed, however, that the prescribed curve cannot be taken as a characteristic curve, since, in general, infinitely many integral surfaces include a characteristic curve. Further, in general there exist exceptional curves through which no integral surface passes (see Section 8.7).

The procedures given above can be extended to the solution of analogous equations in which there are more than two independent variables. Thus, for example, if x, y, and t are independent variables, the general solution of the quasi-linear equation

$$P\frac{\partial z}{\partial x} + Q\frac{\partial z}{\partial y} + R\frac{\partial z}{\partial t} = S$$
(46)

is of the form

$$F(u_1, u_2, u_3) = 0 \tag{47}$$

where $u_1(x,y,z,t) = c_1$, $u_2(x,y,z,t) = c_2$, and $u_3(x,y,z,t) = c_3$ are independent solutions of three of the associated ordinary equations

$$\frac{dx}{P} = \frac{dy}{Q} = \frac{dt}{R} = \frac{dz}{S}.$$
(48)

8.4. Linear and quasi-linear equations of second order. The general quasi-linear equation of second order, involving two independent variables x and y, is of the form

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} + F = 0, \qquad (49)$$

where a, b, c, and F may depend upon x, y, z, $\frac{\partial z}{\partial x}$, and $\frac{\partial z}{\partial y}$. In particular, when a, b, and c are independent of z, $\frac{\partial z}{\partial x}$, and $\frac{\partial z}{\partial y}$, while F is a linear function of those quantities, the equation is *linear* in z. Such an equation thus is of the form

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} + d\frac{\partial z}{\partial x} + e\frac{\partial z}{\partial y} + fz = g,$$
 (50)

where the functions a, \ldots, g depend only upon x and y.

It happens that the terms involving second derivatives are of principal significance. In fact, a very important role is played by the sign of the discriminant $b^2 - 4ac$. From analogy with the terminology associated with conic sections, we say that (49) or (50) is of hyperbolic type when $b^2 > 4ac$, of elliptic type when $b^2 < 4ac$, and of parabolic type in the intermediate case, when $b^2 = 4ac$. The importance of this classification will be indicated in later sections with reference to linear equations, to which our attention will be restricted in most of what follows.

In the linear case, it is clear that, as in the case of *ordinary* differential equations, the most general solution of (50) consists of the sum of any *particular* solution of (50) and the most general solution of the equation obtained by replacing the right-hand member of (50) by zero. This second solution is usually called the *complementary solution* of (50).

It is seen also that any linear combination of two complementary solutions will also be a complementary solution. However, such a combination cannot be expected to be the most general complementary solution, since the general solution should involve arbitrary *functions*. It might be expected that the most general solution of (50) would be a generalization of the form (36),

$$z = s_1(x,y)f[s_2(x,y)] + s_3(x,y)g[s_4(x,y)] + s_5(x,y),$$
(51)

where s_1, \ldots, s_5 are specific functions and f and g are arbitrary functions. However, this condition exists only in special cases. That is, the most general complementary solution of (50) cannot always be written as a sum of terms involving arbitrary functions.*

In Section 8.5 we consider an important special class of linear equations in which the general solution is of simple form, and in Section 8.6 we illustrate other special types of linear equations.

8.5. The homogeneous linear equation of second order, with constant coefficients. We here consider equations of the very special form

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} = 0, \qquad (52)$$

where a, b, and c are constants. An equation of this general type, in which all terms contain derivatives of the same order, is called a homogeneous equation. To obtain the general solution of (52), we assume a solution of the form

$$z = f(y + mx), \tag{53}$$

where f is an arbitrary twice-differentiable function, and attempt to determine values of m for which (53) is a solution of (52). By differentiation, we obtain from (53) the expressions

$$\frac{\partial^2 z}{\partial x^2} = m^2 f''(y + mx), \quad \frac{\partial^2 z}{\partial x \partial y} = mf''(y + mx), \quad \frac{\partial^2 z}{\partial y^2} = f''(y + mx), \quad (54)$$

and the introduction of (54) into (52) gives the condition

$$am^2 + bm + c = 0 \tag{55}$$

which must be satisfied by m.

In general, (55) will determine two distinct values of m, say m_1 and m_2 . Then, because of the linearity of (52), it follows that any expression of the form

$$z = f(y + m_1 x) + g(y + m_2 x)$$
(56)

is a solution of (52). Further, if the new variables $s = y + m_1 x$ and $t = y + m_2 x$ are introduced, it is readily verified that (52) takes the form $\frac{\partial^2 z}{\partial s \partial t} = 0$, from which it follows that (56) is indeed the most general solution of (52).

* For an example establishing this assertion, see Problem 18.

For example, for the equation

$$\frac{\partial^2 z}{\partial x^2} - 3 \frac{\partial^2 z}{\partial x \partial y} + 2 \frac{\partial^2 z}{\partial y^2} = 0, \qquad (57)$$

the determinant equation becomes

$$m^2 - 3m + 2 = 0, \qquad m_1 = 1, m_2 = 2,$$

and the general solution is

$$z = f(x + y) + g(2x + y).$$
 (58)

If a = 0 and $b \neq 0$, only one solution of Equation (55) is obtained. However, in this case it is clear that the second term in the general solution is then an arbitrary function of x alone. If a = b = 0, the general solution is clearly z = f(x) + y g(x).

The remaining exceptional case is that in which (55) is a perfect square, so that the two roots are equal. The second term in the general solution can be found by a method similar to that used in analogous cases in Chapter 1. If m_1 and m_2 are distinct roots of (55), then the expression

$$\frac{h(y + m_2 x) - h(y + m_1 x)}{m_2 - m_1}$$

is clearly a solution of (52). As $m_2 \rightarrow m_1$, this solution approaches the limiting value

$$\left[\frac{\partial}{\partial m}h(y+mx)\right]_{m=m_1}=x h'(y+m_1x).$$

If we write g = h', the general solution in the case of equal roots can then be taken in the form

$$z = f(y + m_1 x) + x g(y + m_1 x).$$
(59)

Thus, for example, the general solution of

$$\frac{\partial^2 z}{\partial x^2} - 2 \frac{\partial^2 z}{\partial x \partial y} + \frac{\partial^2 z}{\partial y^2} = 0$$

is found to be

$$z = f(x + y) + x g(x + y)$$

It may be noticed that since we can write

$$x g(y + m_1 x) \equiv \frac{1}{m_1} \left[(y + m_1 x) g(y + m_1 x) - y g(y + m_1 x) \right]$$

if $m_1 \neq 0$, the second solution can equally well be taken in the form $yg(y + m_1x)$ unless $m_1 = 0$. It is seen that the solutions of (55) will both be real if $b^2 > 4ac$, so that (52) is *hyperbolic*, and will be conjugate complex if $b^2 < 4ac$, so that (52) is *elliptic*. The intermediate case, when $b^2 = 4ac$, so that the equation is of *parabolic* type, is that in which (55) is a perfect square.

Thus, Laplace's equation

$$\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = 0 \tag{60a}$$

is elliptic, with general solution of the form

$$z = f(x + iy) + g(x - iy),$$
 (60b)

the equation

$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} = 0 \tag{61a}$$

is hyperbolic, with general solution

$$z = f(x + y) + g(x - y),$$
 (61b)

and the equation

$$\frac{\partial^2 z}{\partial x^2} - 2 \frac{\partial^2 z}{\partial x \partial y} + \frac{\partial^2 z}{\partial y^2} = 0$$
 (62a)

is parabolic, with general solution

$$z = f(x + y) + x g(x + y).$$
 (62b)

A procedure completely analogous to that given above can be applied in obtaining the general solution of a homogeneous linear equation of any order n with constant coefficients. In such a case, the assumption of a solution of form (53) leads to an equation analogous to (55), but of nth degree in m. If n distinct roots exist, the general solution is of a form similar to (56) but involving n independent terms. If a root m_1 is repeated r times, the part of the solution corresponding to the r equal roots is readily shown to be of the form

$$f_1(y + m_1x) + x f_2(y + m_1x) + \cdots + x^{r-1} f_r(y + m_1x).$$
 (63)

Occasionally it is feasible to solve a problem, governed by a partial differential equation together with appropriate side conditions, by obtaining the general solution of the equation and determining the relevant arbitrary functions by imposing the side conditions, in analogy with the usual procedure relating to *ordinary* differential equations.

Perhaps the best-known situation of this type is that in which the solution $\varphi(x,t)$ of the "one-dimensional wave equation"

$$\frac{\partial^2 \varphi}{\partial t^2} = c^2 \frac{\partial^2 \varphi}{\partial x^2} \tag{64}$$

is required, for all real values of x and t, subject to the conditions

$$\varphi(x,0) = F(x), \qquad \varphi_t(x,0) = G(x),$$
 (65)

prescribed for all x when t = 0. Here c is a positive real constant.

If the general solution of (64) is written in the form

$$\varphi(x,t) = f(x+ct) + g(x-ct), \qquad (66)$$

the conditions (65) require that f and g satisfy the conditions

$$f(x) + g(x) = F(x),$$
 $c[f'(x) - g'(x)] = G(x),$

for all values of x. The elimination of g(x) yields

$$f'(x) = \frac{1}{2}F'(x) + \frac{1}{2c}G(x), \qquad f(x) = \frac{1}{2}F(x) + \frac{1}{2c}\int_0^x G(\xi)\,d\xi + C,$$

where C is an arbitrary constant. The substitution of this result and the corresponding expression for g(x) into (66) gives the desired solution

$$\varphi(x,t) = \frac{1}{2} \left[F(x+ct) + F(x-ct) \right] + \frac{1}{2c} \int_{x-ct}^{x+ct} G(\xi) \, d\xi. \tag{67}$$

This solution is associated with the name of *d'Alembert*. A physical interpretation of the problem and its solution is included in Section 9.12.

Analogous solutions of certain other problems governed by (64) are obtained in Problems 28 and 30.

8.6. Other linear equations. As a simple example of a nonhomogeneous linear equation of second order, we consider the equation

$$\frac{\partial^2 z}{\partial x^2} - z = 0, \tag{68}$$

where x and y are the independent variables. Since y is not involved explicitly, we may integrate (68) as though it were an ordinary equation, holding y constant in the process, and hence replacing the two arbitrary constants of integration by arbitrary functions of y. Thus the general solution of (68) is of the form

$$z = e^{x} f(y) + e^{-x} g(y).$$
 (69)

However, even though the coefficients in the general second-order linear equation (50) be constants, it is not possible in *all* such cases to obtain general solutions of similar forms. The following procedure is frequently useful.

From the analogy with the ordinary equations, we are led to expect that exponential solutions of the equation

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} + d\frac{\partial z}{\partial x} + e\frac{\partial z}{\partial y} + fz = 0$$
(70)

with constant coefficients, may be of importance. If we assume a solution of the form

$$z = A \ e^{\alpha x + \beta y},\tag{71}$$

where A, α , and β are unknown constants, the introduction of (71) into (70) gives the condition

$$a\alpha^2 + b\alpha\beta + c\beta^2 + d\alpha + e\beta + f = 0$$
(72)

to be satisfied by α and β . This equation will, in general, determine β as either of two functions of α , say

$$\beta = \varphi_1(\alpha), \qquad \beta = \varphi_2(\alpha).$$
 (73)

It then follows that any expression of either of the forms

$$A e^{\alpha x + \varphi_1(\alpha)y}, \qquad B e^{\alpha x + \varphi_2(\alpha)y}$$

will satisfy (70) for arbitrary values of A, B, and α , and hence will be a particular solution of (70). The same is true for any linear combination of such expressions or for suitably convergent infinite series of such solutions.

As will be seen, this procedure leads to a precise form of the general solution if (73) expresses β as linear functions of α , as is the case when (70) is homogeneous and in certain other special cases. The algebraic equation (72), associated with (70), represents a conic section in the (α,β) plane, and is a hyperbola, ellipse, or parabola, according as $b^2 - 4ac > 0$, < 0, or = 0. In the special cases just mentioned, the conic section degenerates into two straight lines.

As an application, we consider first the equation

$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} - \frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} = 0.$$
 (74)

The assumption (71) leads to the equation

$$\alpha^2 - \beta^2 - \alpha + \beta = (\alpha - \beta)(\alpha + \beta - 1) = 0.$$

Hence (71) is a solution of (74) if

$$\beta = \alpha$$
 or $\beta = -\alpha + 1$.

Thus, by superposition, any expression of the form

$$z = \sum_{\alpha} A(\alpha) e^{\alpha(x+y)} + \sum_{\alpha} B(\alpha) e^{y+\alpha(x-y)}$$
(75)

is also a solution, the notation \sum_{α} indicating that α may take on arbitrary values in the summation. But since the coefficients A and B are also arbitrary, the first term in (75) may be expected to represent an arbitrary function of x + y, and the second term to represent the product of e^{y} and an arbitrary

function of x - y. Thus we may expect that the general solution can be written in the form

$$z = f(x + y) + e^{y} g(x - y),$$
 (76)

where f and g are twice differentiable, and this can be verified by direct substitution.

This sort of simplification clearly is always possible when the left-hand member of (72) contains linear factors, the condition for which is

$$ae^2 + fb^2 + cd^2 = bde + 4acf.$$
 (77)

In the case of the two-dimensional Helmholtz equation,

$$\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} + k^2 z = 0, \qquad (78)$$

the assumption (71) leads to the equation

$$\alpha^2 + \beta^2 + k^2 = 0,$$

and hence

$$\beta = \pm i \sqrt{\alpha^2 + k^2}.$$

Any expression of the form

$$A(\alpha) e^{\alpha x + i\sqrt{\alpha^2 + k^2}y} + B(\alpha) e^{\alpha x - i\sqrt{\alpha^2 + k^2}y}$$

is then a solution of (78), for any choice of A, B, and α . If we write

$$A+B=C, \quad i(A-B)=D,$$

this expression takes the real form

$$e^{\alpha x}[C\cos\sqrt{\alpha^2+k^2}y+D\sin\sqrt{\alpha^2+k^2}y].$$

Any combination of expressions of this sort, as α , C, and D take on arbitrary values, will also satisfy (78). Finally, any expression of the form

$$z = \int_{-\infty}^{\infty} \left[C(\alpha) \cos \sqrt{\alpha^2 + k^2} \, y + D(\alpha) \sin \sqrt{\alpha^2 + k^2} \, y \right] e^{\alpha x} \, d\alpha, \quad (79)$$

for which the indicated integral is suitably convergent, will be a particular solution of (78). The functions $C(\alpha)$ and $D(\alpha)$ may, of course, be taken to be zero except in a finite range in α , say for $\alpha_1 \leq \alpha \leq \alpha_2$.

8.7. Characteristics of linear first-order equations. In Section 8.2 we have defined a characteristic curve of the first-order linear equation

$$P\frac{\partial z}{\partial x} + Q\frac{\partial z}{\partial y} = R_1 z + R_2, \qquad (80)$$

where P, Q, R_1 , and R_2 are functions of x and y only, as a curve in space whose tangent at any point has the direction ratios $(P,Q, R_1z + R_2)$. Any such curve is the intersection of two surfaces of the form

$$u_1(x,y) = c_1, \quad u_2(x,y,z) = c_2,$$
 (81)

where $u_1(x,y) = c_1$ is an integral of the ordinary equation

$$\frac{dx}{P} = \frac{dy}{Q} \tag{82}$$

and $u_2(x,y,z) = c_2$ is an independent integral of one of the associated equations

$$\frac{dx}{P} = \frac{dy}{Q} = \frac{dz}{R_1 z + R_2}.$$
(83)

Thus, any characteristic curve of (80) is the intersection of the cylinder

$$u_1(x,y) = c_1,$$
 (84)

with elements parallel to the z axis, and a second surface of the form $u_2(x,y,z) = c_2$. It follows that any characteristic curve can be specified by first choosing a particular cylinder of the family (84), and so satisfying (82), and then prescribing z on this cylinder in such a way that another independent equation obtained from (83) is satisfied. We shall find it convenient to speak of any cylinder obtained by specializing c_1 in (84) as a *characteristic cylinder*. Also, we refer to the intersection of this cylinder with the xy plane as a *characteristic base curve*. Such a curve is clearly the *projection* of a characteristic space curve onto the xy plane.

In the case of the equation

$$y\frac{\partial z}{\partial x} - x\frac{\partial z}{\partial y} = y, \tag{85}$$

two associated equations can be written in the form

$$\frac{dx}{y}=-\frac{dy}{x}, \qquad dz=dx,$$

with solutions

$$u_1 = x^2 + y^2 = c_1, \qquad u_2 = z - x = c_2.$$

Thus the characteristic curves are ellipses in space determined as the intersections of the right circular cylinders $x^2 + y^2 = c_1$ and the planes $z - x = c_2$. The cylinders $x^2 + y^2 = c_1$ are the characteristic cylinders, and the circles $x^2 + y^2 = c_1$ in the xy plane are the characteristic base curves, which are the projections of the characteristic ellipses onto the xy plane. Along this base curve we can express x and y in terms of a single variable λ by writing

$$x = \sqrt{c_1} \cos \lambda, \qquad y = \sqrt{c_1} \sin \lambda.$$
 (86a,b)

If along this curve we define z by an equation of the form

$$z = x + c_2 = \sqrt{c_1} \cos \lambda + c_2, \tag{86c}$$

the corresponding locus in space is a characteristic curve. Each characteristic cylinder is seen to include infinitely many characteristic curves.

We now reconsider, from a somewhat different point of view, the question discussed in Section 8.3, as to whether an equation of form (80) has as a particular solution an integral surface which includes an arbitrarily prescribed curve in space.

Along any curve in space, the coordinates x, y, and z can be considered as functions of a single variable, say λ . Thus the curve can be specified by three equations of the form

C:
$$x = x(\lambda), \quad y = y(\lambda), \quad z = z(\lambda).$$
 (87)

The variable λ may be taken as arc length along the curve, for example, or it may be identified with one of the coordinates x or y. The projection C_0 of this curve onto the xy plane is clearly given by

$$C_0: \qquad x = x(\lambda), \qquad y = y(\lambda)$$
 (88)

and z = 0. Thus (87) can be considered as specifying z along the curve C_0 in the xy plane in terms of a parameter λ which varies along C_0 .

Suppose then that z is prescribed along an arbitrary curve C_0 in the xy plane, in accordance with (87), and that $x(\lambda)$, $y(\lambda)$ have continuous first derivatives and $z(\lambda)$ continuous derivatives of all orders with respect to λ . We inquire whether (80) has a solution expressing z as a function of x and y everywhere in a region including C_0 in such a way that z takes on the prescribed values on C_0 . This query is clearly equivalent to asking whether (80) possesses an integral surface which includes the space curve (87).

Let us assume that such a solution does exist and that the solution z = f(x,y) can be expanded in a series of powers of x and y, about a point (x_0,y_0) on C_0 , of the form

$$z = f(x_0, y_0) + \frac{\partial f(x_0, y_0)}{\partial x}(x - x_0) + \frac{\partial f(x_0, y_0)}{\partial y}(y - y_0) + \cdots$$
$$= (z)_{(x_0, y_0)} + \left(\frac{\partial z}{\partial x}\right)_{(x_0, y_0)}(x - x_0) + \left(\frac{\partial z}{\partial y}\right)_{(x_0, y_0)}(y - y_0) + \cdots$$

Then the solution is determined in the neighborhood of a point on C_0 if we know all its partial derivatives at that point.

It may be recalled that in Chapter 3 we made use of such ideas to calculate the series solution of an ordinary differential equation of the form

$$\frac{dy}{dx}=F(x,y)$$

for which y is prescribed as y_0 when $x = x_0$. The differential equation then gives $\frac{dy}{dx}$ when $x = x_0$, and by successive differentiation all higher derivatives are generally obtainable. We then have the required solution

$$y(x) = y(x_0) + y'(x_0)(x - x_0) + \frac{y''(x_0)}{2}(x - x_0)^2 + \cdots$$

if suitable convergence is assumed.

On the curve C_0 , the coefficients P, Q, R_1 , and R_2 , as well as z itself, are now known functions of λ , and hence (80) gives one equation involving the values of the two first partial derivatives of z at points on C_0 ,

$$P(\lambda)\frac{\partial z}{\partial x} + Q(\lambda)\frac{\partial z}{\partial y} = R_1(\lambda) z(\lambda) + R_2(\lambda), \qquad (89)$$

where, for example, $P(\lambda)$ has been written for $P[x(\lambda), y(\lambda)]$. To obtain a second equation, we notice that since z is known on C_0 , its derivative $\frac{dz}{d\lambda}$ along C_0 is also known, and it must satisfy the equation

$$\frac{dz(\lambda)}{d\lambda} = \frac{\partial z}{\partial x}\frac{dx}{d\lambda} + \frac{\partial z}{\partial y}\frac{dy}{d\lambda}.$$

Thus a second equation complementing (89) is of the form

$$\frac{dx}{d\lambda}\frac{\partial z}{\partial x} + \frac{dy}{d\lambda}\frac{\partial z}{\partial y} = \frac{dz}{d\lambda}.$$
(90)

Equations (89) and (90) can be solved uniquely for the unknown quantities $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ everywhere on C_0 if and only if the determinant of their coefficients is never zero,

$$\begin{vmatrix} P(\lambda) & Q(\lambda) \\ \frac{dx}{d\lambda} & \frac{dy}{d\lambda} \end{vmatrix} = P(\lambda) \frac{dy}{d\lambda} - Q(\lambda) \frac{dx}{d\lambda} \neq 0.$$
(91)

But this condition is equivalent to the restriction

$$\frac{dx(\lambda)}{P(\lambda)} \neq \frac{dy(\lambda)}{Q(\lambda)},$$
(92)

and hence (91) requires that the curve C_0 along which z is prescribed never be tangent to a characteristic base curve. If (91) is satisfied at all points on C_0 , the values of $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ can then be calculated by (89) and (90) at all points on C_0 . To calculate the second-order partial derivatives of z on C_0 , we may differentiate (80) with respect to x, and so obtain the equation

$$P\frac{\partial^2 z}{\partial x^2} + Q\frac{\partial^2 z}{\partial x \partial y} = R_1\frac{\partial z}{\partial x} + \frac{\partial R_1}{\partial x}z + \frac{\partial R_2}{\partial x} - \frac{\partial P}{\partial x}\frac{\partial z}{\partial x} - \frac{\partial Q}{\partial x}\frac{\partial z}{\partial y}, \quad (93)$$

where all quantities except $\frac{\partial^2 z}{\partial x^2}$ and $\frac{\partial^2 z}{\partial y^2}$ are now known functions of λ on C_0 . A second equation involving these unknown quantities is obtained by differentiating the known function $\frac{\partial z}{\partial x}$ along C_0 , with respect to λ ,

$$\frac{dx}{d\lambda}\frac{\partial^2 z}{\partial x^2} + \frac{dy}{d\lambda}\frac{\partial^2 z}{\partial x \partial y} = \frac{d}{d\lambda}\left(\frac{\partial z}{\partial x}\right).$$
(94)

Equations (93) and (94) determine $\frac{\partial^2 z}{\partial x^2}$ and $\frac{\partial^2 z}{\partial x \partial y}$ uniquely in terms of known functions if (91) is true, and the remaining derivative $\frac{\partial^2 z}{\partial y^2}$ can then be calculated from the equation obtained by differentiating $\frac{\partial z}{\partial y}$ with respect to λ .

This process can be continued indefinitely, to determine the values of all partial derivatives of z on C_0 in such a case, and it follows that the function z is determined uniquely for values of x and y in a region about C_0 by its prescribed values on C_0 if C_0 is never tangent to a characteristic base curve. This assertion is equivalent to the statement that if C is never tangent to a characteristic cylinder of (80), there exists a unique integral surface of (80) which includes C.

If, however, (91) is violated for all λ on C_0 , so that C is on a characteristic cylinder, then (89) and (90) are *incompatible* unless they are *equivalent*, that is, unless

$$\frac{dx(\lambda)}{P(\lambda)} = \frac{dy(\lambda)}{Q(\lambda)} = \frac{dz(\lambda)}{R_2(\lambda) + zR_1(\lambda)}.$$
(95)

In this case one of the two equations (89) and (90) implies the other, and hence only one restriction on the two derivatives is present. Thus the partial derivatives $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ are not determined uniquely in this case but may be chosen in infinitely many ways. But this result is to be expected, since if (95) is satisfied, C is then a characteristic curve, which clearly does lie on infinitely many integral surfaces. Thus it follows that if C lies on a characteristic cylinder of (80), there is no integral surface of (80) which includes C unless C is a characteristic curve, in which case there exist infinitely many such surfaces. It follows also that if C₀ is a characteristic base curve, then z cannot be prescribed arbitrarily on it but must be taken in such a way that C is a characteristic curve. In such a case the value of z at a point near C₀ is not determined by the values of z on C₀. Thus, in the case of Equation (85), there exists a unique integral surface including any space curve whose projection onto the xy plane is not a circle with center at the origin. The only curves whose projections are of this type and which lie on integral surfaces are the ellipses which are characteristic curves, and these curves each lie on infinitely many integral surfaces.

When (91) is violated only for certain isolated values of λ , so that C_0 is *tangent* to a characteristic base curve at each corresponding point, the relevant solution z(x,y) may behave exceptionally at each such point of tangency.

We next show that if the equation (84) of the characteristic cylinders of (80) is known, the differential equation can be reduced to a simpler form by a suitable change in variables. For this purpose we take as a new independent variable the expression

$$\varphi = u_1(x, y), \tag{96}$$

where $u_1(x,y) = c_1$ is a solution of (82), and retain either x or y as the second independent variable. If $P \neq 0$, then x and u_1 are functionally independent.* If we take them to be the independent variables, there then follows

$$\begin{pmatrix} \frac{\partial z}{\partial x} \end{pmatrix}_{y} = \frac{\partial z(x,\varphi)}{\partial x} + \frac{\partial z(x,\varphi)}{\partial \varphi} \frac{\partial u_{1}}{\partial x} , \\ \begin{pmatrix} \frac{\partial z}{\partial y} \end{pmatrix}_{x} = \frac{\partial z(x,\varphi)}{\partial \varphi} \frac{\partial u_{1}}{\partial y} ,$$

and hence, now considering z as a function of x and φ , Equation (80) becomes

$$P\frac{\partial z}{\partial x} + \left(P\frac{\partial u_1}{\partial x} + Q\frac{\partial u_1}{\partial y}\right)\frac{\partial z}{\partial \varphi} = R_1 z + R_2, \qquad (97)$$

where P, Q, R_1 , and R_2 are now to be expressed as functions of x and φ . But since $u_1 = c_1$ is an integral of (82), the function u_1 satisfies the partial differential equation

$$P\frac{\partial u_1}{\partial x}+Q\frac{\partial u_1}{\partial y}=0.$$

Hence the coefficient of $\frac{\partial z}{\partial \varphi}$ in (97) vanishes, and the equation takes the form

$$P \frac{\partial z(x,\varphi)}{\partial x} = R_1 z + R_2. \tag{98}$$

Thus, in the case of Equation (85),

$$y\frac{\partial z}{\partial x} - x\frac{\partial z}{\partial y} = y, \qquad (99)$$

* If P = 0, Equation (80) is already in the desired form.

we write

$$\varphi = x^2 + y^2, \qquad y = \sqrt{\varphi - x^2}.$$

Then (98) shows that (85) can be put into the simpler form,

$$\sqrt{\varphi-x^2}\frac{\partial z(x,\varphi)}{\partial x}=\sqrt{\varphi-x^2}.$$

Since $\sqrt{\varphi - x^2}$ does not involve z, it may be canceled, and the equation becomes

$$\frac{\partial z(x,\varphi)}{\partial x} = 1, \tag{100}$$

with the obvious general solution

$$z = x + f(\varphi) \equiv x + f(x^2 + y^2).$$

This result is equivalent to the form $u_2 = f(u_1)$, with the notation following (85), as is to be expected.

8.8. Characteristics of linear second-order equations. We next investigate in a similar way the problem of determining solutions of linear second-order partial differential equations satisfying appropriate initial conditions. The most general linear second-order equation with right-hand member zero is of the form

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} + d\frac{\partial z}{\partial x} + e\frac{\partial z}{\partial y} + fz = 0, \quad (101)$$

where the coefficients may be functions of x and y. In the case of ordinary equations of second order, the initial conditions prescribe a point in the plane, through which the integral curve is to pass, and also prescribe the slope of the integral curve at that point. This is equivalent to prescribing the values of the dependent variable y and its derivative $\frac{dy}{dx}$ corresponding to a given value of the independent variable x. In the case of a partial differential equation, the analogous initial conditions prescribe a curve C in space which is to lie in the integral surface, and also prescribe the orientation of the tangent plane to the integral surface along that curve. These conditions are equivalent to conditions which prescribe the values of z and its two partial derivatives $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ along the projection C_0 of the curve C onto the xy plane. However, the values

400

of z, $\frac{\partial z}{\partial x}$, and $\frac{\partial z}{\partial y}$ cannot be prescribed in a completely independent way if z is to be differentiable along the curve C_0 , since if λ is a parameter specifying position along C_0 , we must have

$$\frac{dz}{d\lambda} = \frac{\partial z}{\partial x}\frac{dx}{d\lambda} + \frac{\partial z}{\partial y}\frac{dy}{d\lambda}.$$
 (102)

Thus, for example, if C_0 is the x axis and if z is prescribed as f(x) along C_0 , then the derivative $\frac{\partial z}{\partial x}$ along C_0 cannot also be prescribed but must be given as f'(x). However, the derivative $\frac{\partial z}{\partial y}$ normal to C_0 can be independently prescribed. This limitation is in accordance with (102), which here becomes

$$f'(x) = \frac{\partial z}{\partial x} \cdot 1 + \frac{\partial z}{\partial y} \cdot 0$$

if we identify λ with the distance x along C_0 .

A curve C in space, together with values of $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ prescribed along C in such a way that (102) is satisfied, is called a *strip*, and Equation (102) is

referred to as the strip condition. If it is recalled [see Equation (103), Chapter 6] that the tangent plane to a surface z = z(x,y) has as its normal a vector with direction ratios $\left(\frac{\partial z}{\partial x}, \frac{\partial z}{\partial y}, -1\right)$, we can think of the prescribed values of $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ at points along C as determining the normal direction to differential elements of surface area at these points on the required integral surface. The strip condition (102) requires that the elements join together in a regular way (Figure 8.1).



Figure 8.1

Suppose now that a curve C_0 in the xy plane is given by the equations

$$C_0: \qquad x = x(\lambda), \qquad y = y(\lambda), \qquad (103)$$

where $x(\lambda)$ and $y(\lambda)$ have continuous derivatives, and that z, $\frac{\partial z}{\partial x}$, and $\frac{\partial z}{\partial y}$ are prescribed as functions of λ having derivatives of all orders at all points of C_0 ,

in such a way that (102) is satisfied. For brevity, we introduce the conventional abbreviations

$$p = \frac{\partial z}{\partial x}, \quad q = \frac{\partial z}{\partial y}$$

$$r = \frac{\partial^2 z}{\partial x^2}, \quad s = \frac{\partial^2 z}{\partial x \partial y}, \quad t = \frac{\partial^2 z}{\partial y^2}$$
(104)

Then the differential equation (101) becomes

$$ar + bs + ct + dp + eq + fz = 0,$$
 (105)

and the strip condition takes the form

$$\frac{dz}{d\lambda} = p \frac{dx}{d\lambda} + q \frac{dy}{d\lambda}.$$
 (106)

The prescribed conditions along C_0 are then

$$z = z(\lambda), \quad p = p(\lambda), \quad q = q(\lambda),$$
 (107)

where $z(\lambda)$, $p(\lambda)$, and $q(\lambda)$ satisfy (106).

As in the preceding section, we now attempt to calculate the values of the higher derivatives of z along C_0 in terms of the prescribed values. If all such derivatives can be determined and if z can be expanded in a power series in x and y about points on C_0 , then an integral surface satisfying the prescribed conditions is determined for values of x and y in a region including C_0 , and the values of z at points near C_0 are determined in terms of the known values on C_0 by use of this series.

The first problem, then, is to attempt to determine the values of the second derivatives r, s, and t for points on C_0 . One condition involving these unknown quantities is given by (105),

$$ar + bs + ct = -(dp + eq + fz).$$
 (108)

If we notice that, along C_0 , there follows also

$$\frac{d}{d\lambda}\left(\frac{\partial z}{\partial x}\right) = \frac{\partial}{\partial x}\left(\frac{\partial z}{\partial x}\right)\frac{dx}{d\lambda} + \frac{\partial}{\partial y}\left(\frac{\partial z}{\partial x}\right)\frac{dy}{d\lambda},\\ \frac{d}{d\lambda}\left(\frac{\partial z}{\partial y}\right) = \frac{\partial}{\partial x}\left(\frac{\partial z}{\partial y}\right)\frac{dx}{dy} + \frac{\partial}{\partial y}\left(\frac{\partial z}{\partial y}\right)\frac{dy}{d\lambda},$$

we thus obtain two additional conditions involving r, s, and t, of the form

$$\frac{dx}{d\lambda}r + \frac{dy}{d\lambda}s = \frac{dp}{d\lambda},$$
(109)

$$\frac{dx}{d\lambda}s + \frac{dy}{d\lambda}t = \frac{dq}{d\lambda}.$$
 (110)

If Equations (108), (109), and (110) can be solved for s, then (109) and (110) will determine r and t in terms of s.* But these equations have a unique solution in r, s, and t if and only if the determinant of their coefficients is not zero,

$$\begin{vmatrix} a & b & c \\ \frac{dx}{d\lambda} & \frac{dy}{d\lambda} & 0 \\ 0 & \frac{dx}{d\lambda} & \frac{dy}{d\lambda} \end{vmatrix} = a \left(\frac{dy}{d\lambda}\right)^2 - b \frac{dx}{d\lambda} \frac{dy}{d\lambda} + c \left(\frac{dx}{d\lambda}\right)^2 \neq 0.$$
(111)

If (111) is satisfied for all points on C_0 , then (108), (109), and (110) determine r, s, and t everywhere along C_0 . If we proceed by a method similar to that used in the preceding section, we then find that all higher partial derivations of z also can be calculated at points on C_0 if only (111) is satisfied. In such a case we may say that the strip consisting of the curve C in space and the associated values of $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ along C, satisfying (102), is a proper strip, in the sense that there exists a unique integral surface of (101) which includes this strip.

Now suppose that at all points of C_0 the determinant in (111) is zero,

$$a\left(\frac{dy}{d\lambda}\right)^{2} - b\frac{dx}{d\lambda}\frac{dy}{d\lambda} + c\left(\frac{dx}{d\lambda}\right)^{2} = 0.$$
 (112)

Then (108), (109), and (110) cannot be solved for s unless the numerator determinant in the formal solution by Cramer's rule also vanishes,

a	-(dp + eq + fz)	С	
$\frac{dx}{d\lambda}$	$\frac{dp}{d\lambda}$	0	= 0,
0	$\frac{dq}{d\lambda}$	$\frac{dy}{d\lambda}$	

or

In this case two of the three equations imply the other one, and hence only two restrictions on the three unknown quantities r, s, and t are present. Hence, if (112) and (113) are *both* satisfied, we can determine *infinitely many values* of r, s, and t, satisfying the necessary conditions (108–110). Finally, if (112)

 $a\frac{dp}{d2}\frac{dy}{d2} + c\frac{dq}{d2}\frac{dx}{d2} + (dp + eq + fz)\frac{dx}{d2}\frac{dy}{d2} = 0.$

* Exceptional cases where $\frac{dx}{d\lambda}$ or $\frac{dy}{d\lambda}$ vanishes can be treated separately.

(113)

is satisfied but the left-hand member of (113) is not zero, no solution exists. Equation (113) is known as the equation of compatibility.

If (112) is satisfied at all points on C_0 , then at all such points the slope $\frac{dy}{dx}$ of the curve in the xy plane must satisfy the equation

$$a\left(\frac{dy}{dx}\right)^2 - b\frac{dy}{dx} + c = 0.$$
(114)

That is, we must have

$$\frac{dy}{dx} = \frac{b \pm \sqrt{b^2 - 4ac}}{2a},$$
 (115)

and hence the quantity $b^2 - 4ac$ must be nonnegative. Suppose that $b^2 - 4ac$ is positive and a is not zero. Then (115) determines two families of curves in the xy plane, say

$$\varphi(x,y) = c_1, \quad \psi(x,y) = c_2.$$
 (116)

That is, if (112) is satisfied, the curve C_0 must be a member of one of these families. In such a case no integral surface including the prescribed strip exists unless along the curve C (which projects into C_0) z and its partial derivatives satisfy (113). If, however, (113) is also satisfied, infinitely many integral surfaces including the prescribed strip exist, and the strip is called a *characteristic strip*. The curve C in space which bears the strip is called a *characteristic curve*, and its projection C_0 in the xy plane, which must be a member of one of the families of (116), is called a *characteristic base curve*. The Equations (116) determine two families of cylinders in space, with generators parallel to the z axis, which may be called the *characteristic cylinders*. Some writers refer to the projection C_0 itself as a "characteristic" of the differential equation.

If a = 0, Equation (112) can be solved for $\frac{dx}{dy}$ and two families of characteristic cylinders are again obtained unless also c = 0. If a = c = 0, the cylinders reduce to the planes $x = c_1$ and $y = c_2$. If $b^2 = 4ac$, the two families coincide.

From Equation (115) we see that an *elliptic linear equation of second order* has no real characteristic curves, since for such an equation the discriminant $b^2 - 4ac$ is negative. Thus an integral surface of an elliptic equation can always be determined so as to include a prescribed regular strip in space. Equivalently, if z and its first partial derivatives are prescribed in a sufficiently regular way along any curve in the xy plane, the solution of an elliptic equation which takes on these values is determined uniquely in the neighborhood of this curve.

The situation in the case of a hyperbolic or parabolic equation is, however, quite different. In these cases real characteristics exist, and if a strip is built up along a curve C which lies on a characteristic cylinder, there will in general be

no integral surface including this strip unless the strip is a characteristic strip. In this last case there are then infinitely many integral surfaces including the strip, and the solution in the neighborhood of the strip is not determined uniquely by the prescribed values on the strip.

Thus, for the hyperbolic equation

$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} = 0, \qquad (117)$$

Equation (114) becomes

$$\left(\frac{dy}{dx}\right)^2 - 1 = 0, \qquad \frac{dy}{dx} = \pm 1,$$

and hence the characteristic base curves in the xy plane are the straight lines

$$x + y = c_1, \quad x - y = c_2.$$
 (118)

Along any such line the values of z, $p = \frac{\partial z}{\partial x}$, and $q = \frac{\partial z}{\partial y}$ cannot be independently prescribed in an arbitrary way. The compatibility condition (113),

$$\frac{dp}{d\lambda}\frac{dy}{d\lambda} - \frac{dq}{d\lambda}\frac{dx}{d\lambda} = 0, \qquad (119a)$$

as well as the usual strip condition (102),

$$\frac{dz}{d\lambda} = p \frac{dx}{d\lambda} + q \frac{dy}{d\lambda}, \qquad (119b)$$

must be satisfied.

If we identify λ with arc length, say s, then along a line of the first family $x + y = c_1$ we have

$$\frac{dx}{ds} = -\frac{dy}{ds} = \frac{\sqrt{2}}{2}$$

and hence Equations (119a,b) lead to the requirements

$$p + q = \text{constant}, \quad p - q = \sqrt{2} \frac{dz}{ds}$$
 (along $x + y = c_1$). (120a)

Similarly, along a line of the second family $x - y = c_2$, we have the relations

$$\frac{dx}{ds}=\frac{dy}{ds}=\frac{\sqrt{2}}{2},$$

and hence (119a,b) give

$$p-q = \text{constant}, \quad p+q = \sqrt{2} \frac{dz}{ds} \quad (\text{along } x - y = c_2). \quad (120b)$$

A comparison of these conditions shows that the equation of compatibility requires that at all points of a line of one family, the derivative of z in the direction of the intersecting lines of the other family must be constant. Since here the two families of characteristic lines are perpendicular to each other, the derivative of z normal to a characteristic line must be constant along that line in this particular example.

The general solution of (117) is of the form

$$z = f(x + y) + g(x - y).$$
 (121)

We may notice that each function in (121) is constant along one of the characteristic lines. If we write (118) in the form $\varphi(x,y) = c_1$, $\psi(x,y) = c_2$, the general solution (121) becomes

$$z = f(\varphi) + g(\psi). \tag{122}$$

Also, if φ and ψ are taken as new independent variables, Equation (117) becomes

$$\frac{\partial^2 z}{\partial \varphi \, \partial \psi} = 0. \tag{123}$$

Although the general solution of any given equation of form (101) cannot necessarily be expressed in a form similar to (122), still it is not difficult to show that if the expressions φ and ψ of (116) are taken as new independent variables, in the case where (101) is of *hyperbolic* type, then (101) is transformed to the so-called *normal form*

$$\frac{\partial^2 z}{\partial \varphi \, \partial \psi} = D \, \frac{\partial z}{\partial \varphi} + E \, \frac{\partial z}{\partial \psi} + Fz, \qquad (124)$$

where D, E, and F are functions of φ and ψ .

By choosing suitable new *real* independent variables α and β , any *elliptic* equation of type (101) can be put in the normal form

$$\frac{\partial^2 z}{\partial \alpha^2} + \frac{\partial^2 z}{\partial \beta^2} = D \frac{\partial z}{\partial \alpha} + E \frac{\partial z}{\partial \beta} + Fz$$
(125)

and any *parabolic* equation of this type can be put in the form

$$\frac{\partial^2 z}{\partial \beta^2} = D \frac{\partial z}{\partial \alpha} + E \frac{\partial z}{\partial \beta} + Fz.$$
(126)

8.9. Singular curves on integral surfaces. Let S be a particular integral surface corresponding to the linear first-order equation

$$P\frac{\partial z}{\partial x} + Q\frac{\partial z}{\partial y} = R_1 z + R_2, \qquad (127)$$

where P, Q, R_1 , and R_2 are continuous functions of x and y. Then the existence of the partial derivatives $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ in (127) implies that z itself be continuous; that is, the surface S must be a continuous surface. Hence the equal members of (127) must both be continuous functions of x and y. This condition, however, does not exclude the possibility that the two terms on the left may be *each* discontinuous, so long as their *sum* is continuous. In particular, there may be a curve C on S along which the

surface possesses a "corner" or "edge," as is indicated in Figure 8.2. We denote by C_0 the projection of C onto the xy plane, and consider z, $\frac{\partial z}{\partial x}$, and $\frac{\partial z}{\partial y}$ as functions of x and y. Then, while z is an everywhere continuous

function of x and y, and while the derivative of z in the direction of C_0 is continuous, we suppose that the derivative of z normal to C_0 has a finite jump as the curve C_0 is crossed in the xy plane (Figure 8.2). Surfaces of this general nature are of frequent interest in the



Figure 8.2

study of physical phenomena where abrupt changes of some sort may occur. We now investigate more explicitly the possibility of their existence, as integral surfaces of an equation of type (127).

Let the curve C_0 be specified by the parametric equations

$$x = x(\lambda), \quad y = y(\lambda),$$
 (128)

where λ is a convenient variable indicating position on C_0 . Then λ may also be considered as representing position in the direction of C_0 along parallel curves in the immediate neighborhood of C_0 . In particular, the derivative of z in the direction of C_0 at points on such curves is given by

$$\frac{dz}{d\lambda} = \frac{\partial z}{\partial x}\frac{dx}{d\lambda} + \frac{\partial z}{\partial y}\frac{dy}{d\lambda} = p\frac{dx}{d\lambda} + q\frac{dy}{d\lambda}.$$
(129)

Now consider any point P on the curve C_0 . Since the left-hand member of (127) must be continuous across C_0 at this point, we must have

$$P\,\Delta p + Q\,\Delta q = 0, \tag{130}$$

where Δp and Δq are the values of the jumps associated with the partial derivatives $p = \frac{\partial z}{\partial x}$ and $q = \frac{\partial z}{\partial y}$ across C_0 at P. Since the derivative of z in the direction of C_0 is also assumed to be continuous across C_0 , the right-hand member of (129) must also be continuous, and we obtain a second equation

$$\frac{dx}{d\lambda}\Delta p + \frac{dy}{d\lambda}\Delta q = 0 \tag{131}$$

relating the jumps in p and q across C_0 at P. Equations (130) and (131) are compatible (with Δp and Δq not both zero) only if the coefficients of Δp and Δq are proportional, that is, if

$$\frac{dx}{P} = \frac{dy}{Q} \tag{132}$$

along C_0 . Thus C_0 must lie on a characteristic cylinder of (127), and hence, since we have assumed that C lies on an integral surface of (127), we conclude that the "singular curve" C must be a characteristic curve of (127). That is, corners of the type considered can exist on an integral surface of (127) only along characteristic curves.

In the case of the linear second-order equation

$$a \frac{\partial^2 z}{\partial x^2} + b \frac{\partial^2 z}{\partial x \partial y} + c \frac{\partial^2 z}{\partial y^2} = -\left(d \frac{\partial z}{\partial x} + e \frac{\partial z}{\partial y} + fz\right), \quad (133)$$

the existence of the second derivatives presupposes continuity of z and its first partial derivatives, and hence also continuity of the equal members of (133) if the coefficients are continuous. However, the possibility that the *separate terms* on the left may be individually discontinuous is not excluded. Let C be a curve in an integral surface S, with projection C_0 in the xy plane, and denote position along C_0 by the parameter λ . We investigate the possibility that z, p, and q be continuous on S, and that the derivatives of p and q in the direction of C_0 ,

$$\frac{dp}{d\lambda} = \frac{\partial p}{\partial x}\frac{dx}{d\lambda} + \frac{\partial p}{\partial y}\frac{dy}{d\lambda} = r\frac{dx}{d\lambda} + s\frac{dy}{d\lambda},$$
(134)

$$\frac{dq}{d\lambda} = \frac{\partial q}{\partial x}\frac{dx}{d\lambda} + \frac{\partial q}{\partial y}\frac{dy}{d\lambda} = s\frac{dx}{d\lambda} + t\frac{dy}{d\lambda},$$
(135)

be continuous across C_0 , but that the derivatives of p and q normal to C_0 be discontinuous across C_0 .

The continuity of the right member of (133) and of the left members of (134) and (135) implies continuity of their equivalents, and hence we obtain the three equations

$$\left. \begin{array}{l} a \ \Delta r + b \ \Delta s + c \ \Delta t = 0 \\ \frac{dx}{d\lambda} \Delta r + \frac{dy}{d\lambda} \Delta s = 0 \\ \frac{dx}{d\lambda} \Delta s + \frac{dy}{d\lambda} \Delta t = 0 \end{array} \right\}$$
(136)

relating the jumps in the three partial derivatives r, s, and t across C_0 . These equations are consistent only if the determinant of the coefficients vanishes. But this condition is the same as that of Equation (111), and hence the curve C_0 must lie on a characteristic cylinder of (133). Since C lies on an integral surface and projects into C_0 , it follows that C must be a characteristic curve of (133). In particular, since an equation of elliptic type has no real characteristic curves, an integral surface of such an equation cannot contain a "singular curve" of the type required.

Results of this type are of importance, for example, in the study of twodimensional *compressible* fluid flow around rigid bodies [see Equation (194), Section 6.20]. Here it is found that for low velocities the problem is governed by a partial differential equation of elliptic type, whereas if sufficiently high velocities are attained the governing equation becomes hyperbolic, and discontinuous phenomena may then be present.

8.10. Remarks on linear second-order initial-value problems. In this section we summarize briefly certain preceding results and present additional facts bearing on their significance.

For the general linear second-order equation of the form

$$a\frac{\partial^2 z}{\partial x^2} + b\frac{\partial^2 z}{\partial x \partial y} + c\frac{\partial^2 z}{\partial y^2} + d\frac{\partial z}{\partial x} + e\frac{\partial z}{\partial y} + fz = 0, \quad (137)$$

we define the characteristic base curves, in the xy plane, as those curves for which

$$a(dy)^{2} - b(dx)(dy) + c(dx)^{2} = 0.$$
(138)

In the *elliptic* case, for which $b^2 < 4ac$, no real characteristic base curves can exist. In the *hyperbolic* case, for which $b^2 > 4ac$, two distinct sets of such curves are obtained, whereas in the *parabolic* case, $b^2 = 4ac$, the two sets become coincident.

In the *elliptic* case, if along any curve c in the xy plane z and its derivative in the direction *normal* to that curve are prescribed as functions of position along c in such a way that these prescribed values are *regular* everywhere along c, then a solution to the resulting initial-value problem is determined for values of x and y in *some* neighborhood of c. However, here two additional facts are of importance. *First*, unless the prescribed values are regular along c(in particular, unless derivatives of all orders exist at all points), *no* solution to the defined problem can exist. *Second*, even though a solution exist, in many cases it will exist only in a restricted neighborhood of c and will not be valid over the whole xy plane.

In the hyperbolic and parabolic cases, if c is not a characteristic base curve, the initial-value problem again has a solution. However, here the prescribed initial values need not be regular. The higher derivatives of the prescribed values may have finite jumps at certain points on c. These jumps are then found to be propagated along those characteristic base curves which pass through the relevant points on the initial curve, and the solution obtained is valid throughout the xy plane. If c is a characteristic base curve, then the initial-value problem does *not* have a solution unless the initial values of z and its derivative normal to c satisfy a certain compatibility condition, in which case infinitely many solutions then exist.

Certain of these facts are illustrated in Problems 46 and 47 at the end of this chapter.

8.11. The characteristics of a particular quasi-linear problem. To illustrate the nature of characteristics in other types of problems, we here consider the simultaneous equations

$$\frac{\partial u}{\partial x} + \frac{\partial v}{\partial x}\cos 2v + \frac{\partial v}{\partial y}\sin 2v = 0$$

$$\frac{\partial u}{\partial y} + \frac{\partial v}{\partial x}\sin 2v - \frac{\partial v}{\partial y}\cos 2v = 0$$
(139a,b)

These equations, which are quasi-linear in u and v, are of basic importance in certain two-dimensional problems in the mathematical theory of plasticity. We suppose that u and v are prescribed at all points along a curve C_0 in the xy plane. If λ specifies position along C_0 , we may define C_0 by the equations

$$C_0: \qquad x = x(\lambda), \qquad y = y(\lambda).$$
 (140)

For points on C_0 the dependent functions u and v are then to be considered as given functions of λ .

We now ask whether there exists a curve C_0 , or a set of such curves, with the property that u and v are not uniquely determined for points near C_0 by their prescribed values along C_0 . It is assumed here that these prescribed values are continuously differentiable along C_0 .

Since u and v are known along C_0 , as functions of λ , their derivatives along C_0 are calculable, and must satisfy the equations

$$\frac{du(\lambda)}{d\lambda} = \frac{\partial u}{\partial x}\frac{dx}{d\lambda} + \frac{\partial u}{\partial y}\frac{dy}{d\lambda} \\
\frac{dv(\lambda)}{d\lambda} = \frac{\partial v}{\partial x}\frac{dx}{d\lambda} + \frac{\partial v}{\partial y}\frac{dy}{d\lambda}$$
(141a,b)

Equations (139a,b) and (141a,b), constituting four linear equations in the four first partial derivatives $\frac{\partial u}{\partial x}$, $\frac{\partial u}{\partial y}$, $\frac{\partial v}{\partial x}$, and $\frac{\partial v}{\partial y}$, may be rewritten in the form

$$\begin{array}{cccc} u_{x} & + v_{x}\cos 2v + v_{y}\sin 2v = 0 \\ u_{y} & + v_{x}\sin 2v - v_{y}\cos 2v = 0 \\ u_{x}x' + u_{y}y' & = u' \\ v_{x}x' & + v_{y}y' & = v' \end{array} \right\},$$
(142a-d)

where a prime denotes λ differentiation. A unique solution is assured *unless* the determinant of coefficients vanishes, that is, unless

$$\begin{vmatrix} 1 & 0 & \cos 2v & \sin 2v \\ 0 & 1 & \sin 2v & -\cos 2v \\ x' & y' & 0 & 0 \\ 0 & 0 & x' & y' \end{vmatrix} = 0.$$
(143)

By expansion, this equation can be written in the form

$$y'^{2}\sin 2v + 2x'y'\cos 2v - x'^{2}\sin 2v = 0,$$

from which we obtain the two alternatives

$$\begin{cases} y' = x' \tan v \\ y' = -x' \cot v \end{cases}.$$
 (144a,b)

Thus two systems of curves are determined in the plane, for one of which $\frac{dy}{dx} = \tan v$, and for the second $\frac{dy}{dx} = -\cot v$. The two systems are seen to be orthogonal, and the slope of each curve is seen to depend upon the value of the dependent function v at the point under consideration.

Along any such curve, Equations (142a–d) cannot have a unique solution. In particular, the set cannot be solved for u_x and hence no solution exists unless the numerator determinant of Cramer's rule also vanishes, that is, unless

$$\begin{vmatrix} 0 & 0 & \cos 2v & \sin 2v \\ 0 & 1 & \sin 2v & -\cos 2v \\ u' & y' & 0 & 0 \\ v' & 0 & x' & y' \end{vmatrix} = 0,$$

or, expanding, unless

$$u' = v' \left(\cos 2v - \frac{x'}{y'} \sin 2v \right).$$
 (145)

Along the curves (144a) we then obtain the requirement u' = -v', or

$$\frac{dy}{dx} = \tan v: \qquad u + v = \text{constant}; \qquad (146a)$$

and along the curves (144b) there follows u' = v', or

$$\frac{dy}{dx} = -\cot v: \qquad u - v = \text{constant.}$$
(146b)

If either (146a) or (146b) is satisfied, u_x may be taken *arbitrarily*, and it may be readily verified that Equations (142a–c) then determine u_y , v_x , and v_y in terms of u_x , u', and v'. Thus, in these cases *infinitely many* solutions exist, all corresponding to the same prescribed values of u and v along C_0 .

The curves in the xy plane for which (146a) or (146b) is true may be called the *characteristics* of the simultaneous equations (139a,b). We next show that the equations (139a,b) are simplified if we take as new *dependent variables* the combinations u + v and u - v suggested by (146a,b), say

$$\xi = u + v, \qquad \eta = u - v. \tag{147}$$

The variable ξ is then constant along the first set of characteristics, whereas η is constant along curves in the second set. Then since we also have

$$u = \frac{1}{2}(\xi + \eta), \quad v = \frac{1}{2}(\xi - \eta),$$
 (148)

Equations (139a,b) take the form

$$(\xi_x + \eta_x) + (\xi_x - \eta_x) \cos 2v + (\xi_y - \eta_y) \sin 2v = 0, (\xi_y + \eta_y) + (\xi_x - \eta_x) \sin 2v - (\xi_y - \eta_y) \cos 2v = 0,$$

or, after a rearrangement,

$$\begin{array}{c} (\eta_x - \eta_y \cot v) + \cot^2 v \left(\xi_x + \xi_y \tan v\right) = 0 \\ -(\eta_x - \eta_y \cot v) + \left(\xi_x + \xi_y \tan v\right) = 0 \end{array} \right\}.$$
(149)

These equations imply the simplified relations

$$\begin{cases} \xi_x + \xi_y \tan v = 0\\ \eta_x - \eta_y \cot v = 0 \end{cases}, \tag{150}$$

where $v = \frac{1}{2}(\xi - \eta)$. Equations (150) are still nonlinear, but they are more tractable than the equivalent equations (139) (see Problem 49), and form the basis for further treatment of certain problems in plasticity. Physically, the characteristics in this case are the so-called *shear lines* (lines of maximum shearing stress) in a plastic problem of plane strain for an incompressible material.*

REFERENCES

- 1. Bateman, H., Partial Differential Equations of Mathematical Physics, Cambridge University Press, New York, 1959.
- 2. Courant, R., and D. Hilbert, *Methods of Mathematical Physics*, Vol. 1, Interscience Publishers, Inc., New York, 1953.
- 3. Frank, P., and R. von Mises, Die Differential- und Integralgleichungen der Mechanik und Physik, Rosenberg, New York, 1943.
- 4. Hopf, L., Introduction to the Differential Equations of Physics, Dover Publications, Inc., New York, 1948.
- 5. Jeffreys, H., and B. S. Jeffreys, *Methods of Mathematical Physics*, 3rd ed., Cambridge University Press, London, 1956.

* See W. Prager, *Discontinuous Solutions in the Theory of Plasticity*, Courant Anniversary Volume, pp. 289–300, Interscience Publishers, Inc., New York, 1948.

- 6. Morse, P. M., and H. Feshbach, *Methods of Theoretical Physics*, 2 pts., McGraw-Hill Book Company, Inc., New York, 1953.
- 7. Sommerfeld, A., Partial Differential Equations in Physics, Academic Press, Inc., New York, 1949.
- 8. Webster, A. G., Partial Differential Equations of Mathematical Physics, 2nd ed., Dover Publications, Inc., New York, 1956.

PROBLEMS

Section 8.1

1. Find the differential equation of lowest order which possesses each of the following solutions, with f and g arbitrary functions:

(a)
$$z = (x - y) f(x + y)$$
,
(b) $z = f(ax + by) + g(cx + dy)$ (ad $-bc \neq 0$),
(c) $z = f(ax + by) + x g(ax + by)$.

2. (a) Obtain the partial differential equation of first order satisfied by $z = f(\psi)$, where ψ is a given function of x and y and f is an arbitrary function, in the form

$$\psi_y \frac{\partial z}{\partial x} - \psi_x \frac{\partial z}{\partial y} = 0.$$

(b) With the new independent variables s and t, where $t = \psi(x,y)$ and s is any independent function of x and y, show that, when z is considered as a function of s and t, the differential equation of part (a) takes the form

$$(s_x \psi_y - s_y \psi_x) \frac{\partial z}{\partial s} = 0.$$

Hence deduce that the most general solution of that equation is of the form $z = f(\psi)$, where f is arbitrary.

3. Noticing that $z = \psi$ is a solution of the equation considered in Problem 2(a), deduce that if one solution of the equation

$$P(x,y)\frac{\partial z}{\partial x} + Q(x,y)\frac{\partial z}{\partial y} = 0$$

is of the form $z = \psi(x, y)$, then the most general solution is $z = f(\psi)$, where f is arbitrary.

4. Obtain the partial differential equation of first order satisfied by $x = \varphi f(\psi)$, where φ and ψ are given functions of x and y and f is an arbitrary function, in the form

$$\varphi\left(\psi_{\boldsymbol{y}}\,\frac{\partial z}{\partial x}-\psi_{x}\,\frac{\partial z}{\partial y}\right)=(\varphi_{\boldsymbol{x}}\psi_{\boldsymbol{y}}-\varphi_{\boldsymbol{y}}\psi_{\boldsymbol{x}})z.$$

Section 8.2

5. Determine the general solution of each of the following equations (with a, b, and c constant), writing each solution in a form solved for z:

(a)
$$a \frac{\partial z}{\partial x} + b \frac{\partial z}{\partial y} = c$$
,
(b) $a \frac{\partial z}{\partial x} + b \frac{\partial z}{\partial y} = cz$,
(c) $y \frac{\partial z}{\partial x} - x \frac{\partial z}{\partial y} = 0$,
(d) $\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} + 2xz = 0$,
(e) $x \frac{\partial z}{\partial x} - y \frac{\partial z}{\partial y} = z$,
(f) $x^2 \frac{\partial z}{\partial x} + y^2 \frac{\partial z}{\partial y} = z^2$.

6. (a) Show that if a particular solution of the equation

$$a \frac{\partial z}{\partial x} + b \frac{\partial z}{\partial y} = f_1(x) + f_2(y),$$

where a and b are constant, is assumed in the form $z_p = \varphi_1(x) + \varphi_2(y)$, there follows

$$z_{p} = \frac{1}{a} \int^{x} f_{1}(x) \, dx + \frac{1}{b} \int^{y} f_{2}(y) \, dy \qquad (ab \neq 0),$$

so that the general solution is of the form

$$z=f(bx-ay)+z_p.$$

(b) Illustrate the results of part (a) in the case of the equation

$$\frac{\partial z}{\partial x}-2\frac{\partial z}{\partial y}=2x-e^{y}+1.$$

7. Use the results of Problem 9 of Chapter 7, with an appropriate change in notation, to show that if the dependent and independent variables are interchanged in a pair of simultaneous *quasi-linear* differential equations in u and v, each of the form

 $Au_x + Bu_y + Cv_x + Dv_y = 0,$

where A, B, C, and D depend only on u and v, a pair of equivalent linear equations in x and y is obtained, each of the form

$$Ay_v - Bx_v - Cy_u + Dx_u = 0.$$

Section 8.3

8. (a) Show that the characteristic curves of the equation

$$\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} = 1$$

are straight lines parallel to the vector $\mathbf{i} + \mathbf{j} + \mathbf{k}$, and hence that the characteristic curve passing through a point (x_0, y_0, z_0) is specified by the equations

$$x - x_0 = y - y_0 = z - z_0.$$

(b) Determine the characteristic curve which passes through the point $(0, y_0, z_0)$ in the yz plane, and show that those characteristic curves which pass through points on the curve $z = y^2$ in the plane x = 0 are specified by the equations $y - x = y_0, z - x = y_0^2$. Thus deduce that the surface traced out by these curves is the parabolic cylinder

$$z = x + (y - x)^2.$$

(c) Verify directly that the surface $z = x + (y - z)^2$ is an integral surface of the equation $\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} = 1$ which includes the curve $z = y^2$ in the yz plane.

9. (a) Obtain the general solution of the differential equation of Problem 8(a), in the form

$$z = x + f(y - x).$$

(b) Determine the function f in such a way that this solution is consistent with the equations $z = y^2$, x = 0. Hence rederive the result of Problem 8(c).

10. (a) Show that the solution of the differential equation of Problem 8(a), for which $z = \varphi(x)$ along the line y = 2x in the xy plane, is of the form

$$z = 2x - y + \varphi(y - x).$$

(b) If z is prescribed as $\varphi(x)$ along the line y = x in the xy plane, show that no solution exists unless $\varphi(x)$ is prescribed in the form $\varphi(x) = x + k$, where k is a constant. In this last case, show that z = x + f(y - x) is a solution for any differentiable f such that f(0) = k, so that infinitely many solutions then exist.

(c) Show that the projections of the characteristic curves onto the xy plane are the lines y = x + c, and that if z is prescribed as $\varphi(x)$ along any such curve there is no solution unless $\varphi(x) = x + k$, in which case infinitely many solutions exist.

11. Find the solution of the equation $\frac{\partial z}{\partial x} = \frac{\partial z}{\partial y}$ for which $z = (t + 1)^4$ when $x = t^2 + 1$ and y = 2t.

12. Determine the general solution of each of the following equations, writing each solution in a form solved for z:

(a)
$$(x + y)\left(\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y}\right) = z - 1$$
, (b) $\frac{\partial z}{\partial x} = xy$,
(c) $xz \frac{\partial z}{\partial x} + yz \frac{\partial z}{\partial y} = xy$.

13. (a-f) Determine the integral surface of each equation of Problem 5 which includes the straight line x = y = z, if such a surface exists.

14. (a-f) Determine the solution of each equation of Problem 5 for which $z = x^2$ along the straight line y = 2x in the xy plane, if such a solution exists.

Section 8.4

15. If z = w(x,y) satisfies Equation (50) and z = u(x,y) reduces the left-hand member of (50) to zero, prove that z = w + cu satisfies (50), where c is a constant.

16. If z = u(x,y) and z = v(x,y) each reduce the left-hand member of (50) to zero, prove that the same is true of $z = c_1u + c_2v$, where c_1 and c_2 are constants.

17. (a) Show that the equation

$$\frac{\partial^2 z}{\partial x^2} + 2x \frac{\partial^2 z}{\partial x \partial y} + (1 - y^2) \frac{\partial^2 z}{\partial y^2} = 0$$

is elliptic for values of x and y in the region $x^2 + y^2 < 1$, parabolic on the boundary, and hyperbolic outside the region.

(b) Show that the quasi-linear equation

$$x \frac{\partial^2 z}{\partial x^2} + z \frac{\partial^2 z}{\partial x \partial y} + y \frac{\partial^2 z}{\partial y^2} = 0$$

is elliptic when $z^2 < 4xy$, parabolic when $z^2 = 4xy$, and hyperbolic otherwise. (Notice that the nature of a quasi-linear equation thus may depend not only upon position in the xy plane but also upon the solution z, which in turn depends upon the boundary conditions.)

18. (a) Show that the assumption that the equation

$$\frac{\partial^2 z}{\partial x^2} = \frac{\partial z}{\partial y}$$

possesses a solution of the form $z = \varphi(x,y) f[\psi(x,y)]$, where φ and ψ are specific functions and f is arbitrary, leads to the requirement

$$\varphi \psi_x^2 f''(\psi) + (2\varphi_x \psi_x + \varphi \psi_{xx} - \varphi \psi_y) f'(\psi) + (\varphi_{xx} - \varphi_y) f(\psi) = 0,$$

for arbitrary f, and hence to the three requirements

$$\varphi \psi_x^2 = 0, \qquad 2\varphi_x \psi_x + \varphi \psi_{xx} - \varphi \psi_y = 0, \qquad \varphi_{xx} - \varphi_y = 0.$$

(b) Show that these conditions imply that either $\varphi \equiv 0$ or $\psi = \text{constant}$, so that there exists no solution of the assumed form which actually depends upon an arbitrary function f.

Section 8.5

19. Obtain the general solution of each of the following equations:

(a)
$$\frac{\partial^2 z}{\partial x^2} - 2 \frac{\partial^2 z}{\partial x \partial y} - 3 \frac{\partial^2 z}{\partial y^2} = 0,$$

(b) $\frac{\partial^2 \varphi}{\partial x^2} - 2 \frac{\partial^2 f}{\partial x \partial y} + 2 \frac{\partial^2 \varphi}{\partial y^2} = 0,$
(c) $\frac{\partial^2 \varphi}{\partial x^2} - \frac{\partial^2 \varphi}{\partial x \partial y} = 0,$
(d) $\frac{\partial^2}{\partial r^2} (r^2 w) = \frac{r^2}{c^2} \frac{\partial^2 w}{\partial t^2},$

Problems

(e)
$$(U^2 - V^2) \frac{\partial^2 \varphi}{\partial x^2} + 2U \frac{\partial^2 \varphi}{\partial x \partial t} + \frac{\partial^2 \varphi}{\partial t^2} = 0,$$

(f) $\frac{\partial^4 \varphi}{\partial x^4} - \frac{\partial^4 \varphi}{\partial y^4} = 0,$
(g) $\frac{\partial^4 z}{\partial x^4} - 2 \frac{\partial^4 z}{\partial x^2 \partial y^2} + \frac{\partial^4 z}{\partial y^4} = 0,$
(h) $\nabla^4 \varphi = \frac{\partial^4 \varphi}{\partial x^4} + 2 \frac{\partial^4 \varphi}{\partial x^2 \partial y^2} + \frac{\partial^4 \varphi}{\partial y^4} = 0.$

20. Obtain the general solution of the simultaneous equations

$$\frac{\partial u}{\partial x} = -c^2 \frac{\partial v}{\partial y}, \quad \frac{\partial v}{\partial x} = -k^2 \frac{\partial u}{\partial y}$$

in the form

$$u = f\left(x + \frac{y}{kc}\right) + g\left(x - \frac{y}{kc}\right),$$
$$v = \frac{k}{c} \left[-f\left(x + \frac{y}{kc}\right) + g\left(x - \frac{y}{kc}\right)\right].$$

(First eliminate v and solve the resultant equation for u; then use the original equations to determine v. Notice, for example, that

$$\frac{\partial f}{\partial x} = kc \, \frac{\partial f}{\partial y} \, .$$

An arbitrary constant so introduced can be absorbed into the definitions of f and g.)

21. (a) Prove that the general solution of Laplace's equation, in the form

$$\frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} = 0,$$

can be written in the form

$$\varphi(x,y)=f(x+iy)+g(x-iy),$$

where f and g are twice-differentiable functions of the complex conjugate arguments x + iy and x - iy.

(b) Deduce that the real and imaginary parts of both f(x + iy) and g(x - iy) satisfy Laplace's equation.

(c) By noticing that the real and imaginary parts of g(x - iy) are respectively the real part and the negative of the imaginary part of the conjugate function $f(x + iy) \equiv \bar{g}(x + iy)$, deduce that any solution of Laplace's equation is the real or imaginary part of a twice-differentiable function of the complex variable x + iy, and conversely.

22. (a) Obtain as the real and imaginary parts of the function $(x + iy)^n$, for n = 0, 1, 2, 3, and 4, the functions 1, x, y, $x^2 - y^2$, 2xy, $x^2 - 3xy^2$, $3x^2y - y^3$, $x^4 - 6x^2y^2 + y^4$, and $4x^3y - 4xy^3$, and verify directly that they each satisfy Laplace's equation.

(b) Show that the functions $r^n \cos n\theta$ and $r^n \sin n\theta$, where *n* is integral, are the real and imaginary parts of $(x + iy)^n$ when x and y are expressed in polar coordinates, and verify directly that they each satisfy Laplace's equation. [See Equation (26) of Section 1.5 and Equation (162e) of Section 6.18.]

23. In each of the following cases, first obtain a particular solution of the equation as a function of one variable only, and then obtain the general solution by adding this solution to the general complementary solution:

(a)
$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} = x$$
, (b) $\frac{\partial^2 z}{\partial x^2} - 3 \frac{\partial^2 z}{\partial x \partial y} + 2 \frac{\partial^2 z}{\partial y^2} = \cos y$.

24. Suppose that the right-hand member of a linear partial differential equation is a homogeneous polynomial of degree k in x and y (each term being of degree k), whereas the left-hand member is homogeneous of order n in z (each term involving an nth derivative) with constant coefficients. Show that the assumption of a particular solution in the form of a homogeneous polynomial of degree n + k leads to k + 1 linear equations in n + k + 1 unknown coefficients. (It can be shown that this set always possesses a solution and, indeed, that a certain set of at least n of the coefficients can be assigned arbitrarily, in a convenient way.)

25. Use the procedure outlined in Problem 24 (and/or Problem 23) to obtain any particular solution of each of the following equations:

(a)
$$\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} = x^2 y + x y^2$$
, (b) $\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = x^2 + x y$,
(c) $\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} = x y + x$, (d) $\frac{\partial^2 z}{\partial x^2} - 2 \frac{\partial^2 z}{\partial x \partial y} + \frac{\partial^2 z}{\partial y^2} = x^2 + y$.

[In parts (c) and (d), consider the two terms on the right separately and use superposition.]

26. (a) If $am^2 + bm + c = a(m - m_1)(m - m_2)$, show that the equation

$$a \frac{\partial^2 z}{\partial x^2} + b \frac{\partial^2 z}{\partial x \partial y} + c \frac{\partial^2 z}{\partial y^2} = f(x,y),$$

where a, b, and c are constants, can be written in the operational form

$$a\left(\frac{\partial}{\partial x}-m_1\frac{\partial}{\partial y}\right)\left(\frac{\partial z}{\partial x}-m_2\frac{\partial z}{\partial y}\right)=f(x,y)$$

and deduce the general solution when f(x,y) = 0 by the methods of Section 8.2.

(b) Show that a particular solution of the general equation of part (a) can be obtained as a particular solution of the equation

$$\frac{\partial z}{\partial x} - m_2 \frac{\partial z}{\partial y} = z_1(x,y),$$

where z_1 is a particular solution of the equation

$$a\left(\frac{\partial z_1}{\partial x}-m_1\frac{\partial z_1}{\partial y}\right)=f(x,y).$$

27. Use Equation (67) to find the solution of the equation

$$\frac{\partial^2 \varphi}{\partial t^2} = c^2 \, \frac{\partial^2 \varphi}{\partial x^2}$$

for which $\varphi = x^2$ and $\frac{\partial \varphi}{\partial t} = \cos x$ for all values of x when t = 0.

28. It is required to find the solution of the equation

$$\frac{\partial^2 \varphi}{\partial t^2} = c^2 \frac{\partial^2 \varphi}{\partial x^2}$$

which satisfies the conditions

 $\varphi(0,t) = 0, \quad \varphi(L,t) = 0 \quad \text{(for all values of } t)$

along the boundary of the strip $0 \le x \le L$ in an xt plane, and the conditions

$$\varphi(x,0) = F(x), \quad \frac{\partial \varphi(x,0)}{\partial t} = G(x) \quad \text{(when } 0 < x < L)$$

along the line segment t = 0 in that strip.

(a) Taking the general solution of the equation in the form

$$\varphi = f(x + ct) + g(x - ct),$$

show that the conditions along the boundaries x = 0 and x = L lead to the relations

$$f(u) + g(-u) = 0 \qquad (\text{writing } u = ct),$$

$$f(u + 2L) = f(u) \qquad (\text{writing } u = ct - L),$$

for all values of u, while the conditions along the segment t = 0 for 0 < x < L become

$$\begin{cases} f(u) + g(u) = F(u) \\ c[f'(u) - g'(u)] = G(u) \end{cases}$$
 (when $0 < u < L$).

(b) Deduce that, if we write H'(u) = G(u), then the last two conditions of part (a) give

$$f(u) = \frac{1}{2}F(u) + \frac{1}{2c}H(u),$$
$$g(u) = \frac{1}{2}F(u) - \frac{1}{2c}H(u),$$

when 0 < u < L, and that also, for all values of u, f(u) must be a periodic function, of period 2L, and the condition

$$f(u) = -g(-u)$$

must be satisfied.

(c) Show that these conditions are all satisfied if we take

$$f(u) = \frac{1}{2} \mathscr{F}(u) + \frac{1}{2c} \mathscr{H}(u),$$
$$g(u) = \frac{1}{2} \mathscr{F}(u) - \frac{1}{2c} \mathscr{H}(u),$$

for all values of u, where $\mathscr{F}(u)$ and $\mathscr{G}(u)$ are odd period functions of period 2L, agreeing with F(u) and G(u), respectively, when 0 < u < L, and where $\mathscr{H}'(u) = \mathscr{G}(u)$. Thus, with this notation, obtain the required solution in the form

$$\varphi(x,t)=\frac{1}{2}\left[\mathscr{F}(x+ct)+\mathscr{F}(x-ct)\right]+\frac{1}{2c}\int_{x-ct}^{x+ct}\mathscr{G}(\xi)\,d\xi.$$

29. To illustrate the solution of Problem 28, notice that if $F(x) = \sin \pi x/L$ and G(x) = 0, there follows $\mathscr{F} = F$ and $\mathscr{G} = G = 0$, and obtain the solution in the form

$$\varphi = \frac{1}{2} \left[\sin \pi \frac{x + ct}{L} + \sin \pi \frac{x - ct}{L} \right] = \sin \frac{\pi x}{L} \cos \frac{c\pi t}{L}.$$

30. By using the method of Problem 28, obtain the solution of the problem

$$\frac{\partial^2 \varphi}{\partial t^2} = c^2 \frac{\partial^2 \varphi}{\partial x^2},$$

$$\varphi(x,0) = F(x), \quad \frac{\partial \varphi(x,0)}{\partial t} = G(x) \quad \text{(when } x > 0\text{)}$$

$$\varphi(0,t) = 0$$

in the half plane x > 0 in the form

$$\varphi(x,t)=\frac{1}{2}[\mathscr{F}(x+ct)+\mathscr{F}(x-ct)]+\frac{1}{2c}\int_{x-ct}^{x+ct}\mathscr{G}(\xi)\,d\xi,$$

where here $\mathcal{F}(u)$ and $\mathcal{G}(u)$ are odd functions of u agreeing with F(u) and G(u), respectively, when u > 0.

Section 8.6

31. (a) By assuming a solution of the Helmholtz equation

$$\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} + k^2 z = 0$$

in the form $z = e^{\alpha x + \beta y}$, and satisfying the resultant requirement $\alpha^2 + \beta^2 + k^2 = 0$ by writing $\alpha = ik \cos \varphi$ and $\beta = ik \sin \varphi$, where φ is an arbitrary real parameter, obtain particular solutions in the forms

 $\cos [k(x \cos \varphi + y \sin \varphi)], \quad \sin [k(x \cos \varphi + y \sin \varphi)].$

(b) Deduce that the real and imaginary parts of any expression of the form

$$U(x,y) = \int_{\varphi_1}^{\varphi_2} A(\varphi) e^{ik(x\cos\varphi + y\sin\varphi)} d\varphi$$

is a formal solution, for an arbitrarily chosen function $A(\varphi)$ and arbitrary constant limits φ_1 and φ_2 .

32. By changing to polar coordinates, deduce from the results of Problem 31 that the real and imaginary parts of any expression of the form

$$V(r,\theta) = \int_{\varphi_1}^{\varphi_2} A(\varphi) e^{ikr\cos(\varphi-\theta)} d\varphi$$

formally satisfy the Helmholtz equation in polar coordinates,

$$\frac{\partial^2 z}{\partial r^2} + \frac{1}{r} \frac{\partial z}{\partial r} + \frac{1}{r^2} \frac{\partial^2 z}{\partial \theta^2} + k^2 z = 0.$$

33. By assuming a solution of the equation

$$\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = k^2 \frac{\partial z}{\partial t}$$

in the form $z = e^{\alpha x + \beta y + \gamma t}$, and satisfying the resultant requirement by setting $\alpha = ik\rho \cos \varphi$, $\beta = ik\rho \sin \varphi$, $\gamma = -\rho^2$, deduce particular solutions in the forms

$$e^{-\rho^2 t} \cos [k\rho(x\cos \varphi + y\sin \varphi)], \quad e^{-\rho^2 t} \sin [k\rho(x\cos \varphi + y\sin \varphi)]$$

or as the real or imaginary parts of an expression of the form

$$U(x,y,t) = \iint_{\mathscr{R}} A(\rho,\varphi) e^{-\rho^2 t + ik\rho(x\cos\varphi + y\sin\varphi)} d\rho d\varphi.$$

34. By changing to polar coordinates in Problem 33, deduce particular solutions of the equation

$$\frac{\partial^2 z}{\partial r^2} + \frac{1}{r} \frac{\partial z}{\partial r} + \frac{1}{r^2} \frac{\partial^2 z}{\partial \theta^2} = k^2 \frac{\partial z}{\partial t}$$

in the forms

$$V_1(r,\theta,t) = \iint_{\mathscr{R}} A(\rho,\varphi) e^{-\rho^2 t} \cos \left[k\rho r \cos \left(\varphi - \theta\right)\right] d\rho \, d\varphi$$

and

$$V_2(r,\theta,t) = \iint_{\mathscr{R}} B(\rho,\varphi) \, e^{-\rho^2 t} \sin \left[k \rho r \cos \left(\varphi - \theta \right) \right] \, d\rho \, d\varphi.$$

Section 8.7

35. The function z(x,y) is required to satisfy the differential equation

$$\frac{\partial z}{\partial x} - \frac{\partial z}{\partial y} = z,$$

and to take on the value $z(\lambda) = \sin 2\lambda$ along the straight line C_0 specified by the equations $x = y = \lambda$ in the xy plane.

(a) By using equations corresponding to (89) and (90), show that

$$\frac{\partial z}{\partial x} = \cos 2\lambda + \frac{1}{2}\sin 2\lambda, \qquad \frac{\partial z}{\partial y} = \cos 2\lambda - \frac{1}{2}\sin 2\lambda$$

at points of C_0 .

(b) By using equations corresponding to (93) and (94), show that

$$\frac{\partial^2 z}{\partial x^2} = \cos 2\lambda - \frac{3}{4}\sin 2\lambda, \qquad \frac{\partial^2 z}{\partial x \partial y} = -\frac{5}{4}\sin 2\lambda$$

at points of C_0 .
(c) Use the result of differentiating $\frac{\partial z}{\partial y}$ along C_0 to show that $\frac{\partial^2 z}{\partial y^2} = -\cos 2\lambda - \frac{3}{4}\sin 2\lambda$

at points of C_0 .

(d) Check the result of part (c) by using the result of differentiating the governing differential equation with respect to y.

36. (a) Show that the Taylor expansion of the solution of the problem considered in Problem 35, in the neighborhood of the point (0,0), can be written in the form

$$z(x,y) = \frac{1}{1!}(x + y) + \frac{1}{2!}(x^2 - y^2) + \cdots,$$

in consequence of the results of Problem 35.

(b) Obtain the solution in the closed form

$$z = e^{(x-y)/2} \sin (x + y).$$

(c) Verify the correctness of the result of part (a) by obtaining the leading terms in the expansion of the closed form in a Taylor series about (0,0).

37. Verify directly that the procedure of Problem 35 fails if the curve C_0 is taken instead to be the characteristic base curve $x = -y = \lambda$.

38. (a) If Problem 35 is modified in such a way that C_0 is taken as the characteristic base curve $x = -y = \lambda$ and z is prescribed as $z(\lambda) = e^{\lambda}$ along C_0 , verify directly that $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ then can be determined in infinitely many ways.

(b) Show that the curve $x = \lambda$, $y = -\lambda$, $z = e^{\lambda}$ is a characteristic curve of the differential equation.

39. (a) If Problem 35 is modified in such a way that C_0 is taken as the circle $x = \cos \lambda$, $y = \sin \lambda$, and z is prescribed as $z(\lambda) = f(\lambda)$ along C_0 , show that $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ can be determined uniquely at all points along C_0 except the points

$$(\sqrt{2}/2,\sqrt{2}/2)$$
 and $(-\sqrt{2}/2,-\sqrt{2}/2)$.

(b) Show that C_0 is tangent to characteristic base curves at the exceptional points of part (a).

Section 8.8

40. The solution of Laplace's equation $\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = 0$ is required in the neighborhood of the line C_0 : $x = y = \lambda$, subject to the requirement that z = 0 along C_0 and that $\frac{\partial z}{\partial x} = p = \lambda$ along C_0 .

Problems

(a) Show that the strip condition requires that $\frac{\partial z}{\partial y} = q = -\lambda$ along C_0 .

(b) With the notation of Equation (104), verify that Equations (108-110) take the form r + t = 0, r + s = 1, s + t = -1, so that there must follow r = 1, s = 0, and t = -1 along C_0 .

(c) Deduce that, at the origin (0,0), there follows

$$z = 0, \quad \frac{\partial z}{\partial x} = 0, \quad \frac{\partial z}{\partial y} = 0,$$

 $\frac{\partial^2 z}{\partial x^2} = 1, \quad \frac{\partial^2 z}{\partial x \partial y} = 0, \text{ and } \frac{\partial^2 z}{\partial y^2} = -1,$

and hence show that the Taylor series expansion of z near the origin must be of the form

$$z(x,y) = 0 + \frac{1}{1!}(0x + 0y) + \frac{1}{2!}(x^2 + 2 \cdot 0xy - y^2) + \cdots = \frac{1}{2}(x^2 - y^2) + \cdots$$

(d) Verify the correctness of the result of part (c) by showing that the function $z = \frac{1}{2}(x^2 - y^2)$ in fact satisfies the conditions of the problem.

41. (a) Show that the characteristic base curves of the equation

$$\frac{\partial^2 z}{\partial x^2} + 3 \frac{\partial^2 z}{\partial x \partial y} + 2 \frac{\partial^2 z}{\partial y^2} - \frac{\partial z}{\partial x} = 0$$

are the lines $y = x + c_1$ and $y = 2x + c_2$.

(b) Show that, along the characteristic base curve C_0 : $y = x = \lambda$, the strip condition takes the form $\frac{dz}{d\lambda} = p + q$ and the equation of compatibility becomes

$$\frac{dp}{d\lambda}+2\frac{dq}{d\lambda}=p.$$

(c) Verify directly that, if $z(\lambda)$, $p(\lambda)$, and $q(\lambda)$ are prescribed along C_0 in such a way that these conditions are satisfied, then any one of the second derivatives, say s, can be chosen arbitrarily along C_0 , after which there follows accordingly $r(\lambda) = p'(\lambda) - s(\lambda)$ and $t(\lambda) = q'(\lambda) - s(\lambda)$.

42. Show that, if $\varphi = y - x$ and $\psi = y - 2x$ are taken as new independent variables in Problem 41, then there follows

$$rac{\partial}{\partial x}=-igg(rac{\partial}{\partial arphi}+2\,rac{\partial}{\partial \psi}igg),\qquad rac{\partial}{\partial y}=rac{\partial}{\partial arphi}+rac{\partial}{\partial \psi}$$

and the differential equation takes the normal form

$$rac{\partial^2 z}{\partial \varphi \ \partial \psi} = rac{\partial z}{\partial \varphi} + 2 rac{\partial z}{\partial \psi}.$$

Section 8.9

43. Show that the solution of the equation $\frac{\partial z}{\partial x} + \frac{\partial z}{\partial y} = 0$, for which z = |x| along the x axis, is of the form z = |y - x|. (Notice that the "corner" at the origin is propagated along the characteristic base curve y = x which passes through the origin.)

44. Show that, if z satisfies the equation

$$\frac{\partial^2 z}{\partial x^2} - \frac{\partial^2 z}{\partial y^2} = 0$$

and if $\frac{\partial^2 z}{\partial x^2}$ has a jump of one unit across the line $C_0: y = x$ in the xy plane, then $\frac{\partial^2 z}{\partial x \partial y}$ must have a jump of -1 and $\frac{\partial^2 z}{\partial y^2}$ a jump of +1 across C_0 , if $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ are to be continuously differentiable along C_0 .

45. If z satisfies Laplace's equation, $\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = 0$, show that the assumption that $\frac{\partial^2 z}{\partial x^2}$ has a jump across the line C_0 : y = x in the xy plane leads to a contradiction if $\frac{\partial z}{\partial x}$ and $\frac{\partial z}{\partial y}$ are to be continuously differentiable along C_0 .

Section 8.10

46. (a) By replacing *ct* by *ix*, where $i^2 = -1$, in Equation (67), show that formally we obtain a solution of Laplace's equation $\frac{\partial^2 z}{\partial x^2} + \frac{\partial^2 z}{\partial y^2} = 0$ for which z(x,0) = F(x) and $\frac{\partial z(x,0)}{\partial y} = G(x)$ in the form $z = \frac{1}{2} \left[F(x + iy) + F(x - iy) \right] + \frac{1}{2} \int_{-\infty}^{x+iy} G(\xi) d\xi$

$$z = \frac{1}{2} \left[F(x + iy) + F(x - iy) \right] + \frac{1}{2i} \int_{x-iy}^{x+iy} G(\xi) \, d\xi.$$

(b) Show that if we specify F(x) = x, $G(x) = e^x$, then an expression $z = x + e^x \sin y$

is obtained, and verify that this expression satisfies Laplace's equation everywhere and also satisfies the conditions imposed along the line y = 0.

(c) Notice that the formal solution is *meaningless* if, for example, we specify that F(x) be zero along the negative x axis and unity along the positive x axis. Show also that if

$$F(x) = \frac{1}{1+x^2}, \qquad G(x) = 0$$

the formal solution becomes

$$z = \frac{1 + x^2 - y^2}{(1 + x^2 - y^2)^2 + 4x^2y^2},$$

and that this expression is not defined at the points $(0, \pm 1)$. In particular, show that this formal solution becomes infinite as either of these points is approached along the y axis, so that the differential equation is not satisfied at this point.

[In general, the *initial-value* problem for Laplace's equation (or *any elliptic* equation) has no solution unless the prescribed values are sufficiently regular. Also, even though a solution be determined by the above method, it may be valid only *near the curve* along which the function and its normal derivative are prescribed. A deeper insight into the problem is afforded by the theory of analytic functions of a complex variable (Chapter 10).]

47. Writing the general solution of the equation

$$\frac{\partial^2 z}{\partial x \ \partial y} = 0$$

in the form z = f(x) + g(y), show that the initial conditions

$$z(x,0) = F(x),$$
 $p(x,0) \equiv \frac{\partial z(x,0)}{\partial x} = F'(x),$ $q(x,0) \equiv \frac{\partial z(x,0)}{\partial y} = G(x),$

imposed along the infinite line y = 0, lead to the conditions

$$f(x) + g(0) = F(x), \quad g'(0) = G(x).$$

Hence deduce that the problem has no solution unless G(x) is prescribed as a constant, say G(x) = C, in which case the problem has infinitely many solutions of the form

$$z = F(x) + g(y) - g(0),$$

where g(y) is any (differentiable) function of y for which g'(0) = C. (Notice that the line y = 0 in the xy plane is a characteristic base curve in this case.)

Section 8.11

48. Show by the methods of Section 8.11 that the simultaneous equations

$$\frac{\partial U}{\partial x} + \frac{\partial V}{\partial y} = 0,$$
$$2\frac{\partial U}{\partial x} + \left(\frac{\partial U}{\partial y} + \frac{\partial V}{\partial x}\right) \tan 2V = 0,$$

possess the same characteristics [Equations (144a,b)] as the problem of that section.

49. Use the results of Problem 7 to show that if the Jacobian $\frac{\partial(\xi,\eta)}{\partial(x,y)}$ is not zero, the dependent and independent variables in Equations (150) may be interchanged to give the equations

$$y_{\eta} - x_{\eta} \tan v = 0, \qquad y_{\xi} + x_{\xi} \cot v = 0,$$

where $v = \frac{1}{2}(\xi - \eta)$. [Notice that these equations are *linear* in x and y, whereas Equations (150) are *quasi-linear* in ξ and η , since v depends upon ξ and η .]

CHAPTER 9

Solutions of Partial Differential Equations of Mathematical Physics

9.1. Introduction. Many linear problems in mathematical physics involve the solution of an equation obtained by suitably specializing the form

$$\nabla^2 \varphi + f = \lambda \frac{\partial^2 \varphi}{\partial t^2} + \mu \frac{\partial \varphi}{\partial t}, \qquad (1)$$

where f is a specified function of position and λ and μ are certain specified physical constants. Here the operator ∇^2 is the Laplacian operator in the space of one, two, or three dimensions under consideration and is of the form

$$\nabla^2 = \frac{\partial^2}{\partial x^2} + \frac{\partial^2}{\partial y^2} + \frac{\partial^2}{\partial z^2}$$
(2)

in rectangular coordinates of three-space. The unknown function φ is then, in general, a function of the position coordinates (x,y,z) and the time coordinate t.

In particular, Laplace's equation,

$$\boldsymbol{\nabla}^{2}\boldsymbol{\varphi}=0, \tag{3}$$

is satisfied, for example, by the velocity potential in an ideal incompressible fluid without vorticity or continuously distributed sources and sinks, by gravitational potential in free space, electrostatic potential in the steady flow of electric currents in solid conductors, and by the steady-state temperature distribution in solids.

Also, Poisson's equation,

$$\nabla^2 \varphi + f = 0, \tag{4}$$

is satisfied, for example, by the velocity potential of an incompressible, irrotational, ideal fluid with continuously distributed sources or sinks, by steadystate temperature distributions due to distributed heat sources, and by a "stress function" involved in the elastic torsion of prismatic bars, with a suitably prescribed function f.

The so-called wave equation,

$$\nabla^2 \varphi = \frac{1}{c^2} \frac{\partial^2 \varphi}{\partial t^2} , \qquad (5)$$

arises in the study of propagation of waves with velocity c, independent of the wave length. In particular, it is satisfied by the components of the electric or magnetic vector in electromagnetic theory, by suitably chosen components of displacements in the theory of elastic vibrations, and by the velocity potential in the theory of sound (acoustics) for a perfect gas.

The equation of heat conduction,

$$\nabla^2 \varphi = \frac{1}{\alpha^2} \frac{\partial \varphi}{\partial t}, \qquad (6)$$

is satisfied, for example, by the temperature at a point of a homogeneous body and by the concentration of a diffused substance in the theory of diffusion, with a suitably prescribed constant α .

The telegraph equation,

$$\frac{\partial^2 \varphi}{\partial x^2} = \lambda \frac{\partial^2 \varphi}{\partial t^2} + \mu \frac{\partial \varphi}{\partial t}, \qquad (7)$$

which is a one-dimensional specialization of Equation (1), is satisfied by the potential in a telegraph cable, where $\lambda = LC$ and $\mu = RC$, if leakage is neglected (*L* is inductance, *C* capacity, and *R* resistance per unit length).

Differential equations of higher order, involving the operator ∇^2 , also are rather frequently encountered. In particular, the *bi-Laplacian equation* in two dimensions,

$$\nabla^4 \varphi = \nabla^2 \nabla^2 \varphi = \frac{\partial^4 \varphi}{\partial x^4} + 2 \frac{\partial^4 \varphi}{\partial x^2 \partial y^2} + \frac{\partial^4 \varphi}{\partial y^4} = 0, \qquad (8)$$

is involved in many two-dimensional problems of the theory of elasticity.

The solution of a given problem must satisfy the proper differential equation, together with suitably prescribed *boundary conditions* or *initial conditions* (if time is involved), the nature of these conditions depending upon the problem. In place of attempting to obtain the *most general* solution of the relevant differential equation, and specializing it so as to satisfy the given conditions, the usual procedure consists of determining a suitable set of *particular* solutions, each of which satisfies certain of the conditions, and then of attempting to combine a finite or infinite number of such solutions in such a way that the combination satisfies all the prescribed conditions. In those cases which are to be treated here, the particular solutions are obtained by a certain method of *separation of variables*. It may be mentioned that, although this method is restricted in a mathematical sense to a comparatively narrow range of differential equations, fortunately this range includes a very large number of those equations which arise in practice.

Before illustrating the solution of such problems, we first treat the formulation of the problem of heat flow in space.

9.2. Heat flow. We consider the flow of heat in a region in space such that the temperature T at a point (x, y, z) may depend upon the time t. For any region \mathcal{R} bounded by a closed surface S, the rate at which heat flows outward from \mathcal{R} through a surface element $d\sigma$ with unit outward normal **n** is given by

$$-dQ_1 = -K \frac{\partial T}{\partial n} \, d\sigma = -K(\nabla T) \cdot \mathbf{n} \, d\sigma,$$

where K is the thermal conductivity of the material. Thus the net rate of heat flow *into* \mathcal{R} is given by

$$Q_1 = + \oint_{\mathcal{S}} K(\nabla T) \cdot \mathbf{n} \, d\sigma. \tag{9}$$

However, the rate at which heat is absorbed by a volume element $d\tau$ is given by

$$dQ_2 = s\rho \frac{\partial T}{\partial t} d\tau,$$

where s is the specific heat and ρ the mass density. Hence, if there are no sources or sinks in \mathcal{R} , the rate of heat flow into \mathcal{R} is also given by

$$Q_2 = \iiint_{\mathscr{R}} s\rho \frac{\partial T}{\partial t} d\tau.$$
 (10)

Before equating Q_1 and Q_2 , we first transform Equation (9) to a volume integral by making use of the divergence theorem [Equation (116), Section 6.13], and so obtain*

$$Q_1 = \iiint_{\mathscr{H}} K \, \nabla^2 T \, d\tau. \tag{11}$$

The requirement $Q_1 = Q_2$ then becomes

$$\iiint_{\mathscr{R}} \left[K \nabla^2 T - s \rho \, \frac{\partial T}{\partial t} \right] d\tau = 0; \tag{12}$$

* If K is a function of position, $K \nabla^2 T$ must be replaced by $\nabla \cdot K \nabla T$ in Equations (11) and (12), and (13) is replaced by $\nabla \cdot K \nabla T = s_{\rho} \frac{\partial T}{\partial t}$.

but since this result must be true for any region \mathcal{R} not containing heat sources or sinks, the integrand must vanish. Hence T must satisfy the equation

$$\nabla^2 T = \frac{1}{\alpha^2} \frac{\partial T}{\partial t}, \qquad (13)$$

where we have written

$$\alpha^2 = \frac{K}{s\rho} \,. \tag{14}$$

The quantity α^2 is known as the thermal *diffusivity* of the material.

In particular, in the *steady-state* condition when the temperature at a point no longer varies with time, the temperature satisfies *Laplace's equation*,

$$\nabla^2 T = 0. \tag{15}$$

We now consider two basic problems in the theory of steady-state heat flow. First, it would be expected intuitively that in the steady state the temperatures at points inside a given region \mathcal{R} would be uniquely determinate if the temperature were prescribed along the bounding surface S. Second, one might prescribe the rate of (steady) heat flow outward per unit of area,

$$-\frac{dQ}{d\sigma}=-K\frac{\partial T}{\partial n},$$

at all points of the boundary S and require the temperature at internal points. Let T represent a function which satisfies Equation (15) and one of these boundary conditions. Then Equations (122) and (123) of Section 6.14 give the useful results

$$\iiint_{\mathscr{R}} (\nabla T)^2 d\tau = \oiint_{\mathscr{S}} T \frac{\partial T}{\partial n} d\sigma$$
 (16)

and

$$\oint \int_{S} \frac{\partial T}{\partial n} d\sigma = 0.$$
 (17)

Equation (17) is readily interpreted as requiring that the net flow through the closed boundary S must be zero, as must obviously be the case in steady-state flow without sources or sinks inside \mathscr{R} . Thus $\frac{\partial T}{\partial n}$ cannot be specified in a

perfectly arbitrary way on S, but its mean value on S must be zero.

Assume now that two solutions T_1 and T_2 exist, both satisfying Laplace's equation (15) and both taking on the same prescribed values on the boundary S. Then clearly the difference $T_2 - T_1$ also satisfies (15), and hence may be substituted for T in (16). But since $T_2 - T_1$ takes on the value zero at all points of S, the right-hand side of the resultant equation vanishes, and there follows

$$\iiint_{\mathscr{H}} \left[\nabla (T_2 - T_1) \right]^2 d\tau = 0.$$
 (18)

Now since the integrand is nonnegative, it must itself vanish everywhere in \mathcal{R} , and hence we have

$$\nabla (T_2 - T_1) = 0, \quad T_2 - T_1 = c = \text{constant.}$$
 (19)

Finally, since $T_2 - T_1$ vanishes on S, there follows c = 0 and consequently $T_2 = T_1$. That is, there can be only one solution of Laplace's equation valid in \mathcal{R} and taking on prescribed values on the boundary S. In a similar way we find that two solutions of Laplace's equation valid in \mathcal{R} and having the same specified value of the normal derivative on the boundary S can differ at most by a constant, noticing that here $T_2 - T_1$ is not necessarily zero on S.

The two problems considered are known respectively as the *Dirichlet* and *Neumann problems*. The solution to the Dirichlet problem, where the function itself is prescribed on the boundary, is *unique*, whereas the solution to the Neumann problem, where the normal derivative is prescribed on the boundary, is determined only to within an additive constant. These statements clearly apply to the solution of any problem governed by Laplace's equation, regardless of the nature of the unknown function.

Although other types of boundary-value problems involving Laplace's equation may occur, the two discussed here are of most frequent occurrence.

Next we consider the solution of several simple problems of the Dirichlet type. Although, to fix ideas, we choose to identify the quantity to be determined with steady-state temperature, the results may be equally well interpreted in terms of many other physical quantities which also satisfy Laplace's equation.

9.3. Steady-state temperature distribution in a rectangular plate. Suppose that the three edges x = 0, x = L, and y = 0 of a thin rectangular plate are maintained at zero temperature,

$$T(0,y) = T(L,y) = T(x,0) = 0,$$
 (20a,b,c)

and that the fourth edge y = H is maintained at a temperature distribution f(x),

$$T(x,H) = f(x), \tag{21}$$

until steady-state conditions are realized (Figure 9.1). The temperature distribution throughout the plate is required. Thus we must determine that solution of Laplace's equation in two dimensions,

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} = 0, \qquad (22)$$

which takes on the prescribed boundary values (20) and (21).

The method of *separation of variables* consists of seeking particular "product solutions" of (22) in the form

$$T_{p}(x,y) = X(x) Y(y),$$
 (23)

where X is a function of x alone and Y is a function of y alone. Introducing (23) into (22), there follows

$$\frac{d^2X}{dx^2}Y + X\frac{d^2Y}{dy^2} = 0,$$

or, separating the variables,

$$-\frac{1}{X}\frac{d^{2}X}{dx^{2}} = \frac{1}{Y}\frac{d^{2}Y}{dy^{2}}.$$
 (24)

Since, by hypothesis, the left-hand member of (24) is independent of y and the equivalent right-hand member is independent of x, it follows that both sides must be independent of both x and y, and

hence must be equal to a constant. If we call this arbitrary constant k^2 , there follows

$$\frac{d^{2}X}{dx^{2}} + k^{2}X = 0 \\ \frac{d^{2}Y}{dy^{2}} - k^{2}Y = 0$$
 (25a,b)

Thus we see that the product (23) will satisfy (22) if X and Y are solutions of (25a,b), regardless of the value of k.

Because of the linearity of (22), it follows that any linear combination of such solutions, corresponding to different values of k, will also satisfy (22).

We next notice that three of the boundary conditions are *homogeneous*. Thus, if *each* of the particular product solutions is required to satisfy (20a,b,c), any linear combination will also satisfy the same conditions. Equations (20a,b) will be satisfied if

$$X(0) = X(L) = 0,$$
 (26a,b)

whereas (20c) implies the condition

$$Y(0) = 0.$$
 (27)

Equations (25a) and (26) constitute a previously considered Sturm-Liouville problem for which the characteristic values are

$$k = k_n = \frac{n\pi}{L}$$
 (n = 1, 2, 3, ...), (28)

and the corresponding solutions (characteristic functions) are of the form

$$X = X_n = A_n \sin \frac{n\pi x}{L} \,. \tag{29}$$



Figure 9.1

Corresponding to (28), the solution of (25b) which satisfies (27) is of the form

$$Y = Y_n = B_n \sinh \frac{n\pi y}{L}.$$
 (30)

Thus it follows that any particular solution of the form

$$T_n = a_n \sin \frac{n \pi x}{L} \sinh \frac{n \pi y}{L}$$
 (n = 1, 2, 3, ...), (31)

where we have written $a_n = A_n B_n$, satisfies Equation (22) and the three boundary conditions (20a,b,c). The same is true for any series of the form

$$T = \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} \sinh \frac{n\pi y}{L}$$
(32)

if suitable convergence is assumed. It remains, then, to attempt to determine the coefficients a_n in (32) in such a way that the remaining condition (21) is satisfied, so that

$$f(x) = \sum_{n=1}^{\infty} \left(a_n \sinh \frac{n\pi H}{L} \right) \sin \frac{n\pi x}{L} \qquad (0 < x < L). \tag{33}$$

But, from the theory of Fourier sine series, the coefficients $a_n \sinh(n\pi H/L)$ in this series must be of the form

$$a_n \sinh \frac{n\pi H}{L} = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} dx,$$

and hence, writing $c_n = a_n \sinh(n\pi H/L)$, the required solution (32) takes the form

$$T(x,y) = \sum_{n=1}^{\infty} c_n \sin \frac{n\pi x}{L} \frac{\sinh \frac{n\pi y}{L}}{\sinh \frac{n\pi H}{L}}$$
(34)

$$c_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} \, dx, \qquad (35)$$

where

assuming appropriate convergence.

If f(x) is, in fact, representable by a convergent Fourier series of the form (33) in (0,L), it can be established that the series (34), subject to (35), truly converges to the solution of the stated problem inside the rectangle of definition. This situation exists, in particular, when f(x) is bounded and piecewise differentiable in (0,L).

It is clear that the solution of the more general problem where T is prescribed arbitrarily along all four edges can be obtained by superimposing four solutions analogous to the one obtained here, each corresponding to a problem in which zero temperatures are prescribed along three of the four edges, although more convenient alternative procedures frequently suggest themselves in such cases.

It is of importance to notice that permissible values of the "separation constant" k^2 were here determined by the characteristic-value problem arising from the presence of homogeneous boundary conditions along the *two* edges x = constant.

By virtue of (23) and (25), any expression of the form

$$T_{p} = (c_{1} \cos kx + c_{2} \sin kx)(c_{3} \cosh ky + c_{4} \sinh ky)$$
(36a)

is a particular solution of (22) for arbitrary values of k. It is readily verified that if the equal members of (24) were set equal to $-k^2$ rather than $+k^2$, the signs in (25a,b) would be reversed and particular solutions of the form

$$T_{p} = (c_{5} \cosh kx + c_{6} \sinh kx)(c_{7} \cos ky + c_{8} \sin ky)$$
(36b)

would be obtained. Finally, if k^2 is replaced by zero in (25a,b), the solutions of the resultant equations lead to the further particular product solutions

$$T_{p} = (c_{9}x + c_{10})(c_{11}y + c_{12}).$$
 (36c)

Clearly, exponential forms could be used in (36a,b) in place of the hyperbolic functions, if this procedure were desirable. However, only the product solutions listed in (31) satisfy the homogeneous boundary conditions (20).

The choice of the sign of the separation constant associated with (24) was motivated by the knowledge that solutions of type (36a) would lead to a Fourier series expansion in the x direction along the edge y = H where the nonhomogeneous condition is prescribed.

It was not obvious, *a priori*, that (22) would possess separable solutions of the form (23) or that a solution built up from such solutions could be made to satisfy the prescribed boundary conditions. In fact, only in special cases, in a mathematical sense, do partial differential equations possess product solutions. These cases are fortunately of frequent occurrence.

9.4. Steady-state temperature distribution in a circular annulus. Suppose that the temperature distributions along the inner and outer radii of a circular annulus are maintained as $f_1(\theta)$ and $f_2(\theta)$, respectively,

$$T(r_1, \theta) = f_1(\theta), \qquad T(r_2, \theta) = f_2(\theta),$$
 (37a,b)

until steady-state conditions are realized (Figure 9.2). The temperatures at internal points of the annulus are required.

If polar coordinates are used, Laplace's equation in the plane becomes (see Section 6.18)

$$\nabla^2 T = \frac{\partial^2 T}{\partial r^2} + \frac{1}{r} \frac{\partial T}{\partial r} + \frac{1}{r^2} \frac{\partial^2 T}{\partial \theta^2} = 0.$$
(38)

Assuming a particular product solution of the form

$$T_{p}(r,\theta) = R(r) \Theta(\theta), \qquad (39)$$

Equation (38) becomes

$$R''\Theta + \frac{1}{r}R'\Theta + \frac{1}{r^2}R\Theta'' = 0,$$

where a prime denotes differentiation with respect to the argument. By separating the variables, there follows

$$\frac{1}{R}(r^2R''+rR')=-\frac{\Theta''}{\Theta}=k^2,\quad (40)$$

where k^2 is the separation constant. This condition implies the two ordinary equations

$$r^2 R'' + r R' - k^2 R = 0,$$
 (41)

$$\Theta'' + k^2 \Theta = 0. \tag{42}$$

The sign of the separation constant was chosen in such a way that sines and cosines (rather than exponential functions) will be introduced in the θ direction where expan-

Figure 9.2

sions along the circles $r = r_1$ and $r = r_2$ presumably will be required.

Equation (41) is an equidimensional equation (see Section 1.6) with general solution

$$R = A_k r^k + B_k r^{-k} \qquad (k \neq 0) \\ R = A_0 + B_0 \log r \qquad (k = 0) \end{cases},$$
(43)

whereas (42) has the solution

$$\Theta = C_k \cos k\theta + D_k \sin k\theta \qquad (k \neq 0) \\ \Theta = C_0 + D_0 \theta \qquad (k = 0) \end{cases}.$$
(44)

Thus any expression of the form

$$T = a_0 + b_0 \log r + (c_0 + d_0 \log r) \theta + \sum_k [(a_k r^k + b_k r^{-k}) \cos k\theta + (c_k r^k + d_k r^{-k}) \sin k\theta], \quad (45)$$

where k takes on arbitrary nonzero values, is a solution of Equation (38).



In order that T be *single-valued* in the annulus, we must take

$$c_0 = d_0 = 0. (46)$$

Also, for the same reason, the trigonometric functions must possess a common period 2π . This requirement, in the present case, serves to determine the permissible values of the separation constant,

$$k = n$$
 $(n = 1, 2, 3, ...).$ (47)

Hence we are led to assume the solution of the present problem in the form

$$T = (a_0 + b_0 \log r) + \sum_{n=1}^{\infty} \left[(a_n r^n + b_n r^{-n}) \cos n\theta + (c_n r^n + d_n r^{-n}) \sin n\theta \right] \quad (r_1 \le r \le r_2).$$
(48)

The boundary conditions (37a,b) then take the form

$$f_{1}(\theta) = (a_{0} + b_{0} \log r_{1}) + \sum_{n=1}^{\infty} \left[(a_{n}r_{1}^{n} + b_{n}r_{1}^{-n}) \cos n\theta + (c_{n}r_{1}^{n} + d_{n}r_{1}^{-n}) \sin n\theta \right]$$

$$f_{2}(\theta) = (a_{0} + b_{0} \log r_{2}) + \sum_{n=1}^{\infty} \left[(a_{n}r_{2}^{n} + b_{n}r_{2}^{-n}) \cos n\theta + (c_{n}r_{2}^{n} + d_{n}r_{2}^{-n}) \sin n\theta \right]$$

$$(49)$$

so that, according to the theory of Fourier series,

$$a_{0} + b_{0} \log r_{1} = \frac{1}{2\pi} \int_{0}^{2\pi} f_{1}(\theta) d\theta \\a_{0} + b_{0} \log r_{2} = \frac{1}{2\pi} \int_{0}^{2\pi} f_{2}(\theta) d\theta \right\},$$
(50a,b)

$$a_{n}r_{1}^{n} + b_{n}r_{1}^{-n} = \frac{1}{\pi} \int_{0}^{2\pi} f_{1}(\theta) \cos n\theta \ d\theta$$

$$a_{n}r_{2}^{n} + b_{n}r_{2}^{-n} = \frac{1}{\pi} \int_{0}^{2\pi} f_{2}(\theta) \cos n\theta \ d\theta$$

$$(n = 1, 2, 3, ...), \quad (50c,d)$$

$$c_{n}r_{1}^{n} + d_{n}r_{1}^{-n} = \frac{1}{\pi} \int_{0}^{2\pi} f_{1}(\theta) \sin n\theta \, d\theta$$

$$c_{n}r_{2}^{n} + d_{n}r_{2}^{-n} = \frac{1}{\pi} \int_{0}^{2\pi} f_{2}(\theta) \sin n\theta \, d\theta$$

$$(n = 1, 2, 3, ...). \quad (50e,f)$$

These equations serve to determine in pairs the constants appearing in the required solution (48).

Two limiting cases are of particular interest. First, in the limiting case $r_1 = 0$, the region considered becomes *the interior of the circle* $r = r_2$. In order that the left-hand members of (50a,c,e) remain finite as $r_1 \rightarrow 0$ we must take

$$b_0 = b_n = d_n = 0$$
 (n = 1, 2, 3, ...). (51)

If we also write A_n for $a_n r_2^n$, C_n for $c_n r_2^n$, a for r_2 , and $f(\theta)$ for $f_2(\theta)$, we deduce that the solution of the problem

$$\begin{array}{c} \nabla^2 T = 0 \quad (r \leq a) \\ T(a,\theta) = f(\theta) \end{array} \right\}$$
(52)

is of the form

$$T = A_0 + \sum_{n=1}^{\infty} \left(\frac{r}{a}\right)^n (A_n \cos n\theta + C_n \sin n\theta) \qquad (r \le a) \tag{53}$$

where

$$A_{0} = \frac{1}{2\pi} \int_{0}^{2\pi} f(\theta) d\theta$$

$$A_{n} = \frac{1}{\pi} \int_{0}^{2\pi} f(\theta) \cos n\theta d\theta \qquad (n = 1, 2, 3, ...)$$

$$B_{n} = \frac{1}{\pi} \int_{0}^{2\pi} f(\theta) \sin n\theta d\theta \qquad (n = 1, 2, 3, ...)$$
(54)

from (50b,d,f).

In this case $f_1(\theta)$ is merely the constant value, say T_0 , of the temperature at the center of the circle r = a, so that (50c,e) reduce to the trivial identity 0 = 0. Also, Equation (50a) becomes

$$A_0 = \frac{1}{2\pi} \int_0^{2\pi} T_0 \, d\theta = T_0. \tag{55}$$

Since this result must be compatible with the first equation of (54), which states that A_0 is the mean value of the temperature distribution along the circle r = a, we conclude that the temperature at the center of the circle is the mean of the temperature distribution along the boundary of the circle. This result also follows directly from (53) when r = 0.

In the second limiting case $r_2 \rightarrow \infty$, the region considered becomes the exterior of the circle $r = r_1$. In this case we must take

$$b_0 = a_n = c_n = 0$$
 (n = 1, 2, 3, ...) (56)

in (50b,d,f). Equations (50d,f) then become, in the limit,

$$r_2 \to \infty: \qquad 0 = \int_0^{2\pi} f_2(\theta) \cos n\theta \ d\theta = \int_0^{2\pi} f_2(\theta) \sin n\theta \ d\theta \qquad (n = 1, 2, 3, ...), \qquad (57)$$

where $f_2(\theta)$ is the prescribed temperature distribution on the circle with increasing radius r_2 . These relations can be true only if $f_2(\theta)$ is constant in the limit. Thus the temperature must approach a uniform value, say T_{∞} , as $r \to \infty$. Equations (50a,b) become

$$A_0 \equiv a_0 = \frac{1}{2\pi} \int_0^{2\pi} f_1(\theta) \, d\theta = T_{\infty}, \tag{58}$$

showing that T_{∞} is the mean of the prescribed temperature distribution on the inner boundary, and Equations (50d,f) reduce to 0 = 0.

If we also write B_n for $r_1^{-n}b_n$, D_n for $r_1^{-n}d_n$, a for r_1 , and $f(\theta)$ for $f_1(\theta)$, we deduce that the solution of the problem

$$\begin{array}{c} \nabla^2 T = 0 \quad (r \ge a) \\ T(a,\theta) = f(\theta) \end{array} \right\}$$
(59)

is of the form

$$T = A_0 + \sum_{n=1}^{\infty} \left(\frac{a}{r}\right)^n (B_n \cos n\theta + D_n \sin n\theta) \qquad (r \ge a) \tag{60}$$

where

$$A_{0} = \frac{1}{2\pi} \int_{0}^{2\pi} f(\theta) d\theta$$

$$B_{n} = \frac{1}{\pi} \int_{0}^{2\pi} f(\theta) \cos n\theta d\theta \qquad (n = 1, 2, 3, ...)$$

$$D_{n} = \frac{1}{\pi} \int_{0}^{2\pi} f(\theta) \sin n\theta d\theta \qquad (n = 1, 2, 3, ...)$$

$$(61)$$

9.5. Poisson's integral. We now show that if the values given by (54) are introduced into (53), then the resultant series can be summed, and the solution can be expressed as an integral. For this purpose, we first replace the dummy variable θ in (54) by φ , to distinguish this variable from the current variable θ in (53). The introduction of (54) into (53) then leads to the relation

$$T(r,\theta) = \frac{1}{2\pi} \int_0^{2\pi} f(\varphi) \, d\varphi + \frac{1}{\pi} \sum_{n=1}^\infty \left(\frac{r}{a}\right)^n \left[\cos n\theta \int_0^{2\pi} f(\varphi) \cos n\varphi \, d\varphi + \sin n\theta \int_0^{2\pi} f(\varphi) \sin n\varphi \, d\varphi\right]$$
$$= \frac{1}{\pi} \int_0^{2\pi} f(\varphi) \left\{\frac{1}{2} + \sum_{n=1}^\infty \left(\frac{r}{a}\right)^n \left[\cos n\theta \cos n\varphi + \sin n\theta \sin n\varphi\right]\right\} d\varphi,$$
or

$$T(r,\theta) = \frac{1}{\pi} \int_0^{2\pi} f(\varphi) \left[\frac{1}{2} + \sum_{n=1}^\infty \left(\frac{r}{a} \right)^n \cos n(\varphi - \theta) \right] d\varphi \qquad (r \le a), \qquad (62)$$

assuming the legitimacy of the interchange of summation and integration in the first form. But since*

$$\cos n(\varphi - \theta) = \operatorname{Re} \{ e^{in(\varphi - \theta)} \},\$$

there follows

$$\sum_{n=1}^{\infty} \left(\frac{r}{a}\right)^n \cos n(\varphi - \theta) = \operatorname{Re}\left\{\sum_{n=1}^{\infty} \left[\frac{r}{a} e^{i(\varphi - \theta)}\right]^n\right\}$$
$$= \operatorname{Re}\left\{\frac{\frac{r}{a} e^{i(\varphi - \theta)}}{1 - \frac{r}{a} e^{i(\varphi - \theta)}}\right\} \quad (r < a)$$
$$= \operatorname{Re}\left\{\frac{\frac{r}{a} \left[e^{i(\varphi - \theta)} - \frac{r}{a}\right]}{\left[1 - \frac{r}{a} e^{i(\varphi - \theta)}\right]\left[1 - \frac{r}{a} e^{-i(\varphi - \theta)}\right]}\right\}$$
$$= \frac{\frac{r}{a} \cos\left(\varphi - \theta\right) - \frac{r^2}{a^2}}{1 - 2\frac{r}{a} \cos\left(\varphi - \theta\right) + \frac{r^2}{a^2}} \quad (r < a).$$

Thus the quantity in braces in (62) becomes

$$\frac{1}{2} + \sum_{n=1}^{\infty} \left(\frac{r}{a}\right)^n \cos n(\varphi - \theta) = \frac{1}{2} \frac{a^2 - r^2}{a^2 - 2ar \cos(\varphi - \theta) + r^2} \quad (r < a), \quad (63)$$

and, noticing that

$$f(\theta)=T(a,\theta),$$

we find that Equation (62) takes the form

$$T(r,\theta) = \frac{1}{2\pi} \int_0^{2\pi} \frac{a^2 - r^2}{a^2 - 2ar\cos(\theta - \phi) + r^2} T(a,\varphi) \, d\varphi \qquad (r < a). \tag{64}$$

This remarkable integral formula, due to *Poisson*, expresses the value of a harmonic function T at all points inside a circle of radius a in terms of the values of T on the *circumference* of the circle.

* "Re" means "real part of"; "Im" means "imaginary part of" in the sense that f = Re f + i Im f, so that Im f is real (see Section 10.1).

It can also be derived by methods which do not presume the validity of (53), or of the interchange of summation and integration which led to (62), and it holds, in particular, when $T(a,\theta) = f(\theta)$ is bounded and piecewise continuous in $(0,2\pi)$. The validity of the series representation (53) can be guaranteed only when somewhat more stringent conditions are

imposed, such as the requirement that $f(\theta)$ be bounded and piecewise differentiable.

9.6. Axisymmetrical temperature distribution in a solid sphere. On the surface of a solid sphere, let the temperature distribution be prescribed in such a way that it is symmetrical with respect to a diameter. The temperature distribution inside the sphere is to be determined. Using spherical coordinates (Figure 9.3), with the z axis coinciding with the axis of symmetry, and noticing that the temperature T is indepen-



Figure 9.3

dent of θ , we recall that Laplace's equation takes the form (see Section 6.18)

$$\frac{\partial}{\partial r}\left(r^{2}\frac{\partial T}{\partial r}\right) + \frac{1}{\sin\varphi}\frac{\partial}{\partial\varphi}\left(\sin\varphi\frac{\partial T}{\partial\varphi}\right) = 0.$$
 (65)

The boundary condition on the surface r = a is then

$$T(a,\varphi) = f(\varphi). \tag{66}$$

If a product solution is assumed,

$$T_{p}(r,\varphi) = R(r) \Phi(\varphi), \qquad (67)$$

Equation (65) is readily separable into the form

$$\frac{1}{R}(r^2R')' = -\frac{1}{\Phi\sin\varphi}(\Phi'\sin\varphi)' = k^2, \qquad (68)$$

where k^2 is a separation constant. This condition is equivalent to the equations

$$(r^{2}R')' - k^{2}R = r^{2}R'' + 2rR' - k^{2}R = 0$$
(69)

$$\frac{1}{\sin\varphi}\frac{d}{d\varphi}\left(\sin\varphi\frac{d\Phi}{d\varphi}\right) + k^{2}\Phi = 0.$$
 (70)

Equation (70) is brought into accordance with Equation (169), Section 4.12, if we write

$$k^2 = n(n+1). (71)$$

The general solution of (70) is then of the form

$$\Phi = A_n P_n(\cos \varphi) + B_n Q_n(\cos \varphi), \qquad (72)$$

where P_n and Q_n are Legendre functions. In order that the solution be finite at the z axis $\varphi = 0,\pi$, we must restrict n to integral values and take

$$B_n = 0. \tag{73}$$

This follows from the fact that the Legendre polynomials (for which *n* is an integer) are the only Legendre functions of type (72) which are finite at both $\varphi = 0$ and $\varphi = \pi$ (see Section 4.12). With the notation of (71), the general solution of (69) is obtained in the form

$$R = C_n r^n + D_n r^{-n-1}. (74)$$

To avoid infinite temperature at the center of the sphere (r = 0), we must set

$$D_n = 0. \tag{75}$$

Hence we are led to assume the desired temperature distribution in the form

$$T(r,\varphi) = \sum_{n=0}^{\infty} c_n \left(\frac{r}{a}\right)^n P_n(\cos \varphi) \qquad (0 < \varphi < \pi), \tag{76}$$

where we have written $A_n C_n = c_n/a^n$. The boundary condition (66) requires that the constants c_n be determined so that the representation

$$f(\varphi) = \sum_{n=0}^{\infty} c_n P_n(\cos \varphi) \qquad (0 < \varphi < \pi)$$
(77)

is valid. If we introduce the new variable

$$\mu = \cos \varphi, \tag{78}$$

the function $f(\varphi)$ becomes a new function of μ , say $F(\mu)$, such that

$$F(\mu) = f(\cos^{-1}\mu),$$
 (79)

and (77) becomes

$$F(\mu) = \sum_{n=0}^{\infty} c_n P_n(\mu) \qquad (-1 < \mu < 1).$$
 (80)

The results of Section 5.13 [Equation (204)] then give the result

$$c_n = \frac{2n+1}{2} \int_{-1}^{1} F(\mu) P_n(\mu) d\mu, \qquad (81)$$

or alternatively, returning to the original variable φ and using the relation $F(\cos \varphi) = f(\varphi)$,

$$c_n = \frac{2n+1}{2} \int_0^{\pi} f(\varphi) P_n(\cos \varphi) \sin \varphi \, d\varphi. \tag{82}$$

If the first *n* derivatives of $F(\mu)$ with respect to its argument are continuous for $-1 \le \mu \le 1$, the second alternative form given by Equation (204), Section 5.13,

$$c_n = \frac{2n+1}{2^{n+1}n!} \int_{-1}^{1} (1-\mu^2)^n \frac{d^n F(\mu)}{d\mu^n} d\mu$$
 (83)

may in certain cases be more useful for the determination of c_n .

9.7. Temperature distribution in a rectangular parallelepiped. Let the temperatures of five faces of a rectangular parallelepiped be maintained at zero,

$$T(0,y,z) = T(L_1,y,z) = T(x,0,z) = T(x,L_2,z) = T(x,y,0) = 0,$$
 (84)

(86)

(87)

and suppose that the sixth face is maintained at a prescribed temperature distribution

$$T(x,y,H) = f(x,y) \tag{85}$$

until steady-state conditions are attained (Figure 9.4). We again investigate the resultant distribution of temperature in the interior.

If we assume a product solution of the relevant equation

$$rac{\partial^2 T}{\partial x^2} + rac{\partial^2 T}{\partial y^2} + rac{\partial^2 T}{\partial z^2} = 0$$

in the form

$$T_p = X(x) Y(y) Z(z),$$

the equation may be separated in the form

$$-\frac{X''}{X} = \frac{Y''}{Y} + \frac{Z''}{Z} = k_1^2, \qquad (88)$$



the separation here depending upon the fact that the first member is independent of both y and z and the second equal member is independent of x. Hence we must have

$$X'' + k_1^2 X = 0 (89)$$

and, after a second separation,

$$-\frac{Y''}{Y} = \frac{Z''}{Z} - k_1^2 = k_2^2.$$
(90)

Thus Y and Z are determined by the equations

$$Y'' + k_2^2 Y = 0, (91)$$

$$Z'' - (k_1^2 + k_2^2)Z = 0. (92)$$

The homogeneous boundary conditions (84) are satisfied by the product solution if the factors satisfy the conditions

$$X(0) = X(L_1) = 0,$$
 (93a,b)

$$Y(0) = Y(L_2) = 0,$$
 (94a,b)

$$Z(0) = 0.$$
 (95)

We thus obtain from Equations (89), (91), (93), and (94)

$$k_1 = \frac{m\pi}{L_1}$$
 (*m* = 1, 2, 3, ...), (96)

$$X = X_m = A_m \sin \frac{m\pi x}{L_1}, \qquad (97)$$

$$k_2 = \frac{n\pi}{L_2}$$
 (n = 1, 2, 3, ...) (98)

$$Y = Y_n = B_n \sin \frac{n\pi y}{L_2}.$$
 (99)

If we write further

$$k_{1}^{2} + k_{2}^{2} = \pi^{2} \left(\frac{m^{2}}{L_{1}^{2}} + \frac{n^{2}}{L_{2}^{2}} \right) \equiv k_{mn}^{2}$$

$$k_{mn} = \pi \sqrt{\frac{m^{2}}{L_{1}^{2}} + \frac{n^{2}}{L_{2}^{2}}},$$
(100)

or

the solution of (92) satisfying (95) becomes

$$Z_{mn} = C_{mn} \sinh k_{mn} z. \tag{101}$$

Thus, writing $a_{mn} = A_m B_n C_{mn}$, we are led to assume the desired solution in the form

$$T(x,y,z) = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} a_{mn} \sin \frac{m\pi x}{L_1} \sin \frac{n\pi y}{L_2} \sinh k_{mn} z.$$
 (102)

This expression satisfies (86), as well as conditions (84), for arbitrary values of the coefficients a_{mn} . It remains, then, to determine these coefficients in such a way that the remaining condition (85) is satisfied. If we introduce the abbreviation

$$c_{mn} = a_{mn} \sinh k_{mn} H, \tag{103}$$

this condition takes the form

$$f(x,y) = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} c_{mn} \sin \frac{m\pi x}{L_1} \sin \frac{n\pi y}{L_2} \quad (0 < x < L_1, 0 < y < L_2).$$
(104)

Thus the coefficients c_{mn} are the coefficients of the double Fourier sine-series expansion of f(x,y) over the indicated rectangle.

These coefficients are readily determined by a simple extension of the methods used in earlier work. If both sides of (104) are multiplied by

$$\sin\left(p\pi x/L_1\right)\sin\left(q\pi y/L_2\right),$$

where p and q are arbitrary positive integers, and if the results are integrated over the rectangle, there follows

$$\int_{0}^{L_{1}} \int_{0}^{L_{2}} f(x,y) \sin \frac{p\pi x}{L_{1}} \sin \frac{q\pi y}{L_{2}} dx dy$$
$$= \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} c_{mn} \int_{0}^{L_{1}} \int_{0}^{L_{2}} \sin \frac{p\pi x}{L_{1}} \sin \frac{q\pi y}{L_{2}} \sin \frac{m\pi x}{L_{1}} \sin \frac{n\pi y}{L_{2}} dx dy.$$
(105)

The double integral on the right can be written as the product

$$\left[\int_0^{L_1} \sin \frac{p\pi x}{L_1} \sin \frac{m\pi x}{L_1} dx\right] \left[\int_0^{L_2} \sin \frac{q\pi y}{L_2} \sin \frac{n\pi y}{L_2} dy\right],$$

and hence, by virtue of Equations (134) and (135), Section 5.10, this product vanishes unless p = m and q = n, in which case it has the value

$$\frac{L_1}{2}\frac{L_2}{2} = \frac{L_1L_2}{4}.$$

Thus the double series in the right-hand member of (105) reduces to a single term, for which m = p and n = q, and there follows

.

$$c_{mn} = \frac{4}{L_1 L_2} \int_0^{L_1} \int_0^{L_2} f(x, y) \sin \frac{m \pi x}{L_1} \sin \frac{n \pi y}{L_2} \, dx \, dy.$$
(106)

With these values of c_{mn} , the solution (102) becomes, with the notation of (103),

$$T(x,y,z) = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} c_{mn} \sin \frac{m\pi x}{L_1} \sin \frac{n\pi y}{L_2} \frac{\sinh k_{mn} z}{\sinh k_{mn} H}, \quad (107)$$

where k_{mn} is defined by (100).

9.8 Ideal fluid flow about a sphere. The preceding problems have all been of the Dirichlet type. To illustrate the solution of a Neumann problem, we investigate the effect of the presence of a stationary sphere of radius a in an initially uniform flow of an ideal incompressible fluid (Figure 9.5). We introduce spherical coordinates and require that for large values of r the flow be in the negative direction parallel to the z axis ($\varphi = 0$). The flow will then clearly be independent of the circumferential angle θ . The velocity potential is here

denoted by $P(r,\varphi)$. According to the results of Section 6.15, this function satisfies Laplace's equation, which here takes the form

$$\frac{\partial}{\partial r}\left(r^2\frac{\partial P}{\partial r}\right) + \frac{1}{\sin\varphi}\frac{\partial}{\partial\varphi}\left(\sin\varphi\frac{\partial P}{\partial\varphi}\right) = 0. \tag{108}$$

Also, the velocity vector V is determined as the gradient of P,



$$\mathbf{V} = \mathbf{\nabla}P = \mathbf{u}_r \frac{\partial P}{\partial r} + \mathbf{u}_{\varphi} \frac{1}{r} \frac{\partial P}{\partial \varphi}$$
$$\equiv \mathbf{u}_r V_r + \mathbf{u}_{\varphi} V_{\varphi}. \tag{109}$$

[See Equation (163e), Section 6.18.] At the surface of the sphere (r = a) there must be no component of V normal to the sphere; that is, $V_r = \frac{\partial P}{\partial r}$ must vanish, ∂P

$$r=a:$$
 $\frac{\partial P}{\partial r}=0.$ (110)

Figure 9.5

As $r \to \infty$, the velocity vector must

approach $-V_0 \mathbf{k}$, where V_0 is the undisturbed velocity. Making use of the readily established relation

$$\mathbf{k} = \mathbf{u}_{\boldsymbol{r}} \cos \varphi - \mathbf{u}_{\boldsymbol{\varphi}} \sin \varphi, \tag{111}$$

we see that as $r \to \infty$ we must have $\mathbf{V} \to -V_0 \cos \varphi \, \mathbf{u}_r + V_0 \sin \varphi \, \mathbf{u}_{\varphi}$. Referring to (109), it follows that as $r \to \infty$ the radial velocity must satisfy the condition

$$r \to \infty$$
: $\frac{\partial P}{\partial r} \to -V_0 \cos \varphi.$ (112)

According to Section 9.2, the boundary conditions (110) and (112) are sufficient to determine P, except for an irrelevant additive constant.

As was shown in Section 9.6, particular product solutions of (108) may be superimposed to give solutions of the form

$$P = \sum_{n} \left[(A_n r^n + B_n r^{-n-1}) P_n(\cos \varphi) + (C_n r^n + D_n r^{-n-1}) Q_n(\cos \varphi) \right]$$
(113)

for which also

$$\frac{\partial P}{\partial r} = \sum_{n} \{ [nA_{n}r^{n-1} - (n+1)B_{n}r^{-n-2}]P_{n}(\cos\varphi) + [nC_{n}r^{n-1} - (n+1)D_{n}r^{-n-2}]Q_{n}(\cos\varphi) \}.$$
(114)

To avoid infinite velocities along the axis of symmetry, we must take

$$C_n = D_n = 0, \tag{115}$$

and to avoid infinite velocities as $r \to \infty$, in accordance with (112), we must have

$$A_n = 0$$
 (*n* = 2, 3, ...). (116)

Finally, to satisfy (110) we then must have

$$nA_{n}a^{n-1} - (n+1)B_{n}a^{-n-2} = 0$$

$$B_{n} = \frac{n}{n+1}a^{2n+1}A_{n}.$$
(117)

or

Thus, with (115), (116), and (117), Equations (113) and (114) become

$$P = A_0 P_0(\cos \varphi) + A_1 \left(r + \frac{a^3}{2r^2} \right) P_1(\cos \varphi)$$
 (118)

and

$$\frac{\partial P}{\partial r} = A_1 \left(1 - \frac{a^3}{r^3} \right) P_1(\cos \varphi). \tag{119}$$

If we recall [see Equation (171), Section 4.12] that

$$P_0(\cos \varphi) = 1, \quad P_1(\cos \varphi) = \cos \varphi,$$

it follows that the remaining condition (112) is satisfied if

$$A_1 = -V_0. (120)$$

Hence the velocity potential is of the form

$$P = -V_0 \left(r + \frac{a^3}{2r^2} \right) \cos \varphi + A_0,$$
 (121)

where A_0 is an irrelevant constant and may be set equal to zero. The velocity vector is of the form

$$\mathbf{V} = -\left[V_0\left(1 - \frac{a^3}{r^3}\right)\cos\varphi\right]\mathbf{u}_r + \left[V_0\left(1 + \frac{a^3}{2r^3}\right)\sin\varphi\right]\mathbf{u}_\varphi, \quad (122)$$

by virtue of (109).

To find the streamlines, we make use of Equation (173), Section 6.19, with

$$h_1 = h_r = 1$$
, $h_2 = h_{\varphi} = r$, $h_3 = h_{\theta} = r \sin \varphi$,

in accordance with Equation (163b), Section 6.18. The stream function ψ is then determined from the relation

$$d\psi = -rac{\partial P}{\partial \varphi} \sin \varphi \, dr + rac{\partial P}{\partial r} \, r^2 \sin \varphi \, d\varphi$$

= $-V_0 \bigg[\bigg(r + rac{a^3}{2r^2} \bigg) \sin^2 \varphi \, dr + \bigg(r^2 - rac{a^3}{r} \bigg) \sin \varphi \cos \varphi \, d\varphi \bigg].$

By integration we then obtain

$$\psi = -\frac{V_0}{2} \left(r^2 - \frac{a^3}{r} \right) \sin^2 \varphi + C, \qquad (123)$$

where C is an arbitrary constant. The streamlines are thus the traces of the surfaces $\psi = \text{constant}$, or, equivalently,

$$r^2\left(1-\frac{a^3}{r^3}\right)\sin^2\varphi = ext{constant}$$
 (124)

in the diametral planes $\theta = \text{constant}$.

A considerably shorter but less direct procedure for solving the problem specified by (108), (110), and (112) would consist of *guessing* initially that the solution will be of the form $P(r,\varphi) = R(r) \cos \varphi$, and of then attempting to determine R(r) in such a way that this expression satisfies (108), and such that R(a) = 0 and $R(\infty) = -V_0$.

9.9. Vibration of a circular membrane. If we consider equilibrium of an element (x, x + dx; y, y + dy) of a membrane under a distributed normal load of intensity f(x,y), and denote the deflection in the z direction by w(x,y) and the tension by T(x,y), the differential resultant vertical component of tensile force normal to the xy plane is found to be

$$\frac{\partial}{\partial x}\left(T\frac{\partial w}{\partial x}\right)\,dx\,dy+\frac{\partial}{\partial y}\left(T\frac{\partial w}{\partial y}\right)\,dx\,dy,$$

if only small slopes and deflections are considered. Since the distributed load on the element is given by f dx dy, the differential equation satisfied by w is of the form

$$\frac{\partial}{\partial x} \left(T \frac{\partial w}{\partial x} \right) + \frac{\partial}{\partial y} \left(T \frac{\partial w}{\partial y} \right) = f$$

$$\left(\frac{\partial^2 w}{\partial x^2} + \frac{\partial^2 w}{\partial y^2} \right) + \left(\frac{\partial T}{\partial x} \frac{\partial w}{\partial x} + \frac{\partial T}{\partial y} \frac{\partial w}{\partial y} \right) = f$$
(125a)

or

or, in terms of the invariant operator ∇ ,

T

$$T \nabla^2 w + (\nabla T) \cdot (\nabla w) = f.$$
 (125b)

For small deflections of an initially tightly stretched membrane we may assume that T is uniform in the membrane. In addition, if we consider here only free vibration of the membrane, and hence replace f by the inertia loading $\partial^2 w$

 $\rho \frac{\partial w}{\partial t^2}$, Equation (125b) reduces to the two-dimensional wave equation

$$\nabla^2 w = \frac{1}{\alpha^2} \frac{\partial^2 w}{\partial t^2}, \qquad (126)$$

$$\alpha^2 = \frac{T}{\rho}.$$
 (127)

where

In dealing with vibrations of a *circular* membrane, we introduce polar coordinates and write (126) in the form

$$\frac{\partial^2 w}{\partial r^2} + \frac{1}{r} \frac{\partial w}{\partial r} + \frac{1}{r^2} \frac{\partial^2 w}{\partial \theta^2} = \frac{1}{\alpha^2} \frac{\partial^2 w}{\partial t^2}.$$
 (128)

If a product solution is assumed in the form

$$w_p = R(r) \Theta(\theta) T(t), \qquad (129)$$

the usual process of separation leads to equations determining the factors in the form

$$T'' + \omega^2 T = 0$$

$$r^2 R'' + rR' + \left(\frac{\omega^2}{\alpha^2} r^2 - k^2\right) R = 0$$

$$\Theta'' + k^2 \Theta = 0$$
(130a,b,c)

where ω^2/α^2 and k^2 are arbitrary separation constants. The general solution of (130b) is of the form [see Equation (130a), Section 4.10]

$$R = A J_k \left(\frac{\omega r}{\alpha}\right) + B Y_k \left(\frac{\omega r}{\alpha}\right), \qquad (131)$$

and, if k and ω are not zero, (130c) and (130a) give

$$\Theta = C\cos k\theta + D\sin k\theta, \qquad (132)$$

$$T = E\cos\omega t + F\sin\omega t. \tag{133}$$

In order that the deflection w be single-valued, the functions in (132) must be of period 2π , and hence k must be integral,

$$k = m$$
 (m = 0, 1, 2, ...). (134)

If k = m = 0, (132) does not represent the general solution of (130c). However, the missing solution $D\theta$ is not periodic. Next, to avoid infinite deflections at the center (r = 0), we must take

$$B=0. (135)$$

Finally, since w must vanish on the boundary r = a,

$$w(a,\theta,t)=0, \tag{136}$$

we require that *each* of the product solutions w_p , to be superimposed, vanish when r = a. By virtue of (135) and (134), this condition requires that ω be a solution of the equation

$$J_m\left(\frac{\omega a}{\alpha}\right) = 0. \tag{137}$$

We notice that the solution $\omega = 0$, when m > 0, would reduce the factor R(r) to zero (identically), and hence it need not be considered.

Denoting the *n*th positive solution of this equation by ω_{mn} ,

$$J_m\left(\frac{\omega_{mn}a}{\alpha}\right) = 0 \qquad (n = 1, 2, \ldots), \tag{138}$$

the superposition of permissible product solutions leads to an expression for w in the form

$$w(r,\theta,t) = \sum_{m=0}^{\infty} \sum_{n=1}^{\infty} J_m \left(\frac{\omega_{mn}r}{\alpha}\right) \left[(a_{mn}\cos m\theta + b_{mn}\sin m\theta)\cos \omega_{mn}t + (c_{mn}\cos m\theta + d_{mn}\sin m\theta)\sin \omega_{mn}t \right].$$
(139)

In addition to the *boundary condition* (136), we may be given two *initial* conditions which prescribe the deflection w and the velocity $\frac{\partial w}{\partial t}$ as functions of r and θ at an initial time t = 0,

$$w(r,\theta,0) = f_1(r,\theta), \qquad \frac{\partial w(r,\theta,0)}{\partial t} = f_2(r,\theta).$$
 (140a,b)

Thus the constants in (139) must be determined so as to satisfy the relations

$$f_{1}(r,\theta) = \sum_{m=0}^{\infty} \sum_{n=1}^{\infty} (a_{mn} \cos m\theta + b_{mn} \sin m\theta) J_{m} \left(\frac{\omega_{mn}r}{\alpha}\right)$$
$$f_{2}(r,\theta) = \sum_{m=0}^{\infty} \sum_{n=1}^{\infty} \omega_{mn}(c_{mn} \cos m\theta + d_{mn} \sin m\theta) J_{m} \left(\frac{\omega_{mn}r}{\alpha}\right)$$
, (141a,b)

when 0 < r < a and $0 < \theta < 2\pi$. The determination follows closely the procedure used in Section 9.7, if use is made of Equations (154) of Section 5.11 and of the corresponding results in Section 5.12 (see Problem 39).

For simplicity, we consider further only the special case when f_1 is independent of θ and $f_2 = 0$; that is, we assume that initially the membrane is deflected into a radially symmetrical form and is released from rest. The ensuing motion of interior points is to be determined. In this case only the terms for which m = 0 are independent of θ , so that all other terms may be suppressed. The initial velocities of all points will vanish if the c and d coefficients are set equal to zero. Hence, writing ω_n for ω_{0n} , Equation (139) reduces to the form

$$w = \sum_{n=1}^{\infty} A_n J_0\left(\frac{\omega_n r}{\alpha}\right) \cos \omega_n t \qquad (0 < r < a, t \ge 0), \qquad (142)$$

where ω_n is the *n*th (positive) solution of

$$J_0\left(\frac{\omega_n a}{\alpha}\right) = 0 \qquad (n = 1, 2, 3, \ldots). \tag{143}$$

If the membrane is initially deflected to the form

$$w(r,0) = F(r),$$
 (144)

the coefficients A_n must be determined so as to make

$$F(r) = \sum_{n=1}^{\infty} A_n J_0\left(\frac{\omega_n r}{\alpha}\right) \qquad (0 < r < a). \tag{145}$$

Reference to Equations (169) and (175a) of Section 5.12, with $\mu_n = \omega_n/\alpha$, p = 0, and L = a, then leads to the determination

$$A_n = \frac{2}{a^2 \left[J_1\left(\frac{\omega_n a}{\alpha}\right) \right]^2} \int_0^a r \ F(r) \ J_0\left(\frac{\omega_n r}{\alpha}\right) \ dr.$$
(146)

It may be noticed that the motion specified by (142) is a superposition of modes having frequencies $\omega_n/2\pi$, where ω_n satisfies (143), the amplitude of vibration in each such mode varying in the radial direction as a Bessel function. The smallest solution of (143) is given by

$$\frac{\omega_1 a}{\alpha} = 2.405,$$

to three places (see Appendix B, Table II), so that, for example, the fundamental frequency $f_1 = \omega_1/2\pi$ of a circular drumhead is

$$f_1 = \frac{2.405}{2\pi} \frac{\alpha}{a} = 0.3828 \sqrt{\frac{T}{\rho a^2}}$$

Conversely, the tension T required to produce a desired fundamental note is given by

$$T = 6.825 \ \rho a^2 f_1^2.$$

9.10. Heat flow in a rod. We next consider the one-dimensional problem of heat flow in a homogeneous rod of length L with insulated sides, the temperature depending only on the distance x from one end of the rod and the time t. The heat flow equation (13), Section 9.2, then becomes

$$\frac{\partial^2 T}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial T}{\partial t},$$
(147)

where α is a constant defined by (14). It is apparent from (147) that in the steady state, when $\frac{\partial T}{\partial t} = 0$, T is a *linear* function of the distance x.

Initially, the temperature is prescribed along the rod as a function of x, say

$$T(x,0) = f(x).$$
 (148)

In particular, if steady-state conditions exist initially, f(x) must be a linear function, of the form

$$f(x) = T_1^{(0)} + (T_2^{(0)} - T_1^{(0)})\frac{x}{L}$$

where $T_1^{(0)}$ and $T_2^{(0)}$ are the initial temperatures of the ends.

At the instant t = 0 we suppose that the temperature at the end x = 0 is changed to a new value, say T_1 , and the temperature at the end x = L is changed to the value T_2 , and these constant values are assumed to be maintained thereafter:

$$T(0,t) = T_1, \quad T(L,t) = T_2 \qquad (t > 0).$$
 (149)

The temperature distribution throughout the rod is required as a function of x and t.

In a problem of this sort it is convenient to express the temperature distribution as the sum of two distributions, one of which is to represent the limiting steady-state distribution (independent of t) after transient effects have become negligible, and the other of which is to represent the transient distribution (which must then approach zero as t increases indefinitely). Thus, writing

$$T(x,t) = T_{S}(x) + T_{T}(x,t),$$
(150)

the function $T_S(x)$ must be a linear function of x satisfying (149), and hence is of the form

$$T_{S}(x) = T_{1} + (T_{2} - T_{1})\frac{x}{L},$$
 (151)

and $T_T(x,t)$ is a particular solution of (147). The function T_T must be determined in such a way that it vanishes when $t \to \infty$,

$$T_T(x,\infty) = 0, \tag{152}$$

and so that the sum $T_S + T_T$ satisfies the initial condition (148). Also, since $T_S(x)$ satisfies (149) it follows that T_T must vanish at the ends x = 0 and x = L for all positive values of t,

$$T_T(0,t) = T_T(L,t) = 0$$
 (t > 0). (153)

Thus the transient distribution satisfies homogeneous end conditions. It is for this reason that the steady-state distribution was first separated out.*

Product solutions of (147) satisfying (152) and (153) are readily obtained in the form

$$T_{T,p} = a_n \sin \frac{n \pi x}{L} e^{\frac{-n^2 \pi^2 a^2 t}{L^2}}$$
 $(n = 1, 2, 3, ...).$

* A procedure of the same type was used in Problem 18, Section 9.3.

Thus, by combining (151) and a superposition of solutions of this type, the required function T(x,t) may be assumed in the form

$$T(x,t) = T_1 + (T_2 - T_1)\frac{x}{L} + \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} e^{\frac{-n^2 \pi^2 a^2 t}{L^2}}.$$
 (154)

We may verify that (154) satisfies the end conditions (149), and that this solution approaches the proper steady-state solution as $t \to \infty$. It remains, then, to determine the coefficients a_n in such a way that the *initial condition* (148) is satisfied, and hence

$$f(x) - T_1 - (T_2 - T_1)\frac{x}{L} = \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} \qquad (0 < x < L).$$
(155)

The Fourier coefficients a_n in (155) are determined in the usual way,

$$a_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} \, dx + \frac{2}{n\pi} (T_2 \cos n\pi - T_1). \tag{156}$$

$$\lambda = rac{L^2}{\pi^2 lpha^2} = rac{s
ho L^2}{\pi^2 K},$$

the parameter λ has the units of time.* Since the exponential factor in (154) is then of the form $e^{-n^2 t/\lambda}$, we see that λ is closely related to the time required for the transient effect to become negligible.

We remark that if the temperatures at the ends are not held constant but are required to vary in a specified way with time, the problem is somewhat more difficult. A procedure which can be used in such cases is illustrated in the following section (see also Problems 47-49).

9.11. The superposition integral. We consider here the special case where a rod is initially at zero temperature,

$$T(x,0) = f(x) \equiv 0,$$
 (158)

and where when t > 0 the end x = 0 is kept at zero temperature while the temperature of the end x = L is varied in a prescribed way with time,

$$T(0,t) = 0, \qquad T(L,t) = F(t).$$
 (159)

* In cgs units the relevant quantities are of the following dimensions:

$$[s] = \operatorname{cal}/(g)(^{\circ}C), \quad [\rho] = g/\operatorname{cm}^{3}, \quad [L] = \operatorname{cm},$$
$$[K] = \operatorname{cal}/(\operatorname{cm})(\operatorname{sec})(^{\circ}C), \quad [\lambda] = \operatorname{sec}.$$

If we write

(157)

As a first step toward the solution of this problem, we solve first the problem when F(t) is unity. Denoting this solution by T = A(x,t), we obtain from (155) and (156), with f(x) = 0, $T_1 = 0$, $T_2 = 1$, the result

$$A(x,t) = \frac{x}{L} + \frac{2}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^n}{n} \sin \frac{n\pi x}{L} e^{-n^2 \frac{t}{\lambda}} \qquad (t > 0), \qquad (160)$$

with the notation of (157).

Next, suppose instead that the temperature at the end x = L is maintained at zero until a certain time $t = \tau_1$, and at that instant is raised to temperature unity and is maintained at unit temperature thereafter. Then it is easy to see



that the resulting temperature distribution will be zero everywhere when $t < \tau_1$, and will be given by the result of replacing t by $t - \tau_1$ in (160) when $t > \tau_1$; that is, in this case

$$T = A(x, t - \tau_1)$$
 $(t > \tau_1)$. (161)

Here $t - \tau_1$ is time measured from the instant of change.

Next, suppose instead that the temperature is raised abruptly to the value F(0) when t = 0 and held at this value until $t = \tau_1$, then is again abruptly

raised by an amount $F(\tau_1) - F(0)$ to the value $F(\tau_1)$ at the time τ_1 and held at that value until $t = \tau_2$, then is abruptly raised by an amount $F(\tau_2) - F(\tau_1)$ at the time τ_2 , and so on (Figure 9.6). From the linearity of the problem it is seen that at the instant following $t = \tau_n$, the temperature distribution is given by the sum

$$T = F(0) A(x,t) + [F(\tau_1) - F(0)] A(x, t - \tau_1) + [F(\tau_2) - F(\tau_1)] A(x, t - \tau_2) + \cdots + [F(\tau_n) - F(\tau_{n-1})] A(x, t - \tau_n).$$
(162)

If we write

$$F(\tau_k) - F(\tau_{k-1}) = \Delta F_k, \qquad \tau_k - \tau_{k-1} = \Delta \tau_k, \qquad (163)$$

Equation (162) can be written in the form

$$T = F(0) A(x, t) + \sum_{k=1}^{n} A(x, t - \tau_k) \left(\frac{\Delta F}{\Delta \tau}\right)_k \Delta \tau_k.$$
(164)

Finally, proceeding to the limit as the number n of jumps becomes infinite, in such a way that all jumps and intervals between successive jumps tend to zero, the definition of the integral suggests the limiting form

$$T(x,t) = F(0) A(x,t) + \int_0^t A(x,t-\tau) F'(\tau) d\tau, \qquad (165)$$

assuming that F(t) is differentiable. This is the superposition integral which gives the desired solution in terms of the basic function A(x,t). An alternative form is obtained by an integration by parts,

$$T = F(0) A(x, t) + \left[A(x, t-\tau) F(\tau)\right]_{\tau=0}^{\tau=t} - \int_0^t F(\tau) \frac{\partial}{\partial \tau} A(x, t-\tau) d\tau$$

or, since $\frac{\partial}{\partial \tau} A(x, t - \tau) = -\frac{\partial}{\partial t} A(x, t - \tau)$, we also have $T(x,t) = A(x,0) F(t) + \int_0^t F(\tau) \frac{\partial A(x, t - \tau)}{\partial t} d\tau.$ (166)

If we notice further that in the present case A(x,0) vanishes when $0 \le x < L$, this result reduces to the form

$$T(x,t) = \int_0^t F(\tau) \frac{\partial A(x,t-\tau)}{\partial t} d\tau \qquad (0 \le x < L).$$
(167)

When this form is used, the solution of the given problem takes the form

$$T(x,t) = \frac{2}{\pi\lambda} \sum_{n=1}^{\infty} (-1)^{n+1} n \left[\int_{0}^{t} F(\tau) e^{n^{2} \frac{\tau}{\lambda}} d\tau \right] e^{-n^{2} \frac{t}{\lambda}} \sin \frac{n\pi x}{L} .$$
(168)

When the form (165) is used, the solution takes the equivalent form

$$T(x,t) = \frac{x}{L}F(t) + \frac{2}{\pi}\sum_{n=1}^{\infty} \frac{(-1)^n}{n} \left[F(0) + \int_0^t F'(\tau) e^{n^2 \frac{\tau}{\lambda}} d\tau \right] e^{-n^2 \frac{t}{\lambda}} \sin \frac{n\pi x}{L}.$$
 (169)

Although the second form appears to be somewhat more complicated than the first, it has the advantage that in many cases the convergence of the infinite series in (169) is more rapid than the convergence of (168).

As an example, suppose that the temperature at the end x = L is increased uniformly with time from a zero value, at the rate of T_0 degrees per second,

$$F(t) = T_0 t.$$
 (170)

2 t

Then (168) gives

$$T(x,t) = \frac{2\lambda T_0}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^{n+1}}{n^3} \left[\frac{n^2 t}{\lambda} - \left(1 - e^{-n^2 \frac{t}{\lambda}} \right) \right] \sin \frac{n\pi x}{L}, \quad (171a)$$

and (169) gives the alternative form

$$T(x,t) = T_0 \frac{x}{L} t - \frac{2\lambda T_0}{\pi} \sum_{n=1}^{\infty} (-1)^n \frac{1 - e^{-n^2 \frac{1}{\lambda}}}{n^3} \sin \frac{n\pi x}{L}.$$
 (171b)

The equivalence of these forms is verified by noticing the validity of the expansion

$$\frac{x}{L} = -\frac{2}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^n}{n} \sin \frac{n\pi x}{L} \qquad (0 \le x < L).$$

It is clear that the second form (171b) is better adapted to numerical calculation.

9.12. Traveling waves. It was found in Section 8.5 that the general solution of the one-dimensional wave equation

$$\frac{\partial^2 \varphi}{\partial x^2} = \frac{1}{c^2} \frac{\partial^2 \varphi}{\partial t^2} \tag{172}$$

is of the form

$$\varphi = f(x - ct) + g(x + ct), \qquad (173)$$

where f and g are arbitrary functions. If we consider the graphical representation of φ as a function of x for varying values of the time t, we see that a solution of the form $\varphi = f(x - ct)$ is represented at the time t = 0 by $\varphi = f(x)$, and is represented at any following time t by the same curve moved parallel to itself a distance ct in the positive x direction. That is, a solution $\varphi = f(x - ct)$ is represented by a curve moving in the positive x direction with velocity c. Similarly, a solution $\varphi = g(x + ct)$ is represented by a curve moving in the negative x direction with velocity c. We may speak of these solutions, in a general sense, as traveling waves.

D'Alembert's formula (67), of Section 8.5, in the form

$$\varphi(x,t) = \frac{1}{2} \left[F(x+ct) + F(x-ct) \right] + \frac{1}{2c} \int_{x-ct}^{x+ct} G(\xi) \, d\xi, \qquad (174)$$

defines that solution of (172) which satisfies the conditions

$$\varphi(x,0) = F(x), \qquad \varphi_t(x,0) = G(x)$$
 (175)

for all real values of x, where F and G are any twice-differentiable functions. Here, for example, we may interpret φ and φ_t as the lateral deflection and velocity, respectively, of a tightly stretched uniform string of infinite length with $c = \sqrt{T/\rho}$, where T is tension and ρ is linear mass density, and where small slopes and deflections are assumed. (The case of a string of finite length is treated in Problem 28 of Chapter 8 and in Problem 59 at the end of this chapter.)

In many other applications we are principally interested in solutions of the wave equation which are *periodic in time*. Such solutions, for (172), must be combinations of terms of the form $f_1(x) \cos \omega t$ or $f_2(x) \sin \omega t$ or, equivalently, must be the real or imaginary part of a function of the form

$$\varphi = F(x) e^{i\omega t}, \qquad (176)$$

where F(x) may be complex. By introducing (176) into (172) and canceling the resultant common factor $e^{i\omega t}$, there follows

$$F'' + \frac{\omega^2}{c^2} F = 0.$$
 (177)

If we write the general solution of (177) in the complex form

$$F = c_1 e^{\frac{i\omega_x}{c}x} + c_2 e^{-\frac{i\omega_x}{c}x},$$
 (178)

the required solution (176) becomes

$$\varphi = c_1 e^{i\frac{\omega}{c}(x+ct)} + c_2 e^{-i\frac{\omega}{c}(x-ct)}.$$
(179)

This solution is of the form of (173). By taking real and imaginary parts, we see that linear combinations of terms of the form

$$\cos\frac{\omega}{c}(x-ct), \quad \sin\frac{\omega}{c}(x-ct)$$
 (180)

for arbitrary values of ω are available for the representation of periodic plane waves moving in the *positive* x direction, whereas combinations of terms of the form

$$\cos\frac{\omega}{c}(x+ct), \quad \sin\frac{\omega}{c}(x+ct)$$
 (181)

are solutions of (172) representing periodic plane waves moving in the *negative* x direction. In certain problems it may be more convenient to retain such solutions in the complex exponential form of (179).

We next investigate the existence of analogous spherical wave solutions of the wave equation. In spherical coordinates, with the solution φ dependent only on the radius r from the origin, the wave equation takes the form

$$\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2\frac{\partial\varphi}{\partial r}\right) = \frac{1}{c^2}\frac{\partial^2\varphi}{\partial t^2}.$$
(182)

If we assume a solution periodic in time, of the form

$$\varphi = F(r) e^{i\omega t}, \qquad (183)$$

substitution of (183) into (182) gives the equation

$$rF'' + 2F' + \frac{\omega^2}{c^2}rF = 0$$
 (184)

to be satisfied by the amplitude function F. The general solution of (184) can be expressed in terms of Bessel functions by identifying (184) with Equation

(127), Section 4.10. The solution is then given by Equations (128) and (129) of that section in the form

$$F = \frac{A}{\sqrt{r}} J_{1/2}\left(\frac{\omega r}{c}\right) + \frac{B}{\sqrt{r}} J_{-1/2}\left(\frac{\omega r}{c}\right).$$
(185)

This result is in turn expressible in terms of elementary functions, by virtue of Equation (115), Section 4.9, in the form

$$F = \sqrt{\frac{2c}{\pi\omega}} \left[A \frac{\sin\frac{\omega r}{c}}{r} + B \frac{\cos\frac{\omega r}{c}}{r} \right]$$

or, finally, in the complex form

$$F = c_1 \frac{e^{i\frac{\omega}{c}r}}{r} + c_2 \frac{e^{-i\frac{\omega}{c}r}}{r}.$$
 (186)

Thus Equation (183) becomes

$$\varphi = c_1 \frac{e^{i\frac{\omega}{c}(r+ct)}}{r} + c_2 \frac{e^{-i\frac{\omega}{c}(r-ct)}}{r}.$$
(187)

Linear combinations of such functions, or of their real and imaginary parts, are thus solutions of the wave equation which represent periodic spherical waves moving inward toward or outward from the origin, respectively. It may be noticed that the amplitude of the oscillation is inversely proportional to r.

In cylindrical coordinates, with φ dependent only on the distance r from the z axis, the wave equation becomes

$$\frac{1}{r}\frac{\partial}{\partial r}\left(r\frac{\partial\varphi}{\partial r}\right) = \frac{1}{c^2}\frac{\partial^2\varphi}{\partial t^2}.$$
(188)

The assumption

$$\varphi = F(r) e^{i\omega t} \tag{189}$$

then leads to the equation

$$F'' + \frac{1}{r}F' + \frac{\omega^2}{c^2}F = 0.$$
 (190)

Although the general solution can be written in the form

$$F = A J_0\left(\frac{\omega r}{c}\right) + B Y_0\left(\frac{\omega r}{c}\right),$$

we are guided by the preceding developments in writing this solution instead in the *complex* form

$$F = c_1 H_0^{(1)} \left(\frac{\omega r}{c}\right) + c_2 H_0^{(2)} \left(\frac{\omega r}{c}\right), \qquad (191)$$

where, as defined in Section 4.8, the Hankel functions are given by

$$H_{0}^{(1)}\left(\frac{\omega r}{c}\right) = J_{0}\left(\frac{\omega r}{c}\right) + i Y_{0}\left(\frac{\omega r}{c}\right),$$

$$H_{0}^{(2)}\left(\frac{\omega r}{c}\right) = J_{0}\left(\frac{\omega r}{c}\right) - i Y_{0}\left(\frac{\omega r}{c}\right).$$
 (192)

The solution (189) then becomes

$$\varphi = c_1 e^{i\omega t} H_0^{(1)} \left(\frac{\omega r}{c}\right) + c_2 e^{i\omega t} H_0^{(2)} \left(\frac{\omega r}{c}\right).$$
(193)

The usefulness of the definition of the Hankel functions is now seen if we consider the asymptotic expressions for $H_0^{(1)}$ and $H_0^{(2)}$ as given by Equation (106), Section 4.9. By making use of these results, the behavior of the expression (193) for large values of r is found to be given by the asymptotic expression

$$\varphi \sim \sqrt{\frac{2c}{\pi\omega}} \left[\left(c_1 e^{-i\frac{\pi}{4}} \right) \frac{e^{i\frac{\omega}{c}(r+ct)}}{\sqrt{r}} + \left(c_2 e^{i\frac{\pi}{4}} \right) \frac{e^{-i\frac{\omega}{c}(r-ct)}}{\sqrt{r}} \right] \qquad (r \to \infty).$$
(194)

Hence we see that for large values of r the real and imaginary parts of $e^{i\omega t} H_0^{(1)}(\omega r/c)$ represent cylindrical waves moving inward toward the z axis, whereas the real and imaginary parts of $e^{i\omega t} H_0^{(2)}(\omega r/c)$ represent outward traveling waves. The amplitude of the oscillation here is inversely proportional to \sqrt{r} .

More generally, reference to the asymptotic expressions for $H_p^{(1)}$ and $H_p^{(2)}$ shows that the real and imaginary parts of $e^{i\omega t} H_p^{(1)}(kr)$ represent *inward* traveling cylindrical waves for large values of r (for any nonnegative value of p), whereas the real and imaginary parts of $e^{i\omega t} H_p^{(2)}(kr)$ represent *outward* traveling waves.

Although in most problems these complex forms are preferable, the explicit real and imaginary parts may be listed as follows:

$$e^{i\omega t} H_{p}^{(1)}\left(\frac{\omega r}{a}\right) = \left[J_{p}\left(\frac{\omega r}{a}\right)\cos\omega t - Y_{p}\left(\frac{\omega r}{a}\right)\sin\omega t\right] \\ + i\left[J_{p}\left(\frac{\omega r}{a}\right)\sin\omega t + Y_{p}\left(\frac{\omega r}{a}\right)\cos\omega t\right] \\ e^{i\omega t} H_{p}^{(2)}\left(\frac{\omega r}{a}\right) = \left[J_{p}\left(\frac{\omega r}{a}\right)\cos\omega t + Y_{p}\left(\frac{\omega r}{a}\right)\sin\omega t\right] \\ + i\left[J_{p}\left(\frac{\omega r}{a}\right)\sin\omega t - Y_{p}\left(\frac{\omega r}{a}\right)\cos\omega t\right]$$
(195a,b)
9.13. The pulsating cylinder. As a simple application of these facts, we next consider a problem which is related to certain aerodynamic investigations of nonuniform motion. The lateral boundary of an infinite circular cylinder surrounded by ideal compressible fluid is caused to pulsate radially with



Figure 9.7

circular frequency ω in such a way that the radial velocity of the boundary varies periodically about a zero value, with maximum value V_0 (Figure 9.7) The periodically varying velocities imparted to the points in the surrounding fluid are to be determined.

According to hydrodynamic theory [see Equations (198) and (199), Section 6.20], the velocity vector V is here the gradient of a velocity potential φ which satisfies the wave equation

$$\nabla^2 \varphi = \frac{1}{c^2} \frac{\partial^2 \varphi}{\partial t^2},$$
 (196)

where c is the velocity of sound in the fluid. If we introduce cylindrical coordinates, noticing that φ is independent of z and θ , Equation (196) becomes

$$\frac{\partial^2 \varphi}{\partial r^2} + \frac{1}{r} \frac{\partial \varphi}{\partial r} = \frac{1}{c^2} \frac{\partial^2 \varphi}{\partial t^2}.$$
 (197)

The magnitude V of the (radial) velocity is given by the equation

$$V = \frac{\partial \varphi}{\partial r} . \tag{198}$$

On the boundary the fluid velocity must equal the velocity of the pulsating boundary, say $V_0 \cos \omega t$, whereas for large values of r the velocity must correspond to outward traveling cylindrical waves (in the absence of a reflecting outer boundary).

As will be seen, it is convenient to use complex notation, with the convention that real parts of complex functions or of products of complex functions are to be taken. Thus the boundary condition on the cylinder r = a is written in the form

r = a: $\frac{\partial \varphi}{\partial r} = V_0 e^{i\omega t}$. (199)

Accordingly, if we write

$$\varphi = F(r) e^{i\omega t}, \qquad (200)$$

Equation (193) of the preceding section shows that a suitable expression for φ is of the form

$$\varphi = c_1 e^{i\omega t} H_0^{(1)}\left(\frac{\omega r}{c}\right) + c_2 e^{i\omega t} H_0^{(2)}\left(\frac{\omega r}{c}\right). \tag{201}$$

If we make use of the derivative formula (114), Section 4.9, there then follows

$$V = \frac{\partial \varphi}{\partial r} = -\frac{\omega}{c} \left[c_1 e^{i\omega t} H_1^{(1)} \left(\frac{\omega r}{c} \right) + c_2 e^{i\omega t} H_1^{(2)} \left(\frac{\omega r}{c} \right) \right].$$
(202)

To ensure outward traveling waves for large values of r, we must take

$$c_1 = 0.$$
 (203)

The remaining constant c_2 is determined by (199) in the form

$$V_{0} e^{i\omega t} = -\frac{\omega}{c} c_{2} e^{i\omega t} H_{1}^{(2)} \left(\frac{\omega a}{c}\right),$$

$$c_{2} = -\frac{c}{\omega} \frac{V_{0}}{H_{1}^{(2)} \left(\frac{\omega a}{c}\right)}.$$
(204)

With (203) and (204) the velocity (202) is determined as the real part of the expression

$$V = V_0 e^{i\omega t} \frac{H_1^{(2)}\left(\frac{\omega r}{c}\right)}{H_1^{(2)}\left(\frac{\omega a}{c}\right)}.$$
 (205)

If now we write (205) in the expanded form

$$V = V_0 \operatorname{Re}\left\{ (\cos \omega t + i \sin \omega t) \frac{J_1\left(\frac{\omega r}{c}\right) - i Y_1\left(\frac{\omega r}{c}\right)}{J_1\left(\frac{\omega a}{c}\right) - i Y_1\left(\frac{\omega a}{c}\right)} \right\},\,$$

the usual process of rationalization and separation of real and imaginary parts leads to the explicit form

$$V = \frac{V_0}{J_1^2\left(\frac{\omega a}{c}\right) + Y_1^2\left(\frac{\omega a}{c}\right)} \left\{ J_1\left(\frac{\omega a}{c}\right) \left[J_1\left(\frac{\omega r}{c}\right) \cos \omega t + Y_1\left(\frac{\omega r}{c}\right) \sin \omega t \right] - Y_1\left(\frac{\omega a}{c}\right) \left[J_1\left(\frac{\omega r}{c}\right) \sin \omega t - Y_1\left(\frac{\omega r}{c}\right) \cos \omega t \right] \right\}.$$
 (206)

The amplitude of the velocity oscillation at a given distance r from the axis is given by

$$A(r) = V_0 \left[\frac{J_1^2 \left(\frac{\omega r}{c}\right) + Y_1^2 \left(\frac{\omega r}{c}\right)}{J_1^2 \left(\frac{\omega a}{c}\right) + Y_1^2 \left(\frac{\omega a}{c}\right)} \right]^{\frac{1}{2}}.$$
 (207)

With this notation, Equation (206) can be written in the form

$$V = A(r) \cos \left[\omega t - \alpha(r)\right], \qquad (208)$$

where the phase shift α at distance r from the axis is given by

$$\alpha(r) = \tan^{-1} \left[\frac{J_1\left(\frac{\omega a}{c}\right) Y_1\left(\frac{\omega r}{c}\right) - Y_1\left(\frac{\omega a}{c}\right) J_1\left(\frac{\omega r}{c}\right)}{J_1\left(\frac{\omega a}{c}\right) J_1\left(\frac{\omega r}{c}\right) + Y_1\left(\frac{\omega a}{c}\right) Y_1\left(\frac{\omega r}{c}\right)} \right].$$
(209)

For values of r and a large with respect to c/ω there follows, in particular,

$$A(r) \sim V_0 \sqrt{\frac{a}{r}}, \qquad \alpha(r) \sim \frac{\omega}{c}(r-a),$$

if use is made of Equation (105), Section 4.9. Thus, as $r \to \infty$, (208) becomes

$$V \sim V_0 \sqrt{\frac{a}{r}} \cos\left[\omega\left(t - \frac{r-a}{c}\right)\right]$$

if $a\omega \gg c$.

9.14. Examples of the use of Fourier integrals. When the required solution of a problem is specified along a straight boundary of infinite extent, the superposition of particular solutions is generally accomplished by integration rather than summation.

As a first illustration, we require the solution of the two-dimensional Laplace equation

$$\frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} = 0 \tag{210}$$

valid in the half plane y > 0, taking on prescribed values along the x axis for all values of x,

$$\varphi(x,0) = f(x), \tag{211}$$

and vanishing at large distances from the origin,

$$\lim_{x^2+y^2\to\infty}\varphi(x,y)=0.$$
 (212)

Physically, we may interpret this problem (for example) as determining the steady-state temperature distribution in a thin plane sheet of infinite extent on one side of a straight edge, when the temperature is prescribed along that edge.

Product solutions of (210) which vanish as $y \rightarrow \infty$ are readily obtained in the form

$$\varphi_{p} = e^{-uy} \left(A \cos ux + B \sin ux \right) \qquad (u > 0), \qquad (213)$$

where u^2 has been written for the arbitrary separation constant and A and B are arbitrary. If we think of A and B as functions of u,

$$A = A(u), \qquad B = B(u), \tag{214}$$

we may superimpose solutions of form (213) for all positive values of u by integration, and so write

$$\varphi(x,y) = \int_0^\infty e^{-uy} [A(u)\cos ux + B(u)\sin ux] du. \qquad (215)$$

This expression formally satisfies (210) and (212) for arbitrary forms of the functions A and B. The remaining condition (211) will then be satisfied if we determine A and B such that

$$f(x) = \int_0^\infty \left[A(u) \cos ux + B(u) \sin ux \right] du \qquad (-\infty < x < \infty).$$
 (216)

But, in accordance with Equations (226) and (227) of Section 5.14, the righthand member of (216) reduces to the Fourier integral representation of f(x)if we take

$$A(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(\xi) \cos u\xi \, d\xi, \quad B(u) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(\xi) \sin u\xi \, d\xi.$$
(217a,b)

With these results, Equation (215) takes the form

$$\varphi(x,y) = \frac{1}{\pi} \int_0^\infty e^{-uy} \left\{ \left[\int_{-\infty}^\infty f(\xi) \cos u\xi \, d\xi \right] \cos ux + \left[\int_{-\infty}^\infty f(\xi) \sin u\xi \, d\xi \right] \sin ux \right\} du \qquad (218)$$

or, equivalently,

$$\varphi(x,y) = \frac{1}{\pi} \int_0^\infty \left\{ \int_{-\infty}^\infty e^{-uy} f(\xi) \cos u(\xi - x) \, d\xi \right\} \, du. \tag{219}$$

We remark that only when the integral $\int_{-\infty}^{\infty} |f(x)| dx$ exists is the Fourier integral representation (216) *necessarily* valid, but it *may* be valid when this is not the case. Further, in certain cases when (216) is *not* valid, still, if the *u* integration is carried out before the ξ integration, Equation (219) may nevertheless lead to a definite expression for φ , which, however, may not satisfy the condition (212). The validity of the *formal* solution so obtained should be checked by direct calculation in such cases. Physical considerations will generally indicate whether violation of the condition (212) is permissible in these exceptional cases.

When the order of integration is reversed, the solution (219) takes the form

$$\varphi(x,y) = \frac{1}{\pi} \int_{-\infty}^{\infty} \left\{ \int_{0}^{\infty} e^{-uy} \cos u(\xi - x) \, du \right\} f(\xi) \, d\xi$$

or, after an evaluation of the inner integral,

$$\varphi(x,y) = \frac{1}{\pi} \int_{-\infty}^{\infty} \frac{y f(\xi) d\xi}{y^2 + (\xi - x)^2}.$$
 (220)

It can be proved that this integral converges and defines a solution of Laplace's equation for y > 0 when f(x) is bounded and piecewise continuous for all real x.

For example, if we take

$$f(x) = \begin{cases} 0 & (x < 0) \\ 1 & (x > 0), \end{cases}$$
(221)

there follows, from (220),

$$\varphi(x,y) = \frac{1}{\pi} \int_0^\infty \frac{y}{y^2 + (\xi - x)^2} d\xi$$
$$= \frac{1}{\pi} \left(\frac{\pi}{2} + \tan^{-1} \frac{x}{y} \right)$$

or

$$\varphi(x,y) = 1 - \frac{\theta}{\pi}, \qquad (222)$$

where θ is the polar angle. In this case $\int_{-\infty}^{\infty} |f(x)| dx$ does not exist and, further, the right-hand members of (217) do not exist. However, (222) is easily shown to satisfy (210) and to reduce to (221) when y = 0. Since (212) is violated by the prescribed value of φ on the x axis ($\theta = 0$), it is not surprising that it is also violated by the solution (222) for all values of θ except $\theta = \pi$. Here (222) is the only solution of (210) which reduces to (221) on the x axis and remains bounded as $y \to \infty$.

Other one-dimensional problems of similar type are solved in an analogous way. As a second example, we consider the problem of one-dimensional heat flow in a rod of infinite length with insulated sides (see Section 9.10). We assume that an initial distribution of temperature is prescribed along the rod,

$$T(x,0) = f(x),$$
 (223)

and that at the instant t = 0 the temperature at the end x = 0 is changed to zero and maintained at zero thereafter,

$$T(0,t) = 0$$
 $(t > 0).$ (224)

The temperature at a point x at time t is required.

Product solutions of the relevant differential equation (147) satisfying (224) are of the form

$$T_p = A \ e^{-u^2 a^2 t} \sin u x,$$

where u is arbitrary. Considering A as a function of u, we may superimpose such solutions for all positive values of u by writing

$$T(x,t) = \int_0^\infty A(u) \, e^{-\,u^2 \alpha^2 t} \, \sin \, ux \, du \qquad (x > 0). \tag{225}$$

The condition (223) then requires that A(u) be determined so that

$$T(x,0) = f(x) = \int_0^\infty A(u) \sin ux \, du \qquad (x > 0).$$
 (226)

But, according to Equation (223), Section 5.14, we have the Fourier sineintegral representation

$$f(x) = \frac{2}{\pi} \int_0^\infty \left\{ \int_0^\infty f(\xi) \sin u\xi \, d\xi \right\} \sin ux \, du \qquad (x > 0),$$

and hence the right-hand member of (226) reduces to f(x) if we take

$$A(u) = \frac{2}{\pi} \int_0^\infty f(\xi) \sin u\xi \, d\xi.$$
 (227)

Thus the solution (225) becomes

$$T(x,t) = \frac{2}{\pi} \int_0^\infty \int_0^\infty e^{-u^2 \alpha^2 t} f(\xi) \sin u\xi \sin ux \, d\xi \, du.$$
 (228)

This form can be simplified if we carry out the u integration first. Thus we obtain

$$T(x,t) = \frac{2}{\pi} \int_0^\infty f(\xi) \left\{ \int_0^\infty e^{-u^2 \alpha^2 t} \sin u\xi \sin ux \, du \right\} d\xi.$$
 (229)

The inner integral is first simplified by writing

$$\sin u\xi \sin ux = \frac{1}{2} [\cos u(\xi - x) - \cos u(\xi + x)]$$

Then if use is made of the known result*

$$\int_0^\infty e^{-a^2x^2} \cos bx \, dx = \frac{\sqrt{\pi}}{2a} \, e^{-\frac{b^2}{4a^2}} \qquad (a > 0), \tag{230}$$

Equation (229) is reduced to the form

$$T(x,t) = \frac{1}{2\alpha\sqrt{\pi t}} \int_0^\infty f(\xi) \left[e^{-\frac{(\xi-x)^2}{4\alpha^2 t}} - e^{-\frac{(\xi+x)^2}{4\alpha^2 t}} \right] d\xi, \qquad (231)$$

the validity of which can be established for any bounded and piecewise continuous f.

In particular, if the initial temperature is uniform,

$$T(x,0) = f(x) = T_0,$$
 (232)

* See Equation (99) of Chapter 7.

Equation (231) is reduced by obvious substitutions to the form

$$T(x,t) = \frac{T_0}{\sqrt{\pi}} \left(\int_{-x/2a\sqrt{t}}^{\infty} e^{-u^2} du - \int_{x/2a\sqrt{t}}^{\infty} e^{-u^2} du \right)$$
$$= \frac{T_0}{\sqrt{\pi}} \int_{-x/2a\sqrt{t}}^{x/2a\sqrt{t}} e^{-u^2} du = \frac{2T_0}{\sqrt{\pi}} \int_{0}^{x/2a\sqrt{t}} e^{-u^2} du.$$

The so-called error function, defined as

$$\operatorname{erf}(x) = \frac{2}{\sqrt{\pi}} \int_0^x e^{-u^2} du,$$
 (233)

is of frequent occurrence in the study of heat flow and in other fields, and is a tabulated function. In terms of this function the preceding result takes the form

$$T(x,t) = T_0 \operatorname{erf}\left(\frac{x}{2\alpha\sqrt{t}}\right).$$
 (234)

If we integrate both sides of (230) with respect to the parameter b, from b = 0 to $b = \beta$, we obtain the relation

$$\int_0^\infty e^{-a^2x^2} \frac{\sin\beta x}{x} dx = \frac{\sqrt{\pi}}{2a} \int_0^\beta e^{-\frac{b^2}{4a^2}} db$$
$$= \frac{\pi}{2} \operatorname{erf}\left(\frac{\beta}{2a}\right), \qquad (235)$$

which is frequently useful in dealing with problems which involve the error function.

9.15. Application of the Laplace transform to the telegraph equations for a long line. In dealing with the flow of electricity in a long insulated cable, the potential e (volts) and the current i (amperes) are found to be related approximately by the simultaneous equations

$$\frac{\partial e}{\partial x} = -L \frac{\partial i}{\partial t}, \qquad \frac{\partial i}{\partial x} = -C \frac{\partial e}{\partial t},$$
 (236a,b)

where x is distance (miles) from one end of the cable, if the effects of leakage and resistance are neglected. Here L is inductance per unit length of cable (henries/mile), and C is the capacitance to ground per unit length (farads/mile). If i is eliminated between (236a,b), we find that e satisfies the wave equation

$$\frac{\partial^2 e}{\partial x^2} = LC \frac{\partial^2 e}{\partial t^2}, \qquad (237)$$

and a completely analogous equation is satisfied by *i*.

Initially, the line is here considered to be dead,

$$e(x,0) = \frac{\partial e(x,0)}{\partial t} = 0.$$
(238)

Then, beginning at the instant t = 0, a voltage is impressed at the end x = 0, and thereafter that end is maintained at a prescribed potential which may vary with time,

$$e(0,t) = f(t)$$
 (t > 0). (239)

If the other end (x = l) is open, then i = 0 when x = l. With the use of (236a), this condition becomes

$$\frac{\partial e(l,t)}{\partial x} = 0 \qquad (t > 0), \qquad (240a)$$

whereas if the end x = l is grounded, then e = 0 when x = l,

$$e(l,t) = 0$$
 (t > 0). (240b)

We now illustrate the application of the Laplace transform procedure to the solution of problems of this type. For any fixed value of x, the voltage e(x,t) is a function of time t, and hence we may speak of its transform $\bar{e}(x,s)$, where

$$\bar{e}(x,s) = \int_0^\infty \exp\left(-st\right) e(x,t) \, dt = \mathscr{L}\{e(x,t)\}. \tag{241}$$

The variable x is treated as a constant in the integration. We see that the transform of the x derivative of e is the x derivative of the transform \overline{e} ,

$$\mathscr{L}\left\{\frac{\partial e(x,t)}{\partial x}\right\} = \int_0^\infty \exp\left(-st\right) \frac{\partial e(x,t)}{\partial x} dt = \frac{\partial \bar{e}(x,s)}{\partial x}, \qquad (242)$$

and similarly for higher derivatives. However, as regards time differentiation, the results of Chapter 2 can again be applied to give

$$\mathscr{L}\left\{\frac{\partial e(x,t)}{\partial t}\right\} = s\bar{e}(x,s) - e(x,0), \qquad (243a)$$

$$\mathscr{L}\left\{\frac{\partial^2 e(x,t)}{\partial t^2}\right\} = s^2 \bar{e}(x,s) - se(x,0) - \frac{\partial e(x,0)}{\partial t}, \qquad (243b)$$

and so forth (see formulas T3, T4, T5, page 74). In a similar way, all the results of Chapter 2 are again applicable here, insofar as time variation is concerned.

If we take the transform of Equation (237) and make use of these facts, there follows

$$\frac{\partial^2 \bar{e}(x,s)}{\partial x^2} = LC \left[s^2 \bar{e}(x,s) - se(x,0) - \frac{\partial e(x,0)}{\partial t} \right]$$
(244)

or, taking into account the initial conditions (238),

$$\frac{\partial^2 \bar{e}(x,s)}{\partial x^2} - LC \, s^2 \bar{e}(x,s) = 0. \tag{245}$$

It will be convenient to write

$$v = \frac{1}{\sqrt{LC}},\tag{246}$$

so that (245) becomes

$$\frac{\partial^2 \bar{e}}{\partial x^2} - \frac{s^2}{v^2} \, \bar{e} = 0. \tag{247}$$

We assume that L and C are constant. Since only x differentiation is involved in (247), this equation is integrated as though it were an ordinary differential equation, holding s constant in the process and hence replacing the arbitrary constants of integration by arbitrary functions of s. In this way we obtain the result

$$\ddot{e}(x,s) = A(s)e^{-s\frac{x}{v}} + B(s)e^{s\frac{x}{v}}.$$
 (248)

We consider here only the case where the line is *open* at the end x = l, so that (240a) and (239) are to be satisfied. If we take the transforms of these two equations we find that $\tilde{e}(x,s)$ must then satisfy the end conditions

$$\bar{e}(0,s) = \bar{f}(s), \qquad \frac{\partial \bar{e}(l,s)}{\partial x} = 0,$$
(249a,b)

where f(s) is the transform of f(t). Thus the functions A(s) and B(s) of (248) are determined by the conditions

$$A(s) + B(s) = f(s)$$

$$-A(s)e^{-\frac{sl}{v}} + B(s)e^{\frac{sl}{v}} = 0$$
 (250a,b)

and hence are of the form

$$A(s) = \frac{\tilde{f}(s)}{1 + e^{-2s_{\bar{v}}^{l}}} \\ B(s) = \frac{\tilde{f}(s)e^{-2s_{\bar{v}}^{l}}}{1 + e^{-2s_{\bar{v}}^{l}}} \end{bmatrix}.$$
 (251a,b)

In the special limiting case of an infinite line,

$$l = \infty,$$
 (252)

there follows

$$A(s) = \tilde{f}(s), \qquad B(s) = 0,$$
 (253a,b)

and (248) becomes merely

$$\bar{e}(x,s) = \bar{f}(s)e^{-s\frac{x}{v}}.$$
(254)

Having now determined the *transform* of the desired solution e(x,t), we next make use of formula T9, page 74, to determine e(x,t) in the form

$$e(x,t) = \begin{cases} f\left(t - \frac{x}{v}\right) & \text{when } t > \frac{x}{v} \text{ or } x < vt \\ 0 & \text{when } t < \frac{x}{v} \text{ or } x > vt. \end{cases}$$
(255)

Thus we see that at a given time t the effect of introducing the voltage at x = 0 is present only at distances not greater than $vt = t/\sqrt{LC}$ miles from that end.





That is, a voltage wave is propagated along the line with velocity $v = 1/\sqrt{LC}$, in a manner specified by Equation (255). For example, if e(0,t) = f(t) varies periodically with time as is indicated in Figure 9.8(a), the voltage along the line at time t is as indicated in Figure 9.8(b). In general, we may verify that the representation of e(x,t) as a function of x at time t is obtained by reflecting the curve for e(0,t) about the axis t = 0, replacing the time scale by a $\frac{\text{distance}}{v}$ scale, and then translating the reflected curve to the right through t units of x/v.

In the more general case of a *finite line*, the introduction of (251) into (248) gives the result

$$\ddot{e}(x,s) = \frac{\ddot{f}(s)}{1 + e^{-2s_{v}^{2}}} \left[e^{-s_{v}^{2}} + e^{-\frac{s}{v}(2t-x)} \right].$$
(256)

To determine the solution having this transform, we first expand $1/(1 + e^{-2s\frac{l}{v}})$ in a series of ascending powers of $e^{-2s\frac{l}{v}}$, and so obtain

$$\bar{e}(x,s) = \bar{f}(s) \left[e^{-s\frac{x}{v}} + e^{-\frac{s}{v}(2l-x)} - e^{-\frac{s}{v}(2l+x)} - e^{-\frac{s}{v}(4l-x)} + e^{-\frac{s}{v}(4l+x)} + \cdots \right].$$
(257)

Again making use of formula T9, page 74, we obtain e(x,t) in the form

$$e(x,t) = \begin{cases} f\left(t-\frac{x}{v}\right) & \text{when } t > \frac{x}{v} \\ 0 & \text{when } t < \frac{x}{v} \end{cases}$$

$$+ \begin{cases} f\left(t-2T+\frac{x}{v}\right) & \text{when } t > 2T-\frac{x}{v} \\ 0 & \text{when } t < 2T-\frac{x}{v} \end{cases}$$

$$- \begin{cases} f\left(t-2T-\frac{x}{v}\right) & \text{when } t > 2T+\frac{x}{v} \\ 0 & \text{when } t < 2T+\frac{x}{v} \end{cases}$$

$$- \begin{cases} f\left(t-4T+\frac{x}{v}\right) & \text{when } t > 4T-\frac{x}{v} \\ 0 & \text{when } t < 4T-\frac{x}{v} \end{cases}$$

$$+ \begin{cases} f\left(t-4T-\frac{x}{v}\right) & \text{when } t > 4T+\frac{x}{v} \\ 0 & \text{when } t < 4T+\frac{x}{v} \end{cases} + \cdots$$
(258)

where we have written

$$T = \frac{l}{v} = \sqrt{LC} l, \qquad (259)$$

so that T is the time required for a wave to travel the length of the line.

Noticing that $0 \le x/v \le T$, we see that in the first part of the propagation $(0 \le t \le T)$ only the first brace in (258) may differ from zero, and hence in this part of the process the wave described above is traveling toward the end x = l, reaching it at the time t = T. Then when $T \le t \le 2T$, only the first two

braces in (258) are not zero. In this second interval the second brace represents a *reflected* wave returning from the open end of the line without change in sign. The actual voltage in this time interval thus consists of the superposition of an outward- and an inward-traveling wave. When t = 2T, the reflected wave reaches the closed end (x = 0) and, according to the third brace in (258), is again reflected, but this time with reversal of sign. This process is continued indefinitely, reflections without sign change occurring at the open end and reflections with change of sign occurring at the closed end. The various inward and outward waves so generated may combine in various ways, for various periodic impressed end voltages, the nature of the superposition depending, in particular, upon the relationship between this period and the time interval T.

9.16. Formulation of problems. From the preceding examples it is apparent that the *number* and *nature* of the conditions to be imposed along a physical boundary or at a given time depend upon the type of partial differential equation which governs the problem.

Thus, for example, in the case of Laplace's equation in two dimensions, $\varphi_{xx} + \varphi_{yy} = 0$, we have seen that the solution φ is determined everywhere inside a region \mathscr{R} if the boundary values of φ are prescribed along the closed boundary C of \mathscr{R} . The same can be shown to be true also for other elliptic equations, that is, for any equation $a\varphi_{xx} + b\varphi_{xy} + c\varphi_{yy} + \cdots = 0$ where $b^2 < 4ac$. However, this statement is not true in general for hyperbolic equations, where $b^2 > 4ac$. For example, we readily verify that the expression $\varphi = \sin k\pi x \sin k\pi y$ satisfies the equation $\varphi_{xx} - \varphi_{yy} = 0$, as well as the requirement that φ vanish along the closed boundary of the rectangle $[0 \le x \le 1, 0 \le y \le 1]$ for any integral value of k. Thus for this equation this particular boundary-value problem has infinitely many solutions, whereas for Laplace's equation the only solution is that for which $\varphi = 0$ everywhere inside the rectangle.

In the case of the particular *initial-value* problem for which along the entire x axis the function φ and its normal derivative φ_y are prescribed, say

$$\varphi(x,0) = f(x), \qquad \frac{\partial \varphi(x,0)}{\partial y} = g(x),$$

and a solution valid (say) for all positive values of y is required, it can be shown that for Laplace's equation (see Problem 46, Chapter 8) and other elliptic equations, the solution exists only if f(x) and g(x) satisfy very stringent restrictions. In particular, it is necessary that all derivatives of f(x) and g(x)exist for all real values of x, but even this condition is by no means sufficient to guarantee the existence of a solution valid for all positive values of y. On the other hand, unless the x axis is a characteristic base curve, this problem possesses a proper solution in the case of hyperbolic equations when f(x) and g(x) satisfy only very mild conditions. These examples are typical of the general case. Thus it may be expected that, in general, *elliptic* equations are associated with *boundary*-value problems, whereas *hyperbolic* equations are associated with *initial*-value problems. *Parabolic* equations, where the coefficient discriminant $b^2 - 4ac$ vanishes, are intermediate in nature. We next list certain general types of problems which commonly arise in connection with such equations, many of which have been illustrated in the present chapter. The list is not intended to be exhaustive.

Typical problems associated with *elliptic* equations may be illustrated by considering Laplace's equation in the two-dimensional form

$$\frac{\partial^2 T}{\partial x^2} + \frac{\partial^2 T}{\partial y^2} = 0, \qquad (260)$$

where (for example) T may represent steady-state temperature. Along the closed boundary of a region we may prescribe T (the temperature) or the normal derivative $\frac{\partial T}{\partial n}$ (a quantity proportional to the rate of heat flow through the boundary). In the second case the temperature is determined only within an arbitrary additive constant. Alternatively, a condition of the more general type ∂T

$$aT+b\frac{\partial T}{\partial n}=c$$

may be prescribed along the boundary (as, for example, in the case of heat radiation from the boundary according to Newton's law of cooling). In case the region involved is not simply connected, and in certain other cases, it may be necessary to add requirements of single-valuedness or periodicity. The region in which the solution is to be valid may be the interior or the exterior of the closed boundary. However, in case the region extends to infinity, restrictions concerning desirable or permissible behavior of the solution at large distances from the origin generally must be added. In such cases it is frequently convenient to imagine that the conditions "at infinity" are prescribed along a circle (or an arc of a circle) of infinite radius, forming the outer boundary (or the remainder of a boundary which extends to infinity) of the infinite region involved (Figure 9.9).

Typical problems associated with *hyperbolic* equations may be illustrated in the case of the one-dimensional wave equation

$$\frac{\partial^2 w}{\partial x^2} = \frac{1}{c^2} \frac{\partial^2 w}{\partial t^2} , \qquad (261)$$

where (for example) w may represent the deflection of a vibrating string (x is distance along the string, t time). For a string so long that end conditions are irrelevant, the prescribed *initial values*

$$w(x,0) = f(x), \qquad \frac{\partial w(x,0)}{\partial t} = v(x)$$

of the function w (deflection) and w_t (velocity) at the time t = 0 determine the deflection for all values of x at all following times (t > 0). No additional limiting conditions can be prescribed as $t \to \infty$. However, for a string of finite



Figure 9.9

length L with both ends fixed (at x = 0 and x = L) these conditions are prescribed only over the interval 0 < x < L, and the end conditions

$$w(0,t) = 0, \quad w(L,t) = 0$$

must also be satisfied, for all time such that t > 0. In this case, if we represent x and t as rectangular coordinates in an artificial xt plane, we see that the solution is required in the *semi-infinite strip* ($0 < x < L, 0 < t < \infty$) (Figure 9.10). Along the boundaries x = constant, the function w is prescribed, whereas along the "time boundary" t = 0 the quantities w and $\frac{\partial w}{\partial t}$ are prescribed [for those values of x in the interval (0 < x < L)]. With this interpretation, we may think of $\frac{\partial w}{\partial t}$ as the "normal derivative" of w along the "time boundary."

It can be shown (see Problem 28, Chapter 8) that the solution of this problem, when 0 < x < L, is the same as that for an *infinitely long* string extending along the entire x axis, for which at the time t = 0 the deflection and velocity are prescribed for *all* values of x as odd periodic functions of x, of period 2L, agreeing with f(x) and v(x) in the interval 0 < x < L. That is, the effect of the added semi-infinite portions of the string is then the same as the effect of the physical restraints at the ends. Thus, in a sense, this problem can be considered as purely an initial-value problem.

The prescribed value of w along a boundary (end) x = constant may varywith time; also, an end condition of the more general form $aw + bw_x = c$ may be substituted, as for example in the case of an elastic end support. In many problems in which w is prescribed as varying periodically with time at one boundary, a periodic response is eventually propagated throughout the region, and certain transient effects introduced at the beginning of the motion become negligible at sufficiently later times. In such cases it may be that only the limiting periodic response is of interest. Here we may suppose that the



motion was initiated at $t = -\infty$; then only end conditions need be prescribed, and if a periodic solution is obtained it may be considered in this case as valid for all time. (We remark that if the nature of the propagation and the character of the transient effects are of interest, the use of Laplace transforms is particularly advantageous. In this connection a comparison of Problems 62 and 80 at the end of this chapter is suggested.)

Considerations of this general nature (but obviously with entirely different physical interpretations) apply equally

well to the formulation of problems governed by a hyperbolic equation analogous to the one-dimensional wave equation but in which the *time* variable t is replaced by a second *space* variable y. For example (see Section 6.20), in the linearized theory of steady two-dimensional flow of a nonviscous compressible fluid, if the flow is nearly a uniform flow in the x direction, the deviation in velocity from uniformity is the gradient of a potential φ which satisfies the equation

$$(M^2 - 1)\frac{\partial^2 \varphi}{\partial x^2} - \frac{\partial^2 \varphi}{\partial y^2} = 0, \qquad (262)$$

where M is the ratio of the uniform flow velocity to the velocity of sound in the fluid. When this ratio is less than unity, this equation is elliptic and, in fact, is reducible to Laplace's equation by an obvious change in variables. The associated problem in this case must then be a boundary-value problem. However, when the flow is supersonic (M > 1), the equation is hyperbolic, and conditions of entirely different type must be prescribed. (When $M \approx 1$, the equation is not valid, since then the linearization which leads to this equation is not permissible.) Suitable conditions in certain problems of this sort may be obtained by replacing the time variable t by y in the discussion of the preceding paragraph. In such cases the semi-infinite strip inside which the solution is to be obtained may extend to infinity in either the x or the ydirection. In particular, certain basic problems involve the region consisting of the entire half plane y > 0. The nature of prescribed conditions which make a problem determinate is suggested by the consideration that the half plane may be taken as the limit of either of the semi-infinite strips indicated in Figure 9.11, as the dashed-line boundaries are moved parallel to themselves indefinitely far from the origin.



Figure 9.11

In the first limiting case, where the half plane is considered as the limit of a strip extending in the y direction, the problem tends toward the usual initialvalue problem [Figure 9.12(a)] where φ and its normal derivative are *both* prescribed along the entire x axis. Since this problem is completely determined



Figure 9.12

by the conditions along the boundary y = 0, it appears that the "end conditions" along the sides x = constant must be omitted in the limit. That is, generally there are no additional conditions to be prescribed "at infinity" in this limiting case. However, in the second limiting case [Figure 9.12(b)], where the region is considered as the limit of a strip extending in the x direction, the prescribed conditions along the dashed-line boundaries cannot be completely lost in the limit, since the single condition prescribed along the x axis in this case is not sufficient to determine a unique solution. Although the end condition prescribed along the upper boundary y = constant *is* lost in the limit, as in the preceding case, the same is not true for the initial-value conditions prescribed along the boundary $x = x_0$ as $x_0 \to -\infty$. However, the number of conditions prescribable along this line is, in general, reduced from two to one in the limit. In such a problem nothing can be prescribed with reference to the behavior of the solution as $x \to +\infty$. Obviously, the limiting boundary could be taken instead as the line $x = x_0$ as $x_0 \to +\infty$, in which case no conditions



Figure 9.13

could be prescribed as $x \rightarrow -\infty$. A problem of the type just considered is treated in the following section.

Typical problems associated with *par-abolic* equations may be illustrated in the case of the equation

$$\frac{\partial^2 T}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial T}{\partial t}, \qquad (263)$$

which (for example) governs one-dimensional heat flow in the x direction. When the region is infinite in extent $(-\infty < x < +\infty)$, a prescribed distribution T(x,0) at the time t = 0 determines T everywhere at all following times, t > 0. When the region

is of finite extent (0 < x < L), as for example in the case of a thin rod of length L with insulated sides (Figure 9.13), the distribution T(x,0) may be prescribed in this interval when t = 0, and in addition at each end we may prescribe either the temperature T or the rate of heat flow through that end (KT_x) ; or we may prescribe a condition of the more general form $aT + bT_x = c$ at either end, as, for example, in the case of Newtonian cooling by radiation. As a further alternative, the temperature distribution T(x,0) may not be given explicitly but may be determined by specified values of T at the boundaries of the region, under the assumption that initially (at the time t = 0) steady-state conditions prevail. In still another type of problem, a periodically varying source of heat may be present at one boundary, leading to a temperature distribution throughout the region which varies periodically with time and tends to zero at large distances from the boundary (see Problem 51).

The problems described in this section are typical of those commonly occurring in many diverse fields, where only two independent variables are present. We have chosen to identify the dependent variable with, say, temperature or deflection of a string for the purpose of fixing ideas and making intuitively plausible the suitability of the formulation. The above discussion can be extended readily, in most cases, to problems involving a greater number of space variables, the region \mathscr{R} becoming three-dimensional for Laplace's equation, and the x interval becoming a two- or three-dimensional region in space for the wave and heat flow equations and for their analogies. In the latter cases it may happen that an initial condition (at the time t = 0) itself consists of satisfying a partial differential equation (independent of time) together with associated boundary conditions.

9.17. Supersonic flow of ideal compressible fluid past an obstacle. As a further illustration of physical problems involving hyperbolic differential equations we consider the determination

equations, we consider the determination of the two-dimensional flow of a nonviscous compressible fluid past an obstacle in the form of a small protuberance from an otherwise straight wall. The flow is prescribed as having a uniform velocity V_0 parallel to the wall at large distances upstream from the obstacle (Figure 9.14).

The flow velocity vector at any point can be considered as the sum of the uniform velocity vector V_0 i and a small deviation **u**. According to Equations



Figure 9.14

(189) and (194) of Section 6.20, for an *irrotational* flow the deviation **u** is the gradient of a "potential function" φ ,

$$\mathbf{u} = \boldsymbol{\nabla}\varphi,\tag{264}$$

where, for supersonic flow, φ satisfies the equation (262),

$$(M^2-1)\frac{\partial^2 \varphi}{\partial x^2} - \frac{\partial^2 \varphi}{\partial y^2} = 0 \qquad (M>1).$$
 (265)

Here M, the Mach number of the flow, is the ratio of V_0 to the local velocity of sound in the fluid.

The velocity of flow at any point can thus be expressed in the form

$$\mathbf{V} = V_0 \mathbf{i} + \boldsymbol{\nabla} \varphi. \tag{266}$$

Along the boundary comprising the wall and protuberance, the normal component of V must vanish. If we take the equation of this boundary in the form

$$F(x,y) = 0,$$
 (267)

the normal vector is proportional to ∇F , and hence this condition becomes

$$\left[(V_0 \mathbf{i} + \nabla \varphi) \cdot \nabla F \right]_{\text{boundary}} = 0.$$
(268)

By writing (267) now in the explicit form

$$y = H(x) \tag{269}$$

and setting F(x,y) = y - H(x), we find that this condition becomes

$$\left[\left\{\mathbf{i}\left(V_{0}+\frac{\partial\varphi}{\partial x}\right)+\mathbf{j}\frac{\partial\varphi}{\partial y}\right\}\cdot\left\{-\mathbf{i}H'(x)+\mathbf{j}\right\}\right]_{\text{boundary}}=0$$

$$\left[-\left(V_{0}+\frac{\partial\varphi}{\partial x}\right)H'(x)+\frac{\partial\varphi}{\partial y}\right]_{\text{boundary}}=0.$$
(270)

or

The quantity $\frac{\partial \varphi}{\partial x}$ is the x component, u_x , of the velocity deviation; and hence, in accordance with the linearizing approximations leading to (265), $\frac{\partial \varphi}{\partial x}$ is to be neglected with respect to V_0 in (270). Finally, noticing that H(x) is zero except along the protuberance, say

$$H(x) = 0$$
 unless $0 < x < l$, (271)

and assuming that H(x) is *small* in any case, we may further simplify the analysis by satisfying (270) along the *projection* of the protuberance on the x axis. Thus we replace (270) by the *linearized* boundary condition

$$\frac{\partial \varphi(x,0)}{\partial y} = V_0 H'(x). \tag{272}$$

Along any line $x = x_0$, as $x_0 \to -\infty$ the velocity must tend toward uniformity. In particular, the x component of the velocity deviation must tend to zero,

$$\lim_{x_0 \to -\infty} \frac{\partial \varphi(x_0, y)}{\partial x} = 0.$$
 (273)

Thus, we require the solution of (265) which satisfies (272) and (273) in the upper half plane $y \ge 0$. This problem is seen to be a particular case of that indicated in Figure 9.12(b).

If we write

$$\alpha = \sqrt{M^2 - 1},\tag{274}$$

the general solution of (265) can be written in the form

$$\varphi(x,y) = f(x - \alpha y) + g(x + \alpha y), \qquad (275)$$

where f and g are arbitrary functions (see Section 8.5). The condition (272) then becomes

$$-\alpha f'(x) + \alpha g'(x) = V_0 H'(x),$$

from which there follows

$$f(x) = g(x) - \frac{V_0}{\alpha} H(x) + C,$$
 (276)

where C is an arbitrary constant. If we use (276) to eliminate f from (275), there follows

$$\varphi(x,y) = g(x + \alpha y) + g(x - \alpha y) - \frac{V_0}{\alpha} H(x - \alpha y) + C, \qquad (277)$$

and also

$$\frac{\partial \varphi(x,y)}{\partial x} = g'(x+\alpha y) + g'(x-\alpha y) - \frac{V_0}{\alpha} H'(x-\alpha y).$$
(278)

Now let $y \rightarrow +\infty$ along any "characteristic line" $x + \alpha y = c$. Along any of these lines we have, from (278),

$$\frac{\partial \varphi}{\partial x} = g'(c) + g'(c - 2\alpha y) - \frac{V_0}{\alpha} H'(c - 2\alpha y).$$

But since (see Figure 9.15) any such line intersects the boundary $x = x_0 \rightarrow -\infty$ as $y \rightarrow +\infty$, it follows that $\frac{\partial \varphi}{\partial x}$ must then tend to zero along each such line as $y \rightarrow +\infty$, in accordance with (273). Hence we must have, in the limit,

$$g'(c) + g'(-\infty) - \frac{V_0}{\alpha} H'(-\infty) = 0,$$

or, since H(x) = 0 when x < 0 and hence $H'(-\infty) = 0$, we must have

$$g'(c) = -g'(-\infty).$$
 (279)

But since this result must be true for all positive values of c, we then conclude that g'(c) must be a fixed constant independent of c and hence also that $g'(c) = g'(-\infty)$. This statement contradicts (279) unless g'(c) = 0, so that we conclude that the function g is a constant.

Thus, finally, the deviation potential (277) becomes merely

$$\varphi(x,y) = -\frac{V_0}{\alpha}H(x-\alpha y) + A, \qquad (280)$$

where A is an irrelevant constant. The velocity components are then given by

$$V_{x} = V_{0} + \frac{\partial \varphi}{\partial x} = V_{0} \left[1 - \frac{1}{\alpha} H'(x - \alpha y) \right]$$

$$V_{y} = \frac{\partial \varphi}{\partial y} = V_{0} H'(x - \alpha y)$$
 (281)

From these results we conclude that the velocity components are constant along any line $x - \alpha y = constant$. Hence, in particular, the flow is modified



Figure 9.15

Figure 9.16

from uniformity only in the diagonal strip between the two lines $x - \alpha y = 0$ and $x - \alpha y = l$. The streamlines are as indicated in Figure 9.16.

It is important to notice that any irregularity at a point P of the x axis would propagate a corresponding irregularity in the flow along the particular "characteristic line" $x - \alpha y = c$ which passes through the point P (cf. Section 8.10).

REFERENCES

- 1. References at end of Chapters 2, 5, and 8.
- 2. Carslaw, H. S., and J. C. Jaeger, *Conduction of Heat in Solids*, 2nd ed., Oxford University Press, Fair Lawn, N.J., 1959.
- 3. Courant, R., and K. O. Friedrichs, Supersonic Fluid Flow and Shock Waves, Interscience Publishers, Inc., New York, 1948.
- 4. Kellogg, O. D., Foundations of Potential Theory, Dover Publications, Inc., New York, 1953.

PROBLEMS

Section 9.1

1. Verify that the equation

$$\nabla^2 \varphi + 2a \, \frac{\partial \varphi}{\partial x} = \frac{1}{\alpha^2} \, \frac{\partial \varphi}{\partial t}$$

is transformed into the heat flow equation

$$\nabla^2 U = \frac{1}{\alpha^2} \frac{\partial U}{\partial t}$$

by the substitution

$$\varphi = e^{-(ax+a^2\alpha^2t)} U,$$

if a and α are constant.

2. Verify that the equation

$$\nabla^2 \varphi + b\varphi = \frac{1}{\alpha^2} \frac{\partial \varphi}{\partial t}$$

is transformed into the heat flow equation

$$\nabla^2 U = \frac{1}{\alpha^2} \frac{\partial U}{\partial t}$$

by the substitution

$$\varphi = e^{b\alpha^2 t} U,$$

if b and α are constant.

3. If U(x,y,z) satisfies the equation $\nabla^2 \varphi = 0$, show also that the function (ax + by + cz)U(x,y,z) satisfies the equation $\nabla^4 \varphi = 0$ for any constant values of a, b, and c.

4. If U(x,y,z) satisfies Laplace's equation, show that $\frac{\partial U}{\partial x}$, $\frac{\partial U}{\partial y}$, and $\frac{\partial U}{\partial z}$ are also solutions.

5. If $V(r,\theta,z)$ satisfies Laplace's equation (in circular cylindrical coordinates), show that $\frac{\partial V}{\partial \theta}$ and $\frac{\partial V}{\partial z}$ are also solutions but that $\frac{\partial V}{\partial r}$ generally is *not* a solution.

Section 9.2

6. Suppose that heat is flowing in a uniform rod of cross section a and perimeter p, and that it is assumed that the temperature T does not vary over a cross section, and hence is a function only of time t and distance x measured along the rod. Assume also that heat escapes from the lateral boundary by radiation, in such a way that the rate of heat loss per unit of area is $\mu K(T - T_0)$, where K is the conductivity of the rod material, T_0 the temperature of the surrounding medium, and μ is a constant.

(a) By considering differential thermal equilibrium in an element (x, x + dx) of the rod, show that there must follow

$$\frac{\partial}{\partial x}\left(K\frac{\partial T}{\partial x}\right)a\ dx\ -\mu K(T-T_0)p\ dx\ =\ s\rho a\ \frac{\partial T}{\partial t}\ dx,$$

and hence deduce that T then must satisfy the equation

$$\frac{\partial}{\partial x}\left(K\frac{\partial T}{\partial x}\right) = s\rho \,\frac{\partial T}{\partial t} + \mu K \frac{p}{a}(T-T_0).$$

(b) For a rod of circular cross section, of diameter d and of uniform conductivity K, show that T(x,t) must satisfy the equation

$$\frac{\partial^2 T}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial T}{\partial t} + \frac{4\mu}{d} (T - T_0),$$

where $\alpha^2 = K/s\rho$.

(c) Verify that, if T_0 is assumed to remain constant, the substitution

$$T(x,t) = T_0 + U(x,t) e^{-4\mu \alpha^2 t/d}$$

leads to the normal heat flow equation

$$\frac{\partial^2 U}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial U}{\partial t}$$

7. Suppose that the orthogonal coordinates u_1 , u_2 , and u_3 of Section 6.17 are used to specify position in a heat flow problem, and that the conductivity K of the medium is constant.

(a) Show that the heat flow equation (13) becomes

$$\frac{\partial}{\partial u_1}\left(\frac{h_2h_3}{h_1}\frac{\partial T}{\partial u_1}\right) + \frac{\partial}{\partial u_2}\left(\frac{h_3h_1}{h_2}\frac{\partial T}{\partial u_2}\right) + \frac{\partial}{\partial u_3}\left(\frac{h_1h_2}{h_3}\frac{\partial T}{\partial u_3}\right) = \frac{h_1h_2h_3}{\alpha^2}\frac{\partial T}{\partial t}.$$

(b) If \mathbf{F} is the *flux vector*, representing the rate of heat flow per unit area normal to \mathbf{F} , and \mathbf{K} is the thermal conductivity, show that

$$\mathbf{F} = -K \nabla T = -K \left[\frac{\mathbf{u}_1}{h_1} \frac{\partial T}{\partial u_1} + \frac{\mathbf{u}_2}{h_2} \frac{\partial T}{\partial u_2} + \frac{\mathbf{u}_3}{h_3} \frac{\partial T}{\partial u_3} \right].$$

(c) Show that the *streamlines* (to which \mathbf{F} is tangent) are determined by the equations

$$\frac{h_1^2 du_1}{\frac{\partial T}{\partial u_1}} = \frac{h_2^2 du_2}{\frac{\partial T}{\partial u_2}} = \frac{h_3^2 du_3}{\frac{\partial T}{\partial u_3}}.$$

(d) In the case when the temperature and flow are independent of u_3 , and when h_1 , h_2 , and h_3 are independent of u_3 , show that the stream function $\psi(u_1, u_2)$, determined such that

$$\frac{\partial \psi}{\partial u_1} = -K \frac{h_3 h_1}{h_2} \frac{\partial T}{\partial u_2}, \qquad \frac{\partial \psi}{\partial u_2} = +K \frac{h_3 h_2}{h_1} \frac{\partial T}{\partial u_1},$$

where K is the thermal conductivity, has the property that the streamlines in a surface $u_3 = \text{constant}$ are given by $\psi = \text{constant}$.

(e) Show also that the function ψ of part (d) has the additional property that the total rate of heat flow through a surface based on the arc of any curve in a u_3 surface joining points P_1 and P_2 , and extending through a *unit* increment of u_3 , is given by

$$\int_{P_1}^{P_2} \mathbf{F} \cdot \mathbf{n} \ h_3 ds = \int_{P_1}^{P_2} h_3 \mathbf{F} \cdot d\mathbf{r} \times \mathbf{u}_3 = -\int_{P_1}^{P_2} d\psi = \psi(P_1) - \psi(P_2),$$

and that, when also $h_3 = \text{constant}$, the function ψ also satisfies Laplace's equation.

(See also Section 6.19. Notice that sign differences correspond to the fact that the velocity potential in *fluid* flow has been so defined that flow is from lower to higher potential, whereas temperature is so defined that *heat* flows in the direction of *decreasing* temperature.)

8. Suppose that, in a problem of steady-state heat flow, the temperature T is prescribed as zero over a portion of the boundary S of a region \mathcal{R} , the normal

Problems

derivative $\frac{\partial T}{\partial n}$ is zero over another part of S, and finally a condition of the form $\frac{\partial T}{\partial n} + \mu T = 0$ is prescribed over the remainder of S, say S', where μ is a *positive* constant or function of position on S'.

(a) Show that Equation (16) takes the form

$$\iiint_{\mathscr{R}} (\nabla T)^2 \, d\tau = - \oiint_{S'} \mu T^2 \, d\sigma,$$

and deduce that T then must vanish throughout \mathcal{R} .

(b) Use this result to show that the solution of Laplace's equation $\nabla^2 T = 0$ is uniquely determined in a region \mathscr{R} if either T, $\frac{\partial T}{\partial n}$, or $\frac{\partial T}{\partial n} + \mu T (\mu > 0)$ is prescribed at each point of the closed boundary, except in the case when $\frac{\partial T}{\partial n}$ is prescribed over the complete boundary (in which case an arbitrary additive constant is present in the solution).

9. Suppose that the heat flow equation

$$\nabla^2 T = \frac{1}{\alpha^2} \frac{\partial T}{\partial t}$$

is to be satisfied throughout a region \mathcal{R} bounded by a closed surface S, for all positive values of time t.

(a) Show that Equation (16) then is replaced by the more general form

$$\int \int \int_{\mathscr{R}} (\nabla T)^2 d\tau = -\frac{1}{2\alpha^2} \frac{\partial}{\partial t} \int \int \int_{\mathscr{R}} T^2 d\tau + \bigoplus_{S} T \frac{\partial T}{\partial n} d\sigma$$

(b) Suppose that either T or $\frac{\partial T}{\partial n}$ is maintained as zero at all points of S, for

all time t > 0, and that T is initially zero throughout \mathcal{R} when t = 0. Show that there must follow

$$\int_0^t \left[\int \int \int_{\mathscr{R}} (\nabla T)^2 \, d\tau \right] dt = -\frac{1}{2\alpha^2} \int \int \int_{\mathscr{R}} T^2 \, d\tau$$

for all t > 0. Hence deduce that T must vanish throughout \mathcal{R} for all t > 0.

(c) Use the result of part (b) to show that the solution of the heat flow equation is uniquely determined for t > 0 if T is prescribed in \mathscr{R} when t = 0 and either T or $\frac{\partial T}{\partial n}$ is prescribed on the boundary S for all t > 0.

10. Generalize the result of Problem 9 to the case when either T, $\frac{\partial T}{\partial n}$, or the combination $\frac{\partial T}{\partial n} + \mu T (\mu > 0)$ is prescribed at each point of the boundary S for t > 0 and T is prescribed throughout \Re when t = 0. (See also Problem 8.)

Section 9.3

11. The temperature T is maintained at 0° along three edges of a square plate of length 100 cm, and the fourth edge is maintained at 100° until steady-state conditions prevail. (Small areas near two corners must be considered as excluded.)

(a) Find an expression for the temperature T at any point (x, y) in the plate, using the notation of Section 9.3.

(b) Calculate the approximate value of the temperature at the center of the plate.

12. Suppose that the plate of Section 9.3 is of infinite extent in the y direction, on one side of the boundary y = 0, so that it occupies the semi-infinite strip $0 \le x \le L$, $0 \le y < \infty$.

(a) If the temperature is to vanish on the lateral boundaries x = 0 and x = L, is to tend to zero as $y \to \infty$, and is to reduce to f(x) along the edge y = 0, obtain the temperature distribution in the form

$$T(x,y) = \sum_{n=1}^{\infty} c_n e^{-n\pi v/L} \sin \frac{n\pi x}{L}, \qquad c_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} \, dx.$$

(b) Obtain the same result formally from Equations (34) and (35), by first replacing y by H - y and then considering the limit as $H \rightarrow \infty$.

13. Let Problem 12 be modified in such a way that the rate of heat flow, per unit distance, into the plate through points of the boundary y = 0 is prescribed as g(x), where g(x) may be measured in calories per second per centimeter length along the boundary y = 0.

(a) Show that the condition along the line y = 0 then is of the form

$$-Kh\,\frac{\partial T(x,0)}{\partial y}=g(x),$$

where *h* is the thickness of the plate.

(b) Obtain the solution in the form

$$T(x,y) = \sum_{n=1}^{\infty} c_n e^{-n\pi y/L} \sin \frac{n\pi x}{L}, \qquad c_n = \frac{2}{n\pi Kh} \int_0^L g(x) \sin \frac{n\pi x}{L} dx.$$

14. Suppose that g(x) is prescribed in the form

$$g(x) = \begin{cases} 0 & (0 < x < x_0 - \epsilon) \\ \frac{1}{2\epsilon} & (x_0 - \epsilon < x < x_0 + \epsilon) \\ 0 & (x_0 + \epsilon < x < L). \end{cases}$$

(a) Show that the coefficients in Problem 13(b) are then given by

$$c_n = \frac{2L}{n^2 \pi^2 K h \epsilon} \sin \frac{n \pi \epsilon}{L} \sin \frac{n \pi x_0}{L}.$$

(b) By considering the limit of c_n as $\epsilon \to 0$, so that the function g(x) tends to the unit singularity function $\delta(x - x_0)$, obtain the temperature distribution due to a permanent *unit heat source*, supplying one calorie of heat per second at the boundary point $x = x_0$, in the form

$$T(x,y) = \frac{2}{\pi Kh} \sum_{n=1}^{\infty} \frac{1}{n} \sin \frac{n\pi x_0}{L} \sin \frac{n\pi x}{L} e^{-n\pi y/L}.$$

[Notice that this result also can be obtained directly from the result of Problem 13, by making use of the fact that

$$\int_{a}^{b} \delta(x - x_{0}) \varphi(x) dx = \varphi(x_{0}) \qquad (a < x_{0} < b),$$

for any continuous function φ .]

(c) If the solution of part (b) is denoted by $V(x,y;x_0)$, show that the solution of Problem 13 for any prescribed g(x) can be written in the form

$$T(x,y) = \int_0^L V(x,y;x_0) g(x_0) \, dx_0.$$

Interpret this relation.

15. (a) If f(x) is identified with $\delta(x - x_0)$ in Problem 12, show that the solution of that problem becomes

$$T(x,y) = \frac{2}{L} \sum_{n=1}^{\infty} \sin \frac{n\pi x_0}{L} \sin \frac{n\pi x}{L} e^{-n\pi y/L}.$$

(b) If this solution is denoted by $U(x,y;x_0)$, show that the solution of Problem 12 for any prescribed f(x) can be written in the form

$$T(x,y) = \int_0^L U(x,y;x_0) f(x_0) \, dx_0.$$

(c) Verify that, with the notations of Problems 14(c) and 15(b), there follows

$$U(x,y;x_0) = -Kh \frac{\partial V(x,y;x_0)}{\partial y},$$

and hence that $U(x,y;x_0)$ is the distribution due to a permanent *heat doublet* of strength *Kh* located at $x = x_0$, the doublet being formed by the confluence of a heat source at $(x_0,0)$ and a heat sink at (x_0,ϵ) , each of strength *Kh*/ ϵ , as $\epsilon \to 0$.

16. If Problem 12 is modified in such a way that the lateral edges (x = 0 and x = L) of the strip are *insulated* (so that $\frac{\partial T}{\partial n} = \pm \frac{\partial T}{\partial x}$ vanishes along those lines), obtain the temperature distribution in the strip in the form

$$T(x,y) = c_0 + \sum_{n=1}^{\infty} c_n e^{-n\pi y/L} \cos \frac{n\pi x}{L},$$

where

$$c_0 = \frac{1}{L} \int_0^L f(x) \, dx, \qquad c_n = \frac{2}{L} \int_0^L f(x) \cos \frac{n \pi x}{L} \, dx (n = 1, 2, \ldots).$$

17. By imposing the condition $\nabla^2 \varphi = 0$ on an expression of the form

$$\varphi = A_0 + A_1 x + A_2 y + A_3 x^2 + A_4 x y + A_5 y^2 + A_6 x^2 + \cdots,$$

deduce that *polynomial solutions of Laplace's equation* are specializations of the expression

$$\varphi = a_0 + a_1 x + a_2 y + a_3 x y + a_4 (x^2 - y^2) + a_5 (x^3 - 3xy^2) + a_6 (y^3 - 3x^2 y) + \cdots$$

(See also Problem 22 of Chapter 8.)

18. Find the solution T(x,y) of Laplace's equation in the semi-infinite strip $0 \le x \le L, 0 \le y < \infty$, subject to the conditions

$$T(0,y) = \alpha_1 + \beta_1 y, \qquad T(L,y) = \alpha_2 + \beta_2 y, \qquad T(x,0) = f(x).$$

[Suggestion: Write T = p(x,y) + u(x,y), where p(x,y) is a polynomial solution of Laplace's equation, satisfying the conditions prescribed along x = 0 and x = L, and obtained from the result of Problem 17 in the form

$$p = \alpha_1 + (\alpha_2 - \alpha_1)\frac{x}{L} + \beta_1 y + (\beta_2 - \beta_1)\frac{xy}{L},$$

and where the correction u(x,y) accordingly also satisfies Laplace's equation, but is subject to homogeneous conditions u(0,y) = 0, u(L,y) = 0 along x = 0 and x = L, together with the initial condition $u(x,0) = f(x) - \alpha_1 - (\alpha_2 - \alpha_1)(x/L)$.]

Section 9.4

19. Along the inner boundary of a circular annulus of radii 10 cm and 20 cm the temperature is maintained as $T(10,\theta) = 15 \cos \theta$, and along the outer boundary the distribution $T(20,\theta) = 30 \sin \theta$ is maintained. Find an expression for the steady-state temperature at an arbitrary point (r,θ) in the annulus.

20. Along the circumference of the unit circle r = 1 a solution of Laplace's equation is required to take on the value unity when $0 < \theta < \pi$ and the value zero when $\pi < \theta < 2\pi$.

- (a) Determine an expression for T valid when r < 1.
- (b) Determine a corresponding expression valid when r > 1.

21. (a) Determine the steady-state temperature at points of the sector $0 \le \theta \le \alpha$, $0 \le r \le a$, of a circular plate if the temperature is maintained at zero along the straight edges and at a prescribed distribution $T(a,\theta) = f(\theta)$ when $0 < \theta < \alpha$, along the curved edge.

(b) In the special case when $f(\theta) = T_0$ = constant, show that the temperature at interior points is given by

$$T(r,\theta) = \frac{4T_0}{\pi} \sum_{n \text{ odd}} \frac{1}{n} \left(\frac{r}{a}\right)^{n\pi/\alpha} \sin \frac{n\pi\theta}{\alpha}.$$

22. (a) Show that the solution of Problem 21(b) tends to the form

$$T(r,\theta) = \frac{4T_0}{\pi} \sum_{n \text{ odd}} \frac{1}{n} \left(\frac{r}{a}\right)^{n/2} \sin \frac{n\theta}{2}$$

as the opening angle α of the sector tends to 2π , so that the sector fans out into the interior of the complete circle with a *cut* along the radius which coincides with the positive x axis.

(b) Noticing that $\theta \to 0$ as the cut is approached from above, whereas $\theta \to 2\pi$ as the cut is approached from below, show that the interior temperature distribution of part (a) is *continuous* across the cut.

(c) Show that the derivative of the temperature in the positive y direction tends to (r) -1/2

$$\left[\frac{1}{r}\frac{\partial T}{\partial \theta}\right]_{\theta=0} = \frac{2T_0}{\pi a}\sum_{n \text{ odd}} \left(\frac{r}{a}\right)^{n/2-1} = \frac{2T_0}{\pi}\frac{\binom{r}{a}}{a-r} \qquad (0 < r < a)$$

as the cut is approached from above, whereas it tends to the *negative* of that quantity as the cut is approached from below. (Since $-\frac{\partial T}{\partial y}$ is proportional to the rate of heat flow in the positive y direction, this means merely that heat must be continually drawn off from the plate at all points of the cut in order that the desired temperature distribution may be maintained.)

23. The function $\varphi(r,\theta)$ is required subject to the conditions that φ satisfy Laplace's equation inside the circle r = a and at all finite points outside that circle, that φ be continuous across the circle r = a, that $\frac{\partial \varphi}{\partial r}$ decrease abruptly by a prescribed function $F(\theta)$ as r increases through a, and that $\frac{\partial \varphi}{\partial r}$ remain finite as $r \to \infty$.

(a) Show that φ may be assumed in the form

$$\varphi(r,\theta) = A_0 + \begin{cases} \sum_{n=1}^{\infty} \left(\frac{r}{a}\right)^n (A_n \cos n\theta + B_n \sin n\theta) & (0 \le r \le a) \\ C \log \frac{a}{r} + \sum_{n=1}^{\infty} \left(\frac{a}{r}\right)^n (A_n \cos n\theta + B_n \sin n\theta) & (a \le r < \infty). \end{cases}$$

(b) Show that the additive constant A_0 is arbitrary, and that the remaining coefficients are given by

$$C = \frac{a}{2\pi} \int_0^{2\pi} F(\theta) \, d\theta, \qquad A_n = \frac{a}{2n\pi} \int_0^{2\pi} F(\theta) \cos n\theta \, d\theta,$$
$$B_n = \frac{a}{2n\pi} \int_0^{2\pi} F(\theta) \sin n\theta \, d\theta \qquad (n = 1, 2, \ldots).$$

24. Suppose that the outer boundary $r = r_2$ of an annular plate is insulated, and that the temperature is prescribed as $f(\theta)$ along the inner boundary $r = r_1$.

(a) Show that the expression (48) must be specialized to the form

$$T = a_0 + \sum_{n=1}^{\infty} (r^n + r_2^{2n} r^{-n})(a_n \cos n\theta + c_n \sin n\theta)$$

when $r_1 \leq r \leq r_2$.

(b) By writing $A_n = r_1^n a_n$ and $C_n = r_1^n c_n$, and introducing the abbreviation $\beta = r_2/r_1$, obtain the desired temperature distribution in the annulus in the form

$$T = A_0 + \sum_{n=1}^{\infty} \left[\left(\frac{r}{r_1} \right)^n + \beta^{2n} \left(\frac{r_1}{r} \right)^n \right] (A_n \cos n\theta + C_n \sin n\theta),$$

where

$$A_0 = \frac{1}{2\pi} \int_0^{2\pi} f(\theta) \, d\theta, \quad A_n = \frac{\int_0^{2\pi} f(\theta) \cos n\theta \, d\theta}{\pi(1+\beta^{2n})}, \quad C_n = \frac{\int_0^{2\pi} f(\theta) \sin n\theta \, d\theta}{\pi(1+\beta^{2n})}$$

- -

Section 9.5

25. By making use of the relation

$$\int_0^{2\pi} \frac{d\varphi}{1 - A\cos\varphi} = \frac{2\pi}{\sqrt{1 - A^2}} \qquad (|A| < 1),$$

obtain the result

$$\frac{1}{2\pi}\int_0^{2\pi}\frac{a^2-r^2}{a^2-2ar\cos{(\theta-\varphi)}+r^2}d\varphi = 1 \qquad (|r| < a),$$

and thus verify directly the validity of the Poisson integral formula (64) in the case when $T(a,\theta) = T_0$ = constant. (Notice that the integrand is periodic in φ .)

26. Use the Poisson integral formula (64) to show that, when $0 \le r \le a$,

$$T(r,\theta) \leq \frac{1}{2\pi} \int_0^{2\pi} \frac{a^2 - r^2}{a^2 - 2ar\cos\left(\theta - \varphi\right) + r^2} [T(a,\varphi)]_{\max} d\varphi \leq [T(a,\theta)]_{\max},$$

and hence, by considering also the corresponding inequality for $-T(r,\theta)$, deduce that if T satisfies Laplace's equation inside a circle, then T cannot take on its maximum or minimum value inside the boundary unless it is constant throughout the circle. (The same result can be established for an arbitrary finite region.)

Section 9.6

27. (a) The temperature T on the surface of a sphere of radius a is maintained as $T = T_0(1 - \cos \varphi)$, where φ is the cone angle. Find the steady-state temperature at an arbitrary point inside the sphere.

(b) If, instead, the temperature is maintained at a constant value T_0 over the upper hemisphere ($0 \le \varphi < \pi/2$) and at zero over the remainder of the surface, determine the first three terms in the series solution of Section 9.6.

28. If the temperature of a spherical surface r = a is maintained at $T(a,\varphi) = f(\varphi)$, where φ is the cone angle, and if the temperature tends to zero as $r \to \infty$, show that the temperature distribution *outside* the sphere is given by

$$T(r,\varphi) = \sum_{n=0}^{\infty} c_n \left(\frac{a}{r}\right)^{n+1} P_n(\cos\varphi),$$

where c_n is defined by Equation (82).

29. If a spherical surface r = a is maintained at constant temperature T_0 , show that the temperature at all internal points is also T_0 , whereas the temperature at points outside the sphere is given by $T_0 a/r$ ($r \ge a$).

30. Suppose that the steady-state temperature T in a solid right circular cylinder possesses axial symmetry, and hence is of the form T = T(r,z), where r is distance from the z axis.

(a) Show that T then must satisfy the equation

$$\frac{1}{r}\frac{\partial}{\partial r}\left(r\frac{\partial T}{\partial r}\right)+\frac{\partial^2 T}{\partial z^2}=0$$

inside the cylinder and that, if a product solution is assumed in the form T(r,z) = R(r)Z(z), then R and Z must satisfy equations of the form

$$rrac{d^2R}{dr^2}+rac{dR}{dr}+lpha rR=0,\qquad rac{\partial^2 Z}{dz^2}-lpha Z=0,$$

where α is an arbitrary constant.

(b) If only solutions which remain finite along the z axis are admissible, show that the choice $\alpha = k^2$, where k is real and positive, leads to exponential or hyperbolic functions of z and to the function $J_0(kr)$, whereas the choice $\alpha = -k^2$ leads to circular functions of z and to the function $I_0(kr)$. Finally, show that the choice $\alpha = 0$ leads to $R = c_1 \log r + c_2$ and $Z = d_1 z + d_2$.

31. (a) Suppose that a solid right circular cylinder of radius a is of infinite extent on one side of the plane face z = 0, and that the temperature is maintained at zero along the lateral boundary, whereas the temperature distribution over the face z = 0 is prescribed as T(r,0) = f(r). Show that the steady-state temperature at interior points is given by

$$T(r,z) = \sum_{n=1}^{\infty} A_n e^{-k_n z} J_0(k_n r),$$

where k_n is the *n*th positive root of the equation $J_0(ka) = 0$, and where

$$A_n = \frac{2}{a^2 [J_1(k_n a)]^2} \int_0^a r f(r) J_0(k_n r) dr.$$

(b) If the temperature over the face z = 0 is maintained at a constant value T_0 , show that the resultant interior distribution is given by

$$T(r,z) = \frac{2T_0}{a} \sum_{n=1}^{\infty} e^{-k_n z} \frac{J_0(k_n r)}{k_n J_1(k_n a)}.$$

32. Suppose that the faces z = 0 and z = L of a solid right circular cylinder are maintained at temperature zero, and that the temperature distribution along the lateral boundary r = a is dependent only on z, and is prescribed in the form

T(a,z) = f(z). Obtain the resultant steady-state temperature distribution inside the cylinder in the form

$$T(r,z) = \sum_{n=1}^{\infty} A_n \frac{I_0\left(\frac{n\pi r}{L}\right)}{I_0\left(\frac{n\pi a}{L}\right)} \sin \frac{n\pi z}{L},$$

where

$$A_n = \frac{2}{L} \int_0^L f(z) \sin \frac{n\pi z}{L} \, dz.$$

Section 9.7

33. Suppose that a column with rectangular cross section can be considered to be of infinite height on one side of its base $0 \le x \le L_1$, $0 \le y \le L_2$ in the plane z = 0. If the lateral boundaries are maintained at zero temperature and the temperature distribution over the base is prescribed as T(x,y,0) = f(x,y), obtain the internal temperature distribution in the form

$$T(x,y,z) = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} c_{mn} e^{-k_{mn}z} \sin \frac{m\pi x}{L_1} \sin \frac{n\pi y}{L_2},$$

where k_{mn} and c_{mn} are defined by Equations (100) and (106).

34. If the column considered in Problem 33 is of *circular* cross section, of radius a, and if the temperature over the base is prescribed as $T(r,\theta,0) = F(r,\theta)$, show that the internal temperature distribution is given by

$$T(r,\theta,z) = \sum_{n=1}^{\infty} a_{0n} e^{-k_{0n}z} J_0(k_{0n}r) + \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} e^{-k_{mn}z} J_m(k_{mn}r)(a_{mn}\cos n\theta + b_{mn}\sin n\theta),$$

where k_{mn} is the *n*th positive root of the equation $J_m(ka) = 0$, and where a_{mn} and b_{mn} are to be determined in such a way that

$$\sum_{n=1}^{\infty} \left[a_{0n} J_0(k_{0n}r) + \sum_{m=1}^{\infty} (a_{mn} \cos n\theta + b_{mn} \sin n\theta) J_m(k_{mn}r) \right] = F(r,\theta)$$

when 0 < r < a and $0 < \theta < 2\pi$.

35. By multiplying the equal members of the last equation of Problem 34 by

$$J_p(k_{pq}r) \sin q\theta r dr d\theta$$
,

where p and q are any positive integers, and integrating the result over the base of the column, deduce formally that

$$b_{pq} \frac{\pi a^2}{2} [J_{p+1}(k_{pq}a)]^2 = \int_0^{2\pi} \int_0^a J_p(k_{pq}r) F(r,\theta) \sin q\theta r \, dr \, d\theta.$$

(The remaining coefficients can be determined in a similar way.)

Section 9.8

36. Let a right circular cylinder of radius *a* be placed in an initially uniform flow of an ideal incompressible fluid, in such a way that the axis of the cylinder coincides with the *z* axis and the flow tends to a uniform flow with velocity V_0 in the *x* direction at large distances from the cylinder. Assume that the cylinder can be considered to be of infinite length.

(a) If polar coordinates are used in the xy plane of flow, show that the velocity vector V is related to the velocity potential $\varphi(r,\theta)$ by the equation

$$\mathbf{V} = \mathbf{u}_r \, \frac{\partial \varphi}{\partial r} + \mathbf{u}_\theta \, \frac{1}{r} \, \frac{\partial \varphi}{\partial \theta} \,,$$

and that φ must satisfy the equation

$$\frac{1}{r}\frac{\partial}{\partial r}\left(r\frac{\partial\varphi}{\partial r}\right) + \frac{1}{r^2}\frac{\partial^2\varphi}{\partial\theta^2} = 0$$

and the boundary conditions

$$\frac{\partial \varphi(a,\theta)}{\partial r} = 0, \qquad \frac{\partial \varphi(\infty,\theta)}{\partial r} = V_0 \cos \theta,$$

in addition to the requirement that $V_r = \frac{\partial \varphi}{\partial r}$ be a single-valued function of θ .

(b) By imposing these restrictions on an expression of form (45), obtain the velocity potential in the form

$$\varphi(r,\theta) = c_0 + k\theta + V_0\left(r + \frac{a^2}{r}\right)\cos\theta,$$

where c_0 and k are arbitrary constants, and show that the velocity vector then is given by

$$\mathbf{V} = \left[\mathcal{V}_0 \left(1 - \frac{a^2}{r^2} \right) \cos \theta \right] \mathbf{u}_r - \left[\mathcal{V}_0 \left(1 + \frac{a^2}{r^2} \right) \sin \theta \right] \mathbf{u}_\theta + \frac{k}{r} \mathbf{u}_\theta.$$

(c) Verify that V_r vanishes when r = a and that V tends to $V_0 i$ as $r \to \infty$, regardless of the value of the constant k.

37. (a) Show that the stream function $\psi(r,\theta)$ corresponding to the flow obtained in Problem 36(b) can be expressed in the form

$$\psi(r,\theta) = d_0 - k \log r + V_0 \left(r - \frac{a^2}{r}\right) \sin \theta,$$

where d_0 is an arbitrary constant, and hence that the streamlines are defined by the equation

$$V_{0,V}\left(1 - \frac{a^2}{x^2 + y^2}\right) - \frac{k}{2}\log(x^2 + y^2) = \text{constant},$$

in rectangular coordinates.

(b) Show that the flow is symmetrical about the x axis when k = 0 and sketch typical streamlines in this case.

(c) Show that the added velocity potential $k\theta$ (which is not single-valued) corresponds to a *circulation* of the fluid about the cylinder, in which the velocity V_c is circumferential, with magnitude k/r at distance r from the center of the cylinder section, and that the circulation $\oint_C V_c \cdot d\mathbf{r}$ is given by $2\pi k$, around any circle r = constant.

(Notice that the flow is not determined unless the amount of circulation is prescribed, *in addition to* the values of V, at r = a and of V as $r \to \infty$. Notice also that the flow is indeed vortex-free, since the point r = 0 about which the fluid tends to rotate is not a point in the fluid itself.)

38. The "stagnation points" of a flow are the points at which the flow velocity is zero.

(a) Show that when no circulation is present the stagnation points in the flow of Problem 36 are the points (a,0) and (a,π) at the intersections of the x axis and the boundary of the cylinder section.

(b) With the abbreviation $\alpha = k/(2V_0a)$, show that when circulation is present the stagnation points are $(a, \sin^{-1} \alpha)$ and $(a, \pi - \sin^{-1} \alpha)$ when $|\alpha| < 1$. Show also that when $|\alpha| \ge 1$ there is only one stagnation point in the flow, and that that point is at a distance $a(|\alpha| + \sqrt{\alpha^2 - 1})$ from the center of the cylinder section, on the positive or negative y axis, according to whether α is positive or negative.

Section 9.9

39. By multiplying both sides of Equation (141a) by

$$J_p\left(\frac{\omega_{pq}r}{\alpha}\right)\cos p\theta \cdot r \, dr \, d\theta$$

and integrating over the area of the membrane, obtain the results

$$a_{mn} \frac{\pi a^2}{2} \left[J_{m+1} \left(\frac{\omega_{mn} a}{\alpha} \right) \right]^2 = \int_0^{2\pi} \int_0^a J_m \left(\frac{\omega_{mn} r}{\alpha} \right) f_1(r,\theta) \cos m\theta \cdot r \, dr \, d\theta \qquad (m \neq 0).$$

and

$$a_{0n} \pi a^2 \left[J_1\left(\frac{\omega_{0n}a}{\alpha}\right) \right]^2 = \int_0^{2\pi} \int_0^a J_0\left(\frac{\omega_{0n}r}{\alpha}\right) f_1(r,\theta) \cdot r \, dr \, d\theta \qquad (m=0).$$

(The coefficients b_{mn} , where $m \ge 1$, are determined in a similar way.)

40. For a freely vibrating square membrane of length L, supported along the boundary x = 0, x = L, y = 0, y = L, obtain a permissible expression for the deflection w(x,y,t) by methods analogous to those of Section 9.9, in the form

$$w = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} (a_{mn} \cos \omega_{mn} t + b_{mn} \sin \omega_{mn} t)$$
$$\omega_{mn} = \pi \sqrt{m^2 + n^2} \sqrt{\frac{T}{\rho L^2}}.$$

where

Problems

41. (a) Suppose that the square membrane of Problem 40 is initially deflected in the form

$$w(x,y,0) = f(x,y)$$

and is released from rest. Obtain an expression for the ensuing deflection in the form

$$w = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} a_{mn} \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} \cos \omega_{mn} t$$
$$a_{mn} = \frac{4}{L^2} \int_0^L \int_0^L f(x,y) \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} \, dx \, dy.$$

where

(b) Obtain an expression for w in Problem 40 if initially the membrane is undeflected but if the membrane is in motion such that

$$\frac{\partial \omega(x,y,0)}{\partial t} = v(x,y).$$

42. (a) From the results of Problems 40 and 41, show that free vibrations of a square membrane are compounded of "natural modes" of the form

$$w_{mn}(x,y,t) = A_{mn} \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} \cos \left(\omega_{mn} t + \alpha_{mn} \right),$$

with circular frequencies

$$\omega_{mn} = \pi \sqrt{m^2 + n^2} \sqrt{\frac{T}{\rho L^2}} \equiv \sqrt{\frac{m^2 + n^2}{2}} \Omega$$

where $\Omega = \pi \sqrt{(2T)/(\rho L^2)}$ is the fundamental circular frequency, and with corresponding amplitudes

$$\varphi_{mn}(x,y) = A_{mn} \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L},$$

where m and n are positive integers.

(b) Deduce that in the fundamental natural mode ($\omega = \omega_{11} = \Omega$) the amplitude

$$\varphi = a \sin \frac{\pi x}{L} \sin \frac{\pi y}{L}$$

is zero only along the boundary, so that there are no interior nodal lines.

(c) Show that the second natural circular frequency corresponds to *two* modes, in which either m = 1 and n = 2 or m = 2 and n = 1, and is given by $\omega = \omega_{12} = \omega_{21} = \sqrt{10} \Omega/2$, and that the most general motion with this circular frequency possesses an amplitude of the form

$$\varphi = a \sin \frac{2\pi x}{L} \sin \frac{\pi y}{L} + b \sin \frac{\pi x}{L} \sin \frac{2\pi y}{L}$$
$$= 2 \sin \frac{\pi x}{L} \sin \frac{\pi y}{L} \left(a \cos \frac{\pi x}{L} + b \cos \frac{\pi y}{L} \right),$$

where a and b are constants. Deduce that there is one interior nodal line, along the curve $a \cos \pi x/L + b \cos \pi y/L = 0$, that this curve always passes through the center of the square, and that in the cases $b = \pm a$ the line is a diagonal of the square.

Section 9.10

43. The temperatures at the ends x = 0 and x = 100 of a rod 100 cm in length, with insulated sides, are held at 0° and 100°, respectively, until steady-state conditions prevail. Then, at the instant t = 0, the temperatures of the two ends are interchanged. Find the resultant temperature distribution as a function of x and t.

44. (a) Obtain permissible product solutions of the heat flow equation (147) which satisfy the conditions $\frac{\partial T(0,t)}{\partial x} = 0$, $\frac{\partial T(L,t)}{\partial x} = 0$, corresponding to the requirement that there be no heat flow through the boundaries (ends) x = 0 and x = L.

(b) Use these results to solve the modification of Problem 43 in which at the time t = 0 the two ends of the rod are suddenly insulated.

45. Suppose that the rod considered in Section 9.10 is such that heat escapes from the lateral boundary, according to Newton's law of cooling, so that T(x,t) satisfies the equation

$$\alpha^2 \frac{\partial^2 T}{\partial x^2} = \frac{\partial T}{\partial t} + \beta (T - T_0),$$

where β is a constant and T_0 is the temperature of the surrounding medium. (See Problem 6.) The initial temperature distribution is T(x,0) = f(x) and the ends x = 0 and x = L are maintained at T_1 and T_2 , respectively, when t > 0.

(a) Show that the substitution

$$T(x,t) = T_0 + U(x,t) e^{-\beta t}$$

reduces the problem to the following one:

$$\frac{\partial^2 U}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial U}{\partial t},$$
$$U(x,0) = f(x) - T_0, \quad U(0,t) = (T_1 - T_0) e^{\beta t}, \quad U(L,t) = (T_2 - T_0) e^{\beta t}.$$

(b) In the important special case when $T_1 = T_2 = T_0$, obtain the solution of the original problem in the form

$$T(x,t) = T_0 + \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} e^{-[(\beta + n^2 \pi^2 \alpha^2)/L^2]t},$$

where

$$a_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n \pi x}{L} \, dx - \frac{1 - \cos n \pi}{n \pi} \, 2T_0$$

46. A rod of length L, with insulated lateral boundaries, has the end x = 0 maintained at $T = T_1$ when t > 0, whereas heat escapes through the end x = L according to Newton's law of cooling in the form

$$\left[hL\frac{\partial T}{\partial x}+(T-T_0)\right]_{x=L}=0,$$

where T_0 is the temperature of the surrounding medium and h is a constant. The initial temperature distribution along the rod is prescribed as T(x,0) = f(x).

(a) Obtain the steady-state distribution in the form

$$T_S = T_1 - \frac{T_1 - T_0}{1 + h} \frac{x}{L}.$$

(b) Show that the transient distribution can be assumed in the form

$$T_T(x,t) = \sum_{n=1}^{\infty} a_n \sin \frac{k_n x}{L} e^{-k_n^2 \alpha^2 t/L^2},$$

where k_n is the *n*th positive root of the equation $\tan k + hk = 0$.

(c) Obtain the required temperature distribution in the form $T = T_S + T_S$ T_T , where a_n is determined by the relation

$$a_n \int_0^L \sin^2 \frac{k_n x}{L} dx = \int_0^L \left[f(x) - T_S(x) \right] \sin \frac{k_n x}{L} dx.$$

47. (a) Show that the assumption of a solution of the one-dimensional heat flow equation as a linear function of t, in the form T = t f(x) + g(x), leads to the requirements f'' = 0 and $\alpha^2 g'' = f$. Hence obtain the particular solution

$$T_p(x,t) = c_1(x^3 + 6\alpha^2 tx) + c_2(x^2 + 2\alpha^2 t) + c_3 x + c_4,$$

where the c's are arbitrary constants.

(b) Use this result to obtain a particular solution for which $\frac{\partial T(0,t)}{\partial r} = 0$ and $\frac{\partial T(L,t)}{\partial r} = C \text{ in the form}$

$$T_p = \frac{C}{2L} (x^2 + 2\alpha^2 t).$$

(c) In a similar way, obtain a particular solution for which T(0,t) = 0 and $T(L,t) = T_0 t$ in the form

$$T_{p} = T_{0} \left[\frac{x}{L} t - \frac{1}{6\alpha^{2}L} x(L^{2} - x^{2}) \right].$$

48. A rod with insulated sides has its end x = 0 insulated, whereas heat is introduced into the end x = L at a constant rate, so that $\frac{\partial T(L,t)}{\partial x} = C$, where C is a constant. The initial temperature distribution is prescribed as T(x,0) = f(x).

(a) Show that no steady state can exist in which T is independent of time, but that [see Problem 47(b)] the temperature distribution $T_p = C(x^3 + 2\alpha^2 t)/2L$ satisfies the end conditions (and also happens to specify a state in which the rate of *heat flow* at any point does not vary with time).

(b) Obtain the desired distribution in the form

$$T(x,t) = \frac{C}{2L} (x^2 + 2\alpha^2 t) + a_0 + \sum_{n=1}^{\infty} a_n \cos \frac{n\pi x}{L} e^{-n^2 \pi^2 \alpha^2 t/L^2},$$
where

$$a_{0} = \frac{1}{L} \int_{0}^{L} \left[f(x) - \frac{C}{2L} x^{2} \right] dx, \quad a_{n} = \frac{2}{L} \int_{0}^{L} \left[f(x) - \frac{C}{2L} x^{2} \right] \cos \frac{n\pi x}{L} dx,$$

when n = 1, 2, ...

49. Use the result of Problem 47(c) to obtain the solution of the one-dimensional heat flow equation for which T(0,t) = 0 and $T(L,t) = T_0 t$ when t > 0 and T(x,0) = f(x), in the form

$$T(x,t) = T_0 \left[\frac{x}{L} t - \frac{1}{6\alpha^2 L} x(L^2 - x^2) \right] + \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} e^{-n^2 \pi^2 \alpha^2 t/L^2},$$

where

$$a_n = \frac{2}{L} \int_0^L \left[f(x) + \frac{T_0}{6\alpha^2 L} x(L^2 - x^2) \right] \sin \frac{n\pi x}{L} \, dx.$$

50. The boundary of a circular plate of radius *a* is maintained at a temperature which varies periodically with time, according to the law $T(a,t) = T_0 \cos \omega t$, where T_0 and ω are constants. Determine the steady periodic temperature variation at interior points as follows:

(a) Let T(r,t) be the real part of a complex function $U = F(r) e^{i\omega t}$, and show that F(r) then must satisfy the equation

$$\frac{d^2F}{dr^2}+\frac{1}{r}\frac{dF}{dr}-\frac{i\omega}{\alpha^2}F=0$$

and the condition $F(a) = T_0$, together with the requirement that F(r) be finite at r = 0.

(b) By using the results of Section 4.11, deduce that

$$U(r,t) = T_0 \frac{J_0(i^{3/2}kr)}{J_0(i^{3/2}ka)} e^{i\omega t},$$

where $k = \sqrt{\omega}/\alpha$.

(c) Deduce that

$$T(r,t) = T_0 \operatorname{Re} \left\{ \frac{\operatorname{ber} (kr) + i \operatorname{bei} (kr)}{\operatorname{ber} (ka) + i \operatorname{bei} (ka)} (\cos \omega t + i \sin \omega t) \right\}$$
$$= T_0 \operatorname{Re} \left\{ \frac{M_0(kr)}{M_0(ka)} e^{i[\theta_0(kr) - \theta_0(ka) + \omega t]} \right\}$$
$$= T_0 \frac{M_0(kr)}{M_0(ka)} \cos \left[\theta_0(kr) - \theta_0(ka) + \omega t\right],$$

where

$$M_0(kr) = \sqrt{\operatorname{ber}^2(kr) + \operatorname{bei}^2(kr)}, \qquad \theta_0(kr) = \tan^{-1}\frac{\operatorname{bei}(kr)}{\operatorname{ber}(kr)}.$$

(d) Show that the temperature at the center r = 0 is given by

$$T(0,t) = \frac{T_0}{M_0(ka)} \cos \left[\omega t - \theta_0(ka)\right].$$

Problems

51. Assume that a portion of the earth's surface may be considered as plane and that effects of the periodic heating due to the sun are transmitted only in the x direction, perpendicular to the surface. If the resultant temperature on the surface of the earth is taken in the form $T(0,t) = T_0 \cos \omega t$, show that the temperature at depth x is given by

$$T(x,t) = T_0 e^{-\sqrt{\frac{\omega}{2}} \frac{x}{\alpha}} \cos\left(\omega t - \sqrt{\frac{\omega}{2}} \frac{x}{\alpha}\right)$$

with the notation of Section 9.2. [Consider the boundary value $T_0 \cos \omega t$ as the real part of $T_0 e^{i\omega t}$ and determine the *real part* of an expression $T(x,t) = X(x) e^{i\omega t}$ which satisfies the one-dimensional heat flow equation, reduces to $T_0 e^{i\omega t}$ when x = 0, and vanishes when $x \to \infty$.]

Section 9.11

52. The temperature at the end x = 0 of a rod of length L is held at 0° while the temperature at the other end is varied periodically according to the law

$$T(L,t) = T_0 \sin \omega t$$

when t > 0. Suppose that when t = 0 the temperature is zero at all points in the rod.

(a) Find the temperature distribution in the rod. [Use Equation (169).]

(b) Show also that for prescribed oscillation so slow that $(\lambda \omega)^2 \ll 1$, and for times such that $t \gg \lambda$, the approximation

$$T \approx T_0 \left[\frac{x}{L} \sin \omega t + \frac{2\lambda \omega}{\pi} \left(\sum_{n=1}^{\infty} \frac{(-1)^n}{n^3} \sin \frac{n\pi x}{L} \right) \cos \omega t \right]$$

is valid. [By integrating the Fourier sine expansion of x twice, and determining the constants of integration, it can be shown that the series in parentheses represents the function $\pi^3 x(x^2 - L^2)/12L^3$.]

53. Show that Equation (169) can be written in the form

$$T(x,t) = \frac{x}{L}F(t) + \frac{2}{\pi}\sum_{n=1}^{\infty}\frac{(-1)^n}{n}\left[F(t) - \frac{n^2}{\lambda}\int_0^t F(\tau) e^{-n^2(t-\tau)/\lambda} d\tau\right]\sin\frac{n\pi x}{L}$$

54. (a) If F(t) is the unit singularity function $\delta(t)$, show that the temperature distribution given by Equation (168) takes the form

$$T(x,t) = \frac{2}{\pi \lambda} \sum_{n=1}^{\infty} (-1)^{n+1} n e^{-n^2 t/\lambda} \sin \frac{n \pi x}{L} \qquad (t > 0).$$

(b) If this expression is denoted by U(x,t), verify that $U(x,t) = \frac{\partial A(x,t)}{\partial t}$, where A is defined by (160), and that the temperature distribution (168) corresponding to a general function F(t) can be written in the form

$$T(x,t) = \int_0^t U(x, t - \tau) F(\tau) d\tau.$$

55. If a rod with insulated sides initially is at a temperature distribution T(x,0) = f(x), and if the conditions T(0,t) = 0 and T(L,t) = F(t) are imposed when t > 0, show that the resultant temperature distribution can be expressed as the sum of the right-hand member of (168) or (169) and the right-hand member of (154) with $T_1 = T_2 = 0$.

56. (a) Modify the formal argument of Section 9.11 to show that the temperature distribution for which T(x,0) = 0 and for which T(0,t) = 0 and $\frac{\partial T(L,t)}{\partial x} = F(t)$ when t > 0 is again given by Equation (165) or (167) if A(x,t) is the distribution corresponding to F(t) = 1.

(b) Show that here

$$A(x,t) = x - \sum_{n \text{ odd}} \frac{8L}{n^2 \pi^2} \sin \frac{n\pi}{2} \sin \frac{n\pi x}{2L} e^{-n^2 \pi^2 \alpha^2 t/4L^2}.$$

Section 9.12

57. (a) Show that if $U = F e^{i\omega t}$ is a solution of the wave equation (5) in three dimensions, where F is independent of the time t, then F satisfies the Helmholtz equation

$$\nabla^2 F + \frac{\omega^2}{c^2} F = 0,$$

where ∇^2 is the three-dimensional Laplace operator.

(b) By seeking separable solutions of the preceding equation, show that the real and imaginary parts of the functions

$$U = e^{i(\omega/c)(lx + my + nz + ct)} \qquad (l^2 + m^2 + n^2 = 1),$$

$$V = r^{-1/2} J_{\pm(n+\frac{1}{2})} \left(\frac{\omega r}{c}\right) P_n^m(\cos\varphi) e^{i(m\theta + \omega t)},$$

$$W = e^{i[n\theta + (\omega/c)(lz + ct)]} H_n^{(1 \text{ or } 2)} \left(m\frac{\omega r}{c}\right) \qquad (l^2 + m^2 = 1)$$

are solutions of the wave equation in rectangular, spherical, and circular cylindrical coordinates, respectively. [Here P_n^m is an associated Legendre function (see Section 4.12). The solutions given in Section 9.12 are special cases.]

58. Prove that the standing waves $\cos \frac{\omega x}{c} \cos \omega t$, $\cos \frac{\omega x}{c} \sin \omega t$, $\sin \frac{\omega x}{c} \cos \omega t$, and $\sin \frac{\omega x}{c} \sin \omega t$ each can be expressed as the superposition of two waves traveling with velocity c in opposite directions.

59. The equation governing the lateral displacement w(x,t) in small free vibrations of a tightly stretched string of uniform linear density ρ , under constant tension T, is of the form

$$\frac{\partial^2 w}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial^2 w}{\partial t^2}$$

where $\alpha^2 = T/\rho$. (See Problem 7 of Chapter 5.)

(a) If the string is fixed at the ends x = 0 and x = L, and if the string is initially deflected in the form w(x,0) = f(x) and released from rest, so that also $\frac{\partial w(x,0)}{\partial w(x,0)} = 0$, obtain the ensuing deflection in the form

 $\frac{\partial t}{\partial t} = 0$, obtain the ensuing deflection in the form

$$w(x,t) = \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L} \cos \frac{n\pi a t}{L},$$

where

$$a_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} \, dx.$$

(b) Show that the solution of part (a) can be written in the form

$$w(x,t) = F(x - \alpha t) + F(x + \alpha t),$$

where

$$F(x) = \sum_{n=1}^{\infty} a_n \sin \frac{n\pi x}{L},$$

and hence where F(x) is an odd periodic function of x, of period 2L, which coincides with f(x) when 0 < x < L and is defined for all values of x. (Compare Problem 28 of Chapter 8.)

60. When resistive forces proportional to the velocity are taken into account, the differential equation of Problem 59 is replaced by the equation

$$\alpha^2 \frac{\partial^2 w}{\partial x^2} = \frac{\partial^2 w}{\partial t^2} + 2\gamma \frac{\partial w}{\partial t},$$

where γ is a constant.

(a) Show that this equation admits solutions of the form

$$w = e^{-\gamma t} e^{ik[x \pm \alpha \sqrt{1 - (\gamma^2/k^2 \alpha^2)}t]},$$

where k is an arbitrary nonzero constant, and that, if $\gamma^2 < k^2 \alpha^2$, these solutions represent plane waves which are damped in time, and which move along the x axis with velocity $c = \alpha \sqrt{1 - (\gamma^2/k^2 \alpha^2)}$.

(b) Show that the solution of Problem 59(a) here takes the form

$$w(x,t) = e^{-\gamma t} \sum_{n=1}^{\infty} a_n \left(\cos \omega_n t + \frac{\gamma}{\omega_n} \sin \omega_n t \right) \sin \frac{n \pi x}{L},$$

where

$$\omega_n = \sqrt{\frac{n^2 \pi^2 \alpha^2}{L^2} - \gamma^2}$$

and where

$$a_n=\frac{2}{L}\int_0^L f(x)\sin\frac{n\pi x}{L}\,dx.$$

(Notice that the presence of resistive forces not only causes time damping, but also decreases the frequency of each component oscillation, and that the higher

frequencies are no longer integral multiples of the fundamental frequency. When $\gamma > \pi \alpha / L$, one or more of the ω 's is pure imaginary, and the corresponding time functions then become real exponential functions.)

Section 9.13

61. (a) Verify the identity

$$\frac{1}{r^2}\frac{\partial}{\partial r}\left(r^2\frac{\partial\varphi}{\partial r}\right) = \frac{1}{r}\frac{\partial^2}{\partial r^2}(r\varphi).$$

(b) Hence show that if φ depends only upon distance r from the origin (in spherical coordinates) and upon time t, then the wave equation (5) can be written in the form

$$\frac{\partial^2}{\partial r^2}(r\varphi) = \frac{1}{c^2}\frac{\partial^2}{\partial t^2}(r\varphi).$$

(c) From this result obtain the general solution

$$r\varphi = f(r-ct) + g(r+ct).$$

62. The boundary of a sphere of radius *a*, surrounded by an ideal compressible fluid, is caused to pulsate radially with circular frequency ω , so that the radial velocity of the boundary is given by $V = V_0 \cos \omega t$. By methods analogous to those of Section 9.13, determine the steady-state periodically varying velocity of surrounding points in the form

$$V = \frac{V_0 a^2/r^2}{c^2 + a^2 \omega^2} \left[(c^2 + ar\omega^2) \cos \omega \left(\frac{r-a}{c} - t \right) + c \omega (r-a) \sin \omega \left(\frac{r-a}{c} - t \right) \right],$$

where c is the velocity of sound in the fluid. [Use spherical coordinates with the velocity potential dependent only on r and t, and notice that a suitable assumption for the (complex) potential consists of the second term of Equation (187), with c_2 a *complex* constant to be determined.]

63. Let $\varphi(x,t)$ satisfy the one-dimensional wave equation $\alpha^2 \varphi_{xx} = \varphi_{tt}$ when 0 < x < a, subject to the end conditions $\varphi(0,t) = \cos \omega t$ and $\varphi(a,t) = 0$. Determine a particular solution φ which varies periodically with time in the alternative forms

$$\varphi = \frac{\sin \frac{\omega}{\alpha} (a - x)}{\sin \frac{\omega a}{\alpha}} \cos \omega t$$
$$= \frac{1}{2 \sin \frac{\omega a}{\alpha}} \left[\sin \omega \left(t + \frac{a - x}{\alpha} \right) - \sin \omega \left(t - \frac{a - x}{\alpha} \right) \right],$$

under the assumption that sin $(\omega a/\alpha) \neq 0$, by each of the following procedures:

(a) Assume $\varphi = f(x) \cos \omega t + g(x) \sin \omega t$ and determine f and g.

(b) Assume φ as a linear combination of those terms in Equations (180) and (181) which reduce to $\cos \omega t$ when x = 0.

[Notice that φ could represent a permissible lateral displacement of a string with one end fixed and the other oscillating, or it could represent either velocity potential or velocity of an ideal compressible fluid in a tube, closed at x = a, and with an oscillatory valve at x = 0. Notice also that a steady oscillatory response of the type assumed cannot exist if ω is such that $\sin(\omega a/\alpha) = 0$, that is, if $\omega = n\pi\alpha/a$ where *n* is integral, and that other responses satisfying the prescribed end conditions may exist. (See Problem 64.)]

64. Let the initial conditions $\varphi(x,0) = 0$ and $\frac{\partial \varphi(x,0)}{\partial t} = 0$ be added to the end conditions of Problem 63.

(a) By using the result of Problem 63, show that the complete solution can be assumed in the form

$$\varphi = \frac{\sin \frac{\omega}{\alpha} (a - x)}{\sin \frac{\omega a}{\alpha}} \cos \omega t - \sum_{n=1}^{\infty} A_n \sin \frac{n \pi x}{a} \cos \frac{n \pi \alpha t}{a} \qquad (t > 0)$$

if $\omega \neq k\pi\alpha/a$, where the A's are to be determined in such a way that $\varphi(x,0) = 0$. (b) Deduce that then

$$A_n = \frac{2n\pi}{n^2\pi^2 - (a^2\omega^2/\alpha^2)} \qquad \left(\omega \neq \frac{n\pi\alpha}{a}\right)$$

and hence obtain the solution in the form

20

$$\varphi = \frac{\sin\frac{\omega}{\alpha}(a-x)}{\sin\frac{\omega a}{\alpha}}\cos\omega t - 2\sum_{n=1}^{\infty}\frac{n\pi}{n^{2}\pi^{2} - (a^{2}\omega^{2}/\alpha^{2})}\sin\frac{n\pi x}{a}\cos\frac{n\pi\alpha t}{a}$$

when t > 0, if $\omega \neq k\pi\alpha/a$ (k = 1, 2, ...).

(If $\omega \to k\pi\alpha/a$, the kth term of the sum can be extracted and combined with the particular solution, and the combination can be shown to tend to a function of x and t which involves t as a multiplicative factor, the remaining terms of the sum remaining finite in the limit. This is a case of *resonance*. If small resistive forces were present in the nonresonant case, the portion of the solution represented by the series would be damped out with increasing time. Since such forces always exist in practice, the particular solution can be considered as a *quasi*-steady-state solution, regardless of the initial conditions, in the sense that it is the limit, as resistive forces tend to zero, of the true steady-state solution of the problem in which resistance is present.)

Section 9.14

65. If $\varphi(x,y)$ satisfies Laplace's equation in the infinite strip 0 < y < b, and if $\varphi(x,0) = f(x)$ and $\varphi(x,b) = 0$, obtain φ in the form

$$\varphi(x,y) = \frac{1}{\pi} \int_0^\infty \left[\int_{-\infty}^\infty \frac{\sinh u(b-y)}{\sinh ub} f(\xi) \cos u(\xi-x) d\xi \right] du$$

66. By making use of the formula

$$\int_0^\infty \frac{\sinh px}{\sinh qx} \cos rx \, dx = \frac{\pi}{2q} \frac{\sin \frac{p\pi}{q}}{\cos \frac{p\pi}{q} + \cosh \frac{r\pi}{q}} \qquad (p^2 < q^2),$$

and interchanging the order of integration in the result of Problem 65, express the solution of that problem in the form

$$\varphi(x,y) = \frac{1}{2b} \sin \frac{\pi y}{b} \int_{-\infty}^{\infty} \frac{f(\xi)}{\cosh \frac{\pi}{b} (\xi - x) - \cos \frac{\pi y}{b}} d\xi.$$

67. (a) If $\varphi(x,y)$ satisfies Laplace's equation in the quadrant x > 0, y > 0, and if φ vanishes along the positive y axis and reduces to f(x) along the positive x axis, obtain φ in the form

$$\varphi(x,y) = \frac{2}{\pi} \int_0^\infty \int_0^\infty e^{-uy} f(\xi) \sin ux \sin u\xi \, d\xi \, du.$$

(b) By formally integrating first with respect to u, transform this expression to the form

$$\varphi(x,y) = \frac{1}{\pi} \int_0^\infty f(\xi) \left[\frac{y}{y^2 + (\xi - x)^2} - \frac{y}{y^2 + (\xi + x)^2} \right] d\xi.$$

(c) Deduce the result of part (b) from Equation (220) by replacing f(x) in (220) by an odd function $f_o(x)$ such that $f_o(x) = f(x)$ when x > 0 and $f_o(x) = -f(-x)$ when x < 0.

68. (a) If Problem 67 is modified in such a way that $\frac{\partial \varphi}{\partial x}$ rather than φ vanishes along the positive y axis, show that

$$\varphi(x,y) = \frac{2}{\pi} \int_0^\infty \int_0^\infty e^{-uy} f(\xi) \cos ux \cos u\xi \, d\xi \, du$$
$$= \frac{1}{\pi} \int_0^\infty f(\xi) \left[\frac{y}{y^2 + (\xi - x)^2} + \frac{y}{y^2 + (\xi + x)^2} \right] d\xi$$

(b) Deduce the result of part (a) from Equation (220) by replacing f(x) in (220) by an even function $f_e(x)$ such that $f_e(x) = f(x)$ when x > 0 and $f_e(x) = f(-x)$ when x < 0.

69. A rod with insulated sides extends from $x = -\infty$ to $x = +\infty$. If the initial temperature distribution is given by T(x,0) = f(x), where $-\infty < x < \infty$, show that

$$T(x,t) = \frac{1}{\pi} \int_{-\infty}^{\infty} f(\xi) \left[\int_{0}^{\infty} e^{-u^{2}\alpha^{2}t} \cos u(\xi - x) du \right] dt$$
$$= \frac{1}{2\alpha\sqrt{\pi t}} \int_{-\infty}^{\infty} f(\xi) e^{-(\xi - x)^{2}/4\alpha^{2}t} d\xi.$$

Problems

70. A rod of infinite length, with insulated sides, has its end x = 0 insulated. If the initial temperature distribution is given by T(x,0) = f(x), where $0 < x < \infty$, show that

$$T(x,t) = \frac{1}{2\alpha\sqrt{\pi t}} \int_0^\infty f(\xi) \left[e^{-(\xi-x)^2/4\alpha^2 t} + e^{-(\xi+x)^2/4\alpha^2 t} \right] d\xi.$$

71. (a) If f(x) is the unit singularity function $\delta(x)$ (so that a heat source of intensity ρs is present at x = 0 at the instant t = 0), show that the solution of Problem 69 is of the form

$$T(x,t) = \frac{1}{2\alpha\sqrt{\pi t}}e^{-x^2/4\alpha^2 t} \equiv S(x,t).$$

(b) Deduce that the temperature in a rod extending over $(-\infty, \infty)$, due to a *unit heat source* at $x = x_0$ at the instant $t = t_0$, is given by

$$T(x,t) = \frac{1}{\rho s} S(x - x_0, t - t_0).$$

(c) Show that Equation (231) takes the form

$$T(x,t) = \int_0^\infty [S(x - \xi, t) - S(x + \xi, t)] f(\xi) d\xi,$$

the solution of Problem 70 takes the form

$$T(x,t) = \int_0^\infty [S(x - \xi, t) + S(x + \xi, t)] f(\xi) d\xi,$$

and the solution of Problem 69 becomes

$$T(x,t) = \int_{-\infty}^{\infty} S(x - \xi, t) f(\xi) d\xi.$$

[In two dimensions, the corresponding source function is

$$S(x,y,t) = \frac{1}{4\alpha^2 \pi t} e^{-(x^2+y^2)/4\alpha^2 t}$$

(see Problem 75), whereas the function

$$S(x,y,z,t) = \frac{1}{(4\alpha^2 \pi t)^{3/2}} e^{-(x^2 + y^2 + z^2)/4\alpha^2 t}$$

is the three-dimensional generalization.]

72. The end x = 0 of a rod with insulated sides is maintained at the temperature T(0,t) = F(t) for all t > 0. Initially, all points of the rod are at zero temperature, T(x,0) = 0, where $0 < x < \infty$.

(a) Show that the formal argument of Section 9.11 leads to the solution

$$T(x,t) = \int_0^t F(\tau) \frac{\partial A(x, t-\tau)}{\partial t} d\tau,$$

where A(x,t) is the solution of the problem when F(t) = 1.

(b) Use the result of Equation (234) to deduce that

$$A(x,t) = 1 - \operatorname{erf}\left(\frac{x}{2\alpha\sqrt{t}}\right) \equiv 1 - \frac{2}{\sqrt{\pi}}\int_0^{x/2\alpha\sqrt{t}} e^{-u^2} du.$$

(c) From the results of parts (a) and (b), obtain the solution of the stated problem in the form

$$T(x,t) = \frac{x}{2\alpha\sqrt{\pi}} \int_0^t F(\tau) \, e^{-x^2/4\alpha^2(t-\tau)} \, \frac{d\tau}{(t-\tau)^{3/2}} \, .$$

73. (a) If $F(t) = \delta(t)$ in Problem 72, show that

$$T(x,t) = \frac{x}{2\alpha t \sqrt{\pi t}} e^{-x^2/4\alpha^2 t},$$

and that, if this function is denoted by D(x,t), then the solution of Problem 72 takes the form

$$T(x,t) = \int_0^t D(x, t - \tau) F(\tau) d\tau.$$

(b) Verify that

$$D(x,t) = -2x^2 \frac{\partial S(x,t)}{\partial x},$$

where S(x,t) is defined in Problem 71 and is the solution of Problem 69 which corresponds to the initial function $\delta(x)$. [Hence $D/2\alpha^2$ is the solution of Problem 69 corresponding to the initial unit doublet function $-\delta'(x)$ at x = 0 at the instant t = 0.]

74. At the time t = 0, the temperature in unbounded space is dependent only upon radial distance r from the z axis, and is prescribed as T(r,0) = f(r). The ensuing temperature distribution T(r,t) is required, under the assumption that f(r) behaves satisfactorily for large values of r.

(a) Show that T must satisfy the equation

$$\frac{1}{r}\frac{\partial}{\partial r}\left(r\frac{\partial T}{\partial r}\right) = \frac{1}{\alpha^2}\frac{\partial T}{\partial t}$$

when t > 0, and obtain a product solution which remains finite at r = 0 and as $r \rightarrow \infty$ in the form

$$A e^{-u^2 \alpha^2 t} J_0(ur).$$

(b) Deduce that the required distribution is given formally by the expression

$$T(r,t) = \int_0^\infty A(u) e^{-u^2 \alpha^2 t} J_0(ur) u \, du,$$

where the factor u is inserted for convenience in the following step, and where A(u) is to be determined such that

$$f(r) = \int_0^\infty A(u) J_0(ur) u \, du \qquad (0 < r < \infty),$$

Problems

and hence, by referring to Equations (240a,b) of Section 5.14, obtain the desired solution formally as

$$T(r,t) = \int_0^\infty u \ e^{-u^2 \alpha^2 t} \ J_0(ur) \left[\int_0^\infty \xi f(\xi) \ J_0(u\xi) \ d\xi \right] \ du.$$

(c) By assuming the validity of interchange of order of integration, and making use of the relation

$$\int_0^\infty u \ e^{-a^2 u^2} J_n(bu) \ J_n(cu) \ du = \frac{1}{2a^2} \ e^{-(b^2 + c^2)/4a^2} \ I_n\left(\frac{bc}{2a^2}\right),$$

express the formal solution of part (c) in the form

$$T(r,t) = \frac{e^{-r^2/4\alpha^2 t}}{2\alpha^2 t} \int_0^\infty \xi f(\xi) \ e^{-\xi^2/4\alpha^2 t} \ I_0\left(\frac{\xi r}{2\alpha^2 t}\right) \ d\xi \qquad (t > 0).$$

75. (a) Suppose that the initial temperature in *unbounded space* is zero except throughout an infinitely extended right circular cylinder of radius ϵ , with its axis coinciding with the z axis, and has the constant value T_0 inside that cylinder. Show that the excess heat required per unit length of cylinder is given by

$$2\pi\int_0^\epsilon \rho s T_0 r \, dr = \pi \rho s T_0 \epsilon^2,$$

where ρs is the heat capacity per unit volume. Hence deduce that, if *unit* excess heat is to be present per unit length, there must follow $T_0 = 1/(\pi \rho s \epsilon^2)$, and use Problem 74 to show that the corresponding resultant temperature distribution in space at all following times is given by

$$T(r,t) = \frac{e^{-r^2/4\alpha^2 t}}{2\alpha^2 t} \cdot \frac{1}{2\pi\rho s} \cdot \left[\frac{2}{\epsilon^2} \int_0^{\epsilon} e^{-\xi^2/4\alpha^2 t} I_0\left(\frac{\xi r}{2\alpha^2 t}\right) \xi d\xi\right] \qquad (t > 0).$$

(b) By applying L'Hospital's rule (or otherwise) and recalling that $I_0(0) = 1$, show that the quantity in brackets tends to *unity* as $\epsilon \rightarrow 0$, and deduce that the temperature distribution due to an instantaneous line source along the infinite line r = 0, emitting one calorie of heat per unit length at the instant t = 0, is given by

$$T(r,t) = \frac{e^{-r^2/4\alpha^2 t}}{4\pi\rho s\alpha^2 t} \qquad (t > 0)$$

in unbounded space.

(c) Deduce that if the line source is instead perpendicular to the xy plane at the point (x_0, y_0) , and if the impulse takes place at the time t_0 , then the temperature at any point in unbounded space at any following time is given by

$$T = \frac{1}{4\pi\rho s\alpha^2(t-t_0)} e^{-\frac{(x-x_0)^2+(y-y_0)^2}{4\alpha^2(t-t_0)}} \qquad (t > t_0).$$

(See also Problem 71 for the analogs in one and three dimensions.)

Section 9.15

76. (a) When f(t) = 1, show that the voltage variation given by Equation (258), for any fixed value of x, is given by

$$e(x,t) = \begin{cases} 0, & 0 < t < x/v, \\ 1, & x/v < t < (2l - x)/v, \\ 2, & (2l - x)/v < t < (2l + x)/v, \\ 1, & (2l + x)/v < t < (4l - x)/v, \\ 0, & (4l - x)/v < t < 4l/v, \end{cases}$$

when 0 < t < 4l/v, and is repeated periodically, with period 4l/v = 4T, thereafter.

(b) At the midpoint, x = l/2, show that e = 0 for T/2 seconds, then e = 1 for T seconds, then e = 2 for T seconds, then e = 0 for the remaining T/2 seconds of the period 4T.

(c) At the end x = l, show that e = 0 for T seconds, then e = 2 for 2T seconds, then e = 0 for the remaining T seconds of the period 4T.

77. (a) If the end of the line considered in Section 9.15 is grounded, so that e(l,t) = 0, show that

$$\bar{e}(x,s) = \frac{f(s)}{1 - e^{-2s(l/v)}} \left[e^{-sx/v} - e^{-s(2l-x)/v} \right]$$

= $f(s) \left[e^{-sx/v} - e^{-s(2l-x)/v} + e^{-s(2l+x)/v} - e^{-s(4l-x)/v} + \cdots \right].$

(Notice that reflected waves are reversed at each end of the line.)

(b) When f(t) = 1, show that, for any fixed value of x, this expression takes a form similar to that given in Problem 76(a), with the successive values 0, 1, 0, 1, 0 replacing the values 0, 1, 2, 1, 0 given in that expression, and that, in fact, the voltage here is of period 2T.

78. Let T(x,t) satisfy the heat flow equation $\alpha^2 T_{xx} - T_t = 0$ for t > 0 and for $0 < x < \infty$, subject to the initial condition T(x,0) = 1 and to the end conditions T(0,t) = 0 and $T(\infty,t)$ finite when t > 0.

(a) Show that the Laplace transform $\overline{T}(x,s)$ then must satisfy the equation $\alpha^2 \overline{T}_{xx} - s\overline{T} = -1$, and the end conditions $\overline{T}(0,s) = 0$, $\overline{T}(\infty,s)$ finite, and deduce that

$$T(x,s) = \frac{1}{s}(1 - e^{-\sqrt{s}r/\alpha}).$$

(b) By noticing that the solution of the problem is in fact given by $T(x,t) = \operatorname{erf} \left[\frac{x}{2\alpha \sqrt{t}} \right]$, according to Equation (234), deduce that

$$\frac{1}{s}(1-e^{-qx}) = \mathscr{L}\left\{\operatorname{erf}\frac{x}{2\alpha\sqrt{t}}\right\}$$

and

$$\frac{1}{s}e^{-qx}=\mathscr{L}\left\{1-\operatorname{erf}\frac{x}{2\alpha\sqrt{t}}\right\},\,$$

with the abbreviation

$$q = \frac{\sqrt{s}}{\alpha}$$

(A direct derivation of these results is given in Problem 93 of Chapter 10.)

79. Let T(x,t) satisfy the heat flow equation $\alpha^2 T_{xx} - T_t = 0$ for t > 0 and 0 < x < L, subject to the initial condition $T(x,0) = T_0$ and to the end conditions T(0,t) = 0 and $T(L,t) = T_0$ when t > 0.

(a) Show that Equations (154) and (156) of Section 9.10 give the solution in the form

$$T(x,t) = T_0 \left[\frac{x}{L} + \sum_{n=1}^{\infty} \frac{2}{n\pi} \sin \frac{n\pi x}{L} e^{-n^2 \pi^2 \alpha^2 t/L^2} \right] \qquad (t > 0).$$

(b) Show that the Laplace transform T(x,s) must satisfy the equation

$$\alpha^2 \bar{T}_{xx} - s\bar{T} = -T_0$$

and the end conditions T(0,s) = 0, $T(L,s) = T_0/s$, and deduce that

$$\tilde{T}(x,s) = \frac{T_0}{s} \left\{ 1 - e^{-qx} \left[\frac{1 - e^{-2q(L-x)}}{1 - e^{-2qL}} \right] \right\},\,$$

where $q = \sqrt{s/\alpha}$.

(c) By expanding $1/(1 - e^{-2qL})$ in a series of ascending powers of e^{-2qL} (when q > 0), express this transform in the form

$$\bar{T}(x,s) = T_0 \bigg[\frac{1}{s} - \frac{1}{s} e^{-qx} + \frac{1}{s} e^{-q(2L-x)} - \frac{1}{s} e^{-q(2L+x)} + \cdots \bigg].$$

(d) By evaluating inverse transforms term by term (as can be justified here), and using the results of Problem 78, deduce that

$$T(x,t) = T_0 \left[\operatorname{erf} \frac{x}{2\alpha \sqrt{t}} + \left(1 - \operatorname{erf} \frac{2L - x}{2\alpha \sqrt{t}} \right) - \left(1 - \operatorname{erf} \frac{2L + x}{2\alpha \sqrt{t}} \right) + \cdots \right].$$

[Notice that the series form given in part (a) converges rapidly when $\alpha \sqrt{t/L}$ is large. Since erf *u* approaches unity very rapidly as $u \rightarrow +\infty$, the terms in parentheses in the form of part (d) are very small when $\alpha \sqrt{t/L}$ is *small*. Hence the latter form is particularly convenient when the former is not.]

80. In Problem 62 suppose that until the instant t = 0 the sphere and surrounding medium are at rest in complete equilibrium, and that at that instant the pulsation begins and continues indefinitely. Determine the velocity of surrounding points by using Laplace transforms as indicated below. (a) Make use of the result of Problem 61 [see also Equations (197)–(200), Section 6.20] to obtain the relevant relations involving the velocity potential φ and radial velocity V in the form

$$\frac{\partial^2(r\varphi)}{\partial r^2} = \frac{1}{c^2} \frac{\partial^2(r\varphi)}{\partial t^2};$$

 $r = a: \quad V = \frac{\partial \varphi}{\partial r} = V_0 \cos \omega t; \quad r \to \infty: \quad \frac{\partial \varphi}{\partial r} \text{ finite};$
 $t = 0: \quad \varphi = \frac{\partial \varphi}{\partial t} = 0.$

(b) By calculating the Laplace transforms of these relations, obtain the equations

$$\frac{\partial^2(r\bar{\varphi})}{\partial r^2} - \frac{s^2}{c^2}(r\bar{\varphi}) = 0;$$

$$\bar{V} = \frac{\partial\bar{\varphi}}{\partial r};$$

$$r = a: \quad \bar{V} = \frac{V_0 s}{s^2 + \omega^2}; \quad r \to \infty: \quad \bar{V} \text{ finite.}$$

(c) Show that $r\bar{\varphi}$ must be of the form $A e^{-sr/c}$ and hence also

$$\bar{V} = -\frac{A}{cr}\left(s+\frac{c}{r}\right)e^{-sr/c}.$$

(d) Determine A, as a function of s, so that the condition at r = a is satisfied. Thus obtain the result

$$\overline{V} = \overline{f}(s) \ e^{-[(r-a)/c]s}$$

$$\overline{f}(s) = \frac{aV_0}{r} \frac{s\left(s + \frac{c}{r}\right)}{(s^2 + \omega^2)\left(s + \frac{c}{a}\right)}.$$

where

(e) Expand f(s) in partial fractions and use pairs T9, 12, 15, and 16 of Table 1, Chapter 2, to determine the inverse transform of \vec{V} . The solution is of the form

$$V = 0 \quad \text{when } r - a > ct,$$

$$V = \frac{V_0 a^2/r^2}{c^2 + a^2 \omega^2} \left[(c^2 + ar\omega^2) \cos \omega \left(t - \frac{r - a}{c} \right) - c\omega(r - a) \sin \omega \left(t - \frac{r - a}{c} \right) + \frac{c^2}{a} (r - a) e^{-\frac{c}{a} \left(t - \frac{r - a}{c} \right)} \right] \quad \text{when } r - a < ct.$$

(Notice that the disturbance travels radially outward with the speed c and at time t is present only at distances less than ct from the sphere surface. As $t \rightarrow \infty$ all the surrounding fluid becomes disturbed, the exponential term tends to zero, and the solution approaches that of Problem 62.)

Section 9.17

81. In a one-dimensional irrotational flow of an ideal compressible fluid in the positive x direction, in which the velocity V_x depends upon position x and time t but differs by only a small amount from a constant U, the velocity V_x can be expressed (approximately) in the form

$$V_x = U + \frac{\partial \varphi(x,t)}{\partial x},$$

where φ satisfies the equation

$$(U^2 - V_S^2)\frac{\partial^2\varphi}{\partial x^2} + 2U\frac{\partial^2\varphi}{\partial x \partial t} + \frac{\partial^2\varphi}{\partial t^2} = 0.$$

Here V_S is an effective mean value of the sonic velocity in the fluid. The density ρ is then expressible (approximately) in the form

$$\rho = \rho_0 \left[1 - \frac{1}{V_S^2} \left(U \frac{\partial \varphi}{\partial x} + \frac{\partial \varphi}{\partial t} \right) \right],$$

where ρ_0 is the constant value associated with a uniform flow. (See Section 6.20).

(a) Obtain the general solution for $\varphi(x,t)$ in the form

 $\varphi(x,t) = f[x - (V_S + U)t] + g[x + (V_S - U)t],$

where f and g are arbitrary twice-differentiable functions.

(b) Show that V_x and ρ then are defined by the equations

$$V_{x} = U + F[x - (V_{S} + U)t] + G[x + (V_{S} - U)t],$$

$$\rho = \rho_{0} \left[1 + \frac{1}{V_{S}} \left\{ F[x - (V_{S} + U)t] - G[x + (V_{S} - U)t] \right\} \right],$$

where F = f' and G = g'.

82. (a) Show that the functions V_x and ρ in Problem 81 are determined at the point P with coordinate $x = x_1$ at any time $t = t_1$ by *initial* values of V_x and ρ (at the time t = 0) at the two points Q_1 and Q_2 with coordinates

$$x = x_1 - (V_S + U)t_1$$
 and $x = x_1 + (V_S - U)t_1$.

(b) Show that Q_1 and Q_2 are both upstream from P when the flow is supersonic $(U > V_S)$, whereas they are on opposite sides of P when the flow is subsonic.

(c) Sketch typical "characteristic lines" $x - (V_S + U)t = \text{constant}$ and $x - (V_S - U)t = \text{constant}$ in the upper half (t > 0) of a fictitious xt plane (compare with Figure 9.15) in both the supersonic and subsonic cases, and show that the values of V_x and ρ at any point $P(x_1, t_1)$ of this fictitious plane are determined by values of V_x and ρ at the points where the two characteristic lines passing through P intersect the x axis.

83. (a) If the density is prescribed as $\rho = [1 + \eta(x)]\rho_0$ when t = 0 in Problem 81, where ρ_0 is a constant and where $|\eta(x)| \ll 1$, and where $\eta(x) \to 0$ as $x \to -\infty$, show that there must follow

$$V_x = U + G[x - (V_S + U)t] + G[x + (V_S - U)t] + V_S \eta[x - (V_S + U)t],$$

$$\frac{\rho}{\rho_0} = 1 + \frac{1}{V_S} \{ G[x - (V_S + U)t] - G[x + (V_S - U)t] \} + \eta[x - (V_S + U)t],$$

where G is an arbitrary function.

(b) In addition, let it be prescribed that the velocity must tend to the uniform velocity U at an infinite distance upstream (as $x \to -\infty$) for all positive values of t. Show that, at any positive time t, the velocity at any point for which $x + (V_S - U)t = c$, where c is a positive constant, is given by

$$V_{x} = U + G(c - 2V_{S}t) + G(c) + V_{S} \eta(c - 2V_{S}t).$$

Noticing that $x \to -\infty$ as $t \to +\infty$, when c is held fixed, if the flow is subsonic $(U < V_S)$, show that there then must follow $G(c) = -G(-\infty)$ for all positive values of c, and deduce that the function G must be zero. Hence show that, when the flow is subsonic, the prescribed conditions uniquely determine V_x and ρ in the forms

$$V_x = U + V_S \eta [x - (V_S + U)t], \quad \rho = \rho_0 \{1 + \eta [x - (V_S + U)t]\}$$

when t > 0. [Notice that here we have made $t \to +\infty$ along a characteristic line $x + (V_S - U)t = c$ in the xt plane, and that this line intersects the line on which $x = x_0 \to -\infty$ in the upper half-plane t > 0 if and only if $U < V_S$.]

(c) Show that the same result would follow in the supersonic case if it were prescribed that $V_x \rightarrow U$ as $x \rightarrow -\infty$ for all negative values of t, or that $V_x \rightarrow U$ as $x \rightarrow +\infty$ for all positive time, if $\eta(x) \rightarrow 0$ as $x \rightarrow +\infty$.

84. Determine V_x and ρ in Problem 81, subject to the conditions that $\rho = \rho_0$ and $V_x = [1 + \epsilon(x)]U$ when t = 0, where ρ_0 is a constant and where $|\epsilon(x)| \ll 1$.

Functions of a Complex Variable

10.1. Introduction. The complex variable. A complex number α is an expression of the form $\alpha = a + ib$, where a and b are real numbers and i is the imaginary unit, satisfying the equation

$$i^2 = -1.$$
 (1)

We speak of a as the real part of α and of b as the imaginary part of α , and write

$$\alpha = a + ib$$
: $a = \operatorname{Re}(\alpha), \quad b = \operatorname{Im}(\alpha).$ (2)

A complex number a + ib is said to be *real* when b = 0 and to be *imaginary* when $b \neq 0$. In particular, when $b \neq 0$ and a = 0 the number is said to be *pure imaginary*. In the same way we may define a complex variable z = x + iy, where x and y are real variables.

Geometrically, it is convenient to represent a complex quantity x + iy by the point (x,y) in a rectangular coordinate system known as the *complex plane* (Figure 10.1). Thus the points representing *real* numbers are all located on the x axis in this plane, whereas those representing the *pure imaginary* numbers are located on the y axis. For some purposes it is convenient to think of the complex number z as being represented by a *vector* in the complex plane from the origin (x = y = 0) to the point (x,y). We speak of the length of this vector as the *absolute value* of z, or as the *modulus* of z, and denote this number by |z|,

$$|z| = |x + iy| = \sqrt{x^2 + y^2}.$$
 (3)

The number x - iy is called the *conjugate* of the number z = x + iy and is denoted by the symbol \overline{z} ,

$$\bar{z} = x - iy. \tag{4}$$

We notice that $|z| = |\overline{z}|$ and, further, that

$$z\overline{z} = (x + iy)(x - iy) = x^2 - i^2y^2 = x^2 + y^2 = |z|^2 = |\overline{z}|^2.$$
 (5)

That is, the product of a complex number and its conjugate is a nonnegative real number equal to the square of the absolute value of the complex number.

We say that two complex numbers are *equal* if and only if their real and imaginary parts are respectively equal; that is,

$$x_1 + iy_1 = x_2 + iy_2$$
 implies
 $x_1 = x_2, y_1 = y_2.$ (6)

In particular, a complex number is zero if and only if its real and imaginary parts are both zero.

Addition, subtraction, multiplication, and division of complex numbers are defined as being accomplished according to the rules

governing real numbers when one writes $i^2 = -1$, in accordance with Equation (1). (See also Problem 1.) Thus, if

$$z_1 = x_1 + iy_1, \quad z_2 = x_2 + iy_2,$$
 (7)

there follows

$$z_1 + z_2 = (x_1 + x_2) + i(y_1 + y_2) z_1 - z_2 = (x_1 - x_2) + i(y_1 - y_2)$$
(8a,b)

$$z_1 z_2 = (x_1 + iy_1)(x_2 + iy_2) = (x_1 x_2 - y_1 y_2) + i(x_1 y_2 + x_2 y_1), \quad (8c)$$

and also, using (5),

$$\frac{z_2}{z_1} = \frac{x_2 + iy_2}{x_1 + iy_1} = \frac{(x_2 + iy_2)(x_1 - iy_1)}{x_1^2 + y_1^2}$$
$$= \left(\frac{x_1x_2 + y_1y_2}{x_1^2 + y_1^2}\right) + i\left(\frac{x_1y_2 - x_2y_1}{x_1^2 + y_1^2}\right).$$
(8d)

If we introduce polar coordinates (r,θ) , such that

 $x = r \cos \theta, \quad y = r \sin \theta \qquad (r \ge 0),$

the complex number z can be written in the polar form

$$z = x + iy = r(\cos \theta + i \sin \theta), \qquad (9)$$

where r is the modulus of z. It should be noticed that for a given complex number the angle θ can be taken in infinitely many ways. With the convention that the modulus r be nonnegative, the various possible values of θ differ by



у

integral multiples of 2π . Any one of these angles is known as an *argument* or *amplitude* of z.

If we notice that addition or subtraction of complex numbers follows the parallelogram law of vector combination, the truth of the useful inequalities

$$|z_1| - |z_2| \le |z_1 + z_2| \le |z_1| + |z_2| \tag{10}$$

follows directly from elementary geometrical considerations (Figure 10.2).

10.2. Elementary functions of a complex variable. We now proceed to define functions such as e^z , sin z, log z, and so on, taking care that these definitions reduce to the conventional ones when z becomes the real variable x.

The simplest such function is the integral power function

$$f(z)=z^n,\qquad (11)$$

where n is a positive integer or zero. This function is clearly defined by repeated multiplication according to the law



Figure 10.2

$$z^{n} = (x + iy)^{n} = r^{n}(\cos \theta + i \sin \theta)^{n} \qquad (n = 0, 1, 2, ...).$$
(12)

A *polynomial* is then defined as a linear combination of a finite number of such functions, where the constants of combination may be imaginary,

$$f(z) = \sum_{n=0}^{N} A_n z^n.$$
 (13)

A rational function of z is defined as the ratio of two polynomials.

By considering the limit of expressions of form (13) as $N \rightarrow \infty$ or, more generally, limits of the form

$$f(z) = \lim_{N \to \infty} \sum_{n=0}^{N} A_n (z-a)^n = \sum_{n=0}^{\infty} A_n (z-a)^n, \quad (14)$$

where a may be real or imaginary, we are led to definitions of functions of a complex variable in terms of *power series*. The convergence of such series may be investigated by the *ratio test* (see Section 4.1), just as in the case of series of real terms. Thus, if we write

$$L = \lim_{n \to \infty} \left| \frac{A_{n+1}}{A_n} \right| \tag{15}$$

when that limit exists, the series converges when

$$|z-a| < \frac{1}{L}.$$
 (16)

Geometrically, this restriction is seen to require that z lie inside a circle of radius 1/L with center at the point z = a in the complex plane.

Inside this *circle of convergence* the series can be integrated or differentiated term by term, and the resultant series will represent the integral or derivative, respectively, of the represented function. This fact will be of considerable importance in much of what follows.

The exponential function e^z is defined by the power series

$$e^{z} = \sum_{n=0}^{\infty} \frac{z^{n}}{n!} = 1 + z + \frac{z^{2}}{2!} + \frac{z^{3}}{3!} + \cdots$$
 (17)

This definition is acceptable, since the series converges and hence defines a differentiable function of z for all real or imaginary values of z, and since this series reduces to the usual definition when z is real. If we multiply the series defining e^{z_1} and e^{z_2} together term by term (as is permissible for convergent power series), the resultant series is found to be that defining $e^{z_1+z_2}$; that is, the relation

$$e^{z_1} \cdot e^{z_2} = e^{z_1 + z_2} \tag{18}$$

is true for complex values of z_1 and z_2 . Consequently, if *n* is a positive integer, we have also the relation

$$(e^{z})^{n} = e^{nz}$$
 (n = 1, 2, 3, ...) (19)

for all complex values of z.

The circular functions may be defined, in terms of the already-defined exponential functions, by the relations

$$\sin z = \frac{e^{iz} - e^{-iz}}{2i}$$

$$\cos z = \frac{e^{iz} + e^{-iz}}{2}$$
(20a,b)

together with the relations $\tan z = \sin z/\cos z$, and so on. Consequently, we have, from Equations (20) and (17), the corresponding series definitions

$$\sin z = z - \frac{z^3}{3!} + \frac{z^5}{5!} + \dots = \sum_{n=0}^{\infty} (-1)^n \frac{z^{2n+1}}{(2n+1)!}$$
(21a)

and

$$\cos z = 1 - \frac{z^2}{2!} + \frac{z^4}{4!} + \dots = \sum_{n=0}^{\infty} (-1)^n \frac{z^{2n}}{(2n)!}, \qquad (21b)$$

which reduce to the proper forms when z is real. From these series, or from the definitions (20a,b), it can be shown that the circular functions satisfy the same identities for imaginary values of z as for real values.

Equations (20a,b) imply the important relation

$$e^{iz} = \cos z + i \sin z, \qquad (22)$$

which is known as Euler's formula.

With this relation Equation (9) takes the form

$$z = x + iy = r(\cos \theta + i \sin \theta) = re^{i\theta}.$$
 (23)

In consequence of (23) and (19) we then have

$$z^n = r^n (\cos \theta + i \sin \theta)^n = r^n e^{in\theta}$$
 (n = 1, 2, 3, ...). (24)

But since Equation (22) also implies the relation

$$e^{in\theta} = \cos n\theta + i \sin n\theta$$

we deduce *DeMoivre's theorem*,

$$(\cos\theta + i\sin\theta)^n = \cos n\theta + i\sin n\theta, \qquad (25)$$

and hence may rewrite (24) in the form

$$z^n = r^n(\cos n\theta + i \sin n\theta)$$
 (n = 1, 2, 3, ...). (26)

Geometrically, Equation (26) shows that if z has the absolute value r and the angle θ , then zⁿ has the absolute value rⁿ and the angle n θ , if n is a positive integer.

In a similar way we find that if $z_1 = r_1 e^{i\theta_1}$ and $z_2 = r_2 e^{i\theta_2}$, then

$$z_1 z_2 = r_1 r_2 e^{i(\theta_1 + \theta_2)} \tag{27a}$$

$$\frac{z_2}{z_1} = \frac{r_2}{r_1} e^{i(\theta_2 - \theta_1)}.$$
 (27b)

That is, if z_1 and z_2 have absolute values r_1 and r_2 and angles θ_1 and θ_2 , respectively, then z_1z_2 has the absolute value r_1r_2 and the angle $\theta_1 + \theta_2$, and z_2/z_1 has the absolute value r_2/r_1 and the angle $\theta_2 - \theta_1$.

In particular, we notice that since

$$|e^{i\alpha}| = 1 \qquad (\alpha \text{ real}), \tag{28}$$

the multiplication of any complex number z by a number of the form $e^{i\alpha}$, where α is real, is equivalent to rotating the vector representing the number z through an angle α in the complex plane.

The hyperbolic functions are defined, as for functions of a real variable, by the equations

$$\sinh z = \frac{e^{z} - e^{-z}}{2} \\
 \cosh z = \frac{e^{z} + e^{-z}}{2}
 \right\}$$
(29a,b)

and

(30b)

and by the equations $\tanh z = \sinh z / \cosh z$, and so on. Consequently, we have also

$$\sinh z = z + \frac{z^3}{3!} + \frac{z^5}{5!} + \dots = \sum_{n=0}^{\infty} \frac{z^{2n+1}}{(2n+1)!}$$
(30a)

and

and
$$\cosh z = 1 + \frac{z^2}{2!} + \frac{z^4}{4!} + \dots = \sum_{n=0}^{\infty} \frac{z^{2n}}{(2n)!}$$
. (30b)
From these definitions it can be shown that hyperbolic functions of a complex variable satisfy the same identities as the corresponding functions of a real variable. By comparing Equations (20) and (29), we obtain in particular the relations

$$\begin{array}{ll}
\sinh iz = i \sin z, & \cosh iz = \cos z \\
\sin iz = i \sinh z, & \cos iz = \cosh z
\end{array}$$
(31)

relating the circular and hyperbolic functions.

The results so far obtained permit us to express the functions considered in terms of their real and imaginary parts as follows:

$$e^{z} = e^{x+iy} = e^{x}e^{iy} = (e^{x}\cos y) + i(e^{x}\sin y),$$

$$\sin z = \sin (x + iy) = \sin x \cos iy + \cos x \sin iy$$

$$= (\sin x \cosh y) + i(\cos x \sinh y),$$

$$\cos z = \cos (x + iy) = \cos x \cos iy - \sin x \sin iy$$

$$= (\cos x \cosh y) - i(\sin x \sinh y),$$

$$\sinh z = \sinh (x + iy) = \sinh x \cosh iy + \cosh x \sinh iy$$

$$= (\sinh x \cos y) + i(\cosh x \sin y),$$

$$\cosh z = \cosh (x + iy) = \cosh x \cosh iy + \sinh x \sinh iy$$

$$= (\cosh x \cos y) + i(\sinh x \sin y).$$
 (32)

10.3. Other elementary functions. We next define the complex logarithmic function as the inverse of the exponential function. Denoting this function temporarily by Log z, we see that the equation

$$w = \operatorname{Log} z \tag{33}$$

must then be equivalent to

$$z = e^w. \tag{34}$$

To express Log z in terms of its real and imaginary parts, we write

$$w = u + iv, \tag{35}$$

after which Equation (34) gives

$$z = x + iy = e^{u+iv} = e^u e^{iv} = e^u \cos v + ie^u \sin v.$$

Hence, equating real and imaginary parts, we obtain

$$e^u \cos v = x, \qquad e^u \sin v = y. \tag{36}$$

Solving for u and v, there follows

$$e^{2u} = x^2 + y^2 = |z|^2 = r^2$$
, $\tan v = \frac{y}{x} = \tan \theta$,
 $r \neq 0$,

and hence, if $r \neq 0$,

$$u = \log r = \log |z|, \quad v = \theta, \quad (37)$$

where r represents, as usual, the absolute value of z and log r is the ordinary real logarithm, whereas θ is a particular choice of the infinitely many angles (differing by integral multiples of 2π) which may be associated with z. Thus we have obtained the result

$$\operatorname{Log} z = \log |z| + i\theta, \tag{38}$$

for any $z \neq 0$. To emphasize the fact that θ is determinate only within an integral multiple of 2π , we may here denote that particular value of θ which lies in the range $0 \leq \theta < 2\pi$ by θ_P , and may speak of this value as the *principal value* of θ for the logarithm,

$$0 \leq \theta_P < 2\pi. \tag{39}$$

Then any other permissible value of θ is of the form $\theta = \theta_P + 2k\pi$, when k is integral, and Equation (38) becomes

Log
$$z = \log |z| + i(\theta_P + 2k\pi)$$
 $(k = 0, \pm 1, \pm 2, ...).$ (40)

Thus it follows that the function Log z, defined as the inverse of e^z , is an *infinitely many-valued* function. For example, if z = i, there follows |z| = 1 and $\theta_P = \pi/2$, and hence

Log
$$i = \log 1 + i \left(\frac{\pi}{2} + 2k\pi\right) = \frac{4k+1}{2}\pi i$$
 $(k = 0, \pm 1, \pm 2, ...).$ (41)

The value corresponding to k = 0 is frequently called the *principal value of* the logarithm.

If, in a particular discussion, z is restricted to *real positive* values, say z = x, there follows |z| = x and $\theta_P = 0$, and hence, from (40),

$$\text{Log } x = \log x + 2k\pi i$$
 (x real and positive). (42)

Thus the complex logarithm of a positive real number may differ from the usual real logarithm by an arbitrary integral multiple of $2\pi i$. In order to conform with conventional usage, henceforth we will identify the complex logarithm Log z with the real logarithm log z when, throughout a given discussion, z is real and positive, by taking k = 0 in (42) in such a case. In the

more general case it is also conventional to write $\log z$ in place of Log z, with the understanding that unless z is to take on *only* real positive values, $\log z$ is to be considered as multiply valued. Thus we will write

$$\log z = \log |z| + i(\theta_P + 2k\pi) \qquad (k = 0, \pm 1, \pm 2, \dots), \qquad (43)$$

in place of (40), and avoid resultant contradiction in the case of (42) by taking k = 0 when z is a positive real variable.

Suppose now that for a given point z_1 in the complex plane we choose a particular value of k in (43), say k = 0, and hence determine a particular value of log z_1 . If a point z moves continuously along a path originating at z_1 , the value of log z then varies continuously from the initial value log z_1 . In particular, if z traverses a *closed* path surrounding the origin in the positive (counterclockwise) direction and returns toward the initial point, it may be seen that the angle θ increases by an amount approaching 2π ; and hence as the circuit is completed the logarithm is increased by $2\pi i$, the *real* part of the logarithm returning to its original value. This statement is true, however, only for a path *enclosing the origin* z = 0. If now the point z continues to retrace its first path, the logarithm is now given by a *different* (*one-valued*) function or by a different "branch" of the same (multiple-valued) function. That is, if we write

$$(\log z)_k = \log |z| + i(\theta_P + 2k\pi) \qquad (0 \le \theta_P < 2\pi),$$

and if on the first circuit log z is determined (with k = 0) by the branch $(\log z)_0$, then, if log z is to vary continuously, on the second circuit log z must be determined by the branch $(\log z)_1$, corresponding to k = 1. The point z = 0, which must be enclosed by the circuit if transition from one branch to another occurs, is known as a branch point. We may say that the function log z has infinitely many branches with a single finite branch point at z = 0.

The generalized power function $f(z) = z^a$, where a may be real or imaginary, is now defined in terms of the logarithm by the equation

$$z^a = e^{a \log z}.$$
 (44)

When z is a positive real variable and a is real, this definition is clearly in accord with the usual definition if the logarithm of a positive real number is taken to be real. If a is a positive integer, this definition must be consistent with (24). To see that this is so, we write a = n, where n is an integer, and obtain from (44)

$$z^{n} = e^{n[\log r + i(\theta_{P} + 2k\pi)]} = e^{n \log r} e^{in\theta_{P}} e^{2kn\pi i} \qquad (k = 0, \pm 1, \pm 2, \dots)$$

But since k and n are integers, we have

$$e^{2kn\pi i} = \cos 2kn\pi + i \sin 2kn\pi = 1,$$

and hence there follows, in this case,

$$z^n = r^n e^{i n \theta_P}$$
 (*n* an integer). (45)

This result is in accordance with (24), since $e^{in\theta} = e^{in(\theta_P + 2k\pi)} = e^{in\theta_P}$ when *n* is an integer.

More generally, if a is a real rational number, we can write

$$a=\frac{m}{n}$$

where m and n are integers having no common factor. Then Equation (44) takes the form

$$z^{\frac{m}{n}} = e^{\frac{m}{n}\log r} e^{i\frac{m}{n}\theta_{P}} e^{2k\frac{m}{n}\pi i} \qquad (k = 0, \pm 1, \pm 2, \dots)$$
$$z^{\frac{m}{n}} = [r^{\frac{m}{n}} e^{i\frac{m}{n}\theta_{P}}] e^{2k\frac{m}{n}\pi i} \qquad (k = 0, \pm 1, \pm 2, \dots).$$

or

If we remember that m and n are given integers, whereas k is an arbitrary integer, we can easily see that as k takes on any n successive integral values,

say (k = 0, 1, 2, ..., n - 1), the factor $e^{2k\frac{m}{n}\pi i}$ will take on *n* corresponding different values, but that if further values of *k* are taken, the *n* values so obtained are merely repeated periodically. Hence it follows that for any nonzero value of *z* the function $z^{m/n}$ has exactly *n* different values given by

$$z^{\frac{m}{n}} = r^{\frac{m}{n}} e^{i\frac{m}{n}(\theta_P + 2k\pi)} \qquad (k = 0, 1, 2, \dots, n-1).$$
(46)

It may be seen also that the function $z^{m/n}$ can be considered as having exactly *n* branches, such that if we restrict θ_P between 0 and 2π , and if a point traverses a closed contour *including the origin* (so that θ changes by 2π), then a continuous variation of $z^{m/n}$ is obtained only through transition from one branch to another. We may verify further that if such a closed contour is traversed exactly *n* times, starting initially with a certain branch, the transition from the *n*th branch is back to the initial branch. The point z = 0 is again a branch point.

As an example, suppose that $m/n = \frac{2}{3}$, and that a point z traverses the unit circle in the positive (counterclockwise) direction, starting at the point z = 1. If we arbitrarily start out on the branch k = 0 and note that for z = 1 there follows r = 1, $\theta_P = 0$, the initial value of $z^{2/3}$ at z = 1 is given by 1. As the end of a circuit is approached, the angle θ_P approaches 2π , and as $z \to 1$ the power $z^{2/3}$ approaches the value $e^{i(2/3)(2\pi+0)} = e^{(4/3)\pi i}$. In order that $z^{2/3}$ vary continuously as the point z = 1 is passed, and hence θ_P drops abruptly to zero and then again increases, we must now determine $z^{2/3}$ from the second branch, for which k = 1, since this branch assumes at $\theta_P = 0$ the value approached by the

first branch when $\theta_P \to 2\pi$. As the end of the *second* circuit is approached and again $z \to 1$, the power $z^{2/3}$ approaches the value $e^{i(2/3)(2\pi+2\pi)} = e^{(8/3)m}$; and as this point is passed, the transition to the *third* branch(k = 2)must occur. Finally, as the end of the *third* circuit is approached and once more $z \to 1$, the power $z^{2/3}$ approaches the value $e^{i(2/3)(2\pi+4\pi)} = e^{4\pi i} = 1$, and hence (for continuity) a transition back to the *first* branch (k = 0) must take place. The three values taken on by $z^{2/3}$ at z = 1 are $1, (-1 - i\sqrt{3})/2$, and $(-1 + i\sqrt{3})/2$.

In the general case when *a* is *complex*, of the form

$$a=a_1+ia_2,$$

where a_1 and a_2 are real, Equation (44) becomes

$$z^{a} = e^{(a_{1} + ia_{2})[\log r + i(\theta_{P} + 2k\pi)]}$$

= $e^{[a_{1}\log r - a_{2}(\theta_{P} + 2k\pi)]} e^{i[a_{2}\log r + a_{1}(\theta_{P} + 2k\pi)]}$
= $r^{a_{1}} e^{-a_{2}(\theta_{P} + 2k\pi)} \{\cos [a_{2}\log r + a_{1}(\theta_{P} + 2k\pi)] + i \sin [a_{2}\log r + a_{1}(\theta_{P} + 2k\pi)]\}$ $(k = 0, \pm 1, \pm 2, ...).$ (47)

It is apparent that, in general, the function z^a is infinitely many-valued, with z = 0 as a branch point. If $a_2 = 0$ and a_1 is rational, then only a finite number of branches exist; in particular, if $a_2 = 0$ and a_1 is integral the function is single-valued and z = 0 is *not* a branch point.

In place of choosing as the *principal value* of θ that value which is in the interval $(0,2\pi)$, as was done here in the case of the logarithmic function, one may, for example, equally well adopt the convention $-\pi < \theta_P \leq \pi$. Then transition from branch to branch will take place along the *negative* real axis. The choice to be preferred will depend upon the particular application involved.

The generalized exponential function $f(z) = a^z$, where a may be real or imaginary, is defined similarly by the equation

$$a^z = e^{z \log a}.\tag{48}$$

If we denote by α_P the principal value of the angle corresponding to *a*, such that $0 \leq \alpha_P < 2\pi$, there follows

$$a^{z} = e^{(x+iy)[\log|a|+i(\alpha_{P}+2k\pi)]} = e^{[x \log|a|-y(\alpha_{P}+2k\pi)]} e^{i[y \log|a|+x(\alpha_{P}+2k\pi)]}$$

= $|a|^{x} e^{-y(\alpha_{P}+2k\pi)} \{\cos[y \log|a|+x(\alpha_{P}+2k\pi)] + i \sin[y \log|a|+x(\alpha_{P}+2k\pi)]\} \quad (k = 0, \pm 1, \pm 2, ...).$ (49)

Although this function is again apparently infinitely many-valued, it is seen that here the ambiguity arises only in the choice of the angle to be associated with the constant *a*, and not upon the specification of the angular position of the point z in the plane. That is, here there is no possibility of a continuous transition from an expression corresponding to a given value of k to one corresponding to a second value, as the result of motion of a point z around a curve in the complex plane. Thus, in this sense, each choice of k can be considered as determining a *separate function*, rather than a particular *branch* of a single multivalued function. In any given problem, only one such definition is needed. It is usually convenient to take k = 0, and hence write

$$a^{z} = |a|^{x} e^{-\nu \alpha_{P}} \left[\cos \left(y \log |a| + x \alpha_{P} \right) + i \sin \left(y \log |a| + x \alpha_{P} \right) \right].$$
(50)

We notice in particular that if a = e (and hence $\alpha_P = 0$), the definition (50) reduces to that given in (32). This is the reason for choosing the particular value k = 0.

Finally, to conclude the list of *elementary functions*, we consider the *inverse* circular and hyperbolic functions. In the case of the inverse sine function, the equation

$$w = \sin^{-1} z \tag{51}$$

implies the equation

$$z = \sin w. \tag{52}$$

If we make use of the definition (20a), this equation takes the form

$$z=\frac{e^{iw}-e^{-iw}}{2i},$$

or, equivalently,

$$e^{2iw} - 2iz \ e^{iw} - 1 = 0. \tag{53}$$

Equation (53) is quadratic in e^{iw} , with the solution

$$e^{iw} = iz + (1 - z^2)^{1/2}$$

Solving this result for $w = \sin^{-1} z$, there follows finally

$$\sin^{-1} z = \frac{1}{i} \log \left[iz + (1 - z^2)^{1/2} \right].$$
 (54)

It is important to notice that when $z \neq \pm 1$ the quantity $(1 - z^2)^{1/2}$ has two possible values. Then corresponding to each such value the logarithm has infinitely many values. Hence, for any given value of $z \neq \pm 1$ the function $\sin^{-1} z$ has two infinite sets of values; for $z = \pm 1$, the two sets coincide. This is, of course, already known to be the case when z is real and numerically less than unity so that $\sin^{-1} z$ is real. For example, we have the values $\sin^{-1} \frac{1}{2} = \pi/6 + 2k\pi$ or $5\pi/6 + 2k\pi$, where in either case k may take on arbitrary integral values. In a later section it will be shown that $\sin^{-1} z$ has branch points at $z = \pm 1$. We may verify that Equation (54) gives the known values for $\sin^{-1} \frac{1}{2}$ by making the calculations

$$\sin^{-1}\frac{1}{2} = \frac{1}{i}\log\left[\frac{i}{2}\pm\frac{\sqrt{3}}{2}\right]$$
$$= \begin{cases} \frac{1}{i}\left[\log 1 + i\left(\frac{\pi}{6} + 2k\pi\right)\right] \\ \frac{1}{i}\left[\log 1 + i\left(\frac{5\pi}{6} + 2k\pi\right)\right] \\ = \frac{\pi}{6} + 2k\pi \quad \text{or} \quad \frac{5\pi}{6} + 2k\pi \quad (k = 0, \pm 1, \pm 2, \dots). \end{cases}$$

In an entirely analogous way, expressions may be obtained for the other inverse functions. The results may be written in the form

$$\sin^{-1} z = \frac{1}{i} \log \left[iz + (1 - z^2)^{1/2} \right]
 \cos^{-1} z = \frac{1}{i} \log \left[z + (z^2 - 1)^{1/2} \right]
 \tan^{-1} z = \frac{i}{2} \log \frac{1 - iz}{1 + iz}
 \sinh^{-1} z = \log \left[z + (1 + z^2)^{1/2} \right]
 \cosh^{-1} z = \log \left[z + (z^2 - 1)^{1/2} \right]
 \tanh^{-1} z = \frac{1}{2} \log \frac{1 + z}{1 - z}$$
(56a,b,c)

The functions so far considered in this chapter are the basic *elementary functions*. Any linear combination of such functions or any composite function defined in terms of a finite number of such functions is also known as an elementary function.

The *derivative* of a function of a complex variable is defined, as in the real case, by the equation

$$\frac{df(z)}{dz} = f'(z) = \lim_{\Delta z \to 0} \frac{f(z + \Delta z) - f(z)}{\Delta z}$$
(57)

when the indicated limit exists. It is readily verified that the derivative formulas established for elementary functions of a real variable are also valid for the corresponding functions of a complex variable, as defined in this section. 10.4. Analytic functions of a complex variable. A function f(z) is said to be analytic in a region \mathcal{R} of the complex plane if f(z) has a finite derivative at each point of \mathcal{R} and if f(z) is single-valued in \mathcal{R} . It is said to be analytic at a point z if z is an interior point of some region where f(z) is analytic. It will be shown (Section 10.7) that if a function is analytic at a point z, then f(z) has finite derivatives of all orders at z. This remarkable property of analytic functions is of very great usefulness.

To require that a function w = f(z) have a finite derivative at a point z is equivalent to requiring that the limit

$$\frac{dw}{dz} = \lim_{\Delta z \to 0} \frac{f(z + \Delta z) - f(z)}{\Delta z} = \lim_{\Delta z \to 0} \frac{\Delta w}{\Delta z}$$
(58)

exist uniquely as $\Delta z \rightarrow 0$ from any direction in the complex plane. That is, the value approached by the ratio $\frac{\Delta w}{\Delta z}$ must exist for any direction of approach and must not depend upon the direction.

As an example of a simple function which is not analytic *anywhere*, we consider the relation

$$w = x - iy = \overline{z}.$$
 (59)

Here w can be considered as a function of z = x + iy, since if z is given the real and imaginary parts (x and y) of z are determined and hence w is determined. However, if we examine the ratio

$$\frac{\Delta w}{\Delta z} = \frac{\Delta x - i\,\Delta y}{\Delta x + i\,\Delta y},\tag{60}$$

we see that if Δz approaches zero along a line parallel to the real (x) axis, so that $\Delta y = 0$ throughout the limiting process, the limit of (60) is +1, whereas if Δz approaches zero along a line parallel to the imaginary (y) axis, so that $\Delta x = 0$ throughout the process, the limit is -1. More generally, if Δz approaches zero along a curve with slope $\frac{dy}{dx} = m$ at the point considered in the complex plane, there follows

$$\lim_{\Delta z \to 0} \frac{\Delta w}{\Delta z} = \lim_{\substack{\Delta x \to 0 \\ \Delta y \to 0}} \frac{1 - i \frac{\Delta y}{\Delta x}}{1 + i \frac{\Delta y}{\Delta x}} = \frac{1 - mi}{1 + mi} = \frac{(1 - m^2) - 2mi}{1 + m^2},$$

and hence a different limit is approached for each value of m.

Suppose now that the limit (58) *does* exist uniquely, independently of the manner in which $\Delta z \rightarrow 0$. We indicate the real and imaginary parts of w = f(z) by u and v, respectively, where u and v are certain functions of x and y, and so write

$$w = u(x,y) + iv(x,y).$$
 (61)

Then there follows

$$\frac{dw}{dz} = \lim_{\substack{\Delta x \to 0 \\ \Delta y \to 0}} \frac{\Delta u + i \,\Delta v}{\Delta x + i \,\Delta y}.$$
(62)

For an approach parallel to the real axis we set $\Delta y = 0$ first and so obtain

$$\frac{dw}{dz} = \lim_{\Delta x \to 0} \frac{\Delta u + i \Delta v}{\Delta x} = \lim_{\Delta x \to 0} \left(\frac{\Delta u}{\Delta x} + i \frac{\Delta v}{\Delta x} \right) = \frac{\partial u}{\partial x} + i \frac{\partial v}{\partial x}, \quad (63a)$$

where the partial derivatives are formed with y held constant. For an approach parallel to the imaginary axis, we obtain similarly

$$\frac{dw}{dz} = \lim_{\Delta y \to 0} \frac{\Delta u + i \Delta v}{i \Delta y} = \lim_{\Delta y \to 0} \left(\frac{\Delta v}{\Delta y} - i \frac{\Delta u}{\Delta y} \right) = \frac{\partial v}{\partial y} - i \frac{\partial u}{\partial y}.$$
 (63b)

But if f(z) is analytic at the point considered, the two values (63a) and (63b) must be equal,

$$\frac{dw}{dz} = \frac{d(u+iv)}{dz} = f'(z) = \frac{\partial u}{\partial x} + i\frac{\partial v}{\partial x} = \frac{\partial v}{\partial y} - i\frac{\partial u}{\partial y}.$$
 (64)

Hence the real and imaginary parts of the last two expressions must be respectively equal, so that u and v must satisfy the equations

$$\frac{\partial u}{\partial x} = \frac{\partial v}{\partial y}, \qquad \frac{\partial u}{\partial y} = -\frac{\partial v}{\partial x}.$$
 (65a,b)

These equations are known as the Cauchy-Riemann equations. In order that w = u(x,y) + iv(x,y) be analytic in a region \mathscr{R} of the complex plane, it is thus necessary that these equations be satisfied for values of x and y corresponding to points in that region. It can be shown that if these equations are satisfied and if the four partial derivatives involved are continuous in \mathscr{R} , then the derivative exists uniquely for any method of approach to the limit.

Thus, a single-valued complex function w = f(z) = u + iv, for which the first partial derivatives of u and v are continuous in a region \mathcal{R} , is analytic in \mathcal{R} if and only if the Cauchy-Riemann equations (65a,b) are satisfied. It is clear that the function w = x - iy does not satisfy this condition.

From Equations (65a,b) there follows also

$$\frac{\partial^2 u}{\partial x^2} = \frac{\partial^2 v}{\partial x \, \partial y}, \qquad \frac{\partial^2 u}{\partial y^2} = -\frac{\partial^2 v}{\partial y \, \partial x},$$
$$\frac{\partial^2 v}{\partial x^2} = -\frac{\partial^2 u}{\partial x \, \partial y}, \qquad \frac{\partial^2 v}{\partial y^2} = \frac{\partial^2 u}{\partial y \, \partial x}.$$

But if all these partial derivatives are continuous, the order of differentiation is immaterial and it then follows that u and v satisfy the equations

$$\nabla^{2} u = \frac{\partial^{2} u}{\partial x^{2}} + \frac{\partial^{2} u}{\partial y^{2}} = 0$$

$$\nabla^{2} v = \frac{\partial^{2} v}{\partial x^{2}} + \frac{\partial^{2} v}{\partial y^{2}} = 0$$
(66a,b)

The previously mentioned fact that an analytic function has derivatives of *all* orders (Section 10.7) implies, in particular, that the partial derivatives involved *are* indeed continuous. Hence it follows that *the real and imaginary parts of an analytic function are solutions of Laplace's equation*.

The Cauchy-Riemann equations (65a,b) permit the determination of either u or v if the other is known, since, by making use of these equations, we may write

$$du = \frac{\partial u}{\partial x} dx + \frac{\partial u}{\partial y} dy = \frac{\partial v}{\partial y} dx - \frac{\partial v}{\partial x} dy$$

$$dv = \frac{\partial v}{\partial x} dx + \frac{\partial v}{\partial y} dy = -\frac{\partial u}{\partial y} dx + \frac{\partial u}{\partial x} dy$$

(67a,b)

Thus, for example, if we have

v = xy,

then (67a) gives

$$du = x \, dx - y \, dy,$$

and hence

$$u = \frac{1}{2}(x^2 - y^2) + C,$$

where C is an arbitrary real constant. It follows that xy is the imaginary part of the function

$$\frac{1}{2}(x^2 - y^2 + 2ixy) + C = \frac{1}{2}z^2 + C.$$

We notice that if one part of an analytic function is prescribed, the other part, and hence the function itself, is determinate within an arbitrary additive constant.

Two functions u(x,y) and v(x,y), which are respectively the real and imaginary parts of an analytic function, are known as *conjugate functions*.

10.5. Line integrals of complex functions. If C is a curve in the complex plane joining the points z_0 and z_1 , the line integral of a function f(z) = u + iv along C is defined by the equation

$$\int_{C} f(z) dz = \int_{C} (u + iv)(dx + i dy)$$

= $\int_{C} [(u dx - v dy) + i(v dx + u dy)]$
 $\int_{C} f(z) dz = \int_{C} (u dx - v dy) + i \int_{C} (v dx + u dy).$ (68)

or

It is supposed here, and in the remainder of the chapter, that any curve C, along which such an integration is to be effected, is at least piecewise smooth (see Section 6.10).

The real and imaginary parts of (68) are thus ordinary real line integrals in the plane, of the type considered in Section 6.10. We see that the conditions that the two integrands be exact differentials are precisely the requirement that the Cauchy-Riemann equations (66a,b) be satisfied. The results of Sections 6.10 (or 6.17) and 10.4 then show that the line integral $\int_C f(z) dz$ is independent of the path C joining the end points z_0 and z_1 if C can be enclosed in a simply connected region \mathcal{R} inside which f(z) is analytic.*

In such a case the curve need not be prescribed and we may indicate the integral by the notation $\int_{z_0}^{z_1} f(z) dz$. It follows also that here f(z) dz is the exact differential of a function F(z),

$$f(z) dz = dF(z), \tag{69}$$

and that the integral can then be integrated in the usual way, according to the formula

$$\int_{z_0}^{z_1} f(z) \, dz = F(z_1) - F(z_0), \tag{70}$$

where F(z) is a function whose derivative is f(z).

By considering the case where the end points z_0 and z_1 coincide, we conclude that if C is any closed curve lying in a simply connected region \mathscr{R} where f(z)is analytic, then the line integral of f(z) around C will vanish. If we recall further that when f(z) is analytic at all points on a closed curve C it is also analytic in some neighborhood (on both sides) of the curve, we deduce finally that if f(z) is analytic inside and on a closed curve C, then

$$\oint_C f(z) \, dz = 0. \tag{71}$$

This result is known as Cauchy's integral theorem.

In other cases the line integral of a function f(z) around a closed contour may or may not vanish. The *positive direction* around a closed contour is defined, as before, as the direction along which an observer would proceed in keeping the enclosed area to his *left*.

We notice that if u and v are expressed in polar coordinates, Equation (68) is replaced by the equation

$$\int_{C} f(z) dz = \int_{C} (u + iv)(e^{i\theta} dr + ire^{i\theta} d\theta)$$

=
$$\int_{C} [(u \cos \theta - v \sin \theta) dr - (v \cos \theta + u \sin \theta) r d\theta]$$

+
$$i \int_{C} [(v \cos \theta + u \sin \theta) dr + (u \cos \theta - v \sin \theta) r d\theta].$$
(72)

* In order to apply the results of Section 6.10 (or 6.17), we must also exploit the continuity of f'(z), and hence of u_x , u_y , v_x , and v_y in \mathcal{R} (Section 10.7).

The requirement that the quantities in brackets be exact differentials can be reduced to the conditions

$$\frac{\partial u}{\partial r} = \frac{1}{r} \frac{\partial v}{\partial \theta} \\
\frac{1}{r} \frac{\partial u}{\partial \theta} = -\frac{\partial v}{\partial r}$$
(73)

These equations are equivalent to the forms assumed by the Cauchy-Riemann equations (65) in polar coordinates.

To illustrate these results, we notice that for the single-valued function

$$f(z) = \frac{1}{z}$$

the derivative $f'(z) = -1/z^2$ exists at all points except z = 0. Hence f(z) is analytic in any region \mathcal{R} not including the origin in the complex plane. Any closed curve C not surrounding the origin satisfies the conditions of Cauchy's theorem, and hence for any such curve there follows

$$\oint_C \frac{dz}{z} = 0.$$

However, if C encloses the origin, the line integral need not vanish. If C is taken as the unit circle |z| = 1, with center at the origin, then on C we may write

$$z = e^{i\theta}, \qquad dz = ie^{i\theta} d\theta,$$

and hence for this closed curve we obtain

$$\oint_{C_1} \frac{dz}{z} = \int_0^{2\pi} e^{-i\theta} (ie^{i\theta} d\theta) = \int_0^{2\pi} i d\theta = 2\pi i,$$
 (74)

where \oint_{C_1} indicates integration around the positive direction of the unit circle. In this case we may notice that $\frac{dz}{z}$ is the exact differential of the *multi*valued function log z. The value given by (74) is merely the increase experienced by the imaginary part $i\theta$ of log z as z describes a positive closed circuit about the origin. Now consider any other closed curve C which surrounds the origin. If we make a "crosscut" from C_1 to C, and so determine the simply connected region \mathcal{R} shown in Figure 10.3, we see that since f(z) = 1/z is analytic in \mathcal{R} , the integral around the complete boundary of \mathcal{R} must vanish, by Cauchy's theorem. As the width of the cut is decreased toward zero the integrals along the edges of the cut are taken in opposite directions and hence cancel. Since the part of the complete integration which is carried out along C_1 is taken in the negative direction, there follows in the limit

$$\oint_C \frac{dz}{z} - \oint_{C_1} \frac{dz}{z} = 0 \quad \text{or} \quad \oint_C \frac{dz}{z} = \oint_{C_1} \frac{dz}{z},$$

and hence, for any closed curve C enclosing the origin once in the positive direction, we have

$$\oint_C \frac{dz}{z} = 2\pi i. \tag{75}$$

More generally, if $f(z) = z^n$, where n is an integer, there follows

$$\oint_{C_1} z^n dz = \int_0^{2\pi} e^{ni\theta} (ie^{i\theta} d\theta) = i \int_0^{2\pi} e^{(n+1)i\theta} d\theta$$
$$= i \int_0^{2\pi} \left[\cos\left(n+1\right)\theta + i \sin\left(n+1\right)\theta \right] d\theta$$
$$= 0 \qquad (n \neq -1), \tag{76}$$

where C_1 again represents the unit circle |z| = 1. If *n* is a positive integer or zero, this result is in accordance with Cauchy's theorem, since then $f(z) = z^n$



is analytic for all finite values of z. However, if n is a negative integer, z^n is not analytic at z = 0. Still, Equation (76) states that if $n \neq -1$, the integral $\oint_{C_1} z^n dz$ vanishes even in this case. The preceding argument shows that this condition is true also for any closed curve C surrounding the origin, and hence, since z^n is analytic except at the origin, we conclude that, when n is an integer,

Figure 10.3

 $\oint_C z^n \, dz = 0 \qquad (n \neq -1) \quad (77)$ for any closed curve not passing through the

origin. If n = -1, the integral is zero unless C includes the origin, in which case the value is $2\pi i$.

If we replace z by z - a, where a is any complex number, we deduce from these results the additional results

$$\oint_C (z-a)^n dz = 0 \quad (n \neq -1)$$

$$\oint_C \frac{dz}{z-a} = 2\pi i$$
(78a,b)

where C is a closed curve enclosing the point z = a in the positive direction and n is an integer. These results will be of use in Section 10.12. Fundamentally, the mathematical definition of the complex line integral $\int_C f(z) dz$ along a curve C between the points a and b in the complex plane consists of the limit of the sum

$$\sum_{k=0}^{n} f(z_k) (\Delta z)_k, \qquad (\Delta z)_k = z_{k+1} - z_k$$

as the number n + 2 of the division points $z_0 = a, z_1, \ldots, z_{n+1} = b$, along C increases and the chord lengths $|(\Delta z)_k|$ all tend to zero. This definition can be shown to lead to the evaluation (68), in terms of two real line integrals, whenever f(z) is continuous on C and C is piecewise smooth. Repeated use of the inequality (10) leads to the inequality

$$\left|\sum_{k=0}^{n} f(z_k) \left(\Delta z\right)_k\right| \leq \sum_{k=0}^{n} |f(z_k)| \left| (\Delta z)_k \right|,$$

and hence, proceeding to the limit as $n \to \infty$, there follows

$$\left| \int_{C} f(z) \, dz \right| \leq \int_{C} |f(z)| \, |dz| = \int_{C} |f(z)| \, |dx + i \, dy| = \int_{C} |f(z)| \, ds,$$

where s is arc length along C. If now |f(z)| is bounded on C, say

$$|f(z)| \leq M \text{ on } C, \tag{79}$$

the last member of this expression is numerically not greater than

$$\int_C M\,ds = M\int_C ds = ML,$$

where L is the length of the curve C between the specified end points.

Hence we obtain the very useful inequality

$$\left|\int_{C} f(z) \, dz\right| \leq ML,\tag{80}$$

where M is an upper bound for |f(z)| on C, and L is the length of C.

We remark that if C_1 and C_2 are two nonintersecting closed curves, one enclosing the other, and if f(z) is analytic on C_1 and C_2 and in the region \mathscr{R} between C_1 and C_2 , then by introducing a crosscut as in Figure 10.3 and proceeding as in the development leading to (75), we can show that

$$\oint_{C_1} f(z) \, dz = \oint_{C_2} f(z) \, dz$$

It follows that in dealing with an integral of the form $\oint_{C_1} f(z) dz$, we may deform C_1 in a continuous way to any other closed contour C_2 , enclosing or enclosed by C_1 , without changing the value of the integral, so long as f(z) is analytic on C_1 and C_2 and in the region between C_1 and C_2 . We say that C_1 and C_2 are equivalent contours for f(z). 10.6. Cauchy's integral formula. Let C be a closed contour inside which and along which f(z) is analytic, and let α be a point inside C. Finally, let C_{ϵ} be a small circle of radius ϵ , with center at the point α , inside C (Figure 10.4). Then, clearly, the function

$$\frac{f(z)}{z-\alpha}$$

is analytic between C and C_{ϵ} , and so there follows



$$\oint_C \frac{f(z)}{z-\alpha} dz = \oint_{C_{\epsilon}} \frac{f(z)}{z-\alpha} dz \quad (81)$$

for all positive values of ϵ . In particular, Equation (81) must remain true in the limit as $\epsilon \to 0$. But for points on C_{ϵ} we have

$$z = \alpha + \epsilon e^{i\theta}, \quad dz = i\epsilon e^{i\theta} d\theta$$

 $(0 \le \theta < 2\pi), \quad (82)$

and hence there follows

$$\oint_{C_{\epsilon}} \frac{f(z)}{z - \alpha} dz = \int_{0}^{2\pi} f(\alpha + \epsilon e^{i\theta}) i d\theta.$$
(83)

In the limit, as $\epsilon \rightarrow 0$, we then obtain from (81) and (83) the result

$$\oint_C \frac{f(z)}{z-\alpha} dz = \lim_{\epsilon \to 0} \int_0^{2\pi} f(\alpha + \epsilon e^{i\theta}) \, i \, d\theta = 2\pi i f(\alpha), \tag{84}$$

since f(z) is continuous at $z = \alpha$. Equation (84) expresses the value of f(z) at $z = \alpha$ in terms of the values of f(z) along an enclosing curve C. This result can be written in more convenient form if we interchange z and α in (84),

$$f(z) = \frac{1}{2\pi i} \oint_C \frac{f(\alpha)}{\alpha - z} d\alpha, \qquad (85)$$

where C is now a closed contour enclosing the point z. This is Cauchy's integral formula for a function f(z) analytic inside and along a closed curve C.

If we recall that the real and imaginary parts of f(z) are solutions of Laplace's equation inside C, and hence (see Section 9.2) are determined at points inside C by their values on C, we see that (85) affords exactly this determination, if both parts are given on C.

10.7. Taylor series. We next use this result to show that if f(z) is analytic at a point z = a, then f(z) can be expanded in a Taylor series of powers of

z - a. For this purpose we first notice that the function $1/(\alpha - z)$ can be expanded in the geometric series

$$\frac{1}{\alpha - z} = \frac{1}{(\alpha - a) - (z - a)} = \frac{1}{\alpha - a} \frac{1}{1 - \frac{z - a}{\alpha - a}}$$
$$= \frac{1}{\alpha - a} \sum_{n=0}^{\infty} \left(\frac{z - a}{\alpha - a}\right)^n \quad (|z - a| < |\alpha - a|).$$
(86)

We now take for C any circle, with center at z = a, inside which and on which f(z) is analytic (Figure 10.5). Then for points inside C we have

$$|z-a|<|\alpha-a|=R,$$

where R is the radius of the circle, and hence for such points the power series (86) converges and also can be multiplied by $f(\alpha)$ and integrated around the curve C term by term. Thus (86) can be introduced into (85) to give the result

$$f(z) = \sum_{n=0}^{\infty} \frac{1}{2\pi i} \left[\oint_C \frac{f(\alpha) d\alpha}{(\alpha - a)^{n+1}} \right] (z - a)^n \qquad (|z - a| < R),$$

or, equivalently,

$$f(z) = \sum_{n=0}^{\infty} A_n (z-a)^n \qquad (|z-a| < R), \qquad (87a)$$

$$A_n = \frac{1}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{(\alpha - a)^{n+1}}.$$
(87b)

where

Thus, if f(z) is analytic throughout the circle |z - a| < R, it can be expanded in a Taylor series (87a) converging in this circle, with coefficients given by (87b). By differentiating (87a) k times, setting z = a in the result, and using (87b), we obtain the additional formula

$$f^{(k)}(a) = k! A_k = \frac{k!}{2\pi i} \oint_C \frac{f(\alpha) d\alpha}{(\alpha - a)^{k+1}}$$

or, with a change in notation,

$$f^{(n)}(z) = \frac{n!}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{(\alpha - z)^{n+1}},\tag{88}$$

where C is now the circle with radius R about z, or any equivalent contour (Figure 10.6). We notice that (88) is obtained formally by differentiating (85) n times under the integral sign.
From these results we conclude that if f(z) is analytic at a point z = a (and hence in some circle around this point), derivatives of f(z) of all orders exist at that point, and f(z) can be expanded in the Taylor series

$$f(z) = \sum_{n=0}^{\infty} \frac{f^{(n)}(a)}{n!} (z-a)^n,$$
(89)

the circle of convergence coinciding with (or including) the largest circle with center at z = a throughout which f(z) is analytic.







Conversely, we see also that if f(z) can be represented by a Taylor series in some circle about z = a, then f(z) must be analytic at z = a, since then derivatives of all orders exist at and near z = a.

Generally the radius of convergence R_c of the Taylor series (87) will equal the distance R from z = a to the nearest point where f(z) ceases to be analytic. However, there exist somewhat unusual situations in which $R_c > R$. For example, suppose that f(z) has the definition

$$f(z) = \begin{cases} 1 & \text{inside } C_0 \\ 0 & \text{outside } C_0, \end{cases}$$
(90)

where C_0 is a closed curve enclosing the point z = a. Then R is the distance from z = a to the nearest point on C_0 . But, since f(a) = 1 and $f^{(n)}(a) = 0$ for $n \ge 1$, the series (87) will reduce to the *one-term* series f(z) = 1 (|z| < R). Clearly, the series itself converges for all values of z, so that here $R_c = \infty$. In fact, the *series* defines a function $f^*(z) = 1$ for all z, which is analytic *everywhere* and which is identical with f(z) inside C_0 . Whereas the definition (90) may seem to be somewhat artificial, it may be noted that the more compact definition

$$f(z) = \frac{1}{2\pi i} \oint_{C_0} \frac{d\alpha}{\alpha - z}$$
(90')

is completely equivalent to it when C_0 is respectable.

More generally, the only situations in which $R_c > R$ are those in which f(z) is so defined that the neighborhood \mathscr{R} of z = a inside which f(z) is analytic is finite, but in which it is possible to redefine f(z) outside that region in such a way that f(z) is analytic in an extended region \mathscr{R}^* which includes \mathscr{R} . Such a process of extending the region of definition of an analytic function is called *analytic continuation*.

10.8. Laurent series. A generalization of the Taylor series expansion of f(z), which may involve both positive and negative integral powers of z - a, exists



Figure 10.7

Figure 10.8

whenever there is a circular ring or annulus \mathscr{R} , for which $R_1 < |z - a| < R_2$, such that f(z) is analytic (and single-valued) in \mathscr{R} and on its inner and outer circular boundaries C_1 and C_2 (Figure 10.7).

As a first step toward its derivation, we introduce a crosscut between C_1 and C_2 and apply Cauchy's integral formula to the resultant simply connected region \mathscr{R}' bounded by the curve C' (Figure 10.8). By considering the limit as the width of the cut tends to zero, we thus deduce that, for any point z in the annulus \mathscr{R} , there follows

$$f(z) = \frac{1}{2\pi i} \oint_{C_2} \frac{f(\alpha)}{\alpha - z} d\alpha - \frac{1}{2\pi i} \oint_{C_1} \frac{f(\alpha)}{\alpha - z} d\alpha, \qquad (91)$$

where both C_1 and C_2 are to be traversed in the positive (counterclockwise) direction.

The first integral in (91) now is dealt with exactly as in Section 10.7, by expanding $1/(\alpha - z)$ in powers of $(z - a)/(\alpha - a)$ and integrating term by

term, to yield the relation

$$\frac{1}{2\pi i} \oint_{C_2} \frac{f(\alpha)}{\alpha - z} d\alpha = \sum_{n=0}^{\infty} A_n (z - a)^n \qquad (|z - a| < R_2), \qquad (92)$$

where again

$$A_{n} = \frac{1}{2\pi i} \oint_{C_{2}} \frac{f(\alpha) \, d\alpha}{(\alpha - a)^{n+1}} \qquad (n = 0, 1, 2, ...).$$
(93)

The important point here is that the equal members of (92) now generally do not also equal f(z), nor does $n!A_n$ generally equal $f^{(n)}(a)$, since now f(z) is not necessarily analytic everywhere inside C_2 .

In the second integral of Equation (91), we instead expand $1/(\alpha - z)$ in powers of $(\alpha - a)/(z - a)$,

$$-\frac{1}{\alpha-z} = \frac{1}{(z-a)-(\alpha-a)} = \frac{1}{z-a} \frac{1}{1-\frac{\alpha-a}{z-a}}$$
$$= \sum_{n=1}^{\infty} \frac{(\alpha-a)^{n-1}}{(z-a)^n} \quad (|z-a| > R_1),$$

so that, after multiplying this convergent power series by $f(\alpha)/(2\pi i)$ and integrating term by term around C_1 , there follows

$$-\frac{1}{2\pi i} \oint_{C_1} \frac{f(\alpha)}{\alpha - z} \, d\alpha = \sum_{n=1}^{\infty} B_{-n}(z - a)^{-n}$$
$$= \sum_{n=-\infty}^{-1} B_n(z - a)^n \quad (|z - a| > R_1), \quad (94)$$

where

$$B_n = \frac{1}{2\pi i} \oint_{C_1} \frac{f(\alpha) \, d\alpha}{(\alpha - a)^{n+1}} \qquad (n = -1, -2, \dots). \tag{95}$$

Now, since Equation (92) is true for z inside C_2 and (94) is true for z outside C_1 , both are true for z in the ring \mathscr{R} , and hence then can be introduced into (91). Further, since $f(\alpha)/(\alpha - a)^{n+1}$ is an analytic function of α when α is in \mathscr{R} , when n is zero or any positive or negative integer, it follows that, in (93) and (95), C_1 and C_2 can be deformed into any closed curve C which surrounds z = a and lies in \mathscr{R} .*

Hence, finally, if we write

$$a_n = \frac{1}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{(\alpha - a)^{n+1}} \qquad (n = 0, \, \pm 1, \, \pm 2, \, \dots), \tag{96}$$

* It should be noted that this statement generally does *not* apply to the integrals in Equation (91), since $f(\alpha)/(\alpha - z)$ generally is not an analytic function of α when $\alpha = z$.

there follows $A_n = a_n$ when $n \ge 0$ and $B_n = a_n$ when $n \le -1$, and the result of introducing (92) and (95) into (91) thus becomes

$$f(z) = \sum_{n=-\infty}^{\infty} a_n (z-a)^n \qquad (R_1 < |z-a| < R_2). \tag{97}$$

This expansion is known as the *Laurent series* expansion of f(z), in the specified annulus \mathcal{R} , and is seen to involve both positive and negative powers of z - a, in the general case.

When f(z) also is analytic inside and on C_1 , (96) shows that $a_n = 0$ when n < 0, and (97) then reduces to the Taylor series (87), valid everywhere inside C_2 . On the other hand, when f(z) also is analytic outside and on C_2 , the series involves only a constant term and *negative* powers of z - a and can, in fact, be considered as an ordinary power series in $(z - a)^{-1}$, valid everywhere outside C_1 .

Just as in the case of ordinary power series, it can be proved that, within its annular region of convergence, a Laurent series can be integrated (or differentiated) term by term, and the result will converge to the integral (or derivative) of the function represented by the original series.

Further, it is known that the expansion (97) is *unique*, in the sense that if an expansion of the form

$$f(z)=\sum \alpha_n(z-a)^n$$

can be obtained in any way, and is valid in the specific annulus

$$R_1 < |z-a| < R_2,$$

then necessarily $\alpha_n = a_n$. This fact frequently permits one to obtain a desired Laurent series by elementary methods, without attempting to evaluate the coefficients by using (96).

In illustration, from the familiar expansion $e^t = \sum_{0}^{\infty} t^n/n!$, which is valid for all t we may deduce other relations, such as

valid for all t, we may deduce other relations, such as

$$\frac{e^{z}}{z} = \frac{1}{z} + 1 + \frac{z}{2!} + \frac{z^{2}}{3!} + \cdots + \frac{z^{n}}{(n+1)!} + \cdots \qquad (z \neq 0)$$

and

$$e^{1/z} = 1 + \frac{1}{1! z} + \frac{1}{2! z^2} + \cdots + \frac{1}{n! z^n} + \cdots (z \neq 0).$$

Both these series are Laurent expansions with a = 0, $R_1 = 0$, and $R_2 = \infty$. Also we have, for example, the geometric series expansions

$$\frac{1}{1-z} = 1 + z + z^2 + \cdots + z^n + \cdots \quad (|z| < 1)$$

and

$$\frac{1}{1-z} = \frac{-(1/z)}{1-(1/z)} = -\frac{1}{z} \left(1 + \frac{1}{z} + \cdots \right)$$
$$= -\frac{1}{z} - \frac{1}{z^2} - \frac{1}{z^3} - \cdots - \frac{1}{z^n} - \cdots - (|z| > 1).$$

The first of these expansions is the Taylor series valid inside the circle |z| = 1, the second the Laurent series valid outside that circle.

More generally, if a Laurent expansion of f(z) is required in powers of z - a, in the annulus $R_1 < |z - a| < R_2$, we may attempt to express f(z) as the sum (or product) of two functions $f_1(z)$ and $f_2(z)$, such that $f_1(z)$ is analytic everywhere *outside* the *inner* circle $|z - a| = R_1$ and $f_2(z)$ is analytic everywhere *inside* the *outer* circle $|z - a| = R_2$. The desired representation then can be obtained by expanding $f_2(z)$ in an ordinary power series in z - a, expanding $f_1(z)$ in an ordinary power series in $(z - a)^{-1}$, and adding (or multiplying) together the results.

Thus, for example, the function

$$f(z) = \frac{2}{z(z-1)(z-2)}$$

clearly is analytic except when z = 0, 1, and 2. To expand f(z) in a Laurent series of powers of z in the annulus \Re : 1 < |z| < 2, we may first write

$$f(z) = \frac{1}{z} - \frac{2}{z-1} + \frac{1}{z-2}.$$

Since the first two terms are analytic when |z| > 1, whereas the third term is analytic when |z| < 2, we may expand the first two terms in powers of 1/z and the third in powers of z, to obtain

$$f(z) = \frac{1}{z} - \frac{2/z}{1 - (1/z)} - \frac{1/2}{1 - (z/2)}$$

= $\frac{1}{z} - \frac{2}{z} \left(1 + \frac{1}{z} + \frac{1}{z^2} + \dots + \frac{1}{z^n} + \dots \right)$
 $- \frac{1}{2} \left(1 + \frac{z}{2} + \frac{z^2}{4} + \dots + \frac{z^n}{2^n} + \dots \right).$

Thus there follows

$$f(z) = \sum_{n=-\infty}^{\infty} a_n z^n \qquad (1 < |z| < 2)$$

where $a_n = -1/2^{n+1}$ when $n \ge 0$, $a_{-1} = -1$, and $a_n = -2$ when $n \le -2$. The same function also possesses two other Laurent series in powers of z, valid when 0 < |z| < 1 and when $2 < |z| < \infty$, as well as either three or four Laurent (or Taylor) series in powers of z - a, for any other value of a.

10.9. Singularities of analytic functions. If any region \mathscr{R} exists such that f(z) or a branch of f(z) is analytic in \mathscr{R} , then we speak of f(z) as "an analytic function." Points at which f(z) is not analytic are called singular points or singularities of the function. Also, a point at which a branch of a multivalued function f(z) is not analytic is called a singular point of f(z). From the definition, we see that any point at which $\frac{df}{dz}$ does not exist is a singular point. Also, when f(z) is multivalued, any point which cannot be an interior point of the region of definition of a single-valued branch of f(z) is a singular point.

Points of the latter type are known as branch points. At such points the first derivative may or may not exist, but it can be shown that a derivative of some order will fail to exist. The characteristic feature of a branch point consists of the fact that if a point describes a small closed circuit (enclosing no other singular points) about such a point, the value assumed by the function after the circuit differs from the initial value. In Section 10.3 we have seen that the function log z and functions of the form z^k , where k is not an integer, both

have a branch point at z = 0. Since $\frac{d}{dz} \log z = \frac{1}{z}$ is analytic elsewhere, this is the only singularity of log z in the finite part of the plane. Similarly, since $\frac{d}{dz} z^k = kz^{k-1}$, we see that also z^k has no other finite singularity. Unless the real part of k is smaller than unity, reference to Equation (47) shows that the first derivative of z^k is finite at z = 0. However, it is clear that if successive derivatives are calculated, eventually a power of z with a negative real part

will be obtained (when k is not a positive integer or zero), and hence the corresponding derivative will not exist at the branch point. It is seen that the functions $\log (z - a)$ and $(z - a)^k$ both have a branch point at z = a, with the above restriction on k.

Reference to Sections 10.2 and 10.3 then shows that the elementary functions can have branch points only for those values of z for which a quantity raised to a nonintegral power vanishes, or for which a quantity whose logarithm is taken vanishes or becomes infinite. It is assumed, for this purpose, that inverse trigonometric and hyperbolic functions are expressed in terms of the logarithm, as in Equations (55) and (56).

Thus, for example, it is to be expected that the function

$$f(z) = (1 - z^2)^{1/2} = (1 + z)^{1/2} (1 - z)^{1/2}$$

has branch points at z = 1 and at z = -1. This is readily seen if we notice, for example, that near z = 1 the function f(z) behaves like $(1 + 1)^{1/2}(1 - z)^{1/2}$ $= \sqrt{2} (1 - z)^{1/2}$. Similarly, the function $f(z) = \log (z - z^2)$ can be written in the form

$$f(z) = \log [z(1-z)] = \log z + \log (1-z).$$

Thus we conclude that this function has branch points at z = 0 and at z = 1. In the cases of the inverse circular and hyperbolic functions, reference to Equations (55) and (56) shows that the functions $\sin^{-1} z$, $\cos^{-1} z$, $\cosh^{-1} z$, and $\tanh^{-1} z$ have branch points at $z = \pm 1$, whereas the functions $\tan^{-1} z$ and $\sinh^{-1} z$ have branch points at $z = \pm i$. The expressions for the first derivatives of these functions show that there are no other finite singular points.

In working with multivalued functions (possessing one or more branch points) it is usually desirable to specify one particular *branch* of the functions and artificially prevent the possibility of transition from that branch to another



Figure 10.9

Figure 10.10

branch. Clearly, this result may be accomplished by forbidding curves from surrounding the branch points.

In the case of the function $w = \log z$, it is conventional to imagine the complex plane to be "cut" along the entire positive real axis, so that transition across this part of the axis is impossible (Figure 10.9). In the resultant "cut plane" any convenient branch of $\log z$, say the branch for k = 0,

$$\log z = \log r + i\theta_P \qquad (0 \le \theta_P < 2\pi),$$

is a single-valued function of z, which is *analytic* everywhere except on the *cut*. It is seen that the cut or "barrier" could equally well be made along any other ray or curve extending from the branch point z = 0 to infinity.

For the function $w = z^k$, where k is nonintegral, the principal value of θ is frequently defined by $-\pi < \theta_P \le \pi$, so that the cut is taken along the *negative* real axis (Figure 10.10).

For such a function as $w = \tan^{-1} z = \frac{i}{2} \log \frac{1 - iz}{1 + iz}$, with branch points at $z = \pm i$, two cuts may be made along the imaginary axis, from the two branch points outward to infinity, thus avoiding contours which surround *either* of the branch points [Figure 10.11(a)]. In this case it is readily verified that a circuit which encloses *both* branch points does not introduce a transition from branch to branch. Thus, in place of cutting the plane as described, we could

merely introduce a *finite* cut *joining the two branch points*, and hence avoiding contours which include only one of the branch points [Figure 10.11(b)]. The method of cutting to be preferred depends upon the particular problem considered. Further facts bearing on the nature of the cuts are brought out in Section 10.10.

Thus, in place of dealing with multivalued functions, we may always introduce a suitably cut plane such that only one particular *branch* of the function need be considered. Any such branch is then single-valued in the cut plane.

If it is necessary to retain the possibility of transition from branch to branch, we may think of constructing a series of superimposed planes cut and



Figure 10.11

joined along the cuts in such a way that if a point starts on a given plane, moves around a branch point, and approaches its original position after such a circuit, it moves across a cut from the initial plane onto another superimposed plane corresponding to a second branch of the function. For the function $w = \log z$ an infinite number of such planes would be required, and the resultant configuration would resemble an endless helicoidal surface. For the function $w = z^{1/2}$ only two such planes would be needed, the junctions being so arranged that as a point describes a continuous contour surrounding the origin it moves from the first "sheet" to the second after the first circuit, and returns from the second sheet to the first after a second circuit. Such arrangements are known as *Riemann surfaces*.

In the remainder of this work, however, we will think of the plane as cut in such cases, so that all branch points are removed and the functions considered can be taken to be single-valued. The remaining singular points of a function df

f(z) are then points at which the derivative $\frac{df}{dz}$ does not exist.

In particular, if f(z) is analytic everywhere throughout some neighborhood of a point z = a, say inside the circle C: |z - a| = R, except at the point z = a itself, then z = a is called an *isolated singular point* of f(z). The results of Section 10.8 show that a Laurent expansion of the form

$$f(z) = \sum_{n=-\infty}^{\infty} a_n (z-a)^n \qquad (0 < |z-a| < R)$$
(98)

then is valid.

Now if f(z) were to remain finite as $z \rightarrow a$, it would necessarily follow that all *a*'s with negative subscripts in (98) would be zero. Thus, everywhere inside the circle *C* except *at* z = a, f(z) would be expressible in the form

$$f(z) = a_0 + a_1(z-a) + a_2(z-a)^2 + \cdots$$

If it were also true that $f(a) = a_0$, then f(z) would be analytic everywhere inside C. That is, f(z) would become analytic at z = a if it were suitably defined (or redefined) there. Such a point is often called a "removable singular point." For example, the functions $f_1(z) = z/z$ and $f_2(z) = (\sin z)/z$ are undefined at z = 0, but both functions approach 1 as $z \rightarrow 0$. If we define $f_1(0) = 1$ and $f_2(0) = 1$, then f_1 and f_2 are analytic at z = 0 (and elsewhere). If we were to artificially define $f_1(0)$ or $f_2(0)$ to be 0, say, or to refuse to define $f_1(0)$ or $f_2(0)$ at all, then $f_1(z)$ or $f_2(z)$ would have a "removable singularity" at z = 0.

Henceforth, we will assume that all such "removable singularities" have indeed been removed by suitable definition.

Apart from such artificialities, it thus follows that f(z) cannot remain finite at an isolated singular point. To illustrate such points, we may notice that the function w = 1/z has a singular point at z = 0, the function $w = 1/(z^2 + 1)$ has singular points at $z = \pm i$, and the function $w = \cot z$ has singular points where $z = k\pi$ ($k = 0, \pm 1, \pm 2, ...$). These singularities are examples of points known as poles. Specifically, we say that if f(z) is not finite at z = a but if for some integer m the product

$$(z-a)^m f(z)$$

is analytic at z = a, then f(z) has a pole at z = a.* If m is the smallest integer for which this is so, the pole is said to be of order m.

It follows that any rational function of the form

$$w = \frac{A_0 + A_1 z + \dots + A_M z^M}{B_0 + B_1 z + \dots + B_N z^N} = \frac{N(z)}{D(z)},$$
(99)

where the numerator and denominator are *polynomials* of degree M and N, respectively, without common factors, can have only *poles*, such poles occurring at points where the denominator D(z) vanishes. Thus if $z - a_1$ is a simple

^{*} In accordance with the previously introduced convention, it is meant here that $(z - a)^m f(z)$ becomes analytic at z = a when it is defined to take on its limiting value at that point.

factor of the denominator, the product $(z - a_1)w$ is analytic at $z = a_1$ and the point $z = a_1$ is a pole of order one, or a so-called simple pole. If $z = a_1$ is an *m*-fold zero of the denominator, so that $(z - a_1)^m$ is a factor of D(z), then $z = a_1$ is a pole of order *m*.

Thus the function

$$f(z) = \frac{z^3 - 2z + 1}{z^5 + 2z^3 + z} = \frac{z^3 - 2z + 1}{z(z + i)^2(z - i)^2}$$

has a simple pole at z = 0 and double poles at $z = \pm i$ because zf(z) is clearly analytic at z = 0, whereas $(z - i)^2 f(z)$ is analytic at z = i and $(z + i)^2 f(z)$ is analytic at z = -i. That is, each product listed possesses a derivative at and near the point noted.

Functions other than rational functions may also have poles. Thus the function $f(z) = \csc z$ is not finite at the points $z = 0, \pm \pi, \pm 2\pi, \ldots$ Considering first the point z = 0, we see that

$$z f(z) = \frac{z}{\sin z} = \left(\frac{\sin z}{z}\right)^{-1}$$

is finite at z = 0, since

$$\lim_{z \to 0} \frac{\sin z}{z} = \lim_{z \to 0} \frac{z - \frac{z^3}{3!} + \frac{z^5}{5!} - \cdots}{z} = \lim_{z \to 0} \left[1 - \frac{z^2}{3!} + \cdots \right] = 1.$$

Also, z f(z) has a finite derivative at z = 0, since

$$\frac{d}{dz}[zf(z)] = \frac{\sin z - z \cos z}{\sin^2 z} = \frac{\left(z - \frac{z^3}{3!} + \cdots\right) - z\left(1 - \frac{z^2}{2!} + \cdots\right)}{\left(z - \frac{z^3}{3!} + \cdots\right)^2}$$
$$= \frac{\frac{z^3}{3!} + \cdots}{\frac{z^3}{z^2 + \cdots}} = \frac{z}{3} + \cdots,$$

and hence $\frac{d}{dz} [zf(z)] = 0$ when z = 0. Thus f(z) has a simple pole at z = 0. Similarly, at a singular point $z = k\pi$ we may make the substitution

$$t=z-k\pi.$$

There then follows

$$(z-k\pi)f(z)=\frac{t}{\sin\left(t+k\pi\right)}=(-1)^k\frac{t}{\sin t}.$$

Thus, since $t \to 0$ as $z \to k\pi$, we see that, except for sign, $(z - k\pi)f(z)$ behaves near $z = k\pi$ just as zf(z) behaves near z = 0, and hence f(z) has simple poles at all the points $z = k\pi$ (k an integer). From the expression for f'(z),

$$f'(z)=-\frac{\cos z}{\sin^2 z},$$

we see that f'(z) exists at all other points, and hence these points are the only singular points of the function in the finite part of the plane.

From Equation (98) we may deduce that f(z) has a pole of order m at z = a if and only if f(z) has a Laurent expansion (98) in the neighborhood of z = a, with $a_n = 0$ when n < -m and with $a_{-m} \neq 0$.

All isolated singular points which are not poles (and are not removable) are called *essential singular points*. Thus, when f(z) has an isolated singularity

at a point z = a, that singularity is *essential* if the derivative $\frac{df}{dz}$ does not exist at z = a and if there is no integer *m* for which

$$\frac{d}{dz}\left[(z-a)^m f(z)\right]$$

exists at z = a. It should be noted that a branch point is not an isolated singularity, since any circle with center at that point must also contain points on some branch cut passing through that point, if the branch considered is to be single-valued.

As an example of an essential singular point, we notice that for the function $f(z) = e^{1/z}$ the derivative

$$f'(z) = -\frac{1}{z^2} e^{1/z}$$

exists everywhere except at the origin. Thus the origin is an isolated singular point. For *real* values of z, the function $f(x) = e^{1/x}$ is seen to approach zero as $x \to 0$ from *negative* values, and to become *infinite* as $x \to 0$ from positive values. Thus we see that $e^{1/z}$ approaches zero if $z \to 0$ along the negative real axis and that the absolute value of $e^{1/z}$ increases without limit as $z \to 0$ along the positive real axis. Since the same is true for $z^m e^{1/z}$, for any value of m, we see that $z^m f(z)$ is not analytic at z = 0 for any value of m, and hence the function has an essential singularity at z = 0.

To investigate in more detail the behavior of f(z) near the origin, we write $z = r e^{i\theta}$ and so express f(z) in the form

$$f(z) = e^{1} e^{-i\theta} = e^{\frac{1}{r}(\cos\theta - i\sin\theta)}$$
$$= e^{\frac{\cos\theta}{r}} \left[\cos\left(\frac{\sin\theta}{r}\right) - i\sin\left(\frac{\sin\theta}{r}\right) \right].$$

In particular, we have

$$|f(z)| = |e^{\frac{1}{z}}| = e^{\frac{\cos\theta}{r}}.$$

It follows that if $z \to 0$ along any curve for which $\cos \theta > 0$ as $r \to 0$, the absolute value of f(z) increases without limit, whereas if $z \to 0$ along any curve for which $\cos \theta < 0$ as $r \to 0$, then $f(z) \to 0$. Now suppose that $z \to 0$ along a curve $r = c^{-1} \cos \theta$, that is, along a circle of diameter 1/c tangent to the imaginary axis. For such an approach we have always $|f(z)| = e^c$, and hence |f(z)| can be made to *take on any positive value* as $z \to 0$ along a properly chosen circle of this kind. On such a circle we have also

$$f(z) = e^{c} [\cos (c \tan \theta) - i \sin (c \tan \theta)].$$

Now as $z \to 0$ along this circle the angle θ tends toward $\pi/2$, and hence the argument of the circular functions increases without limit. Consequently, it is seen that f(z) takes on all complex values with absolute value e^c infinitely many times. Thus it follows that in any neighborhood of z = 0 the function $e^{1/z}$ takes on all values except zero an infinite number of times! Further, the exceptional zero value is approached as $z \to 0$ along the negative real axis, or along any curve which passes through the origin along a tangent in the second or third quadrant.

Picard has shown that any analytic function having an isolated essential singularity behaves in this remarkable way near the singularity, all real and imaginary values being *approached* and all values *except one* being actually *taken on infinitely often* in any arbitrarily small region including the singularity.

From Equation (98) we may deduce that f(z) has an isolated essential singular point at z = a if and only if f(z) has a Laurent expansion in the neighborhood of z = a with infinitely many negative powers of z - a.

In addition to the isolated singularities and branch points, still other types of singular points may exist. For example, in the case of the function

$$f(z) = \frac{1}{2\pi i} \oint_{C_0} \frac{d\alpha}{\alpha - z} = \begin{cases} 1 & \text{inside } C_0 \\ 0 & \text{outside } C_0 \end{cases}$$
(100)

mentioned in Section 10.8, all points on the closed curve C_0 are singular points. One might speak of such points as "artificial boundary points" since they comprise a curve across which a transition occurs from one domain, inside which a first analytic function $f_1(z)$ is defined, to another domain in which a second analytic function $f_2(z)$ is defined. In this example there is no difficulty in extending the domain of definition of either $f_1(z)$ or $f_2(z)$ across C_0 , so that points on C_0 are not singularities for either of the two analytic functions f_1 and f_2 .

However, there exist functions, defined on one side of a curve C, such that their definitions cannot be extended analytically across C. Such a curve is called a *natural boundary* for the associated function, and points on such a curve are usually referred to as (nonisolated) *essential singularities*.

When branches of a multivalued function are defined, the points on *branch* cuts are also singular points, but are rather artificially so since (except for the *branch points* themselves) they would no longer be exceptional if the branches of the parent function were redefined.

In addition, it may happen that isolated singular points *cluster* about a point z = a, in such a way that there are such points in *every* circle with center at z = a, no matter how small the radius. Then z = a also is a singular point and is usually included in the category of nonisolated "essential" singularities.

10.10. Singularities at infinity. To study the behavior of a function f(z) for large values of |z|, we may make the substitution

$$z = \frac{1}{t} \,. \tag{101}$$

The function f(z) is then transformed to a new function g(t),

$$g(t) = f\left(\frac{1}{t}\right). \tag{102}$$

Now as a point z moves indefinitely far from the origin *in any direction*, so that $|z| \rightarrow \infty$, the corresponding point t approaches t = 0. It is convenient to say that the point t = 0 corresponds to a "point at infinity" in the complex plane which may be denoted by $z = z_{\infty}$, such that this point is approached by any point z receding from the origin indefinitely far in any direction. Thus we "extend" or "close" the complex plane by adding the point z_{∞} "at infinity." It is then natural to say that f(z) is analytic at z_{∞} if g(t) = f(1/t) is analytic at t = 0, and similarly that if g(t) has a particular type of singularity at t = 0, then f(z) has the same type of singularity at z_{∞} .

If g(t) = f(1/t) is analytic at t = 0, then it can be expanded in a series

$$f\left(\frac{1}{t}\right) = A_0 + A_1t + A_2t^2 + \cdots = \sum_{n=0}^{\infty} A_nt^n$$

in some circle about t = 0, and conversely. Hence, replacing t by 1/z, it follows that f(z) is analytic at z_{∞} if and only if it can be represented by a series of the form

$$f(z) = A_0 + \frac{A_1}{z} + \frac{A_2}{z^2} + \cdots = \sum_{n=0}^{\infty} \frac{A_n}{z^n}$$
(103)

for values of z outside a sufficiently large circle with center at z = 0. In particular, f(z) must approach a limit $f(\infty)$ as $z \to z_{\infty}$, that is, as $|z| \to \infty$ in any way, and Equation (103) shows that

$$A_0 = f(\infty). \tag{104}$$

We notice also that if f(1/t) is analytic at t = 0, then

$$\frac{d}{dt}f\left(\frac{1}{t}\right) = -\frac{1}{t^2}f'\left(\frac{1}{t}\right)$$

must exist at t = 0, and hence $z^2 \frac{df}{dz}$ must exist at z_{∞} .

If f(1/t) has a pole of order m at t = 0, then $t^m f(1/t)$ can be expanded in a power series in t converging near t = 0, and conversely, and hence near t = 0 we may write

$$f\left(\frac{1}{t}\right) = \frac{A_0}{t^m} + \frac{A_1}{t^{m-1}} + \dots + A_m + A_{m+1}t + \dots = \sum_{n=0}^{\infty} A_n t^{-m+n}.$$

Thus we conclude that f(z) has a pole of order m at z_{∞} if and only if we can write

$$f(z) = A_0 z^m + A_1 z^{m-1} + \dots + A_m + \frac{A_{m+1}}{z} + \dots$$
$$= \sum_{n=0}^{\infty} A_n z^{m-n}$$
(105)

for values of z outside a sufficiently large circle with center at z = 0. In particular, $f(z)/z^m$ must approach a limit as $z \to z_{\infty}$, and this limit is identified with the coefficient A_0 in (105),

$$A_0 = \lim_{z \to \infty} \frac{1}{z^m} f(z).$$
(106)

Also, since $\frac{d}{dt} t^m f\left(\frac{1}{t}\right)$ must exist when t = 0, it follows that the expression

$$-z^{2}\frac{d}{dz}\left\lfloor\frac{f(z)}{z^{m}}\right\rfloor = \frac{m}{z^{m-1}}f(z) - \frac{1}{z^{m-2}}\frac{df(z)}{dz}$$

must exist as $z \rightarrow z_{\infty}$.

From (105) we conclude that any polynomial of degree m has a pole of order m at z_{∞} . If we consider the general rational function

$$f(z) = \frac{a_0 + a_1 z + \dots + a_N z^N}{b_0 + b_1 z + \dots + b_M z^M},$$
(107)

with $a_N b_M \neq 0$, we obtain

$$f\left(\frac{1}{t}\right) = \frac{a_0 t^N + a_1 t^{N-1} + \dots + a_N}{b_0 t^M + b_1 t^{M-1} + \dots + b_M} t^{M-N}.$$

Thus it follows that f(1/t) is analytic at t = 0 if $M \ge N$ and that f(1/t) has a pole of order N - M at t = 0 if N > M. The same is thus true for f(z) at z_{∞} .

Since f(z) clearly is analytic at all finite points except at zeros of the denominator, where poles exist, it follows that any rational function possesses no singularities other than poles. If we count a pole of order k as equivalent to k

simple poles, we see that since the denominator has M linear factors, the multiplicity of *finite poles* is M. Similarly, the multiplicity of *finite zeros* is N. If M > N, the function f(1/t) has a zero of order M - N at t = 0, and hence f(z) has a zero of order M - N at z_{∞} . In this case there are exactly M poles and M zeros (counting multiple poles and zeros separately). Similarly, if M < N, there are N poles and N zeros. In any case, for a rational function the total multiplicity of poles is equal to the total multiplicity of zeros, when the point z_{∞} is taken into account.

Since the expansions of e^z , sin z, cos z, sinh z, and cosh z involve all positive powers of z, and converge for all finite values of z, reference to the criterion (105) shows that although these functions have no finite singularities, they each have an essential singularity at z_{∞} . In the case of the exponential function this situation may also be seen by noticing that e^z behaves for large |z| as $e^{1/z}$ behaves for small |z|.

For the function $f(z) = \log z$ there follows $f(1/t) = \log (1/t) = -\log t$. Hence it follows that $f(z) = \log z$ has a branch point at z_{∞} as well as at z = 0. A similar statement applies to $f(z) = z^k$, where k is nonintegral. The "cuts" introduced in Section 10.9 for these functions can be considered as joining the two branch points.

We notice that a *small* circle surrounding the point t = 0 corresponds to a *large* circle with center at z = 0 in the complex plane. Thus, just as we may consider a "circle of zero radius" as enclosing only the origin, we may in a similar sense consider the exterior of a circle C_{∞} "of infinite radius" as consisting only of the point z_{∞} . Consequently, the exterior of a closed contour C may be considered as composed of the region between C and the infinite circle C_{∞} together with an added exterior point z_{∞} at infinity.

Also, a *positive* circuit around the *finite* area enclosed by any closed contour C can be considered also as a *negative* circuit around the *infinite* area *outside* C, *including the point* z_{∞} .

If f(z) is analytic on C and at all *finite* points outside C, it follows that C and C_{∞} are equivalent contours, and hence

$$\oint_C f(z) \, dz = \oint_{C_{\infty}} f(z) \, dz. \tag{108}$$

The right-hand integral may now be considered as taken along a *negative* contour enclosing the exterior of C_{∞} , that is, enclosing only the immediate neighborhood of the point z_{∞} . However, even though f(z) be analytic at z_{∞} , this integral may not vanish. It can be shown to vanish if $z^2 f(z)$ is analytic at z_{∞} (see Problem 59). This fact leads to the result

$$\oint_C f(z) \, dz = 0 \tag{109}$$

in the case when $z^2 f(z)$ is analytic on and outside C and at z_{∞} , as well as in the case when f(z) is analytic on and inside C.

If outside a closed contour C no points, including the point z_{∞} , are branch points, and if we think of a closed circuit around C as surrounding the exterior of C, we may expect that a function f(z) will return to its initial value after such a closed circuit, whether or not there are finite branch points inside C. Thus, for the function

$$f(z) = \tan^{-1} z = \frac{i}{2} \log \frac{1 - iz}{1 + iz}$$

we obtain

$$f\left(\frac{1}{t}\right) = \frac{i}{2}\log\frac{t-i}{t+i}.$$

Hence, since t = 0 is not a branch point for f(1/t), it follows that z_{∞} is not a branch point for f(z). In the first method of cutting the plane for this function (Section 10.9), the cut can be thought of as joining the two branch points $z = \pm i$ along an infinite segment passing through z_{∞} . However, since z_{∞} is not a branch point, closed circuits surrounding z_{∞} need not be prohibited, and the finite cut joining the branch points in the second figure is equally satisfactory. That is, we need only prohibit closed circuits which enclose only one of the branch points. It must follow that a closed circuit surrounding both branch points will return $\tan^{-1} z$ to its initial value. As was stated earlier, this statement can be directly verified (see Problem 39).

It may happen that z_{∞} is a branch point on certain branches of a multivalued function but not on other branches, as for the function $\sin^{-1} z + i \log z$ (see Problem 51).

10.11. Significance of singularities. An important theorem, due to Liouville, states that a function which is analytic at all finite points and at z_{∞} must be constant. To establish this theorem, we consider any two points z_1 and z_2 in the complex plane and apply Cauchy's theorem (Section 10.4) to express $f(z_1)$ and $f(z_2)$ in the form

$$f(z_1) = \frac{1}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{\alpha - z_1}, \qquad f(z_2) = \frac{1}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{\alpha - z_2}, \tag{110}$$

where C is a circle of radius R, with center at the origin, including the points z_1 and z_2 . By subtraction we obtain also

$$f(z_2) - f(z_1) = \frac{z_2 - z_1}{2\pi i} \oint_C \frac{f(\alpha) \, d\alpha}{(\alpha - z_1)(\alpha - z_2)}.$$
 (111)

For points on C we have

$$\alpha = R e^{i\theta}, \quad |\alpha| = R. \tag{112}$$

Since f(z) is assumed to be analytic everywhere (including z_{∞}), it must be bounded everywhere. Denoting this bound by M and making use of the inequality (10) we obtain, for points on C,

$$\left|\frac{f(\alpha)}{(\alpha - z_1)(\alpha - z_2)}\right| \leq \frac{M}{(|\alpha| - |z_1|)(|\alpha| - |z_2|)} = \frac{M}{(R - |z_1|)(R - |z_2|)}$$

Hence, making use of (80), we obtain from (111)

$$|f(z_{2}) - f(z_{1})| \leq \frac{|z_{2} - z_{1}|}{2\pi} \cdot \frac{M}{(R - |z_{1}|)(R - |z_{2}|)} \cdot 2\pi R$$
$$\leq \frac{|z_{2} - z_{1}|}{R} \cdot \frac{M}{\left(1 - \frac{|z_{1}|}{R}\right)\left(1 - \frac{|z_{2}|}{R}\right)}.$$
 (113)

Since f(z) is analytic everywhere, we are permitted to let the radius R of C increase without limit. But since the right member then tends to zero, it follows that the left-hand member must be zero for any two points z_1 and z_2 . Hence f(z) is constant, as was to be shown.

This theorem has many important consequences. In illustration, we have seen that any polynomial of degree *m* has no singularities except a pole of order *m* at z_{∞} . We can now show that, conversely, *a function having no singularities other than a pole of order m at* z_{∞} *is necessarily a polynomial of degree m*. For if f(z) has a pole of order *m* at z_{∞} , then for large |z| it must be representable by a series of form (105). Hence f(z) must be expressible as the sum $P_m(z) + g(z)$ of a polynomial

$$P_m(z) = A_0 z^m + A_1 z^{m-1} + \dots + A_m$$
(114)

and a function whose expression, for large |z|, is of the form

$$g(z) = \frac{A_{m+1}}{z} + \frac{A_{m+2}}{z^2} + \cdots$$
 (115)

But since the function g(z) must then be analytic everywhere, and since it vanishes as $|z| \rightarrow \infty$, it follows from the preceding theorem that g(z) must be zero and hence f(z) must be a polynomial, as stated.

Further, we can show that if a function is analytic except for a finite number of poles it must be a rational function; that is, it must be the ratio of two polynomials. For suppose that f(z) has poles of order k_n at the N finite points $z = z_n$ (n = 1, 2, ..., N). Then, from the definition of a pole, it follows that the function

$$F(z) = (z - z_1)^{k_1} (z - z_2)^{k_2} \cdots (z - z_N)^{k_N} f(z)$$
(116)

is analytic at all finite points. The coefficient of f(z) in (116) is a polynomial of degree K, where

$$K=k_1+k_2+\cdots+k_N,$$

and hence it has a pole of order K at z_{∞} . But since we have supposed that f(z) is either analytic at z_{∞} or, at worst, has a pole at z_{∞} , it follows that F(z) has, at worst, a pole at z_{∞} . Now since F(z) is analytic at all finite points, the preceding theorem shows that it must be a polynomial. Thus f(z) must be the ratio of this polynomial to the polynomial which multiplies it in (116), as was to be shown.

These examples illustrate the fact that analytic functions are, to a certain extent, essentially characterized by their singularities. Those functions which are single-valued, and analytic at all finite points, are known as integral functions or as entire functions. Such a function is either analytic also at z_{∞} , in which case it must be a constant; or it may have a pole at z_{∞} , in which case it must be a polynomial; or, otherwise, it must have an essential singularity at z_{∞} , in which case it is known as an integral transcendental function. The functions e^z , sin z, and sinh z are examples of such functions. We have seen that if an analytic function is expanded in powers of z - a, the circle of convergence extends at least to the nearest singularity. It follows that the Taylor series representation of any integral function converges for all finite values of z. Conversely, if a power series converges for all finite values of z, it represents an integral function.

10.12. Residues. Suppose that the analytic function f(z) has a pole of order *m* at the point z = a. Then $(z - a)^m f(z)$ is analytic and hence can be expanded in the Taylor series

$$(z-a)^{m} f(z) = A_{0} + A_{1}(z-a) + \dots + A_{m-1}(z-a)^{m-1} + A_{m}(z-a)^{m} + \dots, \quad (117)$$

where

$$A_{k} = \frac{1}{k!} \left[\frac{d^{k}}{dz^{k}} \left\{ (z - a)^{m} f(z) \right\} \right]_{z=a},$$
(118)

in accordance with the results of Section 10.7. This series will converge within any circle about z = a which does not include another singularity. If $z \neq a$, Equation (117) can be rewritten in the form

$$f(z) = \frac{A_0}{(z-a)^m} + \frac{A_1}{(z-a)^{m-1}} + \dots + \frac{A_{m-1}}{z-a} + A_m + A_{m+1}(z-a) + \dots$$
(119)

Now let C_a be any closed contour surrounding z = a which lies inside the circle of convergence of (117) and which is such that f(z) is analytic inside

and on C_a , except at z = a. If we integrate (119) around this contour and review Equations (78a,b), we obtain merely

$$\oint_{C_a} f(z) \, dz = 2\pi i \, A_{m-1}$$

the only term contributing to the integration being $A_{m-1}/(z-a)$. The same relation then continues to hold when C_a is deformed into any *equivalent* contour, which need not lie inside the circle of convergence of (117).

We call the coefficient A_{m-1} the residue of f(z) at z = a and denote its value by Res (a), the function f(z) being understood. If a more explicit notation is desirable, we will denote the residue of f(z) at z = a by Res $\{f(z); a\}$. Thus, if f(z) has a pole of order m at z = a, then

$$\oint_{C_a} f(z) \, dz = 2\pi i \operatorname{Res} (a) \tag{120}$$

with

$$\operatorname{Res}(a) = \frac{1}{(m-1)!} \left[\frac{d^{m-1}}{dz^{m-1}} \left\{ (z-a)^m f(z) \right\} \right]_{z=a}, \quad (121)$$

where C_a is a closed contour enclosing z = a but excluding all other singularities of f(z).

We notice that Res (a) is the coefficient of 1/(z - a) in the Laurent expansion of f(z), in powers of z - a, which is valid *near* z = a. The value of the residue can be determined from this fact or can be determined directly from (121).

In the case of a simple pole (m = 1), Equation (121) gives

m = 1: Res (a) =
$$[(z - a)f(z)]_{z=a} = \lim_{z \to a} (z - a)f(z).$$
 (122)

In this case, if f(z) is expressed as the ratio

$$f(z) = \frac{N(z)}{D(z)},$$
(123)

where N(a) is finite, it follows that D(z) must vanish at z = a in such a way that D(z)/(z - a) approaches a finite limit as $z \rightarrow a$. By using L'Hospital's rule, we may then evaluate the limit indicated in (122) in the form

$$m = 1$$
: Res $(a) = \frac{N(a)}{D'(a)}$. (124)

When D(z) is a polynomial, z - a must be a factor, and hence (122) is readily evaluated by merely deleting the factor z - a in the denominator and setting z = a in the remaining ratio.

This procedure is applicable, however, only in the case when z = a is a simple pole. Otherwise, use should be made of Equation (121) unless the series (119) is easily obtained, in which case the residue is read directly from the series.

To illustrate the calculation of residues, we consider first the function

$$f(z) = \frac{1}{z^2 + 1} = \frac{1}{(z + i)(z - i)}.$$
 (125)

There are simple poles at $z = \pm i$ with residues

Res (i) =
$$\lim_{z \to i} (z - i) f(z) = \left(\frac{1}{z + i}\right)_{z = i} = \frac{1}{2i}$$

and

$$\operatorname{Res}\left(-i\right) = \left(\frac{1}{z-i}\right)_{z=-i} = -\frac{1}{2i},$$

from (122). If we use (124), we have, with N = 1 and $D = z^2 + 1$,

Res (i) =
$$\left(\frac{1}{2z}\right)_{z=i} = \frac{1}{2i}$$
, Res (-i) = $\left(\frac{1}{2z}\right)_{z=-i} = -\frac{1}{2i}$,

as before.

For the function

$$f(z) = \frac{1}{z^3 - z^2} = \frac{1}{z^2(z - 1)},$$
 (126)

there is a double pole at z = 0 and a simple pole at z = 1. At z = 1 we obtain, from (122),

Res (1) =
$$\left(\frac{1}{z^2}\right)_{z=1} = 1$$

At z = 0 we use (121) with a = 0, m = 2, and find

$$\operatorname{Res}\left(0\right) = \left[\frac{d}{dz}\left(\frac{1}{z-1}\right)\right]_{z=0} = -1.$$

For the function

$$f(z) = {\sin z - z \over z^6},$$
 (127)

a singularity in the finite part of the plane can occur only at z = 0. By expanding the numerator in powers of z, there follows

$$f(z) = \frac{\left(z - \frac{z^3}{3!} + \frac{z^5}{5!} + \frac{z^7}{7!} + \cdots\right) - z}{z^6}$$
(128)
$$f(z) = -\frac{1}{6z^3} + \frac{1}{120z} - \frac{z}{5040} + \cdots.$$

or

Thus f(z) has a pole of order *three* at z = 0. The residue is merely the coefficient of 1/z in (128),

Res (0) =
$$\frac{1}{120}$$
.

It is easily shown that if f(z) has a pole of order m at z = a, the value of the right-hand member of (121) is unchanged if m is replaced by any integer M such that $M \ge m$. That is, if the order of the pole is *overestimated* when (121) is used, the correct value of the residue will still result. This may be verified in the case just considered by using (121) to calculate Res (0) with m = 6, in



place of the correct value m = 3 which is not readily determined by inspection, and which would lead to a more involved calculation.

If f(z) has an isolated *essential* singularity at z = a, but is single-valued in the neighborhood of that point, then reference to (97) shows that (120) again applies, with

$$\operatorname{Res}(a) = a_{-1},$$
 (129)

Figure 10.12

where a_{-1} is the coefficient of $(z - a)^{-1}$ in the Laurent expansion of f(z) which is valid in the immediate neighborhood of z = a. It

has been seen that this statement is true also when z = a is a *pole*, so that, in fact, it applies when z = a is *any* isolated singularity of a single-valued function. However, the alternative formula (121) applies only when z = a is a *pole*.

In illustration, we note that the function

$$f(z) = z \, e^{1/z} \tag{130}$$

has an isolated essential singularity at z = 0, but has no other singularity at a finite point. From the expansion

$$f(z) = z\left(1 + \frac{1}{z} + \frac{1}{2! z^2} + \cdots\right) = z - 1 + \frac{1}{2! z} + \cdots \qquad (z \neq 0),$$

there follows

Res (0) =
$$\frac{1}{2}$$

Suppose now that C is the boundary of a finite region inside which f(z) is single-valued and has only isolated singularities, at a finite number of points $z = a_1, a_2, \ldots, a_n$. We enclose these points by small nonintersecting closed curves C_1, C_2, \ldots, C_n , each of which lies inside C and encloses only one singularity. Then, by introducing a crosscut from each curve C_k to C, a simply connected region \mathcal{R} is obtained inside which f(z) is analytic (Figure 10.12). Thus the line integral of f(z) around the complete boundary of this region vanishes. Noticing that the integrals along the crosscuts cancel, and that the

integrations taken around the small contours are in the negative sense, there follows

$$\oint_C f(z) \, dz - \left[\oint_{C_1} f(z) \, dz + \oint_{C_2} f(z) \, dz + \dots + \oint_{C_n} f(z) \, dz \right] = 0.$$
(131)

But since the integral taken in the positive sense around C_k is $2\pi i$ times the residue of f(z) at $z = a_k$, Equation (131) leads to the result

$$\oint_C f(z) dz = 2\pi i \sum_{k=1}^n \operatorname{Res} (a_k).$$
(132)

Thus, if f(z) is analytic inside and on a closed curve C, except at a finite number of interior isolated singularities, then $\oint_C f(z) dz$ is given by $2\pi i$ times the sum of the residues of f(z) at those points. This result is known as Cauchy's residue theorem.

10.13. Evaluation of real definite integrals. The residue theorem is useful in evaluating certain types of real definite integrals. In this section a few examples are presented.

Example 1. Any real integral of the form

$$I = \int_0^{2\pi} R(\sin\theta, \cos\theta) \, d\theta, \qquad (133)$$

where R is a rational function of $\sin \theta$ and $\cos \theta$ which is finite for all (real) values of θ , can be evaluated by residue theory as follows. If we make the substitution

$$z = e^{i\theta}, \qquad dz = ie^{i\theta} d\theta \tag{134}$$

there follows also

$$d\theta = \frac{dz}{iz}, \qquad \sin \theta = \frac{z^2 - 1}{2iz}, \qquad \cos \theta = \frac{z^2 + 1}{2z}. \tag{135}$$

Thus, $R(\sin \theta, \cos \theta) d\theta$ takes the form F(z) dz, where F(z) is a rational function of z. Since z describes a positive circuit around the unit circle C_1 in the complex plane as θ varies from 0 to 2π , the integral (133) takes the form

$$I = \oint_{C_1} F(z) dz = 2\pi i \sum_k \operatorname{Res} (a_k), \qquad (136)$$

where the points a_k are the poles of F(z) inside the unit circle.

For example, with (134) and (135) the integral

$$I_1 = \int_0^{2\pi} \frac{d\theta}{A + B\cos\theta} \qquad (A^2 > B^2, A > 0) \tag{137}$$

takes the form

$$I_1 = \oint_{C_1} F(z) dz, \qquad F(z) = \frac{2/i}{Bz^2 + 2Az + B}.$$

The poles of F(z) occur when

$$z = a_1 = -\frac{A}{B} + \frac{\sqrt{A^2 - B^2}}{B}$$

and

$$z=a_2=-\frac{A}{B}-\frac{\sqrt{A^2-B^2}}{B}$$

We see that, since $a_1a_2 = 1$, one pole (that at a_1 , since A is assumed to be positive) is inside the unit circle C_1 , whereas the second pole is outside C_1 (Figure 10.13). Using (124), we obtain

Res
$$(a_1) = \frac{2/i}{2Ba_1 + 2A} = \frac{1}{i\sqrt{A^2 - B^2}}$$

and hence there follows

$$I_1 = 2\pi i \operatorname{Res}(a_1) = \frac{2\pi}{\sqrt{A^2 - B^2}}.$$
 (138)

The restriction $A^2 > B^2$ is necessary in order that the integral (137) exist.



Figure 10.13

Figure 10.14

Example 2. We next indicate how contour integration may be used to evaluate an integral of the form

$$\int_{-\infty}^{\infty} f(x) \, dx, \tag{139}$$

where f(x) is a rational function, whose denominator is of degree at least two greater than the numerator, and which is finite for all (real) values of x. To illustrate the procedure, we consider the integral

$$I = \int_{-\infty}^{\infty} \frac{x^2}{1 + x^4} \, dx. \tag{140}$$

We first write

$$f(z) = \frac{z^2}{1+z^4}$$
(141)

and consider the result of integrating f(z) around the indicated contour (Figure 10.14), consisting of the segment of the real axis extending from -R to R and the semicircle C_R in the upper half-plane. For any value of R there follows

$$\int_{-R}^{R} \frac{x^2}{1+x^4} dx + \int_{C_R} f(z) dz = 2\pi i \sum_k \text{Res} (a_k), \qquad (142)$$

where the points a_k are the poles of f(z) inside the contour. As R increases without limit the first integral on the left approaches the required integral. Also, eventually all finite poles of f(z) in the upper half-plane are absorbed into the contour and hence contribute residues to the right-hand member.

We show next that the integral taken along C_R tends to zero as $R \to \infty$. On C_R we have |z| = R and hence also, using (10),

$$|f(z)| = \left|\frac{z^2}{1+z^4}\right| \le \frac{|z|^2}{|z|^4-1} = \frac{R^2}{R^4-1} \equiv M \quad (R>1).$$

The length of C_R is $L = \pi R$, and hence by Equation (80) there follows

$$\left| \int_{C_R} f(z) \, dz \right| \le ML = \frac{\pi R^3}{R^4 - 1} \qquad (R > 1). \tag{143}$$

Hence the integral along C_R tends to zero as the radius R increases indefinitely.

Thus, proceeding to the limit as $R \to \infty$, Equation (142) gives

$$I = \int_{-\infty}^{\infty} \frac{x^2}{1 + x^4} dx = 2\pi i \sum_{k} \text{Res}(a_k), \qquad (144)$$

where the points a_k are the finite poles of f(z) in the upper half-plane. The poles of f(z) in the upper half-plane are the two values of $(-1)^{1/4}$ which have positive imaginary parts,

$$a_1 = e^{\pi i/4}, \qquad a_2 = e^{3\pi i/4}.$$
 (145)

Also, making use of (124), we obtain

$$\operatorname{Res}\left(a_{k}\right) = \left(\frac{z^{2}}{4z^{3}}\right)_{z=a_{k}} = \frac{1}{4a_{k}}.$$

Thus (144) is evaluated in the form

$$\int_{-\infty}^{\infty} \frac{x^2}{1+x^4} dx = 2\pi i \left(\frac{1}{4} e^{-\pi i/4} + \frac{1}{4} e^{-3\pi i/4} \right) = \frac{\pi}{\sqrt{2}}.$$
 (146)

It should be noticed that the crucial step in this procedure consists of showing that the integral along C_R tends to zero as $R \to \infty$. Considering the more general case described in connection with (139), we see that if the denominator of f(x) is of degree at least two greater than the numerator, then along C_R the maximum value M of |f(z)| is at worst of the order of $\frac{1}{|z|^2} = \frac{1}{R^2}$. Since the length of C_R is $L = \pi R$, it follows that ML is at worst of the order of 1/R and hence, as in the example, the integral tends to zero as $R \to \infty$. Thus in such cases we have

$$\int_{-\infty}^{\infty} f(x) \, dx = 2\pi i \sum_{k} \operatorname{Res} (a_k), \qquad (147)$$

where the points a_k are the poles of f(z) in the upper half-plane. It is easily seen that if f(x) is a rational function, the conditions specified are necessary in order that $\int_{-\infty}^{\infty} f(x) dx$ exist. Thus (147) is true if f(x) is a rational function and if the integral exists.

We note that if f(x) is an even function, there follows

$$\int_0^\infty f(x)\,dx\,=\frac{1}{2}\int_{-\infty}^\infty f(x)\,dx.$$

Thus, for example, Equation (146) also gives

$$\int_0^\infty \frac{x^2}{1+x^4} \, dx = \frac{\pi}{2\sqrt{2}} \,. \tag{148}$$

Example 3. If we attempt to apply the method of the preceding example to the evaluation of an integral of the form

$$\int_{-\infty}^{\infty} f(x) \cos mx \, dx \quad \text{or} \quad \int_{-\infty}^{\infty} f(x) \sin mx \, dx, \tag{149}$$

where f(x) is a rational function of the type described, we encounter difficulty when we attempt to show that the integral of $f(z) \cos mz$ or $f(z) \sin mz$ along C_R tends to zero as $R \to \infty$. Thus, if we notice first that on C_R we have $z = Re^{i\theta}$ and hence

$$\begin{vmatrix} e^{imz} \end{vmatrix} = \begin{vmatrix} e^{imRe^{i\theta}} \end{vmatrix} = \begin{vmatrix} e^{imR(\cos\theta + i\sin\theta)} \end{vmatrix} = e^{-mR\sin\theta} \\ \begin{vmatrix} e^{-imz} \end{vmatrix} = \begin{vmatrix} e^{-imRe^{i\theta}} \end{vmatrix} = \begin{vmatrix} e^{-imR(\cos\theta + i\sin\theta)} \end{vmatrix} = e^{mR\sin\theta} \end{vmatrix},$$
 (150a,b)

it follows that on C_R the functions

$$\cos mz = \frac{e^{imz} + e^{-imz}}{2}, \qquad \sin mz = \frac{e^{imz} - e^{-imz}}{2i}$$
(151)

increase exponentially in magnitude as $R \to \infty$ because of the presence of the term e^{-imz} when m > 0 and because of the term e^{imz} when m < 0. Hence, the maximum value of the integrand is unbounded as $R \to \infty$, and the integral along C_R cannot be shown to tend to zero.

However, if we notice from (150a) that, when m > 0, $|e^{imz}|$ decreases exponentially on C_R as $R \to \infty$ in the upper half-plane (sin $\theta > 0$), we may avoid the difficulty noted by considering the integrals (149) as the real and imaginary parts of the integral

$$\int_{-\infty}^{\infty} e^{imx} f(x) \, dx \qquad (m \ge 0). \tag{152}$$

Then, if the maximum value of |f(z)| on C_R is, at worst, K/R^2 , we have, for points on C_R ,

$$\left|e^{imz}f(z)\right| = e^{-mR\sin\theta}\left|f(z)\right| \leq \left|f(z)\right| \leq \frac{K}{R^2} \qquad (m \geq 0).$$

Thus the integrand is bounded by $M = K/R^2$ and we have

$$\left| \int_{C_R} e^{imz} f(z) \, dz \right| \leq ML = \pi \, \frac{K}{R}$$

so that the integral along C_R again tends to zero as $R \to \infty$.

Hence we conclude that if f(x) is a rational function whose denominator is of degree at least two greater than the numerator, and which is finite for all (real) values of x, then

$$\int_{-\infty}^{\infty} e^{imx} f(x) dx = 2\pi i \sum \operatorname{Res} \left\{ e^{imx} f(z); a_k \right\} \quad (m \ge 0), \quad (153)$$

where the points a_k are the poles of f(z) in the upper half-plane. The integrals (149) are obtained as the real and imaginary parts of this result.

The theory to be developed in the following section shows that if m > 0, the degree of the denominator need be only one greater than that of the numerator.

In illustration, to evaluate the integral

$$\int_{-\infty}^{\infty} e^{imx} \frac{dx}{a^2 + x^2} \qquad (m \ge 0, a > 0),$$

where a and m are real, we notice that $f(x) = \frac{1}{a^2 + x^2}$ is a rational function of the required type. The only pole of $f(z) = \frac{1}{a^2 + z^2}$ in the upper half-plane is at z = ia.

Since the residue of $e^{imz} f(z)$ at z = ia is given by

$$\operatorname{Res}\left\{\frac{e^{imz}}{a^2+z^2}; ia\right\} = \left(\frac{e^{imz}}{z+ia}\right)_{z=ia} = \frac{1}{2ia} e^{-ma},$$

we obtain the result

$$\int_{-\infty}^{\infty} e^{imx} \frac{dx}{a^2 + x^2} = -\frac{\pi}{a} e^{-ma} \qquad (m \ge 0, a > 0). \tag{154}$$

By taking real and imaginary parts, we obtain

$$\int_{-\infty}^{\infty} \cos mx \frac{dx}{a^2 + x^2} = \frac{\pi}{a} e^{-ma}$$

$$\int_{-\infty}^{\infty} \sin mx \frac{dx}{a^2 + x^2} = 0$$

$$(m \ge 0, a > 0). \quad (155a,b)$$

It is seen from the form of the result that the restriction $m \ge 0$ can be removed in this case if we replace e^{-ma} by $e^{-|m|a}$.

10.14. Theorems on limiting contours. In many applications of contour integration it is necessary to evaluate the limit of the result of integrating a function of a complex variable along an arc of a circle as the radius of that circle either increases without limit or tends to zero. In this section we collect and establish certain general results of frequent application. First, however, it is convenient to introduce a useful definition.

If, along a circular arc C_r of radius r, we have $|f(z)| \leq M_r$, where M_r is a bound depending only on r and hence independent of *angular* position on C_r , and if $M_r \to 0$ as $r \to \infty$ (or $r \to 0$), then we will say that f(z) tends to zero uniformly on C_r as $r \to \infty$ (or $r \to 0$). Thus, for example, if C_r is a circular arc with center at the origin and $f(z) = z/(z^2 + 1)$, we have

$$|f(z)| = \frac{|z|}{|z^2 + 1|} \le \frac{|z|}{|z|^2 - 1} = \frac{r}{r^2 - 1} \qquad (r > 1)$$

on C_r , if use is made of the basic inequality (10). Hence, if we then take $M_r = r/(r^2 - 1)$, we conclude that here f(z) tends to zero uniformly on C_r as $r \to \infty$. Also, we may take $M_r = r/(1 - r^2)$ when r < 1 to show that the same is true when $r \to 0$.

It is not difficult to show, in particular, that any *rational function* (ratio of polynomials) whose denominator is of higher degree than the numerator tends uniformly to zero on any C_r as $r \to \infty$.

With this definition, the following theorems may be stated:

Theorem I. If, on a circular arc C_R with radius R and center at the origin, $zf(z) \rightarrow 0$ uniformly as $R \rightarrow \infty$, then

$$\lim_{R\to\infty}\int_{C_R}f(z)\,dz=0.$$

Theorem II. Suppose that, on a circular arc C_R with radius R and center at the origin, $f(z) \rightarrow 0$ uniformly as $R \rightarrow \infty$. Then:

1.
$$\lim_{R\to\infty}\int_{C_R}e^{imz}f(z)\,dz=0\qquad(m>0)$$

if C_R is in the first and/or second quadrants.*

2.
$$\lim_{R\to\infty}\int_{C_R}e^{-imz}f(z)\,dz=0\qquad(m>0)$$

if C_R is in the third and/or fourth quadrants.

3.
$$\lim_{R\to\infty}\int_{C_R}e^{mz}f(z)\,dz=0\qquad(m>0)$$

if C_R is in the second and/or third quadrants.

4.
$$\lim_{R \to \infty} \int_{C_R} e^{-mz} f(z) \, dz = 0 \qquad (m > 0)$$

if C_R is in the first and/or fourth quadrants.

* This result is known as Jordan's lemma.





Theorem III. If, on a circular arc C_{ρ} with radius ρ and center at z = a, $zf(z) \rightarrow 0$ uniformly as $\rho \rightarrow 0$, then

$$\lim_{\rho\to 0}\int_{C_{\rho}}f(z)\,dz=0.$$

Theorem IV. Suppose that f(z) has a simple pole at z = a, with residue Res (a). Then, if C_{ρ} is a circular arc with radius ρ and center at z = a, intercepting an angle α at z = a, there follows

$$\lim_{\rho\to 0}\int_{C_{\rho}}f(z)\,dz=\alpha i\,\operatorname{Res}\,(a),$$

where α is positive if the integration is carried out in the counterclockwise direction, and negative otherwise.

The proof of Theorem I follows from the fact that if $|zf(z)| \leq M_R$, then $|f(z)| \leq M_R/R$. Since the length of C_R is αR , where α is the subtended angle, Equation (80) gives

$$\left|\int_{C_R} f(z) \, dz\right| \leq \frac{M_R}{R} \cdot \alpha R = \alpha M_R \to 0 \qquad (R \to \infty).$$

The proof of Theorem II is somewhat more complicated. To prove part 1, we use the relation

$$\left|\int_{C_R} e^{imz}f(z)\,dz\right| \leq \int_{C_R} |e^{imz}|\,|f(z)|\,|dz|.$$

But on C_R we have $|dz| = Rd\theta$, $|f(z)| \le M_R$, and $|e^{imz}| = e^{-mR\sin\theta}$, according to (150a). Hence there follows

$$|I_R| \equiv \left| \int_{C_R} e^{imz} f(z) \, dz \right| \leq R M_R \int_{\theta_0}^{\theta_1} e^{-mR\sin\theta} \, d\theta,$$

where $0 \le \theta_0 < \theta_1 \le \pi$. Since the last integrand is positive, the right-hand member is not decreased if we take $\theta_0 = 0$ and $\theta_1 = \pi$. Hence we have

$$|I_R| \leq RM_R \int_0^{\pi} e^{-mR\sin\theta} d\theta = 2RM_R \int_0^{\pi/2} e^{-mR\sin\theta} d\theta.$$
(156)

This integral cannot be evaluated in terms of elementary functions of R. However, in the range $0 \le \theta \le \pi/2$ the truth of the relation

$$\sin\theta \geq \frac{2}{\pi}\theta$$

is easily realized by comparing the graphs of $y = \sin x$ and $y = 2x/\pi$. Thus we have also, from (156),

$$|I_R| \le 2RM_R \int_0^{\pi/2} e^{-2mR\theta/\pi} \, d\theta = \frac{\pi}{m} M_R (1 - e^{-mR}), \qquad (157)$$

and hence, if m > 0, I_R tends to zero with M_R as $R \to \infty$, as was to be shown.

The other three parts of Theorem II are established by completely analogous methods.

To prove Theorem III we notice that the integrand is not greater than M_{ρ}/ρ in absolute value and the length of the path is merely $\alpha \rho$, where α is the subtended angle. Hence the integral tends to zero with M_{ρ} .

To establish Theorem IV, we notice that if f(z) has a simple pole at z = a we can write

$$f(z) = \frac{\operatorname{Res}(a)}{z-a} + \varphi(z),$$

where $\varphi(z)$ is *analytic*, and hence bounded, in the neighborhood of z = a. Hence we have

$$\int_{C_{\rho}} f(z) dz = \int_{C_{\rho}} \frac{\operatorname{Res} (a)}{z-a} dz + \int_{C_{\rho}} \varphi(z) dz.$$

On C_{ρ} we can write $z = a + \rho e^{i\theta}$, where θ varies from an initial value θ_0 to $\theta_0 + \alpha$. Hence the first integral on the right becomes

$$\operatorname{Res}\left(a\right)\int_{\theta_{0}}^{\theta_{0}+\alpha}\frac{\rho i e^{i\theta} d\theta}{\rho e^{i\theta}}=i\operatorname{Res}\left(a\right)\int_{\theta_{0}}^{\theta_{0}+\alpha}d\theta=\alpha i\operatorname{Res}\left(a\right).$$

The second integral on the right tends to zero with ρ , in consequence of Theorem III, establishing the desired result.

Several applications of these theorems are presented in the following section.

10.15. Indented contours. In many cases the presence of a pole or branch point may lead to the necessity of *indenting* a contour by introducing an arc of a circle of small radius ρ , to avoid integration *through* a singularity. The desired result is then obtained by considering a limit as $\rho \rightarrow 0$.

In such cases we frequently encounter a new difficulty which can be explained best by considering what is meant by the Cauchy *principal value* of an improper real integral. To introduce this concept, we consider first the function y = 1/x. As is well known, the integral $\int_{-1}^{2} \frac{dx}{x}$ does not exist in the strict sense because of the strong infinity of the integrand at x = 0. We recall that the integral *would* exist, according to the conventional definition of an improper integral, only if the limit

$$\lim_{\delta_1,\delta_2\to 0}\left\{\int_{-1}^{-\delta_1}\frac{dx}{x}+\int_{\delta_2}^2\frac{dx}{x}\right\}$$

were to exist, and have a unique value, as δ_1 and δ_2 independently approach zero (Figure 10.15). But since this limit is of the form

$$\lim_{\delta_1,\delta_2\to 0} (\log \delta_1 + \log 2 - \log \delta_2) = \lim_{\delta_1,\delta_2\to 0} \left(\log 2 - \log \frac{\delta_2}{\delta_1}\right),$$

it clearly does not exist unless δ_1 and δ_2 tend to zero in such a way that $\frac{\delta_2}{\delta_1}$

tends toward a nonzero limit, in which case the value to be assigned to the integral depends upon this limit. However, if δ_1 and δ_2 are taken to be *equal*, so that the gap around the infinity in Figure

10.15 is symmetrical about x = 0, the limit is seen to be log 2. Incidentally, the same value would be obtained by formal substitution in the formula $\int_{a}^{b} \frac{dx}{x} = [\log |x|]_{a}^{b}$. This limit is defined to be the Cauchy

principal value of the improper integral, and we write

$$\mathsf{P}\int_{-1}^{2}\frac{dx}{x}=\log 2,$$

the symbol P denoting a principal value.

More generally, if f(x) is not finite at

Figure 10.15

a point x = A inside the interval of integration, we have the definition

$$\mathsf{P}\int_{a}^{b} f(x) \, dx = \lim_{\rho \to 0} \left\{ \int_{a}^{A-\rho} f(x) \, dx + \int_{A+\rho}^{b} f(x) \, dx \right\}, \tag{158}$$

if that limit exists. If the integral exists in the conventional sense, the true value necessarily agrees with the principal value so defined, and the symbol P may then be omitted.

The consideration of principal values of this kind is frequently necessary in the process of evaluating *proper* integrals, as is seen in the following examples. Further, principal values of improper integrals are not infrequently of physical significance in applications.* Clearly, some care should be exercised in dealing with them in such cases.

Example 4. In order to evaluate the integral

$$I = \int_{-\infty}^{\infty} \frac{\sin x}{x} dx = 2 \int_{0}^{\infty} \frac{\sin x}{x} dx, \qquad (159)$$

we consider the result of integrating the function $F(z) = e^{iz}/z$ around the closed contour of Figure 10.16. As before, we replace the sine by a complex exponential to obtain satisfactory behavior on C_R . In so doing, however, we obtain a function F(z) which then has a *pole* at the origin and hence must introduce the indentation

* For example, they arise frequently in the aerodynamic theory of airfoils. In technical work, the symbol P is often omitted before such integrals. Also, the alternative notation $\oint f(x) dx$, in which a C is superimposed on the integral sign, is sometimes used.



 C_{o} . Since F(z) is analytic inside the contour, Cauchy's residue theorem gives

$$\int_{-R}^{-\rho} \frac{e^{ix}}{x} dx + \int_{C\rho} \frac{e^{iz}}{z} dz + \int_{\rho}^{R} \frac{e^{ix}}{x} dx + \int_{C_{R}} \frac{e^{iz}}{z} dz = 0.$$
(160)

Since |1/z| = 1/R on C_R , the fourth integral tends to zero as $R \to \infty$, by Theorem II.1 of Section 10.14. Also, since F(z) has a simple pole at z = 0, with Res (0) = 1,

Theorem IV states that the second integral tends to $-\pi i \operatorname{as} \rho \to 0$, the negative sign corresponding to the negative sense of C_{ρ} . Thus, proceeding to the limit as $\rho \to 0$ and $R \to \infty$, we have

$$\lim_{\substack{\rho \to 0 \\ R \to \infty}} \left\{ \int_{-R}^{-\rho} \frac{e^{ix}}{x} dx + \int_{\rho}^{R} \frac{e^{ix}}{x} dx \right\} - \pi i = 0.$$
 (161)

Some care must be taken at this stage, since the integral $\int_{-\infty}^{\infty} e^{ix} \frac{dx}{x}$ does not exist in the strict sense because of the fact that the integrand

behaves like 1/x near x = 0. However, we may notice that the limit in (161) is in fact the Cauchy principal value of this *divergent integral*, and hence (161) takes the form

$$\mathsf{P}\int_{-\infty}^{\infty}\frac{e^{ix}}{x}dx = \pi i. \tag{162}$$

By taking imaginary parts of both sides, we obtain the desired result

$$\int_{-\infty}^{\infty} \frac{\sin x}{x} \, dx = \pi, \tag{163}$$

the symbol P now being omitted since this is a *convergent* integral, the integrand being *finite* at x = 0. However, the result of equating *real* parts of (162) should be written in the form

$$\mathsf{P}\int_{-\infty}^{\infty}\frac{\cos x}{x}\,dx\,=0.$$
 (164)

Example 5. We next consider the integral

$$I = \int_{-\infty}^{\infty} \frac{\cos x}{\pi^2 - 4x^2} \, dx,$$
 (165)

noticing that since the integrand is finite at $x = \pm \pi/2$, the question of principal values does not arise in the definition of *I*. To evaluate the integral, we integrate $F(z) = e^{iz}/(\pi^2 - 4z^2)$ around the contour of Figure 10.17, taking into account the poles of F(z) on the real axis. Again making use of the results of Section 10.14, we obtain the result

$$\mathsf{P}\int_{-\infty}^{\infty}\frac{e^{ix}}{\pi^2-4x^2}\,dx\,-\pi i\left[\operatorname{Res}\left(-\frac{\pi}{2}\right)\,+\,\operatorname{Res}\left(\frac{\pi}{2}\right)\right]\,=\,0,$$



where

$$\pi i \left[\operatorname{Res} \left(-\frac{\pi}{2} \right) + \operatorname{Res} \left(\frac{\pi}{2} \right) \right] = \pi i \left[\frac{e^{-\pi i/2}}{4\pi} - \frac{e^{\pi i/2}}{4\pi} \right] = \frac{1}{2},$$

and hence there follows

$$\mathsf{P} \int_{-\infty}^{\infty} \frac{e^{ix}}{\pi^2 - 4x^2} \, dx = \frac{1}{2} \,. \tag{166}$$

By equating real parts of this equation, we obtain the desired result

$$\int_{-\infty}^{\infty} \frac{\cos x}{\pi^2 - 4x^2} \, dx = \frac{1}{2} \,. \tag{167}$$



Example 6. In many cases a real integral can be transformed by contour integration into a more tractable form. To illustrate this procedure and also to consider the treatment of a branch point, we evaluate the integral

$$I = \int_0^\infty \frac{\cos x}{x^{1-m}} \, dx \qquad (0 < m < 1). \tag{168}$$

For this purpose we integrate $F(z) = e^{iz}/z^{1-m}$ around the contour of Figure 10.18. The indentation is necessary to avoid the branch point at the origin. We must, of course, choose that branch of the multivalued function z^{1-m} which is *real* on the positive real axis, that is, the *principal* branch. Cauchy's theorem gives

$$\int_{\rho}^{R} \frac{e^{ix}}{x^{1-m}} dx + \int_{C_{R}} \frac{e^{iz}}{z^{1-m}} dz + \int_{R}^{\rho} \frac{e^{-y}}{(iy)^{1-m}} (i \, dy) + \int_{C_{\rho}} \frac{e^{iz}}{z^{1-m}} dz = 0.$$

The integral along C_R vanishes as $R \to \infty$, by Theorem II.1, and that along C_{ρ} vanishes as $\rho \to 0$, by Theorem III. Hence, proceeding to the limits, we have

$$\int_0^\infty \frac{e^{ix}}{x^{1-m}} \, dx = i^m \int_0^\infty e^{-y} \, y^{m-1} \, dy \qquad (0 < m < 1), \tag{169}$$

where the principal value of i^m must be taken,

$$i^m = e^{im\pi/2} = \cos \frac{m\pi}{2} + i \sin \frac{m\pi}{2}$$
.

The integral on the right in (169) is recognized as that defining $\Gamma(m)$. Thus we have

$$\int_0^\infty \frac{\cos x}{x^{1-m}} dx = \Gamma(m) \cos \frac{m\pi}{2}$$

$$\int_0^\infty \frac{\sin x}{x^{1-m}} dx = \Gamma(m) \sin \frac{m\pi}{2}$$
(0 < m < 1). (170)

We may notice that the right-hand member of (169) is obtained *formally* if we replace x by a new (complex) variable iy in the left-hand member and do not concern ourselves with distinguishing between ∞ and ∞/i . The danger of using such a formal substitution without establishing its validity by contour integration (or otherwise) is illustrated by the consideration that by the same substitution in (168) we should deduce that $I = i^m \int_0^\infty y^{m-1} \cosh y \, dy$. However, this integral does not exist!

10.16. Conformal mapping. In order to represent geometrically the interpretation of a functional relationship w = f(z), it is conventional to employ two planes, in one of which (the z plane) the real and imaginary parts (x and y) of the independent variable z are plotted as the point (x, y), and in the second of which (the w plane) the real and imaginary parts (u and v) of the dependent variable w are plotted as the point (u, v). In this way a correspondence is set up, in general, between points, curves, and regions in one plane and their images in the other plane. We speak of such a correspondence as a mapping between the two planes.

If the function f(z) is single-valued, then corresponding to each point z where f(z) is defined there exists one and only one value of w = u + iv, and hence one and only one point in the w plane. Otherwise, to a given point z there will, in general, correspond two or more points in the w plane. Usually, if f(z) is multiple-valued, we introduce suitable cuts in the z plane in such a way that a given branch of f(z) is single-valued in the cut plane. Then u and v are single-valued functions of x and y. However, the reverse may not be true, in the sense that two or more values of z = x + iy may correspond to the same point w = u + iv. To investigate this possibility, we notice that if we write

$$w = f(z) = f_1(x, y) + i f_2(x, y) = u + iv,$$
(171)

then there follows

$$u = f_1(x,y), \quad v = f_2(x,y).$$
 (172)

Thus, to determine x and y in terms of u and v, we must solve Equations (172) for x and y. According to the results of Section 7.3, these equations can be

solved uniquely for x and y in some region about any point where (171) holds and where the Jacobian determinant

$$J = \frac{\partial(f_1, f_2)}{\partial(x, y)} = \begin{vmatrix} \frac{\partial f_1}{\partial x} & \frac{\partial f_1}{\partial y} \\ \frac{\partial f_2}{\partial x} & \frac{\partial f_2}{\partial y} \end{vmatrix} = \begin{vmatrix} \frac{\partial u}{\partial x} & \frac{\partial u}{\partial y} \\ \frac{\partial v}{\partial x} & \frac{\partial v}{\partial y} \end{vmatrix}$$

is not zero. If f(z) is analytic at $z_0 = x_0 + iy_0$ (and hence in a region including z_0), the Cauchy-Riemann equations are satisfied at z_0 , and hence there then follows, from (64) and (65),

$$[J]_{z_0} = \left[\frac{\partial u}{\partial x}\frac{\partial v}{\partial y} - \frac{\partial u}{\partial y}\frac{\partial v}{\partial x}\right]_{z_0} = \left[\left(\frac{\partial u}{\partial x}\right)^2 + \left(\frac{\partial v}{\partial x}\right)^2\right]_{z_0} = |f'(z_0)|^2.$$
(173)

Thus we conclude that if f(z) is analytic at a point z_0 and if $f'(z_0) \neq 0$, then there exists a region including z_0 in the z plane and a region including $w_0 = f(z_0)$ in the w plane such that the mapping w = f(z) gives a one-to-one correspondence between points in the two regions.

To illustrate such a mapping, we consider the mapping function

$$w = f(z) = z^{1/2}.$$
 (174)

Since f(z) is double-valued, it follows that to each point z = x + iy except the origin there correspond *two* points in the *w* plane. However, since the *inverse* function

$$z = F(w) = w^2 \tag{175}$$

is single-valued, we see that to each point w = u + iv there will correspond a *unique* point in the original z plane. To make the mapping one-to-one, we may cut the z plane along the negative real axis, as in Section 10.9, and consider the *principal branch* of $z^{1/2}$, for which $z^{1/2}$ is real and positive when z is real and positive,

$$w = f(z) = z^{1/2} = \sqrt{|z|} e^{i\theta_P/2} \qquad (-\pi < \theta_P \le \pi).$$
 (176)

To investigate the nature of the mapping, we may introduce polar coordinates in the two planes by writing

$$z = x + iy = re^{i\theta}, \qquad w = u + iv = \rho e^{i\varphi}.$$
 (177)

Then (174) becomes $\rho e^{i\varphi} = \sqrt{r} e^{i\theta/2}$, from which there follows

$$\rho = \sqrt{r}, \qquad \varphi = \frac{\theta}{2}.$$
(178)

We see that as θ varies from $-\pi$ to π , the angle φ varies from $-\pi/2$ to $\pi/2$. Hence the entire z plane is mapped, by the branch of $w = z^{1/2}$ which we have chosen, into only that half of the w plane for which $u \ge 0$. Equation (178) is convenient for plotting the image in one plane of a given *point* in the other plane, or for mapping curves expressed in polar coordinates.

If (175) is expressed in terms of real and imaginary parts, there follows

$$x + iy = (u + iv)^2 = (u^2 - v^2) + i(2uv)$$

and hence we have the relations

$$x = u^2 - v^2, \quad y = 2uv.$$
 (179)

Thus it is seen that any straight line $x = c_1$ is mapped into that portion of the hyperbola $u^2 - v^2 = c_1$ for which $u \ge 0$, whereas the straight line $y = c_2$ is mapped into one branch of the hyperbola $2uv = c_2$. The nature of corresponding regions is indicated in Figure 10.19.



Figure 10.19

If we imagine the z plane to consist of a sort of compressible material and think of the z plane as cut along the negative real axis and then pulled apart along the cut in such a way that the edges of the cut are each rotated through 90 degrees, so that all the material is compressed into the region to the right of the y axis, then straight lines drawn on the plane may be expected to become distorted into the shapes assumed by the corresponding curves sketched in the w plane.

We notice that the cut in the z plane is mapped into the v axis, and that neighboring regions in the z plane are mapped into neighboring regions except for those regions which are separated by the cut, across which transition is prohibited. Also, the lower boundary of regions 7 and 8 maps into the positive v axis, whereas the upper boundary of regions 9 and 10 maps into the negative v axis. Any continuous curve in the z plane not crossing the cut maps into a continuous curve in the w plane.

If neither the function f(z) nor its inverse is multiple-valued, it may happen that each plane maps as a whole into the other plane. If, as in the present case, f(z) is multiple-valued and its inverse is single-valued, a branch of f(z) may map the z plane into only a portion of the w plane. The mapping $w = z^2$ would be described by interchanging the z and w planes in the above sketches. In this case f(z) is single-valued but its inverse is multiple-valued, and a portion of the z plane maps into the entire w plane.

Since in the mapping (174) the derivative f'(z) is not zero for any finite value of z, the theorem proved above states that any point in the cut plane (that is, any point not on the cut) can be included in a region which is mapped into a region in the w plane under a one-to-one correspondence. It is seen, in fact, that *any* region not including points on the cut can be so mapped in this case.

If we indicate the inverse of w = f(z) by z = F(w), we notice that since

$$\frac{dF}{dw} = \frac{dz}{dw} = \frac{1}{\frac{dw}{dz}} = \frac{1}{f'(z)},$$
(180)

the condition $f'(z_0) \neq 0$ ensures that z = F(w) is analytic when $w = w_0$ if F(w) is single-valued.

If f(z) is analytic at $z = z_0$, and if $f'(z_0) \neq 0$, then to any curve C passing through z_0 in the z plane there corresponds a curve C' passing through the point $w_0 = f(z_0)$ in the w plane. If we consider a second point z_1 on C, and its image w_1 on C', and write

$$\frac{\Delta w}{\Delta z} = \frac{w_1 - w_0}{z_1 - z_0} = \frac{\Delta f}{\Delta z},$$
(181)

we see that this ratio is a complex number whose modulus is the ratio of the lengths of the chords (w_0w_1) and (z_0z_1) and whose argument is the angle between the directions of these chords. Thus in the limit as Δz and Δw approach zero, the limiting argument is the angle between the directions of C and C' at corresponding points and the limiting modulus represents a local magnification factor in the neighborhood of z_0 . But since for an analytic function the ratio (181) tends to $f'(z_0)$, independently of the direction of the chord (z_0z_1) , it follows that if f(z) is analytic at z_0 , and if $f'(z_0) \neq 0$, all curves passing through z_0 are mapped into new curves passing through w_0 , all of which are (approximately) rotated through the same angle, with respect to the mapped curves, and are magnified in the same ratio $|f'(z_0)|$ in the neighborhood of w_0 . Thus relative angle and shape are preserved in such a mapping where f(z) is analytic, and the mapping is said to be conformal. That is, a small closed figure will map into a similar closed figure with a certain rotation and magnification. In particular, two curves intersecting at a right angle are in general mapped into two curves intersecting at the same angle.

At points where f'(z) = 0 the magnification factor is zero and the angle of rotation is indeterminate. Such points are often known as *critical points* of the
mapping. It should be noticed that a critical point z_0 corresponds to a point w_0 at which the *inverse* function z = F(w) is *not analytic*.

If we write

$$w = f(z) = u + iv = f_1(x, y) + i f_2(x, y)$$
(182)

and notice that the straight lines u = constant and v = constant are orthogonal in the w plane, we see that the curves $u(x,y) = f_1(x,y) = \text{constant}$ and $v(x,y) = f_2(x,y) = \text{constant}$ in the z plane (of which these lines are the images) must be orthogonal at points where $f'(z) \neq 0$. That is, if u(x,y) and v(x,y) are the real and imaginary parts of an analytic function f(z), then the curves u = constant and v = constant are orthogonal at points of analyticity where $f'(z) \neq 0$. At singular points of f(z) the u and v curves may or may not exist and, in any case, need not be orthogonal. Similarly, at points for which $f'(z) \neq 0$ the inverse function z = F(w) is analytic (if single-valued), and the curve sets x = constant and y = constant in the z plane correspond to orthogonal sets of curves in the w plane. Thus, in general, small rectangles bounded by coordinate lines in either plane correspond to small "curvilinear rectangles" in the other plane except near points where f'(z) is zero or infinite.

10.17. Application to two-dimensional fluid flow. Since the real part of an analytic function of z satisfies Laplace's equation (Section 10.4), it represents the velocity potential of an ideal fluid flow in the xy plane. Thus, if we write

$$\Phi(z) = \varphi(x,y) + i \,\psi(x,y), \tag{183}$$

where Φ is an analytic function of z, and consider $\varphi(x,y)$ as a velocity potential corresponding to a flow velocity with components V_x and V_y in the x and y directions at a point, we must have

$$V_x = \frac{\partial \varphi}{\partial x}, \qquad V_y = \frac{\partial \varphi}{\partial y}.$$
 (184)

Since the curves $\psi = \text{constant}$ are orthogonal to the equipotential lines $\varphi = \text{constant}$, they must be identified with the streamlines of the flow. In fact, since the Cauchy-Riemann equations give

$$\frac{\partial \psi}{\partial x} = -\frac{\partial \varphi}{\partial y} = -V_y, \qquad \frac{\partial \psi}{\partial y} = \frac{\partial \varphi}{\partial x} = V_x \tag{185}$$

it follows [Section 6.19, Equation (175)] that $\psi(x,y)$ can be taken as the *stream* function of the flow. That is, the difference between the values of ψ at two points in the xy plane is numerically equal to the rate of mass flow of fluid with unit density across a curve joining these points.

The complex function $\Phi(z)$ is sometimes called the *complex potential*, since its real part is the velocity potential and its imaginary part is the stream function. We notice that, from Equation (64), we have also

$$\Phi'(z) = \frac{d\Phi}{dz} = \frac{\partial\Phi}{\partial x} = \frac{\partial\varphi}{\partial x} + i\frac{\partial\psi}{\partial x} = V_x - iV_y.$$
(186)

The function $\Phi'(z)$ is frequently called the *complex velocity*; its real part is V_x and its imaginary part $-V_y$. It may be seen that the conjugate function $\overline{\Phi'(z)} = V_x + iV_y$ can be considered as specifying the actual velocity vector.

Now suppose that a second complex variable w = u + iv is defined as an analytic function of the complex variable z = x + iy by the relationship

$$w = f(z) \tag{187}$$

and suppose that this mapping gives a one-to-one correspondence between points in a region \mathscr{R} in the xy plane and a region \mathscr{R}' in the uv plane. Then the equipotential lines and streamlines corresponding to a flow in the region \mathscr{R} will be mapped into a corresponding configuration in the region \mathscr{R}' . If we write the inverse of (187) in the form

$$z = F(w), \tag{188}$$

the original potential function $\Phi(z)$ is then expressible as a function of w in the form $\Phi[F(w)]$. Since we have

$$\frac{d\Phi}{dw} = \frac{d\Phi}{dz}\frac{dz}{dw} = \frac{\Phi'(z)}{f'(z)}$$
(189)

and since $\Phi(z)$ and f(z) are analytic, it follows that Φ is an analytic function of w except at points in the w plane which correspond to points in the z plane where f'(z) = 0. Hence, if we think of u and v as rectangular coordinates in the w plane, we see that the real and imaginary parts of $\Phi[F(w)]$ satisfy Laplace's equation in rectangular coordinates, and hence the new configuration in the w plane represents a new flow pattern of an ideal fluid.

The complex velocity is of the form

$$V_u - iV_v = \frac{d\Phi}{dw}$$
(190)

or, equivalently,

$$V_u - iV_v = \frac{d\Phi}{dz}\frac{dz}{dw} = \frac{\Phi'(z)}{f'(z)} = \frac{V_x - iV_y}{f'(z)}.$$
 (191)

From (191) it follows that

$$\sqrt{V_u^2 + V_v^2} = \frac{\sqrt{V_x^2 + V_y^2}}{|f'(z)|},$$
(192)

so that the absolute velocity at a point w_0 in the w plane is obtained by dividing the absolute velocity at the corresponding point z_0 in the z plane by $|f'(z_0)|$. Moreover, the actual (vector) velocities at the corresponding points will be equal if and only if $f'(z_0) = 1$. The streamlines $\psi(x,y) = \text{constant}$ and equipotential lines $\varphi(x,y) = \text{constant}$ map into corresponding curves in the w plane, with equations obtained by replacing x and y by their equivalent expressions in terms of u and v, or determined from the transformation (187) or (188).

We may verify directly that the velocity potential and stream functions satisfy Laplace's equation in rectangular uv coordinates as follows. Since uand v satisfy the Cauchy-Riemann equations, we have

$$\frac{\partial}{\partial x} = u_x \frac{\partial}{\partial u} + v_x \frac{\partial}{\partial v} = u_x \frac{\partial}{\partial u} - u_y \frac{\partial}{\partial v},$$

and hence

$$\begin{aligned} \frac{\partial \varphi}{\partial x} &= u_x \frac{\partial \varphi}{\partial u} - u_y \frac{\partial \varphi}{\partial v}, \\ \frac{\partial^2 \varphi}{\partial x^2} &= u_{xx} \frac{\partial \varphi}{\partial u} - u_{xy} \frac{\partial \varphi}{\partial v} + u_x \frac{\partial}{\partial x} \frac{\partial \varphi}{\partial u} - u_y \frac{\partial}{\partial x} \frac{\partial \varphi}{\partial v} \\ &= u_{xx} \frac{\partial \varphi}{\partial u} - u_{xy} \frac{\partial \varphi}{\partial v} + u_x^2 \frac{\partial^2 \varphi}{\partial u^2} - 2u_x u_y \frac{\partial^2 \varphi}{\partial u \partial v} + u_y^2 \frac{\partial^2 \varphi}{\partial v^2} \end{aligned}$$

Similarly, we obtain

$$\frac{\partial^2 \varphi}{\partial y^2} = u_{yy} \frac{\partial \varphi}{\partial u} + u_{xy} \frac{\partial \varphi}{\partial v} + u_y^2 \frac{\partial^2 \varphi}{\partial u^2} + 2u_x u_y \frac{\partial^2 \varphi}{\partial u \partial v} + u_x^2 \frac{\partial^2 \varphi}{\partial v^2}.$$

Adding the last two results, there follows

$$\nabla^2 \varphi = \frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} = (u_{xx} + u_{yy}) \frac{\partial \varphi}{\partial u} + (u_x^2 + u_y^2) \left(\frac{\partial^2 \varphi}{\partial u^2} + \frac{\partial^2 \varphi}{\partial v^2} \right).$$

The coefficient of $\frac{\partial \varphi}{\partial u}$ vanishes, since u(x,y) satisfies Laplace's equation in xy coordinates. Also, since $u_x - iu_y = u_x + iv_x = f'(z)$ there follows $u_x^2 + u_y^2 = |f'(z)|^2$, and we have

$$\frac{\partial^2 \varphi}{\partial x^2} + \frac{\partial^2 \varphi}{\partial y^2} = |f'(z)|^2 \left(\frac{\partial^2 \varphi}{\partial u^2} + \frac{\partial^2 \varphi}{\partial v^2} \right). \tag{193}$$

(See also Problem 87, Chapter 6.) Thus, unless f'(z) = 0, the vanishing of the left-hand member implies the vanishing of the parenthesis on the right. The same result is evidently obtained if φ is replaced by ψ .

10.18. Basic flows. In this section we investigate the flows corresponding to a few elementary complex potential functions.

Example 1. The potential

$$\Phi = V_0 z, \tag{194}$$

with V_0 real, corresponds to the complex velocity

$$V_x - iV_y = \frac{d\Phi}{dz} = V_0;$$
$$V_x = V_0, \qquad V_y = 0,$$

and hence represents uniform flow with velocity V_0 along the positive x direction (Figure 10.20). Similarly, if α is a real angle, the potential

$$\Phi = V_0 e^{-i\alpha} z \tag{195}$$

corresponds to flow with velocity components

$$V_x = V_0 \cos \alpha, \qquad V_y = V_0 \sin \alpha,$$

and hence represents uniform flow with velocity V_0 along a direction making an angle α with the positive x axis (Figure 10.21).



Figure 10.20

Figure 10.21

Example 2. For the potential

$$\Phi = k \log z = k(\log r + i\theta), \qquad (196)$$

with k real, the streamlines are

$$\psi = k\theta = \text{constant.}$$

Hence the flow is radial, from a *source* at the origin (Figure 10.22). The *strength* of the source, defined as the rate of mass flow of fluid with unit density across a closed curve enclosing the source, is given by the increase in ψ corresponding to a closed circuit about the origin, and hence has the value $2\pi k$. The velocity is radial, with magnitude k/r at distance r from the source. Similarly, $\Phi = k \log (z - a)$ represents flow from a source at z = a.

Example 3. For the potential

$$\Phi = -ik \log z = k(\theta - i \log r), \qquad (197)$$

with k real, the streamlines are

$$\psi = -k \log r = \text{constant}$$

Hence the flow is a *circulation* about the origin (Figure 10.23). The origin is known as a *vortex* of *strength* $2\pi k$. The velocity is circumferential, with magnitude k/r at distance r from the vortex. Similarly,

$$\Phi = -ik \log (z - a)$$

represents flow about a vortex at z = a.



Figure 10.22

Figure 10.23

Example 4. If a *sink* (negative source) of strength K is located at the origin and a *source* of strength K is located at a neighboring point z = a, the potential corresponding to the combination is given, according to Example 2, by

$$\Phi_a(z) = \frac{K}{2\pi} \left[-\log z + \log \left(z - a \right) \right] = \frac{K}{2\pi} \log \frac{z - a}{z}.$$

As the sink is brought into coincidence with the source, the two will cancel unless at the same time the strength K is continuously increased in inverse proportion to the separation |a|. If we write

$$K=\frac{m}{|a|}, \qquad a=|a|\,e^{i\alpha},$$

the potential can be put in the form

$$\Phi_{a}(z) = -\frac{m}{2\pi} \frac{\log z - \log (z - a)}{a} \frac{a}{|a|} = -\frac{m}{2\pi} e^{ix} \frac{\log z - \log (z - a)}{a}$$

Proceeding to the limit as $a \rightarrow 0$, we obtain

$$\Phi(z) = -\frac{me^{ix}}{2\pi} \frac{1}{z}.$$
 (198)

The limiting combination of source and sink is called a *doublet* of strength m. The angle α , measured from the positive real axis, denotes the direction along which the coincidence was effected, and is called the *orientation* of the doublet. If we write $z = r e^{i\theta}$, Equation (198) can be expressed in the form

$$\Phi(z) = -\frac{m}{2\pi}\frac{i}{r}e^{-i(\theta-\alpha)} = -\frac{m}{2\pi r}[\cos{(\theta-\alpha)} - i\sin{(\theta-\alpha)}].$$

Hence the streamlines are the circles

$$\psi = \frac{m}{2\pi r} \sin \left(\theta - \alpha \right) = \text{constant.}$$

The velocity at any point is of magnitude $m/2\pi r^2$, where r is distance from the doublet (Figure 10.24).

Example 5. By superimposing two or more of the basic flows described, various other flows of interest may be obtained. Thus, for example, if we superimpose upon



Figure 10.24



a doublet with orientation $\alpha = \pi$ and strength $m = 2\pi V_0 a^2$ a uniform flow with velocity V_0 in the positive x direction, we obtain a potential of the form

$$\Phi = V_0 \left(z + \frac{a^2}{z} \right). \tag{199}$$

In terms of real and imaginary parts we then have

$$\varphi + i\psi = \frac{V_0}{r}(a^2 + r^2)\cos\theta + i\frac{V_0}{r}(r^2 - a^2)\sin\theta.$$

The streamlines

$$\psi = \frac{V_0 \sin \theta}{r} (r^2 - a^2) = \text{constant}$$

are seen to include the circle r = a as well as the real axis $\sin \theta = 0$. In fact, it is easily seen that we have obtained a representation of the flow of an initially uniform stream around a circular (cylindrical) barrier (Figure 10.25). The velocity is obtained in the form

$$V_x = V_0 \left(1 - \frac{a^2}{r^2} \cos 2\theta \right), \qquad V_y = -V_0 \frac{a^2}{r^2} \sin 2\theta;$$
$$V = \sqrt{V_x^2 + V_y^2} = |\Phi'(z)| = \frac{V_0}{r^2} \sqrt{r^4 - 2a^2r^2\cos 2\theta} + a^4.$$

At the points $(r = a: \theta = 0, \pi)$ on the surface of the barrier, the velocity is zero. These points are known as *stagnation points*. By superimposing also a suitable circulation of type (197), we may obtain further flows in which the stream function ψ is again constant along the circular boundary and in which the flow tends toward uniformity with large distances from the boundary, but in which the stagnation points are displaced and the flow is no longer symmetrical with respect to the x axis.

We notice from the preceding examples that the presence of a *logarithmic* singularity in the complex potential function corresponds to the presence of a source (or sink) in the flow if the coefficient of the logarithmic term is real, and to the presence of a vortex if the coefficient is purely imaginary. Also, a simple pole at a point corresponds to a doublet at that point. If we recall (see Section 10.10) that a function $\Phi(z)$ behaves near z_{∞} just as $\Phi(1/t)$ behaves near t = 0, we see that the potential $V_0 z$, which represents uniform flow ("streaming") in the finite plane, has a simple pole at z_{∞} , and hence this flow includes a doublet at z_{∞} . Similarly, the potential $k \log z$ represents a flow with a source at the origin and a sink of equal magnitude at z_{∞} ; the potential *ik* log z indicates equal and opposite vortices at the origin and at z_{∞} ; and the potential c/zdefines a flow with a doublet at the origin but with no singularity at z_{∞} .

10.19. Other applications of conformal mapping. As has been seen in the preceding chapter, it is frequently necessary in various fields to determine a function of x and y which satisfies Laplace's equation and which takes on prescribed values at points of a given curve C in the xy plane. Suppose that we have somehow obtained an analytic function w = f(z) = u + iv which maps the curve C onto the real axis (v = 0) of the uv plane. The same mapping relation will transform the prescribed values of φ along C to corresponding values at points along the u axis in the w plane. If now we can find in the w plane a solution of Laplace's equation which takes on these values, this solution can be transformed back into xy coordinates to give the solution of the original problem. Thus, once the proper transformation is known, the problem is reduced from the determination of a function taking on prescribed values at points of a curve to the determination of one which takes on prescribed values at points of a curve to the determination of one which takes on prescribed values at points of a curve to the determination of one which takes on prescribed values at points of a curve to the determination of one which takes on prescribed values at points of fact, has been solved in Section 9.14 [Equation (220)].

In the case of ideal fluid flow, the problem just discussed can be considered as essentially reducing to the determination of a stream function $\psi(x,y)$ which satisfies Laplace's equation, takes on a *constant* value at points along a prescribed streamline, and behaves suitably at infinity. This is true since, once ψ is known, its conjugate $\varphi(x,y)$, the velocity potential, is determined by Equation (67a), Section 10.4, and the flow is completely determined.

Similar problems involving steady-state temperature distributions, electrostatic fields, and so on, are of frequent occurrence. In general, the direct determination of a suitable mapping function is not easily accomplished. However, if the curve C is *polygonal*, that is, made up of straight line segments, such a mapping can be obtained by methods to be presented in the following section.

In certain cases it may be more desirable to map the curve C instead onto the *unit circle* in the w plane and to solve the transformed problem by the methods of Section 9.4. As a single illustration we consider the mapping represented by the relation

$$z = \frac{1}{2} \left(w + \frac{1}{w} \right) \tag{200}$$

[compare Equation (199)]. Here the mapping function is conveniently expressed in a form solved for z. If we introduce polar coordinates (ρ, φ) in the w plane by writing

$$w = u + iv = \rho e^{i\varphi}$$

$$u = \rho \cos \varphi, \quad v = \rho \sin \varphi$$
, (201)

Equation (200) becomes

$$x + iy = \frac{1}{2} \left(\rho e^{i\varphi} + \frac{1}{\rho} e^{-i\varphi} \right),$$

so that we have

$$x = \frac{1}{2} \left(\rho + \frac{1}{\rho} \right) \cos \varphi, \qquad y = \frac{1}{2} \left(\rho - \frac{1}{\rho} \right) \sin \varphi. \tag{202}$$

By eliminating ρ and φ successively from these equations, there follows

$$\frac{\frac{x^{2}}{\left(\rho + \frac{1}{\rho}\right)^{2}} + \frac{y^{2}}{\left(\rho - \frac{1}{\rho}\right)^{2}} = \frac{1}{4}}{\frac{x^{2}}{\cos^{2}\varphi} - \frac{y^{2}}{\sin^{2}\varphi}} = 1$$
(203a,b)

Thus it follows that the circles $\rho = \text{constant}$ in the w plane correspond to the ellipses (203a), whereas the radial lines $\varphi = \text{constant}$ correspond to the hyperbolas (203b), as represented in Figure 10.26. In particular, we verify that the unit circle $\rho = 1$ in the w plane is flattened into a two-sided "slit" in the z plane between z = -1 and z = +1. The other circles in the w plane correspond to ellipses which decrease in flatness with increasing size. We notice that the entire cut z plane maps into the exterior of the unit circle.

Now suppose that it is required to determine a harmonic function T(x,y), say steady-state temperature, which takes on prescribed values along the upper and lower edges of the slit, say

$$T(x,0+) = f_1(x), \quad T(x,0-) = f_2(x) \quad (-1 < x < 1).$$

For points on the slit (202) gives

$$\rho = 1, \qquad x = \cos q,$$

where $0 < \varphi < \pi$ for the upper edge of the slit and $\pi < \varphi < 2\pi$ for the lower edge. Thus in the region $\rho \ge 1$ of the w plane we must determine a harmonic function $T(\rho,\varphi)$ which reduces when $\rho = 1$ to $f_1(\cos \varphi)$ when $0 < \varphi < \pi$ and to $f_2(\cos \varphi)$ when $\pi < \varphi < 2\pi$. This problem is solved in Section 9.4. The



Figure 10.26

value of T determined at any point (ρ, φ) is also the required value of T at the corresponding point (x, y), as determined by (202).

10.20. The Schwarz-Christoffel transformation. In this section we describe and illustrate a method of determining a mapping which sets up a one-to-one correspondence between points on a polygonal boundary in the z plane and points on the real axis (v = 0) in the w plane. It will be convenient to think first of the mapping as transforming the w plane into the z plane.

For this purpose, we consider the mapping z = F(w) for which

$$\frac{dz}{dw} = C(w - u_1)^{k_1}(w - u_2)^{k_2} \cdots (w - u_n)^{k_n}, \qquad (204)$$

where u_1, u_2, \ldots, u_n are any *n* points arranged in order along the real axis in the *w* plane (Figure 10.27), $u_1 < u_2 < \cdots < u_n$, and where the *k*'s are real constants and *C* is a real or complex constant. By taking the logarithm of both sides, Equation (204) can be written in the form

$$\log \frac{dz}{dw} = \log C + k_1 \log (w - u_1) + k_2 \log (w - u_2) + \cdots + k_n \log (w - u_n), \quad (205)$$

it being assumed that principal values of all logarithms are taken. We consider the mapping of the u axis onto the xy plane, remembering (see Section 10.9) that the magnitude of $\frac{dz}{dw}$ is the local magnification factor in the mapping from the w plane to the z plane, and that the angle of $\frac{dz}{dw}$, which can be expressed in the form

$$\preceq \frac{dz}{dw} = \operatorname{Im}\left(\log\frac{dz}{dw}\right),$$

gives the angle through which a small portion of the mapped curve in the w plane is rotated in the mapping. Thus we may obtain from (205) the relation

$$\preceq \frac{dz}{dw} = \preceq C + k_1 \preceq (w - u_1) + k_2 \preceq (w - u_2) + \cdots$$

$$+ k_n \preceq (w - u_n)$$
(206)

by taking imaginary parts of both sides.





Figure 10.28

Let the point at infinity in the w plane be mapped into the point z^* in the z plane. Then, if we consider the image of a point w = u moving from an infinite distance toward the right along the *negative real* axis in the w plane, we notice that so long as $w = u < u_1$ the numbers $w - u_1$, $w - u_2$, ..., $w - u_n$, are all *real and negative*. Hence their angles are all *constantly* equal to π in Equation (206), and so there follows

Thus the portion of the *u* axis to the left of the point u_1 is mapped into a *straight line segment*, making the angle defined by (207) with the real axis in the *z* plane, and extending from z^* to z_1 , the image of $w = u_1$ (Figure 10.28).

Now as the point w crosses the point u_1 on the real axis, the real number $w - u_1$ becomes positive, so that its angle abruptly changes from π to zero. Hence $\preceq \frac{dz}{dw}$ abruptly decreases by an amount $k_1\pi$ and then remains constant as w travels from u_1 toward u_2 . It follows that the image of the segment (u_1u_2) is a line segment (z_1z_2) in the z plane making an angle of $-k_1\pi$ with the segment (z^*z_1) .

Proceeding in this way, we see that each segment $(u_p u_{p+1})$ is mapped into a line segment $(z_p z_{p+1})$ in the z plane, making an angle of $-k_p \pi$ with the segment previously mapped. Thus, if the *interior* angle of the resultant polygonal contour at the point z_p is to have the magnitude α_p , we must set $\pi - \alpha_p = -k_p \pi$ or

$$k_p = \frac{\alpha_p}{\pi} - 1 \tag{208}$$

in Equation (204). After an integration, we then conclude that the mapping

$$z = C \int^{w} (w - u_1)^{k_1} (w - u_2)^{k_2} \cdots (w - u_n)^{k_n} dw + K, \qquad (209)$$

where C and K are arbitrary (complex) constants, will map the real axis (v = 0) of the w plane into a polygonal boundary in the z plane in such a way that the vertices z_1, z_2, \ldots, z_n , with interior angles $\alpha_1, \alpha_2, \ldots, \alpha_n$, are the images of the points u_1, u_2, \ldots, u_n .

We notice that for the final segment $(w = u > u_n)$ the numbers $w - u_i$ are all real and positive and hence possess zero angles, so that this segment is rotated through the angle

$$\preceq \frac{dz}{dw} = \preceq C \qquad (z = u > u_n). \tag{210}$$

For a *closed* polygon the sum of the interior angles is

$$\alpha_1 + \alpha_2 + \cdots + \alpha_n = (n-2)\pi,$$

and hence also

$$k_1 + k_2 + \cdots + k_n = \frac{(n-2)\pi}{\pi} - n = -2.$$

Thus, according to Equations (207) and (210), the two infinite segments of the line v = 0 are rotated through the angles $\angle C - 2\pi$ and $\angle C$, as is clearly necessary for a closed figure.

If the constant C is written in the form $C = ce^{i\beta}$, we see that it comprises an arbitrary magnification factor c and a rotation β and, in fact, that according to (210), the angle β is the angle through which the infinite segment to the right of $w = u_n$ is rotated when it is mapped (into a finite or infinite segment). If we write $K = u_0 + iv_0$, we see that the additive constant K represents an arbitrary translation of the polygon, without rotation or distortion, through the vector $u_0 + iv_0$.

It can be shown that the numbers u_1, u_2, \ldots, u_n , and the complex constants C and K can always be chosen so that *any* prescribed polygon in the z plane is made to correspond point by point to the real axis (v = 0) in the w plane and,

in fact, that the correspondence can be set up in infinitely many ways, in that three of the numbers u_1, u_2, \ldots, u_n , can be determined arbitrarily. Further, the mapping can be shown to establish a one-to-one correspondence between points in the *interior* of the polygon in the z plane and points in the *upper half* of the w plane.

Certain artifices may also be used in obtaining mappings involving the *exterior* of a polygon. As an example, suppose that we wish to investigate the

effect of the barrier of quadrilateral section, sketched in Figure 10.29, on an initially uniform flow of an ideal fluid. From the physical symmetry it is clear that one streamline will consist of the part of the x axis exterior to the polygonal section and, say, the *upper* half of the boundary. Thus we are led to consider a mapping which transforms this streamline into the u axis of the w plane, as is indicated in Figure 10.30. Here we map the region *above* the polygonal contour into the region $v \ge 0$, taking as the "interior angles" those indicated in the figure. The region may be considered as



Figure 10.29

an *infinite* polygon having the contour as its finite boundary. Once the flow is determined in the upper half of the z plane (by mapping a uniform flow in the upper half of the w plane), the flow is determined in the lower half-plane by symmetry.



A similar procedure clearly can be applied in all analogous cases in which there is an axis of symmetry parallel to the direction of flow.

Many illustrations of the Schwarz-Christoffel transformation (209) may be found in the literature. We here consider three such applications.

Example 1. As a simple example, we consider first the case in which the polygon is merely the interior of the sector of the z plane bounded by the positive x axis $(\theta = 0)$ and the line $\theta = \alpha$ (Figure 10.31). If we take the point u_1 , corresponding to the single finite vertex at z = 0, at the origin $(u_1 = 0)$, Equation (209) gives

$$z = C \int_0^w w^k dw,$$

where $k = (\alpha/\pi) - 1$ and C is an arbitrary constant. If, for convenience, we require that the point z = 1 map into the point w = 1, we find that C = k + 1, and there then follows

$$z = w^{\alpha/\pi}, \qquad w = z^{\pi/\alpha} \tag{211}$$

when $\alpha \neq 0$, provided that proper branches of the mapping function and/or its inverse are taken when necessary.

Suppose now that we require the solution T of Laplace's equation in the infinite sector $0 \le \theta \le \alpha$ for which T = 1 when 0 < x < 1 and T = 0 when x > 1, along



Figure 10.31

the edge $\theta = 0$, and for which T = 0 everywhere along the edge $\theta = \alpha$ (Figure 10.31). The mapping (211) transforms this problem to that of determining the solution of Laplace's equation in the half-plane $v \ge 0$ which reduces, when v = 0, to 1 when 0 < u < 1 and to 0 otherwise. But the solution of this problem is given by Equation (220) of Section 9.14 in the form*

$$T = \frac{1}{\pi} \int_0^1 \frac{v \, d\xi}{v^2 + (\xi - u)^2} = \frac{1}{\pi} \left[\tan^{-1} \left(\frac{1 - u}{v} \right) + \tan^{-1} \left(\frac{u}{v} \right) \right]$$
$$= \frac{1}{\pi} \cot^{-1} \left(\frac{u^2 + v^2 - u}{v} \right)$$
(212)

or, in polar coordinates (ρ, φ) ,

$$T = \frac{1}{\pi} \cot^{-1} \left(\frac{\rho - \cos \varphi}{\sin \varphi} \right).$$
(213)

From Equation (211) there follows

$$\rho = r^{\pi/\alpha}, \qquad \varphi = \frac{\pi \theta}{\alpha}$$

* It is useful to notice that the standard conventions $-\pi/2 \le \tan^{-1} x \le \pi/2$ and $0 \le \cot^{-1} x \le \pi$, when x is real, imply the relations

$$\tan^{-1}A + \tan^{-1}B = \begin{cases} -\cot^{-1}[(AB - 1)/(A + B)] & \text{if } A + B \leq 0\\ \tan^{-1}[(A + B)/(1 - AB)] & \text{if } AB \leq 1\\ \cot^{-1}[(1 - AB)/(A + B)] & \text{if } A + B \geq 0 \end{cases}$$

when A and B are real.

and hence the required solution is

$$T = \frac{1}{\pi} \cot^{-1} \left(\frac{r^{\pi/\alpha} - \cos \frac{\pi \theta}{\alpha}}{\sin \frac{\pi \theta}{\alpha}} \right).$$
(214)

The more general case in which T is prescribed *arbitrarily* along the two radial boundaries clearly can be treated in the same way.

Example 2. As a second example, we investigate the effect of a plane barrier of finite length 2a on an initially uniform flow of an ideal fluid perpendicular to it (Figure 10.32). This is the limiting case of the problem discussed in connection with the barrier of Figure 10.29, when the interior angles of the quadrilateral tend to 0 and to 2π . Thus we are led to consider the mapping of Figure 10.33. Here we wish to map the negative x axis into the segment $u < u_1$, the left-hand edge of the barrier into the segment $u_1 < u < u_2$, the right-hand edge



Figure 10.32

of the barrier into the segment $u_2 < u < u_3$, and the positive x axis into $u > u_3$. For purposes of symmetry, we take $u_1 = -b$, $u_2 = 0$, $u_3 = b$, where b is as yet



Figure 10.33

unspecified. With $\alpha_1 = \alpha_3 = \pi/2$, $\alpha_2 = 2\pi$, there follows, from (208), $k_1 = k_3 = -\frac{1}{2}$, $k_2 = 1$, and hence (209) gives

$$z = C \int^{w} \frac{w}{\sqrt{w^{2} - b^{2}}} dw + K = C \sqrt{w^{2} - b^{2}} + K, \qquad (215)$$

where K, C, and b are yet to be determined. The desired correspondence

 $w = -b \leftrightarrow z = 0, \quad w = 0 \leftrightarrow z = ia, \quad w = +b \leftrightarrow z = 0$ (216)

is then evidently obtained if we set

$$K=0, \qquad Cb=a, \qquad (217)$$

so that the mapping becomes

$$z = \frac{a}{b}\sqrt{w^2 - b^2}, \qquad w = \frac{b}{a}\sqrt{z^2 + a^2}.$$
 (218)

Finally we notice that $z_{\infty} \leftrightarrow w_{\infty}$, and hence if a uniform flow in the *w* plane is to be undistorted in the mapping at large distances from the barrier, we must have $\frac{dw}{dz} \rightarrow 1$ when $z \rightarrow z_{\infty}$, or

$$\left(\frac{dw}{dz}\right)_{|z|=\infty} = \left(\frac{b}{a}\frac{z}{\sqrt{z^2+a^2}}\right)_{z=\infty} = \frac{b}{a} = 1, \qquad b = a.$$
(219)

Thus the desired mapping becomes merely

$$z = \sqrt{w^2 - a^2}, \qquad w = \sqrt{z^2 + a^2} = f(z),$$
 (220)

where principal values of the roots are to be taken.

It is readily verified that this mapping gives the specified correspondence. In particular, points in the interior of a quadrant in one plane correspond to points inside the same quadrant in the other plane.

A uniform flow $\Phi(w) = V_0 w$ in the upper half of the w plane then maps into the desired flow in the z plane, with complex potential $\Phi[f(z)]$, or

$$\Phi = V_0 \sqrt{z^2 + a^2} = \varphi(x, y) + i \, \psi(x, y). \tag{221}$$

The streamlines and equipotential lines are obtained by equating to constants the real and imaginary parts φ and ψ .

The complex velocity is given by

$$V_x - iV_y = \frac{d\Phi}{dz} = \frac{V_0 z}{\sqrt{z^2 + a^2}} = \frac{V_0 (x + iy)}{\sqrt{(x^2 - y^2 + a^2)} + i(2xy)},$$
 (222)

where the sign of the radical is most conveniently determined by noticing that, for a given point z, the radical represents the *corresponding point* w. In particular, for points on the y axis there follows

$$V_x - iV_y = \frac{iV_0y}{\pm \sqrt{a^2 - y^2}},$$
 (223)

with a proper choice of sign. At points on the *right-hand* edge of the barrier the radical is real and *positive*, since the corresponding points are on the positive *u* axis, and there follows

$$V_x = 0, \qquad V_y = -\frac{V_0 y}{\sqrt{a^2 - y^2}},$$

whereas along the *left-hand* edge the sign is reversed. Along the remainder of the positive y axis (y > a), we have

$$\sqrt{a^2-y^2}=+i\sqrt{y^2-a^2},$$

and hence here

$$V_x = \frac{V_0 y}{\sqrt{y^2 - a^2}}, \qquad V_y = 0.$$

The flow velocity is seen to be infinite at the ends of the barrier.

Example 3. To illustrate additional features and techniques of the mapping, we consider the determination of the flow of an ideal fluid from a long channel with



Figure 10.34

parallel walls. The width of the channel is denoted by $2\pi L$, and the coordinate axes in the z plane are located as is indicated in Figure 10.34(a). We make use of the symmetry with respect to the x axis and consider only the flow in the upper halfplane. The polygonal boundary to be mapped can then be considered as the limit



Figure 10.35

of the boundary indicated in Figure 10.34(b) as the points A and C tend to infinity and the indicated angles tend to 2π and 0. We next attempt to accomplish the mapping of Figure 10.35, in which the point B is mapped (for convenience) into the point -1 in the w plane and the point (C_1C_2) at infinity is mapped into the origin. We should notice, however, for future reference, that the desired flow around the barrier is in the direction C_1BA , whereas that along the axis of symmetry is in the direction C_2D . This condition suggests that we shall need to map a flow from a source at the origin in the w plane onto the z plane, rather than a uniform flow as in the preceding example. With $u_1 = -1$, $u_2 = 0$ and $k_1 = 1$, $k_2 = -1$, Equation (209) gives

$$\frac{dz}{dw} = C \frac{w+1}{w}, \qquad (224)$$

and hence there follows

$$z = C(w + \log w) + K,$$
 (225)

where K is a constant of integration. Since $\log w$ is multivalued, we must select a particular branch of that function. It is convenient to deal with the principal branch, for which $\log z = \log |z| + i\theta_P$, where here $0 \le \theta_P \le \pi$. We next determine the constants C and K.

It is frequently convenient to obtain conditions determining unknown constants by essentially integrating the basic differential equation between corresponding limits. In illustration, we notice that transition from C_1 to C_2 in the z plane must correspond to transition from C'_1 to C'_2 in the w plane. The transition in the w plane must be considered as taking place along a semicircle of vanishing radius around the branch point at w = 0. If the radius is denoted by ρ , then near w = 0 we have, from (224),

$$dz = C \frac{w+1}{w} dw \sim C \frac{dw}{w} \qquad (as |w| = \rho \to 0),$$

and hence, on the small circle,

$$dz \sim C \frac{i\rho e^{i\varphi} \, d\varphi}{\rho e^{i\varphi}} = iC \, d\varphi. \tag{226}$$

Thus, in the limit as $\rho \rightarrow 0$, we have

$$z\Big|_{C_{1}}^{C_{2}} = iC \int_{\pi}^{0} d\varphi = -i\pi C, \qquad -i\pi L = -i\pi C,$$

$$C = L. \qquad (227)$$

and hence

In order that the point $B:(-L + i\pi L)$ correspond to B':(-1), we must have, from (225),

$$-L + i\pi L = L[-1 + \log(-1)] + K = -L + i\pi L + K,$$

and hence

$$K=0. (228)$$

(The position of the origin in the z plane was chosen in such a way that this convenient result would be obtained.)

Hence the desired mapping is given by

$$z = L(w + \log w). \tag{229}$$

To verify the correctness of this result, we may write $w = \rho e^{i\varphi}$. Then if $\varphi = \epsilon$, a simple calculation shows that $z \approx L[(\rho + \log \rho) + i(1 + \rho)\epsilon]$ except for higher powers of ϵ ; and if $\varphi = \pi - \epsilon$, there follows similarly $z \approx L\{-[\rho - \log \rho] + i[\pi + (\rho - 1)\epsilon]\}$. From these results it is easy to verify the correspondence between the upper boundary of the *u* axis (as $\epsilon \to 0+$) and the path ABC_1C_2D .

Next it is necessary to determine an appropriate flow in the w plane, for which $C'_1B'A'$ and C'_2D' are streamlines. Also it is necessary that the complex potential function Φ have the property that the velocity $\frac{d\Phi}{dz}$ shall reduce to a uniform velocity V_0 in the x direction at the point (C_1C_2) in the z plane. These, however, are the only requirements to be satisfied. Although Φ can be determined from these conditions by purely direct methods, it is suggested that in the present case a flow from a source at the origin in the w plane will have the desired properties. Hence we write

$$\Phi = k \log n$$

and attempt to determine k in such a way that the velocity $\frac{d\Phi}{dz}$ in the z plane will tend to V_0 as $z \to (C_1C_2)$ and, correspondingly, $w \to 0$. Making use of (224) and (227), we obtain the general result

$$\frac{d\Phi}{dz}=\frac{dw}{dz}\frac{d\Phi}{dw}=\frac{1}{L}\frac{w}{w+1}\frac{d\Phi}{dw},$$

and hence, with $\Phi = k \log w$, there follows

$$\frac{d\Phi}{dz} = \frac{k}{L}\frac{1}{w+1}$$

Setting w = 0, we obtain the condition

$$k = LV_c$$

so that the desired potential function takes the form

$$\Phi = LV_0 \log w. \tag{230}$$

The corresponding potential function in the z plane is then to be obtained by eliminating w between Equations (230) and (229), and hence is the solution of the equation

$$z = L\left(e^{\Phi/LV_0} + \frac{\Phi}{LV_0}\right).$$
(231)

Whereas Φ cannot be expressed as a function of z in explicit closed form, we may write

$$\Phi = \varphi + i \, \psi, \tag{232}$$

where, in consequence of (230) and the limitation of permissible values of the logarithm,

$$0 \leq \varphi < \infty, \qquad 0 \leq \psi \leq \pi L V_0, \tag{233}$$

and Equation (231) thus is equivalent to the two real equations

$$\left. \begin{array}{c} \frac{x}{L} = e^{\alpha} \cos \beta + \alpha \\ \frac{y}{L} = e^{\alpha} \sin \beta + \beta \end{array} \right\}, \qquad (234)$$

with the abbreviations

$$\alpha = \frac{\varphi}{LV_0}, \quad \beta = \frac{\psi}{LV_0} \qquad (0 \le \alpha \le \infty, 0 \le \beta \le \pi). \tag{235}$$

These parametric equations are convenient in plotting the streamlines β = constant and the equipotential lines α = constant. It can be verified that the streamlines for which $0 \leq \beta \leq \pi/2$ extend indefinitely in the x direction, whereas on the remaining streamlines a maximum value of x exists, so that these streamlines turn back around the channel walls.

In each of the preceding examples, the region enclosed by the polygonal boundary could instead be considered as a homogeneous conducting sheet carrying an electric current, or as a plane section of an *electrostatic* field between long plane or cylindrical conductors. The *stream function* ψ and the *streamlines* ψ = constant in the problem of *fluid flow* then would correspond to the *potential function* and to the *equipotential lines* in the problem of *electrostatics*, whereas the *potential function* φ and the *equipotential lines* in the flow problem would correspond to the *stream function* and to the *lines of force* in the electrostatic problem.

In two-dimensional electrostatics, the *negative* gradient of the potential function is the electric field intensity E, whose *tangential* component must vanish on the surface of a perfect conductor. The *normal* component of E at such a surface is proportional to the surface charge density σ . The stream function then has the property that the difference between its values at two points on the conducting boundary is proportional to the total charge q, per unit height of the conductor, on the portion of the conductor joining those points. (See Problems 121–126 and Reference 10.)

REFERENCES

- 1. Churchill, R. V., Complex Variables and Applications, 2nd ed., McGraw-Hill Book Company, Inc., New York, 1960.
- 2. Curtiss, D. R., Analytic Functions of a Complex Variable, Second Carus Mathematical Monograph, Open Court Publishing Company, La Salle, Ill., 1926.
- 3. Franklin, P., Functions of Complex Variables, Prentice-Hall, Inc., Englewood Cliffs, N. J., 1958.
- 4. Kaplan, W., A First Course in Functions of a Complex Variable, Addison-Wesley Publishing Co., Inc., Reading, Mass., 1953.
- 5. Knopp, K., Theory of Functions, Dover Publications, Inc., New York, 2 pts., 1945, 1947.
- 6. Kober, H., Dictionary of Conformal Representations, Dover Publications, Inc., New York, 1957.

- 7. Nehari, Z., Conformal Mapping, McGraw-Hill Book Company, Inc., New York, 1952.
- 8. Rothe, R., F. Ollendorf, and K. Pohlhausen, *Theory of Functions as Applied to Engineering Problems*, Dover Publications, Inc., New York, 1961.
- 9. Titchmarsh, E. C., *The Theory of Functions*, Oxford University Press, New York, 1950.
- 10. Walker, M., Conjugate Functions for Engineers, Oxford University Press, New York, 1933.
- 11. Whittaker, E. J., and G. N. Watson, *Modern Analysis*, Cambridge University Press, New York, 1958.

PROBLEMS

Section 10.1

1. Show that, if a complex number a + ib is written as a number pair (a,b), then for two such complex numbers the laws of combination take the form

$$(a_1,b_1) + (a_2,b_2) = (a_1 + a_2, b_1 + b_2),$$

$$(a_1,b_1)(a_2,b_2) = (a_1a_2 - b_1b_2, a_1b_2 + a_2b_1),$$

$$\frac{(a_2,b_2)}{(a_1,b_1)} = \left(\frac{a_1a_2 + b_1b_2}{a_1^2 + b_1^2}, \frac{a_1b_2 - a_2b_1}{a_1^2 + b_1^2}\right),$$

2. (a) Show that, if the real and imaginary parts of $\alpha \equiv a + ib$ are the components of the vector $\mathbf{v} = a\mathbf{i} + b\mathbf{j}$, then the real and imaginary parts of $\alpha_1 \pm \alpha_2$ are the components of the vector $\mathbf{v}_1 \pm \mathbf{v}_2$, but that no such statement applies to multiplication (with reference to either the dot or the cross product of vectors) or to division.

(b) Show, however, that $\alpha \bar{\alpha} \equiv |\alpha|^2$ is equal to the dot product $\mathbf{v} \cdot \mathbf{v} \equiv |\mathbf{v}|^2$, and that $\alpha_1 \bar{\alpha}_2 + \bar{\alpha}_1 \alpha_2 = 2\mathbf{v}_1 \cdot \mathbf{v}_2$. (Notice that these quantities are real scalar numbers).

3. Establish the following results:

(a) Re $(z_1 + z_2)$ = Re (z_1) + Re (z_2) , but Re $(z_1z_2) \neq$ Re (z_1) Re (z_2) in general;

(b) $\text{Im}(z_1 + z_2) = \text{Im}(z_1) + \text{Im}(z_2)$, but $\text{Im}(z_1z_2) \neq \text{Im}(z_1) \text{Im}(z_2)$ in general;

(c) $|z_1z_2| = |z_1| |z_2|$, but $|z_1 + z_2| \neq |z_1| + |z_2|$ in general; (d) $\overline{z_1 + z_2} = \overline{z_1} + \overline{z_2}$ and $\overline{z_1z_2} = \overline{z_1} \overline{z_2}$.

- 4. Establish the following results:
 - (a) $z + \bar{z} = 2 \operatorname{Re}(z)$, (b) $z - \bar{z} = 2i \operatorname{Im}(z)$, (c) $z_1 \overline{z_2} + \overline{z_1} z_2 = 2 \operatorname{Re}(z_1 \overline{z_2}) = 2 \operatorname{Re}(\overline{z_1} z_2)$, (d) $\operatorname{Re}(z) \leq |z|$, (e) $\operatorname{Im}(z) \leq |z|$, (f) $|z_1 \overline{z_2} + \overline{z_1} z_2| \leq 2 |z_1 z_2|$, (g) $(|z_1| - |z_2|)^2 \leq |z_1 + z_2|^2 \leq (|z_1| + |z_2|)^2$.

Section 10.2

5. Express the following quantities in the form a + ib, where a and b are real:

(a)
$$(1 + i)^3$$
, (b) $\frac{1 + i}{1 - i}$, (c) $e^{\pi i/2}$,
(d) $e^{2 + \pi i/4}$, (e) $\sin\left(\frac{\pi}{4} + 2i\right)$, (f) $\cosh\left(2 + \frac{\pi i}{4}\right)$.

6. Prove that the functions $\sin z$ and $\cos z$ are periodic, with real period 2π , whereas e^z , $\sinh z$ and $\cosh z$ are periodic, with pure imaginary period $2\pi i$. What are the periods of the other circular and hyperbolic functions?

7. Use the series definitions to obtain expressions for the derivative of e^z , sin z, cos z, sinh z, and cosh z.

8. Prove that e^z possesses no zeros, that the zeros of sin z and cos z all lie on the real axis, and that those of sinh z and cosh z all lie on the imaginary axis.

9. If
$$f(z) = e^{iz}$$
, show that $f(z) = e^{-iz}$, $f(\overline{z}) = e^{i\overline{z}}$, and $f(\overline{z}) = \overline{f(z)} = e^{-i\overline{z}}$

Section 10.3

10. Show that the *n*th roots of unity are of the form ω_n^k (k = 0, 1, ..., n - 1), where $\omega_n = \cos(2\pi/n) + i \sin(2\pi/n)$.

11. Determine all possible values of the following quantities in the form a + ib, and in each case give also the principal value:

(a) $\log (1 + i)$, (b) $(i)^{3/4}$, (c) $(1 + i)^{1/2}$.

12. Express the roots of the equation $z^4 + 2z^2 + 2 = 0$ in the form a + ib.

13. Express the function z^{π} in the form given by Equation (47) and also find the principal value of this function when $z = (1 + i)/\sqrt{2}$, in the form a + ib.

14. Derive Equations (55c) and (56a).

15. Determine all possible values of the quantities

(a) $\sin^{-1} 2$, (b) $\tan^{-1} (2i)$.

16. (a) Verify that, if a is a positive real constant,

$$\operatorname{coth}^{-1}\frac{z}{a} = \frac{1}{2}\log\frac{z+a}{z-a} = \frac{1}{2}\left[\log\left|\frac{z+a}{z-a}\right| + i\arg\left(\frac{z+a}{z-a}\right)\right].$$

(b) Verify that

$$\frac{z+a}{z-a} = \frac{(z+a)(\bar{z}-a)}{(z-a)(\bar{z}-a)} = \frac{|z|^2 - a^2 - 2ai \operatorname{Im}(z)}{|z|^2 + a^2 - 2a \operatorname{Re}(z)}.$$

(c) Hence deduce that

$$\coth^{-1}\frac{z}{a} = \frac{1}{2}\log\frac{\sqrt{(x^2+y^2-a^2)^2+4a^2y^2}}{(x-a)^2+y^2} + \frac{i}{2}\tan^{-1}\left(\frac{2ay}{a^2-x^2-y^2}\right),$$

when a is real and positive, where an arbitrary additive integral multiple of π is implied in the imaginary part.

17. (a) If (ρ_1, φ_1) are polar coordinates relative to the point (0, a) and (ρ_2, φ_2) are polar coordinates relative to the point (0, -a), show that

$$z = a + \rho_1 e^{i\varphi_1} = -a + \rho_2 e^{i\varphi_2}.$$

(b) Use the result of Problem 16(a) to show that

$$\operatorname{coth}^{-1}\frac{z}{a} = \frac{1}{2}\log\frac{\rho_1}{\rho_2} + \frac{i}{2}(\varphi_2 - \varphi_1),$$

where φ_1 and φ_2 are each defined only within an arbitrary additive integral multiple of 2π .

18. (a) Suppose that the principal values of φ_1 and φ_2 are defined such that $0 \leq \varphi_{1P} < 2\pi$ and $0 \leq \varphi_{2P} < 2\pi$ in Problem 17. Verify that, as the segment of the real axis between -a and a is crossed from above, the angle $\varphi_2 - \varphi_1$ changes abruptly by 2π , whereas no such jump occurs at a crossing outside this segment, so that transition from one branch to another then can and must take place only across the finite segment joining the "branch points" $z = \pm a$.

(b) Suppose that the definitions $0 \le \varphi_{1P} < 2\pi$ and $-\pi < \varphi_{2P} \le +\pi$ are adopted. Verify that $\varphi_2 - \varphi_1$ then is continuous across the finite segment joining the branch points, but that it changes abruptly by 2π as the real axis is crossed at any point external to this segment.

Section 10.4

19. (a) If $3x^2y - y^3$ is the real part of an analytic function of z, determine the imaginary part.

(b) Prove that xy^2 cannot be the real part of an analytic function of z.

(c) Determine whether $2xy + i(x^2 - y^2)$ is an analytic function of z.

20. Let s represent distance in the counterclockwise direction around a closed curve C in the xy plane.

(a) If, at any point P on C, t represents the unit tangent vector in this direction, and **n** represents the unit outward normal vector, show that

 $\mathbf{t} = \mathbf{i} \cos \varphi + \mathbf{j} \sin \varphi$ and $\mathbf{n} = \mathbf{i} \sin \varphi - \mathbf{j} \cos \varphi$, where φ is the slope angle.

(b) If u(x,y) and v(x,y) are the real and imaginary parts of an analytic function of z in a region \mathcal{R} including C, show that

$$\frac{\partial u}{\partial s} = -\frac{\partial v}{\partial n}, \quad \frac{\partial u}{\partial n} = \frac{\partial v}{\partial s}$$

at any point of C. [Recall that $\frac{\partial}{\partial s} = \mathbf{t} \cdot \nabla$ and $\frac{\partial}{\partial n} = \mathbf{n} \cdot \nabla$, and use (65a,b).]

21. By applying the result of Problem 20 to a circle r = constant, obtain the Cauchy-Riemann equations in polar coordinates, in the form

$$\frac{\partial u}{\partial r} = \frac{1}{r} \frac{\partial v}{\partial \theta}, \qquad \frac{1}{r} \frac{\partial u}{\partial \theta} = -\frac{\partial v}{\partial r}.$$

22. Show that the real and imaginary parts of any twice-differentiable function of the form $f(\bar{z})$ satisfy Laplace's equation, but that such a function is nowhere an analytic function of z unless it is a constant. [Compare the values of $\frac{\partial f}{\partial x}$ and $\frac{\partial f}{\partial (iv)}$.]

23. Show that f(|z|) is nowhere an analytic function of z unless it is a constant. (Consider the derivative of f in the θ direction.)

Section 10.5

24. (a) Use the definition (68) to calculate directly the integral $\oint_C z \, dz$, where C is the unit circle $x = \cos t$, $y = \sin t$.

(b) Use the definition (72) to calculate directly the integral $\oint_C \log z \, dz$, where C is the unit circle r = 1, taking the principal value of the logarithm.

25. (a) Show that the value of the integral

$$\int_{-1}^1 \frac{z+1}{z^2} \, dz$$

is $-2 - \pi i$ if the path is the *upper* half of the circle r = 1. [Write $z = e^{i\theta}$, where θ varies from π to 0, or from $(2k + 1)\pi$ to $2k\pi$, where k is any integer.]

(b) Show that the value is $-2 + \pi i$ if the path is the *lower* half of the same circle.

26. Evaluate the integral

$$\oint_C \frac{z+1}{z^2} \, dz,$$

where C is the circle r = 1, first by using the results of Problem 25, and second by using Equations (75) and (77).

27. (a) Prove that the integral

$$\int_{-1}^{2} \frac{dz}{z^2}$$

is independent of the path, so long as that path does not pass through the origin. By integrating along any convenient path (say, around a semicircle and thence along the real axis) show that the value of the integral is $-\frac{3}{2}$.

(b) Show that the real integral

$$\int_{-1}^{2} \frac{dx}{x^2}$$

does not exist, but that the value given by formal substitution of limits in the indefinite integral agrees with that obtained in part (a). (Notice that, in spite of this fact, the integrand is never negative!)

28. Let C represent a *semicircle* of radius R, with center at the origin, where R > 1, and consider the functions

$$f_1(z) = z^2 - 1, \qquad f_2(z) = \frac{1}{z^2 + 1}.$$

(a) Use (10) to show that on C there follows

$$R^2 - 1 \leq |f_1(z)| \leq R^2 + 1, \qquad \frac{1}{R^2 + 1} \leq |f_2(z)| \leq \frac{1}{R^2 - 1}$$

(b) Deduce from (80) that

$$\left|\int_C f_1(z) dz\right| \leq \pi R(R^2 + 1), \qquad \left|\int_C f_2(z) dz\right| \leq \frac{\pi R}{R^2 - 1}.$$

(c) Show also that

$$\left| \int_C f_1(z) f_2(z) \, dz \right| \leq \pi R \, \frac{R^2 + 1}{R^2 - 1} \, .$$

Section 10.6

29. (a) Use the results of Equations (78a,b) to verify Cauchy's integral formula (85) when $f(z) = z^2$. [Express $f(\alpha)$ in the form $(\alpha - z)^2 + 2z(\alpha - z) + z^2$.]

(b) Verify also the derivative formula (88) in this case.

30. If $F(z) = (z + 6)/(z^2 - 4)$, show that $\oint_C F(z) dz$ is 0 if C is the circle $x^2 + y^2 = 1$, is $4\pi i$ if C is the circle $(x - 2)^2 + y^2 = 1$, and is $-2\pi i$ if C is the circle $(x + 2)^2 + y^2 = 1$. [Use (84) in the second and third cases, with the functions f(z) = (z + 6)/(z + 2) and f(z) = (z + 6)/(z - 2), respectively.]

31. Let z_0 denote a point in a region \mathscr{R} where f(z) is analytic, and let C denote any circle, with center at z_0 , which lies inside \mathscr{R} . By writing $\alpha = z_0 + a e^{i\varphi}$ in Cauchy's integral formula (85), show that

$$f(z_0) = \frac{1}{2\pi} \int_0^{2\pi} f(z_0 + a e^{i\varphi}) d\varphi,$$

and deduce that the value of an analytic function at any point z_0 is the average of its values on any circle, with z_0 as its center, which lies inside the region of analyticity.

32. By applying the inequality (80) to the result of Problem 31, deduce that the absolute value of an analytic function at a point z_0 cannot exceed the maximum absolute value of that function along any circle, with center at z_0 , which lies inside the region of analyticity.

Section 10.7

33. If f(z) is analytic inside and on a circle C with center a and radius R, and if $|f(z)| \leq M$ on C, use (88) and (80) to show that

$$\left|f^{(n)}(a)\right| \leq \frac{n! M}{R^n}.$$

(This result is known as *Cauchy's inequality*.)

34. Suppose that, by any method, we have obtained a representation

$$f(z) = \sum_{n=0}^{\infty} a_n (z-a)^n$$

which converges to f(z) when |z - a| < R. By making use of the fact that such a series can be differentiated term by term any number of times inside the circle of convergence, show that there must follow

$$f^{(n)}(a) = n! a_n$$
 $(n = 0, 1, 2, ...),$

so that this series necessarily is the Taylor series of f(z) in that circle.

35. Obtain each of the following series expansions by any convenient method:

(a)
$$\frac{\sin z}{z} = 1 - \frac{z^2}{3!} + \frac{z^4}{5!} - \dots = \sum_{n=0}^{\infty} (-1)^n \frac{z^{2n}}{(2n+1)!}$$
 $(|z| < \infty),$
(b) $\frac{\cosh z - 1}{z^2} = \frac{1}{2!} + \frac{z^2}{4!} + \frac{z^4}{6!} + \dots = \sum_{n=0}^{\infty} \frac{z^{2n}}{(2n+2)!}$ $(|z| < \infty),$
(c) $\frac{e^z}{1-z} = 1 + 2z + \frac{5}{2}z^2 + \frac{8}{3}z^3 + \dots$ $(|z| < 1).$

Section 10.8

36. Expand the function f(z) = 1/(1 - z) in each of the following series:

(a) a Taylor series of powers of z for |z| < 1;

(b) a Laurent series of powers of z for |z| > 1;

(c) a Taylor series of powers of z + 1 for |z + 1| < 2, first writing $f(z) = [2 - (z + 1)]^{-1} = \frac{1}{2}[1 - (z + 1)/2]^{-1};$

(d) a Laurent series of powers of z + 1 for |z + 1| > 2, first writing f(z) = -[1/(z + 1)]/[1 - 2/(z + 1)].

37. Expand the function f(z) = 1/[z(1 - z)] in a Laurent (or Taylor) series which converges in each of the following regions:

(a) $0 < z < 1$,	(b) $ z > 1$,
(c) $0 < z - 1 < 1$,	(d) $ z-1 > 1$,
(e) $ z+1 < 1$,	(f) $1 < z+1 < 2$,
(g) $ z+1 > 2$.	

38. Determine the first three nonzero terms in the Laurent expansion of $f(z) = \csc z$ which is valid when $0 < |z| < \pi$ by first showing that the expansion must be of the form $f(z) = a_{-1}z^{-1} + a_1z + a_3z^3 + \cdots$, then determining a_{-1} , a_1 , and a_3 from the condition

$$1 = \left(\frac{a_{-1}}{z} + a_1 z + a_3 z^3 + \cdots\right) \left(z - \frac{1}{6} z^3 + \frac{1}{120} z^5 - \cdots\right).$$

Section 10.9

39. (a) Show that the function $w = A \log (z - a)$ increases by $2\pi i A$ after a single positive circuit around the point z = a, and is unchanged if the circuit does not include that point.

(b) Show that the expression for $w = \tan^{-1} z \operatorname{can} be written in the form$

$$w = \tan^{-1} z = \frac{i}{2} \log (z + i) - \frac{i}{2} \log (z - i) + \frac{i}{2} \log (-1).$$

If the point z describes a single positive circuit, obtain the increase in w when the circuit includes z = i but excludes z = -i, and when the circuit includes z = -i but excludes z = i. Thus verify that w is unchanged if the circuit surrounds both branch points.

40. With the notation of Problem 17, show that

$$\operatorname{coth}^{-1} \frac{z}{a} = \frac{1}{2} \log \frac{\rho_2}{\rho_1} + \frac{i}{2} (\varphi_2 - \varphi_1)$$

defines a single-valued function in the plane cut along the real axis from -a to a if φ_1 and φ_2 are restricted such that $0 \leq \varphi_1 < 2\pi$ and $0 \leq \varphi_2 < 2\pi$, and a single-valued function in the plane cut by infinite rays extending from the points $z = \pm a$ along the real axis if $0 \leq \varphi_1 < 2\pi$ and $-\pi < \varphi_2 \leq +\pi$.

41. (a) With the notation of Problem 17, show that we may write

$$(z^2 - a^2)^{1/2} = (\rho_1 \rho_2)^{1/2} e^{i(\varphi_1 + \varphi_2)/2},$$

and deduce that transition from one branch to the other can take place only if $(\varphi_1 + \varphi_2)/2$ changes abruptly by an odd integral multiple of π .

(b) Show that, with the restrictions $0 \le \varphi_1 < 2\pi$ and $0 \le \varphi_2 < 2\pi$, the angle $(\varphi_1 + \varphi_2)/2$ has a jump of π over the finite segment between -a and a, is continuous across the real axis to the left of -a, and has a jump of 2π across the real axis to the right of +a, if a is real and positive. Deduce that a cut between -a and +a then is necessary and sufficient to make the function so defined single-valued. Show also that the branch so defined makes f(z) real and of the same sign as z when z is real and $z^2 > a^2$.

(c) In a similar way, show that cuts are needed along infinite rays from the branch points $z = \pm a$ if the restrictions $0 \le \varphi_1 < 2\pi$ and $-\pi < \varphi_2 \le +\pi$ are imposed.

42. Locate and classify the singularities of the following functions:

(a)
$$\frac{z}{z^2+1}$$
, (b) $\frac{1}{z^3+1}$, (c) $\log(1+z^2)$,
(d) $(z^2-3z+2)^{2/3}$, (e) $\tan z$, (f) $\tan^{-1}(z-1)$.

43. Show that the function

$$f(z) = \frac{\cosh z - 1}{\sinh z - z}$$

has a simple pole at the origin.

44. Show that the function $f(z) = \csc(1/z)$ has poles at the points $z = 1/(n\pi)$, where *n* is any integer other than zero, and deduce that f(z) has a nonisolated essential singularity at z = 0.

45. Show that the function $w = 1/(1 + z^{1/2})$ has a branch point at z = 0, and that if the principal branch of $z^{1/2}$ is chosen there is no other finite singularity, whereas if the second branch is taken there is also a pole at z = 1. [Notice that we can write also $w = (1 - z^{1/2})/(1 - z)$; also that $z^{1/2}$ is +1 when z = 1 on the principal branch and is -1 when z = 1 on the second branch.]

46. Using the convergence theorem stated on page 132, determine inside what real interval an infinite Frobenius series of the form

$$y(x) = \sum_{k=0}^{\infty} A_k x^{k+s},$$

satisfying each of the following differential equations, would converge [excluding the point x = 0 when Re (s) < 0]:

(a)
$$(1 - x^2)y'' - 2xy' + p(p + 1)y = 0$$
,
(b) $x(1 + x^2)y'' + y' + xy = 0$,
(c) $x^2y'' + xy' + (x^2 - p^2)y = 0$,
(d) $y'' \sin x + y' \cos x + p(p + 1)y \sin x = 0$,
(e) $x(x^2 + 2x + 2)y'' - y' + (x + 1)y = 0$,
(f) $x e^x y'' + y \tan x = 0$.

Section 10.10

47. (a) If f(z) is analytic at z_{∞} , show that the real and imaginary parts of f(z) must each tend to constant limits (which may be zero) as $\sqrt{x^2 + y^2} \rightarrow \infty$ in any way, and that these limits must be independent of the manner in which this limiting process takes place.

(b) If f(z) has a simple pole at z_{∞} , show that the preceding statement applies instead to the real and imaginary parts of f'(z).

48. (a) Show that $|e^z| = e^x$. Hence deduce that $e^z \to 0$ if $z \to z_{\infty}$ on any curve along which $x \to -\infty$, that $|e^z| \to \infty$ along a curve for which $x \to +\infty$, and that along a curve with an asymptote parallel to the y axis $|e^z|$ tends to a finite limit whereas e^z does not.

(b) Obtain a corresponding result for each of the functions e^{-z} , e^{iz} , and e^{-iz} .

49. (a) Show that $|\sinh z| = \sqrt{\sinh^2 x + \sin^2 y}$. Hence deduce that $|\sinh z| \to \infty$ as $z \to z_{\infty}$ along any curve which is not asymptotic to (or coincident with) a line parallel to the imaginary axis, and that along any of the latter curves $|\sinh z|$ is bounded but does not tend to a limit.

(b) Obtain a corresponding result for each of the functions $\cosh z$, $\sin z$, and $\cos z$.

50. Determine the nature of the point z_{∞} for the following functions:

(a)
$$z^2$$
, (b) $\frac{z}{z+1}$, (c) $z \sin \frac{1}{z}$,
(d) $(1+z)^{1/2}$, (e) $(1+z^2)^{1/2}$, (f) $\log(1+z)$.

51. Show that for the function $w = \sin^{-1} z + i \log z$ [see (55a)] the point z_{∞} is a branch point only on those branches for which w does not remain finite as $z \rightarrow z_{\infty}$.

52. By considering the function f(z) = 1/z, show that the integral $\oint_{C_{\infty}} f(z) dz$ may not vanish even though f(z) is analytic at z_{∞} . [Notice that in this case C_{∞} is equivalent to any other contour enclosing the origin].

Section 10.11

53. Prove that any polynomial of degree N, $f(z) = a_0 + a_1 z + \cdots + a_N z^N$, has at least one zero unless it is constant. [Assume the contrary and apply Liouville's theorem to F(z) = 1/f(z). This result is known as the fundamental theorem of algebra and is assumed in elementary courses.]

54. (a) If f(z) has a pole of order *n* at $z = \alpha_1$, show that

$$f(z) = (z - \alpha_1)^{-n} g_1(z),$$

where $g_1(z)$ is analytic at $z = \alpha_1$ and at all points where f(z) is analytic.

(b) If f(z) has a zero of order m at $z = \beta_1$, show that

$$f(z)=(z-\beta_1)^m\,h_1(z),$$

where $h_1(z)$ is analytic where f(z) is analytic and $h_1(\beta_1) \neq 0$.

(c) Deduce that a function f(z) which is analytic except for a finite number of poles is determined except for a multiplicative constant by the position and order of its poles and zeros.

55. Let f(z) be analytic inside and on a closed curve C except for a finite number of poles inside C, and suppose that $f(z) \neq 0$ on C.

(a) Make use of the results of Problem 54 to show that

$$\frac{1}{2\pi i}\oint_C\frac{f'(z)}{f(z)}\,dz\,=M-N,$$

where M is the number of zeros of f(z) inside C and N is the number of poles, poles or zeros of order k being counted k times.

(b) Show that this result also can be expressed in the form

$$M - N = \frac{1}{2\pi} [\Delta_C \operatorname{amp} f(z)],$$

where the bracketed expression denotes the change in the imaginary part of $\log f(z)$ corresponding to a counterclockwise circuit of C.

56. Use Cauchy's inequality (Problem 33) to give an alternative proof of Liouville's theorem. [Show that $f(z) = \sum_{0}^{\infty} a_n z^n$ and that $|a_n| \leq M/R^n$, where M is independent of R and where R may be increased without limit.]

Section 10.12

57. Calculate the residues of the following functions at each of their finite poles, and in each case find also the sum of these residues:

(a)
$$\frac{e^z}{z^2 + a^2}$$
, (b) $\frac{1}{z^4 - a^4}$, (c) $\frac{\sin z}{z^2}$,
(d) $\frac{\sin z}{z^3}$, (e) $\frac{1 + z^2}{z(z - 1)^2}$, (f) $\frac{1}{(z^2 + a^2)^2}$

58. Show that the substitution z = 1/t transforms the circle |z| = R into the circle |t| = 1/R in such a way that a positive circuit around one circle corresponds to a negative circuit around the other. [Write $z = Re^{i\theta}$, $t = \rho e^{i\varphi}$ and determine ρ,φ in terms of R,θ .]

59. (a) Use the result of Problem 58 to show that

$$\oint_{|z|=R} f(z) dz = \oint_{|t|=1/R} f\left(\frac{1}{t}\right) \frac{dt}{t^2}.$$

By letting $R \to \infty$, deduce that $\oint_{C_{\infty}} f(z) dz$ is given by $2\pi i$ times the residue of $\frac{1}{t^2} f\left(\frac{1}{t}\right) at t = 0.$

(b) Show that this result implies the result of Problem 52.

(c) Use the result of part (a) to show that if $z^2 f(z)$ is analytic at z_{∞} , then $\oint_{C_{\infty}} f(z) dz = 0$.

60. It is conventional to define the residue of f(z) at z_{∞} , when z_{∞} is an isolated singular point, by the equation

$$\oint_{C_{\infty}} f(z) \, dz = -2\pi i \operatorname{Res} (z_{\infty}),$$

the negative sign corresponding to the fact that a positive circuit around C_{∞} is described in a negative sense with respect to the *exterior* of the curve.

(a) Use the result of Problem 59 to show that then

$$\operatorname{Res}\left\{f(z); z_{\infty}\right\} = -\operatorname{Res}\left\{\frac{1}{t^2}f\left(\frac{1}{t}\right); 0\right\}.$$

(b) Use (108) to show that, with the given definition of Res (z_{∞}) , the sum of the residues of f(z) at all finite points and at z_{∞} is zero. [Notice that f(z) may have a nonzero residue at z_{∞} even though f(z) is analytic at z_{∞} , if $z^2 f(z)$ has a pole at z_{∞} .]

61. Use the result of Problem 59(a) (or Problem 60) to evaluate the integral

$$\oint_C \frac{a^2-z^2}{a^2+z^2} \frac{dz}{z},$$

where C is any positive contour enclosing the points z = 0, $\pm ia$, and check the result by calculating the residues at those poles.

62. (a) If f(z) can be represented by a Laurent series

$$f(z) = \sum_{n = -\infty}^{\infty} a_n (z - a)^n$$

when $R < |z - a| < \infty$, for some R, show that a_{-1} is the sum of the residues of f(z) at all singularities in the finite part of the plane, so that the number $-a_{-1}$ is the residue of f(z) at z_{∞} . (See Problem 60.)

(b) By expanding the integrand in increasing powers of 1/z, and identifying the coefficient of 1/z, show that

$$\oint_C \frac{a^2 - z^2}{a^2 + z^2} \frac{dz}{z} = -2\pi i$$

if C is any closed curve enclosing the points z = 0 and $z = \pm ai$. (Compare Problem 61.)

63. Determine the residue of each of the following functions at each singularity:

(a)
$$e^{1/z}$$
, (b) e^{1/z^2} ,
(c) $\cos\left(\frac{\pi}{z-\pi}\right)$, (d) $\sin\left(\frac{z^2-1}{z^2+1}\right)$.

Section 10.13

64. Verify the following evaluations by making use of appropriate results or procedures in Section 10.13:

(a)
$$\int_{0}^{2\pi} \frac{d\theta}{A + B\sin\theta} = \frac{2\pi}{\sqrt{A^{2} - B^{2}}} \quad (A > |B|),$$

(b)
$$\int_{0}^{\infty} \frac{dx}{x^{4} + 4a^{4}} = \frac{\pi}{8a^{3}} \quad (a > 0),$$

(c)
$$\int_{0}^{\infty} \frac{x\sin mx}{a^{2} + x^{2}} dx = \frac{\pi}{2}e^{-am} \quad (a > 0, m > 0),$$

(d)
$$\int_{0}^{\infty} \frac{\cos mx}{x^{4} + 4a^{4}} dx = \frac{\pi}{8a^{3}}e^{-am}(\cos am + \sin am) \quad (a > 0, m > 0).$$

65. Use residue calculus to show that

$$\int_0^{2\pi} \frac{d\theta}{25 - 16\cos^2\theta} = \frac{2\pi}{15},$$

and check the result by first making the change in variables $\theta = \varphi/2$ and then using the result of Equation (138).

66. Use residue calculus to show that

$$\int_{-\infty}^{\infty} \frac{x^2 \, dx}{(x^2+1)(x^2-2x\cos\omega+1)} = \frac{\pi}{2\,|\sin\omega|}$$

if ω is real and sin $\omega \neq 0$.

67. Use residue calculus to show that

$$\int_{-\infty}^{\infty} \frac{\cos mx}{(x^2 + a^2)^2} dx = \pi \frac{1 + am}{2a^3} e^{-am} \qquad (m \ge 0, a > 0).$$

68. It is required to evaluate

$$\int_{-\infty}^{\infty} e^{-x^2} \cos 2ax \, dx,$$

by making use of the fact that

$$\int_{-\infty}^{\infty} e^{-x^2} dx = \sqrt{\pi}.$$

Show that, if x is formally replaced by x + ia in the known integral, there follows

$$\int_{-\infty}^{\infty} e^{-(x+ia)^2} dx = e^{a^2} \int_{-\infty}^{\infty} e^{-x^2} (\cos 2ax - i \sin 2ax) dx = \sqrt{\pi},$$

and hence, since $e^{-x^2} \sin 2ax$ is an odd function of x,

$$\int_{-\infty}^{\infty} e^{-x^2} \cos 2ax \, dx = \sqrt{\pi} e^{-a^2}.$$

(Notice, however, that the validity of such complex substitutions is not established by the familiar rules for real substitutions.)

69. Investigate the validity of the procedure of Problem 68, by considering the integral of $f(z) = e^{-z^2}$ over a closed rectangular path C including the segment of the real axis y = 0 from x = -A to x = +A and the segment of the line y = a from x = +A to x = -A.

(a) Noticing that f(z) is analytic inside and on C, show that

$$\int_{-A}^{A} e^{-x^2} dx - \int_{-A}^{A} e^{-(x+ia)^2} dx + \int_{S_1} e^{-z^2} dz + \int_{S_2} e^{-z^2} dz = 0,$$

where S_1 is the line segment from (A,0) to (A,a) and S_2 is the line segment from (-A,a) to (-A,0).

(b) Show that, on both S_1 and S_2 , there follows

$$|e^{-z^2}| = e^{-(A^2-y^2)} \leq e^{-(A^2-a^2)} = e^{-A^2(1-\frac{a^2}{A^2})},$$

and, by noticing that the lengths of S_1 and S_2 are equal to a, deduce that the integrals along S_1 and S_2 tend to zero as $A \to \infty$ for any fixed value of a. Thus deduce that the relation

$$\int_{-\infty}^{\infty} e^{-x^2} dx = \int_{-\infty}^{\infty} e^{-(x+ia)^2} dx$$

is valid for any real value of a, and hence that the result obtained formally in Problem 68 is indeed correct.

70. Evaluate the integral

$$\int_{-\infty}^{\infty} \frac{e^{px}}{1+e^x} dx \qquad (0$$

by the following method:

(a) Show that the result of integrating $f(z) = e^{pz}/(1 + e^z)$ around a closed rectangular contour C, including the real axis from x = -A to x = +A and the line $y = 2\pi$ from x = +A to x = -A, can be written in the form

$$\int_{-A}^{A} \frac{e^{px}}{1+e^{x}} dx - e^{2p\pi i} \int_{-A}^{A} \frac{e^{px}}{1+e^{x}} dx + \int_{S_{1}} f(z) dz + \int_{S_{2}} f(z) dz$$
$$= 2\pi i \operatorname{Res} \left\{ \frac{e^{pz}}{1+e^{z}}; \pi i \right\},$$

where S_1 and S_2 are the closing segments of the rectangle. (Notice that e^z has the period $2\pi i$, and that f(z) has poles at the points $\pi i \pm 2k\pi i$.)

(b) Show that the integrals along S_1 and S_2 tend to zero as $A \to \infty$ (if 0), and hence deduce that

$$\int_{-\infty}^{\infty} \frac{e^{px}}{1+e^x} \, dx = \left(\frac{2\pi i}{1-e^{2p\pi i}}\right)(-e^{p\pi i}) = \frac{\pi}{\sin p\pi} \qquad (0$$

71. Show that the formula obtained in Problem 70 also is valid when p is complex, provided that 0 < Re(p) < 1. (Only the limits of the integrals along S_1 and S_2 need be reexamined.)

Section 10.14

72. By making the substitution x = iy, and noticing that then $y \to \infty$ as $x \to \infty$, we formally transform the integral

$$\int_0^\infty \frac{x}{x^4+4}\,dx$$

into the integral

$$-\int_0^\infty \frac{y}{y^4+4}\,dy,$$

which is the negative of the original integral, and hence are led to the conclusion that the value of the integral is zero. But the true value is found by elementary methods to be $\pi/8$. What is the fallacy?

[Integrate $f(z) = z/(z^4 + 4)$ around a contour consisting of the portions of the x and y axes for which $0 \le x, y \le R$, and a quadrant of the circle |z| = R, as $R \to \infty$, and interpret the result.]

73. Establish part 3 of Theorem II.

74. If f(z) is analytic everywhere on the imaginary axis and has no singularities to the left of that axis except for a finite number of poles at the points a_1, a_2, \ldots , a_n , and if f(z) tends to zero uniformly on the circle |z| = R as $R \to \infty$ in the third and fourth quadrants, show that

$$\lim_{R\to\infty}\int_{-iR}^{iR} e^{mz}f(z)\,dz\,=2\pi i\sum_{k=1}^{n}\operatorname{Res}\left\{e^{mz}f(z);\,a_{k}\right\}\qquad(m>0),$$

where the integration is carried out along the imaginary axis.

75. Suppose that, on a circular arc C_R with radius R and center at the origin, lying in the sector $-\pi/4 \le \theta \le +\pi/4$, a function f(z) is such that

$$|f(z)| \leq RM_R, \quad \lim_{R\to\infty} M_R = 0,$$

where M_R is independent of θ on C_R . Prove that

$$\lim_{R\to\infty}\int_{C_R}e^{-mz^2}f(z)\,dz\,=0\qquad(m>0)$$

[If the integral along C_R is denoted by I_R , show that

$$\left|I_{R}\right| \leq 2R^{2}M_{R}\int_{0}^{\pi/4}e^{-mR^{2}\cos 2\theta} d\theta,$$

and set $\theta = (\pi - 2\varphi)/4.$]

76. Use the result of Problem 75 to prove that if $|\beta| \le \pi/4$ and if f(z) is analytic throughout the sector bounded by $\theta = 0$, $\theta = \beta$, and r = R, except for a finite number of interior poles, and if $|f(z)| \le RM_R$ on the curved boundary, where M_R is independent of θ and $M_R \to 0$ as $R \to \infty$, then

$$\int_0^\infty e^{-mx^2} f(x) \, dx = \int_0^\infty e^{-mz^2} f(z) \, dz + 2\pi i \sum_k \operatorname{Res} \left\{ e^{-mz^2} f(z); a_k \right\} \qquad (m > 0)$$

where Γ is the radial line $\theta = \beta$, and where the points a_k are the poles of f(z) inside the sector bounded by $\theta = 0$ and $\theta = \beta$. [Notice that hence the formal complex substitution $x = t e^{i\beta}$ in the first (real) integral generally would modify the value of the integral if f(z) were not analytic in the relevant sector.]

77. (a) Show that the integrals

$$C \equiv \int_0^\infty \cos t^2 dt, \qquad S \equiv \int_0^\infty \sin t^2 dt$$

can be combined in the form

$$C - iS = \int_0^\infty e^{-it^2} dt = \int_0^\infty e^{-t^2 e^{i\pi/2}} dt = e^{-\pi i/4} \int_0^\infty e^{-z^2} dz,$$

where Γ is the radial line $\theta = \pi/4$.

(b) From the result of Problem 76, deduce that

$$C - iS = e^{-\pi i/4} \int_0^\infty e^{-x^2} dx = e^{-\pi i/4} \frac{\sqrt{\pi}}{2},$$

Problems

and hence also that

$$\int_0^\infty \cos t^2 \, dt = \int_0^\infty \sin t^2 \, dt = \frac{1}{2} \, \sqrt{\frac{\pi}{2}} \, .$$

Section 10.15

78. By making use of integration around suitably indented contours in the complex plane, evaluate the following integrals:

(a)
$$\int_{-\infty}^{\infty} \frac{\sin x}{x(x^2 + a^2)} dx \qquad (a > 0),$$

(b)
$$\int_{-\infty}^{\infty} \frac{\sin x}{x(\pi^2 - x^2)} dx.$$

79. Evaluate the integral

$$I = \int_0^\infty \frac{x^{m-1}}{x+1} \, dx \qquad (0 < m < 1)$$

by integrating $f(z) = z^{m-1}/(z+1)$ around the contour of Figure 10.36 and proceeding to the limit as the small radius tends to zero and the large radius tends



Figure 10.36

to infinity. (Notice that if along the upper boundary of a cut along the positive real axis we write z = r, then along the lower boundary we must write $z = re^{2\pi i}$ when calculating z^{m-1} .) Show that in the limit we have

$$\int_0^\infty \frac{r^{m-1}}{r+1} dr + \int_\infty^0 \frac{(re^{2\pi i})^{m-1}}{r+1} dr = 2\pi i \operatorname{Res} \left\{ \frac{z^{m-1}}{z+1} ; -1 \right\}$$

or, equivalently,

$$(1 - e^{2\pi m i})I = -2\pi i e^{\pi m i}$$

Hence obtain the result

$$\int_0^\infty \frac{x^{m-1}}{x+1} \, dx = \frac{\pi}{\sin \pi m} \qquad (0 < m < 1).$$

80. Use the result of Problem 45, Chapter 2, to evaluate the integral of the preceding problem 79 in the form

$$\int_0^\infty \frac{x^{m-1}}{x+1} \, dx = \Gamma(m) \, \Gamma(1-m) \qquad (0 < m < 1),$$

and hence, by comparing the two results, deduce the relation

$$\Gamma(m) \ \Gamma(1 - m) = \frac{\pi}{\sin \pi m} \qquad (0 < m < 1).$$

[This relation was stated without proof in Equation (59), Chapter 2. Although the present proof is valid only when 0 < m < 1, the relation is generalized to all nonintegral values of m by making use of the recurrence formula for the Gamma function.]

81. Suppose that $z^m f(z)$ tends uniformly to zero on the circle |z| = R as $R \to \infty$, where m > 0, and that f(z) is analytic except for a finite number of poles a_1, a_2, \ldots, a_n , none of which is on the *positive* real axis or at the origin.

(a) By proceeding as in Problem 79, show that

$$\int_0^\infty x^{m-1} f(x) \, dx = \frac{2\pi i}{1 - e^{2\pi m i}} \sum_{k=1}^n \operatorname{Res} \left\{ z^{m-1} f(z); a_k \right\},$$

where $z^{m-1} = r^{m-1} e^{i(m-1)\theta}$, $0 \le \theta < 2\pi$, when $z = r e^{i\theta}$.

(b) Show that this result can also be written in the form

$$\int_0^\infty x^{m-1} f(x) \, dx = \frac{\pi}{\sin m\pi} \sum_{k=1}^n \operatorname{Res} \left\{ (-z)^{m-1} f(z); a_k \right\}$$

with the definition $(-z)^{m-1} = r^{m-1} e^{i(m-1)(\theta-\pi)}, 0 \leq \theta < 2\pi$, when $z = r e^{i\theta}$.

82. Use the result of Problem 81 to obtain the following evaluations:

(a)
$$\int_0^\infty \frac{x^{m-1}}{x^2+1} dx = \pi \frac{\sin\left[(1-m)\pi/2\right]}{\sin m\pi} = \frac{\pi}{2\sin m\pi/2}$$
 (0 < m < 2),
(b) $\int_0^\infty \frac{x^{m-1}}{(x+1)^2} dx = \frac{(1-m)\pi}{\sin m\pi}$ (0 < m < 2).

83. Suppose that f(z) has a simple pole at a_0 on a closed curve C, but is analytic elsewhere inside and on C except for poles at a finite number of interior points a_1, a_2, \ldots, a_n . If the contour C is indented at a_0 by a circular arc with center at a_0 , show that the limiting form of the integral of f(z) around the indented contour is

$$\mathsf{P} \oint_C f(z) \, dz = \pi i \operatorname{Res} (a_0) + 2\pi i \sum_{k=1}^n \operatorname{Res} (a_k)$$

as the radius of the indentation tends to zero, regardless of whether the indentation excludes or includes the point a_0 .

84. Obtain the evaluation

$$\int_{-\infty}^{\infty} \frac{\cos ax - \cos bx}{x^2} \, dx = \pi(b - a).$$

[Notice that $f(z) = (e^{iaz} - e^{ibz})/z^2$ has a simple pole at the origin.]

85. Obtain the evaluation

$$\int_{-\infty}^{\infty} \frac{\sin\left(x+a\right)\sin\left(x-a\right)}{x^2-a^2} \, dx = \frac{\pi}{2a} \sin 2a.$$

[Notice that $\sin (x + a) \sin (x - a) = \frac{1}{2} (\cos 2a - \cos 2x)$.]

86. (a) Obtain the evaluation

$$\mathsf{P} \int_{-\infty}^{\infty} \frac{e^{px}}{1-e^x} \, dx = \pi \cot p\pi \qquad (0$$

by proceeding as in Problem 70, but also introducing indentations at the poles z = 0 and $z = 2\pi i$.

(b) Deduce the result

$$\int_{-\infty}^{\infty} \frac{e^{px} - e^{qx}}{1 - e^x} dx = \pi(\cot p\pi - \cot q\pi)$$

if 0 and <math>0 < q < 1. (Notice that this integral is convergent, so that principal values are not involved.)

(c) By replacing x by 2x and writing $p = (\omega + 1)/2$ in the result of part (a), deduce that

$$\mathsf{P}\!\int_{-\infty}^{\infty}\frac{e^{\omega x}}{\sinh x}\,dx\,=\pi\,\tan\frac{\pi\omega}{2}\qquad(-1\,<\omega\,<1).$$

(d) By replacing x by bx, where b > 0, and writing $\omega = a/b$, obtain the result

$$\mathsf{P}\!\int_{-\infty}^{\infty}\frac{e^{ax}}{\sinh bx}\,dx\,=\frac{\pi}{b}\tan\frac{\pi a}{2b}\qquad(b>|a|).$$

(e) Deduce that

$$\int_{-\infty}^{\infty} \frac{\sinh ax}{\sinh bx} dx = \frac{\pi}{b} \tan \frac{\pi a}{2b} \qquad (b > |a|).$$

A consideration of Problems 87–94 may serve as an introduction to methods of evaluating inverse Laplace transforms by contour integration. For further information, the references at the end of Chapter 2 may be consulted.

87. (a) Show that for a piecewise differentiable function f(x) which vanishes when x < 0, the complex form of the Fourier integral representation of $e^{-cx} f(x)$ is given, according to Equation (229a) of Section 5.14, by

$$e^{-cx}f(x) = \frac{1}{2\pi}\lim_{R\to\infty}\int_{-R}^{R}e^{iux}\left[\int_{0}^{\infty}e^{-iut}e^{-ct}f(t)\,dt\right]du,$$
for any value of the real constant c for which the integral

$$I = \int_0^\infty e^{-ct} |f(t)| dt$$

exists. Hence deduce the representation

$$f(x) = \frac{1}{2\pi} \lim_{R \to \infty} \int_{-R}^{R} e^{(iu+c)x} \left[\int_{0}^{\infty} e^{-(iu+c)t} f(t) dt \right] du$$

for f(x), with the usual convention that, at a finite jump, f(x) is to be assigned the average of its right- and left-hand limits. (Notice that this representation is more general than the Fourier integral representation, since here the constant c is at our disposal.)

(b) With the substitution

$$iu + c = s$$
,

where s is a new (complex) variable of integration, show that this result becomes

$$f(x) = \frac{1}{2\pi i} \lim_{R \to \infty} \int_{c-Ri}^{c+Ri} e^{sx} \left[\int_0^\infty e^{-st} f(t) dt \right] ds.$$

(c) By making appropriate changes in the notation of part (b), show that if we denote the Laplace transform of f(t) by F(s),

$$F(s) = \int_0^\infty e^{-st} f(t) dt,$$

then f(t) is expressed in terms of its transform by the *complex* integral

$$f(t) = \frac{1}{2\pi i} \lim_{R\to\infty} \int_{c-Ri}^{c+Ri} e^{tz} F(z) dz.$$

[This integral is taken along the infinite line x = c of Figure 10.37(a), parallel to the imaginary axis and c units to the right of it, where c is to be chosen sufficiently large that the integral I of part (a) exists, and is known as the *Bromwich* integral. It is usually to be evaluated by contour integration, as in the following problems.]



Figure 10.37

Problems

88. Suppose that F(z) is analytic except for a finite number of poles, all of which are to the left of some line x = c. Also, suppose that F(z) tends uniformly to zero along the semicircle CDE and the segments BC and EA of Figure 10.37(b).

(a) Show then that the integral of $e^{tz} F(z)$ around the closed boundary B_R of Figure 10.37(b) tends to the Bromwich integral of Problem 87(c) as $R \to \infty$, if t > 0, and hence that under the conditions stated there follows

$$f(t) = \lim_{R\to\infty} \frac{1}{2\pi i} \int_{B_R} e^{tz} F(z) dz.$$

[Use Theorem II.3 of Section 10.14 to treat the semicircle. On each horizontal segment, notice that the length L of the path is constantly c and the modulus of e^{tz} is not greater than e^{tc} , while the modulus of F(z) tends to zero as $R \rightarrow \infty$.]

(b) Deduce that if the real function F(s) is such that the corresponding complex function F(z) satisfies the specified conditions, then F(s) is the Laplace transform of the function f(t) which is given by

$$f(t) = \sum_{k} \operatorname{Res} \{ e^{tz} F(z); a_k \},\$$

where the points a_k are the poles of F(z).

89. Verify the preceding result by determining the inverse Laplace transforms of the following functions, and comparing with the formulas given in Table 1, Chapter 2:

(a)
$$\frac{1}{s+a}$$
, (b) $\frac{1}{s^2+a^2}$, (c) $\frac{1}{s^2-a^2}$.
(d) $\frac{1}{s^4}$, (e) $\frac{s}{(s^2+a^2)^2}$, (f) $\frac{s}{s^4+4a^4}$

90. Suppose that F(z) satisfies the conditions of Problem 88, except that it also possesses a single finite branch point at z = 0.

(a) By introducing a *cut* along the negative real axis, and considering the limit of the contour *ABCDOEFA* of Figure 10.38(a), show that we then have

$$f(t) = \sum_{k} \operatorname{Res} \{ e^{tz} F(z); a_{k} \} + \frac{1}{2\pi i} \int_{L} e^{tz} F(z) \, dz,$$

where the additional integral is carried out over the infinite "loop" L indicated in Figure 10.38(b). [The loop extends to the right along the lower edge of the cut, then around an arbitrarily small circle surrounding the branch point, and back along the upper edge of the cut. Notice that the cut could have been made along any ray (or curve extending from the origin to infinity) in the second or third quadrant.]

(b) Suppose further that z f(z) tends uniformly to zero on the small circle surrounding the origin, as its radius tends to zero. Show then that the loop integral of part (a) becomes

$$\int_L e^{tz} F(z) dz = \int_0^\infty e^{-rt} [F(re^{-i\pi}) - F(re^{i\pi})] dr.$$

(Use Theorem III of Section 10.14 to treat the integral around the circle of vanishing radius.)



Figure 10.38

91. Use the result of Problem 90(b) to determine the inverse Laplace transform of $1/s^m$ where 0 < m < 1 in the form

$$\mathscr{L}^{-1}\left\{\frac{1}{s^m}\right\} = \frac{\sin \pi m}{\pi} \int_0^\infty e^{-rt} r^{-m} dr \qquad (0 < m < 1)$$

and show that this result is equivalent to the result $t^{m-1}/\Gamma(m)$ given in Table 1. Chapter 2, if *rt* is replaced by a new variable and account is taken of Equations (47) and (59) of Chapter 2. [Notice that $F(z) = z^{-m}$ has no finite singularities other than a branch point at z = 0, if 0 < m < 1, and that the conditions of Problem 90 are satisfied. Thus the inverse transform is given entirely by the loop integral.]

92. Use the result of Problem 90(a) to determine the inverse transform of

$$F(s) = \frac{e^{-\beta \sqrt{s}}}{s} \qquad (\beta > 0)$$

in the form

$$f(t) = 1 - \frac{1}{\pi} \int_0^\infty e^{-rt} \frac{\sin \beta \sqrt{r}}{r} dr$$

and, by referring to Equation (235) of Section 9.14, deduce that

$$\mathscr{L}^{-1}\left\{\frac{e^{-\beta\sqrt{s}}}{s}\right\} = 1 - \operatorname{erf}\left(\frac{\beta}{2\sqrt{t}}\right)$$

(Notice that here the integration around the small circle in the loop integral contributes the quantity

$$\frac{1}{2\pi i}\lim_{\rho\to 0}\oint_{C_{\rho}}e^{tz-\beta\sqrt{\frac{1}{z}}}\frac{dz}{z}.$$

On the circle $z = \rho e^{i\theta}$ there follows $\frac{dz}{z} = i d\theta$ and, since the rest of the integrand is continuous at z = 0, it can be replaced by its limit unity as $\rho \rightarrow 0$.)

93 Obtain the inverse transforms

(a)
$$\mathscr{L}^{-1}\left\{\frac{e^{-\beta\sqrt{s}}}{\sqrt{s}}\right\} = \frac{2}{\pi\sqrt{t}} \int_{0}^{\infty} e^{-u^{2}} \cos\frac{\beta u}{\sqrt{t}} \, du = \frac{1}{\sqrt{\pi t}} e^{-\beta^{2}/4t},$$

(b) $\mathscr{L}^{-1}\left\{e^{-\beta\sqrt{s}}\right\} = \frac{2}{\pi t} \int_{0}^{\infty} e^{-u^{2}} \sin\frac{\beta u}{\sqrt{t}} \, u \, du = \frac{\beta}{2\sqrt{\pi t^{3}}} e^{-\beta^{2}/4t}.$

where $\beta > 0$. [Use Equation (235) of Section 9.14 in part (a) and differentiate that equation with respect to b to deal with part (b).]

94. Solve, by use of Laplace transforms, the heat flow problem considered in Section 9.14, for which T(x,t) satisfies the differential equation

$$\frac{\partial^2 T}{\partial x^2} = \frac{1}{\alpha^2} \frac{\partial T}{\partial t},$$

the initial condition $T(x,0) = T_0 = \text{constant}$, and the end condition T(0,t) = 0, with $T(\infty,t)$ finite. [Show that the transform T then satisfies the equation

$$\frac{\partial^2 \bar{T}}{\partial x^2} - \frac{s}{\alpha^2} \bar{T} = -\frac{T_0}{\alpha^2}$$

and hence must be of the form

$$\bar{T} = A(s) e^{-x\sqrt{s}/\alpha} + B(s) e^{x\sqrt{s}/\alpha} + \frac{T_0}{s}.$$

Determine A and B so that the end conditions are satisfied, and show that

$$T = T_0 \left(\frac{1}{s} - \frac{e^{-x\sqrt{s}/\alpha}}{s}\right).$$

Thus, using the result of Problem 92, obtain the result

$$T=T_0 \operatorname{erf}\left(\frac{x}{2\alpha\sqrt{t}}\right),$$

in accordance with the result of Section 9.14.]

Section 10.16

95. Consider the mapping w = az + b, where a and b are complex constants and $a \neq 0$.

(a) Show that the mapping is one-to-one everywhere.

(b) Verify that the mapping consists of a *rotation* of the z plane about the origin through the angle arg a and a homogeneous magnification or *stretching* by a factor |a|, followed by a *translation* through the vector whose components are the real and imaginary parts of b.

(c) Deduce, in particular, that circles map into circles, and straight lines into straight lines.

96. Consider the mapping w = 1/z.

(a) Show that the mapping is one-to-one everywhere when the points z_{∞} and w_{∞} are included.

(b) With the notations $w = \rho e^{i\varphi}$ and $z = re^{i\theta}$, show that $\rho = 1/r$ and $\varphi = -\theta$, and deduce that the mapping consists of an *inversion* with respect to the unit circle combined with a *reflection* about the real axis. [Notice that the point (r,θ) maps into the point $(1/r, -\theta)$.]

(c) Show that the general straight line in the z plane has the polar equation $r \cos(\theta - \theta_0) = p$, where θ_0 is the angle made by its normal with the positive x axis and p is its distance from the origin, and deduce that its image in the w plane is the curve $p_{\rho} = \cos(\varphi + \theta_0)$, where ρ and φ are polar coordinates in the w plane.

(d) Show that the general circle in the z plane, with center at the point with polar coordinates (r_0, θ_0) and radius R, has the polar equation

$$r^{2} - 2r_{0}r\cos(\theta - \theta_{0}) + (r_{0}^{2} - R^{2}) = 0,$$

and deduce that its image in the w plane is the curve

$$(r_0^2 - R^2)\rho^2 - 2r_0\rho\cos(\varphi + \theta_0) + 1 = 0.$$

(e) Deduce from the results of parts (c) and (d) that, under the mapping w = 1/z, circles and straight lines in the z plane not passing through the origin map into circles in the w plane, whereas circles and straight lines in the z plane passing through the origin map into straight lines in the w plane.

97. Consider the mapping corresponding to the general linear fractional transformation

$$w = rac{lpha z + eta}{\gamma z + \delta} \qquad (\Delta \equiv lpha \delta - eta \gamma
eq 0),$$

where α , β , γ , and δ are complex constants.

(a) Show that the mapping is one-to-one everywhere (when z_{∞} and w_{∞} are included) but could not be so if $\Delta = 0$. (Notice that the z plane maps into a single point if $\Delta = 0$.)

(b) Verify that the mapping can be considered as the result of the three successive mappings

$$z' = \gamma z + \delta, \qquad z'' = \frac{1}{z'}, \qquad w = -\frac{\Delta}{\gamma} z'' + \frac{\alpha}{\gamma}$$

when $\gamma \neq 0$, and is of the form

$$w=\frac{\alpha}{\delta}z+\frac{\beta}{\delta}$$

when $\gamma = 0$.

(c) From the results of Problems 95 and 96, and the results of part (b), deduce that any linear fractional transformation always maps circles into circles, with the convention that straight lines are to be considered as limiting forms of circles.

98. Mappings into a half-plane. In each of the following mappings, a specified region in the z plane is mapped into the upper half ($v \ge 0$) of the w plane, with an indicated correspondence of certain boundary points. Verify the correctness of each mapping and show also that $\frac{dw}{dz}$ exists and is nonzero everywhere inside the mapped region.

(a) Infinite strip:

 $(-\infty < x < \infty, \quad 0 \leq y \leq b).$ $w = e^{\pi z/b}.$

Mapping:

Boundary points:

$$(+\infty,b) \to (-\infty,0), \quad (0,b) \to (-1,0), \quad (-\infty,y) \to (0,0),$$

 $(0,0) \to (1,0), \quad (+\infty,0) \to (+\infty,0).$

(b) Semi-infinite strip:

Mapping:

$$(0 \le x \le a, \quad 0 \le y < \infty).$$
$$w = -\cos \frac{\pi z}{a}.$$

Boundary points:

$$(0, +\infty) \rightarrow (-\infty, 0), \quad (0, 0) \rightarrow (-1, 0), \quad (\frac{1}{2}a, 0) \rightarrow (0, 0),$$

 $(a, 0) \rightarrow (1, 0), \quad (a, +\infty) \rightarrow (+\infty, 0).$

(c) Interior of circle:

(
$$|z| \leq R$$
).
Mapping: $w = i \frac{R-z}{R+z}$.

Boundary points:

$$(-R,0-) \rightarrow (-\infty,0), \quad (0,-R) \rightarrow (-1,0), \quad (R,0) \rightarrow (0,0),$$

 $(0,R) \rightarrow (1,0), \quad (-R,0+) \rightarrow (+\infty,0).$

(d) Interior of semicircle:

$$(|z| \leq R, y \geq 0).$$

Mapping: $w = -\frac{1}{2}\left(\frac{z}{R} + \frac{R}{z}\right).$
Boundary points:

$$(0+,0) \rightarrow (-\infty,0), \quad (R,0) \rightarrow (-1,0), \quad (0,R) \rightarrow (0,0),$$

 $(-R,0) \rightarrow (1,0), \quad (0-,0) \rightarrow (+\infty,0).$

(e) Half-plane indented by semicircle:

(
$$|z| \ge R, y \ge 0$$
).
Mapping: $w = \frac{1}{2} \left(\frac{z}{R} + \frac{R}{z} \right).$

Boundary points:

$$(-\infty,0) \rightarrow (-\infty,0), (-R,0) \rightarrow (-1,0), (0,R) \rightarrow (0,0),$$

 $(R,0) \rightarrow (1,0), (+\infty,0) \rightarrow (+\infty,0).$

99. Mappings of sectors. Verify each of the following mappings, showing also that $\frac{dw}{dz}$ exists and is nonzero everywhere inside each specified mapped region, and assuming that principal values are to be taken in each case.

(a) Circular sector into unit semicircle:

$$(|z| \leq R, 0 \leq \theta \leq \alpha) \rightarrow (|w| \leq 1, 0 \leq \varphi \leq \pi).$$

Mapping: $w = \left(\frac{z}{R}\right)^{\pi/\alpha}.$

Boundary points:

$$0 \rightarrow 0, R \rightarrow 1, R e^{i\alpha/2} \rightarrow i, R e^{i\alpha} \rightarrow -1.$$

(b) Circular sector into half-plane:

$$(|z| \leq R, 0 \leq \theta \leq \alpha) \rightarrow (v \geq 0).$$

Mapping: $w = -\frac{1}{2} \left[\left(\frac{z}{R} \right)^{\pi/\alpha} + \left(\frac{R}{z} \right)^{\pi/\alpha} \right].$

Boundary points:

$$0 e^{i0} \rightarrow \infty e^{i\pi}, R \rightarrow -1, R e^{i\alpha/2} \rightarrow 0, R e^{i\alpha} \rightarrow 1, 0 e^{i\alpha} \rightarrow \infty e^{i0},$$

(c) Infinite sector into half-plane:

$$(0 \leq \theta \leq \alpha) \rightarrow (v \geq 0)$$

Mapping:

$$w=z^{\pi/\alpha}.$$

Boundary points:

$$\infty e^{i\alpha} \rightarrow \infty e^{i\pi}, \quad 0 \rightarrow 0, \quad \infty e^{i0} \rightarrow \infty e^{i0}.$$

Section 10.17

100. With the notation of Section 6.18, show that the scale factors associated with the curvilinear coordinates defined by the relation u + iv = f(x + iy) are given by $h_u = h_v = \left|\frac{df}{dz}\right|^{-1}$. Hence, by using the results of that section, deduce that

$$\nabla \varphi = \left| \frac{df}{dz} \right| \left(\frac{\partial \varphi}{\partial u} \mathbf{u} + \frac{\partial \varphi}{\partial v} \mathbf{v} \right),$$

where u and v are unit vectors in the u and v directions, and also that

$$\nabla^2 \varphi = \left| \frac{df}{dz} \right|^2 \left(\frac{\partial^2 \varphi}{\partial u^2} + \frac{\partial^2 \varphi}{\partial v^2} \right)$$

where ∇^2 is the two-dimensional Laplacian operator, in accordance with the results of Section 10.17.

101. Bernoulli's equation [Equation (206) of Section 6.20] states that the pressure at a point in an incompressible fluid is given by

$$p=p_0-\frac{1}{2}\rho V^2,$$

where p_0 is the equilibrium pressure, ρ is the density, and $V = |\mathbf{V}|$ is the absolute velocity at the point.

(a) Show that the *normal force* per unit length, exerted by fluid flowing past a cylindrical surface whose cross section is bounded by a closed curve C in the z plane, has components

$$-\oint_C p \frac{dy}{ds} ds$$
 and $+\oint_C p \frac{dx}{ds} ds$

in the x and y directions, respectively, and show that these components are the real and imaginary parts of the complex quantity $i \oint_C p \, dz$.

(b) By making use of Bernoulli's equation, deduce the relation

$$-D + iL = -\frac{1}{2}i
ho \oint_C \left|\frac{d\Phi}{dz}\right|^2 dz,$$

where Φ is the complex potential function, D is the *drag* component of the force due to pressure in the negative x direction, and L is the *lift* component in the positive y direction, this terminology implying that the preponderant flow is in the positive x direction.

102. (a) Show that the complex potential $\Phi = Aw$ (A real and positive) corresponds to a steady flow, with velocity A in the positive u direction, in the w plane.

(b) Use the result of Problem 98(e) to map the upper half of the w plane into the upper half of the z plane, indented by a semicircle of radius a with center at the origin.

(c) Show that the complex potential takes the form

$$\Phi = \frac{A}{2}\left(\frac{z}{a}+\frac{a}{z}\right).$$

By noticing that the real axis in the w plane corresponds to the portion of the x axis outside the circle $x^2 + y^2 = a^2$, and to the upper half of the boundary of that circle, deduce that the flow has this composite contour as a streamline.

(d) Determine A in such a way that the velocity tends to V_0 at large distances from the origin, and hence obtain

$$\Phi = V_0\left(z + \frac{a^2}{z}\right)$$

as the complex potential in the half-plane y > 0 of a flow, about a circular cylinder, which tends to a uniform flow in the positive x direction at large distances from the origin (see Figure 10.25).

103. For the flow obtained in Problem 102, show that the velocity components are of the form

$$V_x = V_0 \left[1 - \frac{a^2(x^2 - y^2)}{(x^2 + y^2)^2} \right], \qquad V_y = -V_0 \frac{2a^2xy}{(x^2 + y^2)^2}$$

and verify that, because of appropriate symmetry in these expressions, the flow is correctly specified by Φ over the entire z plane.

Section 10.18

104. Suppose that two sources of equal strength m are present at the points $z = \pm a$, where a is real and positive.

(a) Show that the complex potential is given by

$$\Phi = \frac{m}{2\pi} \log (z^2 - a^2),$$

and the complex velocity by

$$V_x - iV_y = \frac{m}{\pi}\frac{z}{z^2 - a^2}.$$

(b) Show (by symmetry or otherwise) that the stream function is constant along the y axis, and deduce that Φ is also the complex potential in the half-plane x > 0 corresponding to a source at z = a and a fixed boundary along the y axis.

(c) Show that the velocity at a point on the boundary x = 0 is given by $V_y = (m/\pi)[y/(y^2 + a^2)]$, and at a point on the line y = 0 by $V_x = (m/\pi)[x/(x^2 - a^2)]$.

105. Suppose that a vortex of strength m is located at z = a, and that an equal and opposite vortex is at z = -a, where a is real and positive.

(a) Show that the complex potential is given by

$$\Phi = \frac{m}{2\pi i} \log \frac{z-a}{z+a},$$

and the complex velocity by

$$V_x - iV_y = \frac{m}{\pi i} \frac{a}{z^2 - a^2}.$$

(b) Show that the vortex at z = -a may be replaced by a fixed boundary along the y axis in part (a), without affecting the flow.

(c) Show that the velocity at a point of the boundary x = 0 is given by $V_y = -(m/\pi)[a/(y^2 + a^2)]$, and that the velocity at a point on the line y = 0 is given by $V_y = (m/\pi)[a/(x^2 - a^2)]$.

106. Fluid is introduced at a steady rate *m* through a small opening at the origin into an infinite channel occupying the region $(-\infty < x < +\infty, 0 \le y \le b)$ of the *z* plane. The flow pattern in the channel is required.

(a) Use the mapping of Problem 98(a) to obtain a transformed problem in which fluid is introduced into the upper half of the w plane at the rate m at the point w = 1, and is withdrawn at the rate m/2 at the points w = 0 and w_{∞} . Noticing that a source of strength 2m then must be present at w = 1, and a sink of strength m at w = 0 (as well as at w_{∞}), since only half the fluid enters the upper

half-plane from a source on its boundary, obtain the complex potential in the w plane in the form

$$\Phi = \frac{m}{\pi} \left[\log (w - 1) - \frac{1}{2} \log w \right] + \text{constant}$$
$$= \frac{m}{\pi} \log (w^{1/2} - w^{-1/2}) + \text{constant.}$$

(b) Obtain the required complex potential in the form

$$\Phi = -\frac{m}{\pi} \log \sinh \frac{\pi z}{2b},$$

neglecting an arbitrary additive constant.

107. (a) Show that the velocity vector in Problem 106 is of the form

$$V_x + iV_y = \frac{m}{2\pi} \coth \frac{\pi(x-iy)}{2b},$$

and that the velocity along the boundary y = b is given by $V_x = (m/2\pi) \tanh(\pi x/2b)$, whereas the velocity along the boundary y = 0 is given by $V_x = (m/2\pi) \coth(\pi x/2b)$.

(b) Show that the stream function is of the form

$$\psi = \frac{m}{\pi} \cot^{-1}\left(\frac{\tanh \pi x/2b}{\tan \pi y/2b}\right),\,$$

and verify that ψ increases by *m* as a small semicircle is traversed about the origin in the counterclockwise direction. Deduce also that the streamlines are the curves

$$\tan\frac{\pi y}{2b}=c\,\tanh\frac{\pi x}{2b}.$$

108. (a) Show that the complex potential

$$\Phi = V_0\left(z + \frac{a^2}{z}\right) + i\frac{m}{2\pi}\log z$$

defines a flow past the circle |z| = a, in which the velocity tends to V_0 as $|z| \rightarrow \infty$, for any constant real value of *m*. Notice that a positive value of *m* corresponds to the presence of a clockwise circulation. (See also Problem 36, Chapter 9.)

(b) By making use of the result of Problem 101(b), show that the drag D and lift L, per unit length, exerted on a cylinder with |z| = a as a cross section is then given by

$$-D + iL = -\frac{1}{2}i\rho \oint_C \left| V_0 \left(1 - \frac{a^2}{z^2} \right) + i \frac{m}{2\pi z} \right|^2 dz,$$

where C is the circle |z| = a.

(c) Show that the integral involved in part (b) can be expressed in the form

$$\int_0^{2\pi} \left[V_0(1-e^{-2i\theta}) + i \frac{m}{2\pi a} e^{-i\theta} \right] \left[V_0(1-e^{-2i\theta}) - i \frac{m}{2\pi a} e^{i\theta} \right] ia e^{i\theta} d\theta.$$

By noticing that all product terms of the form $e^{ir\theta}$ will integrate to zero when r is an integer other than zero, and to 2π when r = 0, obtain the value of the integral in the form $-2V_0m$. Hence deduce that

$$D=0, \qquad L=\rho V_0 m,$$

where *m* is the circulation (positive in the clockwise direction).

Section 10.19

109. (a) Mapping of exterior of ellipse into exterior of circle. Verify that the transformation

$$z=w+\frac{c^2}{4w}$$

maps the exterior of the ellipse

$$\frac{x^2}{a^2} + \frac{y^2}{b^2} = 1$$
 (a > b)

into the exterior of the circle $|w| = \frac{1}{2}(a + b)$, where c is the distance $\sqrt{a^2 - b^2}$ from the center of the ellipse to each focus, in such a way that the mapping is not distorted or rotated at large distances from the ellipse. [Notice that the point $w = \frac{1}{2}(a + b) e^{i\varphi}$ corresponds to the point $z = a \cos \varphi + ib \sin \varphi$.]

(b) Mapping of plane with finite cut into exterior of circle. As a limiting case of part (a) as $b \rightarrow 0$, show that the transformation

$$z = w + \frac{a^2}{4w}$$

maps the z plane with a cut along the real axis from x = -a to x = +a into the exterior of the circle |w| = a/2, with no distortion or rotation at large distances from the cut. (Notice that the point $w = \frac{1}{2}a e^{i\varphi}$ corresponds to the point $z = a \cos \varphi$.)

110. Rotation of mappings of Problem 109.

(a) Suppose that the ellipse defined in Problem 109(a) is rotated about its center through the angle α , where α is positive when the rotation is counterclockwise. By considering the result of replacing z by $e^{-i\alpha}z$ and w by $e^{-i\alpha}w$ in the transformation of that problem, show that the exterior of the resultant ellipse is mapped into the exterior of the circle $|w| = \frac{1}{2}(a + b)$ by the mapping

$$z=w+\frac{c^2\,e^{2i\alpha}}{4w},$$

where again $c^2 = a^2 - b^2$, without distortion or rotation at z_{∞} .

(b) In the limiting case b = 0, show that the transformation

$$z = w + \frac{a^2 e^{2i\alpha}}{4w}$$

maps the z plane, with a cut of length 2a extending from the point $-a e^{i\alpha}$ to the point $a e^{i\alpha}$, into the exterior of the circle |w| = a/2, without distortion or rotation at z_{∞} .

(c) Verify that the point $w = \frac{1}{2}(a + b) e^{i(\alpha + \beta)}$ corresponds to the boundary point $z = e^{i\alpha} (a \cos \beta + ib \sin \beta)$ in the transformation of part (a), and that the point $w = \frac{1}{2}a e^{i(\alpha + \beta)}$ corresponds to the point $z = a e^{i\alpha} \cos \beta$ on the cut in part (b).

111. Show that the complex potential for an ideal fluid flow in the xy plane about a section of an elliptic cylinder, with center at the origin, with semi-major axis of length a along the line $\theta = \alpha$, and with semi-minor axis of length b, which tends to a uniform flow with velocity V_0 in the positive x direction at large distances from the ellipse, is obtained by eliminating w between the two equations

$$\Phi = V_0 \left[w + \frac{(a+b)^2}{4w} \right] + i \frac{m}{2\pi} \log w, \quad z = w + \frac{(a^2 - b^2) e^{2i\alpha}}{4w},$$

where *m* is the clockwise circulation. (Notice that the first equation defines the complex potential for the flow of Problem 108 in the *w* plane about a circle of radius $\frac{1}{2}(a + b)$, and use the result of Problem 110.)

112. By specializing the result of Problem 111 when b = 0 and $\alpha = \pi/2$, deduce that, when no circulation is present, the complex potential for a steady flow about a plane barrier of width 2a, at right angles to the flow and with center at the origin, is of the form

$$\Phi - V_0 \sqrt{z^2 + a^2},$$

where V_0 is the limiting uniform velocity in the x direction. (An alternative derivation of this result is given in Section 10.20).

113. By using the result of Equation (220) of Section 9.14, show that the steadystate temperature distribution in the upper half of the w plane which reduces to unity along the u axis when |u| < 1 and to zero elsewhere along the u axis, is given by

$$T = \frac{1}{\pi} \left[\tan^{-1} \left(\frac{1-u}{v} \right) + \tan^{-1} \left(\frac{1+u}{v} \right) \right] = \frac{1}{\pi} \cot^{-1} \left(\frac{u^2 + v^2 - 1}{2v} \right).$$

(See footnote, page 578.)

114. The steady-state temperature distribution is required in the semi-infinite strip ($0 \le x \le a, 0 \le y < \infty$), subject to the requirements that it reduce to unity along the end y = 0 and to zero along the sides x = 0 and x = a. Use the results of Problems 98(b) and 113 to obtain this distribution in the form

$$T = \frac{1}{\pi} \cot^{-1} \left[\left(\sinh^2 \frac{\pi y}{a} - \sin^2 \frac{\pi x}{a} \right) / 2 \sin \frac{\pi x}{a} \sinh \frac{\pi y}{a} \right] = \frac{2}{\pi} \tan^{-1} \left(\sin \frac{\pi x}{a} / \sinh \frac{\pi y}{a} \right).$$

(Notice that the methods of Chapter 8 would give this solution in the form of an infinite series.)

115. The steady-state temperature is required in the semicircle bounded by the upper half of the circle $x^2 + y^2 = R^2$ and the y axis, subject to the requirements that it reduce to unity along the curved boundary and to zero along the straight

boundary. Use the results of Problems 98(d) and 113 to obtain the distribution in the form

$$T = \frac{1}{\pi} \cot^{-1} \left\{ \frac{\left[\frac{1}{2} \left(\frac{R}{r} - \frac{r}{R}\right)\right]^2 - \sin^2 \theta}{\left(\frac{R}{r} - \frac{r}{R}\right) \sin \theta} \right\} = \frac{2}{\pi} \tan^{-1} \left(\frac{2rR \sin \theta}{R^2 - r^2}\right).$$

Section 10.20

116. Derive the mapping of Problem 98(a).

117. Derive the mapping of Problem 98(b).

118. Obtain the transformation which maps the region in the z plane, above the line y = b when x < 0 and above the x axis when x > 0, into the upper half of the w plane, in the form

$$z = \frac{b}{\pi} [(w^2 - 1)^{1/2} + \cosh^{-1} w],$$

by making the point z = bi correspond to w = -1 and the point z = 0 correspond to w = +1.

119. The infinite strip $(-\infty < x < +\infty, 0 \le y \le b)$ has a semi-infinite cut extending in the positive x direction from the point (0, b/2). It is required to map this cut region into the upper half of the w plane, in such a way that the x axis maps into the segment u < -1, the lower boundary of the cut into -1 < u < 0, the upper boundary of the cut into 0 < u < 1, and the line y = b into the segment u > 1.

(a) Show that the mapping must be such that

$$\frac{dz}{dw} = \frac{Cw}{w^2 - 1}, \qquad z = \frac{C}{2}\log(w^2 - 1) + K.$$

(b) Show that the point z = ib/2 maps into w = 0 if and only if

$$\frac{b}{2}i=\frac{\pi C}{2}i+K,$$

and that the line z = x + ib maps into w = u, where u > 1, if and only if K = ib. Hence obtain the mapping in the form

$$z = ib - \frac{b}{2\pi} \log (w^2 - 1), \qquad w^2 = 1 + e^{-2\pi z/b}.$$

120. Suppose that the temperature is maintained at zero along the boundaries y = 0 and y = b of the strip of Problem 119, and at unity along the semi-infinite cut. Use the results of Problems 119 and 113 to show that the steady-state temperature distribution T(x,y) in the strip is obtained by eliminating u and v from the relations

$$T = \frac{1}{\pi} \cot^{-1} \left(\frac{u^2 + v^2 - 1}{2v} \right),$$
$$u^2 - v^2 = 1 + e^{-2\pi x/b} \cos \frac{2\pi y}{b}, \quad 2uv = -e^{-2\pi x/b} \sin \frac{2\pi y}{b}.$$

121. Electrostatic fields about cylindrical conductors. In a region free of charges, in which an electrostatic field is independent of position normal to the xy plane, the electric field intensity vector E is the negative gradient of an electrostatic potential $\varphi(x,y)$ which satisfies Laplace's equation,

 $\mathbf{E} = -\nabla \varphi$ where $\nabla^2 \varphi = 0$.

(a) Show that the fact that the component of E tangential to a perfect conductor vanishes leads to the conclusion that φ is constant along the boundary of a perfect conductor.

(b) If $\psi(x,y)$ is the function conjugate to $\varphi(x,y)$, so that $\varphi + i\psi$ is an analytic function of x + iy, show that the *lines of force* then are the curves for which $\psi = \text{constant.}$

(c) The surface charge density σ on a conductor is given by $-K \frac{\partial \varphi}{\partial n}$, where K is the electric inductive capacity. Show that the *total charge q* on a cylindrical conductor, per unit length normal to the xy plane, is given by

$$q = \oint_B \sigma \, ds = -K \oint_B \frac{\partial \varphi}{\partial n} \, ds = -K \oint_B \frac{\partial \psi}{\partial s} \, ds = -K \Delta_B \psi,$$

where $\Delta_B \psi$ is the increment in ψ corresponding to a counterclockwise circuit of the boundary *B* of the section of the conductor in the *xy* plane. [See Problem 20(b).]

122. Capacity of a condenser. The capacity C, per unit length, of a condenser formed by two cylindrical conductors B_1 and B_2 , is defined as the absolute value of the ratio of the total charge per unit length on either of the conductors to the potential difference between them. If $\Phi(z) = \varphi + i\psi$ is analytic in the region between the conductors, and if φ is constant on each conductor, show that

$$C = \left|\frac{q}{\varphi_{B_1} - \varphi_{B_2}}\right| = K \left|\frac{\Delta_B \psi}{\varphi_{B_2} - \varphi_{B_1}}\right|.$$

123. By making appropriate use of the function

$$\Phi = \varphi + i\psi = A \log z + B,$$

show that the electrostatic potential outside a *circular* cylindrical conductor is given by

$$\varphi = -\frac{q}{2\pi K} \log r + \text{constant},$$

where r is distance from the axis of the cylinder and q is the charge per unit length on the cylinder, whereas the function ψ is given by

$$\psi = -\frac{q}{2\pi k} \theta + \text{constant},$$

where θ is an angle measured around the axis of the cylinder.

124. Show that the capacity, per unit length, of a condenser formed by concentric *circular* cylinders of radii R_1 and R_2 is given by

$$C = \frac{2\pi K}{\log\left(R_2/R_1\right)}.$$

125. (a) By making use of the results of Problems 109 and 123, show that the result of eliminating w between the relations

$$\Phi = \varphi + i\psi = -\frac{q}{2\pi K}\log w + \text{constant}, \qquad z = w + \frac{c^2}{4w}$$

gives the potential φ and the function ψ for the electrostatic field outside an *elliptical* cylinder with cross section

$$\frac{x^2}{a^2} + \frac{y^2}{b^2} = 1$$
 where $a^2 - b^2 = c^2$,

with charge q per unit length.

(b) Show that the elimination described in part (a) leads to the result

$$\varphi + i\psi = -\frac{q}{2\pi K}\cosh^{-1}\frac{z}{c} + \text{constant.}$$

126. By combining the results of Problems 125 and 124, show that the capacity, per unit length, of a cylindrical condenser whose cross section consists of confocal *ellipses* with focal length c and semi-major axes a_1 and a_2 is given by

$$C = \frac{2\pi k}{\log\left(\frac{a_2 + \sqrt{a_2^2 - c^2}}{a_1 + \sqrt{a_1^2 - c^2}}\right)} \equiv \frac{2\pi K}{\cosh^{-1}\frac{a_2}{c} - \cosh^{-1}\frac{a_1}{c}}$$

[Notice that the ellipse $x^2/a^2 + y^2/(a^2 - c^2) = 1$ corresponds to the circle $|w| = \frac{1}{2}(a + \sqrt{a^2 - c^2})$ in the mapping $z = w + c^2/(4w)$.]

Appendix A

Formulas for Roots of Equations Involving Bessel Functions[†]

1. The nth root, in order of magnitude, of the equation

 $J_{p}(x)=0$

is given by

$$x_n = n\pi \left(1 + \frac{2p-1}{4n}\right) - \frac{c_1}{k} - \frac{4c_2}{3k^3} - \frac{32c_3}{15k^5} - \cdots$$

where

$$k = 2\pi(2p + 4n - 1), \quad m = 4p^2,$$

 $c_1 = m - 1, \quad c_2 = (m - 1)(7m - 31),$
 $c_3 = (m - 1)(83m^2 - 982m + 3779).$

2. The nth root, in order of magnitude, of the equation

$$J_p'(x)=0$$

is given by

$$x_n = n\pi \left(1 + \frac{2p+1}{4n}\right) - \frac{c_1}{k} - \frac{4c_2}{3k^3} - \frac{32c_3}{15k^5} - \cdots$$

where

$$k = 2\pi(2p + 4n + 1),$$
 $m = 4p^2,$
 $c_1 = m + 3,$ $c_2 = 7m^2 + 82m - 9,$
 $c_3 = 83m^3 + 2075m^2 - 3039m + 3537$

3. The nth root, in order of magnitude, of the equation

$$J_{p}(x) Y_{p}(ax) - J_{p}(ax) Y_{p}(x) = 0$$
 (a > 1)

is given by

$$x_n = \frac{n\pi}{a-1} + \frac{c_1}{k} + \frac{c_2 - c_1^2}{k^3} + \frac{c_3 - 4c_1c_2 + 2c_1^3}{k^5} + \cdots$$

[†] See J. McMahon, "On the Roots of the Bessel and Certain Related Functions," Annals of Mathematics, Vol. 9, p. 23 (1895). In certain cases the indicated series may converge slowly for small values of n. where

$$k = \frac{n\pi}{a-1}, \qquad m = 4p^2,$$

$$c_1 = \frac{m-1}{8a}, \qquad c_2 = \frac{4(m-1)(m-25)(a^3-1)}{3(8a)^3(a-1)},$$

$$c_3 = \frac{32(m-1)(m^2-114m+1073)(a^5-1)}{5(8a)^5(a-1)}.$$

4. The nth root, in order of magnitude, of the equation

$$J'_{p}(x) Y'_{p}(ax) - J'_{p}(ax) Y'_{p}(x) = 0$$
 (a > 1)

is given by formula 3, except that here

$$c_{1} = \frac{m+3}{8a}, \qquad c_{2} = \frac{4(m^{2} + 46m - 63)(a^{3} - 1)}{3(8a)^{3}(a - 1)},$$
$$c_{3} = \frac{32(m^{3} + 185m^{2} - 2053m + 1899)(a^{5} - 1)}{5(8a)^{5}(a - 1)}.$$

Appendix B

x	.00	.01	.02	.03	.04	.05	.06	.07	.08	.09
1.0	1.0000	.9943	.9888	.9835	.9784	.9735	.9687	.9642	.9597	.9555
.1	.9514	.9474	.9436	.9399	.9364	.9330	.9298	.9267	.9237	.9209
.2	.9182	.9156	.9131	.9108	.9085	.9064	.9044	.9025	.9007	.8990
.3	.8975	.8960	.8946	.8934	.8922	.8912	.8902	.8893	.8885	.8879
.4	.8873	.8868	.8864	.8860	.8858	.8857	.8856	.8856	.8857	.8859
.5	.8862	.8866	.8870	.8876	.8882	.8889	.8896	.8905	.8914	.8924
.6	.8935	.8947	.8959	.8972	.8986	.9001	.9017	.9033	.9050	.9068
.7	.9086	.9106	.9126	.9147	.9168	.9191	.9214	.9238	.9262	.9288
.8	.9314	.9341	.9368	.9397	.9426	.9456	.9487	.9518	.9551	.9584
.9	.9618	.9652	.9688	.9724	.9761	.9799	.9837	.9877	.9917	.9958

Table I. Values of $\Gamma(x) = (x - 1)!$

Table I	. Zeros	of Bessel	Functions:	$J_{v}(\alpha_{n})$	= 0
---------	---------	-----------	-------------------	---------------------	-----

n	<i>p</i> = 0	p = 1	<i>p</i> = 2	p=3	<i>p</i> = 4	p=5
1	2.405	3.832	5.135	6.379	7.586	8.780
2	5.520	7.016	8.417	9.760	11.064	12.339
3	8.654	10.173	11.620	13.017	14.373	15.700
4	11.792	13.323	14.796	16.224	17.616	18.982
5	14.931	16.470	17.960	19.410	20.827	22.220

Answers to Problems

Chapter 1

1. (a) y' - y = 0. (b) $y' - 2xy = 2 - 4x^2$. (c) $y' - y^2 = 0$. (d) $xy' - y \log y = 0$. (e) yy' + x = 0. (f) $x^2y'^2 = (4y + 1)(xy' - y)$. **2.** (a) y'' - y = 0. (b) y'' - 2y' + y = 0. (c) y'' + y = 0. (d) $yy'' - y'^2 = 0.$ (e) $(y - 1)y'' - 2y'^2 = 0.$ (f) $y''^2 = (1 + y'^2)^2$. 3. (a) $y = C e^{x^2}$. (b) $x - C = \sqrt{1 - y^2}, y = \pm 1$. (c) (y-1)(x-1) = C(y+1)(x+1). (d) $\sqrt{1-y^2} - \sqrt{1-x^2} = C, x = \pm 1, y = \pm 1.$ 13. (a) $y = x^2/(2-k) + Cx^k$ if $k \neq 2$; $y = x^2 \log x + Cx^2$ if k = 2. (b) $y = x \tan x + 1 + C \sec x$. (c) $y = \sin x + C \cos x$. (d) $y = 2 \sin x - 2 + C e^{-\sin x}$. (e) y = (x + 1)(x + C)/(x - 1). (f) $y = (Cx - 1)/(x \log x)$. (g) $y^2 = x(x + C)$. (h) x = y(y + C). 17. (a) $y = c_1 e^{-x} + c_2 e^{2x}$. (b) $y = c_1 e^{-x} + (c_2 + c_3 x) e^{x}$. (c) $y = e^{x}(c_1 \cos x + c_2 \sin x)$. (d) $y = e^{x}(c_1 + c_2x + c_3\cos x + c_4\sin x)$. (e) $y = c_1 e^x + e^{-x/2} [c_1 \cos(\sqrt{3x/2}) + c_2 \sin(\sqrt{3x/2})].$ (f) $y = c_1 e^{(1+i)x} + c_2 e^{-(1+i)x}$.

18. (a) $y = c_1 e^{kx} + c_2 e^{-kx} + c_3 \cos kx + c_4 \sin kx$. (b) $y = e^{kx}(c_1 \cos kx + c_2 \sin kx) + e^{-kx}(c_3 \cos kx + c_4 \sin kx)$. (c) $y = e^{kx}(c_1 + c_2x) + e^{-kx}(c_3 + c_4x)$. 19. (a) $y = c_1 \cos kx + c_2 \sin kx + (\sin x)/(k^2 - 1)$ if $k^2 \neq 0, 1$. (b) $y = c_1 \cos x + c_2 \sin x - \frac{1}{2}x \cos x$. (c) $y = c_1 e^x + c_2 e^{-x} - \frac{1}{2} \sin x$. (d) $y = c_1 e^x + c_2 e^{-x} + \frac{1}{2} x e^x$. (e) $y = c_1 e^x + c_2 e^{-x} + \frac{1}{4}(x^2 - x) e^x$. (f) $y = e^{x}(c_1 \cos x + c_2 \sin x) + \frac{1}{5}(2 \cos x + \sin x)$. (g) $y = e^{x}(c_1 \cos x + c_2 \sin x) - \frac{1}{2}x e^{x} \cos x$. (h) $y = c_1 e^{4x} + c_2 e^{5x} + (2x^2 + 6x + 7) e^{3x}$. (i) $y = c_1 e^{2x} + c_2 e^{-x} - x e^{-x} - 3 \sin x + \cos x + 2x - 1$. 20. (a) $y = c_1 x^k + c_2 x^{-k}$ if $k \neq 0$; $y = c_1 + c_2 \log x$ if k = 0. (b) $y = x(c_1 \cos \log x + c_2 \sin \log x)$. (c) $y = c_1 x^2 + c_2 x^{-1}$. (d) $y = x(c_1 + c_2 \log x)$. (e) $y = c_1 x^{-1} + x(c_2 + c_3 \log x)$. (f) $y = c_1 x^n + c_2 x^{-n-1}$. (g) $y = c_1 x + c_2 x^{-1} + \frac{1}{3} x^2$. (h) $y = c_1 x + c_2 x^{-1} + \frac{1}{2} x \log x$. (i) $y = c_1 x^3 + c_3 x^2 + 3x + 2$. **26.** $x = c_1 t + c_2, y = -\frac{1}{2}c_1 t^2 - (c_1 + c_2)t + c_3$. 27. (a) $x = c_1 e^{2t} + c_2 e^t$, $y = -c_1 e^{2t} + c_2 e^t$. (b) $x = c_1 e^t + c_2 \cos t + c_3 \sin t$, $y = 2(c_1 - 1) e^t + (c_2 - c_3) \cos t + (c_2 + c_3) \sin t.$ (c) $x = e^{t}(c_1 + c_2 t) + c_3 e^{-3t/2} - \frac{1}{2}t$. $y = e^{t}(6c_2 - 2c_1 - 2c_2t) - \frac{1}{3}c_3 e^{-3t/2} - \frac{1}{3}.$ (d) $x = c_1 \cos \omega_1 kt + c_2 \sin \omega_1 kt + c_3 \cos \omega_2 kt + c_4 \sin \omega_2 kt$, $y = \sqrt{2}(c_1 \cos \omega_1 kt + c_2 \sin \omega_1 kt - c_3 \cos \omega_2 kt - c_4 \sin \omega_2 kt),$ where $\omega_1 = \sqrt{2 + \sqrt{2}}$ and $\omega_2 = \sqrt{2 - \sqrt{2}}$. (e) $x = e^{2t} + 1$, $y = e^{2t} - 1$. (f) $x = c_1 e^{-2t}, y = 5c_1 e^{-2t} + c_2 e^{-t}$. (g) $x = c_1 e^{2t}$, $y = \frac{3}{4}c_1 e^{2t} + c_2 e^{-2t}$, $z = -\frac{3}{2}c_1 e^{2t} - \frac{2}{5}c_2 e^{-2t} + c_3 e^{3t}$. **28.** (a) $x = c_1 t + c_2 t^2$, $y = c_1 t - c_2 t^2$. (b) $x = c_1 t^2$, $y = \frac{3}{4} c_1 t^2 + c_2 t^{-2}$, $z = -\frac{3}{2} c_1 t^2 - \frac{2}{5} c_2 t^{-2} + c_3 t^3$. **29.** (a) $py = \int ph \, dx + c$ where $p = e^{\int a_1 dx}$. (b) $y = c_1 \cos x + c_2 \sin x + (\sin x) \log (\tan \frac{1}{2}x)$. (c) $y = c_1 \cos x + c_2 \sin x + (\cos x) \log (\cos x) + x \sin x$. (d) $y = c_1 \cos x + c_2 \sin x + \int_x^x \sin (x - \xi) \log \xi d\xi$. (e) $y = e^{x}[c_1 \cos x + c_2 \sin x - (\cos x) \log (\sec x + \tan x)].$ (f) $y = c_1 x^3 + c_2 x^2 + x^3 \int \frac{\sin x}{x^2} dx - x^2 \int \frac{\sin x}{x} dx.$ (g) $y = c_1 x + c_2 x^2 - x \log x - \frac{1}{2} x (\log x)^2$. 31. $y = c_1 x + c_2 e^x + 1$. 32. $y = c_1(1 + x \tan x) + c_2 \tan x$. **33.** $y = c_1 x + c_2 \left(\frac{x}{2} \log \left| \frac{1+x}{1-x} \right| - 1 \right) + \frac{3}{2} (x^2 + 1).$

622

- 42. $u_1 = \frac{1}{2}(\cosh x + \cos x), u_2 = \frac{1}{2}(\sinh x + \sin x),$ $u_3 = \frac{1}{2}(\cosh x - \cos x), u_4 = \frac{1}{2}(\sinh x - \sin x).$ 43. (a) $k = n\pi/a$, *n* a nonzero integer. (b) $k \neq m\pi/a$, m an odd integer. 44. (a) $x^2y^3 + xy - y^2 = c$. (b) $x^3y - xy^3 = c$. (c) $y e^{x} + x e^{y} = c$. **45.** (a) $y^2 + 2xy - x^2 = c$. (b) $y^2 + xy = cx$. (c) $y + \sqrt{x^2 + y^2} = c$. (d) $y = x \sin^{-1} (cx)$. 46. (a) $xy = e^{cx}$. (b) $x^3 - 3xy = c$. (c) $y^3 - 3xy = c$. (d) $(y-2)^2 + 2(x-1)(y-2) - (x-1)^2 = c$. (e) xy(x + c) = 1. (f) $(x^2 + 1)^2(2y^2 - 1) = c$. 47. (a) $y = \log(x + c_1) + c_2$. (b) $y = c_1 \tan^{-1} (c_1 x) + c_2$. (c) $y^2 = c_1 x + c_2$. (d) $y = c_2 - [1 - (x - c_1)^2]^{1/2}$.
 - (e) $y = c_1 \cos x + c_2 \sin x$.

(f)
$$x = 2p^3 + p + c_1, y = \frac{3}{2}p^4 + \frac{1}{2}p^2 + c_2.$$

Chapter 2

3. (a)
$$(s - a)/[(s - a)^2 + k^2]$$
.
(b) $n!/(s + a)^{n+1}$.
(c) $(1 + e^{-s\pi})/(s^2 + 1)$.
(d) $(e^{-sa} - e^{-sb})/s$.
4. (a) $6/s^4$. (b) $2/(s + 3)^3$. (c) $a(s^2 - 2a^2)/(s^4 + 4a^4)$.
(d) $4(s - 1)/(s^2 - 2s + 5)^2$. (e) $(6as^2 - 2a^3)/(s^2 + a^2)^3$
(f) $(s - a)/[(s - a)^2 - b^2]$.
5. (a) $s^3 \tilde{f}(s) - s^2 f(0) - sf'(0) - f''(0)$. (b) $-\tilde{f}'(s - 1)$.
(c) $\sum_{n=0}^{N} (n! a_n)/s^{n+1}$. (d) $\sum_{n=0}^{N} a_n s/(s^2 + n^2)$.

10. $(\tanh \frac{1}{4}as)/s^2$.

18. (a)
$$1 - \frac{t^2}{2! \, 3!} + \frac{t^4}{4! \, 5!} - \frac{t^6}{6! \, 7!} + \cdots$$

(b) $1 - \frac{(t/2)^2}{(1!)^2} + \frac{(t/2)^4}{(2!)^2} - \frac{(t/2)^6}{(3!)^2} + \cdots$
22. (a) $(1 - \cos at)/a$. (b) $(e^{at} - 1 - at)/a^2$. (c) $(e^{at} - e^{bt})/(a - b)$.

(d)
$$(b \sin at - a \sin bt)/(b^2 - a^2)$$
.

28. (a) $e^{2t} - e^{t}$. (b) $e^{t} \cos 2t + \frac{1}{2}e^{t} \sin 2t$. (c) $\sin T \sinh T + (\sin T \cosh T - \cos T \sinh T)/\sqrt{2}$ where $T = t/\sqrt{2}$.

(d)
$$(1 + 2e^{-t} - 3e^{-3t})/2$$
. (e) $t - \sin t$.
(f) $e^{-(t-1)}$ when $t > 1$; 0 when $0 \le t < 1$.
29. (a) $(\sin at + at \cos at)/(2a)$. (b) $(\sinh at + at \cosh at)/(2a)$.
(c) $-e^{-t} + 5e^{-3t} - 4e^{-3t}$.
(d) $[e^{-at} - e^{at}/(2\cos \frac{1}{2}\sqrt{3} at - \sqrt{3} \sin \frac{1}{2}\sqrt{3} at)]/(3a^{2})$.
(e) 1 when $0 \le t < 1$; 0 when $t > 1$.
(f) $(b \sin at - a \sin bt)/(ab(b^{2} - a^{2})]$ if $b^{2} \ne a^{2}$; $(\sin at - at \cos at)/(2a^{2})$ if $b = a$.
30. (a) $e^{-t} \cos t$. (b) $(\sin at \cosh at + \cos at \sinh at)/(2a)$.
(c) 0 when $0 \le t \le \pi$; $-\sin t$ when $t \ge \pi$. (d) $\frac{1}{2}(2 - 4at + a^{2}t^{2})e^{-at}$.
(e) $\frac{1}{4}[(3 - t^{2})\sin t - 3t \cos t]$. (f) $\frac{1}{4}[(3 + t^{2})\sinh t - 3t \cosh t]$.
31. (a) $y = e^{-kt}$. (b) $y = (1 - e^{-kt})/k$.
(c) $y = e^{-kt}$ when $0 \le t < 1$; $y = (1 + e^{k})e^{-kt}$ when $t > 1$.
(d) $y = e^{-kt} \left[y_{0} + \int_{0}^{t} f(u)e^{4u} du \right]$.
32. (a) $y = e^{-t}\cos t$. (b) $y = 1 - e^{-t}\cos t$.
(c) $y = e^{-t}\cos t + (b - 0) \le t \le 1$;
 $y = e^{-t}\cos t + e^{-(t-1)}\sin (t - 1)$ when $t \ge 1$.
(d) $y = e^{-t} \left[y_{0}\cos t + (y_{0} + y_{0})\sin t + \int_{0}^{t} f(u)e^{u}\sin (t - u) du \right]$.
33. (a) $y = \frac{1}{2}(\sin t \cosh t - \cos t \sinh t)$.
(b) $y = 1 - \cos t \cosh t$.
34. $y = -[\sin t \sin (b - a)]/(\sin b)$ when $0 \le t \le a$;
 $y = -[\sin t \sin (b - a)]/(\sin b)$ when $0 \le t \le a$;
 $y = -[\sin a \sin (b - t)]/(\sin b)$ when $a \le t \le b$.
35. (b) $x = \frac{f_{0}}{k} \left[1 - e^{-\alpha t} \left(\cos \beta t + \frac{\alpha}{\beta} \sin \beta t \right) \right] (\beta^{2} = \omega_{0}^{2} - \alpha^{2} > 0)$,
 $x = \frac{f_{0}}{k} \left[1 - \frac{\alpha + \gamma}{2\gamma} e^{-(\alpha - \gamma)t} + \frac{\alpha - \gamma}{2\gamma} e^{-(\alpha + \gamma)t} \right] (\gamma^{2} = \alpha^{2} - \omega_{0}^{2} > 0)$.
(c) $x = a e^{-\alpha t} (\cos \beta t + \frac{\alpha}{\beta} \sin \beta t) (\beta^{2} = \omega_{0}^{2} - \alpha^{2} > 0)$,
 $x = a e^{-\alpha t} (1 + \alpha t) (\alpha = \omega_{0})$,
 $x = a \left[\frac{\alpha + \gamma}{2\gamma} e^{-(\alpha - \gamma)t} - \frac{\alpha - \gamma}{2\gamma} e^{-(\alpha + \gamma)t} \right] (\gamma^{2} = \alpha^{2} - \omega_{0}^{2} > 0)$.
36. $x = -e^{-(\cos t} + \sin t)$, $y = e^{-t}(1 + \sin t)$.
38. (a) $\sqrt{\pi}/2$. (b) 1.54. (c) 3.33. (d) 1.43. (e) -4.33. (f) $3\sqrt{\pi}/4$.
41. (a) 0 when $\alpha > 0$; 1 when $\alpha = 0$; no finite limit when $\alpha < 0$.
(b) $1/\sqrt{\pi}$.

Chapter 3

- 14. $y(0.1) \approx 0.9949$, $y(0.2) \approx 0.9785$.
- 17. Rounded true values: (a) 1.5527; (b) 1.5841; (c) 0.07842; (d) 1.6082.
- **19.** Rounded true values: (a) 0 3152; (b) 0.3121.
- **24.** Rounded true values: (a) 1 4049; (b) 1.4333; (c) 0.07096; (d) 0.07024; (e) 2.8394; (f) 5.6856.

Chapter 4

1. (a) All values of x. (b) -3 < x < 1. (c) $1 - \sqrt{3} \le x \le 1 + \sqrt{3}$. (d) -1 < x < 1; also x = -1 if k < 1 and x = +1 if k < 0. (e) All values of x. (f) x = 0. (g) a-1 < x < a+1; also x = a+1 if $\alpha > 1$ and x = a-1 if $\alpha > 0$. (h) -4 < x < 4. (i) x > 1 and $x \le -1$. (j) x > -1. 3. (a) $y = c_1 \left(1 + \frac{x^3}{2 \cdot 3} + \frac{x^6}{2 \cdot 3 \cdot 5 \cdot 6} + \cdots \right) + c_2 \left(x + \frac{x^4}{3 \cdot 4} + \frac{x^7}{3 \cdot 4 \cdot 6 \cdot 7} + \cdots \right).$ (b) $y = c_1 \left(1 + \frac{x^2}{2!} - \frac{x^4}{4!} + \frac{1 \cdot 3x^6}{6!} - \frac{1 \cdot 3 \cdot 5x^8}{8!} + \cdots \right) + c_2 x.$ (c) $y = c_1 \left(1 + \frac{x^4}{2!} + \frac{x^8}{4!} + \cdots \right) + c_2 \left(x^2 + \frac{x^6}{3!} + \frac{x^{10}}{5!} + \cdots \right)$ $= c_1 \cosh x^2 + c_2 \sinh x^2.$ 4. (a) $y = A_0 \left(1 - \frac{x^3}{2!} + \frac{x^4}{4!} - \cdots \right) + A_1 \left(x - \frac{x^3}{3!} + \frac{x^5}{5!} - \cdots \right).$ (b) $y = A_0 \left(1 - \frac{3}{2}x^2 + \frac{1}{6}x^3 + \cdots \right) + A_1 \left(x - \frac{1}{2}x^3 + \frac{1}{12}x^4 + \cdots \right).$ (c) $y = A_0 \left(1 + \frac{1}{2} x^2 \right) + A_1 x.$ (d) $y = A_0 \left(1 + \frac{1}{2}x + \frac{1}{2}x^2 + \cdots \right).$ (e) $y = A_0 \left(1 + \frac{x}{1!} + \frac{x^2}{2!} + \cdots \right)$ (f) $y = A_0$.

5. (a) No singular points. (b) No singular points.

- (c) Regular singular points at $x = \pm \sqrt{2}$.
- (d) Irregular singular point at x = 0.
- (e) Regular singular points at x = 0 and x = -1.
- (f) Irregular singular point at x = 0.

b)
$$y = c_1 x^{-1/2} \left(1 - \frac{x}{2!} + \frac{x}{4!} - \cdots \right) + c_2 x^{1/2} \left(1 - \frac{x}{2!} + \frac{x}{4!} - \cdots \right) + c_2 x^{1/2} \left(1 - \frac{x}{2!} + \frac{x}{4!} - \frac{x}{2!} + \frac{x}{4!} - \frac{x}{2!} \right)$$

(c)
$$y = c_1 x^{-1} \left(1 - \frac{x^2}{2!} + \frac{x^4}{4!} + \cdots \right) + c_2 \left(1 - \frac{x^2}{3!} + \frac{x^4}{5!} - \cdots \right)$$

= $x^{-1} (c_1 \cos x + c_2 \sin x).$

$$\begin{aligned} & \text{(d) } y = c_1(1+x+x^3) + c_2x^3(1+x+x^3+x^3+\cdots) \\ &= c_1(1+x+x^3) + c_2x^3(1-x) \quad (|x|<1) \\ &= (C_1+C_2x^3)(1-x) \quad (|x|<1). \end{aligned}$$
7. (a) $y = c_1\left(1+\frac{x^3}{2!}+\frac{x^4}{4!}+\cdots\right) + c_2x^3\left(1+\frac{x^3}{3!}+\frac{x^4}{5!}+\cdots\right).$
(b) $y = c_1\left(1+\frac{x^3}{2!}+\frac{x^4}{4!}-\cdots\right) + c_2\left(x^3-\frac{x^4}{3!}+\frac{x^{10}}{5!}-\cdots\right).$
(c) $y = c_1\left(1-\frac{x^4}{2!}+\frac{x^4}{4!}-\cdots\right) + c_2\left(x^3-\frac{x^4}{3!}+\frac{x^{10}}{5!}-\cdots\right).$
(d) $y = c_1\left(x-\frac{1}{6}x^3+\cdots\right) + c_2\left(x^3-\frac{1}{12}x^4+\cdots\right).$
8. $M(a,c;x) = 1 + \sum_{k=1}^{\infty} \frac{a(a+1)(a+2)\cdots(a+k-1)x^k}{c(c+1)(c+2)\cdots(c+k-1)k!}$
 $&= 1+\frac{a}{c}x+\frac{a(a+1)x^2}{c(c+1)(c+2)\cdots(c+k-1)k!}$
 $&= 1+\frac{a}{c}x+\frac{a(a+1)x^3}{c(c+1)(c+2)}\frac{x^3}{2!}+\cdots.$
9. (b) $u_1(x) = \sum_{k=0}^{\infty} \frac{x^k}{(k!)^3}.$
(d) $v(x) = -\left(2x+\frac{3}{4}x^3+\frac{11}{108}x^3+\cdots\right)$
 $&= -2\left[\frac{1}{(1!)^3}x+\frac{1+\frac{1}{2}}{1!2!}\frac{x}{x^3}+\frac{1+\frac{1}{2}+\frac{1}{2}}{(3!)^3}x^3+\cdots\right].$
10. (b) $u_1(x) = \sum_{k=1}^{\infty} \frac{x^k}{(k-1)!k!}.$
(d) $v(x) = 1 - \left[\frac{1}{0!1!}x+\frac{2(1)+\frac{1}{2}}{1!2!}x^3+\frac{2(1+\frac{1}{2})+\frac{1}{3}}{2!3!}x^3+\cdots\right].$
12. (a) $y = \frac{x^3}{2!}+\frac{x^3}{3!}+\frac{x^6}{6!}+\frac{x^7}{7!}+\cdots.$
(b) $y = \frac{x^3}{2!}+\frac{x^3}{3!}+\frac{x^4}{3!}+\frac{x^4}{3!}+\frac{2x^4}{5!}+\frac{x^4}{5!}+\cdots.$
(c) $y = \frac{x^3}{2(1!)^2}+\frac{x^4}{3(2!)^2}+\frac{x^4}{4(3!)^2}+\frac{x^4}{5!}+\cdots$
(d) $y = x^{-1/2}\left(\frac{4}{7}+\frac{4x}{3}+\frac{2x^2}{7}+\frac{2x^3}{57}+\cdots\right).$
15. $y = c_1u_1(x) + c_2u_1(x) \log x - v(x)],$
where $u_1(x) = 1 + x + \frac{x^2}{2!}+\frac{x^4}{3!}+\cdots = e^x,$
 $v(x) = \sum_{k=1}^{\infty} \frac{\left(1+\frac{1}{2}+\cdots+\frac{1}{k}\right)}{k!}x^4 = x+\frac{3}{4}x^2+\frac{11}{36}x^3+\cdots.$

16. $y = c_1 u_1(x) + c_2[u_1(x) \log x + 1 - v(x)],$ where $u_1(x) = x\left(1 + x + \frac{x^3}{2!} + \frac{x^3}{3!} + \cdots\right) = xe^x$, $v(x) = \sum_{k=1}^{\infty} \frac{\left(1 + \frac{1}{2} + \cdots + \frac{1}{k-1}\right)}{(k-1)!} x^{k} = x^{2} + \frac{3}{4}x^{3} + \frac{11}{36}x^{4} + \cdots$ **19.** (a) 0.1483. (b) -1.081. (c) 0.1924. (d) 0.1357. (e) 0.9900 - 1.081i. (f) 0.4539. **20.** $x = c_1 J_0(t) + c_2 Y_0(t), y = c_1 t J_1(t) + c_2 t Y_1(t).$ 21. $y = c_1 J_0(Ax + B) + c_2 Y_0(Ax + B)$. **22.** (a) $\sqrt[3]{2}/\Gamma(\frac{2}{3})$. (b) $-2/\pi$. (c) 2. (d) $1/(2^n n!)$. (e) $2/\pi$. 31. (a) $y = x^2[c_1 J_2(x) + c_2 Y_2(x)].$ (b) $y = x[c_1 J_{1/2}(x^2) + c_2 J_{-1/2}(x^2)] = C_1 \cos x^2 + C_2 \sin x^2$. (c) $y = c_1 I_{1/2}(x) + c_2 I_{-1/2}(x) = x^{-1/2}(C_1 \cosh x + C_2 \sinh x)$. (d) $y = e^{-x}[c_1 J_0(x) + c_2 Y_0(x)].$ (e) $y = x[c_1 I_1(x) + c_2 K_1(x)].$ (f) $y = c_1 x \cos(a/x) + c_2 x \sin(a/x)$. (g) $y = c_1 x^{1/2} I_{1/4}(x^2/2) + c_2 x^{1/2} I_{-1/4}(x^2/2).$ (h) $y = c_1 e^{-x} I_0(x) + c_2 e^{-x} K_0(x)$. (i) $y = c_1 e^{-x^2} J_0(x) + c_2 e^{-x^2} Y_0(x)$. 32. $y = x^{-1} J_1(2x)$. 33. $y = c_1 J_1(x) - \frac{1}{4}P Y_1(x)$. 34. (a) $y = x^p Z_p(x/L)$, where p = (1 - n)/2. (b) $y = x^{ps} Z_p(x^s/Ls)$, where p = (1 - n)/(2 - n) and s = (2 - n)/2, $n \neq 2$. (c) $y = x^{-1/2}(c_1 + c_2 \log x)$. 35. (a) 0.9844. (b) 0.2496. (d) 0.2483. (c) 1.016. **39.** $U = r^n [c_1 P_n(\cos \theta) + c_2 Q_n(\cos \theta)].$ **47.** (a) $y = c_1 + c_2 \left(1 + \frac{1}{1!x} + \frac{1}{2!x^2} + \cdots \right) = c_1 + c_2 e^{1/x}$. (b) $y = c_1 \left(1 - \frac{1}{2! x^2} + \frac{1}{4! x^4} - \cdots \right) + c_2 \left(\frac{1}{x} - \frac{1}{3! x^3} + \frac{1}{5! x^5} - \cdots \right)$ $=c_1\cos\frac{1}{n}+c_2\sin\frac{1}{n}.$ **48.** $y = c_1 \left(1 - \frac{1}{3! x^2} + \frac{1}{5! x^4} - \cdots \right) + c_2 \left(x - \frac{1}{2! x} + \frac{1}{4! x^3} - \cdots \right)$ $= c_1 x \sin \frac{1}{z} + c_2 x \cos \frac{1}{z}.$

Chapter 5

2.
$$k = n\pi/L$$
, where *n* is any integer other than zero.
4. $y = C J_0(\lambda^{1/2}x)$ where $J_0(\lambda^{1/2}a) = 0$.
8. $\omega_n = \sqrt{\frac{EI}{\rho}} \frac{\mu_n^2}{L^2}$, where $\cos \mu_n = \operatorname{sech} \mu_n$;
 $\varphi_n = \frac{\sinh \mu_n(x/L) - \sin \mu_n(x/L)}{\sinh \mu_n - \sin \mu_n} - \frac{\cosh \mu_n(x/L) - \cos \mu_n(x/L)}{\cosh \mu_n - \cos \mu_n}$

9.
$$\omega_{n} = \sqrt{\frac{Et}{p}} \frac{\mu_{n}^{k}}{L^{k}};$$
Symmetrical modes: $\tan(\mu_{n}/2) = -\tanh(\mu_{n}/2),$
 $\varphi_{n} = \frac{\cosh(\mu_{n}/2)}{\cosh(\mu_{n}/2)} - \frac{\cos(\mu_{n}/2)}{\cos(\mu_{n}/2)}.$
Antisymmetrical modes: $\tan(\mu_{n}/2) = +\tanh(\mu_{n}/2),$
 $\varphi_{n} = \frac{\sinh(\mu_{n}/2)}{\sinh(\mu_{n}/2)} - \frac{\sin(\mu_{n}/2)}{\sin(\mu_{n}/2)}.$
17. Exact value rounds to 6 55.
20. (a) $(x^{2}y')' + (x^{2} + \lambda x)y = 0.$
(b) $(y'\sin x)' + \lambda y\sin x = 0.$
(c) $(e'e''y')' + e''(e^{-1} + \lambda y) = 0.$
(d) $(x^{c}e^{-x}y')' + e''(e^{-1} + \lambda y) = 0.$
25. (a) $A_{n} = 2(1 - \cos(\pi)/(\pi).$
(b) $A_{n} = (-1)^{n+1}2/\pi.$
(c) $A_{n} = \left(1 - \cos(\frac{\pi\pi}{2})\right) / \left(\frac{n\pi}{2}\right).$
27. $1 = \sum_{n=1}^{\infty} A_{n} \sin(\mu_{n}\frac{x}{L}) (0 < x < L),$ where $\tan(\mu_{n}) = -\alpha\mu_{n};$
 $A_{n} = \frac{2(1 - \cos(\mu_{n})}{\mu_{n} - \frac{1}{2}\sin(2\mu_{n})} = \frac{2(1 - \cos(\mu_{n})}{\mu_{n}(1 + \alpha\cos^{2}\mu_{n})}.$
34. (a) $\frac{8L^{s}}{\pi^{s}} \left(\frac{1}{1^{s}}\sin(\frac{\pi x}{L}) + \frac{1}{3^{s}}\sin(\frac{3\pi x}{L}) + \frac{1}{5^{s}}\sin(\frac{5\pi x}{L}) + \cdots \right).$
(c) $\frac{2}{\pi} \left(\sin(\frac{\pi x}{L}) + \sin(\frac{2\pi x}{L}) + \frac{1}{3}\sin(\frac{3\pi x}{L}) + \frac{1}{5^{s}}\sin(\frac{5\pi x}{L}) - \cdots \right).$
(d) $\frac{\pi}{\pi} \sum_{n=1}^{\infty} \frac{1 - \cos(n\pi/2)}{n^{s} - 1}\sin(\frac{\pi x}{L}) + \frac{1}{3^{s}}\sin(\frac{\pi x}{L}) + \frac{1}{2}\sin(\frac{\pi x}{L}) + \frac{4}{\pi} \sum_{m=1}^{\infty} \frac{(-1)^{m+1}m}{4m^{s} - 1}\sin(\frac{2\pi m x}{L}).$
(c) $\frac{1}{\pi} \left(\frac{1}{2}\cos(\frac{2\pi x}{L}) + \frac{1}{4^{s}}\cos(\frac{4\pi x}{L}) + \frac{1}{6^{s}}\cos(\frac{5\pi x}{L}) + \cdots \right).$
(d) $\frac{\mu}{\pi} \frac{L^{s}}{(\frac{1}{2^{s}}\cos(\frac{2\pi x}{L}) + \frac{1}{4^{s}}\cos(\frac{\pi x}{L})} \sin(\frac{\pi x}{L}) + \frac{1}{6^{s}}\cos(\frac{5\pi x}{L}) + \cdots).$
(b) $\frac{L}{4} - \frac{8L^{s}}{(\frac{1}{2^{s}}\cos(\frac{2\pi x}{L}) + \frac{1}{6^{s}}\cos(\frac{5\pi x}{L}) + \frac{1}{10^{s}}\cos(\frac{1}{2}m + \cdots)}.$
(c) $\frac{1}{2} + \frac{2}{\pi} \left(\frac{1}{1^{s}}\cos(\frac{\pi x}{L}) + \frac{1}{3^{s}}\cos(\frac{\pi x}{L}) + \frac{1}{3^{s}}\cos(\frac{5\pi x}{L}) + \cdots \right).$

$$(d) \frac{2}{\pi} \left[1 - 2 \sum_{n=1}^{\infty} \frac{1}{4n^2 - 1} \cos \frac{n\pi x}{L} \right].$$

$$(e) \frac{1}{\pi} \left[1 + \cos \frac{\pi x}{L} + 2 \sum_{n=2}^{\infty} \frac{n \sin (\pi n/2) - 1}{n^2 - 1} \cos \frac{n\pi x}{L} \right].$$

$$(f) \frac{1}{L} \left[1 + \frac{2L}{\pi} \sum_{n=1}^{\infty} \frac{\sin (n\pi n/2)}{n\epsilon} \cos \frac{n\pi x}{L} \right].$$

$$40. (a) \frac{L^2}{12} - \frac{2L^2}{\pi^2} \left(\frac{1}{2^2} \cos \frac{2\pi x}{L} + \frac{1}{4^2} \cos \frac{4\pi x}{L} + \frac{1}{6^4} \cos \frac{6\pi x}{L} + \cdots \right).$$

$$+ \frac{4L^4}{\pi^2} \left(\frac{1}{1^2} \sin \frac{\pi x}{L} + \frac{1}{3^3} \sin \frac{3\pi x}{L} + \frac{1}{5^3} \sin \frac{5\pi x}{L} + \cdots \right).$$

$$+ \frac{4L^4}{\pi^2} \left(\frac{1}{1^2} \sin \frac{\pi x}{L} + \frac{1}{3^3} \sin \frac{3\pi x}{L} + \frac{1}{5^3} \sin \frac{5\pi x}{L} + \cdots \right).$$

$$(b) \frac{L}{8} - \frac{4L}{\pi^2} \left(\frac{1}{2^2} \cos \frac{2\pi x}{L} + \frac{1}{6^4} \cos \frac{6\pi x}{L} + \frac{1}{10^4} \cos \frac{10\pi x}{L} + \cdots \right)$$

$$+ \frac{2L}{\pi^2} \left(\frac{1}{1^3} \sin \frac{\pi x}{L} - \frac{1}{3^2} \sin \frac{5\pi x}{L} + \cdots \right).$$

$$(c) \frac{3}{4} + \frac{1}{\pi} \left(\frac{1}{1} \cos \frac{\pi x}{L} - \frac{1}{3} \cos \frac{3\pi x}{L} + \frac{1}{5} \cos \frac{5\pi x}{L} - \cdots \right)$$

$$- \frac{1}{\pi} \left(\sin \frac{\pi x}{L} - \sin \frac{2\pi x}{L} + \frac{1}{3} \sin \frac{3\pi x}{L} + \frac{1}{5^3} \sin \frac{5\pi x}{L} - \cdots \right).$$

$$(d) \frac{8}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^{n+1}n}{4n^4 - 1} \sin \frac{\pi n x}{L} .$$

$$(e) \frac{1}{2\pi} \left(1 + \cos \frac{\pi x}{L} + \frac{\pi}{2} \sin \frac{\pi x}{L} \right)$$

$$+ \frac{1}{\pi} \sum_{n=2}^{\infty} \frac{1}{n^4 - 1} \left[\left(n \sin \frac{n\pi x}{L} - 1 \right) \cos \frac{n\pi x}{L} - n \cos \frac{n\pi}{2} \sin \frac{n\pi x}{L} \right].$$

$$(f) \frac{1}{2L} + \frac{1}{\pi \epsilon} \sum_{n=1}^{\infty} \frac{1}{n} \left[\sin \frac{\pi \pi c}{L} \cos \frac{n\pi x}{L} + \frac{1}{2} \sin \frac{4\pi x}{R} + \frac{1}{3} \sin \frac{6\pi x}{R} + \cdots \right).$$

$$(b) \frac{1}{2} + \frac{2}{\pi} \left(\frac{1}{1} \sin \pi x + \frac{1}{3} \sin 3\pi x + \frac{1}{5} \sin 5\pi x + \cdots \right).$$

$$(b) \frac{1}{2} + \frac{2}{\pi} \left(\frac{1}{1} \sin \pi x + \frac{1}{3} \sin 3\pi x + \frac{1}{3} \sin 5\pi x + \cdots \right).$$

$$(c) \frac{2}{\pi} - \frac{4}{\pi} \left(\frac{1}{3} \cos 2x + \frac{1}{15} \cos 4x + \cdots + \frac{1}{4n^4 - 1} \cos 2nx + \cdots \right).$$

46.
$$x^{p} = 2L^{p} \sum_{n=1}^{\infty} \frac{J_{p}\left(\alpha_{n} \frac{x}{L}\right)}{\alpha_{n} J_{p+1}(\alpha_{n})}$$
, where $J_{p}(\alpha_{n}) = 0$.
47. $x^{p} = 2L^{p} \sum_{n=1}^{\infty} \frac{\mu_{n} L J_{p+1}(\mu_{n} L)}{(\mu_{n}^{2} L^{2} - p^{2})[J_{p}(\mu_{n} L)]^{2}} J_{p}(\mu_{n} x)$
 $= 2L^{p} \sum_{n=1}^{\infty} \frac{p}{\mu_{n}^{2} L^{2} - p^{2}} \frac{J_{p}(\mu_{n} x)}{J_{p}(\mu_{n} L)}$ where $J_{p}'(\mu_{n} L) = 0$, when $p \neq 0$.

When p = 0, expansion is 1 = 1. 51. $f(x) = \frac{1}{4}P_0(x) + \frac{1}{2}P_1(x) + \frac{5}{16}P_2(x) + \cdots$. 52. $f(\theta) = \frac{1}{4}P_0(\cos \theta) + \frac{1}{2}P_1(\cos \theta) + \frac{5}{16}P_2(\cos \theta) + \cdots$. 67. $A(u) = \frac{2}{\pi} \int_0^\infty f(x) \sin ux \, dx$.

Chapter 6

1. 3; $(\frac{2}{3}, -\frac{2}{3}, -\frac{1}{3})$. 2. c = (na + mb)/(m + n). 6. (a) $\pm (\frac{1}{3}, \frac{2}{3}, -\frac{2}{3})$. (b) $\pm (\frac{1}{3}i + \frac{2}{3}j - \frac{2}{3}k)$. 8. (a) $\frac{2}{3}$. (b) $\cos^{-1} \frac{2}{15}$. **9.** 홏. 12. 45°. 13. (a) $\pm (\sqrt{2}/10)(4i - 3j - 5k)$. (b) $5\sqrt{2}/2$. 14. (a) $\pm (2i - j - k)/\sqrt{6}$. (b) $\sqrt{6}/2$. 15. $\sqrt{\frac{2}{3}}\omega$. 16. -5. 20. ¹/₈. 21. $\pm (i - k)/\sqrt{2}$. 25. $\mathbf{F} \cdot \frac{d\mathbf{F}}{dt} \times \frac{d^3\mathbf{F}}{dt^3}$. 26. (a) $\mathbf{v} = -\omega \mathbf{a} \sin \omega t + \omega \mathbf{b} \cos \omega t$; $\mathbf{r} \times \mathbf{v} = \omega (\mathbf{a} \times \mathbf{b})$. (b) $\mathbf{a} = -\omega^2 \mathbf{r}$. 32. (c) $\mathbf{u} = (-a\mathbf{i}\sin t + a\mathbf{j}\cos t + c\mathbf{k})/\sqrt{a^2 + c^2}$, $\mathbf{n} = -(\mathbf{i}\cos t + \mathbf{j}\sin t),$ $\mathbf{b} = (c\mathbf{i}\sin t - c\mathbf{j}\cos t + a\mathbf{k})/\sqrt{a^2 + c^2}.$ **39.** (a) $\pm (1,1,1)/\sqrt{3}$; $2\sqrt{3}$. (b) $-\frac{3}{5}$. **40.** grad $r = \mathbf{i} \cos \theta + \mathbf{j} \sin \theta$; grad $\theta = -\mathbf{i} \frac{\sin \theta}{r} + \mathbf{j} \frac{\cos \theta}{r}$. **41.** $(l,m,n) = (\varphi_x,\varphi_y,\varphi_z)/\sqrt{\varphi_x^2 + \varphi_y^2 + \varphi_z^2}$ 42. $\pm (2i - j + 2k)/3$. **45.** (a) 2lx + 2my + 2nz. (b) $2lx \mathbf{i} + 2my \mathbf{j} + 2nz \mathbf{k}$. 52. (a) No solution. (b) $\varphi = x^2y + y^2z + z + C$. **53.** (a) $-\frac{4}{13}$. (b) 3. 54. (a) 0. (b) 0 **58.** ³/₄. **59.** π/2. **60.** 24. **61.** 0.

75. Both sides of Equation (129) reduce to π . **76.** $-\pi$. (b) $n r^{n-1} \mathbf{u}_r$. (d) 0. (e) $3\cos\theta$. **90.** (a) \mathbf{u}_{θ}/r . (c) u_z/r . (f) 0. **92.** (a) \mathbf{u}_{σ}/r . (b) $\mathbf{u}_{\theta}/(r \sin \varphi)$ (c) 0. (d) 0. **96.** 0. 97. $\frac{4}{3}\pi$. **99.** (b) $V = 3(x^2 - y^2)i - 6xy j$; $V = 3(x^2 + y^2)$. (c) $\psi = 3x^2y - y^3$; $3x^2y - y^3 = \text{constant}$. (d) 14ρ. 103. (b) V = $-2r \mathbf{u}_r + 4z \mathbf{k}$; $V = 2\sqrt{r^2 + 4z^2}$. (c) $\psi = 2r^2z + c$. Chapter 7 **1.** (a) $\cos \theta$. (b) $-r\sin\theta$. (c) $\sin \theta$. (d) $r \cos \theta$. (e) $\cos \theta$. (g) $-r^{-1}\sin\theta$. (h) $r^{-1}\cos\theta$. (f) sin θ . **2.** (a) $-\cot \theta$. (b) $-\tan \theta$. (c) $\tan \theta$. (d) $\cot \theta$. (f) $r^{-1} \cot \theta$. (g) $-r \cot \theta$. (h) $-r^{-1} \tan \theta$. (e) $r \tan \theta$. 10. (a) $\frac{\partial(f,g)}{\partial(u,v)} \neq 0, h_z \neq 0.$ (e) $z_x = 4x^3 - 4xy^3 - 2x$, $z_y = 6y^5 - 6x^2y^2 - 3y^2$. 16. Yes. 17. Yes. **23.** $1 + x + y + \frac{1}{2}x^2 + xy + \frac{1}{2}y^2 + \cdots$ **24.** $1 - x + y + x^2 - 2xy + y^2 + \cdots$. **26.** 0.906. 27. (a) Min at (0,0) if $b^2 < 1$, saddle point at (0,0) if $b^2 > 1$; min along line y = -x if b = 1; min along line y = x if b = -1. (b) Min at (1,0). (c) Rel. max at (0,0); saddle points at (1,1), (1,-1), (-1,1), (-1,-1). (d) Saddle point at (1,1) **28.** (a) Min at (0,0). (b) Min at (1, -1). (c) Min along lines $y = \pm x$. (d) Max at (1, -1); min along circle $(x - 1)^2 + (y + 1)^2 = 2$. **29.** (a) Min at (0,0). (b) Min along lines $y = \pm x$. (c) Saddle point at (0,0). (d) Min at (0,0). 30. Max = 3 at (-1,0); min = $-\frac{3}{2}$ at $(\frac{1}{2}, \pm \sqrt{\frac{3}{2}})$. **31.** Min at (1,1), (-1,-1). **32.** Max at (1,1), (-1,-1); min at (1,-1), (-1,1). 33. (a) Min = 12 when $x^2 = 4$, $y^2 = 1$, $z^2 = \frac{1}{4}$. (b) Min = 4 when $x^2 = 2$, $y^2 = \frac{1}{2}$, z = 0. (c) Min = 1 at (1,0,0). 34. Furthest at (1,0,0); nearest at $(\frac{1}{3},\frac{2}{3},0)$. **35.** (a) (py')' + qy = f. (b) (Ay')' + (B' - C)y = E - D'. **36.** (a) $(Au'_1)' + (Bu'_2)' = 0$, $(Bu'_1)' + (Cu'_2)' = 0$. (b) $(a_{11}u'_1)' + (a_{12}u'_2)' = b_{11}u_1 + b_{12}u_2$, $(a_{12}u_1')' + (a_{22}u_2')' = b_{12}u_1 + b_{22}u_2.$

37. (a) $z = c_1\theta + c_2$. (b) $\theta = c_3z + c_4$. (c) $z = c_5t + c_6$, $\theta = c_7t + c_8$. [Here a geodesic is either a helix, a circular arc, or a straight line. Part (a) ignores the straight lines θ = constant and part (b) ignores the circles z = constant, while part (c) includes *all* possibilities.]

- **38.** $\theta = \sin^{-1}(c_1 \cot \varphi) + c_2$.
- **39.** Value of integral always $\frac{1}{2}$.
- 41. π^2 , corresponding to $y = \sqrt{2} \sin n\pi x$ with n = 1.
- **46.** $\sqrt{\pi/2}$.
- **47.** y''' = h, y(a) = y'(a) = y''(a) = 0.
- **48.** 2,0945515.
- **49.** 4.493.
- **50.** 3.927.
- **51.** $x \doteq 1.01941, y \doteq 1.03062.$

Chapter 8

1. (a) $(x - y)(z_x - z_y) - 2z = 0$. (b) $bd z_{xx} - (ad + bc)z_{xy} + ac z_{yy} = 0.$ (c) $b^2 z_{xx} - 2ab z_{xy} + a^2 z_{yy} = 0.$ 5. (a) z = cy/b + f(bx - ay) ($b \neq 0$). (b) $z = e^{cx/a} f(bx - ay)$ $(a \neq 0)$. (c) $z = f(x^2 + y^2)$. (d) $z = e^{-x^2} f(y - x)$. (e) z = x f(xy). (f) $z^{-1} = x^{-1} + f(x^{-1} - y^{-1})$. 6. (b) $z = f(2x + y) + x^2 + x + \frac{1}{2}e^y$. 11. $z = (x + y)^2$. 12. (a) $z = \sqrt{x} + y f(x - y) + 1$. (b) $z = \frac{1}{2}x^2y + f(y)$. (c) $z^2 = xy + f(y|x)$. **13.** (a) $z = [(b - c)x + (c - a)y]/(b - a) \quad (b \neq a);$ z = x + f(x - y) where f(0) = 0 if a = b = c. (b) $z = e^{e(y-x)/(b-a)}(bx - ay)/(b - a)$ $(b \neq a)$. (c) $z = \sqrt{(x^2 + y^2)/2}$. (d) No solution. (e) z = x. (f) $z = x / \left[1 + x f \left(\frac{y - x}{xy} \right) \right]$ where f(0) = 0. 14. (a) $z = [(bx - ay)/(b - 2a)]^2 + c(y - 2x)/(b - 2a)$ $(b \neq 2a)$. (b) $z = e^{c(y-2x)/(b-2a)}[(bx - ay)/(b - 2a)]^2$. (c) $z = \frac{1}{6}(x^2 + y^2)$. (d) $z = (y - x)^2 e^{y(y-2x)}$. (e) $z = (\frac{1}{5}x^3y)^{1/2}$. (f) $z = (x^2y^2)/(4x^2 - 8xy + 4y^2 - xy^2 + 2x^2y)$. 19. (a) z = f(3x + y) + g(x - y). (b) $\varphi = f(x + y + ix) + g(x + y - ix)$. (c) $\varphi = f(x + y) + g(x)$. (d) $r^2w = f(r + ct) + g(r - ct)$ $(c \neq 0)$. (e) $\varphi = f[x - (V + U)t] + g[x + (V - U)t] \quad (V \neq 0).$ (f) $\varphi = f_1(x + y) + f_2(x - y) + f_3(x + iy) + f_4(x - iy).$ (g) $z = f_1(x + y) + x f_2(x + y) + f_3(x - y) + x f_4(x - y).$ (h) $\varphi = f_1(x + iy) + f_2(x - iy) + x f_3(x + iy) + x f_4(x - iy).$ 23. (a) $z = f(y + x) + g(y - x) + \frac{1}{6}x^3$. (b) $z = f(y + x) + g(y + 2x) - \frac{1}{2}\cos y$.

25. (a) $z = f(y - x) + \frac{1}{2}x^2y^2$. (b) $z = f(x + iy) + g(x - iy) + \frac{1}{12}x^4 + \frac{1}{6}x^3y$. (c) $z = f(y + x) + g(y - x) + \frac{1}{6}x^3(y + 1)$. (d) $z = f(y + x) + x g(y + x) + \frac{1}{12}x^4 + \frac{1}{6}y^3$. 27. $\varphi = x^2 + c^2t^2 + c^{-1}\cos x \sin ct$.

Chapter 9

11. (a)
$$T = \frac{400}{\pi} \sum_{n \text{ odd}} \frac{\sinh(n\pi y/100)}{n \sinh n\pi} \sin \frac{n\pi x}{100}$$
.

(b) Exact value is 25°.

18.
$$T = \alpha_1 + (\alpha_1 - \alpha_1) \frac{x}{L} + \beta_1 y + (\beta_2 - \beta_1) \frac{xy}{L} + \sum_{n=1}^{\infty} a_n e^{-n\pi y/\theta} \sin \frac{n\pi x}{L}$$

where $a_n = \frac{2}{L} \int_0^L f(x) \sin \frac{n\pi x}{L} dx - \frac{2}{n\pi} (\alpha_1 - \alpha_2 \cos n\pi).$
19. $T = \frac{2}{r} (r^2 - 100) \sin \theta + \frac{1}{2r} (400 - r^2) \cos \theta.$
20. (a) $T = \frac{1}{2} + \frac{2}{\pi} \sum_{n \text{ odd}} \frac{r^n}{n} \sin n\theta$ ($r \le 1$).
(b) $T = \frac{1}{2} + \frac{2}{\pi} \sum_{n \text{ odd}} \frac{r^n}{n} \sin n\theta$ ($r \ge 1$).
21. (a) $T = \sum_{n=1}^{\infty} A_n \left(\frac{r}{a}\right)^{n\pi/\alpha} \sin \frac{n\pi \theta}{\alpha}$, where $A_n = \frac{2}{\alpha} \int_0^{\infty} f(\theta) \sin \frac{n\pi \theta}{\alpha} d\theta.$
27. (a) $T = T_0 \left(1 - \frac{r}{a} \cos \varphi\right).$
(b) $T = T_0 \left[\frac{1}{2} + \frac{3}{4} \frac{r}{a} P_1(\cos \varphi) - \frac{7}{16} \left(\frac{r}{a}\right)^3 P_3(\cos \varphi) + \cdots\right].$
41. (b) $w = \sum_{m=1}^{\infty} \sum_{n=1}^{\infty} b_{mn} \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} \sin \omega_{mn} t,$
where $b_{mn} = \frac{4}{\omega_{mn}L^2} \int_0^L \int_0^L v(x,y) \sin \frac{m\pi x}{L} \sin \frac{n\pi y}{L} dx dy.$
43. $T = 100 - x - \frac{200}{\pi} \sum_{m=1}^{\infty} \frac{1}{m} \sin \frac{m\pi x}{50} e^{-m^2 \pi^2 \alpha^2 t/2500}.$
44. (a) $T_{T,p} = b_n \cos \frac{n\pi x}{L} e^{-n^2 \pi^3 \alpha^2 t/L^2} (n = 0, 1, 2, 3, ...).$
(b) $T = 50 - \frac{400}{\pi^2} \sum_{n \text{ odd}} \frac{1}{n^4} \cos \frac{n\pi x}{\pi 0} e^{-n^2 \pi^3 \alpha^2 t/10^4}.$
52. (a) $T = T_0 \left[\frac{x}{L} \sin \omega t + \frac{2\lambda \omega}{\pi} \sum_{n=1}^{\infty} \frac{(-1)^n \pi}{n^4 + \lambda^2 \omega^3} \left(\cos \omega t + \frac{\lambda \omega}{n^2} \sin \omega t - e^{-n^2 t/\lambda}\right) \sin \frac{\pi \pi x}{L}$

84.
$$\frac{V_s}{U} = 1 + \frac{1}{2} \left[\epsilon [x - (V_s + U)t] + \epsilon [x + (V_s - U)t] \right],$$

 $\frac{p}{p_0} = 1 + \frac{U}{V_s} \left[\epsilon [x - (V_s + U)t] - \epsilon [x + (V_s - U)t] \right].$
Chapter 10
5. (a) $-2 + 2i$. (b) $0 + i$. (c) $0 + i$. (d) $\frac{e^s}{\sqrt{2}} + \frac{e^s}{\sqrt{2}}i$.
(e) $\frac{\cosh 2}{\sqrt{2}} + \frac{\sinh 2}{\sqrt{2}}i$. (f) $\frac{\cosh 2}{\sqrt{2}} + \frac{\sinh 2}{\sqrt{2}}i$.
6. $\sin z$, $\cos z$, $\sec z$, and $\csc z$ have period 2π ; $\tan z$ and $\cot z$ have period π ; $\sinh z$,
 $\cosh z$, $\operatorname{sech} z$, and $\operatorname{csc} z$ have period 2π ; $\tan z$ and $\cot z$ have period π ; $\sinh z$,
 $\cosh z$, $\operatorname{sech} z$, and $\operatorname{csc} z$ have period 2π ; $\tan z$ and $\cot z$ have period π ;
(a) $\log \sqrt{2} + i \left(\frac{\pi}{4} + 2k\pi\right)$, where k is any integer;
principal value is $\log \sqrt{2} + \frac{\pi i}{4}$.
(b) $\cos \frac{3\pi}{8} (1 + 4k) + i \sin \frac{3\pi}{8} (1 + 4k)$, where $k = 0, 1, 2, 3$;
principal value is $\cos \frac{3\pi}{8} + i \sin \frac{\pi}{8}$.
(c) $\sqrt[6]{2} \left[\cos \frac{\pi}{8} (1 + 8k) + i \sin \frac{\pi}{8} (1 + 8k) \right]$, where $k = 0, 1$;
principal value is $\sqrt[6]{2} \left[\cos \frac{\pi}{8} + i \sin \frac{\pi}{8} \right]$.
12. $z = \sqrt[6]{2} (\cos \alpha + i \sin \alpha)$, where $\alpha = \frac{3\pi}{8}, \frac{5\pi}{8}, \frac{11\pi}{8}, \frac{13\pi}{8}$.
13. $z^{\pi} = r^{\pi} [\cos \pi(\theta_P + 2k\pi) + i \sin \pi(\theta_P + 2k\pi)]$, where $z = re^{i\theta}$; $\cos \frac{\pi^a}{4} + i \sin \frac{\pi^4}{4}$.
15. (a) $\left(\frac{\pi}{2} + 2k\pi\right) \pm i \log (2 + \sqrt{3})$, where k is any integer.
(b) $(2k + 1)\frac{\pi}{2} + i \log \sqrt{3}$, where k is any integer.
19. (a) $3xy^2 - x^3 + C$. (c) It is not an analytic function.
24. (a) 0. (b) $2\pi i$.
36. (a) $1 + z + z^2 + \cdots + z^n + \cdots$.
(b) $-\frac{1}{z} - \frac{1}{z^2} - \frac{1}{z^3} - \cdots - \frac{1}{z^n} - \cdots$.
(c) $\frac{1}{2} + \frac{z+1}{2^2} + (\frac{z+1}{2^2})^2 + \cdots + (\frac{z+1}{2^{n+1}})^n + \cdots$.
(d) $-\frac{1}{z-1} - \frac{2}{(z+1)^3} - \frac{2^2}{(z+1)^3} - \cdots - \frac{2^{n-1}}{(z+1)^n} - \cdots$.
(e) $-\frac{1}{z-1}$.

37. (a)
$$\frac{1}{z} + 1 + z + z^{2} + \dots + z^{n} + \dots$$
.
(b) $-\frac{1}{z^{4}} - \frac{1}{z^{3}} - \frac{1}{z^{4}} - \dots - \frac{1}{z^{n}} - \dots$.
(c) $-\frac{1}{z-1} + 1 - (z-1) + (z-1)^{3} - \dots + (-1)^{n}(z-1)^{n} + \dots$.
(d) $-\frac{1}{(z-1)^{3}} + \frac{1}{(z-1)^{3}} - \frac{1}{(z-1)^{4}} + \dots + (-1)^{n+3} \frac{1}{(z-1)^{n}} + \dots$.
(e) $-\frac{1}{z} - \frac{1}{z}(z+1) - \frac{1}{z}(z+1)^{2} - \dots - (1 - 2^{-n-2})(z+1)^{n} - \dots$.
(f) $\dots + \frac{1}{(z+1)^{n}} + \dots + \frac{1}{(z+1)^{2}} + \frac{1}{z+1} + \frac{1}{2} + \frac{z+1}{2^{2}} + \frac{(z+1)^{n}}{2^{n+1}} + \dots$.
(g) $-\frac{1}{(z+1)^{2}} - \frac{3}{(z+1)^{3}} - \frac{7}{(z+1)^{4}} - \dots - \frac{2^{n-1}-1}{(z+1)^{n}} - \dots$.
38. $\frac{1}{z} + \frac{1}{6}z + \frac{7}{360}z^{3} + \dots$.
42. (a) Simple poles at $z = \pm i$.
(b) Simple poles at $z = 1, z = \frac{1}{2}(1 \pm i\sqrt{3})$.
(c) Branch points at $z = 1, z = 2$.
(e) Simple poles at $z = 1, z = 2$.
(f) Branch points at $z = 1, z = 2$.
(g) Simple poles at $z = 1, z = 2$.
(h) Simple poles at $z = 1, z = 2$.
(i) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles at $z = 1, z = 2$.
(j) Simple poles (j) Simple pole. (j) Simple pole.
(j) Res (0) = 1.
(j) Res (0) = 1.
(j) Res (0) = 0.
(j) Res (0) = 1.
(j) Res (0) = 0.
(j) Res (j) = 1, Res ((-1)

Index

(Italicized figures in parentheses refer to problem numbers.)

ABEL'S formula, 29, 31, 42 (8), 47 (30, 36) Absolute value, 509 Acceleration, 273 normal, 274 in polar coordinates, 319 (31) tangential, 274 Acoustics, 427 Adams' method, 96 modified, 100 Admissible function, 355 Algebra, fundamental theorem of, 593 (53) Alternating series, 120 Amplitude, of complex number, 511 Analytic continuation, 531 Analytic function, 521, 535 residues of, 547, 594 (60), 595 (62) singularities of, 535 Angle, between vectors, 265 Angular velocity vector, 267, 294 Approximate solution of differential equations: by graphical methods, 93 by numerical methods, 93 Arc length, differential of, 273, 300 on surface, 371 (22) Area, differential of, 300, 347, 370 (21) Argument, of complex number, 511 Ascending differences, 112

Associated Legendre functions, 165 Associative law, 263 Asymptotic approximation: for Bessel functions, 151 for Gamma function, 81 Asymptotic expansion, 170 of Bessel functions, 170 Average value, 217 Axial load, 198 **BACKWARD** differences, 108 Base curve, characteristic, 395, 404 Beam: bending of, 193 buckling of, 198 rotating, 193 vibrating, 244 (10) Bending moment, 194, 320 (38) Bending stiffness, 320 (38) ber and bei functions, 156 asymptotic approximations for, 181 (37) derivative formulas for, 181 (38) Bernoulli's differential equation, 37, 43 (14)Bernoulli's equation, 314, 609 (101) **Bessel functions:** asymptotic approximations for, 151 asymptotic expansions of, 170 derivative formulas for, 151
Bessel functions (Cont.): differential equations satisfied by, 155 expansions in, 226, 448 of first kind, 143 integral formulas for, 178 (26, 27), 254 (45) integrals of, 177 (24) modified, 150 of order $n + \frac{1}{2}$, 77, 153 of second kind, 147 of third kind, 149 Wronskian of, 179 (29) zeros of, 154, 617, 619 Beta function, 91 (43) Bi-Laplacian equation, 326 (73), 427 Binormal vector, 274 Boundary-value problem: for ordinary differential equation, 187 nonhomogeneous, 213 for partial differential equations, 469 Branch, 516, 535 Branch point, 132, 516, 535, 540 Bromwich integral, 602 (87) Buckling, of long column, 198, 246 (15) Buckling mode, 199 CALCULUS of variations, 355 Capacity, of condenser, 615 (122) Cauchy principal value, 558 Cauchy-Riemann equations, 522, 587 (20)Cauchy's inequality, 590 (33) Cauchy's integral formula, 528 Cauchy's integral theorem, 524 Cauchy's residue theorem, 551 Central differences, 112 Central force, 317 (27) Chain rules, 335 Characteristic base curve, 395, 404 Characteristic curve, 380, 394, 400, 404, 410 Characteristic cylinder, 395, 404 Characteristic equation, 8 Characteristic functions, 188 expansions in series of, 210 orthogonality of, 206 Characteristics, 404, 411 Characteristic strip, 404 Characteristic values, 188 Charge density 584, 615, (121)

Chebyshev polynomials, 142

Circle of convergence, 512, 530 Circular cylindrical coordinates, 298, 303 Circular functions, 512 Circulation, 284, 490 (37), 570 Column, buckling of, 198, 246 (15) Commutative law, 17, 263, 265, 266 Compatibility, equation of, 404 Complementary solution, 7, 388 Completeness, 212 Complete solution, 4 Complex number, 509 Complex plane, 509 extended, 542 Complex potential, 566 Complex variable, function of, 509 Complex velocity, 567 Components, of vector, 263 Compressible fluid, flow of, 309, 409, 475 Condenser, capacity of, 615 (122) Conductor, 584, 615 (121) Confluent hypergeometric function, 141, 169, 174 (8) Conformal mapping, 562 Conjugate complex numbers, 509 Conjugate flows, 307 Conjugate functions, 523 Conservative force, 285 Constraints, 352, 359 Continuity, equation of, 310 Contour integration, 551, 558 Convergence, of series, 119, 511, 530 Convolution, 62 Coordinates: circular cylindrical, 298, 303 curvilinear, 298, 346 nonorthogonal, 346 orthogonal, 298 elliptical cylindrical, 327 (82) parabolic cylindrical, 328 (84) spherical, 304 Cosine-integral function, 32 Cramer's rule, 20 Critical length, of column, 246 (16) Critical load, for column, 199 Critical speed: of rotating shaft, 197 of rotating string, 191 Crosscut, 525 Cross product, 266 Curl, 277, 294 in curvilinear coordinates, 303

Curl theorem, 325(67)Curvature, 195, 273 of beam, 195 radius of, 273 Curvilinear coordinates, 298, 346, 480 (7) arc length, 300 curl, 303 del, 301, 608 (100) divergence, 302 gradient, 301 Laplacian operator, 302 nonorthogonal, 346 orthogonal, 298, 480 (7) surface area element, 300, 370 (Ž1) volume element, 300, 346 Cut, 536 D'ALEMBERT'S solution, 392, 454 Damping, 88 (35) Dashpot, 88 (35) Definite integrals, evaluation of: by calculus of residues, 551, 558 by Laplace transforms, 83 (12) Deflection: of beam, 193 of membrane, 446 of string, 189 Del, 277 in curvilinear coordinates, 301, 608 (100)Delta function, 64 De Moivre's theorem, 513 Dependence: functional, 344 linear, 3 Derivative: of complex function, 520 of vector function, 270, 278 Descending differences, 112 Differences: ascending, 112 backward, 108 central, 112 descending, 112 forward, 98, 108 Differential distance vector, 281 Differential equation: Bernoulli, 37, 43 (14) Bessel (see Bessel functions) Cauchy-Riemann, 522, 587 (20) complementary solution of, 7, 388

Differential equation (Cont.): complete solution of, 4 confluent hypergeometric, 144, 169, 174 (8) equidimensional, 13 exact, 34 Hermit, 176 (18) homogeneous, 35, 37 homogeneous solution of, 7 hypergeometric, 165 Jacobi, 184 (45) Laguerre, 176 (17) Legendre (see Legendre functions) linear, 1, 7, 8, 388, 392 nonlinear, 2, 33 numerical solution of, 93 order of, 2 ordinary, 2 partial, 2, 376 classification of, 388 normal form of, 406 quasi-linear, 376 Riccati, 50 (48) separable, 33, 36 series solutions of, 119 Differential surface area vector, 288, 300, 370 (21) Differentiation: of integral containing parameter, 359 numerical, 111 of vector, 270, 278 Diffusion, 427 Diffusivity, 429 Digamma function, 90 (42) Dirac delta function, 64 Directional derivative, 276 Direction cosines, 264 Direction ratios, 264 Dirichlet problem, 430 Discontinuity, finite, 54 Distributions, theory of, 65 Distributive law, 17, 265, 266, 279 Divergence, 277 in curvilinear coordinates, 302 theorem, 290, 324 (65) Division, of series, 122 Dot product, 265 Double Fourier series, 442 Doublet, 65, 570 orientation of, 570 strength of, 570

Doublet function, 65 Drag force, 609(101)Duhamel's integral (see Superposition integral) Duplication formula, for Gamma function, 92 (48) **EIGENFUNCTION** (see Characteristic function) Eigenvalue (see Characteristic value) Elastic end support, 195 Electric field intensity, 615 (121) Electromagnetic theory, 427 Electrostatic potential, 426, 584, 615 (121)Elementary functions, 511 Elliptical cylindrical coordinates, 327 (82) Elliptic integrals, 192 Elliptic partial differential equation, 388, 409, 469 normal form of, 406 Entire function, 547 Envelope, of curves, 34 Equations of motion, for fluids, 310 Equidimensional differential equation, 13 Equipotential line, 306 Equipotential surface, 286 Equivalent contours, 527 Error function, 464 Essential singularity, 540 nonisolated, 541 Euler equation, 357 Euler load, of column, 199 Euler's constant, 90 (42), 146 Euler's differential equation (see Equidimensional differential equation) Euler's equations, of fluid motion, 310 Euler's formula, 10, 513 Even function, 218 Exact differential equation, 34 Expansion: asymptotic, 170 in series of characteristic functions, 210 Exponent, of differential equation, 131 Exponential function, 512 generalized, 518 Exponential-integral function, 32 Exponential order, function of, 54 Extrapolation, 108 Extremal, 358

FACTORIAL function (see Gamma function) Factorization, of operators, 17 Faltung (see Convolution) Family, 11 Field intensity vector, 615 (121) Fixed end support, 195 Fluid flow, 305, 566 from channel, 581 compressible, 309, 409, 475 about cylinder, 489 (36-38), 571, 609 (102), 611 (108) about plate, 579, 613 (112) about sphere, 443 supersonic, 312, 475 Fluid pressure, 309, 609 (101) Flux, 292, 480 (7) Force, 281 conservative, 285 Forced vibration, 68 Force potential, 285 Formulation of problems, 469 Forward differences, 98 Fourier-Bessel integral, 242 Fourier-Bessel series, 226 Fourier-Bessel transforms, 242 Fourier cosine integral, 238 Fourier cosine series, 220 Fourier integral, 236 applications of, 460 complex form of, 239 Fourier series, 216, 221 complex form, 253 (43) double, 442 Fourier sine integral, 238 Fourier sine series, 217 double, 442 Fourier transforms, 240 Free end condition, 190, 195 Free vibration: of beam, 244 (10) of membrane, 446 of string, 244 (7) Frenet formulas, 274 Frequency, natural, 69, 244 (7, 10) Fresnel integrals, 598 (77) Frobenius, method of, 129 exceptional cases, 135 Functional dependence, 344 Functions: analytic, 521

Functions (Cont.): Bessel (see Bessel functions) Beta, 91 (43) circular, 512 confluent hypergeometric, 141, 169, 174 (8) conjugate, 523 cosine-integral, 32 Digamma, 90 (42) doublet, 65 elementary, 511 error, 464 even, 218 exponential, 512, 518 exponential-integral, 32 Gamma, 73 homogeneous, 35 hyperbolic, 43 (15), 513 Laguerre, 82 (6), 142, 176 (17) Legendre, 161 logarithmic, 514 multivalued, 536 odd, 218 potential, 284 sine-integral, 32 single-valued, 283 singularity, 63 square-wave, 83 (9) stream, 307 triangular-wave, 83 (10) unit impulse, 64 Fundamental set, of solutions, 48 (39) Fundamental theorem, of algebra, 593 (53) **GAMMA** function, 73 asymptotic approximation for, 81 duplication formula for, 92 (48) table of, 619 Gauss's hypergeometric function, 165 Gauss's interpolation formula, 117 (22) Gauss's test, for infinite series, 120 Generalized factorial function (see Gamma function) Generating function: for $J_n(x)$, 179 (28) for $P_n(x)$, 182 (42) Geodesics, 373 (37) Geometry, of space curve, 271 Gradient, 275 in curvilinear coordinates, 301

Gradient theorem, 325 (67) Graphical integration, of differential equations, 93 Gravitational potential, 426 Green's theorem: first form, 293 second form, 293 in the plane, 297 Gregory's backward difference formula, 110 Grounded line, 465 HANKEL functions, 149, 457 Harmonic functions, 295 Harmonics, 217 Heat flow, 428 in column, 488 (33-35) in cylinder, 487 (30-32) in plate, 430, 433, 460 in rectangular parallelepiped, 441 in rod, 449, 462 with radiation, 479(6)in sphere, 439 Heat-flow equation, 427 Heaviside unit step function, 87 (27) Helix, 319 (32) Helmholtz equation, 394, 420 (31-34), 496 (57) Hermite polynomials, 142, 176 (18) Hinged end support, 195 Homogeneous boundary condition, 187, 431 Homogeneous differential equation: of first order, 35, 37 linear, 5, 389 partial, 389 Homogeneous function, 35 Homogeneous solution, 7 Hyperbolic functions, 43 (15), 513 Hyperbolic partial differential equation, 388, 409, 469 normal form of, 406 Hypergeometric function, 165 confluent, 141, 169, 174 (8) Hypergeometric series, 165 **IMAGINARY** number, 509 Imaginary part, of complex quantity, 438, 509 Imaginary unit, 509 Implicit functions, 340

Improper integral, principal value of, 558 Impulse functions, 64 Incompressible fluid, 291 Indented contours, 558 Independence: functional, 344 linear. 3 Indicial equation, 131 Infinite series (see Series) Infinity: point at, 542 residue at, 594 (60), 595 (62) Initial conditions, 32, 384 Initial-value problem, 32, 386, 394, 400, 409, 469 Inner product, 265 Insulated boundary, 483 (16), 492 (44) Integral: of differential equation, 379 elliptic, 192 improper, 558 principal value of, 558 line, 281, 523 surface, 287 Integral equation, 86 (24) Integral function, 547 Integral power function, 511 Integral surface, 380 Integral transcendental function, 547 Integrating factor, 7, 37 Integration, numerical, 111, 116 (18), 117 (23, 25), 118 (27) Intermediate variable, 336 Interpolation, 110, 117 (22) Interval of convergence, 120 Inverse circular functions, 519 Inverse hyperbolic functions, 519 Inverse Laplace transform, 60, 601 (87-94) Inverse tangent, addition formula for. 578 Inversion, of plane, 606 (95) Irregular singular point, of differential equation, 128, 168 Irrotational flow, 294 Isoclines, 93 Isolated singular point, 538 Iteration, methods of: Newton-Raphson method, 362 Picard's method, 106 in solving transcendental equations, 193

Iteration, methods of (Cont.): Stodola-Vianello method, 200, 214 JACOBIAN determinants, 340, 346 Jacobi polynomials, 142, 184 (46) Jacobi's differential equation, 184 (45) Jordan's lemma, 556 ker and kei functions, 158 asymptotic approximations for, 181 (37) derivative formulas for, 181 (38) Kutta (see Runge-Kutta method) LAGRANGE, identity of, 270 Lagrange multipliers, 352, 359 Laguerre polynomials, 82 (6), 142, 176 (17)associated, 142 Laplace's equation, 295, 426, 429, 523 polynomial solutions of, 484 (17) Laplace transforms, 51, 464 inversion of, by contour integration, 601 (87–94) table of, 74 Laplacian operator, 280 in curvilinear coordinates, 302, 608 (100)Laurent series, 531 Least squares, method of, 252 (39), 256 (53) Legendre functions, 159 associated, 165 recurrence formula for, 162 of second kind, 162 Legendre polynomials, 161 Legendre series, 231. 440 Leibnitz's rule, 360 Lerch's theorem, 61 Lift force, 609 (101), 611 (108) Limiting contours, 555 Linear combination, 3 nontrivial. 3 Linear dependence, 3 Linear differential equations (see Differential equations) Linear fractional transformation, 606 (97) Line integral, 281, 523 of a complex function, 523 independent of path, 284, 524 Lines of flow, 306

Lines of force, 584, 615 (121) Liouville's theorem, 545, 593 (56) Logarithm, complex, 514 Long line, equations for, 464 Loop integral, 603 (90)MACH number, 312, 475 Mapping, conformal, 562 Maxima and minima: absolute, 352 of functions, 351 of integrals, 355 relative, 351 Maxwell's equations, 326 (77) Mean value (see Average value) Membrane: deflection of, 446 vibration of, 446 Minimax, 351 Modified Bessel functions, 150 Modulus, of complex number, 509 Moment: bending, 194, 320 (38) of momentum vector, 317 (27) vector, 267 Momentum vector, 317 (27) Multiple products, of vectors, 268 Multiplication, of series, 122 Multivalued function, 536 NATURAL boundary, 541 Natural frequencies: of vibrating beam, 244 (10) of vibrating membrane, 449, 491 (42) of vibrating string, 244 (7) Natural mode, of vibration, 244 (7, 10) Neumann problem, 430, 443 Neumann's function, 146 Newton-Gauss formula, 117 (22) Newton-Raphson method, 362 Newton's backward difference formula, 110 Newton's law, of cooling, 474, 492 (45, **46**) Newton's method, 362 Nodal lines, of membrane, 491 (42) Normal acceleration, 274 Normal derivative, 293 Normal form, of partial differential equation, 406 Normalization, of characteristic functions, 209

Normalizing factor, 210 Normal plane, 275 Normal vector, 287 Numerical methods: of differentiation, 111 of integration, 111, 116 (18), 117 (23, 25), 118 (27) of interpolation, 110, 116 (21, 22) of solving differential equations, 93 characteristic-value problems, 200 of solving nonlinear equations, 193, 362 ODD function, 218 Odd-harmonic function, 226 Open line, 465 Operational calculus (see Laplace transforms) Operator, linear dfferential, 5, 16, 339 Order: of differential equation, 2 of pole, 538 Ordinary differential equation, 2 Ordinary point, of differential equation, 127, 168 Orthogonal coordinates, 298 Orthogonal functions, 206 expansions in series of, 210 Orthogonal set, of functions, 206 Orthonormal set, of functions, 209 Osculating plane, 275 Outer product, 266 PARABOLIC cylindrical coordinates, 328 (84) Parabolic partial differential equation, 388, 409, 470 normal form of, 406 Parabolic rule, 118 (27) Paraboloidal coordinates, 329 (85) Parseval equalities, 252 (42) Partial derivative, 335 Partial differential equation, 376, 426 Partial fractions, 67, 84 (15) Particular solution, 6, 388 Picard's method, 106 Picard's theorem, 541 Piecewise continuity, 54 Piecewise smooth curve, 281 Piecewise smooth surface, 288 Plane, equation of, 315 (10) Plasticity, 412

Poisson's equation, 426 Poisson's integral, 437 Pole, 132, 538 order of, 538 residue at, 548 simple, 539 Polygonal mapping (see Schwarz-Christoffel transformation) Polynomial, 511, 546 existence of zero of, 593 (53) **Polynomials:** Chebyshev, 142 Hermite, 142, 176 (18) Jacobi, 142, 184 (45) Laguerre, 82 (6), 142, 176 (17) Legendre, 161 Position vector, 272 derivative of, 272 Potential, 284 complex, 566 electrostatic, 426, 584, 615 (121) existence of, 285 gravitational, 426 velocity, 305, 566 velocity deviation, 311 Potential energy, 285 Power function: generalized, 516 integral, 511 Power series, 119, 511, 528 convergence of, 119, 511, 529 operations with, 122, 512 solutions of differential equations. 119 Pressure, fluid, 309 Principal normal vector, 273 Principal value: of improper integral, 558 of logarithm, 515 of power function, 518 Proper strip, 403 Pulsating cylinder, 458 **Pulsating sphere:** steady-state solution, 498 (62) transient solution, 505(80)Pure imaginary number, 509 QUASI-LINEAR equations, 357, 376, 388 characteristics of, 380 classification of, 388

RAABE'S TEST, for infinite series, 120 Radiation, from rod, 479 (6) Radius: of convergence, 120, 512, 530 of curvature, 273 of torsion, 274 Rational function, 123, 511, 538, 543, 546 Ratio test, 119, 511 Real number, 509 Real part, of complex quantity, 438, 509 Rectifying plane, 275 Recurrence formula: for Bessel functions, 152 in Frobenius method, 131 for Gamma function, 79 for Legendre functions, 162 Reducible curve, 287 Reduction of order, of differential equation, 29 Reflection: of plane, 606 (96) of waves, 469 **Regular function**, 122 Regular singular point, of differential equation, 128, 168 Removable singular point, 538 Residue, 547 at infinity, 594 (60), 595 (62) at pole, 548 Resistive force, 88 (35), 497 (60) Resonance, 70, 499 (64) Riccati's equation, 50 (48) Riemann surface, 537 Right-handed coordinate system, 299 Rod, deformation of, 320 (38) Rodrigues' formula, 164, 182 (41), 232 Rotating shaft, 193 Rotating string, 189 Rotation: of axes, 318 (29), 329 (86), 605 (95) of fluid, 294 Rouché's theorem, 593 (55) Runge-Kutta method, 102 SADDLE point, 351

Scalar, 262 Scalar product, 265 Scale factor, 299 Schwarz-Christoffel transformation, 574

Semiconvergent series, 171 Separable differential equation, 33, 36, 433 Separation constant, 433 Separation of variables, 1, 428, 430 Series: asymptotic, 170 Fourier (see Fourier series) Fourier-Bessel, 226 Frobenius, 129 geometric, 166 hypergeometric, 165 power, 119 semiconvergent, 171 Taylor, 94, 122, 348 Shaft, rotating, 193 Shearing force, 194, 320 (38) Shear lines, 412 Simple pole, 548 Simple region, 284 Simply connected region, 284 Simpson's Rule, 117 (25), 118 (26, 27) Simultaneous differential equations, 19 Sine-integral function, 32 Single-valued function, 283, 521 Singular curve, on integral surface, 406 Singularities, of analytic functions: branch point, 132, 516, 535, 540 essential, 540, 541 at infinity, 542 isolated, 538 pole, 132, 538 removable, 538 significance of, 545 Singularity functions, 63 Singular point, of differential equation, 127, 168 Singular solution, 5, 34 Sink, 291 Sliding clamped end support, 195 Smooth curve, 281 Smooth surface, 288 Solution, of differential equation, 1 complementary, 7, 388 complete, 4 homogeneous, 7 particular, 6, 388 singular, 5, 34 Sonic velocity, 311

Source, 291, 569 strength of, 569 Source function, in heat flow, 501 (71), 503 (75) Space curve, 271 Specific heat, 428 Speed, 273 Spherical coordinates, 304 Spherical waves, 455 Spring constant, 68 Square-wave function, 83(9)Stability, 199 Stagnation point, 490(38), 572Standard form, of linear ordinary differential equation: first order, 4 second order, 127 Steady state, 429 Stirling's formula, 81, 90 (40) Stodola-Vianello method, 200, 214 Stokes's theorem, 295 Straight line, equations of, 315 (5) Stream function, 307, 480 (7), 566 Streaming, 572 Streamline, 306, 480 (7), 566 Stress function, 427 Stretching, of plane, 605 (95) String: deflection of, 189 rotating, 189 vibrating, 244(7)Strip, 401 characteristic, 404 proper, 403 Strip condition, 401 Sturm-Liouville problem, 208, 373 (40) proper, 209 Substantial derivative, 310 Superposition integral, 451 Supersonic fluid flow, 312, 475 Surface: conical, 384 cylindrical, 383 equation of, 287 equipotential, 286 normal to, 287 Riemann, 537 Surface area, 288, 300, 370 (21) Surface charge, 584, 615 (121) Surface integral, 287

TABLE: of Bessel functions of order $n + \frac{1}{2}$, 77 of Gamma function, 619 of Laplace transforms, 74 of zeros of Bessel functions, 619 Tangential acceleration, 274 Tangent vector, 272 Taylor series, 94, 122, 348, 528 with remainder, 123, 349 uniqueness of, 590(34)Telegraph equations, 427, 464 Temperature, steady state: inside circle, 436 outside circle, 437 in circular annulus, 433 in circular cylinder. 487 (30-32), 488(34)in half-plane, 460 in rectangle, 430 in rectangular parallelepiped, 441, 488 (33)in rod, 450 in sphere, 439 Tension, 189, 320 (38), 446 Thermal conductivity, 428 Thermal diffusivity, 429 Torsion, 274 radius of, 274 Torsional stiffness, 320 (38) Total differential, 335 Transform: Fourier, 240 Fourier-Bessel, 242 Laplace, 51 Transient solution, heat flow, 450 Translation, of axes, 329 (86), 605 (95) Translation properties, of Laplace transforms, 59 Traveling waves, 454 Triangular-wave function, 83 (10) Trigonometric functions, 512 Triple scalar product, 268 Twisting moment. 320 (38)

UNDETERMINED coefficients, method of, 11, 19 Unit doublet function, 65 Unit impulse function, 64 Unit step function, 87 (27) Unit vector, 263 VARIATION, 356 of parameters, 25 Vectors, 262 algebra of, 262 differentiation of, 270, 278 products of, 265 Velocity, 273 Velocity deviation potential, 311 Velocity potential, 305, 566 Vianello (see Stodola-Vianello method) Vibrating beam, 244 (10) Vibrating membrane, 446 Vibrating string, 244 (7) Vibration, 68 natural modes of, 70 Viscosity, coefficient of, 334 (105) Viscous fluid, 334 (105) Volume, element of. 300, 346 Vortex, 305, 570 strength of, 570 WAVE equation, 327 (78), 391, 427, 446, 464 Waves, 454 standing, 496 (58) Weber's function, 146 Weighting function, 206 Work, 281 Wronskian determinant, 4, 27, 29, 31, 41 (7), 179 (29), 370 (20) Young's modulus, 194 ZFROS, of Bessel functions, 154, 617

table of, 619 7ero vector, 263